

User's Guide

NCC

Nebula Control Center

Default Login Details

NCC URL	https://nebula.zyxel.com
User Name	myZyxel account name
Password	myZyxel account password

Version 17.00 Edition 1, 04/2023



Copyright $\ensuremath{\textcircled{\sc c}}$ 2023 Zyxel and/or its affiliates. All Rights Reserved.

IMPORTANT!

READ CAREFULLY BEFORE USE.

KEEP THIS GUIDE FOR FUTURE REFERENCE.

This is a User's Guide for a system managing a series of products. Not all products support all features. Screenshots and graphics in this book may differ slightly from what you see due to differences in release versions or your computer operating system. Every effort has been made to ensure that the information in this manual is accurate.

Related Documentation

• Nebula Device Quick Start Guide

The Quick Start Guide shows how to connect the managed device, such as the Nebula AP, Switch or Security Appliance.

• Nebula Device User's Guide

Refer to the individual Nebula managed device's User's Guide for information about how to set the device to be managed by the NCC and/or configure the device using its built-in Web Configurator,

More Information

Go to the Nebula Control Center to find other information on the NCC.



Table of Contents

Table of Contents	3
Part I: Introduction & Getting Started Tutorials	12
Chapter 1 Introduction	13
	10
1.1.1 MCC Overview	13
1.1.2 Sites Organizations and Croups	14 14
1.1.2 Siles, Olganizations, and Gloups	14
1.1.4 License Concept	15
1.2 Getting Started	13
1.2.1 Connect Nebula Managed Devices	24
1.2.2 Access the NCC Portal	24
1.3 NCC Portal Overview	
1.3.1 Title Bar	31
1.3.2 Navigation Panel	36
1.4 Create Organization	43
1.5 Choose Organization	44
1.6 Cloud-Saving Mode	45
Chapter 2	
Setup Wizard	46
2.1 Setup Wizard	46
2.1.1 Step1: Run the Wizard	46
2.1.2 Step 2: Create an Organization and Site	47
2.1.3 Step 3: Add Your Nebula Devices	48
2.1.4 Step 4: Upgrade your Nebula Device Firmware	49
2.1.5 Step 5: Set up your WiFi Network	49
2.1.6 Step 6: Set up a Guest WiFi Network	50
2.1.7 Step 7: Set up the Deployment Method	51
2.1.8 Step 8: View the Summary	54
2.1.9 Step 9: Activate the Trial License(s)	55
Chapter 3	
Tutorials	57
3.1 Overview	57
3.2 Add a Nebula Device	58

3.3 Activate and Assign a License for a Nebula Device, Site, or Organization	58
3.3.1 Bundled License and Add-on License	59
3.3.2 License States	59
3.3.3 License Activation Process	59
3.4 Monitor a Site	61
3.5 Know What Licenses are Set to Expire in My Site or Organization	62
3.6 Renew an Expired License	63
3.7 Transfer Licenses	64
3.7.1 Select Transferable Licenses	64
3.7.2 Undo Assigning a License	65
3.7.3 Transfer a License to a Different Organization	65
3.7.4 Assign a License to a Nebula Device in the New Organization	66
3.7.5 Transfer a License to a Nebula Device in a New Organization	68
3.8 Reset the Nebula Password	69
3.9 Change an Organization and/or Site Name	73
3.10 Maintain Firmware	74
3.11 Assign an Administrator to Manage a Nebula Device	76
3.12 Transfer the Ownership of the Organization	78
3.13 Manage a Configuration Template	79
3.13.1 Create and Bind a Template Site/Setting	79
3.13.2 Duplicate and Import a Template Setting to a Site	83
3.13.3 Enable the Override Site-wide Configuration (Local Override) Feature	88
3.14 Activate an MSP License	93
3.15 Configure CNP/CNP Plus Security Services	93
3.15.1 Threat Protection	94
3.15.2 Application Visibility & Optimization	95
3.16 Delete an Organization	97
3.16.1 Remove All Nebula Devices	97
3.16.2 Transfer All Licenses	98
3.16.3 Delete All Sites	98
3.16.4 Delete All Administrators	99
3.16.5 Remove All Users	. 100
3.16.6 Delete the Organization	. 100
3.17 Remote Access VPN Setup	. 101
3.17.1 Create a VPN User	. 102
3.17.2 Enable the Remote Access VPN Rule for IPSec VPN Client	. 103
3.17.3 VPN Setup by the VPN Client	. 105
3.17.4 Import the VPN Configuration File	. 105
3.17.5 Open the VPN Tunnel	. 108
3.17.6 Set Up Two Factor Authentication to Bind the User Account	. 110
3.17.7 Check the Connection in Nebula by the Administrator	. 112
3.18 Route L2TP VPN Traffic	. 113
3.19 Resolve WiFi Connection Problems (for Nebula APs only)	. 115

3.20 Configure Voice VLAN (for Nebula Switches only)	. 121
3.20.1 Configure the Nebula Device Ports	. 122
3.20.2 Configure the Voice VLAN	. 122
3.21 Manage IPTV (for Nebula Switches only)	. 123
3.21.1 Set up the VLAN for IPTV	. 124
3.21.2 Define the Role of a Switch	. 125
3.21.3 Configure the Channel Profile and Naming	. 127
3.22 Enable IP Source Guard (for Nebula Switches only)	. 130
3.23 Set Up MAC Authentication With NCAS (for Nebula Switches only)	. 135
3.24 Set Up Dynamic VLAN With RADIUS (for Nebula Switches only)	. 136
3.25 Monitor Dynamic VLAN Using Event Logs (for Nebula Switches only)	. 138
3.26 Register a Nebula Device (mobile router) in Nebula	. 139
3.27 Using Collaborative Detection and Response (CDR)	. 140
3.28 Deploy With Nebula Native Mode (for Security Firewalls in Nebula)	. 142
3.29 Configure DHCP Domain Name (for Security Firewalls in Nebula)	. 147

Part II: Manage by Site Deployment......151

Chapter 4 Site-wide	152
4.1 Dashboard	
4.2 Topology	
4.3 Devices	
4.3.1 Access Points	
4.3.2 Switches	
4.3.3 Security Router	
4.3.4 Firewall	
4.3.5 Security Gateway	
4.3.6 Mobile Router	
4.4 Map & Floor Plans	
4.5 Clients	
4.5.1 WiFi Aid	
4.6 Applications Usage	
4.7 Summary Report	
4.8 Monitor	
4.8.1 Containment List	
4.8.2 Site Features Logs	
4.9 Configure	
4.9.1 WiFi SSID Settings	
4.9.2 Alert Settings	
4.9.3 Firmware Management	

4.9.4 Cloud Authentication	
4.9.5 Collaborative Detection & Response	
4.9.6 Quarantine Interface Configuration	
4.9.7 Site Settings	
Chapter 5	
Access Point	233
5.1 Overview	
5.1.1 Nebula Smart Mesh	
5.1.2 Smart Mesh Network Topology	
5.2 Monitor	
5.2.1 Event Log	
5.2.2 Vouchers	
5.2.3 Wireless Health	
5.2.4 Summary Report	
5.3 Configure	
5.3.1 SSID Advanced Settings	
5.3.2 Captive Portal Customization	
5.3.3 SSID Availability	
5.3.4 Radio Settings	
5.3.5 Traffic Shaping	
5.3.6 Security Service	
5.3.7 AP & Port Settings	
Chapter 6	
Switch	276
6.1 Overview	
6.2 Monitor	
6.2.1 Event Log	
6.2.2 Surveillance	
6.2.3 Surveillance Port Details	
6.2.4 IPTV Report	
6.2.5 Summary Report	
6.3 Configure	
6.3.1 Switch Ports	
6.3.2 ACL	
6.3.3 IP & Routing	
6.3.4 ONVIF Discovery	
6.3.5 Advanced IGMP	
6.3.6 Authentication	
6.3.7 PoE Schedules	
6.3.8 Switch Settings	

Chapter 7		
Security Rouler		
7.1 Overview		
7.2 Monitor		
7.2.1 Event Log		
7.2.2 Threat Report		
7.2.3 VPN Connections		
7.3 Configure		
7.3.1 Interface		
7.3.2 Threat Management		
7.3.3 Traffic Management		
7.3.4 Firewall		
7.3.5 Site-to-Site VPN		
7.3.6 SSID Advanced Settings		
7.3.7 Radio Settings		
7.3.8 Router Settings		
Chapter 8		
Firewall	342	
81 Overview	342	
8 2 Monitor		
8.2.1 Clients		
8.2.2 Event Log	342	
8 2 3 VPN Connections	343	
8.2.4 SecuReporter	344	
8 2 5 Summary Report	345	
8 3 Configure	350	
8.3.1 Port	350	
8.3.2 Interface	351	
8 3 3 Routing	359	
8 3 4 NAT	363	
8 3 5 Site-to-Site VPN	345	
8.3.6 Remote Access VPN	371	
8 3 7 Security Policy	377	
8 3 8 Security Service	385	
8.3.9 Captive Portal	300	
8.3.10 Authentication Method	مر ۸۵2	
	402 103	
8 3 12 Firewall Settings	,	
	405	
Chapter 9		
security Galeway		
9.1 Overview		

9.2 Monitor	
9.2.1 Event Log	
9.2.2 VPN Connections	
9.2.3 NSS Analysis Report	
9.2.4 Summary Report	
9.3 Configure	
9.3.1 Interface Addressing	
9.3.2 Link Aggregation Groups	
9.3.3 Policy Route	
9.3.4 Firewall	
9.3.5 Security Service	
9.3.6 Site-to-Site VPN	
9.3.7 Remote Access VPN	
9.3.8 Captive Portal	
9.3.9 Network Access Method	
9.3.10 Traffic Shaping	
9.3.11 Gateway Settings	
Chapter 10	
Mobile Router	
10.1 Overview	
10.2 Configuration	
10.2.1 Configuration: Edit	
10.2.2 Home Networking	
10.2.3 Cellular IP Passthrough	
10.2.4 Firmware Status	
10.3 Map/Photo	
10.4 Live Tools	
10.4.1 WAN Status	
10.4.2 Cellular Info	
10.4.3 LAN Stations	
10.4.4 WLAN Stations	
10.5 Backup & Restore	

Chapter 11	
Organization-wide	
	100
11.2 License & inventory	

11.2.1 License & Inventory Overview Screen	492
11.2.2 Add Devices Screen	495
11.2.3 Firmware Upgrade Screen	496
11.2.4 Add Licenses Screen	496
11.2.5 Install Wizard	497
11.2.6 License & Inventory Devices Screen	500
11.2.7 License & Inventory Licenses Screen	502
11.2.8 License & Inventory Trial Screen	505
11.2.9 License & Inventory Change Log Screen	507
11.2.10 License & Inventory Purchase History Screen	508
11.3 Administrators	508
11.4 Organization-wide Manage	512
11.4.1 Organization Portal	512
11.4.2 Configuration Management	517
11.4.3 Configuration Templates	520
11.4.4 VPN Orchestrator	523
11.4.5 Security Profile Sync	527
11.4.6 Firmware Management	535
11.4.7 Cloud Authentication	541
11.4.8 Change Log	553
11.4.9 Organization Settings	554

Part IV: Manage by Group Dep	loyment556
------------------------------	------------

Chapter 12 Group-wide

p-wide	
12.1 Introduction	
12.1.1 Creating a Group	
12.1.2 Group-Wide Menu	
12.2 Group Portal	
12.3 Org-to-Org VPN	
12.3.1 Configure Org-to-Org VPN	
12.3.2 Org-to-Org VPN Example	
12.3.3 Org-to-Org VPN Screen	
12.3.4 Add Hub	
12.3.5 Service	
12.4 Inventory	
12.5 Administrators	
12.5.1 Create/Update Administrator	
12.6 Change Log	
12.7 Group Settings	

Part V: MSP	
Chapter 13 MSP	570
	570
13.2 MSP Portal	
13.3 Cross-org synchronization	574
13.3.1 Cross-Org setting sync	574
13.3.2 Cross-Org site clone	574
13.3.3 Cross-org synchronization Screen	575
13.4 MSP Alert Templates	
13.4.1 Alert Settings	
13.5 Admins & Teams	
13.5.1 Administrator Privilege Priority	
13.5.2 Admins Screen	
13.5.3 Create/Update Administrator	
13.5.4 Teams Screen	
13.5.5 Create/Update Team	584
13.6 Change Log	
13.7 MSP Branding	587

Chapter 14 Help	
14.1 Online documents	
14.2 Troubleshooting Tips	
14.2.1 Firewall Information	
14.2.2 Data Policy	
14.3 Device Function Table	
14.4 Support Forum	
14.5 Support Request	593
Chapter 15 Troubleshooting	
15.1 Getting More Troubleshooting Help	
15.2 NCC Live Chat	
Appendix A Customer Support	
Appendix B Legal Information	

dex608

PART I Introduction & Getting Started Tutorials

CHAPTER 1 Introduction

1.1 NCC Overview

The Zyxel Nebula Control Center (NCC) is a cloud-based network management system that allows you to remotely manage and monitor Zyxel Nebula Mobile Routers, Access Points, Ethernet Switches, and Security Appliances. A Nebula Mobile Router is an LTE or NR cellular 5G indoor or outdoor router that can be managed by Nebula. You need to set up a myZyxel account in order to log into the NCC and manage your Nebula Devices, as discussed in Section 1.2.2 on page 24.

NCC feature support includes:

- System accounts with different privilege levels
 - Site Administrator: manage one site, which is a network that contains Nebula Devices
 - Organization Administrator: manage one or more organizations, which are sets of sites
- Multi-tenant management
- Inventory and license management
- Alerts to view events, such as when a device goes down
- Graphically monitor individual devices
- Securely manage Nebula Devices by using the Network Configuration Protocol (NETCONF) over TLS

Note: NCC supports IPv4 address only.

The following table describes the supported Nebula Devices.

CATEGORY	INCLUDED ZYXEL DEVICES
Hybrid Mobile Routers	LTE/NR Indoor/Outdoor Models
Security Router	SCR 50AXE
Security Gateways	NSG Series
Hybrid Security Firewalls	 ZyWALL ATP / USG FLEX / USG20(W)-VPN Series Note: The following Nebula Devices do NOT have a P1 port: USG FLEX 50 USG FLEX 100 rev 2.0 ATP100 rev 2.0
Hybrid Switches	NSW / GS / XGS / XS Series
Hybrid APs (Access Point)	NAP / NWA / WAC / WAX Series

Table 1 Supported Nebula Devices

Note: To view the list of Nebula Devices that can be managed through NCC, go to Help > Device function table.

A hybrid device can operate in either standalone or Nebula cloud management mode. When the hybrid device is in standalone mode, it can be configured and managed by the Web Configurator. When the hybrid device is in Nebula cloud management mode, it can be managed and provisioned by the Zyxel Nebula Control Center (NCC).

1.1.1 MSP (Managed Services Provider) Portal

If you have an MSP license (as discussed in Section 13.1 on page 570), use the MSP menus for crossorganization management and branding.

A Managed Service Provider (MSP) network is a group of organizations that belong to the same organization administrator. With MSP, you can:

- View the organization summary and transfer licenses
- Copy the settings from a source organization to a destination organization
- Create administrators or groups of administrators (teams) and view their login details
- Assign administrators to multiple organizations
- Upload/replace/remove the dashboard logo on NCC
- Set the support contact details
- Configure MSP alerts to monitor Nebula Devices for unexpected events (for example, online/offline events)



1.1.2 Sites, Organizations, and Groups

To manage by how Nebula Devices are deployed, use the Site-wide, Organization-wide and Groupwide menus.

In the NCC, a site is a group of Nebula-managed devices in the same network. An organization is a group of sites. A group is a collection of two or more organizations. To use the NCC to manage your Nebula Devices, each Nebula Device should be assigned to a site and the site must belong to an organization.

- A site can have multiple Nebula Devices, but can only belong to one organization.
- A site can be managed by more than one site or organization administrator.
- An organization can contain multiple sites and can be managed by more than one organization administrator.
- A myZyxel.com account can be an organization administrator and/or site administrator in the NCC (see Section 11.3 on page 508).
- A site administrator can manage more than one site.

1.1.3 Mobile Router, Security Appliance, Switches, and Access Points

To manage by Nebula Device type, use the Security Router, Mobile Router, Firewall, Security Gateway, Switch or Access Point menus.

In the following example, Nebula managed devices, such as the NAP102 or the NSW100-28P, are deployed in two separate networks (Site A and Site B). With the NCC organization administrator account, you can remotely manage and monitor all Nebula Devices even when they are located at different places.



Figure 1 NCC Example Network Topology

1.1.4 License Concept

The following section describes license concepts in NCC. Licenses unlock additional features in NCC. This means you purchase a license, assign the license to a Nebula Device, and you can then use the service in the site or organization that the Nebula Device is in.

1.1.4.1 Summary of NCC Licenses

There are three categories of licenses in NCC:

- Organization: These licenses unlock advanced features for sites and organizations.
- Security Service: These licenses unlock advanced security features on a Security Appliance/Firewall device.
- MSP: This license unlocks the MSP menu for an NCC user account.

The following table gives a summary of all licenses in NCC at the time of writing.

LICENSE	CATEGORY	ASSIGN TO	DESCRIPTION
Nebula Professional Pack	Organization	Any NCC-managed devices	Unlocks all advanced features within the Nebula Device's organization.
			For details on Pro features, see Section 1.1.4.2 on page 17.
Nebula Plus Pack	Organization	Any NCC-managed devices	Unlocks certain advanced features within the Nebula Device's organization.
			Note: Upgrade to Nebula Professional Pack to get all the advanced features.
			For details on Plus features, see Section 1.1.4.2 on page 17.
MSP	MSP	NCC user account	Unlocks the MSP menu and MSP features for an NCC user account.
MSP Trial	MSP	NCC user account	Unlocks the MSP menu and MSP features but is available only once per NCC account for 30 days. Go to More > My devices & services > Services: Activate trial for MSP.
			Note: An MSP Trial license may not be transferred to a different account. A deactivated trial license ends the service and cannot be re-claimed.
Organization Trial	Organization	Organization	Available when creating a new organization. Unlocks all Nebula Professional Pack and Nebula Security Pack (NSS) features in the organization for 30 days. There are no restrictions on the allowed number of Nebula Devices or sites.
			Note: Each Nebula user account can create 10 new organizations with trial licenses every 90 days.
Nebula Security Service N Pack (Nebula		Nebula Security Gateway (NSG)	Unlocks security services, such as anti-virus and anti-malware.
Security Service)		aevices	You can use these security services within the NSG's site.

Table 2 Licenses Summary

LICENSE	CATEGORY	ASSIGN TO	DESCRIPTION
UTM Security Pack	Security Service	USG FLEX devices	Unlocks security services, such as anti-malware, content filtering, URL threat filter, IP reputation, sandboxing, IPS (Intrusion Prevention System), application patrol, SecuReporter, CDR (Collaborative Detection & Response), and security profile sync (see Section 11.4.5 on page 527 for more information), on a Security Firewall. You can then use these security services within the
			Security Firewall's site.
Gold Security Pack	Organization and Security Service	ATP devices	Unlocks security services, such as content filtering, application patrol, DNS/URL threat filter, IPS (Intrusion Prevention System), Reputation filter, anti-malware with hybrid mode, sandboxing, CDR (Collaborative Detection & Response), security profile sync, Secure WiFi, SecuReporter, and all advanced features of a Nebula Professional Pack license.
			For details on Pro features, see Section 1.1.4.2 on page 17.
Gold Security Pack	Organization and Security Service	USG FLEX devices except USG20-VPN / USG20W-VPN / USG FLEX 50	Unlocks security services, such as content filtering, application patrol, DNS/URL threat filter, IPS (Intrusion Prevention System), Reputation filter, anti-malware, sandboxing, CDR (Collaborative Detection & Response), security profile sync, Secure WiFi, SecuReporter, and all advanced features of a Nebula Professional Pack license.
Secure WiFi	Security Service	USG FLEX devices except USG FLEX 50	Unlocks the Remote AP feature.
Content Filter Pack	Security Service	USG VPN devices	Unlocks security services, such as content filtering, SecuReporter, and security profile sync on USG FLEX 50 / USG20-VPN / USG20W-VPN devices.
Connect & Protect (CNP)	Security Service	NWA1123-ACv3, WAC500, WAC500H	Unlocks security services, such as threat protection using DNS and IP reputation filters.
Connect & Protect Plus (CNP+)	Security Service	NWA110AX, NWA210AX, WAX510D, WAX610D, WAX630S, WAX650S	Unlocks security services, such as application visibility and threat protection using DNS and IP reputation filters.
SCR Pro Pack	Organization and Security Service	SCR 50AXE	Unlocks security services, such as web filtering, Ransomware Prevention Premium, and all advanced features within the Nebula Device's organization.
			For details on Pro features, see Section 1.1.4.2 on page 17.

Table 2 Licenses Summary (continued)

1.1.4.2 Organization License Tiers

NCC features the following license tiers for organizations: Base, Plus, Professional.

- The **Base** tier is free and included with every organization.
- The **Plus** and **Professional** tier licenses unlock additional features within the organization. From a **Plus** tier license, upgrade to a **Professional** tier license to unlock all the additional features. These features are marked in the user interface with a diamond icon (\checkmark). Hover the mouse over the licensed features to view the license type.

The feature differences between the license tiers are listed below:

Table 3	NCC License Tier Differences

FEATURE	BASE	PLUS	PROFESSI ONAL	LOCATION	NOTES
Group-wide menu (Monitor – Overview, Inventory, Change log, and Configure – Settings, Org-to-Org VPN, and Administrators)	No	No	Yes	Group-wide	To create a group, you must be an NCC admin and the owner of two or more Professional organizations.
Organization change logs	No	No	Yes	Organization-wide > Organization-wide manage > Change log	
Login IPv4 address ranges for an organization	No	No	Yes	Organization-wide > Organization-wide manage > Organization settings	
Number of admin accounts	5	8	Unlimited	Organization-wide > Administrators	
Number of cloud authentication accounts	50	100	Unlimited	Organization-wide > Organization-wide manage > Cloud authentication	
Cloud authentication users with VLAN attribute	No	No	Yes	Organization-wide > Organization-wide manage > Cloud authentication (Account type: User)	
Cloud Authentication DPPSK account type	No	No	Yes	Organization-wide > Organization-wide manage > Cloud authentication (Account type: DPPSK)	
New site configuration clone	No	No	Yes	Organization-wide > Create site	
Site-wide settings sync	No	No	Yes	Organization-wide > Organization-wide manage > Configuration management	
Switch settings clone	No	No	Yes	Organization-wide > Organization-wide manage > Configuration management	
Site/Switch configuration backup and restore	No	No	Yes	Organization-wide > Organization-wide manage > Configuration management	

Table 3	NCC License Tier Differences	(continued)
---------	------------------------------	-------------

FEATURE	BASE	PLUS	PROFESSI ONAL	LOCATION	NOTES
Configuration templates	No	No	Yes	Organization-wide > Organization-wide manage > Configuration templates	At the time of writing, gateway and mobile router configuration templates are not available
Add client to block list/allow list	No	No	Yes	Site-wide > Clients	
WiFi aid	No	No	Yes	Site-wide > Clients	
Connection log	No	No	Yes	Site-wide > Clients	
Site-wide topology	No	Yes	Yes	Site-wide > Topology	
Summary report email & schedule	No	Yes	Yes	Site-wide > Summary report Site-wide > Monitor > Access point / Switch / Security gateway / Firewall > Summary report	
Time period for summary reports	24 hours	7 days	365 days	Site-wide > Summary report Site-wide > Monitor > Access point / Switch / Security gateway / Firewall > Summary report	
Time period for device monitoring statistics	24 hours	7 days	365 days	Site-wide > Devices > Access point / Switches / Security router / Security gateway / Firewall > [Select Access Points / Switches]	
Time period for client monitoring statistics	24 hours	7 days	365 days	Site-wide > Clients > [Select client]	
Time period for device event log access	24 hours	7 days	365 days	Site-wide > Monitor > Site features logs	
Export data to CSV/XML file	No	No	Yes	All monitoring pages with tables	
Open API	No	No	Yes	All monitoring information	
API access (for example, DPPSK third-party integration)	No	No	Yes	Site-wide > Configure > Site settings	
Smart email alerts	No	Yes	Yes	Site-wide > Configure > Alert settings	
Per-device firmware upgrade schedules	No	Yes	Yes	Site-wide > Configure > Firmware management	

Table 3 NCC	License Tier Differen	ices (continued)
-------------	-----------------------	------------------

FEATURE	BASE	PLUS	PROFESSI ONAL	LOCATION	NOTES
Org-wide firmware upgrade	No	Yes	Yes	Organization-wide > Organization-wide manage > Firmware management	
Priority support requests from NCC portal or Nebula app	Yes	No	Yes	Help center > Support request	
Web chat with tech support directly from NCC portal	No	No	Yes	Website footer	
Maximum uploaded photos from phone through NCC app	1	1	5	Site-wide > Devices > [select Nebula Device for example, Access points] > Photo	
Remote CLI access	No	No	Yes	Site-wide > Devices > Access Points / Security gateway / Firewall [Select AP] Live tools	
Wireless health monitor and report	No	No	Yes	Site-wide > Monitor > Access points > Wireless health	
Programmable SSID/PSK	No	No	Yes	Site-wide > Configure > WiFi SSID settings	
Dynamic Personal Pre-Shared Key (DPPSK)	No	No	Yes	Site-wide > Configure > Access points > SSID advanced settings	
Vouchers as WiFi authentication credentials	No	Yes	Yes	Site-wide > Monitor > Access points > Vouchers	
				Configure > Site settings	
				Site-wide > Configure > Access points > SSID advanced settings	
				Site-wide > Configure > Access points > Captive portal customization > [portal theme]	
Facebook WiFi	Configure in NCC	No	Yes	Site-wide > Configure > Access points > SSID advanced settings	
RADIUS accounting for captive portal	No	No	Yes	Site-wide > Configure > Access points > SSID advanced settings	

Table 3	NCC License Tier Differences	(continued)
---------	------------------------------	-------------

FEATURE	BASE	PLUS	PROFESSI ONAL	LOCATION	NOTES
Customize RADIUS NAS ID	No	No	Yes	Site-wide > Configure > Access points > SSID advanced settings	
Customize portal redirect URL parameter	No	No	Yes	Site-wide > Configure > Access points > Captive portal customization	
Smart steering per AP	Νο	Νο	Yes	Site-wide > Configure > Access points > Radio settings > [Edit the selected Access Point]	
Bandwidth Management by VLAN interface	No	Νο	Yes	Site-wide > Configure > Access points > Traffic shaping	Currently supported on NWA1123ACv3, WAC500, WAC500H, NWA110AX, NWA210AX, WAX510D, WAX610D, WAX630S, WAX650S
AP traffic log	No	No	Yes	Site-wide > Configure > Site settings	
IPTV report	No	No	Yes	Site-wide > Monitor > Switches > IPTV report	
Advanced IGMP	No	No	Yes	Site-wide > Configure > Switches > Advanced IGMP	
Switch Surveillance Monitoring with ONVIF	No	No	Yes	Site-wide > Monitor > Switches > Surveillance	Currently only supported on GS1350 series switches
Extended PoE range	Yes	Yes	Yes	Site-wide > Configure > Switches > Switch ports > [select port]	Currently only supported on GS1350 series switches
Automatic PoE device recovery	No	Yes	Yes	Site-wide > Configure > Switches > Switch ports > [select port]	
Port bandwidth control	Yes	Yes	Yes	Site-wide > Configure > Switches > Switch ports > [edit the selected port]	
Vendor ID-based VLAN	No	Yes	Yes	Site-wide > Configure > Switches > Switch settings	

Table 3	NCC License	Tier Differences	(continued)
---------	-------------	------------------	-------------

FEATURE	BASE	PLUS	PROFESSI ONAL	LOCATION	NOTES
IP interface and static route	No	No	Yes	Site-wide > Configure > Switches > IP & routing	
Remote SSH in Live tools	No	No	Yes	Site-wide > Devices > Switches: Switch Details > Live tools > Remote access	
IP Source Guard	No	No	Yes	Site-wide > Configure > Switches > Switch settings	
Nebula cloud authentication	Yes	Yes	Yes	Site-wide > Configure > Switches > Authentication	
Time period for security service (AV/App Patrol/CF/IDP/NSS) analysis report	24 hours	7 days	365 days	Site-wide > Monitor > Security gateway > NSS analysis report	Requires Nebula Security Gateway (NSG) Nebula Security Service (NSS) – Security Pack (SP) license
Traffic log archiving	No	No	Yes	Site-wide > Monitor > Firewall > SecuReporter	
VPN topology with traffic usage	No	No	Yes	Organization-wide > Organization-wide manage > VPN Orchestrator	
Smart VPN	No	No	Yes	Organization-wide > Organization-wide manage > VPN Orchestrator	
VPN provision script email	No	No	Yes	Site-wide > Configure > Security gateway / Firewall > Remote access VPN (L2TP/IPSec)	
Collaborative Detection & Response (CDR) with automatic respond action	No	No	Yes	Site-wide > Configure > Collaborative detection & response	Requires Security Firewall UTM Security Pack license
Smart mesh with manual select of mesh controller (root) and automatic fall back to auto mode	Yes	Yes	Yes	Site-wide > Devices > Access points	Currently supported on NWA110AX, NWA210AX, WAX510D, WAX610D, WAX630S, WAX650S, NWA1123ACv3, WAC500, and WAC500H APs

FEATURE	BASE	PLUS	PROFESSI ONAL	LOCATION	NOTES
Traffic logs to SecuReporter	No	No	Yes	Site-wide > Configure > Site settings	Also available for Gold Security Pack, UTM Security Pack, and Content Filter Pack
Cellular IP Passthrough	No	No	Yes	Site-wide > Devices > Mobile Router > Configuration	Currently only supported on NR7101 and LTE7461
Remote configurator in Live tools	No	No	Yes	Site-wide > Devices > Mobile Router > Live tools > Remote configurator	Requires LTE or NR cellular 5G indoor or outdoor router running the latest firmware

 Table 3
 NCC License Tier Differences (continued)

Organization License Grace Period

If a Professional or Plus license expires while assigned to a Nebula Device or you add an unlicensed Nebula Device to the organization, you have a 15-day grace period during which the organization's license remains active. During the grace period, you must perform one of the following actions:

- Assign a valid Plus or Professional license to the unlicensed Nebula Device.
- Remove the unlicensed Nebula Device from the organization.

If the expired Nebula Device is still in the organization after the grace period elapses, the organization automatically downgrades to the Base tier.

The grace period status can be any of the following:

- Near Expiring: Any Nebula Devices with licenses expiring within 15 days before the grace period has started.
- License Expired: Any Nebula Devices with expired licenses after the grace period.
- Insufficient Licenses: Any Nebula Devices that are unlicensed, or lower tier licensed Nebula Devices added during the grace period.

1.1.4.3 General License Information

License Validity

Each license has a validity period, for example: 6 months, 1 year, 2 years. After being activated, a license also has an expiry date, which is calculated as Activation Date + Validity Period. For example, if a 1-year license is activated on January 1st 2022, then its expiry date is January 1st 2023.

Note: A license cannot be deactivated. An activated license continues counting towards its expiry date, even if its licensed service is deactivated.

Bundled and Renewal Licenses

A **bundled license** is a license that is included when you purchase a Nebula Device. The bundled license is automatically assigned to the purchased Nebula Device when you add the Nebula Device to NCC.

A **renewal license** is a license purchased separately from a Nebula Device as a license key, from Zyxel or a third-party reseller. To assign a renewal license to a Nebula Device, go to **Organization-wide** > License & inventory > License and then click +Add. See Section 11.2.7 on page 502 for more information.

1.2 Getting Started

You can perform network management with the NCC using a web browser. Use a browser that supports HTML5, such as Microsoft Edge, Mozilla Firefox, or Google Chrome. The recommended browser is Google Chrome.

View the browser in full screen mode to display the NCC portal properly.

1.2.1 Connect Nebula Managed Devices

Connect your Nebula managed devices (such as the NAP102 or the NSW100-28P) to your local network. Your local network must have Internet access. See the corresponding Quick Start Guides for hardware connections.

1.2.2 Access the NCC Portal

Go to the NCC portal website.

1 Enter *http://nebula.zyxel.com* in a supported web browser. Click **Get Started**.



Note: The NCC requires a myZyxel account before you can register and manage Nebula Devices. Log into the NCC with your myZyxel account. Click **Create account** if you do not have a myZyxel account and create an account with your existing email address.

2 Enter the Email Address and Password, and then click Sign In.

	Sigr	۱In		
	Don't have	an account? Cr	eate account	
	Email Addr	ess		
	samuel vu@	zvxel com tw		
	Password			
				(jii)
	Rememb	ber me. Sig	gn In	
			or	
		Try	Demo	
	l have signe Resend Act	ed up but haven't i vation Email	activated my accou	nt.
Legal No	tice	Terms of Use	Privacy	Cookle Settings
	2.50.13 Copy	rright © 2023 Zyxel ar	nd/or its affiliates. All Rights	s Reserved.

Note: Click **Try Demo** to enter the **Demo Site**. The **Demo Site** allows you to explore the NCC Portal.

3 Click Go under Nebula Control Center to log in to NCC.



Alternatively, click **Go** under **Nebula Orchestrator** to go to the Nebula SD-WAN (Orchestrator) web portal to configure ZyWALL VPN devices. This is only available if you have purchased the SD-WAN license for Orchestrator Management.

Crchestrator			Control Center →)?	:	: T
GROUP	Select / Create Group Organization	Define Create Sites Org Plan and AutoVPN	Deploy Device Data and ZTP Confirm		
Monitor Information	Group Name*		¥		
Configuration					
	TION If a group only has one or For example:	ganization, you can use the same nam	ne for both.		
Description		Group: Zyxel			
Monitor		Organization: Zyxel			
Configuration create a new group or organization of an exist	a new	Site: HQ			
SITE group.					
× 1					
Monitor					
Configuration					
-					
				Concel	Next >

Figure 2 Nebula SD-WAN (Orchestrator)

You can click Control Center to go back to the NCC platform.

4 Click Create organization to create a new organization. If this is the first time you have logged into NCC, proceed to step 10.

If you have more than one organization, click a row to select the organization you want to manage.

and the second sec	nebula Control Center					?	\odot	තු	 8
		Accounts for san	nuel.yu@	zyxel.com.t	w				
		Choose organization	1						
		Q Search	-		① Create organization				
		Name		Туре					
		a		Nebula Base Po	ack				
		pro12		Nebula Profess	ional Pack				

5 The NCC supports two-factor authentication (2FA) to add a second layer of security to your account. Click Manage account to enable Two-factor authentication on the following page. Otherwise, you can skip 2FA and go to step 10 directly.

ZYXEL	Ad	ccount Information	S Sama enmine % Manag		5
Account Information	Profile				
Two-Factor Authentication	First Name Last Name Email Address Country / Region	Samuel V. consultar@zyxel.com.tw Taiwan, Republic Of China	Edit Profile		
	Password			_	
	Last changed: N/A		Change Password		

6 Click Two-Factor Authentication and then click the switch to enable Two-Factor Authentication.

ZYXEL		:	#	T
	Two-Factor Authentication			
Account Information Two-Factor Authentication				
	Two-Factor Authentication			
	Use Two-Factor Authentication to secure login every time!			

7 The following screen appear. Activate the two-step verification service using the Google Authenticator app or your email address. If you select **Google Authenticator**, install the app on your smartphone and scan the QR code on the NCC web screen to get a 6-digit one-time code. Then enter the code and click **Verify** to authenticate your identity.

Two-Factor Authentication	
Google Authenticator	Email Verification
Step 1 Install Google Authenticator on you	r mobile device
ANDROID APP ON Google Play	
Step 2 Open Google Authenticator and sco	an this QR code.
Enter Key Manually	
Step 3 Enter the six-digit code shown in Go	oogle Authenticator.
Enter Code	
	Cancel Verify

Alternatively, click **Email Verification** to use your email to authenticate.

If you select **Email Verification**, an email is sent to your myZyxel account's email address. Enter the code exactly as it appears in the email and click **Verify**.

Google Authenticator	Email Verification
Ve have send a verification emai Please enter the six-digit code in	l to @***el.com.tv the email.
Enter Code	

8 Enter the verification code to get 10 backup codes, which help regain access to your account in case your smartphone is not available for 2FA the next time you need to log in again.

Note: If you generate a new set of backup codes, the old set will become inactive.

	Two-Fact	or Auther	tication			
Two-	Factor Auth	entication				
6	Google Auth Get code on G	enticator oogle Authenti	cator app			0
	Backup Code These one-off- away from you generate more	passcodes allo r mobile phone codes as you r	w you to sign ir . Each backup weed.	n when you use code can only	Google Authentic be used once. You	cator I may
	122220	482019	064804	716777	867627	
	485769	496888	306540	556545	164640	
			Downle	oad Gene	erate New Code	,
	Email Verific Get code via e	ation mail				0

Write down or print out the backup codes for your account. You can enter the backup code on the NCC web page to authenticate your identity at the next login. Each code can only work once. Click **Download** to download the backup codes.

9 To re-log in Nebula after the **Two-Factor Authentication** is enabled. Go to **Applications** > **Nebula** and then enter a code to log in your Nebula account.

← → C (≜ a	ccounts.myzyxel.com/users/sign_in	Q	$\dot{\mathbf{x}}$	
ZYXE	L			
	Two-Factor Authenticator			
	Enter the six-digit code for "myZyxel" shown in Google Authenticator.			
•	Verify			•
•	Legal Notice Terms of Use Privacy Cooke Settings 2.49.6 Copyright © 2021 Zysel and/or its affiliates. All Rights Reserved.	•		

10 If this is the first time you have logged into NCC, the setup wizard welcome screen displays. You need to create your organization and sites, register Nebula Devices and associate them with a site. See Chapter 2 on page 46 for how to use the wizard.



1.3 NCC Portal Overview

The following summarizes how to navigate the Nebula web site from the **Dashboard** screen. The NCC portal screen is divided into these parts:

Figu	re 3 NCC O	verview					
s.	nebula Control Center Organiz	ation: pro12	▼ > Site: pro123	- A		Q 🤉 💬	A 🎕 🎞 🛛
			Welcome to Nebula Profession	al Pack! Make the most of you	ur network without limitations.		×
8 % [Site-wide > <u>Dashboard</u> Dashboard						음우 Customize
8, 0(AP Status	Wireless Clients	🛲 Switch Status	PoE power	Appliance Status	Mobile router	WAN utilization
11 B) 🖗	Online O% Heavy loading	0	No Switches	No Switches	No Gateways	No Mobile router	No Gateways
er 🛞 's	()00 SSIDs (by Usage)	No Data	9	Clients (by Usage)	Ne	Data	9
в			@	Appliance Clients (by	Usage)		99
		No Gateways			No G	ateways	
	O Clients	O bytes Traffic = 20:22:05:1	99) 20035	Clients manufacturer	o Data	Clients OS	99) o Doto
	AP application usage	No Dota	90)				
		Co	Last login: 2 pyright © 2023 Zyxel and/or its a	hour ago from 612228626 (Ti ffiliates. all Rights Reserved.) Terms of Use Privacy Polic	aiwan, Hsinchu) Version: 16.00 (build 230321.0701, Sy	local)	▣

- A Title Bar
- B Navigation Panel
- C Main Screen

1.3.1 Title Bar

The title bar provides common links and is always at the top of NCC.

Figure 4 NCC Title Bar

nebula Control Center Organization: pro12 - > Site: pro123 -	५ ७ 💬 म 🕸 🏭 💿
---	---------------

The icons provide the following functions.

Table 4 NCC Title Bar

LABEL	DESCRIPTION
Group	This shows the name of the groups you are managing, if your NCC account has an MSP license. Click to choose another group if you have multiple groups.
	Note: To create a group, you must be the owner of two or more Pro pack organizations that are not currently assigned to a group, as discussed in Section 12.1.1 on page 557.
Organization	This shows the name of the organization you are managing. Click to choose another organization, access the MSP portal or create a new organization.
Site	This shows the name of the site you are managing. Click to choose another site if you have multiple sites in the selected organization.
Search	Use this to search for managed Nebula Devices by model, description or MAC address.
Help	Click this to view the documentation for NCC and NCC-compatible devices. For example, to view the Security Firewall Series configuration and hardware information, locate the documents under Security Appliance.
More	Click this to view your account information, login history and active sessions. You can also view your Nebula Devices and manage NCC licenses linked to your account.
Notification	Click this to view log messages.
Settings	Click this to select a display language for the screens, or change the theme between dark and light mode.
Applications	Click this to open a list of links to different Zyxel sites, such as myZyxel, Nebula, SecuReporter, Astra, Circle, Marketplace, Store, Education, and the Community.
Account	Click this to manage your NCC account settings, or to sign out of NCC.

Note: If the browser window is too narrow, the layout of the title bar changes and some settings are hidden under the More menu.

Figure 5 Layout of the Title Bar



1.3.1.1 Site/Organization/Group

Select the site, organization and group that you want to manage.

- If you select a group, you can only select organization in that group. Select List all Groups from the Group drop-down list to view all organizations and group.
- If you have multiple organizations, select MSP Portal from the Organization drop-down list box to view your organization summary (see Section 13.2 on page 570).

Note: You need to have an MSP license to view the MSP Portal.

- If you need to have more organizations, select **Create organization** from the **Organization** drop-down list box to create a new one (see Section 1.4 on page 43).
- If you need to have more sites, select **Create site** from the **Site** drop-down list box to create a new one.

Figure 6 NCC Title Bar: Group/Organization/Site

Group: List All Groups	•	Organization: Ne	bula_Org	•	Site:	Hsinchu	•
------------------------	---	------------------	----------	---	-------	---------	---

1.3.1.2 Search

Click this to search for NCC-managed devices by model, description or MAC address. You can enter partial search criteria.

Figure 7 Search

	Q
Q Search for	×
 Start typing the name of clients or de 	vices
🚺 Start typing the name of clients or de	vices

1.3.1.3 More

Click the More icon at the top right-hand corner of the **Dashboard** screen to view and configure account settings.

Figure 8 More



The following table describes this menu.

Table 5Login Account Menu

LABEL	DESCRIPTION
Profile	This shows account information, such as name, address, and phone number.
My devices & services	This shows a list of all Nebula Devices in NCC that have your login account as the owner. You can filter the list of Nebula Devices by name, serial number, model, or organization. You can also register licenses to your account, such as an MSP license.
Active sessions	Shows all active web browser sessions for this login account. Click End Session to close a session and force the user to log into NCC again in that browser.
Recent logins	Shows the login history for this user account, including IPv4 address, location, and time.

Click **My devices & services** and the following screen appears. Click **Devices** to view all Nebula Devices of the user account which can be managed by NCC, and/or all Nebula Devices not registered to this user account but with a Full (Delegated) administrator privilege. See the table on MSP cross-org > MSP cross-org manage > Admins & teams > Admins in Section 13.5.2 on page 581 for details on the organization privileges.



he list of all Nebula d	devices and services t	that have been owne	ed by your acco	ount.			
Devices Servic	ces Purchase hi	istory					
a armt	- (7	E) davidante					C. Franke
Q, Search	• (o) devices.					[] Export ♥
Q search	MAC address	Serial Number	Name	Organization	Site	Device owner	
Model USG FLEX 500	MAC address	Serial Number	Name	Organization	Site	Device owner	E)
Model USG FLEX 500 NWA110AX	MAC address RRECARDSTRA ROOFARESTOOD	Social Number Steal Asociation Social Asociation	Name	Organization	Site	Device owner soon uite (Bhrout soon tu) soom uite (Bhrout soon try	

Click **Services** to view and configure the start dates, end dates, registered dates, activated dates and statuses of an MSP license, purchase or register a license key, and export the list of MSP licenses in CSV/XML format.

Figure 10 My Services



Click **Purchase history** to view the order ID, purchase date, number of licenses, statuses of purchased MSP license(s), and export the information in CSV/XML format.

Figure 11 Purchase History



1.3.1.4 Notifications

Click this alert icon to view log messages for the selected site.

Figure 12 NCC Notification



1.3.1.5 Settings

Click the Settings icon at the top right-hand corner of the screen to view and configure NCC settings.





The following table describes this menu.

Table 6 Settings Menu

LABEL	DESCRIPTION
Dark mode	Click this to apply a black background and white text to the white background and black text on the NCC screen.
Language	Select the NCC display language.
	At the time of writing, the following languages are available: English, Chinese, Japanese, German, Russian, French.

Figure 14 Dark Mode

1	nebula Organiz Control Center	ation: pro12	• Site: pro123			୧ ଡ 💬	A 🅸 🏭 💿
			Welcome to Nebula Professio	nal Packi Make the most of you	r network without limitations.		×
å B	Site-wide > <u>Dashboord</u> Dashboard						문후 Customize
1 6 6 6 6 1	AP Status O/1 Oran Heavy loading	Clients Office	No Switches	PoE power No Switches	Appliance Status No Gateways	Mobile router	WAN utilization
⊞ ₹⊗ 49	() _{0[}] SSIDe (by Usage)	No Data	9	(荷), Clients (by Usage)	No	Doto	Θ

1.3.1.6 Applications

Click this to display a list of related NCC links.

Figure 15 Related NCC Links



1.3.1.7 Account

Click the **Account** icon at the top right-hand corner of the screen to view and configure NCC account settings.

Figure 16 Account



The following table describes this menu.

Table 7 Account Menu

LABEL	DESCRIPTION
Manage account	Click this to edit your account settings at myZyxel.
Sign out	Sign out of NCC.

1.3.2 Navigation Panel

Use the NCC menu items to configure network management for each site, organization and/or Nebula Device. Click the arrow (\ll) on the upper right corner of the navigation panel to collapse or expand the navigation panel menus.

Table 8	Navigation Menus Overview	

LABEL	DESCRIPTION	
Use these menus to set up customer networks.		
Site-wide	Manage Nebula Devices in a site.	
Organization -wide	Manage multiple network sites within an organization.	
Table 8	Navigation Menus Overview	(continued)
---------	---------------------------	-------------

LABEL	DESCRIPTION
Group-wide	Manage settings for multiple organizations and create VPN links between groups in the organization. Two or more Pro tier organizations can be a group.
MSP	Create multiple organizations and change the branding and assign administrators to multiple organizations.
Use these menus	to set up customer Nebula Devices.
Mobile router	Manage the Zyxel LTE/NR devices.
Security router	Manage the SCR 50AXE.
Security gateway	Manage the ZyWALL NSG devices.
Firewall	Manage the ZyWALL ATP, USG FLEX, and USG20(W)-VPN devices (firewalls).
Switch	Manage the Zyxel Switches.
Access point	Manage the Zyxel APs (Access Points).
Help center	Access the Zyxel community forum, submit a support ticket, view User Guides for Nebula managed devices, view ports used by Nebula, view Nebula privacy policies, and view devices/ features that can be managed by Nebula.

This is a summary of the menu details.

Table 9 NCC Menu Summary

LEVEL 1	LEVEL 2 / LEVEL 3	FUNCTION
Site-wide	Dashboard	Use this menu to view Nebula Device connection status and traffic summary.
	Topology	Use this menu to view Nebula managed-device connections in your network.
	Devices	
	Add devices	Click + to register a Nebula Device and add it to the site.
	Access points / Switches / Security router / Firewall / Security gateway / Mobile router	Use this menu to view Nebula Device connection status and traffic summary.
	Map & floor plans	Use this menu to locate Nebula Devices on a world map or on a floor plan.
	Clients	
	Client list	Use this menu to view the connection status and detailed information of all wired and WiFi clients connected to Nebula Devices (Access Points, Switches, Security Appliances) in the site.
	WiFi Aid	Use this menu to display an overview of the AP's WiFi clients connection issues, as an aid to troubleshooting.
	Connection log	Use this menu to view all related event logs between Access Points and WiFi clients, and DHCP logs of Nebula Security Appliances (NSG, ZyWALL USG FLEX, ATP, and USG20(W)-VPN). Association, Authentication, Disconnection, and DHCP event logs that occur are summarized in chronological order to aid in troubleshooting.
	Applications usage	Use this menu to view usage of applications such as Social Network, Telephony (VoIP), Advertising, News, Web Services in the network.
	Summary report	Use this menu to view network statistics for a site, such as bandwidth usage, power usage, top Nebula Devices, top clients and/or top SSIDs.

NCC User's Guide

LEVEL 1	LEVEL 2 / LEVEL 3	FUNCTION	
	Monitor		
	Access points		
	Event log	Use this menu to view all events on the Access Point. An event is something that has happened to a Nebula managed device.	
	Vouchers	Use this menu to create and manage vouchers that allow WiFi network access	
	Wireless health	Use this menu to view health of the WiFi networks for the supported Access Points and connected clients.	
	Summary report	Use this menu to view network statistics specific to Access Points in the site.	
	Switches		
	Event log	Use this menu to view all events on the Switch. An event is something that has happened to a Nebula managed device.	
	Surveillance	Use this screen to view information about Powered Devices (PDs) connected to ports on the Switch.	
	IPTV report	Use this menu to view available IPTV channels and client information.	
	Summary report	Use this menu to view network statistics specific to Switches in the site.	
	Security router		
	Event log	Use this menu to view all events on the Security router. An event is something that has happened to a Nebula managed device.	
	Threat report	Use this menu to view statistics for threat management categories.	
	VPN connections	Use this menu to view status of the site-to-site VPN connections.	
	Firewall		
	Event log	Use this menu to view all events on the Security Firewall. An event is something that has happened to a Nebula managed device.	
	VPN connections	Use this menu to view status of the site-to-site VPN connections.	
	SecuReporter	Use this menu to view the statistics report for NSS (Nebula Security Service), such as content filtering, Intrusion Detection and Prevention (IDP), application patrol, and anti-virus.	
	Summary report	Use this menu to view network statistics specific to the Security Firewall in the site.	
	Security gateway		
	Event log	Use this menu to view all events on the Security Appliance. An event is something that has happened to a Nebula managed device.	
	VPN connections	Use this menu to view status of the site-to-site VPN connections.	
	NSS analysis report	Use this menu to view the statistics report for NSS (Nebula Security Service), such as content filtering, Intrusion Detection and Prevention (IDP), application patrol, and anti-virus.	
	Summary report	Use this menu to view network statistics specific to the Security Appliance in the site.	
	Containment list	Use this menu to view and manage Nebula Devices contained by CDR (Collaborative Detection & Response).	
	Site features logs	Use this menu to view log messages about configuration changes made by the NCC for the site.	

LEVEL 1	LEVEL 2 / LEVEL 3	FUNCTION
	Configure	
	WiFi SSID settings	Use this menu to view and configure SSID settings and authentication methods.
	Access points	
	SSID advanced settings	Use this menu to configure network access, traffic options and advanced settings for SSID profiles.
	Captive portal customization	Use this menu to configure captive portal settings for SSID profiles.
	SSID availability	Use this menu to configure SSID visibility settings and set whether the SSID is enabled or disabled on each day of the week.
	Radio settings	Use this menu to configure global radio settings, such as maximum output power or channel width, and enable smart client steering for all Access Points in the site.
	Traffic shaping	Use this menu to configure the maximum bandwidth and load balancing.
	Security service	Use this menu to enable application visibility and optimization, and IP reputation filter on the managed Access Point.
	AP & port settings	Use this menu to configure load balancing settings and enable or disable a port on the managed Access Point and configure the port's VLAN settings.
	Switches	
	Switch ports	Use this menu to view the Switch port statistics and configure Switch settings for the ports.
	ACL	Use this menu to configure the access control list in order to control access to the Switches.
	IP & Routing	Use this menu to configure layer 3 features such as creating IP interfaces and static routes on the Switch.
	ONVIF discovery	Use this menu to enable ONVIF and configure ONVIF VLAN ID for the selected Switch.
	Advanced IGMP	Use this menu to enable and configure IGMP snooping and create IGMP filtering profiles.
	Authentication	Use this menu to configure authentication servers and policies.
	PoE schedules	Use this menu to set the schedule for Switches in distributing power to powered devices.
	Switch settings	Use this menu to configure global Switch settings, such as (R)STP, QoS, port mirroring, voice VLAN and DHCP white list.

Table 9	NCC Menu Summary	(continued)
---------	------------------	-------------

LEVEL 1	LEVEL 2 / LEVEL 3	FUNCTION	
	Security router		
	Interface	Use this menu to configure interface address, subnet mask and VLAN ID settings on the Security Router.	
	Threat management	Use this menu to enable threat management categories, configure exception list using client's name/IP address, and allowed/blocked domain name list.	
	Traffic management	Use this menu to manage the use of various applications on the network and control access to specific web sites or web content.	
	Firewall	Use this menu to configure firewall rules for outbound traffic, create new NAT rules and edit/delete existing NAT rules.	
	Site-to-Site VPN	Use this menu to configure VPN rules between Security Routers.	
	SSID advanced settings	Use this menu to configure WiFi security, band selection, assisted roaming and U-APSD (Unscheduled automatic power save delivery) settings for the SSID profiles.	
	Radio settings	Use this menu to configure global radio settings, such as maximum output power or channel width, and enable smart client steering for all Security Routers in the site.	
	Router settings	Use this menu to configure DNS settings.	
	Firewall		
	Port	Use this menu to configure network mode and port grouping on the Security Firewall.	
	Interface	Use this menu to configure interface address, subnet mask and VLAN ID settings on the Security Firewall.	
	Routing	Use this menu to view and configure policy routes, static routes and WAN load balancing.	
	NAT	Use this menu to view and configure virtual servers and NAT settings.	
	Site-to-Site VPN	Use this menu to configure VPN rules between Security Firewalls.	
	Remote access VPN	Use this menu to enable and configure IPsec VPN or L2TP VPN rules from off-site clients to an on-site Security Firewall.	
	Security policy	Use this menu to configure firewall rules for outbound traffic, application patrol, schedule profiles and port forwarding rules for inbound traffic.	
	Security service	Use this menu to enable content filtering and block access to specific web sites. You can also enable Anti-virus and Intrusion Detection and Prevention (IDP) on the Security Firewall.	
	Captive portal	Use this menu to configure captive portal settings for each Security Firewall interface.	
	Authentication Method	Use this menu to configure network access settings through a captive portal or Nebula Cloud Authentication.	
	Wireless	Use this menu to configure different SSID profiles for your ZyWALL USG FLEX 100W and USG20W-VPN.	
		Note: This menu only appears for the ZyWALL USG FLEX 100W and USG20W-VPN.	
	Firewall settings	Use this menu to configure the DNS server and address records and also set the external AD (Active Directory) server or RADIUS server that the Security Firewall can use in authenticating users. You can also specify walled garden web site links for all interfaces on the Security Firewall.	

LEVEL 1	LEVEL 2 / LEVEL 3	FUNCTION
	Security gateway	
	Interface addressing	Use this menu to configure network mode, port grouping, interface address, static route and DDNS settings on the Security Appliance.
	Policy route	Use this menu to view and configure policy routes.
	Firewall	Use this menu to configure firewall rules for outbound traffic, application patrol, schedule profiles and port forwarding rules for inbound traffic.
	Security service	Use this menu to enable content filtering and block access to specific web sites. You can also enable Anti-virus and Intrusion Detection and Prevention (IDP) on the Security Appliance.
	Site-to-Site VPN	Use this menu to configure VPN rules.
	Remote access VPN	Use this menu to enable and configure IPsec VPN or L2TP VPN settings.
	Captive portal	Use this menu to configure captive portal settings for each Security Appliance interface.
	Network access method	Use this menu to enable or disable web authentication on an interface.
	Traffic shaping	Use this menu to configure the maximum bandwidth and load balancing.
	Gateway settings	Use this menu to configure the DNS server and address records and also set the external AD (Active Directory) server or RADIUS server that the Security Appliance can use in authenticating users. You can also specify walled garden web site links for all interfaces on the Security Appliance.
	Alert settings	Use this menu to set which alerts are created and emailed or sent by the Zyxel Nebula Mobile app. You can also set the email addresses to which an alert is sent.
	Firmware management	Use this menu to upgrade firmware or schedule firmware upgrades for Nebula Devices in the site.
	Cloud authentication	Use this menu to add user accounts and grant user access to the selected site through different authentication methods, such as the MAC-based authentication, captive portal or the IEEE 802.1x authentication method.
	Collaborative detection & response	Use this menu to view and configure the policies and notification settings for malware, IDP and web threats and corresponding containment actions to quarantine, alert or block. This is only available for ZyWALL USG Flex Series and ZyWALL ATP Series at the time of writing.
	Site settings	Use this menu to change the general settings for the site, such as the site name, Nebula Device login password, captive portal reauthentication, SNMP, AP traffic logs to a Syslog server, traffic logs to SecuReporter, WiFi network authentication voucher settings, and API access for DPPSK third- party integration.

 Table 9
 NCC Menu Summary (continued)

LEVEL 1	LEVEL 2 / LEVEL 3	FUNCTION	
Organization- wide	License & inventory	Use this menu to manage your licenses and view the summary of Nebula Devices which have been registered and assigned to the sites in the selected organization.	
	Administrators	Use this menu to view, remove, or create a new administrator account for this organization.	
	Organization-wide n	nanage	
	Organization portal	Use this menu to view a list of sites belonging to the selected organization and detailed information about the Nebula Devices connected to the sites.	
	Configuration management	Use this menu to synchronize the configuration between sites or switch ports and back up or restore a configuration file.	
	Configuration templates	Use this menu to create or delete a configuration template or bind a site to the template.	
	VPN orchestrator	Use this menu to view and manage VPNs created for the selected organization.	
	Security profile sync	Use this menu to synchronize the settings of URL threat filter, anti-malware and content filtering on the selected gateways.	
	Firmware management	Use this menu to upgrade firmware or schedule firmware upgrades for Nebula Devices in the organization.	
	Cloud authentication	Use this menu to create or remove user accounts and grant user access to all sites in the selected organization through different authentication methods, such as MAC-based authentication, captive portal, or the IEEE 802.1x authentication method.	
	Change log	Use this menu to view log messages about configuration changes in this organization.	
	Organization settings	Use this menu to configure security settings or delete the organization.	
Group-wide	Group-wide manage		
	Group portal	Use this menu to view organization and license details of a selected group.	
	Org-to-Org VPN	Use this menu to view and manage VPNs between members in the group.	
	Inventory	Use this menu to view Nebula Devices belonging to organizations. You may also export the list of Nebula Devices found to your computer.	
	Administrators	Use this menu to view, remove, or create a new administrator account for the selected group.	
	Change log	Use this menu to view log messages about configuration changes in the group.	
	Group settings	Use this menu to configure group information and group members.	

 Table 9
 NCC Menu Summary (continued)

LEVEL 1	LEVEL 2 / LEVEL 3	FUNCTION	
MSP	MSP cross-org manage		
	MSP portal	Use this menu to create multiple organizations and change the branding and assign administrators to multiple organizations.	
	Cross-org synchronization	Use this menu to sync or clone organization-wide settings from a source organization to a destination organization.	
	Alert templates	Use this menu to configure MSP alert templates to monitor Nebula Devices for unexpected events (for example, online or offline events).	
	Admins & teams	Use this menu to create administrators or groups of administrators (teams) and view their login details.	
	Change log	Use this menu to view log messages about configuration changes in the Admins & teams and Cross-org synchronization screens.	
	MSP branding	Use this menu to upload/replace/remove the dashboard logo. You can also set the support contact details.	

 Table 9
 NCC Menu Summary (continued)

1.4 Create Organization

Use this screen to first create an organization, then create a site (network) in the organization, and finally add Nebula Devices to the site.

- Note: You have to contact Zyxel customer support if you need to change the device owner at myZyxel or remove an Organization from the NCC. But an administrator can remove sites without customer support. Configure your Nebula Device owners and organizations carefully. See also Section 11.2 on page 492.
- Note: There is no limit as to how many organizations you can create, but you can only activate a trial license up to 10 new organizations every 90 days. The expiration date of the organization created using a trial license is shown.
- 1 Click **Create Organization** from the **Organization** drop-down list box in the title bar. The Wizard starts. See Chapter 2 on page 46 for detailed information about how to use the wizard to create an organization and site. Otherwise, click **Exit Wizard** to close the wizard and display the **Create organization** screen.
- 2 Enter a name for your organization.
- 3 If you already have one or more than one organization under your account and you want to copy the organization settings of an existing one, select the organization name from the Copy setting from field and also Add this Org to MSP Teams by selecting existing teams before clicking the Create organization button.
- 4 Click the Create organization button to add a new organization.

Figure 17	Create Organization
-----------	---------------------

S.	nebula Control Center	← > Organiz	zation: Create organization	• Site: Select site	*	Q	•	ŝ	 5
	New Organization								
		Clone a new or	ganization from one of your ex	isting organization.					
		Organization-w	vide settings for your new orga	inization will be copied from the one y	ou specify.				
10		This operation	cannot be undone.						
	Organization name:	Sample	×						
	Country:	Talwan	•						
	Copy setting from	(None)							
	Add this Org to MSP Teams:								
"©		Select teams							
\otimes		Create organi	ization						
da.									

5 Choose whether to activate a one-month trial of Nebula Pro Pack and Nebula Security Services for the organization. For example, USG FLEX 700, Secure WiFi License, 1MO; USG FLEX 700, UTM Security Pack License, 1MO; Nebula Professional Pack License, 1MO.

1.5 Choose Organization

When you have more than one organization on your account, the following screen displays right after you log in. Select the organization you want to manage now, access the **MSP Portal** or click **Create organization** to add a new one.

Note: You need to purchase an MSP license to see the MSP Portal menu.

Accounts for somuely	/แตรงหยุ่ากา	m tw
ŵ MSP Portal		
Choose organization		
Q Search	•	\oplus Create organization
Name	Туре	
End-OCT	Nebula Bas	se Pack
Hotel Maeir	Nebula Bas	e Pack
MY HOME	Nebula Bas	e Pack
Nebula_Org	Nebula Pro	fessional Pack
NSBU	Nebula Pro	fessional Pack
Switch_AE	Nebula Pro	fessional Pack
Test July	Nebula Pro	fessional Pack
Test_October	Nebula Bas	se Pack

Figure 18 Choose Organization

1.6 Cloud-Saving Mode

If you do not log into a base (free) license tier organization for over 30 days, the organization automatically enters Cloud-saving mode to save your network bandwidth and cloud resources.

When Cloud-saving is enabled, NCC does not record any data traffic statistics, except the following:

- Event logs
- Security Appliance WAN interface logs between the Nebula Device and NCC, and
- NSS (Nebula Security Service) analysis report (requires Nebula Security Pack (Nebula Security Service) license).

To disable Cloud-saving mode, click the **Cloud-saving mode** switch or click the link in the NCC banner when notified.

Figure 19 Cloud-saving mode

Welcome back	×
You haven't logged in to this Organization over 30 days. NCC has deactivated the collection of the traffic stats (except for the event log for troubleshooting) to conserve bandwidth & cloud resource You may disable cloud-saving mode on the banner and NCC will resur collection.	device's es. ne data
	Close

CHAPTER 2 Setup Wizard

2.1 Setup Wizard

- The setup wizard helps you create an organization and site, add Nebula Devices, upgrade your Nebula Device firmware, and set up WiFi networks quickly.
- The wizard appears automatically after you log in the first time or if there is no organization created under your account.
- The wizard also starts when you click **Create Organization** from the **Organization** drop-down list box in the title bar.



2.1.1 Step1: Run the Wizard

1 After logging in to https://nebula.zyxel.com, the following screen appears. Click GO to start the NCC wizard.



2 The welcome screen displays when you are creating the first organization under your account. Click Let's Start to begin.



Note: This screen will appear only if you have not created a new organization.

2.1.2 Step 2: Create an Organization and Site

1 Enter a descriptive name for your organization and site. Both names must consist of 1 – 64 characters.

- 2 Select the time zone of your location. This will set the time difference between your time zone and Coordinated Universal Time (UTC).
- 3 Click Next to continue.

		Exit Wilcord
	First step is to create yo	our Organization and Site
<u>01</u>		
Nebula is organized into Organizations, for example,	Organization	× *
"YourCompany" or "YourClient", and Sites, for example, "London	(and a	
Branch" or "Factory". You can create as many Organizations and	- DILU	×
Sites as you need once you're up and running. The country allows	Country	
us to set the correct time zone for your site and the legal	Taiwan	•
requirements for settings like radio power on occess points.	Timezone	
	Asia - Taipei (UTC +8.0)	*
Please enter your Organization and Site names and select the		
correct Country and Time Zone.		Nant

2.1.3 Step 3: Add Your Nebula Devices

1 Enter your device's MAC address and serial number.

You can also leave the fields blank and click **Next** to move on to the next step without adding a Nebula Device.

- 2 Click the + Add button to register and add the Nebula Device to the site. You can register multiple Nebula Devices at a time.
- 3 Click Next to proceed.

		Exit Wizord
_ 02	Let's now add your device(s) to Nebula	
To add your device(s) you will need to input the MAC address, which is the number that looks like this. 7C 99 DD 39 AC E0, and	And a second	
the Serial Number that looks similar to: S891345239054. These are	MAC address Serial number Name	
located on the box and at the bottom of each device, it may	× × ×	
appear as:		
Serial Number	C Add	
MAC address	Bock Next	
You might just click Next to skip this step.		

2.1.4 Step 4: Upgrade your Nebula Device Firmware

You should always use the most recent firmware to get the latest features, improvements, and bug fixes by clicking **Yes** (default setting).

Even if you choose not to upgrade the firmware, NCC will still perform a mandatory upgrade to **Stable** firmware version if the Nebula Device's firmware have security vulnerabilities, and/or lack key performance improvements. See Table 176 on page 538 for the description of a **Stable** firmware.

The following table shows when a mandatory firmware upgrade occurs for the different Nebula Device types.

NEBULA DEVICE TYPE	MANDATORY FIRMWARE UPGRADE TIME
Access Points	The mandatory firmware upgrade occurs when the Nebula Device is online with NCC.
Switches / Firewalls / Security Gateways	The mandatory firmware upgrade occurs after registering the Nebula Device on NCC.

Table 10 Mandatory Firmware Upgrade Behavior

Click **Next** to proceed.



2.1.5 Step 5: Set up your WiFi Network

1 Configure the WiFi settings for the managed APs. Enter the WiFi network name (SSID) and the WiFi password.

You can also leave the fields blank and click **Next** to move on to the next step without setting up the main WiFi network.

2 Configure the ID number of the VLAN to which the SSID belongs.

The VLAN ID 1 is generated automatically by the NCC and reserved for a gateway's LAN 1 and LAN 2 by default. The IPv4 subnets 192.168.1.0/24 and 192.168.2.0/24 are also reserved for these two LAN interfaces.

If you enter a different VLAN ID other than the default one ("1") in the VLAN field, click the Set up VLAN interface link to create a gateway interface with the specified VLAN ID. You need to configure an IPv4 address and subnet mask and enable the DHCP server function for this interface.

3 Click **Next** to proceed.

	Exit Witcord
	Let's get your WiFi set up
04	ES SCHEE PASCAL WAR DEC BARD
Enter your WiFi name. This is what you will select from a device	WIFI Name (SSID)
when connecting to your network. If you leave the password	Proteword (Pre-Shared Key)
empty then anyone will be able to access your network without the need to enter a password. If a password is entered, we will	VLAN
automatically add WPA2 security so that every device will need to	1 x
enter this password to connect to your network.	Set up VLAN interface Sotuwork
Foreway Optionally, you could configure the IP address settings of	Bock Next
the WIFI VLAN in case a Nebula gateway is installed in this site.	Skip WiFi settings
You might just click Next to skip this step.	

2.1.6 Step 6: Set up a Guest WiFi Network

1 Configure WiFi and VLAN settings for guest users who can wirelessly access the Internet or networks through Nebula Devices.

You can also leave the fields blank and click **Next** to move on to the next step without setting up the guest WiFi network.

- 2 If you want to enable web authentication, select Clicking "Agree" to access the network to block network traffic until a client agrees to the policy of user agreement. Otherwise, select Using their Facebook account to join the network to block network traffic until the client logs in using his/her existing Facebook account.
 - Note: If you do not enable any WiFi security, your network is accessible to any WiFi networking device that is within range.
 - Note: The guest network function and Layer 2 isolation between clients are enabled on this WiFi network by default.

If you enter a different VLAN ID other than the default one ("1") in the VLAN field, click the Set up VLAN interface link to create a gateway interface with the specified VLAN ID. You can set the gateway interface as a guest interface, configure the IPv4 address and subnet mask and enable the DHCP server function for this interface.

Note: If you set the guest WiFi network to use the same VLAN ID as the WiFi network and have already configured the gateway interface, the gateway interface configuration fields will be grayed out in this screen.

3 Click Next to proceed.



	Exit Witton
	Need to set up a Guest WiFi?
	WW/ Harve DDDD
05	Guest Test ×
Enter your Guest WiFi name. If you leave the password empty,	12345678 ×
then anyone will be able to access your network without the need	How do you prefer quest to access your quest network (Captive partal)?
to enter a password. Additionally, you can choose to add a	
captive portal that will redirect the guests to either click "I agree"	O No captive web portai
or by using their Facebook account to access your guest network.	Clicking "Agree" to access the network
	Using their Facebook account to join the network
Bateway Optionally, you could configure the IP address settings of	VAN I X
the Guest WIFI VLAN in case a Nebula gateway is installed in this	
site. The interface can also be set as Guest to restrict devices	🔺 Set up VLAN interface Bateway
access to Internet only.	Bock Next
You might just click Next to skip this step.	

2.1.7 Step 7: Set up the Deployment Method

If you added a ZyWALL USG FLEX / ATP / USG20(W)-VPN Series device in step 3, you need to select a deployment method for management by Nebula. Select **Nebula native mode** if available. If not, select **Zero Touch Provision mode** and configure an email address to send an activation link to the administrator who is in charge of managing the Nebula Device.

		Deployment	Method	Exit Wizord
	Model Name	ATP200	Show device information	
	Deployment Method 0			
Configure WAN settings for the gateway device that you added	Nebula native mo Connect your a Internet access Login GW GU a	omputer to the GW LAN port and	connect WAN port to a modem or router that has on settings.	
earlier in the wizard. Nebula Control Center (NCC) then assigns the device you added as the gateway device for the new site. NCC also sends the WAN settings to the specified email address, as an encoded URL.	EXTEREL			
Dateway After you have finished this wizard, follow the instructions in the email to apply the WAN settings to the gateway device.	Post.			
You might just click Next to skip this step.				
	🔘 Zero Touch Provisi	lon mode .		
			Ask	Question

2.1.7.1 Nebula Native Mode

To use the Nebula native mode deployment method, perform the steps described in On the Nebula Device.

2.1.7.2 Zero Touch Provision Mode

To configure the Zero Touch Provisioning (ZTP) settings, do the following in NCC:

- 1 Enable VLAN Tag and configure the VLAN ID (1 4094) for the WAN port.
- 2 Select Static/DHCP/PPPoE/PPPoE with static IP for the WAN type of the Nebula Device.
- 3 If you select DHCP, enter the MTU (Maximum Transmission Unit) to set the maximum size (1280 1500) of each data packet, in bytes, that can move through this interface.

If you select Static, enter the IP Address, Subnet Mask, Default Gateway, First/Second DNS Server, and MTU (1280 – 1500).

If you select **PPPoE**, select the **Authentication Type**, enter the **Username**, **Password**, and **MTU** (1280 – 1492).

If you select PPPoE with static IP, select the Authentication Type, enter the Username, Password, IP Address, Default Gateway, First DNS Server and MTU (1280 – 1492).

- Note: Configure the VLAN ID and WAN interface for the Nebula Device exactly as your ISP gave it to you.
- 4 Click Next.
- 5 Select I will install Firewall by myself to receive an activation email and activation link/file. Alternatively, if you want another administrator to activate the Nebula Device, enter the recipient's Email Address.
- 6 Click Next.
- 7 Select where the Nebula Device will get and install the activation file, from a computer or through a USB drive.

		Deployment M	ethod	Exit Wizord
	Model Name	USG FLEX 100	Show device information	
Configure WAN settings for the gateway device that you added earlier in the wizard. Nebula Control Center (NCC) then assigns the device you added as the gateway device for the new site. NCC also sends the WAN settings to the specified email address, as an encoded URL. Forteact After you have finished this wizard, follow the instructions in the email to apply the WAN settings to the gateway device. You might just click Next to skip this step.	Model Name Deployment Method Mobilizin notative model Can Figure your WAR VLAN Tog VLAN Tog	Deployment M	ethod show device information × (1 - 4000) • Port P2 × × × × × × × × × × × × ×	
		Internet		

On the Nebula Device

- 1 Back up the current configuration (in case you want to return to On Premises mode later).
- 2 Reset the Nebula Device if it was previously configured.
- 3 Connect the Nebula Device's WAN port to a modem/router that has Internet access.
- 4 Connect your computer to the Nebula Device's LAN port.
- If you select Nebula native mode, go directly to step 7. Click the activation link in the email. Alternatively, save the activation file in the root directory of a USB drive. Then insert the USB drive into your Nebula Device. Wait until Nebula Zero Touch Provisioning is successful.





- 6 Click Go to Nebula Control Center to configure the Nebula Device using NCC.
- 7 When you log into the Web Configurator for the first time or when you reset the Nebula Device to its default configuration, the Initial Setup Wizard screen displays. Choose Nebula Mode to manage your Nebula Device remotely using Nebula Control Center (NCC).
- 8 Follow the wizard to configure the Nebula Device network settings to connect to NCC. The screens vary depending on the encapsulation type. Refer to information provided by your ISP to know what to enter in each field. Leave a field blank if you do not have that information.

Note: Refer to the Nebula Device User's Guide for more information.

2.1.8 Step 8: View the Summary

- 1 A summary of the wizard configuration will display after you complete the deployment method.
- 2 You can click a section's edit icon (\square) to modify its setting.
- **3** You must click **Go to Nebula Dashboard** to save your changes in the wizard; otherwise click **Exit Wizard** to close the wizard screen without saving the settings.



Organization Summary 🛛				Devices		
Organization name for Mod Image: Site name for Mockup with Taiwan Asia - Taipei (UTC +8.0)	skup a long name descrip	tion		2 1 12 22 22	Mobile router(s) Firewall(s) 2 Switch(es) 2 Access point(s)	
Overview for your Wifi configurati	ion					
🛜 Wifi setting 🛛		Î	🔓 Guest Wifi setting 🕜			
WiFi Name (SSID):	Wifi name for Mock	up	Guest WiFi Name (SSII	D):	Wifi name for Mockup	
WiFi Password:	12345678		Guest WiFi Password:		12345678	
VLAN Interface (Gateway):	VLAN 2 - 232.22.123	.2	Authentication:		Click-to-continue	
	DHCP server ON		VLAN Interface (Gate	eway):	VLAN 2 - 232,22,123,2 DHCP server ON	
Overview for your Security Applic	ance configuration [2				
Model Name: USG20-VPN	WAN Setting:	• WAN Port: 22	Recipient:	vn.zyxe	el@gmail.com	
		• WAN Type: STATIC				
		• VLAN ID: 2				
	Everythi	ng seems fine,	ready to go?			

2.1.9 Step 9: Activate the Trial License(s)

After setting up the wizard, the following screen will appear. You can decide if you want to activate a one-month trial period of Nebula Pro Pack and Nebula Security Services for the organization. Before deciding on the trial license to activate, see Section 11.2.8 on page 505 for more information.

Note: Before activating a trial license, make sure the services in the license can be used by a Nebula Device in the organization.

If you choose to activate a trial license, click to select the trial license(s) and then click **OK**. NCC will send you an email reminding you to purchase the full license when the trial is close to expiring.



Note: To set the administrator privileges, see Section 13.5.1 on page 581 for more information.

CHAPTER 3 Tutorials

3.1 Overview

This chapter shows you how to use the NCC's various features.

- Add a Nebula Device
- Activate and Assign a License for a Nebula Device, Site, or Organization
- Monitor a Site
- Know What Licenses are Set to Expire in My Site or Organization
- Renew an Expired License
- Transfer Licenses
- Reset the Nebula Password
- Change an Organization and/or Site Name
- Maintain Firmware
- Assign an Administrator to Manage a Nebula Device
- Transfer the Ownership of the Organization
- Manage a Configuration Template
- Activate an MSP License
- Configure CNP/CNP Plus Security Services
- Delete an Organization
- Remote Access VPN Setup
- Route L2TP VPN Traffic
- Resolve WiFi Connection Problems (for Nebula APs only)
- Configure Voice VLAN (for Nebula Switches only)
- Manage IPTV (for Nebula Switches only)
- Enable IP Source Guard (for Nebula Switches only)
- Set Up MAC Authentication With NCAS (for Nebula Switches only)
- Set Up Dynamic VLAN With RADIUS (for Nebula Switches only)
- Monitor Dynamic VLAN Using Event Logs (for Nebula Switches only)
- Register a Nebula Device (mobile router) in Nebula
- Using Collaborative Detection and Response (CDR)
- Deploy With Nebula Native Mode (for Security Firewalls in Nebula)
- Configure DHCP Domain Name (for Security Firewalls in Nebula)

3.2 Add a Nebula Device

This section shows you how to add a Mobile Router, Security Gateway, Nebula Firewall, Access Point or Switch to a selected organization and site on NCC for management.

1 Go to the Site-wide > Devices > + > Add devices screen. Click +Add.

dd devices using MAC	C Address and Serial Number. Whe	n you register a device, that device	vill be added to your organi	zation's inventory and assigned to your sit
	Q Search	(0) devices.		(+A
Device name	Serial number	MAC Address	Model	Registered On

- 2 Enter the Serial number, MAC address, and a descriptive Name of the Nebula Device you want to add. Click the Finish button to save the changes.
 - Note: When a Nebula Device is added to a site other than a Nebula Device owner, the **Acknowledge** button appears. Click this button first to confirm that the **Serial number** and **MAC Address** information are correct. Then click the **Next** button to check the Nebula Device firmware.

Add devices		[] ×
Add devices	Devices	
Firmware upgrade	Enter one or more MAC address and serial number.	
	Or you can download the template here and import multiple records for faster registration.	
	What Zyxel devices support Nebula?	
	Where can I find these numbers?	
	MAC address Serial number Name Model License info Expiration date Assign licen	ses from invent
	+ Add another device	
	Registered device will be added to Organization Creator account in myZyxel.com. Acknowledge	
	<	Þ
	Next	Cancel

3.3 Activate and Assign a License for a Nebula Device, Site, or Organization

This section shows you how to activate and assign a license for a Nebula Device, site, or organization. See Section 1.1.4.2 on page 17 for a summary of NCC licenses.

The following table describes the license types at the time of writing.

LOCATION	LICENSE TYPE	APPLICATION
MSP (Managed Services Provider)	MSP	NCC (Nebula Control Center) user account
Organization-wide	Professional / PLUS	AP (Access Point) / NSG (Nebula Security Gateway) / Switch / USG FLEX device
Organization-wide	Gold Security	ATP device
Site-wide	NSS (Nebula Security Service)	NSG device
Site-wide	UTM (Unified Threat Management) Security / Secure WiFi	USG FLEX device
Site-wide	Content Filter	USG FLEX 50 / USG20-VPN / USG20W-VPN device
Site-wide	Connect & Protect (CNP) / Connect & Protect Plus (CNP+)	NWA1123ACv3, WAC500, WAC500H / NWA110AX, NWA210AX, WAX510D, WAX610D, WAX630S, WAX650S device
Site-wide	SCR Pro	SCR 50AXE

Table 11 License Types

3.3.1 Bundled License and Add-on License

A bundled license is a license that is included when you purchase a Nebula Device (Mobile Router, Access Point, Switch, NSG, USG FLEX, ATP, and USG20(W)-VPN). The bundled license is automatically assigned to the purchased Nebula Device when you add the Nebula Device to NCC. A bundled license cannot be transferred to another Nebula Device.

An add-on license is a license purchased separately from a Nebula Device as a license key, from Zyxel or another vendor. An add-on license can be applied to any Nebula Device.

3.3.2 License States

The following are the license states in NCC.

- Active the license pack is assigned to a Nebula Device, is activated, and is in use (expiration countdown/timer has started).
- Queued the license pack is assigned to a Nebula Device, is activated, but not yet in use.
- Inactive the license pack is assigned to a Nebula Device, but is not yet activated in NCC.
- Unused The license pack is assigned to an organization, but is not yet assigned to a Nebula Device and not yet activated in NCC.

3.3.3 License Activation Process

You must have a Nebula Device and a license pack to activate a license. Perform the following to activate a license.

1 In the Organization-wide > License & inventory, click Action > Add more licenses.

Overview	Devices	Licenses		Trial	Change log	Purchase History		
Organization status	à 🚺						Actions -	
Organization type:	Professional	Pack (Expire on: 2023-06-23	1)				Add more devices	
NCC license:	Device(s)	with over 90 days but less th	ian 1 year licer	nse.			Add more licenses	
Security license:	Device(s)	expired or unlicensed. Chec	k license issue	2.		L		
							Install wizard	
Device status by ex	piration date							
Device detail status								
Device detail status		Nebula Professional Pac	k 🕶 🟮		4 - F 4	"	#	
Device detail status		Nebula Professional Pac	k 🕶 🟮	# in org	# unlicensed (expired)	# expires within 90 days	# expires after 90 days	# înactive
Device detail status		Nebula Professional Pac Device type	k - 0	# in org 1	# unlicensed (expired) 0	# expires within 90 days 0	# expires after 90 days 1	# inactive
Device detail status)	Nebula Professional Pac Device type C Access Point - NAPI02	k • 0	# in org 1	# unlicensed (expired) 0	# expires within 90 days 0	# expires after 90 days 1 1	# inactive 0
Device detail status)	Nebula Professional Pac Device type C Access Point - NAPI02 Switch	k v 0	# in org 1 1	# unlicensed (expired) 0 0	# expires within 90 days 0 0	# expires after 90 days 1] 3	# inactive 0 0
Device detail status	d) 0	Nebula Professional Pac Device type C Access Point - NAPIO2 Switch - GS1350-6HP	k ▼ 0 ^	# in org 1 1 3	# unlicensed (expired) 0 0 0	# expires within 90 days 0 0	# expires after 90 days 1] 3	# inactive 0 0 0
Device detail status	d) 0 lays 0 two 5	Nebula Professional Pac Device type Contemporal Access Point - NAPIO2 Contemporal Switch - GS1350-6HP	k • 0	# in org 1 1 3 1	# unlicensed (expired) 0 0 0	# expires within 90 days 0 0 0	# expires after 90 days 1 1 3 1	# inactive 0 0 0 0 0 0 0 0
Unlicensed (expire Expires within 90 c Expires within 90 c Inactive	d) 0 lays 0 iyys 5 0	Nebula Professional Pace Device type Content of the second seco	k • 0	# in org 1 1 3 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	# unlicensed (expired) 0 0 0 0 0	# expires within 90 days 0 0 0 0 0	# expires after 90 days 1 1 3 1 1	# inactive 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
Device detail status Unlicensed (expire Expires within 90 de Expires after 90 de Inactive	d) 0 lays 0 iys 5 0	Nebula Professional Pac Device type Contemport Access Point - NAPIO2 Switch - GS1350-6HP - XG52220-30HP - XS1930-12HP	k ▼ 0	# in org 1 1 3 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	# unlicensed (expired) 0 0 0 0 0 0	# expires within 90 days 0 0 0 0 0 0 0	# expires after 90 days 1 1 3 1 1 1 1	# inactive 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
Unlicensed (expire Expires within 90 of Expires after 90 do	d) O fays O iys 5 O	Nebula Professional Pac Device type C Access Point - NAPIO2 Switch - GS1350-6HP - XG52220-30HP - XS1930-12HP C Security Appliance	k ▼ 0 ∧ ∧	# in org 1 1 3 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	# unlicensed (expired) 0 0 0 0 0 0 0	# expires within 90 days 0 0 0 0 0 0 0 0 0	# expires after 90 days 1 1 3 1 1 1 1	# inactive 0 0 0 0 0 0 0 0 0 0 0 0 0

2 Enter the License key and the License information will display.

Add licenses	×
Add licenses	Add licenses Enter one more license keys. Or You can download the <u>template</u> here and <u>import</u> multiple license keys for faster registration.
	License key License information
	LIC-PRO-4YR-202106170006 X Nebula Professional Pack License, 4YR
	+ Add
	Cancel Finish

3 Click Finish. The license is now assigned to your organization and site.

Note: A newly assigned license will not start its expiration countdown/timer until activated. Multiple add-on Plus Pack and Pro Pack licenses can be assigned to the same Nebula Device managed by NCC.

- 4 In the Organization-wide > License & inventory, select the Devices tab.
- 5 Locate the Nebula Device to assign a license(s). Click the Actions button and select Assign license on the device row.
- 6 Clear any license that you do not want added to the Nebula Device.

- 7 For multiple licenses of the same type to be added to the Nebula Device, set the number of licenses in the Select # of license field.
- 8 Set the expected expiration date criteria from the License assignment mode.
 - Assign min. period NCC assigns one of each license type with the shortest duration to each Nebula Devices.
 - Assign all NCC assigns all selected license type equally to each Nebula Device.
 - Target expiration date Set a future date. NCC assigns an equal number of licenses to each Nebula Devices until the expiration date (future date) is reached or exceeded.
- 9 Click Please check this box if you want to activate licenses and upgrade. Then, click Finish.

Assign licenses	
Assign licenses	Assign licenses You are currently assigning unused licenses which can be assigned to any matching devices Nebula Professional Pack License, 4YR I Q Search I er MAC address Expiration date i Selected licenses Select # of license is in the select # of
	204 99:CF.4F.E3:7C:99 NCC N/A → 2025-08-17 Nebula Professional Pack License, 4YR * 1 Select # of license → ↑ ↑ ↑ ↑ ↑ ↑ ↑ ↑ ↑ ↑ ↑ ↑ ↑ ↑

The features that will be unlocked depends on the license type purchased.

Figure 20 License Activation Process



3.4 Monitor a Site

This section shows you how to view and monitor your Nebula Devices and WiFi/wired networks within a site.

1 Go to the Site-wide > Dashboard screen. To change the default view, click Customize to show the Widget, Reset, and Close buttons.

Dashboard						
AP Status	Wireless Clients		Switch Status	PoE Power	Gateway Status	WAN Throughput
15/19 Online Os, Heavy loading	17		No Switches	No Switches	1/1 Online 57% CPU Usage	<pre>160.86↓ < 315.55↑ </pre>
Gateway Network Applicatio	ons		(241)	👼 Gateway Clients (by Usage)		(24h)
254 MB Sada HTT	ube 1775 MB desk 31.4 MB o 18 MB Discovery 75 MB P 6.5 MB	Nbr Mdr Dhc HTT Win	rs 2 MB ns 17 MB pp 14 MB PPS 12 MB dows Marketplace 12 MB	DC85DFEA8D29 00004001242E DE5KTOP-000K666 00004001244E 9453208EEC58		O bytes O bytes O bytes O bytes O bytes O bytes
[n]] SSIDs (by Usage)		(24h)	Wireless Clients (by Usage)	(241)	(Wireless Clients Manufactur	rer (24h)
NSBU_SVD3_6	(14.2	2 GB	TWNBWLANSVD38	14.2 GB	Intel	Corporate
NSBU_SVD3_4	10.6	5 GB	TWNBWLANSVD06	10.4 GB	Zyxe	I Communications Corporation
NSBU_SVD3_3	8.7	GB	TWNBWLANSVD05	8.7 GB	Vizic	, Inc
NSBU_SVD3_5 NSBU_SVD3_10	6.9	GB GB	TWNBWLANSVD24 TWNBWLANSVD13	8.7 GB	Athe	ros Communications, Inc. e, Inc.
Wireless Clients OS		(241)	APs (by Usage)		(241)	
Wind Othe Apple Apple Andr	lows 7 or newer r e iOS e macOS X oid		7 Clients Tro	KSBU_SVD3_10(WAC5 KSBU_SVD3_6(WAX61 KSBU_SVD3_6(WAX61 KSBU_SVD3_4(WAX65 KSBU_SVD3_1(WAX65 KSBU_SVD3_1(WAX65 KSBU_SVD3_1(WAX65	000H) 00D) 00S) 00S) 00D)	

2 Click Widget to select which widgets to display. For example, clicking SSIDs (by Usage) will show the top 5 SSIDs with the highest percentage of bandwidth usage in the past 24 hours. Click Reset to restore the dashboard back to the default view.

Click Close to hide the Widget. Reset, and Close buttons and show the Customize button.

Dashboard	문 Widget	े Reset Close
AP Google Map Map	X AP Status X Mobile router SSIDs (by Usage) X AP Google Map X AP Traffic PoE Power X Wireless Clients (by Usage) X Hit for Threat Protection by CNP Service Wireless Clients X WAN Utilization X Wireless Clients Manufacturer Switch Status X Appliance Network Applications X Wireless Clients OS Appliance Network Climate (furthers) X Devices	WAN Utilization
Hsin Ocity	Applicitie dutus Mapplicitie Clients (by Usuge) Note: widgets include top information	No Gateways

3.5 Know What Licenses are Set to Expire in My Site or Organization

Use the **Overview** tab in the **Organization-wide** > **License & inventory** to keep track of what licenses are set to expire to prevent a cut in services.

Overview	Devices	Licenses		Trial	Change log	Purchase History		
Organization status	0						Actions +	当 Purchase licens
Organization type:	Professional Pa	ck (Expire on: 2023-02-18))					
NCC license:	Device(s) will	expire in 90 days. Check	license issue,					
Security license:	Device(s) ex	pired or unlicensed. Check	k license issue.					
Secure WiFi license:	Device(s) exp	pired or unlicensed. Chec	k license issue.					
CNP license:	Device(s) exp	pired or unlicensed. Checl	k license issue.					
Device status by exp Device detail status	iration date	Nehula Professional Packa						
Device status by exp	iration date	Nebula Professional Paci Device type	k 🕶 🕄	#in org	# unlicensed	# expires within 90 draws	# expires after 90	# inactive
Device status by exp	iration date	Nebula Professional Paci Device type	k ▼ 0	#inorg 4	# unlicensed (expired) 0	# expires within 90 days 0	# expires after 90 days 4	# inactive
Device status by exp	iration date	Nebula Professional Paci Device type Access Point - NWA110AX	k - 0	# in org 4	# unlicensed (expired) 0 0	# expires within 90 days 0 0	# expires after 30 days 4	# inactive 0
Device status by exp	iration date	Nebula Professional Paci Device type Access Point - NWA110AX - NWA220AX-6E	k • 0	# in org -4 1	# unlicensed (expired) 0 0	# expires within 90 days 0 0	# expires after 90 days 4 1	# inactive 0 0
Device status by exp Device detail status	iration date	Nebula Professional Paci Device type Access Point - NWA110AX - NWA220AX-6E - WAX650S	k • 0	# in org 4 1 1	# unlicensed (expired) 0 0 0	# expires within 90 days 0 0 0	# expires after 90 days 4 1 1	# inactive 0 0
Device status by exp Device detail status Unlicensed (expired Expires within 90 de Expires within 90 de	iration date	Nebula Professional Paci Device type Access Point - NWA110AX - NWA220AX-6E - WAX650S	k - 0	# in org 4 1 1 2	# unlicensed (expired) 0 0 0 0	# expires within 90 days 0 0 0 0	# expires after 90 days 4 1 1 2	# inactive 0 0 0

The license health is shown in the Device detail status and the following are the definition:

- Red Nebula Device with expired license.
- Orange Nebula Device with license that will expire in 90 days.
- Blue Nebula Device with license that will expire in less than a year but over 90 days.
- Green Nebula Device with license that will not expire within a year.

If a Pro or Plus tier license expires while assigned to a Nebula Device or you add an unlicensed Nebula Device to the organization, you have a 15-day grace period during which the organization's license remains active. See Section on page 23 for details on a Nebula Device entering the grace period and what actions you must take.

3.6 Renew an Expired License

An administrator account should have read and write (Full) access privilege to add or renew licenses for Nebula Devices in the organization. Go to **Organization-wide** > **License & inventory** to view the available (unused) licenses assigned to your organization.

C	Overview	Device	License	Change log		
	5 assigned	(Pro Pack, 1MO)	9 unused (Pro Pack, 1YR)	1 unused (Plus Pack, 1MO)) 2 unused (Plus Par	ck, 1YR)
Actic	ons* Q (license	eStatesFilter=ACTIVE 🔻	18 matches in 18 licen	ises.	+	Add 🕒 Export 🕶
	License Key		License states	Associated device	Activate date	Action E
	WTECT 7IAVE ALL	VILVII ACTIEDOV	Active	20:21:03:21:13:46	2021-06-15	Action -
		210202160647	Queued	20:21:03:21:13:40	2021-06-11	Action -
	110 DI LIC 1140 01	0100001000 AE	Active	20:21:03:21:13:40	2021-06-07	Action -
		IOL II - 22070-1420	Inactive	20:21:03:21:13:41	≅.	Action -
		10000160641	Unused			Action -

In the example figure above, four kinds of licenses are available for assigning to your Nebula Device: Pro Pack 1MO / 1YR and Plus Pack 1MO / 1YR. Click any one of the license. For example, if you click Plus Pack 1YR, then only the two Plus Pack **License Keys** with 1-year validity will display in the table.

Select the check box and click **Action**. Then click **Assign license**. See Section 3.3.3 on page 59 for details on assigning a license to a Nebula Device.

If the expired Nebula Device is still in the organization after the grace period elapses, the organization automatically downgrades to the Base tier. See Section on page 23 for details on a Nebula Device entering the grace period and what actions you must take.

3.7 Transfer Licenses

A license assigned to an organization and Nebula Device can be transferred to another Nebula Device in the same or different organization. The following guidelines apply when transferring licenses:

- The Nebula Devices must have the same owner.
- Bundled, Trial, and Promotion licenses cannot be transferred. (See Table 159 for more information.)
- If the license transfer causes the Nebula Devices in the organization to be without a valid license, the organization automatically downgrades to the Base tier.

3.7.1 Select Transferable Licenses

To select a transferable license(s), do the following:

- 1 Go to the Organization-wide > License & inventory > Licenses screen.
- 2 Select the license you want to transfer. Click Actions, and then click Transfer license.

Overview	Devices	Licenses	Trial	Change log	Purchase History				
			2 assign	ned					
Actions - Q. Sea	irch	• 1 selected , 3 matches	in 3 licenses.			Show expired I	icenses	+ Add	🕒 Expo
Actions + Q Sea	irch	T selected , 3 matches Service	in 3 licenses. License states	License expiration date	Remaining days	Show expired I Claim date	icenses Associat	+ Add ted device	C Exp
Actions + Q Sea License Key LIC-NPRO-ZZIYO	rch	T selected , 3 matches Service Nebula Professional Pack License, 1YR	in 3 licenses. License states Active	License expiration date	Remaining days	Show expired I Claim date 2021-03-26	icenses Associat B8:EC:A3	+ Add ted device 3:AE:EA:14	C Exp Actions Action
LIC-NPRO-ZZ1YC	00F202103261313	T selected , 3 matches Service Nebula Professional Pack License, 1YR Nebula Professional Pack License, 1YR	in 3 licenses. License states Active Expired	License expiration date 2023-04-11 2022-04-10	Remaining days 91 days	Show expired I Claim date 2021-03-26 2021-03-31	icenses Associat B8:EC:A3 B8:EC:A	+ Add ted device 3:AE:EA:14 Change or	Actions Actions
License Key License Key License Key License Key License Key License Key License Key	00F202103261313 00F202103261311 00F202104091025	T selected , 3 matches Service Nebula Professional Pack License, 1YR Nebula Professional Pack License, 1YR Nebula Professional Pack License, 1YR	In 3 licenses.	License expiration date 2023-04-11 2022-04-10 Queued	Remaining days 91 days - 366 days	Show expired I Claim date 2021-03-26 2021-03-31 2021-04-09	icenses Associat B8:EC:A B8:EC:A B8:EC:A	+ Add ted device 3:AE:EA:14 Change of Assign lice	Actions Action Action

3.7.2 Undo Assigning a License

An administrator account should have read and write (Full) access privilege to un-assign licenses. Only an **Inactive** license (license is assigned to a specific Nebula Device but not activated) can be un-assigned.

To un-assign a license, do the following:

- 1 Go to the Organization-wide > Configure > License & inventory > License screen.
- 2 Select the License Key with an Inactive license state that you want to undo assign. Click Action, then click Undo assign. The license will return to the Unused license state.

Overview	Device	License Cl	hange log		
5 assigned	1 unused (Pro Pack, 1MO)	9 unused (Pro Pack, 1YR)	unused (Plus Pack, 1MO)	2 unused (Plus Pac	ok, 1YR)
Actions Q (licer	nseStatesFilter=ACTIVE -	18) matches in 18) license	S.	Activate date	Add 🕒 Export +
	11 Y 1.41 A & J IEDOM	Active	20:21:03:21:13:46	2021-06-15	Action -
	743031000160647	Queued	20:21:03:21:13:40	2021-06-11	Action -
	21210202160645	Active	20:21:03:21:13:40	2021-06-07	Action -
	B IOLII VVOZO MVIEO	Inactive	20:21:03:21:13:41	-	Action -
	11/20210001010	Unused			Action -

3.7.3 Transfer a License to a Different Organization

Only an **Unused** license (a license which is assigned to an organization but not assigned to a specific Nebula Device) can be transferred. Both source and destination organizations should belong to the same owner.

To transfer a license to another organization, do the following:

- 1 Perform the steps described in Select Transferable Licenses.
- 2 With the licenses you want to transfer selected, click Actions and then click Change organization.

Overview	Devices	Licenses Tr	ial	Change log	Purchase History			
		4 assigned		4 unus (Pro	sed Pack, 2YR)			
Actions - Q. Searc	h	▼ ④ selected , (15) matches in	15 licenses.		ء 💽	Show expired I	icenses + A	dd 🕒 Expo
Change organization		Service	License states	License expiration date	Remaining days	Claim date	Activate date	Associated de
Assign license	CF4F477DF1-01	Nebula Professional Pack License, 1MO	Expired	2021-04-19	-	2021-03-19	2021-03-19	BC:CF:4F:47:70
Undo assign	202106040001-11	Nebula Professional Pack License, 2YR	Unused	0-1	731 days	2022-03-23	-	
Transfer license	202106040001-12	Nebula Professional Pack License, 2YR	Unused		731 days	2022-03-23	×.	
LIC-NPRO-ZZ2Y00	F202106040001-13	Nebula Professional Pack License, 2YR	Unused	-	731 days	2022-03-23		
			Unused		791 days	2022.02.22	2	

3 Select the **Organization** you want to transfer the licenses to. The current organization will be excluded from the list. Then click **Yes**.

Change organization	×
You are going to move license(s) from organization.	
License Key	
WITECT DTVTT EVI NIA ECLACE LAEUVD	
Organization MY HOME	-
	Cancel Yes

You have successfully transferred a license to another organization, but without assigning it to a Nebula Device yet.

3.7.4 Assign a License to a Nebula Device in the New Organization

To assign a license(s) to a Nebula Device in the new organization, do the following:

- 1 Perform the steps mentioned in Transfer a License to a Different Organization.
- 2 Select the Organization and Site where the license is transferred.

Organizatio	e: Quitch AE	Cita: AF Test
Organizatio	Q	Site. AE_lest
	End-OCI	ack! Make the most of your network
	Hotel Maeir	
<u>se & inventory</u>	MY HOME	
	PRO Nebula_Org	
evices	PRO shawn	I Change log
	PRO Switch_AE	
	PRO Test July	1
	Test_October	- 4

- 3 Go to the Organization-wide > Configure > License & inventory > Device screen.
- 4 Select the **Devices**, click **Actions**, then click **Assign license**.

	verview	Devices	Licenses	Tric	al Cho	inge log	Purchase History		
		1 Access Point		3 Switch		1 Security App	liance	О м	obile Router
ctio	ns▼ In use Uni	used Both Q S	earch_	• 1	selected in 5 devic	es.			+ Add 🕒 Exp
	Device name	Device type	Model	Serial number	MAC address	Claim date	License expiration date	License info	o Actions
	60-21-07-94-E1-94	Access Point	NAP102	0160706100040	R0-01-07-04-E1-04	2022-12-20	2024-07-22	Nebula Pro	fessional Pack Actions
2									
	50-5300-50-00-80	Security Gateway	NSG50	01701 07100056	50-53-00-50-00-70	2022-12-20	2023-07-22	Nebula Pre	Change organization
	50-53-00-50-00-80 30-06-30-11-05-0 <i>8</i>	Security Gateway Switch	NSG50 XS1930-12HP	01701 07100056	50-50-00-50-00-40 20-06-20-11-05-84	2022-12-20 2023-01-10	2023-07-22 2025-01-10	Nebula Pra	Change organization Change site assignment
• • • • • •	50-5300-50-00-40 20-06-20-11-05-94 80-06-46-47-70-51	Security Gateway Switch Switch	NSG50 XS1930-12HP GS1350-6HP	01791 07100058 00069011050 <i>1</i> 01091 11000096	50-53-60-50-00-48 20-06-29-11-05-94 ВО-ОБ-45-47-70-51	2022-12-20 2023-01-10 2022-12-20	2023-07-22 2025-01-10 2023-07-05	Nebula Pre Nebula Pre Nebula Pre	Change organization Change site assignment Remove from organization
• - - - - - -	E0.E0.80.50.00.40 20.06:20.11.05:04 80.05:46:47:70:51 XGS220-30HP_Test	Security Gateway Switch Switch Switch	NSG50 XS1930-12HP GS1350-6HP XGS2220-30HP	00000000000000000000000000000000000000	50-50-60-50-00-48 20-06-20-11-05-84 ВС-05-46-47-70-51 ВЯ-50-А3-55-55-1А	2022-12-20 2023-01-10 2022-12-20 2023-01-05	2023-07-22 2025-01-10 2023-07-05 2023-06-23	Nebula Pro Nebula Pro Nebula Pro Nebula Pro	Change organization Change site assignment Remove from organization Assign license

5 Select the License assignment mode to have NCC filter licenses that can be assigned.

License assignment mo	de	Assign min. period 🕶
Expiration date Se	electe	Assign min. period
NCC 2024-07-22 -> 2026-07-23 N	ebulc	🔿 Assign all
		Target expiration date
		2023-01-31
		O Custom assignment
		Apply

- Assign min. period one month license packs for your Nebula Device will be picked and displayed.
- Assign all all licenses that can be assigned are displayed.
- Target expiration date all licenses that meet the expiry criteria you set and can be assigned are displayed.
- Custom assignment any change in value to Assign min period and Assign all licenses above will become a Custom assignment and are displayed.

6 Click Select # of license. In the pop-up window, confirm or edit the value appearing beside the license type based on the criteria set in License assignment mode. Click Select to confirm. Then click Finish.

Assign licenses			$\Box \times$
Assign licenses	Assign licenses You are currently assigning unused licenses which can be assigned to any matching Nebula Professional Pack License, 2YR	Nebula Professional Pack License, 2YR 1 Reset Select tula Professional Pack License, 2YR *1 Select # of license	
		Cancel	Finish

3.7.5 Transfer a License to a Nebula Device in a New Organization

To transfer a license(s) to a Nebula Device in the new organization, do the following:

- 1 Perform the steps mentioned in Assign a License to a Nebula Device in the New Organization.
- 2 Click Organization-wide > License & inventory > Device tab.
- **3** Select the devices with the license to be transferred.
- 4 Click Actions and select Transfer License.

Overview	Devices	Licenses	Tria	Il Cho	inge log	Purchase History		
1	Access Point		3 Switch		Security App	bliance	O Mobile Router	
Actions - In use Unuse	ed Both QS	earch_	• 1	selected in 5 devic	es.		+ Add	🕒 Ехр
Change organization								
Change organization	evice type	Model	Serial number	MAC address	Claim date	License expiration date	License info	Actions
Change organization Change site assignment	evice type	Model NAP102	Serial number S162Z25100249	MAC address 60:31:97.84:E1:84	Claim date 2022-12-20	License expiration date	License info Nebula Professional Pack	Actions Actions
Change organization Change site assignment Remove from organization	evice type pocess Point ecurity Gateway	Model NAP102 NSG50	Serial number S162Z25100249 S172L37100056	MAC address 60:31:97:84:E1:84 5C:E2:8C:5C:00:48	Claim date 2022-12-20 2022-12-20	License expiration date 2024-07-22 2023-07-22	License info Nebula Professional Pack Nebula Professional Pack	Actions Actions Actions
Change organization Change site assignment Remove from organization Assign license	evice type access Point ecurity Gateway witch	Model NAP102 NSG50 XS1930-12HP	Serial number S162Z25100249 S172L37100056 200629110584	MAC address 60:31:97.84:E1:84 5C:E2:8C:5C:00:48 20:06:29:11:05:84	Claim date 2022-12-20 2022-12-20 2023-01-10	License expiration date 2024-07-22 2023-07-22 2025-01-10	License info Nebula Professional Pack Nebula Professional Pack Nebula Professional Pack	Actions Actions Actions Actions
Change organization Change site assignment Remove from organization Assign license Undo assign	avice type ccess Point acurity Gateway witch	Model NAP102 NSG50 XS1930-12HP GS1350-6HP	Serial number S162225100249 S172L37100056 200629110584 S192L11090036	MAC address 60:31:97.84:E1:84 5C:E2:8C:5C:00:48 20:06:29:11:05:84 BC:CF:4F:477D:F1	Claim date 2022-12-20 2022-12-20 2023-01-10 2022-12-20	License expiration date 2024-07-22 2023-07-22 2025-01-10 2023-07-05	License info Nebula Professional Pack Nebula Professional Pack Nebula Professional Pack Nebula Professional Pack	Actions Actions Actions Actions Actions

5 The License transfer window appears. Click Search to set the filter to select the licenses.

License transfe	r							×
<u>License transfer</u>	License transfer Transfer add-on license(s) fr Select target device • ✓ License Key ✓ LiC-NPRO-ZZ2Y00F2 <	rom device(s) to othe Q Search License state Any Service Any Associated device Any	er device.	1 selected in (al Pack License, 2YR	licenses. License states Active	Associated device	Target device 🕚 Not set yet	Select •
		Days remaining Any	▼ Clear filter	_			Cancel	Finish

6 Click Select target device to transfer all licenses to one Nebula Device by selecting the same/different Organization and target Device. Then click OK. Or select the devices individually.

License transfer						×
<u>License transfer</u>	License transfer Transfer add-on license(s) from device(s Select target device -	to other device.	1) licenses.			
	Organization:	се	License states	Associated device	Target device 🖒	
	Select organization	la Professional Pack License, 2YR	Active	60-31-07-97-F1-97	Not set yet	Select
	Device:					۱.
		Ŧ				
	Reset					
					Cancel	

3.8 Reset the Nebula Password

If you forget your Nebula portal login password and need to reset it, do the following:

1 In the Nebula portal Sign In page, click Forgot Password.



ZYXEL NETWORKS	
	Sign In
	Don't have an account? Create account
	samual vis®zvval com tor
	Password (************************************
•	Forgot Password
	or
	Try Demo
	I have signed up but haven't activated my account. Resend Activation Email
	Legal Notice Terms of Use Privacy Coakie Settings
	2.50.8 Copyright © 2022 Zyxei and/or its affiliates. All Rights Reserved.

2 Enter your myZyxel account's email address, and then click Send.

ZYXEL Networks	
	Forgot Password No worries! Enter your email address, and we will send you a password reset email shortly.
	Email Address Email
	Send Remember your password now? Sign In Don't have an account? Create account

A reset password email has been sent notification appears.



3 Click the link in the **Request Password Reset to Login Zyxel** email.

ြို့ Reply ြို့ Reply All ြို့ Forward ပြီ IM
Represent Parameter Parameter Login Zingl
To Camiled Viu jim III B
This email notification was generated by Zyxel. Do not reply to this email.
52111 4146777 7122282222 2472218 511 2472181 2672146717 22788 2672146717 22787 2672146717 22787 2672146717 22787 2672146717 22787 2672146717 22878 2672146718 11 2672146718 11 2672146718 11 2672146718 11 2146718 11 11 <
Dear Customer,
You have recently requested to reset your password for your Zyxel account. Click the following link to set your new
password.
https://accounts.myzyxel.com/users/password/edit#KLZwWLKLFyKpPz6DsxzL
Need Help?
<u>Contact</u> us and we will be happy to help.
Sincerely,
If you wish to stay updated on the latest news, insights and special offers, please click here to receive our newsletter.

4 The following screen appears. Click Reset Password.



5 Enter the New Password. Use a minimum of 8 characters, including 0–9 a–z A–Z `~!@#\$%&*(_+-={} | [];"'./ <> ?). Then click Continue.

ZYXEL	
	Reset Password
	New Password Password Minimum 8 characters required, including (A-2)/(a-2), (0-9), and pencelebrbetic characters (a L § 1, 1, 1).
	Continue
	Remember your password now? Sign In
	Don't have an account? Create account

6 You will be transferred to the myZyxel portal. Click Update password.


3.9 Change an Organization and/or Site Name

To change your organization name or site name, do the following:

Organization Name

1 Go to Organization-wide > Organization-wide manage > Organization settings.

S	ettings		
	Organization information		
	Name:	Test_July	× *
	Country:	Taiwan	•

2 Enter a new descriptive name, 1 – 64 characters including 0–9 a–z A–Z `~!@#\$%&*(_+-={} | [];"'./<> ?) in Name.

Note: NCC does not allow duplicate organization name.

Note: Changing the organization name will not affect the Nebula Devices configuration in NCC.

3 Then, click **Save** at the bottom of the screen.

Site Name

1 Go to Site-wide > Configure > Site settings.

+8.0) 💌
×

2 Enter a descriptive name, 1 – 64 characters including 0–9 a–z A–Z `~!@#\$%&*(_+-={} | [];"'./<> ?) in Site name.

Note: NCC does not allow duplicate site name.

Note: Changing the site name will not affect the Nebula Devices configuration in NCC.

3 Then, click Save at the bottom of the screen.

3.10 Maintain Firmware

This section shows you how to update and maintain a Nebula Device's firmware.

1 Go to the Site-wide > Configure > Firmware management > Overview screen. Under Settings, you can set different times to upgrade firmware for your Access Points, Switches, Firewalls, Security Gateways, and Mobile Routers in your site. Select the day and time of the week when NCC will detect if any new firmware is available. NCC will send out a reminder email to the administrator for the available updates. If the administrator does NOT perform the update, after the set period of time is over, NCC will automatically upgrade the firmware for the Nebula Devices in the site. Or select Upgrade now to upgrade immediately.

	Devices	
Access Point	Upgrade available Newer firmware is available that may contain security fixes, new features, and performance improvements. We recommend that you upgrade to the stable or latest firmware version.	What is this?
	Settings Upgrade policy Auto upgrade at Monday O2:00 UTC+8.0 Upgrade at 2023-01-11 Th:00 UTC+8.0 Upgrade now	
	Ignore upgrade Firmware type Stable	
Switch	Upgrade available There's newer firmware available but not your preferred firmware type settings or per device scheduled settings. No further actions required. Settings	What is this?
	Upgrade policy Auto upgrade at Monday O2:00 Upgrade policy Upgrade at 2023-01-11 Th:00 UTC+8.0 Upgrade now	
	O Ignore upgrade	
Security Gatewo	Ignore upgrade Firmware type Stable -	What is this?

2 You can set different times to upgrade firmware for your Nebula Devices to overwrite the site-wide Settings by going to the Site-wide > Configure > Firmware management > Devices screen. Or select Upgrade now to upgrade immediately.

Overview	Dev	ices											
tatus	Device type	Tag		Model	Current version	Firmware st	tatus	Firmwar	e type	Availab	oility I	Locked	d
Any	Any	▼ Any	•	Any 👻	Any 👻	Any	•	Any	•	Any	•	Any	
Status	Device type Switch	Model	MAC address	S/N	Current version	2019	Firmware	status	Availability	ailable	Firmware type	9	Upgrade
Status	Device type Switch	Model NSW100-10P	MAC address	S/N C:BA S172L13000021	Current version V3.00(ABGO.2) 11/19,	/2019	Firmware Good 🚺	status	Availability Upgrade av	vailable	Firmware type Stable	e	Upgrade No
Status	Device type Switch Access point	Model NSW100-10P WAX510D	MAC address <u>B8:EC:A3:28:4</u> <u>D8:EC:E5:78:E</u>	S/N EC:BA S172L13000021 C:BE S212L40102451	Current version V3.00(ABGO.2) 11/19, V6.00(ABTF.0)IT_202;	/2019 21019121600	Firmware Good 1 Custom	status	Availability Upgrade av Upgrade av	vailable vailable	Firmware type Stable General Availa	ability	Upgrade No No
Status	Device type Switch Access point Access point	Model NSW100-10P WAX510D NWA50AX	MAC address <u>B8:EC:A3:2B:4</u> <u>D8:EC:E5:78:E</u> <u>B8:EC:A3:DD1</u>	S/N CEE \$172L13000021 CEE \$212L40102451 19:10 \$211842002072	Current version V3.00(ABGO.2) 11/19, V6.00(ABTF.0)IT_2022 V1.00(ABTW.0)	<mark>/2019</mark> 21019121600	Firmware Good () Custom () Warning	status D	Availability Upgrade av Upgrade av	vailable vailable vailable	Firmware type Stable General Availa General Availa	a ability ability	Upgrade No No No
	Device type Switch Access point Access point Access point	NSW100-10P WAX510D NWA50AX WAX650S	MAC address BSEC:A32B:4 DS:EC:E578:E BSEC:A3:DD1 BC:CF:4F:56E	S/N CEE \$172L13000021 CEE \$212L40102451 191C \$211842002072 3D:6D \$192L29290035	Current version V3.00(ABGO.2) 11/19, V6.00(ABTF.0) T_2022 V1.00(ABYW.0) V6.50(ABRM.0)b5	/2019 21019121600	Firmware Good 1 Custom 1 Warning Custom 1	status D D	Availability Upgrade av Upgrade av Upgrade av	vailable vailable vailable vailable	Firmware type Stable General Availa General Availa Beta	ability ability	Upgrade No No No
Status Image: Status	Device type Switch Access point Access point Access point Access point	Nodel NSW100-10P WAX510D NWA50AX WAX650S NWA50AX PRO	MAC address B3:EC:A3:2B:4 D3:EC:E5:78:E B3:EC:A3:DD1 B3:EC:F4F:56:E FC:22:F4:91:EF	S/N LC:BA \$172L13000021 C:BE \$212L40102451 191C \$211842002072 3D:6D \$192L29290035 E62 \$220Y51018102	Current version V3.00(ABGO.2) V6.00(ABTF.0) T.00(ABYW.0) V6.50(ABRM.0) V6.50(ACGE.0)	/2019 21019121600	Firmware Good () Custom () Warning Custom ()	status D D D D D	Availability Upgrade av Upgrade av Upgrade av Upgrade av	vailable vailable vailable vailable vailable	Firmware type Stable General Availa General Availa Beta General Availa	ability ability ability	Upgrade No No No No
Status	Device type Switch Access point Access point Access point Access point Access point	Nodel NSW100-10P WAX510D NWA50AX WAX650S NWA50AX PRO NWA90AX PRO	MAC address B8EC:A32B34 D8EC:E578E B8EC:A3:DD1 BC:CF4F:56E FC:22F4:91EF FC:22F4:91EF	S/N 10:EA \$172_1300.0021 CEE \$212_40102451 19:10 \$21842002072 3D:6D \$192_22920035 522 \$220Y51018102 EDD \$220Y51018132	Current version V3.00(ABGO.2) 11/19; V6.00(ABTF.0)T_2022 V1.00(ABYW.0) V6.50(ABRM.0)b5 V6.50(ACGE.0)b6	/2019 21019121600	Firmware Good 1 Custom 1 Custom 1 Custom 1 Custom 1	status D D D D D D	Availability Upgrade av Upgrade av Upgrade av Upgrade av Upgrade av	vailable vailable vailable vailable vailable	Firmware type Stable General Availa General Availa General Availa General Availa	ability ability ability ability	Upgrade No No No No No

- 3 If you do not want to upgrade the firmware immediately, you can click +Schedule Upgrade to create a schedule for your Nebula Device.
 - Select Follow device type settings to upgrade the Nebula Device according to the site-wide schedule configured for all Nebula Devices in the site.
 - Select Auto upgrade at every Week/Month on Sunday–Saturday at hh:mm to set up a routine schedule for upgrades.
 - Select **Upgrade at** to set up a specific date and time for a one time upgrade. This option can be enabled only when the selected Nebula Devices have a new firmware available.

Note: Due to network bandwidth and number of Nebula Devices per site, not all Nebula Devices may get the firmware upgrade on the specified date/time.

- Select Upgrade now to immediately install the firmware. Then select the Firmware type (Stable or Latest (default)).
- Note: When a firmware is officially released by Zyxel, it is the Latest firmware. For example, V6 is the Latest firmware. When the next firmware, V7, is released by Zyxel, V7 becomes the Latest firmware, and V6 will be classified as General Availability. Your Nebula Device firmware can be upgraded to V7 to use the new features. Zyxel will select a previous version (for example, V3) as a Stable release if no major issues have been reported by users.
- Note: The **Upgrade at** and **Upgrade now** options can be enabled only when the selected Nebula Devices have a new firmware available.
- 4 Click Add to save the settings.

Schedule up	grade					×
Upgrade policy	Follow devic Selected devic	ce type settings What is this? e(s) will be updated to site-wid	2 de per device's type settinț	gs. Device(s) will remove lo	ocked status and clear recurrent schedule	э.
	 Auto upgra Upgrade at Upgrade no 	de at every Week on N 2022-10-11	tonday ▼ at 02:00 ▼	UTC+8.0		
Below device(s) wi	II be upgraded as re	equired time.				
Device type	Model	MAC address	s/N	Current version	Schedule upgrade version	
Access point	WAX510D	D0-E0-E5-70-E0-40	CO101 40100450	V6.40(ABTF.4)	N/A	
					Cancel	Add

3.11 Assign an Administrator to Manage a Nebula Device

This section shows you how to assign an administrator to manage your Nebula Device.

1 Go to the Organization-wide > Administrators screen. Click +Add.

	t 🗍 Delete 🔍 Se	arch administrators	• 1	administrators			🕒 Import 🛛 + Ad
Name	Email address	Merged privilege 👔	Privilege	Account status	Last access time (UTC)	Create date (UTC)	Status change d [
S Yu	samualuu @zyxel.com.tw	Owner	Owner	ок	2021-10-05 08:21:16	2021-07-12 06:44:24	2021-07-12 06:44:2

2 Enter the Name and Email of a myZyxel account. Assign the Organization access (Full, Read-Only, None). See Table 202 on page 581 for information on organization privileges.

If you select **Full** for **Organization access**, select **Delegate owner's authority** to grant owner privileges to the new administrator except deleting/transferring organization ownership. Otherwise, do not select this option.

Select Yes if you wish to Activate the account administrator. Alternatively, select No if you wish to create an account administrator, but activate at a later time. The click Create admin.

Create adminis	trator		×
Name:	Jon		*
Email:	jon@zyxel.com		*
Organization access:	Full		•
Activate:	Yes		•
		Close Create	e admin

3 The Account status field will show Unverified. Click Save.

	jout 📋 Delete 🔍	Search administrators	2 administrators			mer 🕒 Import +	- Ado
Name	Email address	Merged privilege 👔	Privilege	Account status	Last access time (UTC)	Create date (UTC)	
SYu	syu@zyxel.com.tw	Owner	Owner	OK	2021-10-05 08:21:16	2021-07-12 06:44:24	20
Jon	jon@zyxel.com.tw	Organization (Full)	Organization (Delegated)	Unverified	Never	-	20

The Account status field will show OK after saving. The new administrator will receive an email notification.

	logout 📋 Delete 🔍	Search administrators	2 administrators			wner 📴 Import 🕂	- Ada
Name	Email address	Merged privilege ¦	Privilege	Account status	Last access time (UTC)	Create date (UTC)	E
SYu	syu@zyxel.com.tw	Owner	Owner	ок	2021-10-05 08:21:16	2021-07-12 06:44:24	202
Jon	jon@zyxel.com.tw	Organization (Full)	Organization (Delegated)		2021-10-01 02:14:07	2021-10-05 09:16:15	202
				-			

3.12 Transfer the Ownership of the Organization

This section shows you how to transfer an organization's ownership, which includes transfer ownership of the Nebula Devices.

Note: Only the owner can transfer ownership of an organization to another administrator. See Section 3.11 on page 76 if you want to transfer management of your Nebula Devices only.

1 The new owner must be an administrator in the same organization. Go to the Organization-wide > Administrators screen. Click Change owner.

	Delete Q Search administ	trators • 6 admin	istrators		🙎 Change	owner 🕒 Import 🕂 A
Email address	Merged privilege 📋	Privilege	Account status	Last access time (UTC)	Create date (UTC)	Status change date (UTC)
sel.yu@zyxel.com.tw	Owner	Owner	OK	2022-11-02 01:04:16	2021-07-12 06:44:24	2021-07-12 06:44:24
j@zyxel.com.tw	Organization (Full)	Organization (Delegated)	OK	2022-10-21 01:28:43	2021-10-06 02:10:50	2021-10-06 02:45:23
zhen@zyxel.com.tw	Organization (Full)by MSP		OK	2022-06-07 01:42:02	2022-01-10 05:09:15	2022-01-10 05:09:15
wuang@zyxel.com.tw	Organization (Full)	Organization (Full)	OK	2022-11-01 06:30:04	2022-03-16 02:08:48	2022-03-16 02:08:48
hen@zyxel.com.tw	Organization (Full)	Organization (Full)	OK	2022-07-11 03:33:51	2022-03-16 03:26:09	2022-03-16 03:26:09
ckuo@zyxel.com.tw	Organization (Full)	Organization (Full)	OK	2022-11-01 06:39:09	2022-10-21 01:51:44	2022-10-21 01:51:44

2 Select the new owner from the other administrators in this organization from the drop-down menu. Select the checkbox to continue, and click **Yes** to confirm transfer of ownership.



The new owner will be notified by email and must accept ownership of the organization.

3.13 Manage a Configuration Template

This section shows you how to use a configuration template to manage sites for your organization. Create a site and then bind a site to a template. You may enable the local override function if you want to configure some specific settings directly in a site after a site is bound to a template.

Note: This feature is available to an organization administrator with full privileges only (see Table 202 on page 581 for details on organization privileges).

- 1 Create and Bind a Template Site/Setting
- 2 Duplicate and Import a Template Setting to a Site
- 3 Enable the Override Site-wide Configuration (Local Override) Feature

3.13.1 Create and Bind a Template Site/Setting

1 Go to the Organization-wide > Organization-wide manage > Configuration templates screen. Click +Create.

Conf	figu	ration templates			
	+ c	reate — Delete	Q Search	▼ ③ Templat	te
		Name	Description	# bound sites	Bound sites
		Site-wide general		1	Hsinchu
		SSID Template1	Zyxel	2	Taipei •••
		Switch Template1		1	Site01
					Save or Cancel

2 The following screen appears. Enter a **Template name** and **Template description** for the template site or setting you want to create.

To create a new configuration template, select **Create new configuration template**. To import an existing template from a site or template, select **Import settings from**.



Create a new template X						
Template name: Template description:	SSID Template 2 Zyxel	×	*			
Import settings from:		-				
Create new configuration	2					
	Sites					
You could also bind sites durin	Hsinchu					
Target sites:	Kaohsiung					
	Site01					
	Taipei		Create			
1	Templates					
	Site-wide general		•			

Note: Under Import settings from, select a site from Sites to copy a site's settings. Under Import setting from, select a template from Templates to copy a site's site-wide general setting, an Access Point's SSIDs setting or a Switch's port setting.

3 Select a site from the Target sites drop-down list box to bind the template to a site. Click Create and then click Save to save the changes.

Create a new template					
Template name: Template description: Import settings from: Create new configuration ten	SSID Template2 Zyxel	× * ×			
You could also bind sites during c Target sites:	reate template: Hsinchu Kaohsiung				
	Site01 Taipei	Create			

If you skip this step, you can bind a template to a site later. Go to the **Organization-wide** > **Organization-wide** manage > **Configuration templates** screen. Select the template you want to use and then click the row with the template that you want to bind to a site.



Configu	uration templates			
+ 0	Create — Delete	Q Search	• 1 selected	n 5 Template
	Name	Description	# bound sites	Bound sites
	Site-wide general		1	Hsinchu
	SSID Template 2	Zyxel	0	
	SSID Template1	Zyxel	2	Taipei
	SSID Template3		0	
	Switch Template1		1	Site01
				ave or Cancel

4 The following screen appears. Click **Bind additional site** to select the site you want to bind the template to.

Configuration templates			
<u>Configuration template list</u> / <u>SSID Template2</u> 1 site are bound to this configuration template.			
Bind additional site Unbind Revert to templa	e setting Q Search	• 1) Site	
	Tags	Device	Local Override
ZyNet TW		7	AP SWITCH SITEWID

5 The following screen appears. Click the Target sites drop-down list box.

Select sites to follow "SSID Template 2"		
Target sites:		
	Hsinchu	
	Kaohsiung	
	Site01	
	Taipei	

6 Select a site from the Target sites drop-down box list and then click Bind.

Select sites to follow "S	×	
Target sites:	Kaohsiung 😫	× •
		Close Bind

7 Click Save to save the changes.

onfigure	ation template list / St	SID Template2			
sites are Bind ada	bound to this configur	ation template. Revert to template setting	Q Search	1 selected in 2 Site	
	Name		Tags	Device	Local Override
	ZyNet TW			7	AP SWITCH SITEW
	testSite-02-23			0	
				Save pr Cancel	
			(Please allow 1-2	2 minutes for changes to take effect.)	

8 A configuration template is created as shown in the Organization-wide > Organization-wide manage > Configuration templates screen.

+ Create Delete Q. Search • 5 Template Name Description # bound sites Bound sites Site-wide general 0 SSID Template 2 Zyxel 1 Kaohsiung SSID Template1 Zyxel 0 SSID Template3 0 Switch Template1 0
Name Description # bound sites Bound sites Site-wide general 0 0 SSID Template 2 Zyxel 1 Kaohsiung SSID Template1 Zyxel 0 SSID Template3 0 0 Switch Template1 0
Site-wide general 0 SSID Template 2 Zyxel 1 SSID Template1 Zyxel 0 SSID Template3 0 Switch Template1 0
SSID Template 2 Zyxel 1 Kaohsiung SSID Template1 Zyxel 0 SSID Template3 0 Switch Template1 0
SSID Template1 Zyxel 0 SSID Template3 0 Switch Template1 0
SSID Template3 0 switch Template1 0
Switch Template1 0

9 To release a site from using a configuration template, select a site and then click **Unbind** to unbind the site. The site which is unbound from the template still retains the settings applied from the template. The following screen appears. Click **Confirm** to confirm the changes.



10 Click Save to save the changes.



Configuration template list / SSID Te sites are bound to this configuration	mplate2 a template.		
Bind additional site Unbind Rev	ert to template setting Q Search	1 selected in 2 Site	
Name	Tags	Device	Local Override
ZyNet TW		7	AP SWITCH SITE
Z testSite-02-23		7	

3.13.2 Duplicate and Import a Template Setting to a Site

This section shows you how to duplicate and then import the following template settings to a site:

- The site-wide general setting includes the device configuration, SNMP and captive portal reauthentication.
- An Access Point's SSID setting.
- A Switch's port setting.

The site-wide general setting

1 Select a bound site from the Site drop-down list box.

Group: TW Test • Org	anization: Test July	•	itite: ZyNet TW ▼	•		Q	\odot	Ļ	තු	••••	s
Configuration management			testSite-02-23 TW Temp								
Synchronization			ZyNet TW								
	Settings:	Site-wo	e ç Template								
	From source site:	TW Tem	D Site-wide general								
	To site(s):	Select	SSID Template2	÷	•						
	What will be synchr	onized?			Sync						

2 Go to the Organization-wide > Organization-wide manage > Configuration management screen. Under Synchronization, select the Site-wide settings in Settings to copy a site's general setting to another site.

Configuration management			
Synchronization			
	Settings:	Site-wide settings	
	From source site:	Site-wide settings SSIDs	
	To site(s):	Kaohsiung	
	What will be syn	sync Sync	
Switch settings clone			
	From source device:	D0-E0-A0-AE-EA-14	
	To device(s):	•	
		Include uplink port settings	
	What will be clo	ned? Clone	

3 From the From source site drop-down list box, select the site you want to copy the Site-wide settings from.

Configuration management		
Synchronization		
	Settings:	Site-wide settings
	From source site:	Hsinchu
	To site(s):	٩
	<u>What will be syn</u>	Hsinchu Kaohsiung
		Site01
Switch settings clone		Taipei
	From source device:	
	To device(s):	•
		Include uplink port settings
	<u>What will be clor</u>	Clone
Switch settings clone	From source device: To device(s): <u>What will be clor</u>	Taipei

4 From the To site(s) drop-down list box, select the site you want to import the Site-wide settings to. Click Sync to save the changes.

Configuration management			
Synchronization			
	Settings:	Site-wide settings 🔹	
	From source site:	FLEX100W_0630	•
	To site(s):	Select some sites	•
	<u>What will be s</u>	which ronized?	Sync
Switch settings clone			
	From source device:		•
	To device(s):		•
		Include uplink port settings	
	<u>What will be a</u>	cloned?	Clone

An Access Point's SSID Setting

1 Go to Organization-wide > Organization-wide manage > Configuration management screen. Under Synchronization, select SSIDs to copy a site's SSIDs settings to another site. The duplicated SSIDs include the authentication and captive portal settings.

Configuration management			
Synchronization			
	Settings:	SSIDs 🔻	
	From source site:	Site-wide settings	-
		SSIDs	•
	To site(s):	Kaohsiung	•
	<u>What will be syn</u>	chronized?	Sync
Switch settings clone			
	From source	RR-EC-AR-AE-EA-14	•
	To device(s):		•
		Include uplink port settings	
	What will be clo	ned?	Clone

2 From the From source site drop-down list box, select the site you want to copy the SSIDs from.

Configuration management		
oynom on 2 a con	Settings:	SSIDs
	From source site:	Hsinchu
	To site(s):	٩
		Hsinchu
	<u>What will be syn</u>	Kaohsiung
		Site01
Switch settings clone		Idibei
	From source device:	D0-E0-A0-AE-EA-14
	To device(s):	•
		Include uplink port settings
	What will be clor	Clone

3 From the To site(s) drop-down list box, select the site you want to import the SSIDs to. Click Sync to save the changes.

Configuration management		
Synchronization		
	Settings:	SSIDs 👻
	From source site:	FLEX100W_0630
	To site(s):	Select some sites
	What will be s	synchronized?
Switch settings clone		
	From source device:	
	To device(s):	
		Include uplink port settings
	<u>What will be a</u>	cloned?

A Switch's Port Setting

1 Go to the Organization-wide > Organization-wide manage > Configuration management screen. Under Switch settings clone, select the Nebula Device's MAC address from the From source device drop-down list box. The cloned switch setting includes the port setting, IGMP advanced settings and STP bridge priority.

Configuration management			
Synchronization			
	Settings:	SSIDs 🔹	
	From source site:	Hsinchu	-
	To site(s):	Kaohsiung	•
	What will be sync	chronized?	Sync
Switch settings clone			
	From source device:	D0-E0-A0-AE-EA-14	-)
	To device(s):		•
		Include uplink port settings	
	What will be clon	ed?	Clone

2 From the To device(s) drop-down list box, select the Nebula Device's MAC address you want to import the Switch setting to. Click Clone to save the changes.

Configuration management		
Synchronization		
	Settings:	SSIDs 👻
	From source site:	Hsinchu
	To site(s):	Kaohsiung 👻
	What will be sync	shronized? Sync
Switch settings clone		
	From source device:	R9-E∩-A3-AE-EA1/ ▼
	To device(s):	
	What will be clon	Include uplink port settings

3.13.3 Enable the Override Site-wide Configuration (Local Override) Feature

A configuration template is a list of common settings that you can bind (apply) to a site. If you do not want to apply any new settings from the template to a site, just unbind that site. If you want to configure some specific settings directly in a site after the site is bound to a template, turn on the local override function. This feature is available to an organization administrator with full privileges only.

This section shows you how to enable the **Override site-wide configuration** feature to update site information. Select a bound site from the **Site** drop-down list box to edit the details of the selected site.

Group: TW Test 👻	Organization: Test July	•	Site	ZyNet TW 👻		•	Q	\odot	Ļ	තු	 S
Configuration management				testSite-02-23 TW Temp							
Synchronization				ZyNet TW							
	Settings:	Site-w	de g	ZyNet TW-2							
				Template							
	From source site:	TW Te	np	Site-wide general							
	To site(s):	Select		SSID Template2	*						
	What will be synchro	onized?				Sync	ĺ.				

1 Go to a page under Site-wide > Configure and then select the Override site-wide configuration box. The Configuration page of a bound site contains an Override site-wide configuration box.

	This site is bound to template <u>SSID Template2</u>	
		Override site-wide configu
Site information		
Site name	ZyNet TW ×	
Local time zone	Taiwan 🔹 Asia - Taipei (UTC +8.0) 🔹	
Site location	×	
	What is this?	
	This site location will apply to your new added device(s) as address on map automatically.	
Configuration template	This site uses the configuration of the template <u>SSID Template2</u> Unb	ind

2 The following screen appears. Click **Confirm** to continue.



3 In the Site-wide > Configure > Site settings screen, edit the Site information, Device configuration, Captive portal reauthentication, SNMP and Voucher settings on the following page. Click Save to save the changes.

	This site is bound to template SSID Template2	
		Verride site-wide configuration
Site information		
Site name	ZyNet TW 🛛 🗙 🔭	
Local time zone	Taiwan 👻 Asia - Taipei (UTC +8.0) 👻	
Site location	×	
	What is this?	
	This site location will apply to your new added device(s) as address on map automatically.	
Configuration template	This site uses the configuration of the template <u>SSID Template2</u> Unbind	

4 To verify the local override setting of a site, go to Organization-wide > Organization-wide manage > Configuration templates. The Local Override field may show that AP/SWITCH/GATEWAY/SITE-WIDE settings in the template do not apply to the site. A tag for AP, as shown in the following figure, indicates that Access Point settings have a local override and any further changes in the template's AP settings will not be synchronized to the site.



Configu	ration templa	te list / <u>Site Ter</u>	mplote1		
Bind ad	ditional site	Unbind Rev	ert to template setting	h • 1) selected in 1) Site	
Z	Name		Тадз	Device	Local Override
	<u>Hsinchu</u>			0	AP

5 If you decide to go back to the original template settings, clear the **Override site-wide configuration** box on any page under **Site-wide > Configuration**. The following screen appears. Click **Confirm** to continue.

Revert override	×
Configuration in this page will follow template Site Tem Please click confirm to continue	plate1.
Close	nfirm

Overwrite the Access Point / Switch Setting

1 Go to any page under Site-wide > Configure > Access points / Switch and then select the Override access point configuration box. Every Configuration page of a bound site contains an Override site-wide configuration box.

Note: If the local override configuration is enabled on one page, all configuration pages of the Nebula Devices in the selected site will be enabled.



SID: SSID1	· · · · · · · · · · · · · · · · · · ·
Network access	
Security options	Open
	Users can connect without entering a password
	User can connect without password. Enhanced open provides improved data encryption in open Wi-Fi networks.
	WPA Personal With WPA2
	Users must enter the password to associate
	MAC-bread Authentication with
	MAC-bused Audientication with Nebula cloud authentication
	Use MAC address as a username and password
	WPA Enterprise with WPA2 -
	Use 802.1X authentication that requires a unique username and password
	WPA Enterprise with Nebula cloud authentication 🔹
Sign-In method	O Disabled
	Users can access the network without any web authentication
	O Click-to-continue
	Users must view and agree the captive portal page then can access the network
	○ Voucher
	Users must enter a voucher code then can access the network
	Create and manage voucher passcode on the <u>Vouchers</u> page.
	○ Sign-on with Nebula cloud authentication 👻

2 This allows a specific type of Nebula Device setting override. The following screen appears. Click **Confirm** to continue.



3 In Site-wide > Configure > Access point > SSID settings, edit your SSIDs, authentication or captive portal settings on the following page. Click Save to save the changes.

SSID settings	-				Override access point configuration
Advanced mode: Beta	What is this?				
+ Add SSID network					
No.	i.		2		3
Name	SSID1	× *	SSID2	× *	SSID3
Enabled					
Programmable SSID Beta					
Tagging	Tag		Tag		Tag
	Enable SSID on APs with any of the specified to	gs	Enable SSID on APs with any of the specified t	ags	Enable SSID on APs with any of the specifie
Guest Network	•		0		0
SSID advanced settings		Edit		Edit	
WLAN security	Open		Open		Open
Sign-in method	Disable		Disable		Disable
Band mode	2.4 GHz S GHz 6 GHz		2.4 GHz 5 GHz 6 GHz		2.4 GHz 5 GHz 6 GHz
VLAN ID	1		1		1
Rate limiting	⊕unlimited Kb/s⊕unlimited Kb/s		⊕unlimited Kb/s⊕unlimited Kb/s		@unlimited Kb/s®unlimited Kb/s
Captive portal customization		Edit		Edit	
Theme	Modern		Modern		Modern
	4.				

In the Site-wide > Configuration > Switches > Switch settings screen, edit VLAN configuration, STP configuration, Quality of service, or Port mirroring settings on the following page. Click Save to save the changes.

Switch settings	Verride switch configuration
Auto configuration recovery Model list	eto
Auto configuration recovery 🕕	
VLAN configuration Management VLAN	1 * Before changing management VLAN, please check that uplink port enable management control and belongs to management VLAN member for avoiding disconnect with NCC. To configure management control port in <u>Switch ports.</u>
STP configuration Rapid spanning tree protocol (RSTP):	
STP bridge priority: 0	Switches Bridge priority Default 32768

4 To go back to the original template settings, clear the Override switch configuration box on any page under Site-wide > Configuration > Access points / Switches. The following screen appears. Click Confirm to continue.



3.14 Activate an MSP License

You must have an NCC account and an MSP license pack to activate an MSP license.

To activate an MSP pack, do the following:

1 Click the More icon (upper right) and select My devices & services.

		Q 💮
itł	8	My devices & services New
	\sim	Active sessions
	Ð	Recent logins

2 Select the Services tab.

er > My devices & services									
y devices & services									
ne list of all Nebula devices and se	ervices that have been owned by yo	ur account.							
Devices Services Purch	nase history								
Actions - Q. Search	• 1 Licenses					省 Pu	rchase MSP licenses	+ Register	Export -
		Start data	End date	Status	Actions	Registered date	Activated date		
🗹 License key	Service description	Sturt uute	Endude						
License key LIC-NMSP-2YR-20220623	Service description 0916 Nebula MSP Pack License; 2Y	R 2022-07-01	2024-07-01	Activated	Transfer license	2022-06-23	2022-07-01		
License key LiC-NMSP-2YR-20220623	Service description 0916 Nebula MSP Pack License; 2Y	R 2022-07-01	2024-07-01	Activated	Transfer license	2022-06-23	2022-07-01		

3 Select the MSP Pack license, click Actions, and select Activate. The MSP menus can now unlock the MSP branding, Admins & teams, Cross-org synchronization, and MSP alerts features (see Chapter 13 on page 570 for details on the MSP menus).

3.15 Configure CNP/CNP Plus Security Services

Different features are enabled depending on the type of trial license you purchased.

If you activate the CNP trial license, only the IP reputation filter is enabled. If you activate the CNP Plus trial license, IP reputation filter and application visibility & optimization are enabled.

3.15.1 Threat Protection

An IP address with a bad reputation is an IP address associated with suspicious activities, such as spam, virus, and phishing. These are stored in a database. IP reputation checks the reputation of an IPv4 (only) IP address from the database. When there are packets coming from an IPv4 address with bad reputation, you can set the Nebula Device to respond by blocking these packets. You can change the response action set in NCC. You can also configure an exempt list to allow packets from specific IP addresses regardless of their content rating.

Both the CNP/CNP Plus licenses enable the IP reputation filter feature. To configure IP reputation filter, do the following:

- 1 Go to Site-wide > Configure > Access points > Security service.
- 2 Refer to Section 5.3.6 on page 268 for details on how to configure the Threat Protection fields.

Threat Protection Beta Model list	
Enabled	DNS Threat/IP Reputation Filter screens out unsafe phishing sites or botnets to provide users a trustworthy wireless service.
Block log	
Click to proceed	Allow users to browse unsafe sites. Proceed to unsafe website at user own risk:
Denied access message	This high risk page is blocked by Zyxel Connect & Protect service due it may contain maliclous $ imes$
Redirect external URL	URL: ×
	To use custom captive portal page, please download the zip file and edit them.
	Download the customized captive portal page example.
Notification page	
Enable on	e-Nebula-FT99 👻
Access message	Zyxel AP proactively secure your network and establish a trustworthy Wireless LAN to protect $_{\rm V}$ $_{\rm X}$
Category list	Yor Proxy Image: Mobile Threats Anonymizers Phishing Molicious Downloads Denial of Service Scanners BotNets Web Attacks and Malicious Sites Exploits Spyware and Adware Keyloggers Spam URLs
IP Reputation exempt list	IP or CIDR X
DNS Threat exempt list	FQDN(support wildcard)
	×

3 Then click Save.

Go to Site-wide > Dashboard: Hit for Threat Protection by CNP Service to view the following:

- total number of times packets coming from an IPv4 address with a bad reputation occur, and
- the number of times connection attempts to an IPv4 address with a bad reputation occur.



3.15.2 Application Visibility & Optimization

Application visibility provides a way for a Nebula-managed Access Point to manage applications in WiFi network. It can detect the type of applications used by WiFi clients and how much bandwidth they use.

Application optimization is a way to limit the bandwidth usage of applications in the WiFi network. For example, applications that need real time traffic such as video streaming may use more resources. Use application optimization to limit the bandwidth used to stream video to prevent it from slowing down your WiFi network. Application optimization limits the applications bandwidth usage by their categories. You can manage and view the applications and their categories in **Site-wide > Applications usage > Application View by Access Point**.

You need to purchase the CNP Plus license to enable application visibility & optimization. To configure application visibility & optimization, do the following:

- 1 Go to Site-wide > Applications usage.
- 2 Make sure you are in Application View (--> Category View is displayed)
- 3 Select Application View by Access Point in the Applications field.
- 4 Hover the mouse pointer anywhere on an application row. Click the Limit icon to set its Bandwidth limit.



plications	Application view by Acces	ss Point Search		Last 24 hours 🔹 🖒	$2 \rightarrow Category View$
	2.82 GB	551.05 MB 367.37 MB			
	Application	183.68 MB	9.30 21.00 22.30 11.Feb 01.3	0 0300 0430 0600 0730	0800 1030 1200 1330 1500 1630 181
Q Search Status		41) applications	Bandwidth limit	Usaae	%Usage
Q Search Status	Application	41) applications Category Web Services	Bandwidth limit 5Mb (<u> Limit</u>) 4	Usage 1.32 GB	%Usage 46.80%
Q Search Status	Application Apple Microsoft SharePoint	(41) applications Category Web Services Microsoft Office	Bandwidth limit 5Mb () Limit 4 20Mb	Usage 1.32 GB 287.36 MB	%Usage 46.80% 9.96%
Q Search Status O	Application Apple Microsoft SharePoint Akamai	41) applications Category Web Services Microsoft Office Web Services	Bandwidth limit 5Mb (B Limit) 4 20Mb 5Mb	Usage 1.32 GB 287.36 MB 258.27 MB	%Usage 46.80% 9.96% 8.96%
Q Search Status O O O	Application Apple Microsoft SharePoint Akamai Apple iTunes	41) applications Category Web Services Microsoft Office Web Services Web Services	Bandwidth limit 5Mb (B Limit) 4 20Mb 5Mb 5Mb	Usage 1.32 GB 28736 MB 258.27 MB 163.67 MB	%Usage 46.80% 9.96% 8.96% 5.68%
Q, Search Status © © © ©	Application Apple Microsoft SharePoint Akamai Apple ITunes Microsoft Outlook	41) applications Category Web Services Microsoft Office Web Services Web Services Web Services Mail	Bandwidth limit 5Mb () Limit 4 20Mb 5Mb 5Mb 10Mb	Usage 132 GB 287.36 MB 258.27 MB 163.67 MB 94.71 MB	%Usage 46.80% 9.96% 8.96% 5.88% 3.28%
Q Search Status O O O O O O O	Application Apple Microsoft SharePoint Akamai Apple iTunes Microsoft Outlook Facebook		Bondwidth limit 5Mb () Limit 4 20Mb 5Mb 5Mb 10Mb 20Mb	Usage 132 GB 287.36 MB 258.27 MB 163.67 MB 94.71 MB 94.71 MB	%Usage 46.80% 9.96% 8.96% 5.68% 3.28% 2.71%
Q Search Status O O O O O O O O O O O O O	Application Apple Microsoft SharePoint Akamai Apple iTunes Microsoft Outlook Facebook Microsoft Office		Bandwidth limit 5Mb B Limit 4 20Mb 5Mb 5Mb 10Mb 20Mb 20Mb	Usage 1.32 GB 267.36 MB 258.27 MB 163.67 MB 94.71 MB 94.71 MB 78.06 MB 68.31 MB	%Ubage 46.80% 9.96% 8.96% 5.68% 3.28% 2.71% 2.87%
Q Search Status C C C C C C C C C C C C C C C C C C C	Application Apple Microsoft SharePoint Akamai Apple iTunes Microsoft Outlook Facebook Microsoft Office		Bandwidth limit 5Mb B Limit 4 20Mb 5Mb 5Mb 10Mb 20Mb 20Mb 20Mb 20Mb	Usage 1.32 GB 28736 MB 258.27 MB 258.27 MB 163.67 MB 94.71 MB 78.06 MB 68.31 MB 57.80 MB	%Ubage 46.80% 9.96% 8.96% 5.68% 3.28% 2.71% 2.37% 2.00%
Q Search Status C C C C C C C C C C C C C C C C C C C	Application Apple Microsoft SharePoint Akamai Apple ITunes Microsoft Outlook Facebook Microsoft Office QUIC HTTPS		Bandwidth limit 5Mb B Limit 4 20Mb 5Mb 5Mb 10Mb 20Mb 20Mb 20Mb 5Mb 5Mb	Usage 1.32 GB 28736 MB 258.27 MB 258.27 MB 163.67 MB 94.71 MB 94.71 MB 78.06 MB 68.31 MB 65.38 MB	%Usage 46.80% 9.96% 8.96% 5.68% 3.28% 2.71% 2.87% 2.00% 1.95%

5 Use the slider or enter the Traffic allowed in Mb/s (1 – 30 or Unlimited).

Web Services	×
Traffic	5 (Mb/s)(1 - 30)
Bandwidth limit applies on application category. The setting helps on smooth wireless experience by limiting the applications consuming large o network bandwidth.	mounts of
(Per client device traffic rate)	
	Cancel Ok

6 Then click Ok.

To monitor the application bandwidth usage, go to **Site-wide** > **Dashboard**: **Access points application usage** to view the top ten applications that use the most bandwidth in the site.

(AF	P Traffic					(24h)
		MPEG Tra	ns3 GB	Apple	8.2 MB	
		Facebook	37 MB	📕 Microsoft		
	3.1 GB	Spotify	25 MB	Microsoft	T6.1 MB	
		HTTPS	13.1 MB	Google	6 MB	
		Zyxel	13 MB	Microsoft	4.1 MB	

3.16 Delete an Organization

Only the Organization owner can delete an Organization. An Organization can be deleted only when it has no site(s), administrator(s), user(s), license(s), or Nebula Device(s) in the Organization.

To delete an Organization from the NCC, do the following:

3.16.1 Remove All Nebula Devices

- 1 Go to Organization-wide > License & inventory > Devices tab (1).
- 2 Click the check box (2) to select all Nebula Devices.
- **3** Click the **Actions** button (3).

ense	e & inventory								
(Overview] Devices	Licenses	Tria	al Cho	inge log	Purchase History		
	(2)	1 Access Point		3 Switch		1 Security App	bliance	O Mobile Router	
_		_							0.0
Actio	ions - In use Un	used Both Q S	earch_	• (5)	selected in (5) devic	es.		-+ Add	[] Expo
Actio	Device name	Device type	earch Model	Serial number	MAC address	es. Claim date	License expiration date	+ Add	Actions
Actio	Device name 60:31:97:84:E1:84	Device type Access Point	Model NAP102	• (5) Serial number \$162Z25100249	MAC address 60:31:97:84:E1:84	Claim date 2022-12-20	License expiration date	+ Add License info Nebula Professional Pack	Actions
Actio	In use Un Device name 60:31:97:84:E1:84 5C:E2:8C:5C:00:48	Device type Access Point Security Gateway	Model NAP102 NSG50	(5) Serial number S162Z25100249 S172L37100056	MAC address 60:3197:84:E1:84 50:E2:80:50:00:48	Claim date 2022-12-20 2022-12-20	License expiration date 2024-07-22 2023-07-22	License info Nebula Professional Pack Nebula Professional Pack	Actions Actions Actions
	In use Un Device name 60:31:97:84:E1:84 5C:E2:8C:5C:00:48 20:06:29:11:05:84	Device type Access Point Security Gateway Switch	Model NAP102 NSG50 XS1930-12HP	Serial number S162225100249 S172L37100056 200629110584	MAC address 60:31:97.84:E1:84 50:E2:80:50:00:48 20:06:29:11:05:84	Claim date 2022-12-20 2022-12-20 2023-01-10	License expiration date 2024-07-22 2023-07-22 2025-01-10	License info Nebula Professional Pack Nebula Professional Pack Nebula Professional Pack	Actions Actions Actions Actions
	In use Un Device name 60:3197.84:E1:84 5C:E2:8C:5C:00:48 20:06:29:11:05:84 BC:CF:4F:477D:F1 80:00:00:00:00:00:00:00:00:00:00:00:00:0	Used Both Q S Device type Access Point Security Gateway Switch Switch	Model NAP102 NSG50 XS1930-12HP GS1350-6HP	 Serial number S162Z25100249 S172L37100056 200629110584 S192L11090036 	MAC address 60:3197.84:E184 5C:E28C:5C:00:48 20:06:29:11:05:84 BC:CF:4F:47:7D:F1	Claim date 2022-12-20 2022-12-20 2023-01-10 2022-12-20	License expiration date 2024-07-22 2023-07-22 2025-01-10 2023-07-05	License info Nebula Professional Pack Nebula Professional Pack Nebula Professional Pack Nebula Professional Pack	Actions Actions Actions Actions Actions

- 4 Click Remove from organization.
- 5 Click the Yes button to confirm, or click the delete icon to remove each devices individually.

Remove device:	s		×
You are going to remove network services will be	e device(s) from organiz stopped.	ation. Those device(s	s) belong to site will be removed and
Device name	MAC address	Serial number	Site
£∩-91-07-01-E1-01	£∩-01-07-0 <i>1</i> -⊑1-0 <i>1</i>	e160705100040	2F_Office
50-50-00-50-00-40	50-50-00-50-00-40	e1701 07100056	2F_Office
<u> </u>	20-06-20-11-05-0 <i>1</i>	200620110507	Alex test PoE schedules
DO-OE-4E-47-7D-E1	DO-OE-8E-87-7D-E1	e1021 11000026	2F_Office
KUGUULUHP Test		60001 16000010	2F_Office
Do you want to continue	9?		
			Cancel Yes

3.16.2 Transfer All Licenses

See Section 3.7 on page 64 in this chapter for information on how to transfer licenses assigned to an organization and Nebula Device to another Nebula Device in a different organization.

3.16.3 Delete All Sites

- 1 Go to Organization-wide > Organization-wide manage > Organization portal > Sites tab (1).
- 2 Click the check box (2) to select all sites.
- 3 Click the **Delete** button (3) to remove all sites.



4 Click the **Delete sites** button to confirm.

Delete site confirmation	×
Deleting sites is irreversible; all configuration and client data associate with those sites will be	lost!
Close Delete s	ites

3.16.4 Delete All Administrators

- 1 Go to Organization-wide > Administrators (1).
- 2 Click the check box to select all administrators (2).
- 3 Click the **Delete** button (3).
- 4 Click the Save button (4) to confirm.



ctivatio	m 🔻 🕞 F	Force logout 🚺 🔒 Delete	Q Search administ	rators • (3) selected in (3	administrators	🖇 Change own	er 🖹 Import 🕂 Ad
	Name	Email address	Merged privilege	Privilege	Account status	Last access time (UTC)	Create date (UTC)	Status change date (
	Sam	sam@zyxel.com.tw	Owner	Owner	ОК	2022-02-07 00:56:07	2021-07-12 06:44:24	2021-07-12 06:44:24
2	John	john@zyxel.com.tw	Organization (Full)	Pending deletion	n <mark>OK</mark>	2022-01-26 06 15:41	2021-10-06 02 10 50	2021-10-06 02:45:23
2				Pending deletion	n <mark>OK</mark>			

3.16.5 Remove All Users

- 1 Go to Organization-wide > Organization-wide manage > Cloud authentication (1).
- 2 Select the User tab (2).
- 3 Click the check box to select all users (3).
- 4 Click the **Remove users** button (4).
- 5 Click the Save button (5) to confirm.

Cloud a User Auth	MAC [orization - 4	tion DPPSK Remove use	ərs VPN a	ICCESS -	VLAN attribute +	Q Se	arch users) selected in (2)Users	•			i Impo	ort + Add	🕒 Export •
3 🔽	Email	Userna	Descri	802.1X	VPN ac	Authori	Expire i	Login by	DPPSK 🗞 VLAN a	2FA St	Bypass	Authori	Create	Create
	nsbusvd3	nsbusvd3	svdtest	No	No	No		Email		Not Enroll	No		sdd9.rd@	2022-12-27 06.
	test@gm	test1234		No	No	No		Email		Not Enroll	No		sdd9.rd@	2022-12-27 06.
4														
					(5)		5	or Can	el					

3.16.6 Delete the Organization

- 1 Go to Organization-wide > Organization-wide manage > Organization settings (1).
- 2 Enter the Name of the organization you wish to remove (2).
- **3** Click the **Delete organization** button (3).

organization information	
Name:	Test_July × 2
Country:	Taiwan 👻
ecurity	
Idle Timeout (0 × * minutes of inactivity will logout users.
💎 Login IP ranges	Only allow access to this organization from IP addresses in the specified ranges.
	This computer is using IP address : 61222.86.26
	What do Lenter here?
Import certificate	
Delete this organization	You can delete this organization only if it has no sites, administrators, users, licenses, or devices registered in this inventory.
	Please check your setting as below: <u>sites</u> , <u>administrators</u> , <u>users</u> , <u>licenses/devices</u> of devices.

4 Click the **OK** button to confirm.

WARNING	×
WARNINGI Deleting an organization cannot be reversed! Are you sure you want to delete this orgo	inization?
Canc	

3.17 Remote Access VPN Setup

The following figure illustrates a secure VPN channel configured through Nebula. The VPN client (C) remotely accesses the office server (A) through the Nebula Device (S) in a typical work from home scenario.



To set up a remote access VPN on Nebula, do the following:

- Create a VPN user
- Enable the remote access VPN rule for IPSec VPN client
- Check the connection in Nebula.

The user needs to do the following:

- Set up the VPN using Zyxel's SecuExtender (only), a VPN client software
- Import the VPN configuration file
- Open the VPN tunnel
- Set up two Factor Authentication on a mobile device to bind the user account.

3.17.1 Create a VPN User

1 Go to the Site-wide > Configure > Cloud authentication screen. Click +Add to create a user.

Cloud User	Cloud authentication (Site :AE_Test) User MAC DPPSK													
						Q Sec	arch users		• 2 Users			🕒 Impo	ort + Add	🕒 Export 🕶
	Email	Userna	Descri	802.1X	VPN ac	Authori	Expire in	Login by	DPPSK 🔌 VLAN a	2FA St	Bypass	Authori	Create	Create
	nsbusvd3	nsbusvd3	svdtest	No	No	No		Email	••••••	Not Enroll	No		sdd9.rd@	2022-12-27 14:1
	test@gm	test1234		No	No	No	_	Email	••••••	Not Enroll	No		sdd9.rd@	2022-12-27 14:

2 Enter an Email, Username, generate or enter a Password (4 – 31 characters, including 0–9 a–z A–Z `~!@#\$%&*(_+-={} | [];"./<> ?). Click Allow to use Remote VPN access. Click Does not expire to set no time limit for this user account. Select Username or Email in Login by. Click to select Email account information to user. Then click Create user.

Create user		×
Account type:	USER	*
Email:	vpnuser@zyxel.com \times *	
Username:	vpnuser ×*	
Description:	×	
Password:	vfMMB4In * 🖒 Generate	
DPPSK:	Generate	
802.1X:	Allow to use WPA-Enterprise to access networ	'k
VPN Access:	Allow to use Remote VPN access	
Authorized:	Yes 💌	
Expires:	O Does not expire	
	◯ Expires in: × * minutes ▼	
Login by:	Username or Email 🔹	
VLAN assignment: Beta	×	
Two-Factor Auth.:	Bypass two-factor authentication.	
Email to user:	Email account information to user.	
	Close Print	Create user

3 Click Save.

					Q Sec	arch users		• 3 Users			🕒 Impo	rt + Add	Export
Email	Userna	Descri	802.1X	VPN ac	Authori	Expire in	Login by	DPPSK 🗞 VLAN a	2FA St	Bypass	Authori	Create	Create
nsbusvd3	nsbusvd3	svdtest	No	No	No		Email		Not Enroll	No		sdd9.rd@	2022-12-27
test@gm	test1234		No	No	No		Email	••••••	Not Enroll	No		sdd9.rd@	2022-12-27
vpnuser	vpnuser		No	Yes	Yes	Never	Usernam		Not Enroll	No	samuel.yu	samuel.yu	-

3.17.2 Enable the Remote Access VPN Rule for IPSec VPN Client

1 Go to the Site-wide > Configure > Firewall > Remote access VPN screen. Click IPSec VPN server to enable VPN. Enter the IP address range in Client VPN subnet. Select IKEv2 in IKE version. Click **Two-factor authentication with Captive Portal** to enable two-factor authentication with the Google authenticator app. The VPN client will be asked to provide a Google authenticator verification code, so must install the Google Authenticator app. Then click **Save**.

note access VPN		
WAN interface	Auto 👻	
Domain name	alpha-Bebe4d35.d2ns-nbl.com	
IPSec VPN server		2
Client VPN subnet	192.168.100.0/24 × *	
IKE version	IKEv2	
DNS name servers	Firewall •	
Upload bandwidth limit	× Mbps 0	
Policy	Default	
Authentication	Nebula Cloud Authentication	
Constitutendor IVEN VDN configuration provision	Two-factor authentication with Captive Portal 0	
declaration inclusion provident	vpnuser@zyzel.com e somuel.yu@zyzel.com.tw e Etitiend i molt.	
L2TP over IPSec VPN server		
	(Diassa allau 1.0 minutes far algement to tota offect)	

2 Click Send Email to give your VPN client the configuration instructions through email.

note access VPN			
WAN Interface	Auto	*	
Domain name:	alpha-6ebe4d35.d2ns-nbl.com		
IPSec VPN server			e batel
Client VPN subnet	192168.100.0/24	×	
IKE version	IKEV2	*	
DNS name servers	Firewall	*	
Upload bandwidth limit		× Mbps 0	
Policy	Default		
Authentication	Nebula Cloud Authentication	*	
	Two-factor authentication w	ith Captive Portal 0	
SecuExtender IKEv2 VPN configuration provision	vpnuser@zyxet.com @ sam	nuelyu@xyxel.com.tw 🍙	
L2TP over IPSec VPN server			
		Dave or Cancel	
	(Please all	ow 1-2 minutes for changes to take effect.)	

NCC User's Guide

3.17.3 VPN Setup by the VPN Client

- 1 The VPN client should receive the following emails:
 - Configuration for SecuExtender IPSec VPN Client email with attached VPN configuration file (.tgb). Save the configuration file in your computer.
 - Nebula Cloud Account Information email with the following login information: Email, Username, Password, and Expired time (validity = NEVER).
- 2 Click the link in the **Configuration for SecuExtender IPSec VPN Client** email for instructions on installing the SecuExtender and activating the license key. The **How to activate SecuExtender license key after your online purchase** webpage appears.
 - Click Download.
 - Select the SecuExtender app based on your computer's operating system to install it.
 - Follow the online prompts to activate the SecuExtender license.

3.17.4 Import the VPN Configuration File

- 1 Save the attached VPN configuration file (.tgb) from the Configuration for SecuExtender IPSec VPN Client email on the VPN user's computer.
- 2 On your computer, open SecuExtender. Click the menu icon.

VPN Connections	Х
? 📃	

3 Click Configuration > Import.

SecuExtender IPSec	VPN Client	- U	X
nfiguration Tools	?		
Save	Ctrl+S		
Import		VPN (
Export		IKE 1/2	
Get from Server			
Move to USB Drive			
Wizard		IKE V2 Configuration	
Quit			
		This folder enables the creation of IKE V2 tunnels. It is possible to create as many IKE Auth SA and Child SA as required. The contextual menu (right click on IKE V2) enables to create, copy or paste IKE Auth SA and Child SA.	
		IKE V2 tunnel creation wizard	
		Export all IKE V2 tunnels	

4 Locate in your computer and click **Open** to import the VPN configuration file from the Configuration for SecuExtender IPSec VPN Client email.

💙 SecuExter	nder IPSec VPN C	lient				\times						
Configuration	n Tools ?											
ZYX	EL				VPN	CLIENT						
		IKE V	2									
VPN Confi	🐭 Open	¹ Open										
	Look in:	P15	G 🤌 📂 🛄 -									
IKE V2	Quick access Desktop Libraries This PC	Name How to S PDF Files SecuExter UG VG File name: File name:	<pre>ctup Zyxel Nebula Remote Acce inder VPN Client_IPSec_6.6.87.10 ha-6ebe4d35.d2ns-nbl.com.tgb IKEv2-alpha-6ebe4d35.d2ns-nbl.com Compatible files (*tgb)</pre>	Date modified 2022/5/30 下午 05:34 2022/4/12 下午 02:55 2022/6/16 上午 10:37 2022/6/16 下午 01:29 2022/6/16 下午 01:26 2022/6/16 上午 10:56	Type File fc File fc File fc File fc File fc TheG							
l			Upen as read-only									
VPN Client	tready											

5 Click RemoteAccessVPN in VPN Configuration > IKE V2 > RemoteAccessVPN.



3.17.5 Open the VPN Tunnel

1 Right-click RemoteAccessVPN in VPN Configuration > IKE V2 > RemoteAccessVPN and click Open tunnel.
💙 SecuExtender	IPSec VPN Client				- 🗆	×
Configuration To	pols ?					
ZIAEL					VPN CLIE	ENT
		RemoteA	ccessVPN: Child	I SA		
VPN Configuratio	on	Child SA Adv	anced Automation Re	emote Sharing	IPV4 IF	vV6
	arameters	Traffics	electors			
in the v2	AccessVPN		VPN Client address	0.0.0.0	1	
o Ren	Open tunnel	Ctrl+0	VFIV CIETT address	0.0.0.0]	
	Export		Address type	Subnet address \sim		
	Conv	Chilli C	Remote LAN address	0.0.0.0		
	Rename	F2	Subnet mask	0.0.0.0]	
	Delete	Del			the gateway	
		Cryptog	raphy		, one gaterial	
			Encryption	AES CBC 256 V		
			Integrity	SHA2 256 $$		
			Diffie-Hellman	DH21 (ECP 521) 🛛 🗸		
		Exte	ended Sequence Number	No 🗸		
		Lifetime				
			Child SA Lifetime	28800 sec.		
VPN Client read	ły					

2 On the next screen, enter the Login: Username and Password from the Nebula Cloud Account Information email. Then click OK.

R 1	nter Auth Innel.	entication login and password to ope	n the
	Login:	vpnuser	
Pa	essword:	•••••	

IKEV2 Auth sent will appear on the lower right of the screen.



Wait until **Tunnel opened** appears on the lower right of the screen.



An IP address will now appear in VPN Client address to replace the previous 0.0.0.0. The button lights green in front of RemoteAccessVPN in VPN Configuration > IKE V2 > RemoteAccessVPN.

3 When Your connection isn't private appears on the web browser, click Advanced to continue.



4 Click the Continue to xxx.xxx.x.x (unsafe) link on the bottom of the screen.



3.17.6 Set Up Two Factor Authentication to Bind the User Account

1 On the Two factor authentication screen, click Setup.

two tactor auto	entication
Your network req	uire two-factor authenticator, please setup your Google
Serve	

The prompt to download and install the **Google Authenticator** app on a mobile device appears. Install the **Google Authenticator** app. Then click **Next**.



2 Use the Google Authenticator app to scan the QR code. The QR code contains the user account information created in step 2 of Create a VPN User. Enter the code. Then click Verify.

nebula Control Center	
	Link your account
	1. Scan this QR code with the authentication app on your phone.
	If you can't scan the QR code, enter the following code (tap to copy) info the app: V INFACMAETEL CONFLICTED INFORMATION
	2. Enter the code generated by the authenticator app to finish linking your account.
	cosnort
	Cancel Verty

Note: Two Factor Authentication needs to be set up by the user only once. On the next login, just enter the Two Factor Authentication passcode.

The following screen will appear in the user's web browser.

Welcome to Zyxel
Successi
Powered by ZYXEL

3.17.7 Check the Connection in Nebula by the Administrator

Go to the Site-wide > Monitor > Firewall > VPN connections screen. The remote VPN connection should appear in Client to site VPN login account table.

Connection status						
Configuration		This security gateway is ex	oporting 2 subnet over the VPN	19216810/24, 19216820/24		
Site connectivity						
Location	Subnet	Status	Inbound	Outbound	Tunnel Up Time	Last Heartbeat
Non-Nebula VPN peer	s connectivity					
Non-Nebula VPN peer	s connectivity Subnet	Status	Inbound	Outbound	Tunnel Up Time	Last Heartbeat
Non-Nebula VPN peer	s connectivity Subnet	Status inbound	inbound Out	Outbound	Tunnel Up Time Tunnel Up Time	Last Heartbeat
Non-Nebula VPN peer Location Remote AP VPN Nome	s connectivity Subnet Status	Status Inbound	Inbound	Outbound	Tunnel Up Time Tunnel Up Time	Last Heartbeat Last Heartbeat
Non-Nebula VPN peer Location Remote AP VPN Name Client to site VPN login	s connectivity Subnet Status	Status	inbound Cut	Outbound	Tunnel Up Time Tunnel Up Time	Last Heartbeat Last Heartbeat
Non-Nebula VPN peer Location Remote AP VPN Name Client to site VPN login User Name	s connectivity Subnet Status	Status Inbound	inbound Cut	Outbound bound ligned IP	Tunnel Up Time Tunnel Up Time Public IP	Last Heartbeat Last Heartbeat

3.18 Route L2TP VPN Traffic

L2TP (Layer 2 Tunneling Protocol) is a tunneling protocol used to support virtual private networks (VPNs). L2TP works at layer 2 (the data link layer) to tunnel network traffic between two Nebula Devices over another network (like the Internet). In L2TP VPN, establish an IPSec (Internet Protocol Security) VPN tunnel first and then build an L2TP tunnel inside it. IPSec VPN connects IPSec routers or remote users using an IPSec software such as SecuExtender.

The following example figure shows a VPN client (C) connecting to a Nebula Device (R1) through an L2TP VPN (V1). Nebula Device (R1) connects to Nebula Device (R2) using site-to-site VPN (V2). The VPN client (C) can access a server (S) inside the Nebula Device (R2) through the two VPN tunnels (V1, V2).



You can set up a VPN site-to-site tunnel to a cloud computing service like Microsoft Azure. To route L2TP traffic between your site and Microsoft Azure site, do the following:

Nebula Device (Firewall device) IP address	192.168.1.1
L2TP VPN (source site)	192.168.3.0/24
Microsoft Azure network (destination site)	172.10.1.0/24

Go to Site-wide > Configure > Firewall > Routing: Policy Route/Traffic Shaping: Add.

- Enter a definition for the rule in **Description**: for example, L2TP_Routing.
- Enter the L2TP IP address range to which this rule applies in Source IP: 192.168.3.0/24.
- Enter the **Destination** IP address range to which this rule applies: 172.10.1.0/24.
- Select Any protocol to apply the policy route to in Service.
- Click to enable Policy Route.
- Select VPN Traffic in Type to route the matched packets through the VPN tunnel you specified in the Next-Hop field.
- Select the remote VPN gateway's site name in Next-Hop.

Then click **Update**. Network traffic can now pass between your site and Microsoft Azure site through the L2TP tunnel.

Update polic	cy route / Traffic Shaping Rule	×
Matching Criter	L2TP_Routing	×
Source:	192.168.3.0/24 ⊗	*
Destination:	172.10.1.0/24 😵	*
Service:	Any	-
Policy Route	VPN Traffic	* *
Traffic Shaping 🤇		
	Close	Update

3.19 Resolve WiFi Connection Problems (for Nebula APs only)

The **WiFi Aid** tab in **Site-wide** > **Clients** helps you identify connection problems between WiFi clients and supported AP(s) for a selected time range.

Note: Make sure your Nebula AP is using the latest firmware.

The following tables allow you to view and identify connection problems using the following categories.

- Connection Issues by SSID
- Connection Issues by Client
- Connection Issues by Access Point
- Captive Portal Login Issues by Client

Client list	🖤 WiFi Aid 🛛 Beta	Connection log			
is feature is designed	to provide an overview to	summarize the issue from a w	ireless client connection. It's only available for a	ccess points. <u>Model list</u>	
Time range:	SSID	AP tog			
Last 24 hours •	All SSIDs		•		
Client devices affe	ited by connection proble	ms 1/26 devices		Client devices affect	ed by captive portal problems
(→(<u>;</u>)-	\rightarrow		
Wireless 1		O	O		O
failures		failures	failures		failures
Connection issues by	SSID		Connection issues by cli	ent	
Connection issues by	SSID # Clients affected by c	connection problems	Connection issues by cli Client device	cnt # Failed/total connection	ns Latest failed issue
Connection issues by SSID e-Nebula-FT	SSID # Clients affected by c	connection problems	Connection issues by cli Client device D2350EFE71F9	ent # Failed/total connection 2/19	ns Latest failed issue Wireless connection
Connection issues by SSID e-Nebula-FT	SSID # Clients affected by c]	connection problems	Connection issues by cli Client device D2350FEE71F9 BE795104D518	ent # Failed/total connection 2/19 1/14	ns Latest failed issue Wireless connection Wireless connection
Connection issues by SSID e-Nebula-FT	SSID # Clients affected by o	connection problems	Connection issues by cli Client device D2350FEE7/F9 BE795104D518 SEEA F9 ED IB 83	ent # Failed/total connection 2/19 1/14 1/16	ns Lotest failed issue Wreless connection Wreless connection Wreless connection
Connection issues by SSID e-Nebula-FT	SSID # Clients affected by o	connection problems	Connection issues by cli Client device 02350EEE77.E9 EE79410405.B 26500 EF5610 B6500 EF5610	ent # Failed/lotal connection 2/19 1/14 1/16 1/17	ts Latest failed issue Wireless connection Wireless connection Wireless connection Wireless connection
Connection issues by SSID e-Nebula-FT	# Clients affected by o	connection problems	Connection issues by cli Client device D2350EEE77F9 BE795104D538 3E5AF9ED18B3 B65E06756030 F6119A3EA9EF CharburgteWatch	ent # Folled/total connection 2/19 1/14 1/16 1/17 1/9 1/1	ns Latest foiled issue Wireless connection Wireless connection Wireless connection OHCP Wireless connection
Connection issues by ssp e-Nebulo-FT Connection issues by	# Clients affected by o	connection problems	Connection issues by cli Client device D2250EFE71F9 BE796104 D5 18 SERA F9 ED 18 B3 B658 D9 F76010 F6119A SE AB EF ChuchusheWatch	ent # Failed/total connection 2/19 1/14 1/16 3/17 1/3 3/1 is by client	ns Latest foiled issue Wireless connection Wireless connection Wireless connection DHCP Wireless connection
Connection issues by SSID e-Nebulo-FT Connection issues by Access point	# Clients affected by o	connection problems	Connection issues by cli Client device D2350E EE77E9 EE796104 D238 DE58208E76010 E618845EABEE ChurbustleWatch Captive portal login issue Client device	ent # Failed/total connection 2/19 1/14 1/16 3/17 1/9 1/1 1/1 i/1 # Failed outhent # Failed outhent	ns Latest foiled issue Wireless connection Wireless connection Wireless connection DHCP Wireless connection
Connection issues by ssip e-Nabula-FT Connection issues by Access point Product team.	# Clients affected by o I access point. # Clients affected by 1	connection problems	Connection issues by cli Client device D2350E FE71F9 BE796104 D2 88 BE58 D9 F76010 F6118A 5E A8 EF ChurburtleWatch Captive portal login issue Client device Zyzel5E	ent # Failed/total connection 2/19 1/14 1/16 3/17 1/9 1/1 1/1 1/1 es by client # Failed outhent 1	ns Latest foiled issue Wireless connection Wireless connection Wireless connection DHCP Wireless connection
Connection issues by sisto e-Nabula-FT. Connection issues by Access point Product team. PMM	# Clients affected by o 2 access point # Clients affected by 1 3	connection problems	Connection issues by di Client device D2350E EE77E9 BE79504 D383 BE5820E7603C E61844E48EF ChurburcleWatch Captive portal login issue Client device ZVX015F example	ent # Failed/total connection 2/19 1/14 1/16 3/17 1/9 1/1 1/9 1/1 1/9 1/1 1/9 1/1 1/9 1/1 1/9 1/1 1 1 1	ns Latest foiled issue Wireless connection Wireless connection Wireless connection DHCP Wireless connection
Connection issues by sisto e-Nebula-FT. Connection issues by Access point Product team. PMM	# Clients affected by o 2 access point # Clients affected by 1 3	connection problems	Connection issues by dir Client device D2350E EE77E9 BE79504 D38 3E5A 59 ED 1983 Be58 D8F7603C F618A 5EA9 EF ChurburgteWatch Captive portal login issue Client device ZXX015F example Netbula	ent # Foiled/total connection 2/19 1/14 1/16 3/17 1/9 1/1 1/9 1/1 1/9 1/1 1/1 1/1 1/1 1/1	ns Latest foiled issue Wireless connection Wireless connection Wireless connection DHCP Wireless connection
Connection issues by sisto e-Nebulo-FT Connection issues by Access point Product team PMM	# Clients affected by o 2 access point # Clients affected by 1 3	connection problems	Connection issues by dir Client device D2350EEE77F9 BE79504D938 3E5A59ED9883 Be520E76030 F6119A5EA9EF CharburgteWatch Captive portal login issue Client device Zyze15E example Network test	ent # Failed/total connection 2/19 1/14 1/16 1/17 1/9 1/1 1/1 1/1 1/1 ess by client # Failed outherst 1 1 1 1 2	ns Latest foiled issue Wireless connection Wireless connection Wireless connection DHCP Wireless connection

Connection Issues by SSID

This table displays the number of WiFi clients with WiFi connection/DHCP client/DNS failures in each WiFi network. The list displays the WiFi network with the most connection failures first, in descending order.

1 Click a hyperlink in the # Clients affected by connection problems column.

Connection issues by S	ŝiD
SSID	# Clients affected by connection problems
e-Nebula-FT	1

The Site-wide > Monitor > Site features logs screen appears showing all related event logs for WiFi clients in the e-Nebula-FT WiFi network in the last 24 hours.

Client list	💎 WiFi Aid Be	ta 🛛 💎 Connection log Beta	
Last 24 hour	s 🔻 e-Nebule	a-FT 🔹 All APs	▼ Wireless failed connecti ▼ All clients ▼
Connection time	Connected to	Event type	Detail Issue
2023-01-16 17:37:32	Product team	Wireless failed connection [WiFi Aid]	Station: a0:78:17:8d:4d:b9 blocked by key handshake fail on Channel: 112, SSID: e-Nebula-FT, 5GHz, Signal: -88d
2023-01-16 17:26:21	Product team	Wireless failed connection [WiFi Aid]	Station: 8e:34:72:56:40:74 blocked by key handshake fail on Channel: 6, SSID: e-Nebula-FT, 2:4GHz, Signal: 0dBn
2023-01-16 17:15:58	Product team	Wireless failed connection [WiFi Aid]	Station: 22:10:a8:0b:66:c8 blocked by key handshake fail on Channel: 112, SSID: e-Nebula-FT, 5GHz, Signal: 0dBn
2023-01-16 16:33:49	Product team	Wireless failed connection [WiFi Aid]	Station: f6:11:ba:5e:ab:ef blocked by key handshake fail on Channel: 112, SSID: e-Nebula-FT, 5GHz, Signal: 0dBm,
2023-01-16 16:33:48	Product team	Wireless failed connection [WiFi Aid]	Station: f6:11:ba:5e:ab:ef blocked by key handshake fail on Channel: 6, SSID: e-Nebula-FT, 2.4GHz, Signal: OdBm,
023-01-16 16:28:59	Product team	DHCP client [WiFi Aid]	F6:11:BA:5E:AB:EF failed to receive an IP address due to DHCP failure/timeout (No response after offer) with DH
2023-01-16 14:50:09	PMM	Wireless failed connection [WiFi Aid]	Station: f6:6c:06:d2:51:af blocked by key handshake fail on Channel: 1, SSID: e-Nebula-FT, 2.4GHz, Signal: -75dBr
2023-01-16 14:41:58	PMM	Wireless failed connection [WiFi Aid]	Station: b6:5b:d8:f7:60:1c blocked by key handshake fail on Channel: 112, SSID: e-Nebula-FT, 5GHz, Signal: -76dB
2023-01-16 14:28:49	PMM	Wireless failed connection [WiFi Aid]	Station: 3a:9c:59:3a:f5:c3 blocked by key handshake fail on Channel: 112, SSID: e-Nebula-FT, 5GHz, Signal: -93dt
2023-01-16 13:03:34	Product team	Wireless failed connection [WiFi Aid]	Station: 7c:04:d0:f1:50:51 blocked by key handshake fail on Channel: 6, SSID: e-Nebula-FT, 2.4GHz, Signal: 0dBm

- 2 Use the following information listed in chronological order to resolve WiFi connection issues.
 - Connection time. This shows the starting time period from which the event log occurred.
 - Connected to. This shows the name (if available) or MAC address of the connected client.
 - Event type. This shows the event type (Association, Authentication, Disconnection, DHCP server, Wireless failed connection, DHCP client, DNS failure, Captive portal) that occurred.
 - Detail issue. This shows a summary of the APs event logs in chronological order.

Connection Issues by Client

This table displays the number of WiFi clients with failed connection attempts (WiFi connection/DHCP client/DNS failures – numerator) over the number of total connection attempts (denominator). The list displays the WiFi client with the most connection failures first, in descending order.

1 Click a hyperlink in the **Client device** column.

Client device	# Failed/total connections	Latest failed issue
D2:35:0E:EE:71:F9	2 / 16	Wireless connection
8E:79:61:04:D5:1B	<u>1</u> /14	Wireless connection
3E:BA:F9:ED:1B:B3	1/16	Wireless connection
B6:5B:D8:F7:60:1C	1/17	Wireless connection
F6:11:BA:5E:AB:EF	1/3	DHCP
ChiuhuipleWatch	1/1	Wireless connection





- 2 Use the information in this screen to identify the WiFi client with connection issues. See Table 24 on page 196 for the description of the fields.
- 3 Click History: Event log to view Nebula AP log messages. Enter the Nebula AP's name or a key word, select one or multiple event types, or specify a date/time or even a time range to display only the log messages related to it.
- 4 Click **Ping** to ping the client's IP address from the Nebula AP to test connectivity.
- 5 Click the numerator hyperlink in the **# Failed/total connections** column.

lient device	# Failed/total connections	Latest failed issue
09-95-0E-EE-71-ED	<u>2</u> ∕19	Wireless connection
JE-70-61-04-05-10	1 14	Wireless connection
E-DA-EN-EN-10-00	1 16	Wireless connection
8-50-D0-E7-80-10	1 17	Wireless connection
R-11-0 A-5E- A D-EE	1 3	DHCP
ChiuhuipleWatch	1 1	Wireless connection

The **Site-wide** > **Monitor** > **Site features logs** screen appears showing all related event logs between APs and WiFi clients. See Section on page 115 on using the information listed in chronological order to resolve WiFi connection issues.

Client list	💎 WiFi Aid Beta	💎 Connection log Beta	
Last 24 hour	s 🔻 All SSIDs	✓ All APs	✓ Association, Disconnecti ▼ D2:35:0EEE:71:F9 ▼
Connection time	Connected to	Event type	Detail Issue
023-01-16 13:44:10	Product team	DHCP client [WiFi Aid]	D2:35:0E:EE:71:F9 succeeded to receive IP address 173.16.2.121, SSID: e-Nebula-FT.
023-01-16 13:44:08	Product team	Association	Station: d2:35:0e:ee:71:f9 connected on Channel: 112, SSID: e-Nebula-FT, SGHz, Signal: -52dBm. Interface:wlan-
023-01-16 11:33:51	Product team	Association	Station: d2:35:0e:ee:71:f9 connected on Channel: 112, SSID: e-Nebula-FT, 5GHz, Signal: -41dBm. Interface:wlan-
023-01-16 10:48:05	Product team	DHCP client [WiFi Aid]	D2:35:0E:EE:71:F9 succeeded to receive IP address 173:16:2:121, SSID: e-Nebula-FT.
023-01-16 10:48:04	Product team	Association	Station: d2:35:0e:ee:71:f9 connected on Channel: 112, SSID: e-Nebula-FT, 5GHz, Signal: -46dBm. Interface:wlan
023-01-16 10:03:21	Product team	Wireless failed connection [WiFi Aid]	Station: d2:35:0e:ee:71:f9 blocked by group rekey handshake fail on Channel: 112, SSID: e-Nebula-FT, 5GHz, Si
023-01-16 10:03:17	PMM	Wireless failed connection [WiFi Aid]	Station: d2:35:0e:ee:71:f9 disconnected by group rekey handshake fail on Channel: 112, SSID: e-Nebula-FT, 5G
023-01-16 10:03:07	Product team	DHCP client [WiFi Aid]	D2:35:0E:EE:71:F9 succeeded to receive IP address 173:16:2:121, SSID: e-Nebula-FT.
023-01-16 10:03:03	Product team	Association	STA fast roamed, MAC:D2:35:0E:EE:71:F9, From:PMM, To:Product team, SSID:e-Nebula-FT.
023-01-16 10:03:03	Product team	Association	Station: d2:35:0e:ee:71:f9 connected on Channel: 112, SSID: e-Nebula-FT, 5GHz, Signal: -79dBm. Interface:wlan

Connection Issues by Access Point

This table displays the number of WiFi clients with WiFi connection/DHCP client/DNS failures listed according to access point. The list displays the access point with the most connection failures first, in descending order.

1 Click a hyperlink in the # Clients affected by connection problems column of a specific AP.

Access point	# Clients affected by connection problems
Product team	1
РММ	1

The **Site-wide** > **Monitor** > **Site features logs** screen appears showing all related event logs between a specific AP (for example, Product team) and its WiFi clients. See Section on page 115 on using the information listed in chronological order to resolve WiFi connection issues.

	WITTAID BOO	Connection log Beta	
Last 24 hour	s 🔻 All SSIDs	▼ Product team	
Connection time	Connected to	Event type	Detail Issue
2023-01-16 14:17:51	Product team	Association	Station: f6:6c:06:d2:51:af connected on Channel: 112, SSID: e-Nebula-FT, 5GHz, Signal: -73dBm. Interface:wlan-
2023-01-16 13:57:05	Product team	Association	STA fast roamed, MAC:3A:9C:59:3A:F5:C3, From:PMM, To:Product team, SSID:e-Nebula-FT.
2023-01-16 13:57:05	Product team	Association	Station: 3a:9c:59:3a:f5:c3 connected on Channel: 112, SSID: e-Nebula-FT, 5GHz, Signal: -57dBm. Interface:wlan
2023-01-16 13:45:25	Product team	DHCP client [WiFi Aid]	A0:78:17:8D:4D:B9/TWNBNT03245-MBP succeeded to receive IP address 173.16.2.40, SSID: e-Nebula-FT.
2023-01-16 13:45:25	Product team	Association	Station: a0:78:17:8d:4d:b9 connected on Channel: 112, SSID: e-Nebula-FT, 5GHz, Signal: -65dBm. Interface:wlar
2023-01-16 13:44:23	Product team	Association	Station: 22:10:a8:0b:66:c8 connected on Channel: 112, SSID: e-Nebula-FT, 5GHz, Signal: -56dBm. Interface:wlar
2023-01-16 13:44:10	Product team	DHCP client [WiFi Aid]	D2:35:0E:EE:71:F9 succeeded to receive IP address 173:16:2:121, SSID: e-Nebula-FT.
2023-01-16 13:44:08	Product team	Association	Station: d2:35:0e:ee:71:f9 connected on Channel: 112, SSID: e-Nebula-FT, 5GHz, Signal: -52dBm. Interface:wlan
2023-01-16 13:43:37	Product team	DHCP client [WiFi Aid]	62:94:AB:B8:FF:27/Free-Wifi succeeded to receive IP address 173.16.2.56, SSID: e-Nebula-FT.
022-01-16 12:42:25	Product team	Association	Station: 62:94:ab:b8:ff:27 connected on Channel: 112, SSID: e-Nebula-FT, 5GHz, Signal: -69dBm. Interface:wlar

Captive Portal Login Issues by Client

This table displays the list of WiFi clients with the corresponding number of failed hotspot authentication. The list displays the WiFi client that failed hotspot authentication the most number of times first, in descending order.

1 Click a hyperlink in the Client device column.

Client device	# failed authentication	
zvxelSF	1	
example	1	
Nebula	1	
test	٥	
lobby	Q	

The Site-wide > Clients > Client list: WiFi client details screen appears showing individual client statistics. See Section on page 116 on setting the filters and using the information listed in chronological order to resolve WiFi connection issues.



- 2 Use the information in this screen to identify the WiFi client with connection issues. See Table 24 on page 196 for the description of the fields.
- 3 Click History: Event log to view Nebula AP log messages. Enter the Nebula AP's name or a key word, select one or multiple event types, or specify a date/time or even a time range to display only the log messages related to it.
- 4 Click **Ping** to ping the client's IP address from the Nebula AP to test connectivity.
- 5 Click the hyperlink in the **# Failed Authentication** column.

zvxel5F 1 example 1	
example 1	
Nebula 1	
test Q	

The **Site-wide** > **Monitor** > **Site features logs** screen appears showing all related event logs of a specific client device with failed hotspot authentication event type.

Client list WiFi Aid Beta Connection log Beta Last 24 hours All SSIDs All APs Captive portal [WiFi Aid] D6:20:21:43:41:F6	Clie	nts			
Last 24 hours All SSIDs All APs Captive portal [WiFi Aid] D6:20:21:43:41:F6		Client list	💎 WiFi Aid Beta	Connection log Beta	
		Last 24 hours	✓ All SSIDs	✓ All APs	▼ Captive portal [WiFi Aid] ▼ D6:20:21:43:41:F6 ▼

- 6 Use the following information listed in chronological order to resolve failed hotspot authentication issues.
 - Connection time. This shows the starting time period from which the event log occurred.
 - Detail issue. This shows a summary of the APs event logs in chronological order.

3.20 Configure Voice VLAN (for Nebula Switches only)

VoIP (voice over Internet protocol) devices are commonly in use in office environments. When designing a network, assign a higher priority to voice traffic. Use voice VLAN to prioritize voice packets from a VoIP device, and separate data packets from a computer.

As shown in the next figure, connect the VolP device (P) to the Nebula Device (S) on one end. Connect the computer (C) to the VolP device (P) on the other end. The VolP device (P) serves as a bridge for both the Nebula Device (S) and computer (C).



The Nebula Device will add a VLAN tag for voice packets (V) and data packets (D) separately after receiving them. Then forward the voice packets (V) and data packets (D) to the uplink port (U). This section shows you how to separate data packets (D) and voice packets (V) between a VoIP device (P) and computer (C), without having to assign a VLAN tag.

- Configure the Nebula Device Ports
- Configure the Voice VLAN

3.20.1 Configure the Nebula Device Ports

- 1 Go to Site-wide > Configure > Switches > Switch ports.
- 2 Select the port that connects to a VoIP device and click Edit.

Switch po	rts Last 2 hours 🔹 🖒										
Edit	ggregate≁ <mark>x[×]Split Tag▼</mark> Reset Q, 1	Search ports.		•	1)selected in (6)Switch ports					Export -
	Switch / Port	Port name	# Port	LLDP	Received bytes	Sent bytes	Connection	PoE	Tag	IPSG protected	Management contrc 📃
	BC:CF:4F:47:7D:F1(GS1350-6HP)/1 details	Port1	1	Enabled	0 bytes	0 bytes	1	Enabled	li -	N/A	Enabled
	BC:CF:4F:47:7D:F1(GS1350-6HP)/2 details	Port2	2	Enabled	0 bytes	0 bytes	1	Enabled		N/A	Enabled
	BC:CF:4F:47:7D:F1(GS1350-6HP)/3 details	Port3	3	Enabled	0 bytes	0 bytes	1	Enabled		N/A	Enabled
	BC:CF:4F:47:7D:F1(GS1350-6HP)/4 details	Port4	4	Enabled	0 bytes	0 bytes		Enabled		N/A	Enabled
	BC:CF:4F:47:7D:F1(GS1350-6HP)/5 details	Port5	5	Enabled	0 bytes	0 bytes		Enabled		N/A	Enabled
	BC:CF:4F:47:7D:F1(GS1350-6HP)/6 details	Port6	6	Enabled	0 bytes	0 bytes		N/A		N/A	Enabled
4											

- 3 Select Access for the port Type.
- 4 Select Voice VLAN for the VLAN type.
- 5 Assign a **PVID** for the port. Use the PVID to tag data packets with the VLAN ID.
- 6 Then click Update.

Switch ports	BC:CF:4F:47:7D:F1(GS1350-6F	HP)/1				
Name	Port1	×	Bandwidth control	Disabled		•
Tags	None		Loop guard	Disabled		•
Port enabled	Enabled	-	Storm control	Disabled		
RSTP	Enabled	-		Broadcast Limit (pps)	100)
LLDP	Enabled	•		Multicast Limit (pps)	100	>
Link	Auto	•		DLF Limit (pps)	100)
Extended range 🐧	Disabled	•				
Port isolation	Disabled	•	Туре	Access		
			Management control	Enabled		•
			VLAN type	Voice VLAN		
			PVID	5		;

3.20.2 Configure the Voice VLAN

1 Go to Site-wide > Configure > Switches > Switch settings.

- 2 Scroll to the Voice VLAN part of the screen.
- 3 Click the switch to enable the voice VLAN feature in the Nebula Device.
- 4 Enter a Voice VLAN ID.
- 5 Select the **Priority** of the voice VLAN from 1 to 6.
- 6 Select OUI in Assign VLAN by. The Nebula Device assigns the port connected to the VoIP device to the voice VLAN if the connected VoIP device's OUI matches any OUI in the list.
- 7 Enter the **OUI** address of the VoIP device. The OUI (Organizationally Unique Identifier) is the first three octets of the VoIP device's MAC address. By specifying the MAC address, the Nebula Device can identify voice traffic accordingly.

Voice VLAN					
Voice VLAN 🚺					
Voice VLAN ID:	10	×			
Priority:	5	*			
Assign VLAN by:	OUI	-			
OUI:	oui		Description		
	1 00:50:04		× SCOM	× * 💼	
	+ Add OUI on this net	twork			
Vendor ID based VLAN					
Vendor ID based VLAN Model list					
Access management					
Access management Model list					
DHCP Server Guard					
DHCP Server Guard: 🏮					
IP source guard Model list				(Please allow 1-2 minutes for chance	aes to take effect.)
IP source guard					Ask Question

Note: The Nebula Device supports up to six vendor OUIs.

8 Then click Save.

3.21 Manage IPTV (for Nebula Switches only)

This section shows you how to configure IPTV settings and view IPTV reports:

- Set up the VLAN for IPTV
- Define the Role of a Switch

• Configure the Channel Profile and Naming

3.21.1 Set up the VLAN for IPTV

1 Go to the Site-wide > Configure > Switches > Advanced IGMP screen. Click IGMP snooping to enable IGMP snooping on all Switches in the site. Under IGMP-snooping VLAN, select Auto-detect to automatically detect which VLANs are used for IPTV. Otherwise, manually enter the VLAN IDs (1 – 4094, up to 16 VLANs, separated by commas, no spaces) in the User Assign VLANs field. Click Save when you are finished.

Vancea IGIVIP		Verride switch configuration
GMP snooping		
GMP-snooping VLAN Model list	Auto-detect 1,2,4 × User Assign VLANs.	
Unknown multicast drop Model list Drop on VLAN	All ×	
GMP filtering profiles 0		O IGMP filtering profile
+ Add		
IPTV topology setup		
	s IGMP topology tips	Save or Cancel

2 If you have not defined the IP address of the Switch, go to the Site-wide > Configure > Switches > IP & Routing screen and click +Add under IP interface. The following screen appear. Enter the Interface IP, Subnet mask and ID number of the VLAN used for IPTV. Click Create to save the setting.

Interface ed	itor	×	
Switch	101012-02-02-020-04		
	This switch only supports in and monitor purpose. No rou switch.	nterfaces for management ting capability on this	
Name	Interface VLAN 2	×	
Interface IP		*	
Subnet mask		*	
VLAN	2		
		Close Create	

3.21.2 Define the Role of a Switch

1 Go to the Site-wide > Configure > Switches > Advanced IGMP screen. Under IPTV topology setup, select a Switch you want to configure and select a Role to define the role of your Switch from the drop-down list box.

Note: Click the **IGMP topology tips** link to view information about Switch roles. If the role of the Switch is not defined accordingly, the IPTV performance will be greatly affected.

IGMP	snooping - Role - Port set	tings IGMP topology tips		
~	Switch name	IGMP snooping	Role	Port settings
~	RR-EC-12-14		-Select role-	Advanced setup
			-Select role-	
			Querier	
			Aggregator	
			Access	

2 After you define the role of the Switch, click Advanced setup and the following screen appears. The Leave mode will show the default setting based on the role you select. But you can still go back to the Advanced IGMP screen to configure the Role and Leave mode. Under Maximum group, you can select Enable and enter the maximum number of channels allowed at a time. Otherwise, select Disable. Click Save to save the changes.

Note: You can click **Reset** to reset the port settings to default.

Port settings			×
Switch name	\$1101.40.42.54.W		
Role	Querier		
Leave mode	Normal leave	×*	
Maximum group	Enable -	×	
IGMP filtering profile	No select 👻		
Reset		Close	

3 If a reminder of Network analytic alert appears on the Site-wide > Monitor > Switches > IPTV report page, click the Update filter rules link below to use the default ACL rules to block UPnP packets. In the example screen below, a Network analytic alert indicates that your IPTV traffic flow is affected by unneeded UPnP packets. Click the Update filter rules link to define IP filtering rules in the Site-wide > Configure > Switches > ACL screen to block these packets.



4 The Update filter rules link will lead you to the following screen. Click Save to save the default setting to block UPnP packets.

lanager	nent rule	s <u>What is thi</u>	2					
ebula co	ntrol cente	er IP address						
19.85.221								
ustomiz		es <u>Model list</u>	-					
	Enabled	Policy	Protocol	Source MAC		Source IP	Source port	Destination MAC
ۇ 1		Deny 🔻	UDP 🔻	any	× *	any ×*	any × *	any
- ∲→ 1		Deny 🔻	UDP 👻	any Any	× *	any × *	any × *	any

3.21.3 Configure the Channel Profile and Naming

A channel profile is the IP address range allowed to receive IPTV channels. An IPTV channel is used to send video traffic to the IP addresses in the channel profile.

1 To set up a range of available IPTV channels, go to the Site-wide > Configure > Switches > Advanced IGMP screen. Under IGMP filtering profiles, click +Add and the following screen appear. Enter a Profile name and enter the Start IP address and End IP address. Click Save & Back to save the changes.

IGMP filter			×
Profile name	New Name	*	
Start IP address		End IP address	
1	*		× *
+ Add			
			Close Save & Back

2 To edit the naming of the IPTV channels, go to the Site-wide > Monitor > Switches > IPTV report screen and click Channel management under Channel information.

		0	0	0	
		lotal channels	Channel in use	Current viewers	
hanne	el summary Top	o 11 to 20 channels -			
Before	▼ 2021-03-10	💼 16:18 🔻 24 ho	▼ UTC+8 Search		
	٦'				
	0.9 -				
	0.7 -				
0	0.6 -				
ity (%	0.5 -		No data to display		
oulari	0.4 -				
Pop	0.3 -				
	0.1-				
	o				
etwor	k analytic alert				
o abno	rmality detected				

- **3** You can choose to either import an updated channel list (channels.xlsx), or enter/edit each **Channel** address and **Channel name** individually.
 - Under **Channel management**, click **channel list** to download a blank Excel file template, edit accordingly and save it, and then click **import** to import the complete channel list to Nebula. Or,
 - Click +Add to add and then add/edit a Channel address and Channel name at a time.

Control Center	anization: Nebula_O	rg 🔹 Site: Site01 👻	
~~	0.9 -		
6	0.8 -		
ADT WEP	0.7 -		
Organization-wide	(%) 0.5-	No c	data to display
	-4.0 riguit	Channel management	×
Site-wide	do 0.3 -	shamermanagement	~
~	0.2 - 01 - Y	ou can download the channel list here and import multipl	le records for faster channel namina
USG FLEX	0+	Channel address Channel name	
switch		×	× 💼
Access point	Network ana	+ Add	
	No abnormality		Close
	Channel information	ion	
	Channel managen	ent	
P Help	No channel to displo	ry	
Nebula Technology Alliance Partner			

4 To view the summary of the IPTV report, go to the Site-wide > Monitor > Switches > IPTV report screen. Click Channel summary to see the top or bottom viewed channels within the specified time period you choose.

3 Total channels	s O Channel in use	O Current viewers		
2020-01-0 💼	10:15 🔻 Last day 👻	Bef 🔻		
hannel summe	ary			
Top 10 cnannes				
	100	.42		
	90			
(%	70			
arity ('	50			
Popul	40			
	20			
	0		3.82	0.76
	224.0.	.0.251	239.255.255.250	224.0.0.252
etwork analy	ytic alert			
Network analy 1020-01-07 07:48 IPnP packets ho IPnP packets m	ytic alert 8:36 ave been detected on the IPTV netw hay interfere with IPTV traffic and ca	ork. use pixilation. You can use IP Filter	ing to block UPnP packets. <u>Update filter rules</u> to drop U	IPnP traffic by destination address.
Vetwork analy 1020-01-07 07:48 IPnP packets ho IPnP packets m	ytic alert 8:36 ave been detected on the IPTV netwo way interfere with IPTV traffic and ca	ork. use pixilation. You can use IP Filter	ring to block UPnP packets. <u>Update filter rules</u> to drop U	PnP traffic by destination address.
Network analy 020-01-07 07:48 IPnP packets ho IPnP packets m Shannel inforr	ytic alert 836 ave been detected on the IPTV netwo way interfere with IPTV traffic and ca mation	ork. use pixilation. You can use IP Filter	ing to block UPnP packets. <u>Update filter rules</u> to drop U	iPnP traffic by destination address.
letwork analy 020-01-07 07:48 IPnP packets ho IPnP packets m Shannel inforr	ytic alert 8:36 ave been detected on the IPTV netwo Interfere with IPTV traffic and ca mation gement 3 Channels	ork. use pixilation. You can use IP Filter	ring to block UPnP packets. <u>Update filter rules</u> to drop U	IPnP traffic by destination address.
Network analy 020-01-07 07:48 IPnP packets ho IPnP packets m Shannel inforr Channel manage	ytic alert 8:36 ave been detected on the IPTV netwo hay interfere with IPTV traffic and ca mation gement 3 Channels	ork. use pixilation. You can use IP Filter	ring to block UPnP packets. <u>Update filter rules</u> to drop U	PnP traffic by destination address.
Vetwork analy 020-01-07 07:48 IPnP packets hr IPnP packets m Channel inforr Channel	ytic alert 8:36 ave been detected on the IPTV netwo way interfere with IPTV traffic and ca mation gement 3 Channels	ork. uuse pixilation. You can use IP Filter Switch Por	ring to block UPnP packets. <u>Update filter rules</u> to drop U rt name Port VID	IPnP traffic by destination address.
Network analy 1020-01-07 07-48 JPnP packets he JPnP packets m Channel inforr Channel manage Channel 22400.251	ytic alert 8:36 are been detected on the IPTV netwo hav interfere with IPTV traffic and ca mation Igement ③ Channels	ork. Iuse pixilation. You can use IP Filter Switch Por	ting to block UPnP packets. <u>Update filter rules</u> to drop U	IPnP traffic by destination address.
Network analy 2020-01-07 07:4E UPnP packets hr JPnP packets m Channel inforr Channel manag Channel 224 0 0 251 (2 224 0 0 252 (2)	ytic alert 8:36 ave been detected on the IPTV netwo way interfere with IPTV traffic and ca mation gement ③ Channels	ork. Iuse pixilation. You can use IP Filter Switch Por	ring to block UPnP packets. <u>Update filter rules</u> to drop L thame Port VID	IPnP traffic by destination address.
Network analy 2020-01-07 07:48 UPnP packets he Channel inforr Channel manage 224.0.0251 (224.0.0252 (2239.255.255.250	ytic alert B36 ave been detected on the IPTV netwo nay interfere with IPTV traffic and ca mation gement Channels Channel	ork. Iuse pixilation. You can use IP Filter Switch Por	'ing to block UPnP packets. <u>Update filter rules</u> to drop U t name Port VID	IPnP traffic by destination address.
Network analy 2020-01-07 07-45 UPnP packets in Channel inforr Channel manas Channel 224.0.0.251. (224.0.0.252. (239.255.255.250	ytic alert 8:36 ave been detected on the IPTV netwo way interfere with IPTV traffic and ca mation gement ③ Channels 2 2 2 2 2 2 2 2 3 4 4 4 4 4 4 4 4 4 4 4 4 4	ork. Iuse pixilation. You can use IP Filter Switch Por	ring to block UPnP packets. <u>Uodate filter rules</u> to drop U rt name Port VID	IPnP traffic by destination address.
Network analy 2020-01-07 07.4E UPnP packets in Channel inforr Channel manage Channel 224.0.0.251 (239.255.255.250	ytic alert B36 ave been detected on the IPTV netwo inay interfere with IPTV traffic and ca mation gement 3 Channels 2 2 2 2	ork. Iuse pixilation. You can use IP Filter Switch Por	ring to block UPnP packets. <u>Update filter rules</u> to drop U t name Port VID	IPnP traffic by destination address.
Network analy 2020-01-07 07-48 JPnP packets m Channel inforr Channel manage 224 0.0 251 (2) 224 0.0 252 (2) 239 255 255 250	ytic alert B36 ave been detected on the IPTV network invo Interfere with IPTV traffic and ca mation igement ③ Channels 2 2 2 2 2 2 2 3 4 4 5 5 5 6 6 7 7 7 7 7 7 7 7 7 7 7 7 7	ork. Iuse pixilation. You can use IP Filter Switch Por	ting to block UPnP packets. <u>Update filter rules</u> to drop U t name Port VID	IPnP traffic by destination address.

3.22 Enable IP Source Guard (for Nebula Switches only)

IP source guard consists of the following features:

- DHCP snooping. Use this to filter unauthorized DHCP server packets on the network and to build a binding table dynamically.
- ARP inspection. Use this to filter unauthorized ARP packets on the network.
- Static IP bindings. Use this to create static bindings in the binding table.

Binding Table

IP source guard uses a binding table to distinguish between authorized and unauthorized ARP packets in your network. The Nebula Device builds the binding table by snooping DHCP packets (dynamic bindings) and from information provided manually by administrators (static bindings).

DHCP Snooping

The Nebula Device only allows an authorized DHCP server on a trusted port to assign IP addresses. Unauthorized DHCP servers will not be able to assign IP addresses to network clients. When the Nebula Device receives a DHCP server packet from an authorized DHCP server, it inspects the packet and records the DHCP information in a binding table. The binding records are used in ARP inspection to filter unauthorized ARP packets.

ARP Inspection

When the Nebula Device receives an ARP packet, it looks up the appropriate MAC address, VLAN ID, IP address, and port number in the binding table. If there is a binding, the Nebula Device forwards the packet. Otherwise, the Nebula Device discards the packet.

If you want to use dynamic bindings to filter unauthorized ARP packets (typical implementation), you have to enable DHCP snooping before you enable ARP inspection.

The following figure demonstrates a scenario with DHCP snooping and ARP inspection enabled. In this scenario, we connect an authorized DHCP server (A) and the client devices on the ARP trusted ports (T). A client device (B) is assigned the IP address 192.168.1.56 by the authorized DHCP server (A). A malicious host (C) on an untrusted port (UT) puts a wrong MAC address with the IP address 192.168.1.56 in an ARP reply packet pretending to be client device (B) (192.168.1.56). The Nebula Device snoops DHCP packets sent from the authorized DHCP server (A) and creates bindings in the binding table. When the Nebula Device receives ARP packets from an untrusted port (UT), it compares the IP and MAC addresses with the existing bindings. Since the IP and MAC binding is different from the existing bindings, the Nebula Device blocks the unauthorized ARP packets sent from the malicious host (C). The malicious host (C) therefore cannot disguise as client device (B) to build connections with other client devices on your network.



To setup IP source guard on the Nebula, do the following:

1 Go to Site-wide > Configure > Switches > Switch settings. Slide the switch to enable IP source guard for the Nebula Devices in your site. Then click Save. The Protected switch and Allowed client list will appear. The Protected switch information synchronizes with the port's IPSG Protected setting in Site-wide > Configure > Switches > Switch ports. It will display the enabled ports.

P source guard				
Protected switch	IPSG adds protec address will need IP address to suc	ction to allow only author d to be inserted to "Permi ccessfully access the netv	ized client traffic in the ne tted client entry", others ne vork.	twork. Client with static and to renew their DHCF
	Switch name	IP source guard	Protected ports	Client table
Allowed client list 🚯	Action Q IP, M	IAC, VLAN	• Oclients	+ Add cl
	IPv4 addre	ess MAC (address VI	AN

NCC User's Guide

2 Click the IP Source Guard switch to enable/disable IP source guard for the specific registered Nebula Device(s) in your site.

Switch Name	IP Source Guard	Protected	ports	Client table
DC-00-11-D0-24-44		1,3,7	Ø	🕨 Run
XS3800-30		1,4		🕨 Run
XG\$2220-30		Null		► Run

3 Click the edit icon to go to Site-wide > Configure > Switches > Switch ports to configure Protected ports for the Nebula Device. A port is protected if IPSG protected is enabled on this port.

Switch Name	IP Source Guard	Protected	d ports	Client table
R00-11-DR-2A-AA		1,3,7		► Run
XS3800-30		1,4		► Run
XG\$2220-30		Null		► Run

4 Click to select the port you want to enable IP source guard.

dit A	ggregate 🔹 📌 Split 🛛 Ta	ng • Reset	Q Sear	ch ports_	- (2) selected in	n (28) Switc	h ports						💎 🕒 Exp
	Switch / Port	Port name	# Port	LLDP	Received bytes	Sent bytes	Enabled	Connection	PoE	Status	Туре	Tag	Number of IGMP Group	
	XS3800-1-1/1 <u>details</u>	Port1	1	Enabled	0 bytes	0 bytes	Enabled	_	N/A	Disabled	Trunk		0	
	XS3800-1-1/2 details	Port2	2	Enabled	0 bytes	0 bytes	Enabled		N/A	Disabled	Trunk		0	
	XS3800-1-1/3 details	Port3	3	Enabled	0 bytes	0 bytes	Enabled		N/A	Disabled	Trunk		0	
	XS3800-1-1/4 details	Port4	4	Enabled	0 bytes	0 bytes	Enabled		N/A	Disabled	Trunk		0	
1	XS3800-1-1/5 details	Port5	5	Enabled	0 bytes	0 bytes	Enabled		N/A	Disabled	Trunk		0	
]	XS3800-1-1/6 details	Port6	6	Enabled	0 bytes	0 bytes	Enabled		N/A	Disabled	Trunk		0	
1	XS3800-1-1/7 details	Port7	7	Enabled	0 bytes	0 bytes	Enabled		N/A	Disabled	Trunk		0	
]	XS3800-1-1/8 details	Port8	8	Enabled	0 bytes	0 bytes	Enabled		N/A	Disabled	Trunk		0	
)	XS3800-1-1/9 details	Port9	9	Enabled	0 bytes	0 bytes	Enabled		N/A	Disabled	Trunk		0	
1	XS3800-1-1/10 details	Port10	10	Enabled	0 bytes	0 bytes	Enabled		N/A	Disabled	Trunk		0	

Note: Do NOT configure IPSG on an uplink port as this may cause disconnection between the client device and Nebula.

To restore connection on an uplink port, go to **Site-wide > Configure > Switches > Switch ports** to select the uplink port. In the **Update 1 port** screen select **Disabled** in **IPSG protected**. Then reset the Nebula Device to its factory-default setting (see the Nebula Device's User's Guide for more information).

5 In the Update port screen, select Enabled in IPSG protected. The IPSG protected field in the Site-wide > Configure > Switches > Switch ports table for the updated port will display Enabled.

ral settings						
Switch ports	XS3800-1-1/1					
Name	Port1	×	Bandwidth control	Enabled		•
Tags	None			Ingress 1000	000	Kbps ×
Port enabled	Enchled			Egress 10000	000	Kbps 🗙
PSTP	Enablea	•	Loop guard	Englied		
Kall	Enabled	•		Endbled		•
STP guard	Root guard	•	Storm control	Enabled		•
LLDP	Enabled	•		Broadcast Limit (pps)	100	×
Link	Auto	•		Multicast Limit (pps)	100	×
Media type	SFP+	•		DLF Limit (pps)	100	×
Port isolation	Enabled	•	Туре	Access		•
IPSG protected	Enabled	•	VLAN type	Vendor ID based VLAN	1	•
Radius policy	Open	•	PVID	1		×
lettings	advanced IGMP setting					
Leave mode 🚺	Normal leave 🔻 4000	ms×				
Maximum Group 🚺	Enabled 💌 1	×				
IGMP filtering profile	No Select	•				
Fixed router port	Auto	•				

6 Click Run.

Switch Name	IP Source Guard	Protected p	Protected ports		
00.00.01.00.04.05		1,3,5,7		🕨 Run	
XS3800-30		1,4		🕨 Run	
XG\$2220-30		1		► Run	

7 A merged list window appears. Click to select the port and then click Transfer.

The learning table of 1. It takes about 5 n 2. Protected port is	could be empty if: ninutes to refresh the ad not specified.	dress table o	after you app	bly the Switch setting.
IP address	MAC address	VLAN	(N) Entries Port	Туре
192.168.1.100	0000000110100	1	1	dhcp-snooping

8 The port with the particular IP and MAC addresses is added to the Allowed client list. Click Save.

Allowed client list 🚺	Acti	ON Q. IP, MAC, VLAN	• 1 clients	+ Add client
		IPv4 address	MAC address	VLAN
		192.168.1.100	02-00-00-11-01-02	1
			Save or Cancel	
		(Please allow 1	-2 minutes for changes	to take effect.)

3.23 Set Up MAC Authentication With NCAS (for Nebula Switches only)

To set up MAC authentication with NCAS (Nebula Cloud Authentication Server), do the following:

- 1 Go to Site-wide > Configure > Switches > Authentication: Server type to select the authentication server.
- Click +Add to create the Authentication policy.
 Enter the Name (for example, Trusted Device) and select MAC-Base in Authentication type.
- 3 Go to Site-wide > Configure > Switches > Switch ports to bind the authentication policy to the access port(s).
 - **3a** Select the port(s) and click Edit.
 - 3b In the Update # port screen, select Access in Type. Select MAC-Base/Trusted Device in Auth. policy. Then click Update.
- 4 Go to Organization-wide > Organization-wide manage > Cloud authentication > MAC to add MAC addresses in the cloud authentication list.
 - 4a Click +Add to create to create a new user account.
 - 4b In the Create user screen, enter the MAC address for this account.

- 4c In the Authorized field, select the user's access to All sites or Specified sites. If you select Specified sites, a field displays allowing you to specify the sites to which the user access is authorized.
- 4d Then click Create user.

3.24 Set Up Dynamic VLAN With RADIUS (for Nebula Switches only)

In this example, VLAN10 is configured on port 1 (P1) of the Nebula Device. The user creates the following two accounts in the RADIUS server (R):

- Account with VLAN100 assignment
- Account without VLAN assignment.

Scenario 1:

The login account passes IEEE 802.1x port authentication with dynamic VLAN assignment. Client (C) will connect to the network through VLAN100.

Scenario 2:

The login account passes IEEE 802.1x port authentication without dynamic VLAN assignment. Client (C) will connect to the network through VLAN10.





To set up dynamic VLAN with RADIUS, do the following:

1 Configure the client list in the RADIUS server. In the example screen below, enter the management IP address of the Nebula Device in NAS. Enter the shared Secret (password) in your Site-wide > Configure > Switches > Authentication screen. Then click the add (+) button.

	RAD	IUS N	lanager										_				-		×
Fi	le Serv	/ice	Help Groups	Clients	🗐 Set	ttings	🗐 Log	Active Sess	sions 📕	Dictionary	🔳 Re	eports	<u>(</u> о	ounters	BØ S	QL			
ſ	NAS			Secret		Vend	lor	Username Part	Enabled	I Interim L	Jpdate	PoD &							
	100 100	1 22		admin1234		ie			Yes	0									
	107 168	1 25		admin1234		ie			Yes	0									
	RADIUS	Clie	nt Prope	rties Se	ecret		Usernam	e Part			Vend	dor		Enabl	led I	nterim (Jpdate P	Period	
	102 169	1 22	,	× a	dmin123	4					ietf			V Yes	~	0		onds	
	Kill Comm	and				-					PoD	& CoA K	ey Att	ributes		-		- +	×

2 Create a user with dynamic VLAN attributes in the RADIUS server. In the example screen below, 10 in the **Tunnel-Private-Group-ID** is the value of the dynamic VLAN of this user account.

SADIUS Manager			_	
File Service Help				
👗 Users 🔱 Groups 腪 Clients 🗔 Settings 🗐 Log	🔄 Active Sessions 📒 Dictionary	🖩 Reports 🔊 Count	ers 🗗 SQL	
Browse Users	User vlan10 (Enabled)			
All 🗸 User-Name 🗸 like 🗸 🗸 🗸	Attribute Ty	/pe	Value	
Username Group	User-Password C	heck	****	
vlan10 Default	Service-Type St	uccess-Reply	Administrative	
vlan20 Default	Tunnel-Type Si	uccess-Reply	VLAN	
	Framed-Protocol Si	uccess-Reply	802 PPP	
	Tunnel-Private-Group-ID St	uccess-Reply	10	
F0-DE-F1-F9-E2-45 Default 🗸 🔡 📑 🗹 🗙	Check V User-Password	~	()	∔ ×

- 3 Go to Site-wide > Configure > Switches > Authentication to create the authentication policy.
 - **3a** Select the authentication server in **Server type**.
 - **3b** Click +Add in Authentication server to create a new RADIUS server entry.
 - **3c** Enter the IP address of the external RADIUS server in Host. Enter the port of the RADIUS server for authentication (default 1812) in Port.

Enter a password (up to 32 alphanumeric characters) as the key to be shared between the external RADIUS server and the Nebula Device in **Secret**.

- 3d Click +Add in Authentication policy to create a new policy.
- 3e Enter a descriptive name for the policy in Name.
 Select 802.1x in Authentication type to validate access to the ports based on the user name and password provided by the client.
- 4 Go to Site-wide > Configure > Switches > Switch ports to bind the authentication policy to the Nebula Device access ports.
 - 4a Select the port(s) and click Edit.
 - 4b In the Update # port screen, select Access in Type. Select 802.1X/VLAN Assignment in Auth. policy. Then click Update.
- 5 Go to Site-wide > Configure > Switches > Switch ports to add the dynamic VLAN list to the allowed VLAN list of uplink ports.
 - 5a Select the uplink port and click Edit.
 - 5b In the Update # port screen, select Trunk in Type. Enter the dynamic VLAN(s) in Allowed VLANs. Then click Update.

3.25 Monitor Dynamic VLAN Using Event Logs (for Nebula Switches only)

Go to Site-wide > Monitor > Switches > Event log to monitor dynamic VLANs. The following are example dynamic VLAN-related event logs:

• User submits an incorrect 802.1X credential (wrong user name on the client port 'Port4').

Kewer Older)) 1 mat	ches in 212 Event I	ogs		💎 🕒 Export 🔻
Time 🔺	Priority	Switch	Category	Detail	Ę
2022-09-16 16:20:07	Notice	XGS1930-52HP	AAA	802.1x Authentication failure User-Name wronguser NAS-Port 4 - Port4	

• The dynamic VLAN attribute received is without a corresponding static VLAN (missing static VLAN 10 for the user name 'vlan10' on the client port 'Port4').

	> 2 m	atches in 161 Event	t logs		💎 🕒 Export 🕶
Time	 Priority 	Switch	Category	Detail	R
2022-09-16 15:18:20	Warning	XGS1930-52HP	AAA	802.1x - <mark>Static VLAN 10 does not exist [User-Name vlan10</mark> NAS-Port 4 - Port4]	
2022-09-16 15:18:20	Notice	XGS1930-52HP	AAA	802.1x Authorization failure [username: vlan10]	

• The Nebula Device cannot connect with an external RADIUS server.

Newer Older Older	13) matches i	n 55 Event la	ogs 💽 🖓	Export -
Time 👻	Switch	Category	Detail	Ð
2022-09-27 14:54:37	XGS1930-52HP	AAA	802.1x Authentication - retransmit EAPOL-START packet [User-Name wronguser] [NAS-Port 20 - Port20]	^
2022-09-27 14:47:15	XGS1930-52HP	AAA	RADIUS server 1 is unreachable	
2022-09-27 14:47:15	XGS1930-52HP	AAA	802.1x RADIUS server timeout [User-Name wronguser] [NAS-Port 20 - Port20]	
2022-09-27 14:46:45	XGS1930-52HP	AAA	802.1x Authentication - retransmit EAPOL-START packet [User-Name wronguser] [NAS-Port 20 - Port20]	

• The Nebula Device re-establishes connection with an external RADIUS server.

Kewer Older) (13) matches in	55 Event lo	gs	💎 🕒 Export 🔻
Time 🗨	 Switch 	Category	Detail	Ð
2022-09-27 14:54:37	XGS1930-52HP	AAA	8021x Authentication - retransmit EAPOL-START packet [User-Name wronguser] [NAS-Port 20 - Port20	0]
2022-09-27 14:45:07	XGS1930-52HP	AAA	RADIUS server 1 becomes reachable	

3.26 Register a Nebula Device (mobile router) in Nebula

To manage a Nebula Device (mobile router) and monitor its status in Nebula, do the following:

Nebula Configuration

- 1 Use the Setup Wizard to create an organization and a site, and add the Nebula Device. See Setup Wizard on page 46 for more information on using the wizard.
- 2 After configuring the Setup Wizard, close the Nebula Control Center welcome message to go to the Nebula portal dashboard. **0/1 Online** will show on **Mobile router**. This means that one Nebula Device (mobile router) is registered in Nebula but not yet online.

Dashboard						🗄 Customize
AP Status	Hit for AP Networ	rk IP Reputation Filter	Appliance Status	PoE Power	Security (24h)	Mobile router
O/1 Online O% Heavy loading		No Data	O/1 Online O% CPU Usage	No Switches	0	O/1 Online
📟 Switch Status	Utilization	Appliance Clients	s (by Usage)	(24h)	Wireless Clients	
No Switches	< O↓ bps > O↑ bps		No Data		0	
[] ₀ [] SSIDs (by Usage)	(24h)	Wireless Clients (by Usage)	Wireless Clients	Manufacturer (241)	Hit for Collaborativ Response	re Detect & 🛛 🗃
No	Data	No Data	Ne	o Data	No D	ata
						Ask Question

Insert the SIM Card

Insert the SIM card and do the hardware connections on the Nebula Device. Refer to the Nebula Device's QSG (Quick Start Guide) for more information.

Check the Connection in Nebula

1 Go to Site-wide > Dashboard. 1/1 Online will show in Mobile router. This means that one Nebula Device (mobile router) is registered in Nebula and is online.

Dashboard							B: Customize
AP Status	Hit for AP Networ	k IP Reputation Filter	(24h)	Appliance Status	PoE Power	Security (24h)	Mobile router
O/1 Online O _% Heavy loading		No Data			No Switches	0	1/1 Online
📟 Switch Status	WAN Utilization	Applia	nce Clients	(by Usage)	(24b)	Wireless Clients	
No Switches	< O↓ bps > O↑ bps			No Data		0	
SSIDs (by Usage)	(24h)	Wireless Clients (by Usage)	(24h)	Wireless Clients N	Manufacturer (24b)	Hit for Collaborativ Response	ve Detect & (7d)
No	Data	No Data		No) Data	No E	Data
							Ask Question

2 Click Mobile router to monitor the Nebula Device's status.

The Nebula Device goes into Nebula-managed mode automatically after it is successfully registered in the Nebula web portal and can be accessed there.

Note: Its login password and settings are then overwritten with what you have configured in the Nebula web portal. To access the Web Configurator when the Nebula Device is in Cloud mode, use the Nebula Local credentials password to login. The Local credentials: Password can be found in Site-wide > Configure > Site settings > Device configuration.

3.27 Using Collaborative Detection and Response (CDR)

Use CDR to block client IP traffic when an unsafe connection is detected and reaches the pre-set threshold. See on page 232 for more information.

To configure CDR, do the following:

1 Go to Site-wide > Configure > Collaborative detection & response. Click Enable to activate CDR (refer to the A part in the below figure).

Collaborative detection & respon	se A				
Policy	Category	Event type	Occurrence	Duration (Min	utes) Containment <mark>1</mark>
B	Malware	Malware detected	2	× [*] 60	× * Alert •
-	IPS	Vulnerability exploit detected	2	× * 10	× * Alert •
	Web Threats	Connections to malicious web sites detected	1	* 30	× Alert
ontainment					
General					
С	O Default	BUTTON Modern			
Logo		No logo	pload a logo		
Notification message	There are m	alicious network activities found on yo	ur device. Please conta	ct network administrator.	
Redirect external URL	To use custo	http://www.google.com ×	d the zip file and edit th	nem.	
Containment period	Download ti	ne customized captive portal page examples and the second s	nple.		
Block					
Block wireless client 1					
Quarantine Quarantine VLAN	Edit vlan	id: 44, 10.254.252.1/255.255.254.0			
kempt list					

- 2 Configure the criteria (Occurrence, Duration) and the Containment action (Alert, Block, Quarantine) for each Category (Malware, IDP, Web Threat) (refer to the B part in the above figure). See Table 39 on page 226 for more information.
- 3 Configure the containment alert (Theme), customized pop-up (Notification message) for the client blocked by CDR, and the (Containment Period) time interval (refer to the C part in the above figure).

- 4 In Block, set how long a suspect client should be blocked or quarantined (1 minute to 1 day (1,440 minutes)). Enter 0 to block a suspect client until released in Site-wide > Monitor > Containment list. In Quarantine, configure a VLAN in order to isolate traffic from suspect clients (refer to the D part in the figure for step 1).
- 5 Enter the IPv4 and/or MAC addresses of client device(s) that are exempt from CDR checking in **Exempt** list (refer to the E part in the figure for step 1).
- 6 To unblock a suspect client, go to Site-wide > Monitor > Containment list. Select a client, then
 - click Release to free the client from CDR containment, or
 - select an IPv4 address or MAC address, click Add to Exempt List and then click OK to release the client device from CDR containment. The client device's IP or MAC address is exempt from future CDR checking.

ntainmer	nt list						
Q IP add	iress, MAC addre:	15 👻					
Time	IP address	MAC add	User	Event type	Contain	Time rema	Connect
elease/Ad IP add MAC a	ld to Exempt List ress iddress						

3.28 Deploy With Nebula Native Mode (for Security Firewalls in Nebula)

Nebula native mode means the Security Firewall has a certificate (ZTP (Zero Touch Provision) or factory) to connect with Nebula.

- Note: Make sure the Nebula Device can connect to NCC through the Internet by using any of the following methods:
 - DHCP WAN, or
 - configure WAN through the Nebula Device's Web Configurator.

If you are adding a ZyWALL USG FLEX / ATP / USG20(W)-VPN Series Security Firewall (SF) with v5.10 and later firmware to a site, or if your SF has run ZTP before, do the following to deploy the SF using Nebula native mode:

- Reset the SF to factory-default settings
- Select the Nebula management mode.

Reset the SF to Factory-Default Settings

Note: You only need to do this if you have configured the SF before.

Press the **RESET** button on the SF panel (see the SF user's guide for more information).

Note: Apply the factory-default settings on the SF before switching to cloud mode. Only the following two settings can be changed after resetting:

- Default administrator account password
- WAN settings

Select the Nebula Management Mode

- 1 Log into the SF Web Configurator (see the SF user's guide for more information). When you log into the Web Configurator, the **Initial Setup Wizard** screen displays.
- 2 Select Nebula Mode and click Next.

c	On Premises Mode	Nebula Mode
	MR	
		nebula
Choose On Premis Device directly usi Configurator or the	es Mode to manage your Zyxel ng either the browser-based Web e Command Line Interface (CLI).	Choose Nebula Mode to manage your Zyxel Device remotely using Nebula Control Center (NCC). Select this mode if you want to configure and monitor one or more Zyxel Devices through the cloud.

3 Configure the WAN settings and click Next.

Connect to Internet (WAN) > 1 2 Internet Access - First WAN Interface VLAN Togged VLAN ID: C IF Parameters Encopsulation: Etheme! MTU: 1500 Bytes IP Address Assignment: PAddress Assignment: Auto DHCP Option 60:	Initial Setup Wizard			
ISP Setting I have two ISPs Internet Access - First WAN Interface VLAN Togged VLAN ID: ISP Parameters Encopsulation: Hame MTU: ISO IP Address Assignment First WAN Interface: ge2 Zone: WAN IP Address Assignment: Auto MTO IP Address Assignment: Auto MTO MTO MTO MTO MTO MTO MTO MTO	Connect to Internet (WAN)	> Add Device		
ISP Setting I have two ISPs Internet Access - First WAN Interface VLAN Togged VLAN ID: (1-4080) ISP Parameters Encopsulation: Etherme MTU: 1500 Bytes IP Address Assignment First WAN Interface: ge2 Zone: WAN IP Address Assignment: Auto IP Address Assignment: IP Address Assignment: IP Address Assignment: IP Address Assignment: IP Address Assignment: IP Address Assignment: IP Address Assignment: IP Address Assignment				
Internet Access - First WAN Interface VLAN Togged VLAN ID: (1-4080) ISP Parameters Encopsulation: Etherme MTU: 1500 Bytes IP Address Assignment: First WAN Interface: ge2 Zone: WAN P DHCP Option 60:	ISP Setting			
Internet Access - First WAN Interface	I have two ISPs			
VLAN Togged VLAN ID: ISP Parameters Encapsulation: Etherme MTU: 1500 Bytes IP Address Assignment IP Address Assignment: Auto IP Address Assignment: Auto IP Address Assignment: Auto IP Address Assignment: Auto IP Address Assignment: IP Address	Internet Access - First WA	N Interface		
VLAN ID: I=4080) ISP Parameters Encapsulation: Encapsulation: Isthermet MTU: 1500 Bytes IP Address Assignment First WAN Interface: ge2 Zone: WAN IP Address Assignment: Auto DHCP Option 60: Image: Content of the second	VLAN Togged			
ISP Parameters Encapsulation: Etherme MTU: 1500 Bytes IP Address Assignment: ge2 Zone: WAN IP Address Assignment: Auto DHCP Option 60: Ethermet	VLAN ID:		< >	(1-4080)
Encapsulation: Etherme v MTU: 1500 Bytes IP Address Assignment First WAN Interface: ge2 Zone: WAN v IP Address Assignment: Auto v DHCP Option 60:	ISP Parameters			
MTU: 1500 Bytes IP Address Assignment First WAN Interface: ge2 Zone: WAN Y IP Address Assignment: Auto Y DHCP Option 60:	Encapsulation:	Ethernet	*	
IP Address Assignment First WAN Interface: ge2 Zone: WAN IP Address Assignment: Auto DHCP Option 60:	MTU:	1500]	Bytes
First WAN Interface: ge2 Zone: WAN IP Address Assignment: Auto DHCP Option 60:	IP Address Assignment			
Zone: WAN IP Address Assignment: Auto DHCP Option 60:	First WAN Interface:	ge2		
IP Address Assignment: Auto	Zone:	WAN	*	
DHCP Option 60:	IP Address Assignment:	Auto	~	
	DHCP Option 60:			

4 Click Connection Test to check that you can access the Internet and then click Next.

Initial Setup	Wizard		E		
	Initial Setup Wizard				
	Connect to Internet (WAN)	Add Device			
	Congratulations. The Inter Summary of Internet Acce	net Access wizard is completed. sss configuration:			
	First Setting				
	VLAN ID:				
	Encapsulation:	Ethernet			
	First WAN Interface:	wani			
	IP Address Assignment:	Static			
	IP Address:	192.168.69.35			
	IP Subnet Mask:	255.255.255.0			
	Gateway IP Address:	192.168.69.1			
	First DNS Server:				
	Second DNS Server:				
	Connection Test				
	Note: Once you complete mode menu.	e WAN configuration, you cannot go back to the initial management			
		< Back Next >			

5 Click Go to Nebula.
Connect to Internet (WAN) > Add Device 1 2 You can use the Nebula portal or the app to manage your Zyxel Device remotely using Nebula Control Center (NCC). Nebula portal: 1. Log into the Nebula portal (http://nebula.zyxel.com) with your myZyxel account. 2. Follow the wizard to create an organization and a site for your Zyxel Device. 3. Enter the MAC address and serial number (S/N) on the device label when prompted. 4. Click Go To Nebula to finish the Wizard Nebula app: 1. Download the Nebula Mobile app from App Store or Google Play. 2. Run the app and select a site for your Zyxel Device. 3. Scan the QR code below to register the Zyxel Device using its MAC address and serial number.	Initial Setup Wizard	Î				
You can use the Nebula portal or the app to manage your Zyxel Device remotely using Nebula Control Center (NCC). Nebula portal: 1. Log into the Nebula portal (<u>http://nebula.zyxel.com</u>) with your myZyxel account. 2. Follow the wizard to create an organization and a site for your Zyxel Device. 3. Enter the MAC address and serial number (S/N) on the device label when prompted. 4. Click Go To Nebula to finish the Wizard Nebula app: 1. Download the Nebula Mobile app from App Store or Google Play. 2. Run the app and select a site for your Zyxel Device. 3. Scan the QR code below to register the Zyxel Device using its MAC address and serial number.	Connect to Internet (WAN) > Add Device					
You can use the Nebula portal or the app to manage your Zyxel Device remotely using Nebula Control Center (NCC). Nebula portal: 1. Log into the Nebula portal (http://nebula.zyxel.com) with your myZyxel account. 2. Follow the wizard to create an organization and a site for your Zyxel Device. 3. Enter the MAC address and serial number (S/N) on the device label when prompted. 4. Click Go To Nebula to finish the Wizard Nebula app: 1. Download the Nebula Mobile app from App Store or Google Play. 2. Run the app and select a site for your Zyxel Device. 3. Scan the QR code below to register the Zyxel Device using its MAC address and serial number.						
Nebula portal: 1. Log into the Nebula portal (http://nebula.zvxel.com) with your myZyxel account. 2. Follow the wizard to create an organization and a site for your Zyxel Device. 3. Enter the MAC address and serial number (S/N) on the device label when prompted. 4. Click Go To Nebula to finish the Wizard Nebula app: 1. Download the Nebula Mobile app from App Store or Google Play. 2. Run the app and select a site for your Zyxel Device. 3. Scan the QR code below to register the Zyxel Device using its MAC address and serial number.	You can use the Nebula portal or the app to manage your Zyxel Device remotely using Nebula Control Center (NCC).	1				
 Log Into the Nebula portal (http://nebula.zvxel.com) with your myZyxel account. Follow the wizard to create an organization and a site for your Zyxel Device. Enter the MAC address and serial number (S/N) on the device label when prompted. Click Go To Nebula to finish the Wizard Nebula app: Download the Nebula Mobile app from App Store or Google Play. Run the app and select a site for your Zyxel Device. Scan the QR code below to register the Zyxel Device using its MAC address and serial number. 	Nebula portal:					
 2. Follow the wizard to create an organization and a site for your Zyxel Device. 3. Enter the MAC address and serial number (S/N) on the device label when prompted. 4. Click Go To Nebula to finish the Wizard Nebula app: Download the Nebula Mobile app from App Store or Google Play. Run the app and select a site for your Zyxel Device. S. Scan the QR code below to register the Zyxel Device using its MAC address and serial number. 	1. Log into the Nebula portal (http://nebula.zyxel.com) with your myZyxel account.					
 3. Enter the MAC address and serial number (S/N) on the device label when prompted. 4. Click Go To Nebula to finish the Wizard Nebula app: Download the Nebula Mobile app from App Store or Google Play. Run the app and select a site for your Zyxel Device. Scan the QR code below to register the Zyxel Device using its MAC address and serial number. 	2. Follow the wizard to create an organization and a site for your Zyxel Device.					
 4. Click Go To Nebula to finish the Wizard Nebula app: Download the Nebula Mobile app from App Store or Google Play. Run the app and select a site for your Zyxel Device. Scan the QR code below to register the Zyxel Device using its MAC address and serial number. 	3. Enter the MAC address and serial number (S/N) on the device label when prompted.					
Nebula app: 1. Download the Nebula Mobile app from App Store or Google Play. 2. Run the app and select a site for your Zyxel Device. 3. Scan the QR code below to register the Zyxel Device using its MAC address and serial number.	4. Click Go To Nebula to finish the Wizard					
 Download the Nebula Mobile app from App Store or Google Play. Run the app and select a site for your Zyxel Device. Scan the QR code below to register the Zyxel Device using its MAC address and serial number. 	Nebula app:					
2. Run the app and select a site for your Zyxel Device. 3. Scan the QR code below to register the Zyxel Device using its MAC address and serial number.	1. Download the Nebula Mobile app from App Store or Google Play.					
3. Scan the QR code below to register the Zyxel Device using its MAC address and serial number.	2. Run the app and select a site for your Zyxel Device.					
	3. Scan the QR code below to register the Zyxel Device using its MAC address and serial number.					
見た認識が見						
		-				

Nebula Configuration

1 You will be redirected to the Nebula portal. Click Get Started.

ZYXEL	
	Prote Overview of Zvvel Nebula Control Center.
Welcome to Nebula Cloud!	NETWORKS
Begin your journey towards one easy network and security management.	
Get Started	Overview of Nebula Control

2 Use the Setup Wizard to create an organization and a site, and add the Nebula Device. See Setup Wizard on page 46 for more information on using the wizard.

Note: Make sure to select **Nebula native mode** as the **Deployment Method** in the Setup Wizard.

Nebula native mode may be un-clickable (grayed-out) if your Nebula Device do NOT support it. Select **Zero Touch Provision mode** instead (see Section 2.1.7 on page 51 for more information on the ZTP deployment method).

			Exit Wize
	Deployment Me	thod	
Model Name	USG FLEX 500	Show device information	
Deployment Method 🟮			
Nebula native mode			
1. Connect your com	puter to the GW LAN port and conne	ct WAN port to a modern or router the	xt has
2. Login GW GUI and	configure your WAN connection sett	ings.	
			AULTIN
ZYXEL		<u>n n n n n</u>	
	*	*	
Front			
	Ę	E C	
	0	0	
	\sim		
	65		
	\bigcirc		
	Internet	PC	
C Zero Touch Provision	mode		
	Back		

3 After configuring the Setup Wizard, close the Nebula Control Center welcome message to go to the Nebula portal dashboard. 1/1 Online will show on Appliance Status. This means that one SF is registered in Nebula and is online.

Dashboard							🗄 Customize	
AP Status	Hit for AP Network IP Reputation Filter			Appliance Status	PoE Power	Security (24h)	Mobile router	
O/1 Online O% Heavy loading		No Data	1/1 Online No Switche 10% CPU Usage		0	O/1 online		
📟 Switch Status	Utilization	App	liance Clients	(by Usage)	(24b)	Wireless Clients		
No Switches	<			No Data	No Data			
000 SSIDs (by Usage)	(245)	I Wireless Clients (by Usage)	(24h)	Wireless Clients I	Manufacturer (24b)	Hit for Collaborati Response	ve Detect & (78)	
No Data		No Data	No Data		b Data	No Data		
Wireless Clients C	os @	APs (by Usage)		(24h)	AP Traffic		(24h) Ask Question	

3.29 Configure DHCP Domain Name (for Security Firewalls in Nebula)

You can configure a DHCP domain name to map to a specific IP address on a specific interface. For this example, to add a domain name for the IP address 192.168.8.1 in the **Ian1** interface, do the following.

1 Go to Firewall > Configure > Interface. Click the Edit icon for the Ian1 interface to open the Site-wide > Configure > Firewall > Interface > LAN interface configuration screen.

		WAN	Interface			
Status	IP address	Subnet mask	VLAN ID	Port group		
				WAN Group 1		- Z
		LAN	l interface			
Status	IP address	Subnet mask	VLAN ID Port grou	р	Guest	
	192.168.8.1	255.255.255.0	LAN Gro	oup1 .		
	192168.9.1	255.255.255.0	LAN Gr	pup 2		1
	Status Status	Status IP address Status IP address Status IP address 192168.81 192168.91	Status IP address Subnet mask	Status IP address Subnet mask VLAN ID Image: Constraint of the state of the s	Status IP address Subnet mask VLAN ID Port group Image: Status Image: Sta	Status IP address Subnet mask VLAN ID Port group WAN Group 1 WAN Group 1 WAN Group 1 Status IP address Subnet mask VLAN ID Port group Guest 192168.81 255.255.0 LAN Group 1 Image: Comp 1 Image: Comp 1 Image: Comp 1

2 Click ADVANCED OPTIONS. Then click +Add new to open the Site-wide > Configure > Firewall > Interface > LAN interface configuration: DHCP option screen.

LAN interface configuration		×
	ADVANCED OPTIONS	*
DHCP extended options		
First WINS server	×	
Second WINS server	×	
PXE server	×	
PXE Boot loader file	×	
Default gateway	×	
+ Add new		
IGMP proxy		
	IGMP upstream	
	O IGMP downstream	
		•
		Close OK

3 Select User defined as the DHCP Option that you want to add in the DHCP packets sent through the LAN interface. Select TEXT for the Type, enter a descriptive Name to identify and the Code number of the selected DHCP option (15, for setting the Domain Name). See https://www.iana.org/assignments/bootp-dhcp-parameters/bootp-dhcp-parameters/bootp-dhcp-parameters.xhtml for the list of code numbers. Enter the DNS domain name of the IP address in Value. Then click OK.

DHCF	option		×
	Option	User defined	•
	Name	DomainName	×
	Code	15	× (1-254)
	Туре	TEXT	-
	Value	cs.com	×
			Close

4 A new user-defined DHCP option appears in LAN interface configuration. Click OK.

LAN in	terface configurat	ion:				×	
DHC	P extended options						*
	First WINS server				×		
	Second WINS server				×		
	PXE server				×		
	PXE Boot loader file				×		
	Default gateway				×		
	Name	Code	Туре	Value			
	DomainName	15	TEXT	cs.com	2 💼		T
	+ Add new						l
IGM	IP proxy	3					
			O IGMP upst				
)	O IGMP dow				
							×
						Close OK	

5 Go to Site-wide > Configure > Firewall > Firewall settings and click +Add in DNS to create an Address Record. This record specifies the mapping of a Fully-Qualified Domain Name (FQDN) to an IP address.

Firewall settings	
DNS	
Address Record	
+ Add	
Domain Zone Forwarder	
+ Add	

6 Enter the FQDN (cs.com) and IP Address (192.168.8.1). Then click Save to finish mapping the FQDN to the IP address.

rewall settings		
Address Record		
FQDN	IP Address	
cs.com × *	192168.8.1 × *	.
+ Add Domain Zone Forwarder		
+ Add		

To check if the domain name configuration is successful.

- 1 Connect a computer to the lan1 interface (with IP address 192.168.8.1).
- 2 Run the **Command Prompt** and enter **ipconfig**. Check the value for **Connection-specific DNS Suffix** to confirm.



PART II Manage by Site Deployment

CHAPTER 4 Site-wide

4.1 Dashboard

If a site is created and selected, the **Dashboard** is always the first menu you see when you log into the NCC. You can also click **Site-wide** > **Dashboard** to access this screen.

It shows the status and information for all types of Nebula Devices connected to the selected site by default.

Note: The banner N Switches are currently protected by Auto Configuration Recovery will display when the Nebula Switch(es) is locked by NCC. Click N Switches to go to Sitewide > Monitor > Switches for more information.

Click **Customize** to show the **Widget**, **Reset** and **Close** buttons. You can then rearrange widgets by selecting a block and holding it to move around. You can also click the **Widget** button to collapse, add and close individual widgets. Click **Reset** to return the widget settings to the defaults.

Site-wide > Doshboard						
Dashboard					88	Widget 🖒 Reset Close
e	e st This site has no mobile router.	Online CPU	1/1 Online 1 4% Loading 0	Access point Da Online 1 Des Loading Or	A Wireless C) 3
APs by Usage		(246)	()00) SSIDs by Usage		(24)	PoE power
O Clients	2.6 IB Iffic Zynapio2			No Dota		1% cansumed 375w Total
APs application usage			(24)	🔊 Wireless clients by usag	e	(a))
	No Data				No Data	
Wireless clients by OS No Data	9	Clients by manufactur	yxel Communications C 6	Contraction of the security gateway client Annr-052220 N S XM XM XM	s by usage 2014P 19192 19195 1920 1920	70.3 MB 79 MB 55 MB 54 MB 53 MB
WANT Utilization	Security gateway network a	oplications		(24)	Wireless Clients	Mobile router
9.4↓ < > 38.8↑ ^{Kbps}	0 bytes		No Data		0	No Mobile router
Security gateway	Switch status	APs Status	Threat protection by CN	IP services	9	
I/1 Online 1%. CPU Usage	I/1 Online On Loading	I/1 Online O _% Heavy loading		No Data		

Figure 22 Site-wide > Dashboard

LABEL	DESCRIPTION
APs status	This shows the number of assigned and connected Nebula access points, and what percentage of the access points become overloaded, that is, the number of online access points that exceed the maximum client device number (in Site-wide > Configure > Access points > Traffic shaping) by total number of online access points in the site.
Wireless clients	This shows the number of WiFi clients currently connected to the managed access points.
Switch status	This shows the number of Nebula Switches assigned and connected, and what percentage of the Switches become overloaded, that is, the number of online Nebula Switches that exceed 70% of their upstream bandwidth by total number of online Nebula Switches in the site.

LABEL	DESCRIPTION
PoE power	This shows the total PoE power budget on the Switch and the current amount of power consumed by the powered devices.
Security router / Firewall / Security Gateway status	This shows the number of Nebula Security Appliances assigned and connected, and what percentage of the Security Appliance's processing capability is currently being used if the CPU goes over 93% usage.
WAN utilization	This shows the data rate of inbound/outbound traffic in Kbps (kilobits per second) or Mbps (megabits per second) that has been transmitted through the WAN interface. If the Security Appliance supports multiple WAN interfaces and more than one are active, use the arrow to switch and view the throughput of each WAN interface.
Security alert	This shows the total number of the latest alerts sent to the administrator in the last 24 hours.
Mobile router	This shows the number of Nebula mobile routers assigned and connected.
Security router / Firewall / Security Gateway network applications	This shows the top ten applications used by the Nebula Security Appliance in the past 24 hours.
Security router / Firewall / Security Gateway clients by usage	This shows the top five clients of the Nebula Security Appliance with the highest percentage of bandwidth usage in the past 24 hours.
Wireless clients	This shows the number of WiFi clients connected (clients of the access points only).
SSIDs by usage	This shows the top five SSIDs with the highest percentage of bandwidth usage in the past 24 hours. You can click a WiFi network name to go to the Site-wide > Monitor > Access Point > Summary report screen.
Wireless clients by usage	This shows the top five WiFi clients (clients of the access points only) with the highest percentage of bandwidth usage in the past 24 hours. You can click a client's name to go to the Site-wide > Clients: Client list screen.
Clients by manufacturers	This shows the top five manufacturers of WiFi client devices in the past 24 hours. You can click a manufacturer name to go to the Site-wide > Clients screen and view the client devices which are made by the manufacturer.
Collaborative detection & response hit	This shows the total number of malicious traffic detected from wired and WiFi clients that are blocked and quarantined using Collaborative Detection & Response (CDR) in the past 7 days.
Wireless clients by OS	This shows the top five operating systems used by WiFi client devices in the past 24 hours. You can click an operating system to go to the Site-wide > Clients screen and view the client devices which use this operating system.
APs by usage	This shows the top five managed access points with the highest percentage of bandwidth usage in the past 24 hours. This also shows the number of WiFi clients associated with the access points. You can click an access point's name to go to the Site-wide > Devices > Access Points : Access Points Details screen.
APs application usage	This shows the usage statistic of the top ten applications used in the site in the past 24 hours.
APs locations	This shows the locations of access points on the Google map.
Threat protection by CNP services	This shows the total number of times packets coming from an IPv4 address with a bad reputation occur and the number of times connection attempts to an IPv4 address with a bad reputation occur in the past 24 hours.

Table 12 Site-wide > Dashboard (continued)

4.2 Topology

Use this screen to view the links between Nebula Devices in the site. Click **Site-wide** > **Topology** to access this screen.

The icon of a node in the network topology indicates its Nebula Device type and the color shows whether the Nebula Device is online (green), has alerts (amber), or is offline (red).

Move the pointer over a node to view detailed Nebula Device information, such as its name, model number, number of connected clients, and MAC address. Click **Reboot** to restart the Nebula Device.

Move the pointer over a link to view link details, such as type (Ethernet or wireless mesh), speed, and data usage from the past 24 hours. If the link is supplying power to a node using Power over Ethernet (PoE), you can click **Reset** to perform a power cycle on the port. This action temporarily disables PoE and then re-enables it, in order to reboot connected PoE devices.

Enable Label all devices to show Nebula Device information, such as MAC address in the network topology diagram.

Enable Show redundant links to display the secondary connection between two nodes, if any.

Enable **Show other devices** to also display the Nebula Devices that are connected to your network but cannot be identified by the NCC. This on/off switch is configurable only when there is a non-Nebula Device installed in the network and detected by the NCC through LLDP packets.

Zyxel device is a device manufactured by Zyxel but not registered at the NCC or unable to work in Nebula cloud management mode.



Figure 23 Site-wide > Topology

4.3 Devices

Use the **Devices** menus to check Nebula Device information, client information, event log messages and summary report for Nebula Devices in the selected site.

4.3.1 Access Points

This screen allows you to view the detailed information about a Nebula Device in the selected site. Click **Site-wide > Devices > Access points** to access this screen.

Figure 24	Site-wide >	Devices >	Access	points
-----------	-------------	-----------	--------	--------

c <mark>tion →</mark>) selec	Tag • ted in (Move APR 6 access point	Q Sear	ch	•		•	Online 🖲 Offli	ne 🖲 Alert 🖲 Offline mo	ore than 6 days 💙 🄁	Export
Sto	itus	Name	LAN IP	Remote AP	AP Role Capab	Public IP	Model	Client	Current client	MAC address	Usag
	1	shawn-620-6E	173.16.2.107	Disabled	Remote AP	210.61.209.2	WAX620D-6E	7	2	10:71:B3:1B:73:1C	0 byte
	6	B8:EC:A3:DD:19			Standard		NWA50AX	0	0	B8:EC:A3:DD:19:1C	0 byte
2	1	BC:CF:4F:56:B	173.16.2.93	Disabled	Remote AP	210.61.209.2	WAX650S	13	6	BC:CF:4F:56:BD:6D	1.34 G
	1	shawn seat2			Standard		NWA90AX PRO	0	0	FC:22:F4:91:EF:DC	0 byte
	1	Marketing2	173.16.2.92		Standard	210.61.209.2	NWA50AX PRO	10	3	FC:22:F4:91:EF:82	954.71
	1	marketing		Disabled	Remote AP		WAX510D	0	0	D8:EC:E5:78:EC:BE	0 byte
	-			_							

Table 13 Site-wide > [Devices > Access points
------------------------	-------------------------

LABEL	DESCRIPTION		
Access points	Select to view device information and connection status in the past two hours, day, week or month.		
5	Click this button to reload the data-related frames on this page.		
Action	Perform an action on the selected Nebula Devices.		
Reboot	Select this to restart the Nebula Device.		
Upgrade	Select this to upgrade the firmware on the Nebula Device.		
Change PSK	Select this to generate a random Pre-Shared Key, or use a custom Pre-Shared Key. This allows a user to access the WiFi network through the Nebula Device. Update programmable pre-shared key X Re-generate a random pre-shared key X		
	Custom pre-shared key WPA2 pre-shared-key supports 8-63 characters Cancel Key Note: Programmable SSID must be enabled in Site-wide > Configure > WiFi SSID.		

LABEL	DESCRIPTION
Tag	Select one or multiple Nebula Devices and click this button to create a new tag for the Nebula Devices or delete an existing tag.
Move	Select one or multiple Nebula Devices and click this button to move the Nebula Devices to another site or remove the Nebula Devices from the current site.
AP Role	Select one or multiple Nebula Devices and click this button to enable or disable the Remote AP feature.
	Remote Nebula Device enables the site's Security Appliance to connect to the Nebula Device through a secure VPN tunnel. This allows you to set up VPN-enabled WiFi Nebula Devices in remote locations, such as in a branch office or at home. Clients connected to these Nebula Devices can securely access your network through the VPN tunnel.
	Note: Enabling Remote Nebula Device automatically enables Ethernet and wireless storm control on the Nebula Device.
Search	Specify your desired filter criteria to filter the list of Nebula Devices.
access points	This shows the number of Nebula Devices connected to the site network.
Export	Click this button to save the access point list as a CSV or XML file to your computer.
*	Click this to select all the rows in this table.
Status	This shows the status of the Nebula Device.
	 Green: The Nebula Device is online and has no alerts. Amber: The Nebula Device has alerts. Red: The Nebula Device is offline. Gray: The Nebula Device has been offline for 7 days or more. (a): The Nebula Device is acting as a repeater.
	For example, an alert is created and the status color is amber when the Nebula Device is transmitting data at 100 Mbps in full duplex mode or when the Nebula Device is in a Limited Power mode .
Name	This shows the descriptive name of the Nebula Device.
LAN IP	This shows the local (LAN) IP address of the Nebula Device.
Remote AP	This shows whether the Remote Nebula Device function is Enabled or Disabled .
2.4GHz	This shows the number of WiFi clients in the 2.4 GHz band.
5GHz	This shows the number of WiFi clients in the 5 GHz band.
6GHz	This shows the number of WiFi clients in the 6 GHz band.
AP Role Capability	This displays whether the Nebula Device can act as a remote Nebula Device (Remote AP) or not (Standard AP).
Public IP	This shows the global (WAN) IP address of the Nebula Device.
Model	This shows the model number of the Nebula Device.
Client	This shows how many clients are connected to the Nebula Device within the specified time period.
Current client	This shows how many clients are currently connecting to the Nebula Device.
MAC address	This shows the MAC address of the Nebula Device.
Channel	This shows the channel ID the Nebula Device is using.
Channel Utilization 2.4GHz	This shows the percentage of the 2.4 GHz channel ID usage.
Channel Utilization 5GHz	This shows the percentage of the 5 GHz channel ID usage.
Channel Utilization 6GHz	This shows the percentage of the 6 GHz channel ID usage.

Table 13 Site-wide > Devices > Access points (continued)

LABEL	DESCRIPTION	
Usage	This shows the amount of data consumed by the Nebula Device's clients.	
% Usage	This shows the percentage of the Nebula Device's data usage.	
Description	This shows the user-specified description for the Nebula Device.	
Tag	This shows the user-specified tag for the Nebula Device.	
Serial number	This shows the serial number of the Nebula Device.	
Configuration status	This shows whether the configuration on the Nebula Device is up-to-date.	
Connectivity	This shows the access point connection status.	
	The red time slot indicates the connection to the NCC is down, and the green time slot indicates the connection is up. Move the cursor over a time slot to see the actual date and time when an Nebula Device is connected or disconnected.	
Ethernet 1	This shows the speed and duplex mode of the Ethernet connection on the Nebula Device's up- link port. It shows Down if the Nebula Device is connected to a mesh controller wirelessly.	
Neighbor Info	This shows the LLDP information received on the up-link port.	
Production information	This shows the production information of the Nebula Device.	
Нор	This shows the hop count of the Nebula Device. For example, "1" means the Nebula Device is connected to a mesh controller directly. "2" means there is another mesh extender between this Nebula Device and the mesh controller.	
IP type	This shows whether the IP address was assigned automatically (DHCP), or manually (Static IP).	
Uplink AP	This shows the role and descriptive name of the Nebula Device to which this Nebula Device is connected wirelessly.	
	When Smart Mesh is enabled and the mesh extender losses connection to the mesh controller, click Reconnect to re-establish connection.	
	Note: Make sure to enable Manual uplink in Site-wide > Devices > Access points: Details > Status > Smart mesh > Edit. You also need to specify the mesh controller in select an AP. See Table 14 on page 161 for more information.	
Uplink signal	Before the slash, this shows the signal strength the uplink Nebula Device (a mesh controller or a mesh extender) receives from this Nebula Device (in repeater mode). After the slash, this shows the signal strength this Nebula Device (in repeater mode) receives from the uplink access point.	
Uplink Tx/Rx rate	This is the maximum transmission/reception rate of the mesh controller or mesh extender to which the Nebula Device is connected.	
Wireless bridge	This shows whether wireless bridge is enabled on the Nebula Device.	
	For more information about wireless bridge, see Section 5.1.2.2 on page 235.	
Uplink	This shows whether the Nebula Device is connected to the gateway through a wired Ethernet connection or WiFi connection.	

Table 13 Site-wide > Devices > Access points (continued)

LABEL	DESCRIPTION
Power mode	This shows the Nebula Device's power status.
	Full – the Nebula Device receives power using a power adapter and/or through a PoE switch/ injector using IEEE 802.3at PoE plus. The PoE device that supports IEEE 802.3at PoE Plus can supply power of up to 30W per Ethernet port.
	Limited – the Nebula Device receives power through a PoE switch/injector using IEEE 802.3af PoE even when it is also connected to a power source using a power adapter. The PoE device that supports IEEE 802.3af PoE can supply power of up to 15.4W per Ethernet port.
	When the Nebula Device's power mode is Limited , the Nebula Device throughput decreases and has just one transmitting radio chain.
	It always shows Full if the Nebula Device does not support power detection.
Firmware availability	This shows whether the firmware on the Nebula Device is Up to date , there is firmware update available for the Nebula Device (Upgrade available), or a specific version of firmware has been installed by Zyxel customer support (Locked).
Firmware status	This shows whether the firmware installed on the Nebula Device is up-to-date.
Firmware type	This shows Stable when the installed firmware may not have the latest features but has passed Zyxel internal and external testing.
	This shows Latest when the installed firmware is the most recent release with the latest features, improvements, and bug fixes.
	This shows General Availability when the installed firmware is a release before Latest , but is still undergoing Zyxel external testing.
	This shows Dedicated when the installed firmware is locked and Zyxel support is monitoring. Contact Zyxel customer support if you want to unlock the firmware in order to upgrade to a later one.
	This shows Beta when the installed firmware is a release version for testing the latest features and is still undergoing Zyxel internal and external testing.
	This shows N/A when the Nebula Device is offline and its firmware status is not available.
Current version	This shows the firmware version currently installed on the Nebula Device.
Remote AP VPN	This shows which VPN the Remote Nebula Device tunnel is configured to use.
	If Remote Nebula Device is disabled, this field shows Disconnected.
R	Click this icon to display a greater or lesser number of configuration fields. For faster loading of data, select only the configuration fields listed that do NOT take a long time to fetch data.

Table 13 Site-wide > Devices > Access points (continued)

4.3.1.1 Access Point Details

Click a Nebula Device entry in the **Site-wide** > **Devices** > **Access points** screen to display individual Nebula Device statistics.



Figure 25 Site-wide > Devices > Access points: Details Part 1



Figure 26 Site-wide > Devices > Access points: Details Part 2

The following table describes the labels in this screen.

Table 14	Site-wide > Devices >	> Access points: Details
----------	-----------------------	--------------------------

LABEL	DESCRIPTION			
C	Click this button to reload the data-related frames on this page.			
Configuration				
Click the edit configuration icon to change the Nebula Device name, description, tags, load balancing, and				

Click the edit configuration icon to change the Nebula Device name, description, tags, load balancing, and address. You can also move the Nebula Device to another site or remove.

LABEL	DESCRIPTION				
Remote AP	Click this to enable or disable the Remote AP feature.				
	Remote AP enables the site's Security Appliance to connect to the Nebula Device through a secure VPN tunnel. This allows you to set up VPN-enabled WiFi Nebula Devices in remote locations, such as in a branch office or at home. Clients connected to these Nebula Devices can securely access your network through the VPN tunnel.				
	With the Remote AP feature (in the Secure WiFi license) the connection is from the Nebula Device to a managed access point using NVGRE (Network Virtualization using Generic Routing Encapsulation) over IPSec tunnel. This encapsulates and encrypts traffic from the remote access point to the Nebula Device. The clients connected to the remote access point do not need IPSec client software installed.				
	Note: Enabling Remote AP automatically enables Ethernet and wireless storm control on the Nebula Device. At the time of writing, Ethernet Secure Tunnel Setting for Remote AP Setting is available for WAC500H only.				
	Remote AP Setting X				
	Local SSID Setting				
	Enabled SSID Security Mode Key Band				
	1				
	2 X * WPA2-Perso • @ * Concurrent •				
	Ethernet Secure Tunnel Setting 🕴 Boot				
	Enabled Tunnel to gateway Interface				
	Cancel Save				
	Configure and enable up to two SSID(s) in Local SSID Setting. WiFi clients connected to these SSIDs are forwarded to the local network of the remote site. The Local SSID Setting are different from the SSIDs you configured in Site-wide > Configure > Access points > SSID advanced settings. See Section 5.3.1 on page 247 for the description of the fields.				
	Select from the available LAN or VLAN interface in Tunnel to gateway interface to enable it, and click Save .				
Name	This shows the descriptive name of the Nebula Device.				
MAC address	This shows the MAC address of the Nebula Device.				
Serial number	This shows the serial number of the Nebula Device.				
Description	This shows the user-specified description for the Nebula Device.				
Address	This shows the user-specified address for the Nebula Device.				
Тад	This shows the user-specified tag for the Nebula Device.				
Load balancing	This shows the load balancing group name that the Nebula Device belongs (up to two groups per access point). Nebula Devices in the same group should be within the proximity. This allows them to share the load.				
Status					

Table 14	Site-wide >	Devices >	Access	noints	Details	(continued)
		Devices -	ACCESS	points.	Deruis	(COI IIII IOEU)

LABEL	DESCRIPTION			
LAN IP	This shows the local (LAN) IP address of the Nebula Device. It also shows the IP addresses of the gateway and DNS server.			
	Click the edit icon to a number and tagging s	open a screen where you can change the IP addresses, VL setting.	AN ID	
	Set IP Address	;	×	
	IP type	Static IP 💌		
	IP	×		
	Management VLAN ID	1 (1~4094)		
		• Untagged Tagged		
	Subnet mask	×		
	Gateway	X		
	Primary DNS	×		
		Close Of	ĸ	
Public IP	This shows the global (WAN) IP address of the Nebula Device.		
Usage	This shows the amount	of data consumed by the clients.		
Current clients	This shows the number its details.	of clients which are currently connecting to the Nebula D	evice and	
Topology	Click Show to go to the	e Site-Wide > Topology screen. See Section 4.2 on page 15	55.	
Neighbor info	This shows the LLDP info	ormation received on the up-link port.		
Link	This shows the speed a ports.	nd duplex mode of the Ethernet connection on the Nebula	a Device's	
	It shows Uplink: Wireles controller wirelessly.	ss if the access point is an mesh extender and connected	to a mesh	
	A warning icon display	rs when the Nebula Device is running at 100 Mbps or a low	ver speed.	
Ports	This is available only fo port (except the uplink	r the Nebula Device that has one or more than one Etherr < port).	net LAN	
	This shows the PVID of belongs. See Section 5	the LAN port and the ID number of VLANs to which the LAN 5.3.7 on page 271 for how to change the port's VLAN settin	N port ngs.	
Storm control	Storm control limits the packets received per s number of allowable b subsequent packets an and/or DLF packets in	number of broadcast, multicast and destination lookup for second on the Nebula Device's Ethernet ports. When the r proadcast, multicast and/or DLF packets is reached per se re discarded. Enabling this feature reduces broadcast, mu your network.	ailure (DLF) maximum cond, the ulticast	
Channel (Band)	This shows the channel Device.	I ID and WiFi frequency band currently being used by the I	Nebula	
Channel utilization	This shows the percent	age of the channel ID usage.		

Table 14 Site-wide > Devices > Access points: Details (continued)

LABEL	DESCRIPTION				
Power mode	This shows Full when the Nebula Device receives power directly through a power outlet.				
	This shows Full (Power by DC) when the Nebula Device receives power using a power adapter.				
	This shows Full (Power by PoE) when the Nebula Device receives power through a PoE switch/injector using IEEE 802.3at PoE plus. The PoE device that supports IEEE 802.3at PoE Plus can supply power of up to 30W per Ethernet port.				
	This shows Limited (Require 802.3bt power) when the Nebula Device receives power through a PoE switch/injector using IEEE 802.3bt PoE even when it is also connected to a power source using a power adapter. The PoE device that supports IEEE 802.3bt PoE can supply power of up to 71.3W per Ethernet port.				
	This shows Limited (Require 802.3at power) when the Nebula Device receives power through a PoE switch/injector using IEEE 802.3at PoE even when it is also connected to a power source using a power adapter. The PoE device that supports IEEE 802.3at PoE can supply power of up to 15.4W per Ethernet port.				
	This field is blank when the access point's firmware is older than version 5.50 or (WAX650S / WAX510D firmware is older than version 6.00P4C0). Or when the access point is offline.				
	Click the edit icon to open a screen where you can enable full power mode.				
	Power Setting X				
	on O Force override the power mode to full power				
	Note: Please make sure the power source can provide full power to avoid the system interrupt issue.				
	Close Update				
	Note: As of this writing, the following is a list of models that will show the edit icon for enabling full power mode: NAP303, NAP353, NWA1302-AC, NWA1123- AC HD, NWA5123-AC HD, WAC6303D-S, WAC6502D-E, WAC6502D-S, WAC6503D-S, WAC6552D-S, WAC6553D-S, WAX650S, NWA110AX, WAX510D.				
Antenna	This displays the antenna orientation settings for the Nebula Device that comes with internal antennas and also has an antenna switch.				
Smart mesh	This shows whether Nebula Smart Mesh is enabled on the Nebula Device.				
	For more information about Smart Mesh, see Section 5.1.1 on page 233.				
	To view the list of Nebula Devices that support smart mesh, go to Help > Device function table.				

Table 14 Site-wide > Devices > Access points: Details (continued)

LABEL	DESCRIPTION				
Edit	Edit the Nebula Device's Smart Me	sh settings.			
	Smart mesh	×			
	Enabled				
	Band: Beta 🕦	Auto (high band preferred)			
	Downlink: Beta 🚺				
	Manual uplink: Beta				
	Uplink auto failover:				
	्र select a AP:	· · · · · · · · · · · · · · · · · · ·			
	Note: Configure smart mesh here will override	global setting for this access point.			
		Cancel Save			
Enabled	Enable or disable Smart Mesh on th	e Nebula Device.			
	This setting overrides the Smart Mes NCC.	h settings configured for the Nebula De	evice's site in		
Lock	When enabled, the Nebula Device's local Smart Mesh settings overrides the Smart Mesh settings configured for the Nebula Device's site in NCC.				
	Example 1: If Smart Mesh is enabled for the site in NCC, you can disable Smart Mesh on the Nebula Device by setting Lock to on and Enabled to off.				
	Example 2: If Smart Mesh is disabled for the site in NCC, you can enable Smart Mesh on the Nebula Device by setting Lock to on and Enabled to on.				
Band	This setting will apply to mesh extender.				
	Select Auto (high band preferred) to allow the mesh extender to select a higher radio band mesh controller				
	Select 2.4 GHz to use the 2.4 GHz band for regular Internet surfing and downloading.				
	 Select 5 GHZ or 6 GHZ to use the 5 or 6 GHZ band for time sensitive frattic like high- definition video, music, and gaming. 				
	Note: 6 GHz will display only for mesh extender that support it.				
Downlink	When enabled, the mesh extender can provide downlink capability to another mesh extender.				
Manual uplink	When enabled, this allows you to select a mesh controller or mesh extender.				
Uplink auto failover	When enabled, an mesh extender after 5 tries, will automatically conr	that cannot connect to the selected n nect to another mesh controller or mesh	nesh controller n extender.		
select a AP	Select a mesh controller or mesh ex	ktender.			
Wireless bridge	This shows whether wireless bridge is enabled on the Nebula Device.				
	For more information about wireless bridge, see Section 5.1.2.2 on page 235.				
	Note: Wireless bridge can only	work when smart mesh is enabled	in this screen.		
Edit	Edit the Nebula Device's wireless b	ridge settings.			

Table 14 Site-wide > Devices > Access points: Details (continued)

LABEL	DESCRIPTION				
Enabled	Enable or disable wireless bridge on the Nebula Device.				
	Note: If Smart Mesh is disabled for the site in NCC, then enabling wireless bridge automatically enables Smart Mesh on the Nebula Device.				
Allowed VLANs	Enter the IDs of the VLANs that the Nebula Device will forward over the wireless bridge.				
	By default, this field uses the VLANs allowed for LAN1 at Site-wide > Configure > Access points > AP & port settings. For details, see Section 5.3.7 on page 271.				
History	Click Event log to go to the Site-wide > Monitor > Access points > Event log screen.				
Configuration status	This shows whether the configuration on the Nebula Device is up-to-date.				
Firmware availability	This shows whether the firmware on the Nebula Device is up-to-date or there is firmware update available for the Nebula Device.				
Current version	This shows the firmware version currently installed on the Nebula Device.				
Мар	This shows the location of the Nebula Device on Google map (Map view or Satellite imagery view) or on a floor plan. Click Floor plan to display a list of existing floor plans. Each floor plan has a drawing that shows the rooms scaled and viewed from above. Drag-and- drop your Nebula Device directly on the Google map or click Position device to update the Nebula Device's address (physical location).				
	Position device X				
	Update my device's location. What is this? Use the device's IP address (GEO IP). Get my location from web browser. Use the following address or coordinates. X				
	Cancel Update				
	 Select GEO IP to use the public IP address of the Nebula Device. Select Get my location from web browser to use the public IP address of the computer accessing the NCC portal. 				
	Select Use the following address or coordinates to enter the complete address or coordinates of the Nebula Device.				
	Note: Nebula Devices that are offline cannot use GEO IP.				
Photo	This shows the photo of the Nebula Device. Click Add to upload one or more photos. Click x to remove a photo.				
Live tools					
	This shows the Nebula Device traffic statistics.				
Current stations	This shows the Nebula Device's connected WiFi clients' MAC address, SSID name, IPv4 Address, Signal strength, Security, Channel, Tx rate, Rx rate, Association time, and Capability.				
Ping	Enter the domain name or IP address of a computer that you want to perform ping from the Nebula Device in order to test a connection and click Ping .				
	This can be used to determine if the Nebula Device and the computer are able to communicate with each other.				

Table 14	Site-wide > Devices >	Access	points: Details	(continued)	
		100000	pointis. Dorails		

LABEL	DESCRIPTION	
Traceroute	Enter the domain name or IP address of a computer that you want to perform traceroute from the Nebula Device and click Run . This determines the path a packet takes to the specified computer.	
Reboot AP	Click the Reboot button to restart the Nebula Device.	
	Note: All connected clients will be temporarily disconnected during reboot.	
Locator LED	Enter a time interval between 1 and 60 minutes. The locator LED will blink for the number of minutes set here once you turn on the locator LED.	
	Click the button to turn on the locator feature, which shows the actual location of the Nebula Device between several devices in the network.	
Remote Access	This allows you to establish a remote connection to this Nebula Device by specifying the port number. Then click Establish .	
	This feature is available to the organization owner, organization administrators with full privileges, and site administrators with full privileges.	
Wired stations	This shows the Nebula Device's connected wired clients' MAC address, IPv4 Address, Port number, and the VLAN ID assigned to the wired station.	
	Note: At the time of writing Wired stations is available for WAC500H only.	
Access point usage ar	nd connectivity	
Move the cursor over t	the chart to see the transmission rate at a specific time.	
Zoom	Select to view the statistics in the past 2 hours, day, week, or month.	
Pan	Click to move backward or forward by one day or week.	

Table 14 Site-wide > Devices > Access points: Details (continued)

4.3.2 Switches

This screen allows you to view the detailed information about a Nebula Device in the selected site. Click **Site-wide > Devices > Switches** to access this screen.

Figure 27 Site-wide > Devices > Switches

Switches	Last 2 hours	•	C					
Action	Tag 🕶 Mo	ve▼ Q Sec	rch Switch_	• (1) selected	d in 1 switch	es	Onl	line 🖲 Offline 🖲 Alert 🖲 Offline more than 6 days 🛛 🌱 🔁 Export 🗸
2	Status	Name	Tag	MAC address	LAN IP	Public IP	Model	Configuration status
	4 🔵	Shawn_NSW10	00-10P	B8:EC:A3:2B:4C:BA	192.168.11.37	210.61.209.2	NSW100-10P	Up to date
2023-01-04 1 UPnP packet	6:04:17 is detected in I	PTV networks. <u>C</u>	heck detail					

Table 15 Site-wide > Devices > Switches

LABEL	DESCRIPTION
Switches	Select to view the Nebula Device information and connection status in the past two hours, day, week or month.
5	Click this button to reload the data-related frames on this page.
Action	Perform an action on the selected Nebula Devices.
Reboot	Restart the Nebula Device.
Upgrade	Upgrade the firmware on the Nebula Device.

LABEL	DESCRIPTION	
Tag	Select one or multiple Nebula Devices and click this button to create a new tag for the Nebula Devices or delete an existing tag.	
Move	Select one or multiple Nebula Devices and click this button to move the Nebula Device to another site or remove the Nebula Device from the current site.	
Search	Specify your desired filter criteria to filter the list of Nebula Devices.	
Switch	This shows the number of Nebula Devices connected to the site network.	
Export	Click this button to save the Nebula Device list as a CSV or XML file to your computer.	
Status	This shows the status of the Nebula Device. Hover the mouse over the icon for a brief description.	
	 Green: The Nebula Device is online and has no alerts. Amber: The Nebula Device has alerts. Red: The Nebula Device is offline. Gray: The Nebula Device has been offline for 7 days or more. With lock: The Nebula Device is locked by Auto Configuration Recovery. See Table 75 on page 312 for more information. 	
	Move the cursor over an amber alert icon to view the alerts the NCC generates when an error or something abnormal is detected on the IPTV network.	
Name	This shows the descriptive name of the Nebula Device.	
Tag	This shows the user-specified tag for the Nebula Device.	
MAC address	This shows the MAC address of the Nebula Device.	
LAN IP	This shows the local (LAN) IP address of the Nebula Device.	
Public IP	This shows the global (WAN) IP address of the Nebula Device.	
Model	This shows the model number of the Nebula Device.	
# Port	This shows the number of the Nebula Device port which is connected to the NCC.	
Configuration status	This shows whether the configuration on the Nebula Device is up-to-date.	
Bandwidth Utilization (Uplink port)	This shows what percentage of the upstream/downstream bandwidth is currently being used by the Nebula Device's uplink port.	
Production information	This shows the Nebula Device's product description to explain what this Nebula Device is and also provides information about its features.	
Connectivity	This shows the Nebula Device connection status. Nothing displays if the Nebula Device is offline.	
	The gray time slot indicates the connection to the NCC is down, and the green time slot indicates the connection is up. Move the cursor over a time slot to see the actual date and time when a Nebula Device is connected or disconnected.	
Description	This shows the user-specified description for the Nebula Device.	
Serial number	This shows the serial number of the Nebula Device.	
Firmware status	This shows whether the firmware installed on the Nebula Device is up-to-date.	

Table 15 Site-wide > Devices > Switches (continued)

LABEL	DESCRIPTION
Firmware type	This shows Stable when the installed firmware may not have the latest features but has passed Zyxel internal and external testing.
	This shows Latest when the installed firmware is the most recent release with the latest features, improvements, and bug fixes.
	This shows General Availability when the installed firmware is a release before Latest , but is still undergoing Zyxel external testing.
	This shows Dedicated when the installed firmware is locked and Zyxel support is monitoring. Contact Zyxel customer support if you want to unlock the firmware in order to upgrade to a later one.
	This shows Beta when the installed firmware is a release version for testing the latest features and is still undergoing Zyxel internal and external testing.
	This shows N/A when the Nebula Device is offline and its firmware status is not available.
Firmware availability	This shows whether the firmware on the Nebula Device is Up to date , there is firmware update available for the Nebula Device (Upgrade available), or a specific version of firmware has been installed by Zyxel customer support (Locked).
Current version	This shows the firmware version currently installed on the Nebula Device.
Usage	This shows the amount of data that has been transmitted or received by the Nebula Device's clients.
IP type	This shows whether the IP address was assigned automatically (DHCP), or manually (Static IP).
Ð	Click this icon to display a greater or lesser number of configuration fields. For faster loading of data, select only the configuration fields listed that do NOT take a long time to fetch data.

Table 15 Site-wide > Devices > Switches (continued)

4.3.2.1 Switch Details

Click a Nebula Device entry in the **Site-wide** > **Devices** > **Switches** screen to display individual Nebula Device statistics.

10000		Map Photo
20nflguration2		
		📾 💿 🔍 Position device 👔 🛞 Floor plan 📈 Map Satellit
Name:	Shawn_NSW100-10P	This device is being leasted by GEO IR
MAC address:	B8:EC:A3:2B:4C:BA	IP-based geolocation services can only provide an approximate measure of geolocation accuracy. Acknowledge.
Serial number:	S172L13000021 (NSW100-10P)	
Description:		
Address:		ng Ro
Tag:		Thongyla
status		* *
LAN IP:	192.168.11.37 (via DHCP) 12	Hsinchu Moat Park 3 関本公開 3 間本公開
	Gateway: 192.168.11.1	Genter
	DNS: 8.8.8.8	Art Gallery
	VLAN: 1	ion Hall 術館
DHCP server:	192.168.11.1 🖻	E
Public IP:	210.61.209.2	The
💎 Topology:	Show	\$
RSTP status:	root is <u>Shawn NSW100-10P</u> / root bridge priori	y: 32768
IGMP status:	Enabled	
PoE status:	Consumption 10.2 / 180 W	
History:	Event log	
Configuration status:	Up to date	
Firmware availability:	Upgrade available	
Current version:	V3.00(ABGO.2) 11/19/2019 (Stable)	Conda no Rd
Configure ports		
2 4 6 8	10 10	
💼 🔒 💼 👝	1Gbps	
	2.5Gbps	
	10Gbps	
1 3 5 7	9 9 STP blocking	
	P PoE	
in tools		
e coois	MAC table Reboot switch Locator L	ED
ing Port power cycle		
ing Port power cycle	-	
ing Port power cycle	55.	-

Figure 28 Site-wide > Devices > Switches: Switch Details



Note: The banner **This switch is currently protected by Auto Configuration Recovery** will display when this Nebula Device is locked by NCC. Click the **Unlock** button to continue using the Nebula Device.

LABEL	DESCRIPTION	
0	Click this button to reload the data-related frames on this page.	
Unlock	This button only appears when the Nebula Device is locked by NCC.	
	Click this button to continue using the Nebula Device.	
Configuration		
Click the edit icon to a Nebula Device to ano Nebula Device and ca	change the Nebula Device name, description, tags and address. You can also move the ther site. After modifying a Nebula Device name, the new name will be synchronized to the an be seen by protocols such as SNMP and LLDP.	
Name	This shows the descriptive name of the Nebula Device.	
MAC address	This shows the MAC address of the Nebula Device.	
Serial number	This shows the serial number of the Nebula Device.	
Description	This shows the user-specified description for the Nebula Device.	
Address	This shows the user-specified address for the Nebula Device.	
Tag	This shows the user-specified tag for the Nebula Device.	
Status		

Table 16 Site-wide > Devices > Switches: Switch Details

LABEL	DESCRIPTION				
LAN IP	This shows the local (LAN) IP address of the Nebula Device. It also shows the IP address the gateway and DNS servers.				
	Click the edit icon to open a screen where you can change the IP address, VLAN ID number and DNS server settings.				
	Set IP address		×		
	IP type	Static IP	•		
	IP		×		
	VLAN	1	×		
		on Follow site-wide setting.	Edit		
	Subnet mask	×			
	Gateway	×			
	Primary DNS	×			
	Secondary DNS	×			
			Cancel OK		
DHCP server	This shows the IP address of t	he DHCP server.			
Public IP	This shows the global (WAN) IP address of the Nebula Device.				
Topology	Click Show to go to the Site-wide > Topology screen. See Section 4.2 on page 155.				
RSTP status	This shows Disabled when RSTP is disabled on the Nebula Device. Otherwise, it shows the name or MAC address of the Nebula Device that is the root bridge of the spanning tree, and the bridge priority.				
IGMP status	This shows whether IGMP is enabled on the Nebula Device. If IGMP is enabled, it also shows the ID number of the VLAN on which the Nebula Device learns the multicast group membership and the IP address of the Nebula Device interface in IGMP querier mode.				
PoE status	This shows the power management mode, the amount of power the Nebula Device is currently supplying to the connected PoE-enabled devices and the total power the Nebula Device can provide to the connected PoE-enabled devices on the PoE ports. N/A displays if the Nebula Device does not support PoE.				
	Click the edit icon to open t	he PoE Configuration screen. See	Section 4.3.2.2 on page 175.		
History	Click Event log to go to the S	Site-wide > Monitor > Switches > E	Event log screen.		
Configuration status	This shows whether the confi	guration on the Nebula Device is	sup-to-date.		
Firmware availability	This shows whether the firmw update available for the Ne	vare on the Nebula Device is up-t bula Device.	o-date or there is firmware		
Current version	This shows the firmware version	on currently installed on the Neb	ula Device.		

Table 16 Site-wide > Devices > Switches: Switch Details (continued)

LABEL	DESCRIPTION	
Мар	This shows the location of the Nebula Device on Google map (Map view or Satellite imagery view) or on a floor plan. Click Floor plan to display a list of existing floor plans. Each floor plan has a drawing that shows the rooms scaled and viewed from above. Drag-and- drop your Nebula Device directly on the Google map or click Position device to update the Nebula Device's address (physical location).	
	Position device X	
	Update my device's location. What is this? Use the device's IP address (GEO IP). Get my location from web browser. Use the following address or coordinates. X	
	Cancel Update	
	 Select GEO IP to use the public IP address of the Nebula Device. Select Get my location from web browser to use the public IP address of the computer accessing the NCC portal. Select Use the following address or coordinates to enter the complete address or coordinates of the Nebula Device. 	
	Note: Nebula Devices that are offline cannot use GEO IP.	
Photo	This shows the photo of the Nebula Device. Click Add to upload one or more photos. Click x to remove a photo.	
Ports This shows the ports on 4.3.2.3 on page 175. N connection status of th	the Nebula Device. You can click a port to see the individual port statistics. See Section to ver a port to see additional port information. The port color indicates the ne port.	
 Gray (#888888): Th Orange (#FF8900): Green (#64BE00): 1 Azure (#0079FF): Th Violet (#8800FF): Th Blue (#004FEE): The 	The port is disconnected. The port is connected and is transmitting data at 10 or 100 Mbps. The port is connected and is transmitting data at 1000 Mbps (1 Gbps). The port is connected and is transmitting data at 2.5 Gbps. The port is connected and is transmitting data at 5 Gbps. The port is connected and is transmitting data at 10000 Mbps (10 Gbps).	
When the port is in the blocked icon displays	STP blocking state, failed LACP negotiation state, or failed port authentication state, a on top of the port (
Name	This shows the Nebula Device name configured in NCC.	
Status	This shows the connection status of the port.	
Туре	This shows the port type (Trunk or Access), PVID, and allowed VLANs.	
Speed	This shows the current connection speed of the port. If the speed is unavailable, this displays "Ethernet".	
LLDP	This shows the LLDP information received on the port.	
Reset	This button only appears when the PoE port is connected to a PD (powered device). Follow the prompt and click Confirm to reboot the PD connected to this port. Note: This button is not available for an uplink port.	

Table 16 Site-wide > Devices > Switches: Switch Details (continued)

LABEL	DESCRIPTION
Configure ports	Click this button to go to the Site-wide > Configure > Switches > Switch ports screen, where you can view port summary. See Section 6.3.1 on page 288.
Live tools	
Ping	Enter the host name or IP address of a computer that you want to perform ping in order to test a connection and click Ping .
Port Power Cycle	Enter the number of the ports and click the Reset button to disable and enable the ports again.
MAC table	This shows what device MAC address, belonging to what VLAN group (if any) is forwarded to which ports.
	You can define how it displays and arrange the data in the summary table below.
	Note: This tab will appear for NSW100 and NSW200 Series only.
Switch tables	Import the following data into NCC:
	• MAC table. Click Run to show what device MAC address, belonging to what VLAN group (if any) is forwarded to which ports. You can define how it displays and arrange the data in the summary table.
	 Routing table. Click Run to show the routing destination, gateway, interface IP addresses, hop count, and routing methods. The routing table is only displayed for L3 Nebula Devices.
	ARP table. Click Run to show the IP-to-MAC address mappings. The ARP table is only displayed for L3 Nebula Devices.
	• IP source guard. Click Run to show the static, DHCP snooping, blocked client entries, and expiration time of DHCP snooping and blocked entries on the Nebula Device.
	After clicking Run in IP source guard , the IPSG (IP source guard) table could be empty if:
	 It takes about 5 minutes to refresh the address table after you apply the Nebula Device settings Protected part is not appeiling
	 Protected points not specified NCC may not get completed data from Nebula Device due to unstable network. Please retry.
Reboot switch	Click the Reboot button to restart the Nebula Device.
Locator LED	Enter a time interval between 1 and 60 minutes to stop the locator LED from blinking. The locator LED will start to blink for the number of minutes set here.
	Click the \odot button to turn on the locator feature, which shows the actual location of the Nebula Device between several Nebula Devices in the network.
Remote Access	Select to use TCP (Transmission Control Protocol) Port 22 or 443 to establish a remote connection to this Nebula Device. The Nebula Device will create a reverse SSH (Secure SHell) connection. Then click Establish .
	After clicking Ok , NCC will provide a remote connection IPv4 address and service port number. For example, Remote connection: 34.247.173.104:27086. Use this IPv4 address and port to connect to the Nebula Device using an SSH terminal emulator (for example, Putty). The remote session will be available for 30 minutes.
	In case the connection cannot be established, confirm that the network allows Port 22 or 443 .
	Note: Use Remote Access for troubleshooting only.
Uplink usage	
Move the cursor over	the chart to see the transmission rate at a specific time.
Zoom	Select to view the statistics in the past 12 hours, day, week, month, 3 months or 6 months.
Pan	Click to move backward or forward by one day or week.
Power Consumption	

Table 16 Site-wide > Devices > Switches: Switch Details (continued)

LABEL	DESCRIPTION
	Select to view the Nebula Device power consumption in the past two hours, day, week or month.
	This shows the current, total, maximum and minimum power consumption of the Nebula Device.
y-axis	The y-axis shows how much power is used in Watts.
x-axis	The x-axis shows the time period over which the power consumption is recorded.

Table 16 Site-wide > Devices > Switches: Switch Details (continued)

4.3.2.2 PoE Configuration

Use this screen to set the PoE settings for the Nebula Device. To access this screen, click the edit icon next to **PoE Status** in the **Site-wide** > **Devices** > **Switches: Switch Details** screen.

Note: To set PoE settings for an individual port, such as schedule, priority, and power mode, edit the Nebula Device's port settings. For details, see Section 6.3.1 on page 288.

Figuro 20	Sita wida >	Davicas >	Switches	Switch	Datails	POF	[^] onfigu	ration
rigule 29	slie-wide /	Devices /	SWIICHES.	SWIICH	Derails.	FUE	Jornigu	anon

F	PoE configuratio	n		×
	Modifications to POE connect to it. Reference before any change is of Please contact suppo	configuration on this page have se the "Help page" carefully for a applied to it. rt team for any inquiries.	severe impact to POE device detail functional description	is X
	PoE mode	Consumption mode	•	
			Car	ncel Saving

The following table describes the labels in this screen.

LABEL	DESCRIPTION
PoE Mode	Select the power management mode you want the Nebula Device to use. Classification mode – Select this if you want the Nebula Device to reserve the Max Power (mW) to each powered device (PD) according to the priority level. If the total power supply runs out, PDs with lower priority do not get power to function.
	Consumption mode – Select this if you want the Nebula Device to manage the total power supply so that each connected PD gets a resource. However, the power allocated by the Nebula Device may be less than the Max Power (mW) of the PD. PDs with higher priority also get more power than those with lower priority levels.
Close	Click this button to exit this screen without saving.
Saving	Click this button to save your changes and close the screen.

Table 17	Site-wide >	Devices >	Switches:	Switch	Details:	PoF	Configur	ation
		D 0 11000 ·	0.0000	0,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	Dorano		coringoi	anon

4.3.2.3 Switch Port Details

Use this to view individual Nebula Device port statistics. To access this screen, click a port in the **Ports** section of the **Site-wide** > **Devices** > **Switches**: **Switch Details** screen or click the **details** link next to a port in the **Site-wide** > **Configure** > **Switches** > **Switch ports** screen.



Table 18	Site-wide > Devices >	Switches: Switch	Details: Port Details
		01110110010011011	Dorans, Forr Dorans

LABEL	DESCRIPTION
3	Click this button to reload the data-related frames on this page.
Switch / Port	Select to view the port information and connection status in the past two hours, day, week or month.
Port	This figure shows the ports on the Nebula Device.
	Click a port to go to the corresponding port details screen. The selected port is highlighted. Move the pointer over a port to see additional port information, such as its name, MAC address, type, and connection speed.
	The port color indicates the connection status of the port.
	 Gray (#888888): The port is disconnected. Orange (#FF8900): The port is connected and is transmitting data at 10 or 100 Mbps. Green (#64BE00): The port is connected and is transmitting data at 1000 Mbps (1 Gbps). Azure (#0079FF): The port is connected and is transmitting data at 2.5 Gbps. Violet (#8800FF): The port is connected and is transmitting data at 5 Gbps. Blue (#004FEE): The port is connected and is transmitting data at 10000 Mbps (10 Gbps).
	When the port is in the STP blocking state, failed LACP negotiation state, or failed port authentication state, a blocked icon displays on top of the port (for example) in the diagram.
Name	This shows the descriptive name of the port.
Status	This shows the connection status of the port.
MAC address	This shows the MAC address of the port.
Туре	This shows the port type (Trunk or Access), PVID, and allowed VLANs.
Speed	This shows the current connection speed of the port. If the speed is unavailable, this displays "Ethernet".
LLDP	This shows the LLDP information received on the port.
Configuration Click the edit icon to a port number). See Sec	open the Switch ports screen and show the ports that match the filter criteria (the selected stion 6.3.1 on page 288.
Summary	This shows the port's VLAN settings.
RSTP	This shows whether RSTP is disabled or enabled on the port.
Port mirroring	This shows whether traffic is mirrored on the port.
Status	
Name	This shows the name of the port.
Status	This shows the status of the port.
LLDP	This shows the LLDP (Link Layer Discovery Protocol) information received on the port.
History	Click Event log to go to the Site-wide > Monitor > Switches > Event log screen.
Bandwidth Utilization	
Current Utilization	This shows what percentage of the upstream/downstream bandwidth is currently being used by the port.
Maximum Utilization	This shows the maximum upstream/downstream bandwidth utilization (in percentage).
Minimum Utilization	This shows the minimum upstream/downstream bandwidth utilization (in percentage).
y-axis	The y-axis represents the transmission rate in Kbps (kilobits per second).
x-axis	The x-axis shows the time period over which the traffic flow occurred.
Power Consumption	

LABEL	DESCRIPTION
Total	This shows the total power consumption of the port.
Current Consumption	This shows the current power consumption of the port.
Maximum Consumption	This shows the maximum power consumption of the port.
Minimum Consumption	This shows the minimum power consumption of the port.
y-axis	The y-axis shows how much power is used in Watts.
x-axis	The x-axis shows the time period over which the power consumption is recorded.
Packets Counters	
TX/RX Unicast	This shows the number of good unicast packets transmitted/received on the port.
TX/RX Multicast	This shows the number of good multicast packets transmitted/received on the port.
TX/RX Broadcast	This shows the number of good broadcast packets transmitted/received on the port.
TX/RX Pause	This shows the number of 802.3x Pause packets transmitted/received on the port.
IGMP V2/V3	
Query Rx	This shows the number of IGMP query packets received on the port.
Report Rx	This shows the number of IGMP report packets received on the port.
Report Tx	This shows the number of IGMP report packets transmitted on the port.
Report Drops	This shows the number of IGMP report packets dropped on the port.
Leave Rx	This shows the number of IGMP leave packets received on the port.
Leave Tx	This shows the number of IGMP leave packets transmitted on the port.
Leave Drops	This shows the number of IGMP leave packets dropped on the port.
Error Packets	
RX CRC	This shows the number of packets received with CRC (Cyclic Redundant Check) errors. CRC errors indicate packet errors in the network, potentially caused by mismatching Ethernet speed/duplex, bad cables or transceivers, or malfunctioning client devices.
Length	This shows the number of packets received with a length that was out of range.
Runt	This shows the number of packets received that were too short (shorter than 64 octets), including the ones with CRC errors.
IPv4 Address	This shows the IP address of the incoming frame which is forwarded on the port.
	Note: The IP address is obtained using one of the following three methods:
	 LLDP remote information Information collected by the Nebula Security Gateway (NSG) in this site Information collected by NCC when the client connected to Nebula
MAC Address	This shows the MAC address of the incoming frame which is forwarded on the port.
VLAN	This shows the VLAN group to which the incoming frame belongs.
Cable Diagnostics	
Diagnose	Click Diagnose to perform a physical wire-pair test of the Ethernet connections on the port. The following fields display when you diagnose a port.
Channel	An Ethernet cable usually has four pairs of wires. A 10BASE-T or 100BASE-TX port only use and test two pairs, while a 1000BASE-T port requires all four pairs.
	This displays the descriptive name of the wire-pair in the cable.

Table 18	Site-wide >	Devices >	Switches: Sw	itch Details:	Port Details	(continued)
----------	-------------	-----------	--------------	---------------	--------------	-------------

LABEL	DESCRIPTION				
Pair Status	OK: The physical connection between the wire-pair is okay.				
	Open : There is no physical connection (an open circuit detected) between the wire-pair.				
	Short: There is a short circuit detected between the wire-pair.				
	Unknown : The Nebula Device failed to run cable diagnostics on the cable connected to this port.				
	Unsupported : The port is a fiber port or it is not active.				
Cable Length	This displays the total length of the Ethernet cable that is connected to the port when the Pair Status is OK and the Nebula Device chipset supports this feature.				
	This shows N/A if the Pair Status is Open or Short. Check the Distance to fault.				
	This shows Unsupported if the Nebula Device chipset does not support to show the cable length.				
Distance to fault (m)	This displays the distance between the port and the location where the cable is open or shorted.				
	This shows N/A if the Pair Status is OK .				
	This shows Unsupported if the Nebula Device chipset does not support to show the distance.				
DDMI	This section is available only on an SFP (Small Form Factor Pluggable) port.				
DDMI	Click DDMI (Digital Diagnostics Monitoring Interface) to display real-time SFP transceiver information and operating parameters on the port. You can also see the alarm and warning thresholds for temperature, voltage, transmission bias, transmission and receiving power.				
Port	This shows the number of the port on the Nebula Device.				
Vendor	This shows the vendor name of the transceiver installed in the port.				
PN	This shows the part number of the transceiver installed in the port.				
SN	This shows the serial number of the transceiver installed in the port.				
Revision	This shows the firmware version of the transceiver installed in the port.				
Date-code	This shows the date the installed transceiver's firmware was created.				
Transceiver	This shows the type and the Gigabit Ethernet standard supported by the transceiver installed in the port.				
Calibration	This shows whether the diagnostic information is internally calibrated or externally calibrated.				
Current	This shows the current operating parameters on the port, such as transceiver temperature, laser bias current, transmitted optical power, received optical power and transceiver supply voltage.				
High Alarm Threshold	This shows the high alarm threshold for temperature, voltage, transmission bias, transmission and receiving power. A trap is sent when the operating parameter is above the threshold.				
High Warn Threshold	This shows the high warning threshold for temperature, voltage, transmission bias, transmission and receiving power.				
Low Warn Threshold	This shows the low alarm threshold for temperature, voltage, transmission bias, transmission and receiving power. A trap is sent when the operating parameter is below the threshold.				
Low Alarm Threshold	This shows the low warning threshold for temperature, voltage, transmission bias, transmission and receiving power.				

 Table 18
 Site-wide > Devices > Switches: Switch Details: Port Details (continued)

4.3.3 Security Router

This screen allows you to view the detailed information about the Nebula Device in the selected site. Click **Site-wide** > **Devices** > **Security router** to access this screen.

de » Devices » Security r rity router	outer						
nflauration 🖗			Map Photo				
					diseta se a la constante de la constante		
iame.				baltion device	tigs Ploor plan	Map Sa	telli
AC dooress:	DATADIOFEEPO		This device is being lo	cated by GEO IP		inclosoftion	
enai number:	5220Y44039035 (SCH 504XE)		occuracy Acknowled	98.	Anovinione miscassie or s	addition and	
escription.							
ddress:				0			
aðe:							
1 2 3 4 5			原語科技設份有限公司				
			Google	Keyboard ahortouta Mop doto 02023 Go	ogle 10 m T	erms of Use Rep	ortar
itus							
			Usage	No usage in the last 24 ho	iurs		1
Public IP:	0.0.0.0		Topology.	Show			
Channel (Band):			History:	Event log			
			Configuration status:	Not up to date			
			Firmware availability.	N/A			
			Current version:	N/A (General Availability)			
letwork usage and a Zoom: 2 hours	1 day 7 days 30 days				Por: 44 4 C	k b	
16.40	1700	1720	17.40	100	10	20	
-		1				-	
1040	1200	17:20	12:40	18:00	10	20	
ive tools Ping Traceroute	DNS lookup Reboot gateway	150	1740	. 1820	107	U.	
accale.com						~ [Pie
the address and all						~	- 88

Figure 31 Site-wide > Devices > Security router
Table 19	Site-wide > Devices > Security router

LABEL	DESCRIPTION		
Configuration			
Click the edit icon to a also move the Nebula	change the Nebula Device name, description, tags and address (physical location). You can a Device to another site or remove.		
Name	This shows the descriptive name of the Nebula Device.		
MAC address	This shows the MAC address of the Nebula Device's WAN port.		
Serial number	This shows the serial number of the Nebula Device.		
Description	This shows the user-specified description for the Nebula Device.		
Address	This shows the user-specified address (physical location) for the Nebula Device.		
Tags	This shows the user-specified tags for the Nebula Device.		
Port	This shows the ports on the Nebula Device.		
	The port is highlighted in green color when it is connected and the link is up.		
	Move the pointer over a port to see additional port information, such as its name, connection status, MAC address, and connection speed.		
Мар	This shows the location of the Nebula Device on Google Maps (Map view or Satellite imagery view) or on a floor plan. Click Floor plan to display a list of existing floor plans. Each floor plan has a drawing that shows the rooms scaled and viewed from above. Drag-and- drop your Nebula Device directly on the Google map or click Position device to update the Nebula Device's address (physical location).		
	Position device X		
	Update my device's location. What is this? Use the device's IP address (GEO IP). Get my location from web browser. Use the following address or coordinates. X		
	Cancel Update Select GEO IP to use the public IP address of the Nebula Device. Select Get my location from web browser to use the public IP address of the computer accessing the NCC portal. Select Use the following address or coordinates to enter the complete address or coordinates of the Nebula Device.		
	NOTE: NEDUIA DEVICES THAT ARE OTTIINE CANNOT USE GEO IP.		
Photo	This shows the photo of the Nebula Device. Click Add to upload one or more photos. Click x to remove a photo.		
Status			
Public IP	This shows the IPv4 address of the WAN interface, and whether it was assigned automatically (DHCP), manually (Static IP), or by PPPoE.		

LABEL	DESCRIPTION
Channel (Band)	This shows the channel ID and WiFi frequency band currently being used by the Nebula Device.
	Note: This field only appears for ZyWALL ATP100W, USG FLEX 100W, and USG20W- VPN.
Usage	This shows the amount of data that has been transmitted or received by the Nebula Device's clients.
Topology	Click Show to go to the Site-Wide > Topology screen. See Section 4.2 on page 155.
History	Click Event log to go to the Site-wide > Monitor > Security router > Event log screen.
Configuration status	This shows whether the configuration on the Nebula Device is Up-to-date .
Firmware availability	This shows whether the firmware installed on the Nebula Device is Up-to-date .
Current version	This shows the firmware version currently installed on the Nebula Device.
Network usage and co	onnectivity
Move the cursor over t	the chart to see the transmission rate at a specific time.
Zoom	Select to view the statistics in the past 2 hours, 24 hours, 7 days, or 30 days.
Pan	Click to move backward or forward by one day or week.
Live tools	
Ping	Enter the host name or IP address of a computer that you want to perform ping in order to test a connection and click Ping . You can select the interface (WAN, LAN, or VLAN) through which the Security Firewall sends queries for ping.
	Note:
	 To ping for VPN/routing issues, it is not necessary to connect an end-device on the LAN interface of the Nebula Device.
	 A routing problem is possible if the WAN interface can reach the Internet but not the LAN interface.
Traceroute	Enter the host name or IP address of a computer that you want to perform the traceroute function. This determines the path a packet takes to the specified computer.
DNS lookup	Enter a host name and click Run to resolve the IP address for the specified domain name.
Reboot gateway	Click the Reboot button to restart the Nebula Device.

Table 19 Site-wide > Devices > Security router (continued)

4.3.4 Firewall

This screen allows you to view the detailed information about the Nebula Device in the selected site. Click **Site-wide** > **Devices** > **Firewall** to access this screen.



Figure 32 Site-wide > Devices > Firewall

NCC User's Guide

LABEL	DESCRIPTION
Configuration	
Click the edit icon t also move the Neb	to change the Nebula Device name, description, tags and address (physical location). You can ula Device to another site or remove.
Name	This shows the descriptive name of the Nebula Device.
MAC address	This shows the MAC address of the Nebula Device's WAN port.
Serial number	This shows the serial number of the Nebula Device.
Description	This shows the user-specified description for the Nebula Device.
Address	This shows the user-specified address (physical location) for the Nebula Device.
Tags	This shows the user-specified tags for the Nebula Device.
Port	This shows the ports on the Nebula Device.
	The port is highlighted in green color when it is connected and the link is up.
	Move the pointer over a port to see additional port information, such as its name, connection status, MAC address, and connection speed.
Port	This shows the identity number of the selected port.
Port Group	This shows the name of the port group that the port belongs to.
Status	This shows the connection status of the port.
	the Nebula Device's address (physical location). Position device X
	Update my device's location. <u>What is this?</u> Use the device's IP address (GEO IP). Get my location from web browser. Use the following address or coordinates.
	 Select GEO IP to use the public IP address of the Nebula Device. Select Get my location from web browser to use the public IP address of the computer accessing the NCC portal. Select Use the following address or coordinates to enter the complete address or coordinates of the Nebula Device. Note: Nebula Devices that are offline cannot use GEO IP.
Photo	This shows the photo of the Nebula Device. Click Add to upload one or more photos. Click
	x to remove a photo.

Table 20 Site-wide > Devices > Firewall

LABEL	DESCRIPTION	
CPU usage	This shows what percentage of the Nebula Device's processing capability is currently being used.	
Memory usage	This shows what percentage of the Nebula Device's RAM is currently being used.	
Session	This shows how many sessions the Nebula Device currently has. A session is a unique established connection that passes through, from, to, or within the Nebula Device.	
Channel (Band)	This shows the channel ID and WiFi frequency band currently being used by the Nebula Device.	
	Note: This field only appears for ZyWALL ATP100W, USG FLEX 100W, and USG20W- VPN.	
Usage	This shows the amount of data that has been transmitted or received by the Nebula Device's clients.	
Тороlоду	Click Show to go to the Site-Wide > Topology screen. See Section 4.2 on page 155.	
History	Click Event log to go to the Site-wide > Monitor > Firewall > Event log screen.	
Configuration status	This shows whether the configuration on the Nebula Device is Up-to-date .	
Firmware availability	This shows whether the firmware installed on the Nebula Device is Up-to-date .	
Current version	This shows the firmware version currently installed on the Nebula Device.	
WAN status		
WAN Interface	This shows the descriptive name of the active WAN connection.	
Status	This shows the connection status of the WAN interface (up or down).	
IP	This shows the IP address of the WAN interface, and whether it was assigned automatically (DHCP), manually (Static IP), or by PPPoE.	
Gateway	This shows the IP address of the default Nebula Device assigned to the WAN interface.	
DNS Server	This shows the IP addresses of the DNS servers assigned to the WAN interface.	
Network usage and co	onnectivity	
Move the cursor over t	he chart to see the transmission rate at a specific time.	
Zoom	Select to view the statistics in the past 2 hours, 24 hours, 7 days, or 30 days.	
Pan	Click to move backward or forward by one day or week.	
Live tools		
Traffic	This shows the WAN port statistics.	
	The y-axis represents the transmission rate for uploads and downloads.	
	The x-axis shows the time period over which the traffic flow occurred.	
DHCP leases	This shows the IP addresses currently assigned to DHCP clients.	
Ping	Enter the host name or IP address of a computer that you want to perform ping in order to test a connection and click Ping . You can select the interface (WAN, LAN, or VLAN) through which the Security Firewall sends queries for ping.	
	Note:	
	 To ping for VPN/routing issues, it is not necessary to connect an end-device on the LAN interface of the Nebula Device. 	
	 A routing problem is possible if the WAN interface can reach the Internet but not the LAN interface. 	
Traceroute	Enter the host name or IP address of a computer that you want to perform the traceroute function. This determines the path a packet takes to the specified computer.	
DNS lookup	Enter a host name and click Run to resolve the IP address for the specified domain name.	

 Table 20
 Site-wide > Devices > Firewall (continued)

LABEL	DESCRIPTION
Reboot device	Click the Reboot button to restart the Nebula Device.
Remote Access	This option is available only for the Nebula Device owner. Establish a remote command line interface (CLI) connection to the Nebula Device by
	specifying the Port number and clicking Establish .

Table 20 Site-wide > Devices > Firewall (continued)

4.3.5 Security Gateway

This screen allows you to view the detailed information about a Nebula Device in the selected site. Click **Site-wide > Devices > Security gateway** to access this screen.



Figure 33 Site-wide > Devices > Security gateway

NCC User's Guide

Table 21 Site-wide > Devices > Security gates	wav
---	-----

LABEL	DESCRIPTION		
Configuration			
Click the edit icon to a Nebula Device to and	change the Nebula Device name, description, tags and address. You can also move the other site or remove.		
Name	This shows the descriptive name of the Nebula Device.		
MAC address	This shows the MAC address of the Nebula Device.		
Serial number	This shows the serial number of the Nebula Device.		
Description	This shows the user-specified description for the Nebula Device.		
Address	This shows the user-specified address for the Nebula Device.		
Tags	This shows the user-specified tag for the Nebula Device.		
Port	This shows the ports on the Nebula Device.		
	The port is highlighted in green color when it is connected and the link is up.		
	Move the pointer over a port to see additional port information, such as its name, connection status, MAC address, and connection speed.		
Name	This shows the descriptive name of the port.		
Status	This shows the connection status of the port.		
MAC address	This shows the MAC address of the port.		
Speed	This shows the current connection speed of the port. If the speed is unavailable, this displays "Ethernet".		
LLDP	This shows the LLDP information received on the port.		
	imagery view) or on a floor plan. Click Floor plan to display a list of existing floor plans. Each floor plan has a drawing that shows the rooms scaled and viewed from above. Drag-and- drop your Nebula Device directly on the Google map or click Position device to update the Nebula Device's address (physical location).		
	Position device X		
	Update my device's location. <u>What is this?</u>		
	• Use the device's IP address (GEO IP).		
	Get my location from web browser.		
	Use the following address or coordinates.		
	×		
	Cancel Update		
	 Select GEO IP to use the public IP address of the Nebula Device. Select Get my location from web browser to use the public IP address of the computer accessing the NCC portal. Select Use the following address or coordinates to opter the complete address or 		
	coordinates of the Nebula Device.		
	Note: Nebula Devices that are offline cannot use GEO IP.		

LABEL	DESCRIPTION	
Photo	This shows the photo of the Nebula Device. Click Add to upload one or more photos. Click \mathbf{x} to remove a photo.	
Status		
WAN1/WAN2	This shows the IP address, gateway, DNS, and VLAN ID information for the active WAN connection.	
Public IP	This shows the global (WAN) IP address of the Nebula Device.	
CPU usage	This shows what percentage of the Nebula Device's processing capability is currently being used.	
Memory usage	This shows what percentage of the Nebula Device's RAM is currently being used.	
Security Service	This shows whether Nebula Security Services (NSS) are enabled on the Nebula Device. Click What is this? to view the type of enabled security services.	
	When the gateway's NSS license expires, NSS is automatically disabled. This field displays an edit button which you can use to re-enable the services after renewing the NSS license.	
Usage	This shows the amount of data that has been transmitted or received by the Nebula Device's clients.	
Тороюду	Click Show to go to the Site-Wide > Topology screen. See Section 4.2 on page 155.	
History	Click Event log to go to the Site-Wide > Monitor > Security gateway > Event log screen.	
Configuration status	This shows whether the configuration on the Nebula Device is up-to-date.	
Firmware availability	This shows whether the firmware installed on the Nebula Device is up-to-date.	
Current version	This shows the firmware version currently installed on the Nebula Device.	
Live tools		
Internet traffic	This shows the WAN port statistics.	
	The y-axis represents the transmission rate in Kbps (kilobits per second).	
	The x-axis shows the time period over which the traffic flow occurred.	
DHCP leases	This shows the IP addresses currently assigned to DHCP clients.	
Ping	Enter the host name or IP address of a computer that you want to perform ping in order to test a connection and click Ping . You can select the interface through which the Nebula Device sends queries for ping.	
Traceroute	Enter the host name or IP address of a computer that you want to perform the traceroute function. This determines the path a packet takes to the specified computer.	
DNS lookup	Enter a host name and click Run to resolve the IP address for the specified domain name.	
Reboot gateway	Click the Reboot button to restart the Nebula Device.	
Remote Access	This option is available only for the Nebula Device owner.	
	Establish a remote connection by specifying the Port number and clicking Establish .	
Network usage and connectivity		
Move the cursor over	the chart to see the transmission rate at a specific time.	
Zoom	Select to view the statistics in the past 2 hours, day, week, or month.	
Pan	Click to move backward or forward by one day or week.	

 Table 21
 Site-wide > Devices > Security gateway (continued)

4.3.6 Mobile Router

This screen allows you to view the detailed information about a Nebula Device in the selected site. Click **Site-wide > Devices > Mobile router** to access this screen. See the Mobile Router chapter for more information.

4.4 Map & Floor Plans

This screen allows you to locate a Nebula Device on the world map and use a floor plan to show where Nebula Devices are physically located. Click **Site-Wide** > **Map & floor plans** to access this screen.





Place device on map

You can mark on the map the places where the Nebula Devices are located. Click the **Place device on map** tab to display the Nebula Device list for the selected site. Click the arrow (<<) on the upper left corner of the **Map & floor plans** screen to collapse or expand the list.

Click the **Placed** button to show the Nebula Devices that you have pinned on the map and/or the floor plan. Click the **Un-placed** button to show the Nebula Devices that remain to be pinned on the map. To pin a Nebula Device, select the Nebula Device from the **Un-placed** list, then drag and drop it on the map.

The pin icon next to a Nebula Device name is green (\bigcirc) if you have marked the Nebula Device on the map. Otherwise, the pin icon is gray (\bigcirc). Click the \bigotimes icon to remove a Nebula Device from the map.

Place devic	e on map	Edit floor plans	
Placed 4	Un-placed (1	\supset	$\langle \langle$
Status	Name	Action	
•	HomeNAP.	. 💡 🔕	
•	Home NS	<mark>9</mark> 🛛	
•	OfficeNAP.	. 💡 🔕	
•	Home GW	❷ ⊗	

Place devic	e on map	Edit floor plans	
Placed 4	Un-placed (1	
Status	Name	Action	
•	Office NS	s 🛛	

Figure 35 Site-wide > Map & floor plans: Place device on map

Edit floor plans

Click the Edit floor plans tab to display the list of existing floor plan, a drawing that shows the rooms scaled and viewed from above. Click the arrow (\ll) on the upper left corner of the Map & floor plans screen to collapse or expand the list.

Use the **Create+** button to upload a new floor plan. The floor plan then shows on the Google map at the right side of the screen. Use your mouse to move the floor plan, and use the icons at the top of the map to rotate, change the transparency, resize or hide the floor plan. Click **Set position** to apply your changes. If you want to relocate the floor plan, select the floor plan from the list and click its edit icon.



Figure 36 Site-wide > Map & floor plans: Edit floor plans

LABEL	DESCRIPTION
Floor plan	This shows the descriptive name of the floor plan.
Devices	This shows the number of Nebula Devices marked on this floor plan.
	Click this icon to open a screen, where you can modify the name, address and/or dimension of the floor plan.
1	Click this icon to delete the floor plan.

Table 22 Site-Wide > Map & floor plans: Edit floor plans

4.5 Clients

This screen shows a list of all wired and WiFi clients connected to Nebula Devices (access points, Switches, Security Appliances, mobile routers) in the site. You can also block or allow clients. Click **Sitewide** > **Clients** to access this screen.

Figure 37 Site-wide > Clients > Client list

	▪ Last 2 h	ours 👻	C					Show all a	lients Show policy
Q s	earch clients	• (21) clien	ts						💎 🕒 Ехр
Status	Description	Connected to	MAC address	IPv4 address	First seen	Last seen	Manufacturer	Policy	Note
	MR32 - NSBU	Shawn NSW100	00:18:0A:2F:0A:C0		2023-01-10 11:09:	2023-01-10 13:04:	Cisco Meraki	Normal	
) (îr	OPPO-Reno5-5G	BC:CF:4F:56:BD:6D	1A:9F:C2:CF:ED:F5	173.16.2.118	2023-01-10 11:08:05	2023-01-10 13:05:10	Unspecified	Normal	
) 🛜	22:10:A8:08:66:C8	BC:CF:4F:56:BD:6D	22:10:A8:0B:66:C8		2023-01-10 11:59:05	2023-01-10 13:05:10	Unspecified	Normal	
] 🛜	3A-9C-59-3A-F5-C3	BC:CF:4F:56:BD:6D	3A:9C:59:3A:F5:C3	173.16.2.88	2023-01-10 11:20:06	2023-01-10 13:05:10	Unspecified	Normal	
) 🛜	TWNBNT03196-M	Marketing2	3C:06:30:43:37:4B	173.16.2.78	2023-01-10 11:09:	2023-01-10 13:06:11	Apple, Inc.	Normal	
] 🋜	Kell-iPhone-2	BC:CF:4F:56:BD:6D	70:B3:06:20:C0:5F	173.16.2.80	2023-01-10 11:54:06	2023-01-10 12:02:	Apple, Inc.	Normal	
) 🌾	shumantkiiPhone	BC:CF:4F:56:BD:6D	72:D3:1D:99:EB:93	173.16.2.50	2023-01-10 11:08:01	2023-01-10 13:05:10	Unspecified	Normal	
) 🛜	86:EE:D1:02:99:A4	shawn-620-6E	86:EE:D1:02:99:A4	173.16.2.85	2023-0 <mark>1-1</mark> 0 11:08:05	2023-01-10 13:05:14	Unspecified	Normal	
) 🛜	8E:34:72:56:40:74	BC:CF:4F:56:BD:6D	8E:34:72:56:40:74	173.16.2.69	2023-01-10 12:38:	2023-01-10 13:05:10	Unspecified	Normal	
	TWNBNT03245-M.	BC:CF-4F-56:BD:6D	A0:78:17:8D:4D:B9	173.16.2.40	2023-01-10 11:08:01	2023-01-10 13:05:10	Apple, Inc.	Normal	

LABEL	DESCRIPTION
Client list	Select to filter the list of clients, based on what type of Nebula Device (access point, Switch, Security Appliance, mobile router) the client is connected to.
	You can also set a time; the list shows each client's connection status in the past two hours or past 24 hours.
\mathcal{C}	Click this button to reload the data-related frames on this page.
Show all clients	Click this to show all clients that have been online during the selected time period.

Table 23 Site-wide > Clients > Client list

Table 23	Site-wide > Clients > Client	list	(continued)
		1131	

LABEL	DESCRIPTION
Show policy clients	Click this to show clients that have a white-listed or blocked policy applied to them, regardless of when they were last online. The client's usage data is calculated according to the selected time period.
Policy	Select the clients from the table below, and then choose the security policy that you want to apply to the selected clients. Choose one of the following policies, then click Apply policy .
	 Allow list: The selected clients to bypass captive portal authentication. Block list: The selected clients cannot connect to the site. How a client is blocked depends on the connected Nebula Device type selected under Client list. AP: The client is blocked by MAC address from connecting to any AP in the site. Switch: The client is blocked by MAC address from sending or receiving network traffic. Gateway: The Security Appliance will not route traffic for the client's IP address. To specific SSID: Selectively apply captive portal authentication to specific_SSIDs on an AP. Normal: The selected clients have no policies applied to them.
Search clients	Specify your desired filter criteria to filter the list of clients.
N clients	This shows the number of clients (N) connected to the gateway in the site network.
Export	Click this button to save the client list as a CSV or XML file to your computer.
General fields	
	Select an entry's check box to select a specific client. Otherwise, select the check box in the table heading row to select all clients.
Status	This shows whether the client is online (green) or offline (red), and whether the client is wired or wireless.
	Clients connected to an Access Point are reported as wireless.Clients connected to a Switch or Security Appliance are reported as wired.
Description	 This shows the descriptive name of the client. By default, this is the client's MAC address. The client description can be obtained through the following: User customized description Hostname detected from client's LLDP (Link Layer Discovery Protocol) System Name Hostname detected from the Nebula-managed access point
	 Hostname detected from the Nebula-managed Security Appliance.
	Click the name to display the individual client statistics. See wireless: Section 4.5.0.1 on page 194 and wired: Section 4.5.0.2 on page 196.
Connected to	This shows the name of the Nebula Device to which the client is connected in this site.
	Click the Nebula Device name to display the screen where you can view detailed information about the Nebula Device.
MAC address	This shows the MAC address of the client.
	Click the MAC address to display the individual client statistics. See wireless: Section 4.5.0.1 on page 194 and wired: Section 4.5.0.2 on page 196.
IPv4 address	This shows the IPv4 address of the client. By default, the field is blank. The client IPv4 address can be obtained through the following:
	 IPv4 address detected from client's LLDP (Link Layer Discovery Protocol) Management Address IPv4 address detected from the Nebula-managed access point IPv4 address detected from the Nebula-managed Security Appliance.
First seen	This shows the first date and time the client was discovered over the specified period of time.
Last seen	This shows the last date and time the client was discovered over the specified period of time.
Manufacturer	This shows the manufacturer of the client hardware.
Policy	This shows the security policy applied to the client.

Table 23 Site-wide > Clients > Client list (continued)

LABEL	DESCRIPTION
Note	This shows additional information about the client.
E	Click this icon to display a greater or lesser number of configuration fields.

4.5.0.1 WiFi Client Details

Click a WiFi client entry in the Site-wide > Clients > Clients list screen to display individual client statistics.



Figure 38 Site-wide > Clients > Clients list: WiFi Client Details

LABEL	DESCRIPTION
Client	Click the edit icon to change the client name.
Status	This shows whether the client is online (green), or goes offline (red). It also shows the last date and time the client was discovered.
SSID	This shows the name of the Access Point's WiFi network to which the client is connected.
Connected to	This shows the name of the Nebula managed Access Point to which the client is connected.
	Click the name to display the individual Access Point statistics. See Section 4.3.1.1 on page 159.
Signal	This shows the RSSI (Received Signal Strength Indicator) of the client's WiFi connection, and an icon showing the signal strength.
	Icon default thresholds:
	• Green/5 blocks: signal is greater than -67 dBm, strong signal
	Amber/4 blocks: signal –67 to –73 dBm, average signal
	 Amber/3 blocks: signal -/4 to -80 dBm, below average signal Red/2 blocks: signal is less than -80 dBm, weak signal
Security	This shows the encryption method used to connect to the Access Point
Captive portal	This shows the web authentication method used by the client to access the network
	This shows the number of users currently connected to the network through the client
0301	device.
Manufacturer	This shows the manufacturer of the device connected to the Access Point.
OS	This shows the operating system running on the client device, if known.
Capability	This shows the WiFi standards supported by the client or the supported standards currently being used by the client.
Note	This shows additional information for the client. Click the edit icon to change it.
History	Click Event log to go to the Site-wide > Monitor > Access points > Event log screen.
Мар	This shows the location of the client on the Google map.
Period	Select to view the statistics in the past two hours, day, week or month.
Pan	Click to move backward or forward by two hours or one day.
y-axis	The y-axis shows the transmission speed of data sent or received by the client in kilobits per second (Kbps).
x-axis	The x-axis shows the time period over which the traffic flow occurred.
Network	
IPv4 address	This shows the IP address of the client.
MAC address	This shows the MAC address of the client.
VLAN	This shows the ID number of the VLAN to which the client belongs.
Ping	Click the button to ping the client's IP address from the Nebula AP to test connectivity.
Loss rate	This shows the rate of packet loss when you perform ping.
Average latency	This shows the average latency in ms when you perform ping.

Table 24	Site-wide > Clients	: > Clients list.	WiFi Client Details

4.5.0.2 Wired Client Details

Click a wired client's descriptive name in the Site-wide > Clients > Clients list screen to display individual client statistics.

Figure 39	Site-wide >	· Clients >	Clients list:	Wired Client	t Details
-----------	-------------	-------------	---------------	--------------	-----------

Basic information		Network	
Status:	📋 (Last seen 2021-04-27 14:15:58)	IPv4 address 🕕 :	192.168.137.46
Connected to:	BC: DE ME HE HE DE	MAC address:	and the service of the
Manufacturer:	Zyxel Communications Corporation	VLAN:	1
Note:		Port:	1
Ping	Loss rate: 0% Average latency: 2.348ms		
6.6	Loss rate: 0% Average latency: 2.348ms		
Ping	Loss rate: 0% Average latency: 2.348ms		

Table 25	Site-wide >	Clients >	Clients list:	Wired	Client Details

LABEL	DESCRIPTION
Client	Click the edit icon to change the client name.
Status	This shows whether the client is online (green) or offline (red). It also shows the last date and time the client was discovered, and whether the client is wired or wireless.
Connected to	This shows the name of the Security Appliance to which the client is connected.
Manufacturer	This shows the manufacturer of the client device.
Note	Enter information about this Nebula Device, for yourself or for other administrators.
LLDP information	This shows the LLDP (Link Layer Discovery Protocol) information received from the remote device.
Network	
IPv4 address	This shows the IPv4 address of the client.
MAC address	This shows the MAC address of the client.
VLAN	This shows the VLAN ID for this client.
Port	This shows the port number of the Nebula Device the client is connected.
Ping	Click the button to ping the client's IP address from the gateway to test connectivity.
	Note: This button is grayed-out when client is not assigned an IP address.

4.5.1 WiFi Aid

The **WiFi Aid** screen displays the number of WiFi clients that cannot connect to an AP(s) in a site. It also displays the number of WiFi clients who cannot authenticate in a hotspot (captive portal) or who have timed out.

Use this screen to identify connection problems between WiFi clients and supported AP(s). Click Sitewide > Clients > WiFi Aid to access this screen.

Note: This feature is available if you have the Nebula Pro Pack license only.

Note: After a WiFi client successfully connects to the Nebula Device, NCC will not count past connection failures.

Figure 40 Site-wide > Clients > WiFi Aid

Chent list	WiFi Aid Betn	Connection log Beta		
feature is designed to	provide an overview to sum	marize the issue from a wireless client connection. It's	only available for access point	nte Model list
		ABles		
Last hour • All	sside 💌	All togs - C		
Client devices affecte	d by connection problems			Client devices affected by captive portal problems
		1 / 8 Client devices		O / 0 Client devices
	N			
)(→(<u>2</u>)		
Wireless)	\rightarrow $($ $)$ $()$ $($		Portal
Wireless)	\rightarrow \bigcirc		Portel
Wireless 1 follures)	→ () O foilures	O failures	Portal O failures
Wireless 1 follures)	→ () O foilures	O follures	Portal O failures
Virietess 1 follures)	→ () O foilures	O foilures	Portal O failures
Wireleas 1 failures)	\rightarrow \bigcirc \bigcirc \bigcirc \bigcirc failures	O foilures	Portal O failures
Wireless 1 failures	# Failed/lotal connec	O failures	O failures Pailed connection by SSID SSID e-Nebula-FT	Portal O failures # Failed connections
Wireless 1 follures called clients Cfient device AEDF EP A9 DO 47	# Failed/total connect 4/8	C Show history within the time range Kions Last failed issue Wireless connection	Contraction by SSID SSID e-Nebula-FT	"Failed connections
Client device AEDF EX A DO 07 Koll JPhone-2	# Failed/total connect 4/8 1/3	Constant of the second se	Contraction by SSID	# Failed connections
Client device AEDF ER A 900 47 Koll JPhone - 2 F66C 06 D2 51 AF	# Failed/total connec 4/8 1/3 1/7	Control of the second sec	Contraction by SSID	# Failed connections

Table 26	Site-wide >	> Clients >	> WiFi Aic
		0.01.01	

LABEL	DESCRIPTION
WiFi Aid	Select a Time range . The overview will show all WiFi clients' connection issues in the Last hour , Last 12 hours , Last 24 hours , or Custom range (from 15 minutes to one day).
	Select to filter the overview of the client's WiFi connection issues based on one AP WiFi network (SSID), or all WiFi networks (All SSIDs, default).
	Select to filter the overview of all WiFi clients' connection issues based on one AP tag , or All tags (default). This is the tag you create in Access point > Monitor > Access points .
	Click the Refresh icon to update this screen.
Client devices	This chart displays the number of WiFi clients with the following connection problems.
affected by connection	• Wireless failures. This displays the number of WiFi clients that failed association to an AP or failed authentication.
problems	• DHCP failures. This displays the number of WiFi clients that failed to receive an IP address due to DHCP failure/timeout with the DHCP server.
	 DNS failures. This displays the number of WiFi clients that failed DNS query due to DNS timeout from a DNS server.
Client devices affected by captive portal problems	This chart displays the number of WiFi clients that failed hotspot authentication. This includes entering the wrong user credentials or an authentication timeout.
Failed clients	This table displays the number of WiFi clients with failed connection attempts (WiFi connection/ DHCP client/DNS failures – numerator) over the number of total connection attempts (denominator). The list displays the WiFi client with the most connection failures first, in descending order.
	Clicking the hyperlink in the Client device column will direct you to the Site-wide > Monitor > Client: Client device screen. See Section 4.5 on page 192 for more information on this screen.
	Clicking the numerator hyperlink in the # Failed/total connections column will direct you to the Site-wide > Monitor > Connection log screen. See Section 4.8 on page 204 for more information on this screen.
	The Last failed issue column displays the reason for the last connection failure.
Failed connection by SSID	This table displays the number of WiFi clients with WiFi connection/DHCP client/DNS failures in each WiFi network. The list displays the WiFi network with the most connection failures first, in descending order.
	Clicking the hyperlink in the # Failed connections column will direct you to the Site-wide > Monitor > Connection log screen. See Section 4.8 on page 204 for more information on this screen.
Captive portal login issues by client	This table displays the list of WiFi clients with the corresponding number of failed hotspot authentication. The list displays the WiFi client that failed hotspot authentication the most number of times first, in descending order.
	Clicking the hyperlink in the Client device column will direct you to the Site-wide > Monitor > Client: Client device screen. See Section 4.5 on page 192 for more information on this screen.
	Clicking the hyperlink in the # Failed authentication column will direct you to the Site-wide > Monitor > Connection log screen. See Section 4.8 on page 204 for more information on this screen.
Failed connection by AP	This table displays the number of WiFi clients with WiFi connection/DHCP client/DNS failures in each access point. The list displays the access point with the most connection failures first, in descending order.
	Clicking the hyperlink in the # Failed connection column will direct you to the Site-wide > Monitor > Connection log screen. See Section 4.8 on page 204 for more information on this screen.

4.6 Applications Usage

This screen displays usage statistics for applications used in the site. An application can be a specific app or service (for example, Facebook) or a general protocol (for example, HTTP). You can also block or restrict bandwidth for applications at the gateway, and for multiple applications by category.

Click **Site-wide** > **Applications Usage** to access this screen.

Note: You can view this screen by application or by category.

Figure 41 Site-wide > Applications usage: Application View

catio	Ins Usage Application view by	y Access Point	•	Last 24 hours - C		-y couse
	AND NO.	53338 MB				
	2.73 GB	172.73 MD	$\sqrt{1}$	_		
						\land
	Application	0 system 10:0	10 12:00 12:00 16:00 16:00	18:00 19:00 21:00 22:00 30 Me	sr 0130 0300 0430 0600 0	12:00 05:00
Sear	reh • (2	12) applications				
Sear	rch (2) Application Facebook	2) applications Category Social Network	Bandwidth limit 20Mb	Usoge 424.65 MB	%Usoge 1519%	
Sear tus	rch (2 Application Facebook Microsoft Office	2) applications Category Social Network Microsoft Office	Bandwidth limit 20Mb 20Mb	Usoge 424.65 MB 359.39 MB	%Usoge 1519% 1287%	
Sear	rch (2 Application Facebook Microsoft Office HTTPS	2) applications Category Social Network Microsoft Office Web Services	Bandwidth limit 20Mb 20Mb 5Mb	Usoge 424 65 MB 359 59 MB 343 46 MB	503609 1519% 12.87% 12.29%	
Sear	rch (2 Application Facebook Microsoft Office HTTPS Microsoft Outlook	2) applications Category Social Network Microsoft Office Web Services Mail	Bandwidth limit 20Mb 20Mb 5Mb 10Mb	Usoge 424 65 MB 359 59 MB 343.46 MD 22720 MB	%Usage 1519% 12.87% 12.29% 8.13%	
Sear	rch (2 Application Facebook Microsoft Office HTTPS Microsoft Outlook Apple iTunes	2) applications Category Social Network Microsoft Office Web Services Mail Web Services	Bandwidth limit 20Mb 20Mb 5Mb 10Mb 5Mb	Usoge 424.65 MB 359.59 MB 343.46 MB 22720 MB 200.02 MB	%Usoge 1519% 12.87% 12.29% 8.13% 2.16%	
Sear	rch (2 Application Facebook Microsoft Office HTTPS Microsoft Outlook Apple iTunes Google	2) applications Category Social Network Microsoft Office Web Services Mail Web Services Web Services	Bandwidth limit 20Mb 20Mb 5Mb 10Mb 5Mb 5Mb	Usoge 424.65 MB 359.59 MB 343.46 MB 22720 MB 200.02 MB 200.02 MB 142.85 MB	%Usoge 1519% 12.87% 12.29% 8.13% 2.16% 5.11%	
Sear	rch (2 Application Facebook Microsoft Office HTTPS Microsoft Outlook Apple iTunes Google Alibabo CDN	2) applications Category Social Network Microsoft Office Web Services Mail Web Services Web Services	Bandwidth limit 20Mb 20Mb 5Mb 10Mb 5Mb 5Mb 5Mb	Usope 424.65 MB 359.59 MB 343.46 MD 22720 MB 200.02 MB 142.05 MD 11427 MB	505000 1519% 12.87% 12.29% 813% 736% 5.11% 4.09%	
Sear	rch (2 Application Facebook Microsoft Office HTTPS Microsoft Outlook Apple ITunes Google Alibabo CDN Apple ICloud	Category Social Network Microsoft Office Web Services Mail Web Services Web Services Web Services File Sharing	Bandwidth limit 20Mb 20Mb 5Mb 5Mb 5Mb 5Mb 5Mb 5Mb	Usope 424.65 MB 359.59 MB 043.46 MD 22720 MB 200.02 MB 142.05 MB 142.05 MB 114.27 MB	505000 1519% 12.87% 12.29% 813% 736% 5.11% 4.09% 2.69%	
Sear	rch (2 Application Facebook Microsoft Office HTTPS Microsoft Outlook Apple ITunes Google Alibabo CDN Apple ICloud Google Marketing	Category Social Network Microsoft Office Web Services Mail Web Services Web Services Web Services File Sharing Advertising	Bandwidth limit 20Mb 20Mb 5Mb 5Mb 5Mb 5Mb 5Mb 5Mb 5Mb	Usope 424.65 MB 359.59 MB 343.46 MD 22720 MB 200.02 MB 142.05 MB 142.05 MB 114.27 MB 103.08 MB 80.66 MD	505000 1519% 1229% 813% 716% 5.11% 4.09% 3.89% 3.09%	





LABEL	DESCRIPTION
Applications	In Application view, select to view all applications of Nebula Security Appliances / Nebula Access Points, or only applications with bandwidth or block policies applied to Nebula Security Appliances.
	In Category view, select to view all applications of Nebula Security Appliances / Nebula Access Points only.
	Select to view the report for the past day or week. Alternatively, select Custom range to specify a time period the report will span. You can also select the number of results you want to view in a table.
	 Last 24 hours Last 7 days Custom range Update
C	Click this button to reload the data-related frames on this page.
Category View / Application View	Click this button to view statistics by application or category.
y-axis	The y-axis shows the total amount of data used by applications or categories in the site.
x-axis	The x-axis shows the time period over which the data usage occurred.
Keyword	Enter a keyword to filter the list of log entries.
N applications/ categories	This shows the number of applications/categories (N) in the list.
Application/Category	-View Fields
Status	This shows whether the application or category is blocked or allowed within the current site.
Application	This shows the application name.
Category	This shows the name of the category to which the application belongs.
	Note: Click this field in Category view to see all applications in the category.
Bandwidth limit	This shows the bandwidth restriction policy for the application.
Usage	This shows the amount of data consumed by the application, or all applications in the category.
% Usage	This shows the percentage of usage for the application or category.
Limit	Click this to limit the bandwidth for the application on the site's gateway.
	You can apply the restrictions per gateway interface, or for all interfaces.

Table 27 Site-wide > Applications usage

4.7 Summary Report

Use this screen to view statistics for the Nebula Devices and networks in the selected site.

Click **Site-wide** > **Summary report** to access this screen.

d	evices by usage				Location	
2	Name	Model	Usage	# Client	CO O C C M	Amart Map Satellit
	HELEAS	NWASOAX PRO	9.54 GB	17	ФЕ 6 °	Sinchu Zoo
	Product team	WAX650S	3.37 GB	22		1122 @DW (1
	Shawn NSW100-10P	NSW100-10P	79.82 MB	6	Ledus PA MEE	COSTCO Hainer
					Coogle	is Lake
S	SIDs by usage		# Client	% Client	Usage	% Usoge
ī	e-Nebula-FT		28	8750%	21.91 GB	100.00%
	e-Nebula-FT		4	12.50%	0 bytes	0.00%
s	witches by power us Name Shawn NSW100-10	age 2	_	Model NSW100-10P	Power Uso 111.65 Wh	ge

Figure 43 Site-wide > Summary report

The following table describes the labels in this screen.

Table 28 Site-wide> Summary report

LABEL	DESCRIPTION
Summary report	Select to view the report for the past day, week or month. Alternatively, select Custom range to specify a time period the report will span. You can also select the number of results you want to view in a table.
Email report	Click this button to send summary reports by email, change the logo and set email schedules.
Top devices by usage	
	This shows the index number of the Nebula Device.
Name	This shows the descriptive name of the Nebula Device. You can click the name to view the Nebula Device details.
Model	This shows the model number of the Nebula Device.

NCC User's Guide

LABEL	DESCRIPTION					
Usage	This shows the amount of data that has been transmitted by or through the Nebula Device.					
Client	This shows the number of clients currently connected to the Nebula Device.					
Location						
This shows the location	of the site's gateway device on the map.					
Top SSIDs by usage						
# This shows the ranking of the SSID.						
SSID This shows the SSID network name.						
Encryption This shows the encryption method use by the SSID network.						
# Client This shows how many WiFi clients are connecting to this SSID.						
% Client This shows what percentage of associated WiFi clients are connecting to this SSID.						
Usage	nis shows the total amount of data transmitted or received by clients connecting to this SID.					
% Usage	This shows the percentage of usage for the clients connecting to this SSID.					
Top switches by powe	r usage					
#	This shows the ranking of the Nebula Switch.					
Name	This shows the descriptive name of the Nebula Switch.					
Model	This shows the model number of the Nebula Switch.					
Power Usage This shows the total amount of power consumed by the Nebula Switch's conr devices during the specified period of time.						
Ethernet power	This graph shows power used by all PoE Switch ports in the site within the specified time, in Watts.					
Avg	This shows the average power consumption for all Switch ports.					
Мах	This shows the maximum power consumption of the Switch ports.					
Min	This shows the minimum power consumption of the Switch ports.					
y-axis	The y-axis shows how much power is used by all Switches in the site, in Watts.					
x-axis	The x-axis shows the time period over which power consumption is recorded.					

Table 28 Site-wide> Summary report (continued)

4.8 Monitor

Use the **Monitor** menus to check the site features logs and containment list of the Nebula Devices for the selected site. Please click the following links to go to the respective Nebula Devices **Monitor** menus.

- Access points (Section 5.2 on page 235)
- Switches (Section 6.2 on page 276)
- Security router (Section 7.2 on page 316)
- Mobile router (Section 10.4 on page 479)
- Firewall (Section 8.2 on page 342)
- Security gateway (Section 9.2 on page 414)

4.8.1 Containment List

This screen shows a list of clients that are currently blocked in the site by the CDR security service. You can use this screen to release blocked clients. Click **Site-wide** > **Monitor** > **Containment list** to access this screen.

Figure 44 Site-wide > Monitor > Containment list

IP add	iress, MAC addre	55 👻					
ne	IP address	MAC add	User	Event type	Contain	Time rema	Connect
ease/Ac	dd to Exempt List						
IP add	Iress						

The following table describes the labels in this screen.

LABEL	DESCRIPTION		
Search	Enter a MAC or IP address to filter the list of clients.		
Time	This field displays the date and time CDR contained this client.		
IP address	This field displays the IPv4 address of the client contained by CDR.		
MAC address	This field displays the MAC address of the client contained by CDR.		
User This field displays the user name of a client contained by CDR who has been authentical for Internet access. The field is blank if user authentication is not required.			
Event type This field displays details on the category of signature that triggered CDR: Web Filte Anti-Malware or IPS (IDP).			
Containment	This field displays if the client is blocked, quarantined or just triggers an alert.		
Time Remaining (mins.)	This field displays the amount of time left until this client is released by CDR.		
Connect to	This field displays the description of the Access Point or the interface of the Nebula Device that the contained client is connected to.		
Release/Add to Exemp	Dt List		
Release	Select a client and then click this to release this client device from CDR containment.		
Add to Exempt List	Select a client, select an IPv4 address or MAC address, and then click OK to release this client device from CDR containment. This client device's IP or MAC address is exempt from future CDR checking.		

Table 29	Site-wide > Monitor >	Containment list

4.8.2 Site Features Logs

This screen displays events from the Security Appliance within the selected site, such as CDR service events, alerts, and firmware management.

Click Site-wide > Monitor > Site features logs to access this screen.

Figure 45	Site-wide $> N$	Aonitor > Site	features logs
			1001010010090

Feature:			Keyword:			Category:				
Any			Any X			c Any				
	From:				To:					
Range 🔻 2021-03-01				14:16 ▼ 2021-03-31			▼ UTC+8	UTC+8 🗵 🔍 Sea		
	Max ra	nge is 30 days, the d	ates will be aut	:o-adjuste <mark>d</mark> .						
Kewer Older	> (40786	Logs								💎 🕒 Expo
Time	Feature	Category	Detail							
2021-03-29 14:35:32	CDR	Block	Releas	e contain	ed client: Tim	ne's up: IP:192.1	68.2.37,	65.984	NO TABLE	8
2021-03-29 14:35:32	CDR	Block	CDR e	vent detec	oted: IP:192.16	8.2.37,	10.00.00	84.(H		
2021-03-29 09:29:56	CDR	Block	Releas	e contain	ed client: Tim	ne's up: IP:192.1	68.47.160,	nini il	的标准的	
2021-03-29 09:29:56	CDR	Block	CDR e	CDR event detected: IP:192.168.47160,						
2021-03-29 09:29:26	CDR	Block	Releas	e contain	ed client: Tim	ne's up: IP:192.1	68.47.159,	ana) 14	1710.000.000	18
2021-03-29 09:29:26	CDR	Block	CDR e	vent detec	cted: IP:192.16	8.47.159,	ki find			
2021-03-29 09:29:26	CDR	Block	Releas	e contain	ed client: Tim	ne's up: IP:192.1	68.47.158,	ion_7	175438(8)	10
2021-03-29 09:29:26	CDR	Block	CDR e	vent detec	cted: IP:192.16	8.47.158,	1000	1. Sector		

Table 30	Site-wide >	Monitor >	Site	features	logs
					~

LABEL	DESCRIPTION	
Feature	Select the features that you want to view logs for.	
Keyword	Enter a keyword to filter the list of log entries.	
Category	Select the type of log messages you want to view. The available categories will depend on the features you have selected under Feature .	
Range/Before	Select filtering options, set a date, and then click Search to filter log entries by date.	
	Range: Display log entries from the first specified date to the second specified date.	
	Before: Display log entries from the beginning of the log to the selected date.	
Reset filters 🖂	Click this to return the search criteria to the previously saved time setting.	
Search	Click this to update the list of logs based on the search criteria.	
Newer/Older	Click to sort the log messages by most recent or oldest.	
N Logs	This shows the number of log messages (N) in the list.	
Export	Click this button to download the log list as a CSV or XML file to your computer.	
Time	This shows the date and time when the log was recorded.	
	It uses the local time set for the site at Site-wide > Configure > Site settings .	
Feature	Select the feature that created the log message.	
Category	This shows the type of log message, for example "Block". The available categories will depend on the feature.	

LABEL	DESCRIPTION
Detail	This shows the details of the event.
	Note: Click the Nebula Device name link for an Auto configuration recovery alert to go to Site-wide > Devices > Switches: Switch Details screen for more information.
Ę	Click this icon to display a greater or lesser number of configuration fields.

Table 30 Site-wide > Monitor > Site features logs (continued)

4.9 Configure

Use the **Configure** menus to set the WiFi security settings for Nebula Devices of the selected site. Please click the following links to go to the respective Nebula Devices **Configure** menus.

- Access points (Section 5.3 on page 247)
- Switches (Section 6.3 on page 288)
- Security router (Section 7.3 on page 319)
- Mobile router (Section 10.2 on page 471)
- Firewall (Section 8.3 on page 350)
- Security gateway (Section 9.3 on page 422)

4.9.1 WiFi SSID Settings

This screen allows you to configure up to eight different SSID profiles for your Nebula Devices. An SSID, or Service Set IDentifier, is basically the name of the WiFi network to which a WiFi client can connect. The SSID appears as readable text to any device capable of scanning for WiFi frequencies (such as the WiFi adapter in a laptop), and is displayed as the WiFi network name when a person makes a connection to it.

Click Site-wide > Configure > WiFi SSID settings to access this screen.

Igule 46 Sile-wide	> Compute > WH 33D serings	
Advanced mode: Beta	What is this?	
+ Add SSID network		
No.	1	
Name	GSBU_Test	
Enabled		
🂎 Programmable SSID 🛛 Beta		
	Name:	× *
	PSK:	× (optional)
Broadcasting APs	All APs	•
Tagging	Tag	
	Enable SSID on APs with any of the specified tags	
Guest Network		
SSID advanced settings		Edit
WLAN security	WPA3-Personal	
Sign-in method	Disable	
Band mode	2.4 GHz 5 GHz 6 GHz	
VLAN ID	1	
Rate limiting	@unlimited Kb/s®unlimited Kb/s	
Captive portal customization		Edit
Theme	Modern	

Figure 46 Site-wide > Configure > WiFi SSID settings

LABEL	DESCRIPTION	
Advanced mode	Select Off to disable Advanced mode.	
	This allows you to create SSID profiles by only specifying an SSID name and optional password. NCC sets all other WiFi settings to default.	
+ Add SSID network	Click this button to configure up to eight different SSID profiles for your Nebula Device	
No.	This shows the index number of this profile.	
delete	Click this icon to remove the SSID profile.	
Name	This shows the SSID name for this profile. Click the text box and enter a new SSID if you want to change it.	
Enabled	Click to turn on or off this profile.	
Programmable SSID	Select On to have each Nebula Device that uses this SSID generate a unique SSID name and pre-shared key (PSK) based on the Nebula Device's model name, serial number, or MAC address.	
	For example, a hotel can install a Nebula Device in each room and then have each Nebula Device broadcast a unique SSID based on the room number: FreeWiFi_Room1, FreeWiFi_Room2, FreeWiFi_Room3, and so on.	

LABEL	DESCRIPTION
Name	Name : Enter a programmable SSID name in the format PREFIX+VALUE(X). This name overrides the original SSID name.
	 PREFIX: Optional prefix to add to the SSID, for example "FreeWiFi_". To use "\$" in the SSID name, enter "\$\$"
	 VALUE: Specify a Nebula Device value to use to generate the SSID name. Use one of the following: \$AP = Nebula Device device name. \$MAC = Nebula Device MAC address. \$SN = Nebula Device serial number.
	 X: Specify how many characters of the Nebula Device value to use in the SSID. A positive number means the first X characters, and a negative number means the last X characters.
	Example: <i>FreeWiFi_Room\$AP(-3)</i> generates an SSID called "FreeWiFi_Room" + the last three characters of the access point device name.
PSK	PSK : Enter an optional programmable PSK in the format GENTYPE(Y).
	 GENTYPE: Specify how the Nebula Device will generate a random PSK. \$GENMIX = The Nebula Device generates a mix of random letters and numbers. \$GENNUM = The Nebula Device generates a mix of random numbers only. \$AP = Nebula Device device name. \$MAC = Nebula Device device address. \$SN = Nebula Device serial number. Y = Specify the length of the PSD. The minimum length is 8.
	Example 1: \$GENNUM(10) generates a unique 10-character PSK for this SSID, consisting only of numbers.
	Example 2: \$MAC(-5)\$SN(-5) uses the MAC address's last 5 characters and the serial number's last 5 characters (for example, 8E3AE02451).
	Example 3: ZYXEL-\$GENMIX(4) appends the fixed characters 'ZYXEL' and generates a unique 4-character mix of random letters and numbers (for example, ZYXEL-3c4d).
	Note: You can specify a fixed PSK for this SSID at Site-wide > Configure > Access points / Security router > SSID advanced settings.
Broadcasting APs	Select All APs or specify the AP to use this SSID profile.
Tagging	Enter or select the tags you created for Nebula Devices in the Site-wide > Devices > Access points / Security router / Mobile router screen. The SSID profile will only be applied to Nebula Devices with the specified tag.
	If you leave this field blank, this SSID profile will be applied to all Nebula Devices in the site.

Table 31 Site-wide > Configure > WiFi SSID settings (continued)

LABEL	DESCRIPTION
Guest Network	Select On to set this WiFi network as a guest network. Layer 2 isolation and intra-BSS blocking are automatically enabled on the SSID. WiFi clients connecting to this SSID can access the Internet through the Nebula Device but cannot directly connect to the LAN or the WiFi clients in the same SSID or any other SSIDs.
	Note: In your VLAN-enabled network, if the SSID's gateway MAC address and the Nebula Device's gateway MAC address are different and belong to different VLANs, you need to manually add the SSID's gateway MAC address to the layer 2 isolation list. See Section 5.3.1 on page 247.
	Note: If you have a Nebula Security Appliance installed in the site but the gateway interface with the same VLAN ID is not configured as a guest interface, Smart Guest/VLAN network tip , click here . displays after you select On . Click here to open a screen where you can directly select to use the interface as a Guest interface.
	Smart VLAN X
	This SSID has Guest network turned ON. To limit the access to internet only, Guest function can also be enabled on the gateway VLAN interface.
	Note: This setting is not recommended if wired connections or SSIDs using the same VLAN need access to other interfaces.
	VLAN ID 1 (2-4094)
	Guest (Enable Internet access only)
	Close Continue
SSID advanced setting	IS
Edit	Click this button to go to the SSID advanced settings screen and configure WiFi security and advanced settings, such as band selection, enable assisted roaming and U-APSD (Unscheduled automatic power save delivery). See Table 48 on page 249 for more information on assisted roaming and U-APSD.
WLAN security	This shows the encryption method used in this profile.
Sign-in method	This shows the authentication method used in this profile or Disable .
Band mode	This shows whether the SSID use either 2.4 GHz band, 5 GHz band, or the 6 GHz band.
VLAN ID	This shows the ID number of the VLAN to which the SSID belongs.
Rate limiting	This shows the maximum incoming/outgoing transmission data rate (in Kbps) on a per- station basis.
Captive portal custom	ization
Edit	Click this button to go to the Captive portal screen and configure the captive portal settings. See Section 5.3.2 on page 256.
Theme	If captive portal is enabled, this shows the name of the captive portal page used in this profile.

Table 31 Site-wide > Configure > WiFi SSID settings (continued)

4.9.2 Alert Settings

Use this screen to set which alerts and reports are created and emailed. You can also set the email addresses to which an alert is sent. Click **Site-Wide** > **Configure** > **Alert settings** to access this screen.

Note: NCC's Smart Alert Engine uses knowledge of network topology and cross-device functionality to only generate alerts for unexpected events. This helps avoids unnecessary emails and notifications.

For example, an Access Point is receiving power from a PoE switch. If the Access Point loses power because its Ethernet cable is disconnected, NCC generates an alert. If the Access Point loses power because the Switch has a PoE schedule that disables power to the Access Point, NCC does not generate an alert.

Figure 47 Site-Wide > Configure > Alert settings

Site-wide > Configure > <u>Alert settings</u> Alert settings		
Recipient		
All site administrators	Email to all site administrators	
Custom email recipient	✓ E.g. nebula@zyxel.com	
System alerts 👔		
Wireless	Both 🔹 60 💌 minutes after AP goes offline	
	Hide additional recipients	
	E.g. nebula@zyxel.com	
Switches	Both Both	
	Hide additional recipients	
	E.g. nebula@zyxel.com	
	Email G0 minutes Any switch port goes down	
	Hide additional recipients	
	E.g. nebula@zyxel.com	
Security Appliance	Both 🔹 15 💌 minutes after the appliance goes offline	
	Hide additional recipients	
	E.g. nebula@zyxel.com	
	Email A VPN connection is established or disconnected 	
	Hide additional recipients	
	Eg.nebula@zyxel.com	
Mobile router	Both Both Both Both Both B	
	Hide additional recipients	
	E.g. nebula@zyxel.com	
Other	Email Configuration settings are changed	
	Hide additional recipients	
	Eg. nebula@zyxel.com	

Table 32	Site-Wide >	Configure >	Alert setting	JS

LABEL	DESCRIPTION
Recipient	
All site administrators	Select this to send alerts to all site administrators for the current site.
Custom email recipient	Enter the email addresses to which you want to send alerts.

LABEL	DESCRIPTION	
Notification Type	 For each alert, you can set how to receive alert notifications: Email: Alert notifications are sent by email to configured administrators, custom email 	
	 recipients, and additional recipients. In-app Push: Alert notifications are sent to site administrators who are logged into the 	
	 Nebula Mobile app. This type of notification is not available for some features. Both: Alert notifications are sent by email and app notification 	
	 Disabled: No alerts are sent. 	
Show additional recipients	Add additional user accounts who will receive email and in-app notifications for the alert.	
System Alerts		
Wireless	Specify how long in minutes the NCC waits before generating and sending an alert when an AP becomes offline.	
Switches	Specify how long in minutes the NCC waits before generating and sending an alert when a port or a Switch goes offline.	
Security Appliance	Select the check box to have the NCC generate and send an alert by email when the following events occur:	
	 A Security Appliance goes offline. Any DHCP pool on the Security Appliance runs out of IP addresses. A VPN connection to or from the Security Appliance is established or disconnected. The WAN connectivity status changed. 	
Mobile router	Specify how long in minutes the NCC waits before generating and sending an alert when a mobile router goes offline.	
Other	Specify whether to send an alert each time configuration settings are changed.	

Table 32 Site-Wide > Configure > Alert settings (continued)

4.9.3 Firmware Management

Use this screen to schedule a firmware upgrade. You can make different schedules for different types of Nebula Devices in the site or create a schedule for a specific Nebula Device. Click **Site-Wide** > **Configure** > **Firmware management** to access these screens.

4.9.3.1 Firmware Management Overview Screen

Use this screen to schedule a firmware upgrade for each Nebula Device type. You can make different schedules for different types of Nebula Devices in the site. Click **Site-wide > Configure > Firmware management > Overview** to access this screen.

Site-wide > Configure > Firmware management Firmware management		
Overview Devices		
Access Point	Upgrade available	What is this?
	1 of device with per devices schedules, please check in <u>Devices</u> tab.	device type schedule is selected, the administrators will
	Settings Upgrade policy O Auto upgrade at Monday ▼ 02.00 ▼ UTC+8.0 Upgrade at 2023-03-29 14:00 UTC+8.0 Upgrade now	Nebula portal indiriculation in davani in davanie. The Nebula portal initi display the next upcoming schedule based on your settings. You can chose upgrade now, reschedule or ignore upgrade if you don't want to upgrade. If no action is taken, the upgrade will be performed as shown in the defined schedule. For more information, please visit our community.
	Ignore upgrade Firmware type Latest	
Switch	Upgrade available	What is this?
	Settings	When your device have a newer firmware available and per device type schedule is selected, the administrators will
	Upgrade policy O Auto upgrade at Monday VIC+8.0 UTC+8.0	receive an email notification 14 days in advance. The Nebula portal will display the next upcoming schedule based on your settings.
	Upgrade at 2023-03-29 🛅 14:00 👻 UTC+8.0	You can choose upgrade now, reschedule or ignore upgrade if you don't want to upgrade. If no action is taken the upgrade will be performed as shown in the defined
	O Ignore upgrade	schedule. For more information, please visit our <u>community</u> .
	Firmware type Latest 👻	
Firewall	Upgrade available	What is this?
	Settings	When your device have a newer firmware available and per device type schedule is selected, the administrators will
	Upgrade policy Auto upgrade at Monday UTC+80 Upgrade at 2023-03-29 Interview UTC+80 Upgrade now Informe upgrade	receive an email notification 14 days in advance. The Nebula portal will display the next upcoming schedule based on your settings. You can choose upgrade now, reschedule or ignore upgrade if you don't want to upgrade. If no action is taken, the upgrade will be performed as shown in the defined schedule. For more information, please visit our <u>community</u> .
	Firmware type	
Mobile Router	Upgrade available	What is this?
	Settings	When your device have a newer firmware available and per device type schedule is selected, the administrators will
	Upgrade policy O Auto upgrade at Monday V 02:00 V UTC+8.0	receive an email notification 14 days in advance. The Nebula portal will display the next upcoming schedule based on your settings.
	Upgrade at 2023-03-29	You can choose upgrade now, reschedule or ignore upgrade if you dan't want to upgrade. If no action is taken, the upgrade will be performed as shown in the defined schedule. For more information, please visit our community.
	Firmware type Latest	

Figure 48 Site-Wide > Configure > Firmware management > Overview

Table 33	Site-Wide >	· Confiaure >	Firmware	management >	> Overview
10010-00		Connigoro	1 11 11 11 01 0	managomorn	01011011

LABEL	DESCRIPTION
Access Point / Switch /	Security router / Firewall or Security Gateway / Mobile Router
Upgrade available	 This shows the status of the Nebula Device's firmware in your site. Up to date is displayed if all the Nebula Device(s) of a particular type (for example, all Switches) in your site are using the latest firmware version. Upgrade available is displayed if there is firmware update available for any of the Nebula Device(s) of a particular type in your site. Click Devices to see a table list of your Nebula Device(s) that can receive this upgrade.
	Applicable devices X
	Firmware Type: Stable
	Device type Model MAC address S/N Current version Schedule upgrade version
	Access point NWA1123-AC HD 50-6A-80-ED-4A-62 \$172V37001024 V6.25(ABIN.4) V6.25(ABIN.6)
	Access point WAX610D BC-CE-VE-DC-ER-CE \$2021 36240587 V6.30(ABTE.0) V6.30(ABTE.4)
	Close
	 Locked is displayed if all the Nebula Device(s) of a particular type (for example, all Switches) in your site are using a specific version of firmware that Zyxel customer support is monitoring for troubleshooting. No devices is displayed if there is no Nebula Device of a particular type (for example, Mobile Router) registered in your site.

LABEL	DESCRIPTION
Settings	Create a schedule for each Nebula Device type. The following Upgrade policy are available:
	 Select Auto upgrade at to create a recurring schedule. With a recurring schedule, NCC will check and install the firmware when a new firmware release is available for each Nebula Device type.
	 Select Upgrade at to install the firmware at a specific date and time (up to 1 month from now) when firmware update is available for each Nebula Device type.
	Note: Due to network bandwidth and number of Nebula Devices per site, not all Nebula Devices may get the firmware upgrade on the specified date/ time
	This field's setting will change to the Auto upgrade at schedule after performing the firmware update.
	 Select Upgrade now to immediately install the firmware for each Nebula Device type. Then select the Firmware type (Stable or Latest (default)).
	Note: This button is selectable only when there is firmware update available. This field's setting will return to it's previous setting (Auto upgrade at or Ignore upgrade) after performing the firmware update.
	• Select Ignore upgrade if you choose not to install the firmware.
	Note: NCC will still perform a mandatory upgrade if the Nebula Device's firmware have security vulnerabilities, and/or lack key performance improvements. When the schedule for Auto upgrade at is earlier than the mandatory upgrade schedule, then the Auto upgrade at schedule has priority.
Firmware type	Set the type of firmware to be installed for each Nebula Device type.
	 Select Stable to install a firmware that may not have the latest features but has passed Zyxel internal and external testing. Select Latest to install the most recently release firmware with the latest features, improvements, and bug fixes.
	Note: This field is hidden when Ignore upgrade is selected in Settings . We generally recommend updating to the Latest firmware type so that you get the latest features, improvements, and bug fixes. All firmware releases are thoroughly tested internally by our engineers. If your requirements are such that you prefer fewer updates, go with the Stable firmware type.

Table 33 Site-Wide > Configure > Firmware management > Overview (continued)

4.9.3.2 Firmware Management Devices Screen

Use this screen to make different firmware upgrade schedules for different types of Nebula Devices in the site. Click **Site-wide > Configure > Firmware management > Devices** to access this screen.

Note: While installing a firmware update, the Nebula Device will continue to operate normally until it reboots. The reboot will take 3 to 5 minutes, so it is best to pick an upgrade time that has minimal impact on your network.



Overview	Dev	ices									
atus	Device type	Tag		Model	Current version	Firmware	status	Firmware	type Av	ailability	Locked
Any	Any	▼ Any	•	Any 👻	Any 👻	Any	•	Any	· •	ny 👻	Any
Status	Device type	Model	Reset (1) selected in (7) dev S/N	rices Current version		Firmware s	tatus A	vailability	Firmware type	Upgrade sc
Status	Device type Switch	Model NSW100-10P	MAC address	Selected in (7) dev	Current version V3.00(ABGO.2) 11/19/2	2019	Firmware s Good ()	tatus A U	vailability pgrade availab	Firmware type	Upgrade so No
Status	Device type Switch Access point	Model NSW100-10P WAX510D	MAC address	S/N S/N SE \$1731 19000031 SE \$3731 40103451	Current version V3.00(ABGO.2) 11/19/2 V6.00(ABTF.0)IT_20221	2019 1019121600	Firmware s Good (1) Custom (1)	totus A U U	vailability pgrade availab pgrade availab	Firmware type le Stable le General Availabi	Upgrade so No ility No
Status	Device type Switch Access point Access point	Model NSW100-10P WAX510D NWA50AX	Reset (1 MAC address Do.EC.A2.00.44 Do.EC.A2.00.44 Do.EC.A2.00.44	selected in (7) dev S/N CBA \$1701 SE \$2101 SE \$2101 SE \$2101	Current version V3.00(ABGO.2) 11/19/2 V6.00(ABTF.0) T_20221 V1.00(ABYW.0)	2019 1019121600	Firmware s Good () Custom () Warning ()	tatus A U U U U	<mark>vailability</mark> pgrade availab pgrade availab pgrade availab	Firmware type le Stable le General Availabi le General Availabi	Upgrade sc No ility No ility No
Status	Device type Switch Access point Access point	Model NSW100-10P WAX510D NWA50AX WAX650S	Reset (1 MAC address D0-EC-A0-00-44 D0-EC-A0-00-44 D0-EC-A0-00-44	selected in (7) dev S/N SBA<	ices Current version V3.00(ABGO.2) 11/19/2 V6.00(ABTF.0) T_20221 V1.00(ABYW.0) V6.50(ABRM.0)b5	2019 1019121600	Firmware s Good () Custom () Warning (Custom ()	tatus A U U U U U	vailability pgrade availab pgrade availab pgrade availab pgrade availab	Firmware type le Stable le General Availabi le General Availabi	Upgrade so No ility No ility No No
	Device type Switch Access point Access point Access point	Model NSW100-10P WAX510D NWA50AX WAX650S NWA50AX PRO	Reset 1 MAC address 0000,4000,400 0000,4000,400,400,400 0000,400,400 0000,400,400,400,400 0000,400,400	selected in (?) dev S/N OEA ettail 10000031 OEE ettail 10000031 OEE ettail 10000031 OE ettail 10000031 OE ettail 10000031 OE ettail 10000031 OE ettail 10000033 OE ettail 10000033 OE ettail 10000035	Current version V3 00(ABGO 2) 11/19/2 V6 00(ABTF 0) 1T_20221 V1.00(ABTV 0.) V6 50(ABRM 0) b5 V6 50(ACGE 0) b6	2019 1019121600	Firmware s Good () Custom () Warning () Custom ()	tatus A U U U U U U U	vailability pgrade availab pgrade availab pgrade availab pgrade availab	Firmware type le Stable le General Availabi le General Availabi le Beta le General Availabi	Upgrade so No ility No ility No No ility No
	Device type Switch Access point Access point Access point Access point	Model NSW100-10P WAX510D NWA50AX WAX650S NWA50AX PRO NWA90AX PRO	Reset 1 MAC address B000,4000,44 B000,4000,44 B000,4000,44 B000,4000,4000,4000,4000,4000,4000,4000	selected in (?) dev S/N OBA 01201 120000031 DBC 02001 40100451 DBC 02001 40100451 DBC 02001 40100451 DBC 02001 40100451 DBC 02001 40100035 DBC 02001 401000035 DBC 02001 401000035	Current version V3 00(ABGO.2) [11/19/2 V6.00(ABTF.0)IT_20221 V1.00(ABYW.0) V6.50(ABRM.0)b5 V6.50(ACGE.0)b6	2019 1019121600	Firmware s Good () Custom () Warning () Custom () Custom ()	totus A U U U U U U U U U	vailability pgrade availab pgrade availab pgrade availab pgrade availab pgrade availab	Firmware type le Stable le General Availabi le General Availabi le Beta le General Availabi	Upgrade so No lility No No lility No lility No
--											

LABEL	DESCRIPTION							
Upgrade Now	Click this to immediately install the firmware on the selected Nebula Devices.							
	This button is selectable only when there is firmware update available for all the selected Nebula Devices.							
	Then, select the Firmware type to be installed.							
	Select Stable to install a firmware that may not have the latest features but has passed							
	 Zyxel internal and external testing. Select Latest to install the most recently release firmware with the latest features, improvements, and bug fixes. 							
	Upgrade now X							
	You are going to upgrade your device's firmware now. While installing a firmware update, your service will continue to operate normally until they reboot as the final step in the upgrade process. The reboot takes 3-5 minutes, so it is the best to pick an upgrade time with minimal expected network usage. Please click QK to continue							
	Firmware type: Latest 💌							
	Cancel Ok							
	Nebula Devices. You can select to upgrade firmware according to the site-wide schedule configured for the Nebula Device type in the site, create a recurring schedule, edit the schedule with a specific date and time when firmware update is available for all the selected Nebula Devices, or immediately install the firmware. With a recurring schedule, the NCC will check and perform a firmware update when a new firmware release is available for any of the selected Nebula Devices. If the NCC service is downgraded from Nebula Professional Pack to Nebula Base, the Nebula Devices automatically changes to adhere to the side-wide schedule.							
	Schedule upgrade X							
	Note: Schedule upgrade will follow each site's time zone.							
	Upgrade policy Follow device type settings <u>What is this?</u> Auto upgrade at every Week on Monday at 02.00 Upgrade at 2023-03-29 IA 30							
	Upgrade now Device(s) below will be upgraded as required time.							
	Device type Model MAC address S/N Current version Schedule upgrade version							
	Firewall USG FLEX 500 BREC A 212/37 EA C1891 45/00/173 N/A N/A							
	AULOTOS (JUDIT), NYVAHUARA SI SISTERICI NY SITTERICI NY NYA Access point WAX650S 20.21.07.03.0128 202107030128 N/A N/A							
	Access point WAX650S 20/2107/03/01/29 202107/03/01/29 N/A N/A							
	Access point NWA220AX-6E 20:22:06:25:00:08 202206250008 N/A N/A							
	Mobile router NR7101 DDPOFERODECE DMOSTER N/A N/A Switch GS2220-10HP 20.22.09.03.20.01 202209032001 N/A N/A							
	Cancel Update							

NCC User's Guide

LABEL	DESCRIPTION
Reset	Select one or more Nebula Devices, and then click Reset to allow the Nebula Devices to follow the site-wide firmware management settings.
Status	This shows the status of the Nebula Device.
	 Green: The Nebula Device is online and has no alerts. Amber: The Nebula Device has alerts. Red: The Nebula Device is offline. Gray: The Nebula Device has been offline for 7 days or more.
Device type	This shows the type of the Nebula Device.
Model	This shows the model number of the Nebula Device.
Tag	This shows the tag created and added to the Nebula Device.
Name	This shows the descriptive name of the Nebula Device.
MAC address	This shows the MAC address of the Nebula Device.
S/N	This shows the serial number of the Nebula Device.
Current version	This shows the version number of the firmware the Nebula Device is currently running. It shows N/A when the Nebula Device goes offline and its firmware version is not available.
Firmware status	The status shows Good if the Nebula Device is running a stable firmware and no immediate action is required. The installed firmware does not have the latest features but provides the smoothest operation.
	The status shows Warning if a newer firmware is available and immediate action is recommended. The newer firmware may contain security enhancements, new features, and performance improvements.
	The status shows Critical if a newer firmware is available and immediate action is required. The firmware may have security vulnerabilities and/or lack key performance improvements.
	The status shows Custom if the Nebula Device is running a firmware with specialized features that is not available to the general public.
	The status changes to Upgrading after you click Upgrade Now to install the firmware immediately.
Firmware type	This shows Stable when the installed firmware may not have the latest features but has passed Zyxel internal and external testing.
	This shows Latest when the installed firmware is the most recent release with the latest features, improvements, and bug fixes.
	This shows General Availability when the installed firmware is a release before Latest , but is still undergoing Zyxel external testing.
	This shows Dedicated when the installed firmware is locked and Zyxel support is monitoring. Contact Zyxel customer support if you want to unlock the firmware in order to upgrade to a later one.
	This shows Beta when the installed firmware is a release version for testing the latest features and is still undergoing Zyxel internal and external testing.
	This shows N/A when the Nebula Device is offline and its firmware status is not available.
Availability	This shows whether the firmware on the Nebula Device is Up to date , there is firmware update available for the Nebula Device (Upgrade available), or a specific version of firmware has been installed by Zyxel customer support (Locked). Contact Zyxel customer support if you want to unlock the firmware in order to upgrade to a later one.

Table 21	Sita wida >	Configuro	> Eirmuuara	management	Daviana	(continued)
10016 54	Sile-wide >	Conngole		munuyemeni	Devices	(commoed)

LABEL	DESCRIPTION
Upgrade scheduled	This shows the date and time when a new firmware upgrade is scheduled to occur. Otherwise, it shows Follow upgrade time and the Nebula Device sticks to the site-wide schedule or No when the firmware on the Nebula Device is up-to-date or the Nebula Device goes offline and its firmware status is not available.
	A lock icon displays if a specific schedule is created for the Nebula Device, which means the Nebula Device firmware will not be upgraded according to the schedule configured for all Nebula Devices in the site.
Last upgrade time	This shows the last date and time the firmware was upgraded on the Nebula Device.
Schedule upgrade version	This shows the version number of the firmware which is scheduled to be installed.
	Click this icon to display a greater or lesser number of configuration fields.

Table 34 Site-wide > Configure > Firmware management > Devices (continued)

4.9.4 Cloud Authentication

Use this screen to view and manage the user accounts which are authenticated using the NCC user database, rather than an external RADIUS server. Click **Site-wide** > **Configure** > **Cloud authentication** to access these screen.

Note: The changes you made in this screen apply only to the current site. To change the cloud authentication settings for all sites in the organization, go to **Organization-wide** > **Organization-wide manage** > **Cloud Authentication** (see Section 4.9 on page 207).

Note: For more information on user account types, see Section 11.4.7.1 on page 541.

4.9.4.1 Cloud Authentication User Screen

Use this screen to view and manage regular NCC network user accounts. Click **Site-wide** > **Configure** > **Cloud Authentication** > **User** to access this screen.

Figure 50 Site-wide > Configure > Cloud Authentication > User

uthorization •	Remove users	/PN access -	VLAN attribute -	Q. Search i	Jsers_,	• 1) selecti	ed in 1 Use			🕒 In	nport + Ac	id 🕒 Expor
🗸 Email	Userna Des	scrip 80	2.1X VPN ac	Authori	Expire in	Login by DPPSK 🕸	VLAN a	2FA Sta	Bypass	Authori	Create	Created at

Note: Some of the actions on this screen are only available if your administrator account has full access to the organization.

Table 35	Site-wide >	Configure >	Cloud Authentication >	User
10010-00		Gormgoro		0001

LABEL	DESCRIPTION
Authorization	Select one or more than one user account and click this button to configure the authorization settings for the selected user accounts. Authorize users (this site only) Does not expire
	 Revoke authorization (this site only) Update
Remove users	Select one or more than one user account and click this button to remove the selected user accounts.
VPN access	Select one or more than one user account and click this button to configure whether the accounts can be used to connect to the organization's networks through VPN.
VLAN attribute	Select one or more than one user account and click this button to assign the users to a specific VLAN ID, or clear the VLAN ID. Then click Update .
Search users	Enter a key word as the filter criteria to filter the list of user accounts.
N User	This shows how many user accounts (N) match the filter criteria and how many user accounts of the selected type are created in total.
Import	Click this button to create user accounts in bulk by importing a complete list of all new users in an Excel file. Bulk Import Bulk Import Browse Or drag file here Close
Add	Click this button to create a new user account. See Section 4.9.4.2 on page 221.
Export	Click this button to save the account list as a CSV or XML file to your computer.
Email	This shows the email address of the user account.

NCC User's Guide

LABEL	DESCRIPTION
Username	This shows the user name of the user account.
Description	This shows the descriptive name of the user account.
802.1X	This shows whether 802.1X (WPA-Enterprise) authentication is enabled on the account.
VPN access	This shows whether the accounts can be used to connect to the organization's networks through VPN.
Authorized	This shows whether the user has been authorized in this site or not.
Expire in (UTC)	This shows the date and time that the account expires.
	This shows if authentication is disabled for this account.
	This shows Never if the account never expires.
	This shows Multiple value if the account has different Expire in values across different sites.
Login by	This shows whether the user needs to log in with the email address and/or user name.
DPPSK	This shows the account's dynamic personal pre-shared key (DPPSK), if one is set.
VLAN assignment	This field is available only when the account type is set to User .
	This shows the VLAN assigned to the user.
2FA Status	This shows whether the account has set up two-factor authentication yet.
Bypass 2FA	This shows whether the account is allowed to bypass two-factor authentication, if two-factor authentication is enabled on a captive portal or VPN gateway.
Authorized by	This shows the email address of the administrator account that authorized the user.
	If the account has been authorized by different administrators across different sites, it shows Multiple value .
Created by	This shows the email address of the administrator account that created the user.
Created at	This shows the date and time that the account was created.
R	Click this icon to display a greater or lesser number of configuration fields.

 Table 35
 Site-wide > Configure > Cloud Authentication > User (continued)

4.9.4.2 Cloud Authentication MAC Screen

Use this screen to view and manage Nebula Device user accounts, used for MAC-based authorization. Click **Site-wide** > **Configure** > **Cloud Authentication** > **MAC** to access this screen.

Figure 51 Site-wide > Configure > Cloud Authentication > MAC

Site-wide > Configure > <u>Cloud authentication</u> Cloud authentication (Site :Mesh)									
User MAC DPPSK*									
	e users Q Searc	ch users	• 1) selected	in 2 Users	ि In	nport + Add 🕒 Exp	port		
MAC address	Description	Account type	Authorized	Authorized by	Expire in	Created at	[
MAC address 80:EA:96:3B:1F:80	Description shaw-iphone	Account type MAC	Authorized All sites	Authorized by shaw.hsiao@zyxel	Expire in Never	Created at 2019-11-01 09:40	[
MAC address 80:EA:96:3B:1F:80 80:6E:BF:12:A2:B0	Description shaw-iphone ZF4	Account type MAC MAC	Authorized All sites All sites	Authorized by shaw.hsiao@zyxel shaw.hsiao@zyxel	Expire in Never Never	Created at 2019-11-01 09:40 2019-12-16 13:15			

Note: Some of the actions on this screen are only available if your administrator account has full access to the organization.

Table 36 Site-wide > Configure > Cloud Authentication > MAC

LABEL	DESCRIPTION				
Authorization	Select one or more than one account and click this button to configure the authorization settings for the selected user accounts.				
	• Authorize users (this site only)				
	O Does not expire				
	O Expires in: X minutes -				
	Revoke authorization (this site only)				
	Update				
Remove users	Select one or more than one user account and click this button to remove the selected user accounts.				
Search users	Enter a key word as the filter criteria to filter the list of user accounts.				
N User	This shows how many user accounts (N) match the filter criteria and how many user accounts of the selected type are created in total.				
Import	Click this button to create user accounts in bulk by importing a complete list of all new users in an Excel file.				
	Bulk Import X				
	"Bulk Import" supports for faster inputting. Please follow this template to import				
	Browse				
	Or drag file here				
	Close				
Add	Click this button to create a new user account. See Section 4.9.4.3 on page 223.				
Export	Click this button to save the account list as a CSV or XML file to your computer.				
MAC address	This shows the MAC address of the user account.				
Description	This shows the descriptive name of the user account.				
Account type	This shows the type of user account: USER, MAC, or DPPSK.				
Authorized	This shows whether the user has been authorized in this site or not.				
Authorized by	This shows the email address of the administrator account that authorized the user.				
	If the account has been authorized by different administrators across different sites, it shows Multiple value .				
Expire in	This shows the date and time that the account expires.				
	This shows if authentication is disabled for this account.				
	This shows Never if the account never expires.				
	This shows Multiple value if the account has different Expire in values across different sites.				

T O (/ I ² II
Table 36	Site-wide > Contigure > Cloud Authentication > MAC	(continued)

LABEL	DESCRIPTION
Created at	This shows the date and time that the account was created.
	Click this icon to display a greater or lesser number of configuration fields.

4.9.4.3 Cloud Authentication DPPSK Screen

Use this screen to view and manage DPPSK network user accounts. Click **Site-wide** > **Configure** > **Cloud Authentication** > **DPPSK** to access this screen.

Figure 52 Site-wide > Configure > Cloud Authentication > DPPSK

e-wide > Configure > <u>Cloud authentication</u> ioud authentication (Site :Mesh)									
ser 1	MAC DPPSK	. 🖤							
Auth	norization + Ren	move users Prin	nt Q Search	h users	• 1	selected in 35 (Jsers	ि Import	+ Add 🕒 Export •
	Email	Username	Account ty	DPPSK 🕸	VLAN ID	Authorized	Expire in	Created by	Created at
	kellyyang@zy	kellyyang	DPPSK	••••		No			2020-08-05 18:15
	CvKtTKFchrF	TeacherDemo	DPPSK	•••••	P.I	No			2020-08-05 19:17
	Wiflex-Yx64px	Wiflex-Yx64px	DPPSK		D1	No			2020-08-05 19:17
			DPPSK	*******	D+	No		shawn.hsiao@	2021-08-09 15:22
			DPPSK		D-1	No		shawn.hsiao@	2021-08-09 15:22
			DPPSK	*******	D+	No		shawn.hsiao@	2021-08-09 15:22
			DPPSK	•••••	D+	No		shawn.hsiao@	2021-08-09 15:22
			DPPSK		D f	No		shawn.hsiao@	2021-08-09 15:22
			DPPSK	•••••	D+	No		shawn.hsiao@	2021-08-09 15:22
			DPPSK	*******	D (No		shawn.hsiao@	2021-08-09 15:22
4						K K	Page 1 of	4 > > Resul	ts per page: 10 👻

Table 37 Site-wide > Configure > Cloud Authentication > DPPSK

LABEL	DESCRIPTION
Authorization	Select one or more than one user account and click this button to configure the authorization settings for the selected user accounts.
	• Authorize users (this site only)
	O Does not expire
	O Expires in: X minutes -
	Revoke authorization (this site only)
	Update
Remove users	Select one or more than one user account and click this button to remove the selected user accounts.

LABEL	DESCRIPTION						
Print	Click this button to print the unique dynamic personal pre-shared key (DPPSK) and expiry time of each selected user account.						
	The account details can be cut into cards, and then given to users in order to grant them WiFi network access.						
	DPPSK						
	?; nduzjauv9f ?; paatdtcgh4						
	Expired in: Never Never						
Search users	Enter a key word as the filter criteria to filter the list of user accounts.						
N Users	This shows how many user accounts (N) match the filter criteria and how many user accounts of the selected type are created in total.						
Import	Click this button to create user accounts in bulk by importing a complete list of all new users in an Excel file.						
	Bulk Import ×						
	"Bulk Import" supports for faster inputting. Please follow this template to import						
	Browse						
	Or drag file here						
	Close						
Add	Click this button to create a single new account, or a batch of accounts.						
	• Single DPPSK: See Section 11.4.7.7 on page 550.						
	Batch create DPPSK: See Section 11.4.7.8 on page 551.						
Export	Click this button to save the account list as a CSV or XML tile to your computer.						
Email	This shows the email address of the user account.						
Username	This shows the user name of the user account.						
	This shows the appound's dynamic percend pro shared key (DPPSK)						
	This shows the VLAN assigned to the account						
	This shows the descriptive name of the user account						
Authorized	This shows whether the user has been authorized in this site or not						
Expire in	This shows the date and time that the account expires.						
	This shows if authentication is disabled for this account.						
	This shows Never if the account never expires.						
	This shows Multiple value if the account has different Expire in values across different sites						
Created by	This shows the email address of the administrator account that created the user.						
- /							

Table 37 Site-wide > Configure > Cloud Authentication > DPPSK (continued)

LABEL	DESCRIPTION
Created at	This shows the date and time that the account was created.
	Click this icon to display a greater or lesser number of configuration fields.

Table 37 Site-wide > Configure > Cloud Authentication > DPPSK (continued)

4.9.5 Collaborative Detection & Response

Collaborative Detection & Response (CDR) allows you to detect wired and WiFi clients that are sending malicious traffic in your network and then block or quarantine traffic coming from them. In this way, malicious traffic is not spread throughout the network. Secure policies can block malicious traffic for specific traffic flows, but CDR can block malicious traffic from the sender. Malicious traffic is identified using a combination of Web Filtering, Anti-Malware and IPS (IDP) signatures.

Note: To use the CDR feature, a Gold/UTM Security Pack license and a Nebula Pro Pack license is required.

The following table shows the CDR feature with/without a Gold/UTM Security Pack license.

CDR	WITHOUT GOLD/UTM SECURITY PACK	WITH GOLD/UTM SECURITY PACK	AFTER GOLD/UTM SECURITY PACK EXPIRES			
With Nebula Pro Pack	CDR will not function. CDR settings will be grayed-out.	CDR full functionality.	 CDR will disable its full functionality. CDR fields in an "Enabled/ Disabled" state will show "Enabled/ Disabled" but grayed-out. The Policy rule settings, Quarantine VLAN, and Exempt list will be kept in Site-wide > Configure > Collaborative detection & response. Previously quarantined clients will be released. 			
With Nebula Base/Plus Pack	CDR will not function. CDR settings will be grayed-out.	 User is notified that CDR is with partial functionality only. CDR event detection is available CDR triggered events are logged in the Site-wide > Monitor > Site features logs Containment actions (Alert/Block/Quarantine) is not available Previously blocked/ quarantined clients will be released in Site-wide > Monitor > Containment list. 	 CDR will disable its full functionality. CDR fields in an "Enabled/ Disabled" state will show "Enabled/ Disabled" but grayed-out. The Policy rule settings, Quarantine VLAN, and Exempt list will be kept in Site-wide > Configure > Collaborative delection & response. Previously quarantined clients will be released. 			

Table 38 CDR Feature With/Without a Gold/UTM Security Pack License

-wide = Configure = <u>ColloborativeDetectionAndlee</u> laborativeDetectionAndReponse	ponse							
Collaborative detection & response								
Enable								
Policy	Category	Event type	Occurrence		Duration (Minutes)		Containment	
	Malware	Malware detected	5	×	60	×	Alert	
	IPS	Vulnerability exploit detected	2	× *	10	× *	Alert	
	Web Threats	Connections to malicious web sites detected	3	× *	30	×	Alert	
ontainment								
General								
Theme		Ø						
		0						
		BUTTON						
	O Contestin	Modern			Sav	or Cance	d.	
				(Plec	ise allow 1-2 minute	es for chai	nges to take eff	fe
Logo		Up	load a logo					
		No logo						
Notification message	There are m	alicious network activities found an your d	wice. Please contact netw	ork administra	itor			
Redirect external URL		×						
	To use custo	m captive portal page, please download the	zip file and edit them.					
	Download ti	ne customized captive partal page example						
Containment period	60							
Block								
Block wireless client ()								
Quarantina								
Quarantine VLAN	Edit Man	d- 44, 10 254 252 1/255 255 254 0						
	cont want							
xempt list								
xempt list								

Figure 53	Site-wide >	Configure >	Collaborative	Detection &	& Respons
-----------	-------------	-------------	---------------	-------------	-----------

Table 39 Site-wide > Configure > Collaborative Detection & Response	se
---	----

LABEL	DESCRIPTION
Collaborative detection	on & response
Enable	Select this check box to activate Collaborative Detection & Response. Make sure you have active Web Filtering, Anti-Malware, IPS (Intrusion Prevention System), and CDR (Collaborative Detection & Response) licenses.
Policy	

LABEL	DESCRIPTION
Category	Category refers to the signature type that identified the malicious traffic: Malware (Anti-Malware, Anti-Virus), IDP (IPS), and Web Threat (Content Filtering and URL Threat Filtering).
Event Type	This displays some details on the category of malicious traffic detected.
Occurrence (1– 100)	Enter the number of security events that need to occur within the defined Duration to trigger a CDR Containment action.
Duration (1–1440)	Enter the length of time in minutes the event should occur from a client the Occurrence number of times to trigger a CDR Containment action.
	For example, Occurrence is set to 10, and Duration is set to 100. If the NCC detects 10 or more occurrences of malicious traffic in less than 100 minutes, then CDR Containment is triggered.
Containment	Select the action to be taken when the number of security events exceed the threshold within the defined duration.
	Alert: Select this if you just want to issue a notification in NCC.
	Block : Select this if you want to block traffic from a suspect client at the NCC, or from a suspect WiFi client at the AP connected to the NCC. Traffic is still broadcast to other clients in the same subnet. A 'notification' web page is displayed when this action is triggered.
	Quarantine: Select this if you want to isolate traffic from a suspect client at the NCC in a quarantine VLAN. Traffic is not broadcast to other clients in the same subnet. A 'notification' web page is displayed to the client when this action is triggered.
Containment	Use this section to configure the selection containment action.
General	
Theme	Configure the CDR block page.
	 Click the Preview icon at the upper right corner of a theme image to display the block page in a new frame. Click the Copy icon to create a new custom theme (block page).
Logo	This shows the logo image that you uploaded for the customized block page.
	Click Choose File and specify the location and file name of the logo graphic or click Browse to locate it. You can use the following image file formats: GIF, PNG, or JPG. File size must be less than 200 KB, and images larger than 244 x 190 will be resized.
Notification message	Enter the message that is displayed on the CDR block page. The client is redirected here when a Block or Quarantine action is triggered. For example, "Malicious traffic is coming from your device so traffic is temporarily stopped. Please contact the network administrator."
	Redirect external URL : Enter a URL in "http://domain" or "https://domain" format to an external notification page. The client is redirected here when a Block or Quarantine action is triggered. Make sure the external notification page is accessible from the NCC.
Redirect external URL	Enable this setting, and then enter a URL in "http://domain" or "https://domain" format to an external notification page. The client is redirected to this page when a Block or Quarantine action is triggered. You can download a sample block page by clicking Download .
	Note: The external notification page must be accessible from NCC.
Containment Period	Enter how long the client should be blocked or quarantined. This should be at least twice the DHCP server lease time in order to prevent false positives.
Block	Enter how long a suspect client should be blocked or quarantined. You can enter from 1 minute to 1 day (1,440 minutes). 0 means the suspect is blocked forever until released in Site-wide > Monitor > Containment list.
Block wireless client	Select this to have traffic from the suspect client blocked at the AP. Clear this to have traffic from the suspect client blocked at the NCC.

Table 39	Site-wide > Configure > Collaborative Detection & Response (continued)
	one mae eeningere eenaberante Bereenen a kespense (001111100001

LABEL	DESCRIPTION
Quarantine	
Quarantine VLAN	Click Set to configure a VLAN in order to isolate traffic from suspect clients. Traffic from a suspect client is broadcast to all members in the VLAN.
Exempt list	Enter IPv4 and /or MAC addresses of client devices that are exempt from CDR checking.

Table 39 Site-wide > Configure > Collaborative Detection & Response (continued)

4.9.6 Quarantine Interface Configuration

Click Set at Site-wide > Configure > Collaborative detection & response > Containment > Quarantine to configure the VLAN and interface used to isolate a client when a quarantine action is triggered. The following screen appears.

Note: Only IPv4 addresses can be used in quarantine VLANs.

Figure 54 Site-wide > Configure > Collaborative detection & response > Containment > Quarantine

Quarantine interface configuration				
Interface Properties				
Interface name	Quarantine			
Port group	LAN Group 1	•		
VLAN ID	44	× (1 - 4094)		
IP address assignment				
IP address	10.254.252.1	×		
Subnet mask	255.255.254.0	×		
DHCP server				
IP pool start address	10.254.252.2	× Pool size 510 ×		
		Cancel	Ok	

Each field is explained in the following table.

Table 40 Site-wide > Configure > Collaborative detection & response > Containment > Quarantine

LABEL	DESCRIPTION
Interface Properties	
Interface Name	This field is read-only. The default name is "Quarantine".
Port group	Select the name of the port group to which you want the interface to belong.
Base Port	Select the Ethernet interface on which the VLAN interface runs.
VLAN ID	Enter the VLAN ID. This 12-bit number uniquely identifies each VLAN. Allowed values are 1 – 4094. (0 and 4095 are reserved)

LABEL	DESCRIPTION
IP address assignment	This is a 3-bit field within a 802.1 Q VLAN tag that is used to prioritize associated outgoing VLAN traffic. "0" is the lowest priority level and "7" is the highest.
IP address	Enter the IP address for this interface.
Subnet mask	Enter the subnet mask of this interface in dot decimal notation. The subnet mask indicates what part of the IP address is the same for all computers in the network.
DHCP Server	
Get Automatically	Enter the IP address from which the Security Appliance begins allocating IP addresses. If you want to assign a static IP address to a specific computer, click Add new under Static DHCP Table.
IP pool start address	Enter the IP address from which the Security Appliance begins allocating IP addresses for this VLAN.
Pool size	Enter the total number of IP addresses the DHCP server will hand out.
OK	Click OK to save your changes back to the NCC.
Cancel	Click Cancel to exit this screen without saving.

Table 40 Site-wide > Configure > Collaborative detection & response > Containment > Quarantine

4.9.7 Site Settings

Use this screen to change the general settings for the site, such as the site name, Nebula Device login password, captive portal reauthentication, SNMP, AP traffic logs to a Syslog server, traffic logs to SecuReporter, and API access for DPPSK third-party integration. Click **Site-wide > Configure > Site settings** to access this screen.

Figure 55	Site-wide >	Configure	> Site	settinas
riguie JJ		Connigore	- 3110	Johnga

Site-wide > Configure > <u>Site settings</u>	
Site information	
Site name	ZyNet TW ×
Local time zone	Toiwan
Site location	What is this?
	This site location will apply to your new added device(s) as address on map automatically.
Device configuration	
Local credentials	Username: admin (Firewall username is "support") Password:
	Password must be at least 8 characters in length and consists of letters and numerals. The valid characters are letters, numerals and symbols as follow : ~ ! @ # \$ % ^ & * ()_+ * -= {};; <> .
Smart guest/VLAN network Beta	What is this?
Captive portal reauthentication	(Please allow 1-2 minutes for changes to take efference)
For my AD server users	Every day 👻
For my RADIUS server users	Every day 👻
For click-to-continue users	Every day 👻
For cloud authentication users 👔	Every day 👻
SNMP	
SNMP access:	V1/V2c +
SNMP community string:	×
Reporting	
Syslog server	Server IP Types Action
	× AP traffic log ⊗
Use timezone for syslog server logs 🚯	Model list
AP traffic log 1	
SecuReporter 1	Sending Security Appliance traffic logs to SecuReporter.
API access 1	
API token:	DKLEIdONDPU7/gw/NRY

LABEL	DESCRIPTION		
Site Information			
Site name	Enter a descriptive name for the site.		
Local time zone	Choose the time zone of the site's location.		
Site location	Enter the complete address or coordinates (physical location) of the Nebula Devices in the site. All newly added Nebula Devices will automatically use this as the default address and location on the Google map.		
	Note: You can edit each Nebula Device's location on the Google map.		
Device configuration			
Local credentials	The default password is generated automatically by the NCC when the site is created. You can specify a new password to access the status page of the Nebula Device's built-in web- based configurator. The settings here apply to all Nebula Devices in this site.		
Smart guest/ VLAN network	Click On to enable this feature. This allows the NCC to check if the VLAN ID and guest network settings are consistent on the APs and Security Appliance in the same site to ensure guest network connectivity.		
	The guest settings you configure for a gateway interface (in Site-wide > Configure > Security gateway > Interface addressing) will also apply to the WiFi networks (SSIDs) associated with the same VLAN ID (in Site-wide > Configure > Access points > SSID settings). For example, if you set a gateway interface in VLAN 100 as a guest interface, the SSID that belongs to VLAN 100 will also act as a guest network.		
Captive portal reauthe	entication		
For my AD server users	Select how often the user (authenticated by an AD server) has to log in again.		
For my RADIUS server users	Select how often the user (authenticated by a RADIUS server) has to log in again.		
For click-to- continue users	Select how often the user (authenticated through the captive portal) has to log in again.		
For cloud authentication users	Select how often the user (authenticated using the NCC user database) has to log in again.		
SNMP			
SNMP access	Select V1/V2c to allow SNMP managers using SNMP to access the Nebula Devices in this site. Otherwise, select Disable.		
SNMP community string	This field is available when you select V1/V2c . Enter the password for the incoming SNMP requests from the management station.		
Reporting			
Syslog server	Click Add to create a new entry.		
Server IP	Enter the IP address of the server.		
Types	Select the type of logs the server is for.		
	Note: Besides sending Gateway traffic log to a Syslog server, you can also set the Security Appliance (through its Web Configurator) to save a copy of the logs to a connected USB storage device. Gateway traffic log includes the traffic information (such as its source, destination or usage) of the Security Appliance clients.		
Action	Click the Delete icon to remove the entry.		

 Table 41
 Site-wide > Configure > Site settings

LABEL	DESCRIPTION				
Use timezone for	Click On to enable this feature. This allows the Syslog server logs to use the site's timezone.				
syslog server logs	If disabled, the Syslog server logs will show GMT 0 time. GMT does not adjust automatically for Daylight Savings Time (DST). You must adjust for Daylight Savings directly in the Syslog server.				
AP traffic log	Log traffic for access points in the site that have NAT mode enabled. You can also send the logs to a Syslog server, by selecting AP traffic log under Syslog server > Types .				
	For details on configuring NAT mode , see Section 5.3.1 on page 247.				
SecuReporter	Click On to enable this feature. This allows the NCC to send traffic logs to SecuReporter.				
	Note: Disable this option if you have configured sending of traffic logs to an external syslog server.				
API access	API access allows third-party software to integrate with the DPPSK feature in NCC. For more information, please contact Zyxel.				
API token	Generate an API token for DPPSK third-party integration.				
Сору	Click this button to copy the API key to the system's clipboard.				
Delete	Click this button to delete the API key.				

 Table 41
 Site-wide > Configure > Site settings (continued)

CHAPTER 5 Access Point

5.1 Overview

This chapter discusses the menus that you can use to monitor the Nebula-managed APs (Access Points) in your network and configure settings even before an AP is deployed and added to the site.

Nebula Device refers to Zyxel Hybrid APs (NAP / NWA / WAC / WAX Series) in this chapter. To view the list of Nebula Devices that can be managed through NCC, go to **Help** > **Support tools** > **Device function table**.

The following features in the Access Point menus apply to specific models only.

FEATURES/FIELDS	INCLUDED NEBULA DEVICES	LOCATION
Ethernet Secure Tunnel Setting in Remote AP Setting	WAC500H	Click a Nebula Device entry in the Site-wide > Devices > Access points screen to display
Wired stations		4.3.1 on page 156 for more information.
WPA3 in Security options	NWA110AX, WAX510D, WAX650S	Click Site-wide > Configure > Access points > SSID advanced settings. See Section 5.3.1 on page 247 for more information.
Ethernet Traffic options Forwarding Mode	WAC500H	Click an entry in the Port setting table of the Site-wide > Configure > Access points AP & port settings screen to access the Edit – AP & port settings screen. See Section 5.3.7.1 on page 274 for more information.

Table 42 Features/Fields Supported on Specific Nebula Devices Only

5.1.1 Nebula Smart Mesh

Nebula Smart Mesh, also called Smart Mesh or AP Smart Mesh, is a WiFi mesh solution for Nebula Devices. With Smart Mesh, you can have two or more Nebula Devices automatically create a mesh network within your home or office, ensuring there are no areas with a weak WiFi signal.



Smart Mesh assigns a role to each Nebula Device depending on its connection method.

- Root AP: A Nebula Device (mesh controller) that is connected to the network by Ethernet and can reach the gateway device.
- **Repeater AP:** A Nebula Device (mesh extender) that is connected to the network wirelessly, or that is connected to the network by Ethernet but cannot reach the gateway device.

The mesh extender rebroadcast the mesh controller's SSID, and then relay WiFi traffic back to the gateway.

To create a Smart Mesh network, add two or more Nebula Devices to the same Nebula-managed site and ensure that each Nebula Device has Smart Mesh enabled. Then connect one or more Nebula Devices to your network's gateway using an Ethernet cable, so that you have at least one mesh controller. Finally, place one or more non-wired Nebula Devices in areas where you want to extend WiFi coverage.

5.1.2 Smart Mesh Network Topology

After you add a Nebula Device to an NCC site and then turn it on, the new Nebula Device automatically connects to a mesh network called the **default mesh**. The Nebula Device then tries to connect to a mesh controller and contact NCC. After the Nebula Device successfully contacts NCC and joins the site, the Nebula Device stops using the default mesh and instead connects to other Nebula Devices in the site using a dedicated network called the **site mesh**.

5.1.2.1 Smart Mesh Wireless Hops

Each mesh extender tries to connect to the site gateway through a mesh controller. If a mesh extender cannot connect directly to a mesh controller, then the mesh extender relays its WiFi traffic through another mesh extender. Each time traffic passes through a WiFi connection in the mesh network, it counts as one **hop**.

Nebula Smart Mesh supports an unlimited number of hops. However, each hop in a mesh network reduces network throughput by up to half. Therefore, we recommend only allowing a maximum of two hops within your Smart Mesh network.





5.1.2.2 Wireless Bridge

Wireless bridge is a Smart Mesh feature that allows two Nebula Devices to automatically connect two network segments together over a WiFi connection. This is useful when you want to extend your wired network to a new area, but it is difficult to run cables to that area.

To use wireless bridge, enable **Wireless Bridge** on two Nebula Devices in NCC. Then connect wired clients to one of the Nebula Device's LAN port. These wired clients form a new network segment and are able to reach the site gateway through the Nebula Device's WiFi connection.



Figure 58 Nebula Smart Mesh Wireless Bridge

5.2 Monitor

Use the **Monitor** menus to check Nebula Device event log messages and summary report for Nebula Devices in the selected site.

5.2.1 Event Log

Use this screen to view WiFi Nebula Device log messages. You can enter the Nebula Device name or a key word, select one or multiple event types, or specify a date/time or even a time range to display only the log messages related to it.

Click Site-wide > Monitor > Access points > Event log to access this screen.

Access Point:		Keyword:		С	Category:
Any	×	Any		×	Any
Bef	ore 🔻	2019-10-30		17	7.12 • 1h • UTC+8 🗵 🔾 Searc
< Newer 🛛 Older >	135 Ev	ent log			💎 🕒 Export
Time	Access p	point	Category		Detail
2019-10-30 16:14:23	61,21,27	14.17.19	Wireless LAN		Station: 9c:5c:f9:61:f6:c1 has associated on Channel: 6, SS
2019-10-30 16:14:27	62,21,27	84.2719	Wireless LAN		Station: 9c:5c:f9:61:f6:c1 has blocked by Hostapd3 on Ch
2019-10-30 16:14:27	62,3137	84.2719	Wireless LAN		Station: 9c:5c:f9:61:f6:c1 has blocked by prev-Auth Failed
2019-10-30 16:14:27	62,31,37	84.2719	Wireless LAN		WPA authenticator requests disconnect: reason 1. Interf
2019-10-30 16:14:27	62,21,37	84.2719	Wireless LAN		WPA authenticator requests disconnect: reason 2. Interf.
2019-10-30 16:19:26	61,31,37	84.2719	Wireless LAN		Station: 9c:5c:f9:61:f6:c1 has associated on Channel: 6, SS
2019-10-30 16:19:30	61,31,37	84.2719	Wireless LAN		Station: 9c:5c:f9:61:f6:c1 has blocked by Hostapd3 on Ch
2019-10-30 16:19:30	61,31,37	84.2719	Wireless LAN		Station: 9c:5c:f9:61:f6:c1 has blocked by prev-Auth Failed
2019-10-30 16:19:30	00.3137	DE.1733	Wireless LAN		WPA authenticator requests disconnect: reason 1. Interf
2019-10-30 16:19:30	60.2130	84.2738	Wireless LAN		WPA authenticator requests disconnect: reason 2. Interf.

Figure 59 Site	-wide > Monitor >	Access points >	Event log
----------------	-------------------	-----------------	-----------

5.2.2 Vouchers

A voucher is a unique printable code that allows a user to authenticate with a WiFi network for a limited period of time. A user connects to the WiFi network's SSID and then enters the code in a captive portal. After a successful login, the expiry time of the voucher starts counting down.

Vouchers are useful in situations where you want to give individual users time-limited WiFi access. For example: A customer can purchase a voucher for 2 hours of Internet access in a hotel or coffee shop.

Note: You can only enable voucher authentication for one SSID per site.

5.2.2.1 Using Vouchers

- Go to Site-wide > Configure > WiFi SSID, and create a dedicated SSID for voucher-based WiFi access. For example, "Hotel_Guest_Network".
 For details on configuring SSIDs, see Section 12.3.1 on page 538.
- 2 Go to Site-wide > Configure > WiFi SSID, select the SSID, and then under Sign-in method select Voucher. For details, see Section 12.3.2 on page 540.
- 3 Go to Site-wide > Monitor > Access points > Vouchers > Settings to configure how the vouchers will look when printed. For details, see Section 5.2.2.4 on page 239.
- 4 Go to Site-wide > Monitor > Access points > Vouchers, and then click Create to create one or more vouchers.

5.2.2.2 Vouchers Screen

This screen allows you to create and manage vouchers for WiFi network authentication.

Click **Site-wide > Monitor > Access points > Vouchers** to access this screen.

Figure 60 Site-wide > Monitor > Access points > Vouchers

Site-wide > Monitor > Access po	ints > <u>Vouchers</u>						
Vouchers							
Vouchers can be set to speci	fic time increments to limit acc	cess to users who you autho	rize to use the network for a spe	cific period of time.			
Vouchers	Settings						
Reset Delete Print	Q Search	• 1) selected in (1) voucher			+ Create	Export -
Voucher	Comments	Duration	Remaining	Expires	Status	Created	R
33288189	For hotel guests use	only.	10 hours 37 mins	2023-04-01 00:00:00	Unused	2023-03-31 13:22:27	
	Deleted and exp	ired vouchers will be auto-p	urged daily. Unused vouchers w	vill be auto-purged after the sp	ecified number of day	'S.	

The following table describes the labels in this screen.

LABEL	DESCRIPTION
Reset	Select one or more vouchers and then click this button to reset the vouchers back to their original states. Each voucher's status is set to Unused and time remaining is reset to the time configured in Duration .
Delete	Select one or more vouchers and then click this button to delete the vouchers.
Print	Select one or more vouchers and then click this button to print the vouchers.
	You can modify how vouchers look when printed at Site-wide > Configure > General settings .
Search	Use this field to search for vouchers, by voucher code, duration, and/or status.
Create	Click this button to create one or more vouchers.
	For details, see Section 5.2.2.3 on page 238.
Export	Click this button to export the voucher table and all information in it to a CSV or XML file.
Voucher	This displays the voucher's unique authentication code.
Comments	This displays information about the voucher.
Duration	This displays how long the voucher is valid from when it is activated, in hours.
Remaining	This displays how much time is left before the voucher expires.
	NCC only starts counting this time after the voucher has been activated.
Expire in	This displays the date and time that the voucher will expire.

Table 43 Site-wide > Monitor > Vouchers

Table 43	Site-wide > Monitor >	Vouchers	(continued)

LABEL	DESCRIPTION
Status	This displays the current status of the voucher:
	Unused: The voucher has not yet been used for authentication.
	Active: A user has used the voucher for authentication. NCC has started counting down the duration.
	Expire: The voucher has reached the end of its duration period and can no longer be used.
	Delete: The voucher is unused and has reached the time set under Purge after (days).
	Note: NCC automatically deletes vouchers with the status Expire or Delete after 24 hours. You can see a history of these automatic deletions in the NCC event log.
Created	This displays the date and time that the voucher was created.

5.2.2.3 Create Vouchers Screen

Use this screen to create one or more new vouchers.

	Figure 61	Site-wide > Monitor > Access	points > V	ouchers >	Create
--	-----------	------------------------------	------------	-----------	--------

Create Vouchers		×
Quantity	1	
Code length	8	
Comment	×	
Valid period 👔		
O Duration (hours)	12	
Purge after (days)	30	
O Expires on		
Print after created		
Save as default		
	Cancel	reate

LABEL	DESCRIPTION
Quantity	Sets the number of vouchers you want to create.
	The valid range for this setting is $1 - 999$.
Code length	Sets the length of the unique code on each voucher.
	The valid range for this setting is $6 - 10$.
Comment	Enter information about the voucher that might be useful for other administrators.
Valid period	There are two ways to set your voucher's validity.
Duration (hours)	Sets how long the voucher is valid after it has been activated, in hours.
	The valid range for this setting is $1 - 72$.
Purge after	Sets how long a non-activated voucher is valid for, in days.
(ddys)	The valid range for this setting is $1 - 180$.
Expires on	Sets the date and time for the expiration of this voucher.
Print after created	Select this to print the vouchers immediately after clicking Create.
Save as default	Click this to make the settings on this page the default settings for new vouchers.

Table 44 Site-wide > Monitor > Vouchers > Create

Note: Dynamic Personal Pre-Shared Keys (DPPSKs) also allow you to give individual users a printable password and time-limited WiFi access. For details, see Section 12.3.2 on page 540.

5.2.2.4 Voucher Settings Screen

Use this screen to change the voucher settings for the Nebula Device. Click **Site-wide** > **Monitor** > **Access points** > **Voucher** > **Settings** to access this screen.

Figure 62	Site-wide >	Monitor >	Access	points >	Voucher >	Settings
-----------	-------------	-----------	--------	----------	-----------	----------

Site-wide > Monitor > Access po	oints > <u>Vouchers</u>			
Vouchers				
Vouchers can be set to spec	ific time increments to lim	it access to users who you authorize to us	e the net	work for a specific period of time.
Vouchers	Settings			
Voucher settings				
Duration text:		Hours Valid: X	This te	xt will precede the duration on the printed voucher
Date Text:		Expire on: X	This te	xt will precede the voucher expiration date on the printed voucher
Access text:		Login Code: X	This te	xt will precede the voucher code on the printed voucher
Show image:				
Promotion text:		×	Option	al (Maximum Is 64 character)
Promotion URL:		×	Option	al (Maximum is 64 character)
Voucher image:				Upload an Image
		No image		

|--|

LABEL	DESCRIPTION			-	
Voucher settings	Use these settings to configure how WiFi network authentication vouchers for this site look when printed.				
		V	oucher		
		SSID:			
		Hours Valid:	Login Code:		
		12	42786823		
	For more informat	ion on vouche	rs, see Section 5	.2.2 on page 236.	
Duration text	Sets the text that	orecedes the c	luration on the	voucher.	
	The text must consist of 1 – 16 characters.				
Date text	Sets the text that precedes the expiration date on the voucher.				
	The text must con	sist of 1 – 16 ch	aracters.		
Access text	Sets the text that	orecedes the v	oucher code o	n the voucher.	
	The text must con	sist of 1 – 16 ch	aracters.		
Show image	Sets whether to d	splay an image	e at the top-left	of the voucher. This image is optional.	
Promotion text	Sets the promotional text on the voucher. This text is optional.				
	The text must consist of 1 – 64 characters.				
Promotion URL	Sets the promotional URL on the voucher. This URL is optional.				
	The URL is displayed as a QR code on the voucher.				
Voucher image	This shows the uploaded image that will be displayed at the top-left of the voucher.				
Upload an image	Click this button to appears. Click thi image file. The mo	o upload an im s button to loco aximum image	age from your I ate the PNG (pr file size is 200 KE	ocal computer. The Choose File button eferred for its transparency) / JPEG/GIF 3.	
Replace this image	Click this button to	o change the u	ploaded image	Ð.	
Remove this image	Click this button to	o delete the up	loaded image.		

5.2.3 Wireless Health

This screen lets you monitor the health of WiFi networks for your Nebula Devices and connected WiFi clients.

You can improve WiFi network performance by doing the following:

- Enable DCS (Dynamic Channel Selection) to select a radio channel with least interference
- Enable client steering to use a stronger WiFi signal
- Change channel bandwidth to reduce radio interference from other WiFi devices

Click Site-wide > Monitor > Access points > Wireless health to access this screen.



Figure 63 Site-wide > Monitor > Access points > Wireless health



LABEL	DESCRIPTION
Auto optimization acti	on
6G radio	 Select ON to enable and specify how the Nebula Device improves the WiFi network performance. Otherwise, select OFF to disable it. Adaptive channel width – select this option to have the Nebula Device change the channel bandwidth from 160 MHz to 80 MHz to reduce the radio interference with other WiFi devices. If adaptive channel width does not improve WiFi performance then the Nebula Device also performs Dynamic Channel Selection (DCS). DCS (Dynamic Channel Selection) – select this option to have the Nebula Device scan and choose a radio channel that has least interference.

Table 46 Site-wide > Monitor > Access points > Wireless health

LABEL	DESCRIPTION
5G radio	Select ON to enable and specify how the Nebula Device improves the WiFi network performance. Otherwise, select OFF to disable it.
	 Adaptive channel width – select this option to have the Nebula Device change the channel bandwidth from 80 MHz to 20 MHz to reduce the radio interference with other WiFi devices. If adaptive channel width does not improve WiFi performance then the Nebula Device also performs Dynamic Channel Selection (DCS).
	 DCS (Dynamic Channel Selection) – select this option to have the Nebula Device scan and choose a radio channel that has least interference.
2.4G radio	Select ON to enable the Nebula Device to improve the WiFi network performance. Otherwise, select OFF to disable it.
	 DCS (Dynamic Channel Selection) – select this option to have the Nebula Device scan and choose a radio channel that has least interference.
Client	Select ON to have the Nebula Device try to steer the WiFi clients in poor health to a Nebula Device or SSID with a strong signal. Client steering to improve the signal strength is done every 30 minutes. Otherwise, select OFF to disable steering.
Optimization	The Nebula Device optimizes the WiFi network performance by doing the following:.
aggressiveness	 Change the channel bandwidth from 160 MHz to 80 MHz, or 80 MHz to 20 MHz to reduce radio interference from other wireless devices (Adaptive Channel Width). Select a radio channel with least interference (DCS, Dynamic Channel Selection). Direct clients to an AP with a stronger WiFi signal.
	There might be some disruption to the client's WiFi connections while the Nebula Device is optimizing the WiFi network. To minimize disruption, you can decide to optimize the WiFi network only when the WiFi network is below a certain level of busyness. Low, Standard, and High stand for different levels of busyness. The busyness level you select decides when the Nebula Device takes action to optimize the WiFi network.
	Low: Only perform WiFi network optimization action when the WiFi network traffic is below Low.
	Standard: Only perform WiFi network optimization action when the WiFi network traffic is Low.
	High: Only perform WiFi network optimization action when the WiFi network traffic is Standard, or Low.
AP wireless health ove	rview
Move the cursor over t	the information icon to view the supported Nebula Device model list.
Current status	This shows the number of supported Nebula Devices that are currently online, using the specified frequency band that are in Good , Fair or Poor wireless health threshold as detected by Nebula.
y-axis	The y-axis represents the state of wireless health.
x-axis	The x-axis shows the time period over which the Nebula Device health state is recorded.
Top APs by health aler	t
Name	This shows the descriptive name of the Nebula Device.
Model	This shows the model number of the Nebula Device.
Alert	This shows how many times the Nebula Device is in a poor state of wireless health.
	The NCC generates a log when the Nebula Device is in poor wireless health. You can view the log messages in the Site-wide > Monitor > Access points > Event log screen.
Clients wireless health	overview
Current status	This shows the number of connected WiFi clients that are currently online, using the specified frequency band and in Good , Fair or Poor wireless health threshold as detected by Nebula.

Table 11	Site wide > Monitor >	Accors points >	Wirdloss boalth	(continued)
	3116-MIGE - MOLIIIOI -	ACCESS DOILIIS /		ICOMMUEU

LABEL	DESCRIPTION	
Client health	Select to view the health of all WiFi clients which are connected to the supported Nebu Devices using the 6 GHz, 5 GHz or 2.4 GHz band.	
	You can select to view the health report for the past day, week or month, as well as filter the WiFi station to view.	
y-axis	The y-axis represents the state of wireless health.	
x-axis	The x-axis shows the time period over which the client health state is recorded.	
Top clients by health a	lert	
Description	This shows the descriptive name of the client.	
Alert	This shows how many times the client is in a poor state of wireless health.	
	The NCC generates a log when the client is in poor wireless health. You can view the log messages in the Site-wide > Monitor > Access points > Event log screen.	

Table 46 Site-wide > Monitor > Access points > Wireless health (continued)

5.2.4 Summary Report

This screen displays network statistics for Nebula Devices of the selected site, such as bandwidth usage, top clients and/or top SSIDs.

Click Site-wide > Monitor > Access points > Summary report to access this screen.



Figure 64 Site-wide > Monitor > Access points > Summary report

Table 47 Site-wide > Monitor > Access points > Summary report

LABEL	DESCRIPTION			
Summary report				
Usage				
y-axis	The y-axis shows the transmission speed of data sent on this port in megabits per second (Mbps).			
x-axis	The x-axis shows the time period over which the traffic flow occurred.			
Top APs by usage				
#	This shows the ranking of the Nebula Device.			
Name	This shows the descriptive name of the Nebula Device.			
Model	This shows the model number of the Nebula Device.			
Usage	This shows the amount of data transmitted or received by the Nebula Device.			
Client	This shows how many clients are currently connecting to the Nebula Device.			
Location				
This shows the location	of the Nebula access points on the map.			
Top applications by us	age			
#	This shows the ranking of the application.			
Application	This shows the application name.			
Category	This shows the category of the application, for example email, file sharing.			
Usage	This shows the amount of data consumed by the application.			
% Usage	This shows the percentage of usage for the application.			
Top SSIDs by usage				
#	This shows the ranking of the SSID.			
SSID	This shows the SSID network name.			
Encryption	This shows the encryption method used by the SSID network.			
# Client	This shows how many WiFi clients are connecting to this SSID.			
% Client	This shows what percentage of associated WiFi clients are connecting to this SSID.			
Usage	This shows the total amount of data transmitted or received by clients connecting to this SSID.			
% Usage	This shows the percentage of usage for the clients connecting to this SSID.			
Clients per day				
y-axis	The y-axis represents the number of clients.			
x-axis	The x-axis represents the date.			
Top clients by usage				
#	This shows the ranking of the client.			
Description	This shows the descriptive name or MAC address of the client.			
Usage	This shows the total amount of data transmitted and received by the client.			
% Usage	This shows the percentage of usage for the client.			
Top operating systems	by usage			
#	This shows the ranking of the operating system.			
OS	This shows the operating system of the client device.			
# Client	This shows how many client devices use this operating system.			

LABEL	DESCRIPTION
% Client	This shows the percentage of top client devices which use this operating system.
# Usage	This shows the amount of data consumed by the client device on which this operating system is running.
% Usage	This shows the percentage of usage for top client devices which use this operating system.
Top client device man	ufacturers by usage
#	This shows the ranking of the manufacturer.
Manufacturer	This shows the manufacturer name of the client device.
# Client	This shows how many client devices are made by the manufacturer.
% Client	This shows the percentage of top client devices which are made by the manufacturer.
# Usage	This shows the amount of data consumed by the client device.
% Usage	This shows the percentage of usage for the client device.

Table 47 Site-wide > Monitor > Access points > Summary report (continued)

5.3 Configure

Use the **Configure** menus to set the WiFi security settings for Nebula Devices of the selected site.

5.3.1 SSID Advanced Settings

Use this screen to configure the WiFi security, L2 isolation, intra-BSS traffic blocking and walled garden settings for the SSID profiles.

Click Site-wide > Configure > Access points > SSID advanced settings to access this screen.

e-Nebula-FT	·				
etwork access					
Security options ()	Open				
	Users can connect without entering a password				
	C Enhanced-open				
	User can connect without password. Enhanced open provides improved data encryption in open Wi-Fi networks.				
	WPA Personal With WPA2				
	Users must enter this key to associate:				
	WI-FI Access QR Code Print				
	V Dynamic personal psk 1 Model list				
	MAC-based Authentication with Nebula cloud authentication - Model list				
	Use MAC address as a username and password				
	WPA Enterprise with WPA2 🔹				
	Use 802.1X authentication that requires a unique username and password				
	WPA Enterprise with Nebula cloud authentication 👻				
Sign-in method					
	Users can access the network without any web authentication				
	Click-to-continue				
	Users must view and agree the captive portal page in order to access the network				
	Voucher				
	Users must enter a voucher code in order to access the network				
	Create and manage voucher prescode on the Vouchers page				
	offente and manage obtainer passible of the <u>rotations</u> page.				
	Sign-on with Nebula cloud authentication				
	Users must enter a username and password in order to access the network				
aptive portal advance setting					
Walled garden					
and the design of the second					
walled garden ranges:					
	x				
	What do I enter here?				
Self-registration:	Allow users to create accounts with manual authorized				
Simultaneous login limit	Unlimited				
Strict Policy	Block all access until sign-on				
Reauth time	Follow site-wide setting				
NCAS disconnect behavior 🜖	Allowed Client devices can access the network without signing in, except they are explicitly blocked. Limited				

Figure 65 S	Site-wide >	Configure >	Access I	points >	SSID	advanced	settinas P	art 1
-------------	-------------	-------------	----------	----------	------	----------	------------	-------

anne epanene	
Forwarding mode	O Local bridge
	NAT mode ① Modellist Use Zyxel DHCP & NAT Clients receive IP addresses in an isolated network. Client cannot communicate with other clients associated with different AP. Turnel mode: ① Modellist APs send traffic over a tunnel to Zyxel Security gateway Turneled to a specified VLAN at the Zyxel Security gateway.
Rate-limit	Download One Unlimited (Mb/s) (1 - 160) 1M 2M 3M 4M 5M 6M 7M 8M 9M 10M Image: Second
dvanced settings	
VLAN ID	1 × (1-4094)
Band mode	 24GHz band 5GHz band 6GHz band <u>Why cont I see WIFI In 6GHz?</u>
	MAC Description 1 202107030123 × + Add
Intra-BSS traffic blocking	Enable Intra-BSS traffic blocking ()
Intra-BSS traffic blocking Band select	Enable Intra-BSS traffic blocking 🕕
Intra-BSS traffic blocking Band select Assisted roaming	Enable Intra-BSS traffic blocking ()
Intra-BSS traffic blocking Band select Assisted roaming 80211r	 Enable Intra-BSS traffic blocking () Enable 80211K/v Enable 80211K/v

Figure 66 Site-wide > Configure > Access points > SSID advanced settings Part 2

Table 48	Site-wide >	Configure >	Access points >	SSID advanced settings
----------	-------------	-------------	-----------------	------------------------

LABEL	DESCRIPTION
SSID advanced settings	Select the SSID profile to which the settings you configure here is applied.
Network access	Note: You cannot enable MAC authentication, 802.1X authentication and web authentication at the same time.
	Note: User accounts can be created and authenticated using the NCC user database. See Section on page 555.

Security options	Select Open to allow any client to associate this network without any data encryption or authentication.
	Select Enhanced-open to allow any client to associate this network without any password but with improved data encryption.
	Upon selecting Enhanced-open or WPA Personal With WPA3 , transition mode generates two VAP so devices that do not support Enhanced-Open/WPA Personal With WPA3 can connect using Open/WPA Personal With WPA2 network. This is always on at the time of writing.
	Select WPA Personal With (WPA1/WPA2/WPA3) and enter a pre-shared key from 8 to 64 case-sensitive keyboard characters to enable WPA1/2/3-PSK data encryption. Upon selecting WPA Personal With WPA3 , Nebula Devices that do not support it will revert to WPA2.
	 Turn on 802.11r to enable IEEE 802.11r fast roaming on the access point. 802.11r fast roaming reduces the delay when the clients switch from one Nebula Device to another by allowing security keys to be stored on all Nebula Devices in a network. Information from the original association is passed to the new Nebula Device when the client roams. The client does not need to perform the whole 802.1x authentication process.
	Click Print to display the QR code that includes the password for quick access. You can save the QR code as PDF.
	Select Dynamic personal psk to have every user connect to the SSID using a unique pre- shared key (PSK) that is linked to their user account. This allows you to revoke a user's WiFi network access by disabling their account.
	After enabling this option, you must create one or more DPPSK users in the site or organization at Site-wide > Configure > Cloud authentication > Account Type > DPPSK .
	 For details on creating a site DPPSK user, see Section 11.4.7.3 on page 544. For details on creating organization DPPSK users, see Section 4.9 on page 207.
	Turn on MAC-based Authentication with to authenticate WiFi clients by their MAC addresses together with My RADIUS server to use an external RADIUS server. Or select Nebula cloud authentication to use the NCC for MAC authentication.
	Select WPA-Enterprise with to enable 802.1X secure authentication. You can select My RADIUS server to use an external RADIUS server or select Nebula cloud authentication to use the NCC for 802.1X authentication.
	 Turn on 802.11r to enable IEEE 802.11r fast roaming on the Nebula Device. 802.11r fast roaming reduces the delay when the clients switch from one Nebula Device to another by allowing security keys to be stored on all Nebula Devices in a network. Information from the original association is passed to the new Nebula Device when the client roams. The client does not need to perform the whole 802.1x authentication process.
	 Select Two-Factor Authentication to require that the user log in using both their password and a Google Authenticator code. To log in, users must have Two-Factor Authentication enabled on their account and have setup Google Authenticator on their mobile device. Select Enable on RAP only to only require Two-Factor Authentication when accessing the network through a remate accessing to (10.10).

 Table 48
 Site-wide > Configure > Access points > SSID advanced settings (continued)

LABEL	DESCRIPTION
Sign-in method	Select Disabled to turn off web authentication.
	Select Click-to-continue to block network traffic until a client agrees to the policy of user agreement.
	Note: After enabling Click-to-continue , the Nebula Device creates a user account with user name "clicktocontinue_X_Y", where X is the radio type (1 = 2.4 GHz, 2 = 5 GHz) and Y is the SSID number (1–8) of the SSID profile. The Nebula Device uses this account to authenticate clients who agree to the terms of the click-to-continue page.
	Select Voucher to require that a user logs in with a voucher code. For details on vouchers, see Section 5.2.2 on page 236.
	Note: Vouchers cannot be enabled if Dynamic Personal Pre-Shared Key (DPPSK) or WPA Enterprise are enabled. You can only enable voucher authentication for one SSID per site.
	Select Sign-on with and:
	 select Nebula cloud authentication to block network traffic until a client authenticates with the NCC through the specifically designated web portal page. select My RADIUS server to block network traffic until a client authenticates with an external RADIUS server through the specifically designated web portal page. Enable MAC authentication fallback when both RADIUS-based MAC authentication and web authentication are implemented.
	Scenario 1: When MAC authentication fails. A WiFi client tries to connect to the WiFi network using MAC authentication (RADIUS server). If MAC authentication fails, he will fall back to web authentication. The WiFi client needs to provide a user name and password for web authentication.
	Scenario 2: When MAC authentication is successful. A WiFi client tries to connect to the WiFi network and passes MAC authentication. Web authentication is then skipped.
	Note: When MAC authentication fallback is enabled, the WiFi client can avoid network disassociations due to MAC authentication failure.
	• select Facebook to block network traffic until a client authenticates with the NCC using Facebook Login.
	Facebook Login is a secure and quick way for users to log into your app or website using their existing Facebook accounts. If you get the App ID for your app at the Facebook developers site, you can enter your Facebook app ID to obtain more information about your users using Facebook Analytics, such as user activity, age, gender, and so on.
	 select Facebook Wi-Fi to let users check in to a business on Facebook for free Internet access after connecting to the Nebula Device's WiFi network. Users then have the option to like the Facebook fan page. You should already have set up a Facebook fan page associated with the business location.
	Click here to open the Facebook WiFi configuration screen in a new window, where you can select the Facebook Page associated with your location and configure bypass mode and session length.

 Table 48
 Site-wide > Configure > Access points > SSID advanced settings (continued)

LABEL	DESCRIPTION	
Sign-in method (continued)	Facebook Wi-Fi Configuration S1321.32200016 Facebook Page To use Facebook Wi-Fi you need to be the admin of a local business Page that has a valid location associated with it. Select a Page What with it. Select a Page Wrow route mem sharps have the option to skip checking in. They can do this by clicking on a link that lets them skip check-lin, or by entering a Wi-Fi code that you provide to them. Stip check-lin link 1/1 Require Wi-Fi code 1/1 Select the lingth of time your customers will have Wi-Fi for after they check in. If we hours Terms of Service [?] Optional: Add your own Terms of Service [?] Visit Help Center Note: When the NCC license of the organization expires, the SSID configured with Facebook WiFi will be disabled automatically. To enable the SSID again, change its authentication method or register with a new license key.	
RADIUS server	This field is available only when you select to use the following:	
	 MAC-based Authentication with My RADIUS server or WPA2-Enterprise with My RADIUS server in the WLAN security field, or when you select Sign-on with My RADIUS server in the Sign-in method field. Click Add to specify the IP address/domain name, port number, and shared secret password of the RADIUS server to be used for authentication. Note: User must enter the Account Format and Calling Station ID when MAC authentication fallback field is enabled. Note: Nebula Devices with firmware version 5.50 or older will turn OFF this SSID when the Host field is configured with a domain name. 	
NAS Identifier	If the RADIUS server requires the Nebula Device to provide the Network Access Server identifier attribute with a specific value, enter it here.	
RADIUS accounting	This field is available only when you select to use WPA2-Enterprise with My RADIUS server in the WLAN security field, or when you select Sign-on with My RADIUS server in the Sign-in method field. Select RADIUS accounting enabled to enable user accounting through an external RADIUS server. Select RADIUS accounting disabled to disable user accounting through an external RADIUS	
RADIUS	server.	
accounting	and shared secret password of the RADIUS server to be used for accounting.	
Captive portal advance	e setting	
Walled garden	Select On to enable Walled garden.	

 Table 48
 Site-wide > Configure > Access points > SSID advanced settings (continued)
LABEL	DESCRIPTION				
Walled garden ranges	This field is not configurable if you set Sign-in method to Disable . With a walled garden, you can define one or more web site addresses that all users can access without logging in. These can be used for advertisements for example.				
	Select to turn on or off the walled garden feature.				
	Specify walled garden web site links, which use a (wildcard) domain name or an IP address for web sites that all users are allowed to access without logging in.				
Self-registration	This field is available only when you set Sign-in method to Sign-on with Nebula Cloud authentication .				
	Select Allow users to create accounts with auto authorized or Allow users to create accounts with manual authorized to display a link in the captive portal login page. The link directs users to a page where they can create an account before they authenticate with the NCC. For Allow users to create accounts with manual authorized, users cannot log in with the account until the account is authorized and granted access. For Allow users to create accounts with auto authorized, users can just use the registered account to log in without administrator approval.				
	Select Don't allow users to create accounts to not display a link for account creation in the captive portal login page.				
Simultaneous Iogin limit	This field is available only when you set Sign-in method to Sign-on with My RADIUS server or Sign-on with Nebula Cloud authentication .				
	Select Unlimited if you allow users to log in as many times as they want as long as they use different IP addresses.				
	Select 1 to 10 if you do NOT allow users to have simultaneous logins.				
Strict Policy	Select Allow HTTPS traffic without sign-on to let users use HTTPS to access a web site without authentication.				
	Select Block all access until sign-on to block both HTTP and HTTPS traffic until users authenticate their connections. The portal page will not display automatically if users try to access a web site using HTTPS. They will see an error message in the web screen.				
Reauth time	Select Follow site-wide setting or select a specific time the user can be logged in through the captive portal in one session before having to log in again.				
NCAS disconnect	This field is available only when:				
	 you set Sign-in method to Sign-on with Nebula Cloud authentication you enable MAC-based Authentication with and you select Nebula cloud authentication 				
	Select Allowed to allow any users to access the network without authentication when the NCAS (Nebula Cloud Authentication Server) is not reachable.				
	Select Limited to allow only the currently connected users or the users in the white list to access the network.				
Traffic options					

 Table 48
 Site-wide > Configure > Access points > SSID advanced settings (continued)

LABEL	DESCRIPTION
Forwarding mode	Select Local bridge if you only want to access the Internet. Network traffic from clients connected to the Nebula Device is sent directly to the network through the access point's local gateway.
	Select NAT mode to have the Nebula Device create a DHCP subnet with its own NAT for the SSID. This simplifies WiFi network management, as you do not need to configure a separate DHCP server.
	The following Nebula Device features do not work when NAT mode is enabled:
	 802.11r Layer2 isolation Dynamic VLAN (cloud authentication, RADIUS server)
	Note: In NAT mode, clients cannot communicate with clients connected to a different Nebula Device.
	Select Tunnel mode to forward broadcast and multicast traffic using an existing VLAN interface in the Nebula Device (Security Firewall device). This is the interface you configured in Site-wide > Configure > Security gateway > Interface addressing .
	Note: Tunnel mode is available for Nebula Device (Security Firewall device) only. In Tunnel mode, make sure the ICMP protocol is enabled. See Site-wide > Configure > Firewall: Policy routes/Traffic shaping and Site-wide > Configure > Firewall > Security policy: Action for information.
	Select Tunnel mode for clients that want to access the network behind the Nebula Device. Select Local bridge for clients that want to access the Internet, but you do not want them to access the network behind the Nebula Device.
Rate-limit	Set the maximum data download and upload rates in Kbps, on a per-station basis.
	Click a lock icon to change the lock state. If the lock icon is locked, the limit you set applies to both download and upload traffic. If the lock is unlocked, you can set download and upload traffic to have different transmission speeds.
Advanced settings	

 Table 48
 Site-wide > Configure > Access points > SSID advanced settings (continued)

LABEL	DESCRIPTION				
VLAN ID	Enter the ID number of the VLAN to which the SSID belongs.				
	Note: If you have a Nebula Security Appliance installed in the site but did not configure an identical VLAN interface on the gateway, Smart Guest/VLAN network tip , click here . displays. Click here to open a screen where you can create a gateway interface with the specified VLAN ID.				
	Smart VLAN X				
	Nebula detected that VLAN 1000 has not been created as gateway interface. <i>S</i> Fill-up the VLAN settings and click Continue to proceed with the interface creation, or click Close to skip. VLAN ID (1-4094)				
	IP address				
	Subnet mask ×				
	Port group Port Group 1 👻				
	DHCP None 👻				
	Guest (Enable internet access only)				
	Close Continue				
	Note: If you select Tunnel mode in Forwarding mode , the Tunnel to gateway interface field appears. Select LAN1 as the default.				
Band mode	Select to have the SSID use either 2.4GHz band, 5GHz band, or 6GHz band only.				
Layer 2 isolation	This field is not configurable if you select NAT mode.				
	Select to turn on or off layer-2 isolation. If a device's MAC addresses is NOT listed, it is blocked from communicating with other devices in an SSID on which layer-2 isolation is enabled.				
	Click Add to enter the MAC address of each device that you want to allow to be accessed by other devices in the SSID on which layer-2 isolation is enabled.				
Intra-BSS traffic blocking	Select on to prevent crossover traffic from within the same SSID. Select off to allow intra-BSS traffic.				
Band select	Select to enable band steering. When enabled, the Nebula Device steers WiFi clients to the 5 GHz band.				
	Note: Band mode must be set to Concurrent operation (2.4 GHz and 5 GHz).				
Assisted roaming	Select to turn on or off IEEE 802.11k/v assisted roaming on the Nebula Device.				
	When the connected clients request 802.11k neighbor lists, the Nebula Device will response with a list of neighbor Nebula Devices that can be candidates for roaming. When the 802.11v capable clients are using the 2.4 GHz band, the Nebula Device can send 802.11v messages to steer clients to the 5 GHz band.				
802.11r	Select to turn on or off IEEE 802.11r fast roaming on the Nebula Device.				
	802.1 Ir fast roaming reduces the delay when the clients switch from one Nebula Device to another, by allowing security keys to be stored on all Nebula Devices in a network. Information from the original association is passed to the new Nebula Device when the client roams. The client does not need to perform the whole 802.1x authentication process.				
U-APSD	Select to turn on or off Automatic Power Save Delivery. This helps increase battery life for battery-powered WiFi clients connected to the Nebula Device.				

 Table 48
 Site-wide > Configure > Access points > SSID advanced settings (continued)

5.3.2 Captive Portal Customization

Use this screen to configure captive portal settings for SSID profiles. A captive portal intercepts network traffic until the user authenticates his or her connection, usually through a specifically designated login web page.

Click Site-wide > Configure > Access points > Captive portal customization to access this screen.



aptive portal castornization	
SID	RAP_Field_Trial_Stress (Disabled)
	aptive portal on this SSID is disabled. You can change this setting <u>here</u>
BUTTON	
Click-to-continue/Voucher/Sign	on page
Logo	Upload a logo No logo
Message	×
Success page	
Messoge	Success!
External captive portal URL	
Use URL:	To use custom captive portal page, please download the zip file and edit them. Download the customized captive portal page example.
Captive portal behavior	
After the captive portal page where should go?	the user O Stay on Captive portal authenticated successfully page
	O To promotion URL: X

Table 49	Site-wide >	Configure >	Access points >	Cantive	nortal customization
	3116-MIGE >	Coninguie /	Access points -	Cupilve	pondi costornization

LABEL	DESCRIPTION			
SSID	Select the SSID profile to which the settings you configure here is applied.			
Themes	 This section is not configurable when External captive portal URL is set to ON. Click the Preview icon at the upper right of a theme image to display the portal page a new frame. Click the Copy icon to create a new custom theme (login page). Click the Edit icon of a custom theme to go to a screen where you can view and configure the details of the custom theme pages. See Section 5.3.2.1 on page 258. Click the Remove icon to delete a custom theme page. 			
	Select the theme you want to use on the specified SSID.			
Click-to-continue/Vou This section is not confi	cher/Sign-on page gurable when External captive portal URL is set to ON .			
Logo	This shows the logo image that you uploaded for the customized login page. Click Upload a logo and specify the location and file name of the logo graphic or click			
Browse to locate it. You can use the following image file formats: GIF, PNG, or .				
Message Enter a note to display below the title. Use up to 1024 printable ASCII characte are allowed.				
Success page	Success page			
Message	Enter a note to display on the page that displays when a user logs in successfully. Use up to 1024 printable ASCII characters. Spaces are allowed.			
External captive porta	External captive portal URL			

LABEL	DESCRIPTION				
Use URL	Select On to use a custom login page from an external web portal instead of the one b into the NCC. You can configure the look and feel of the web portal page. Specify the login page's URL; for example, http://IIS server IP Address/login.asp. The Inte Information Server (IIS) is the web server on which the web portal files are installed. Click Download to download a ZIP file containing example captive port files. Edit these then upload them to a webserver which is accessible from NCC.				
	Edit		×		
	URL format: http(s)://external_htm usermac=aa:11:bb:22: apip=192.168.1.35&user //192.168.1.35	nl?gw_addr=http(s)://192.168.1.35&apmac=aa:bb:cc:ee:ff:gg& :c:33& rip=192.168.1.37&ssid_name=MySSID&auth_path=/login.cgi&apurl=http(s):	^		
	Attribute Name	Customized Name			
	gw_addr	gw_addr	*		
	apmac	apmac	× *		
	usermac	usermac	×*		
	apip	apip	× *		
	userip	userip	×*		
	ssid_name	ssid_name	×*		
		c	lose OK		
	ь				
Captive portal behavi	ior				
After the captive portal page where the user should go?	Select To promotion redirected after a su successfully page.	URL and specify the URL of the web site or page to which uccessful login. Otherwise, select Stay on Captive portal a	n the user is uthenticated		

Table 49 Site-wide > Configure > Access points > Captive portal customization (continued)

5.3.2.1 Custom Theme Edit

Use this screen to check what the custom portal pages look like. You can also view and modify the CSS values of the selected HTML file. Click a custom login page's **Edit** button in the **Site-wide** > **Configure** > **Access points** > **Captive portal** screen to access this screen.

Captive portal / Copy of Modern	← Back to c	onfig
Theme name	success.html user_login.html click_to_continue.html social_login.html	cc
Copy of Modern 🗹		•
Font	Save Apply	
Arial 🗘 13px	Welcome to connect	
$B \ \mathbf{I} \ \underline{U} \ \mathbf{\underline{A}} \ \mathbf{\underline{M}} \ \mathbf{\underline{E}} \ \mathbf{\underline{E}}} \ \mathbf{\underline{E}} \ \mathbf{\underline{E}} \ \mathbf{\underline{E}} \ \mathbf{\underline{E}} \ \mathbf{\underline{E}} \ \mathbf{\underline{E}}} \ \mathbf{\underline{E}} \ \mathbf{\underline{E}} \ \mathbf{\underline{E}} \ \mathbf{\underline{E}} \ \mathbf{\underline{E}} \ \mathbf{\underline{E}} \ \mathbf{\underline{E}}} \ \mathbf{\underline{E}} \ \mathbf{\underline{E}} \ \mathbf{\underline{E}}} \ \mathbf{\underline{E}} \ \mathbf{\underline{E}} \ \mathbf{\underline{E}} \ \mathbf{\underline{E}} \ \mathbf{\underline{E}} \ \mathbf{\underline{E}}} \ \mathbf{\underline{E}} \ \mathbf{\underline{E}}} \ \mathbf{\underline{E}} \ \underline{E$	Ē	
Color	This is a message that can be set on NCC.	
R 100 G 240 B 0 # 64F000 Se	e Agree	
	Powered by ZYXEL	
•	•	

Figure 68 Site-wide > Configure > Access points > Captive portal: Edit

LABEL	DESCRIPTION
Back to config	Click this button to return to the Captive portal screen.
Theme name	This shows the name of the theme. Click the edit icon the change it.
Font	Click the arrow to hide or display the configuration fields.
	To display this section and customize the font type and/or size, click on an item with text in the preview of the selected custom portal page (HTML file).
Color	Click the arrow to hide or display the configuration fields.
	Click an item in the preview of the selected custom portal page (HTML file) to customize its color, such as the color of the button, text, window's background, links, borders, and so on.
	Select a color that you want to use and click the Select button.
HTML/CSS	This shows the HTML file name of the portal page created for the selected custom theme. This also shows the name of the CSS files created for the selected custom theme.
	Click a HTML file to display the portal page. You can also change colors and modify the CSS values of the selected HTML file.
$\langle \rangle$	Click this button to view and modify the CSS values of the selected HTML file. It is recommended that you do NOT change the script code to ensure proper operation of the portal page.
\bigcirc	Click this button to preview the portal page (the selected HTML file).
Save	Click this button to save your settings for the selected HTML file to the NCC.
Apply	Click this button to save your settings for the selected HTML file to the NCC and apply them to the access points in the site.

5.3.3 SSID Availability

Use this screen to configure SSID availability and the schedules which can be applied to the SSIDs. The SSID is enabled or disabled at the specified time. Click **Site-wide > Configure > Access points > SSID availability** to access this screen.

Figure 69 Site-wide > Configure > Access points > SSID availability

D:	Youwontbeabletoconnect	v	
sı	D availability		
	Visibility:	Broadcast this SSID 👻	
	Tagging:		
		Enable SSID on APs with any of the specified tags.	
sı	D schedule		
	Enabled		
	Schedule:	NotAtHome 👻 🗹	
	Schedule template:	Custom schedule 🔹	
		Local time zone: Asia - Taipei (You can set this on <u>General settings</u>)	
		Day Availability	
		Sunday 00:00 02:00 04:00 06:00 08:00 10:00 12:00 14:00 16:00 18:00 20:00 22:00 24	• 1:00
		Monday Off 00:00 02:00 04:00 06:00 08:00 10:00 12:00 14:00 16:00 18:00 20:00 24:00 2	4:00
		Tuesday Off 00:00 02:00 04:00 06:00 08:00 10:00 12:00 14:00 16:00 18:00 20:00 24:00 24:00 24:00 16:00	4:00
		Wednesday off 00:00 02:00 04:00 06:00 08:00 10:00 12:00 14:00 16:00 18:00 20:00 22:00 24	4:00
		Thursday Off 00:00 02:00 04:00 06:00 08:00 10:00 12:00 14:00 16:00 18:00 20:00 24:00	4:00
		Friday 00:00 02:00 04:00 06:00 08:00 10:00 12:00 14:00 16:00 18:00 20:00 24:00 24:00 20:00 24:00 20:00	1:00
		Saturday on 00:00 02:00 04:00 06:00 08:00 10:00 12:00 14:00 16:00 18:00 20:00 24:00	1:00
		+ Add Each site can have at most 5 SSID schedules. This schedule also used in SSID(s): Guests- HonduGerma	n

Table 51 Site-wide > Configure > Access points > SSID availability

LABEL	DESCRIPTION
SSID	Select the SSID profile to which the settings you configure here is applied.
SSID availability	

LABEL	DESCRIPTION						
Visibility	Select Hide this SSID if you want to hide your SSID from WiFi clients. This tells any WiFi clients in the vicinity of the Nebula Device using this SSID profile not to display its SSID name as a potential connection. Not all WiFi clients respect this flag and display it anyway. Otherwise, select Broadcast this SSID .						
	When an SSID is "hidden" and a WiFi client cannot see it, the only way you can connect to the SSID is by manually entering the SSID name in your WiFi connection setup screens (these vary by client, client connectivity software, and operating system).						
Tagging	Enter the tags you created for Nebula Devices in the Site-wide > Devices > Access points screen. The SSID profile will only be applied to Nebula Devices with the specified tag.						
	If you leave this field blank, this SSID profile will be applied to all Nebula Devices in the site.						
SSID schedule							
Enabled	Click On to enable and configure a schedule.						
Schedule	Select a schedule to control when the SSID is enabled or disabled. You can click the edit icon to change the schedule name.						
Schedule templates	Select a pre-defined schedule template or select Custom schedule and manually configure the day and time at which the SSID is enabled or disabled.						
Day	This shows the day of the week.						
Availability	Click On to enable the SSID at the specified time on this day. Otherwise, select Off to disable the SSID on the day and at the specified time.						
	Specify the hour and minute when the schedule begins and ends each day.						
Add	Click this button to create a new schedule. A window pops up asking you to enter a descriptive name for the schedule for identification purposes.						
	New Schedule X						
	Name: New Schedule X Close Create						
Delete	Click this button to remove a schedule which is not used in any SSID profile.						

Table 51 Site-wide > Configure > Access points > SSID availability (continued)

5.3.4 Radio Settings

Use this screen to configure global radio settings for all Nebula Devices in the site. Click **Site-wide** > **Configure** > **Access points** > **Radio settings** to access this screen.

io settings		
Country	Taiwan	▼ ● The 6GHz supported country list can be found. <u>Here</u>
Deployment selection	Custom	• 0
Maximum output power	2.4 GHz 30 dBm	v
	5 GHz 30 dBm	*
	6 GHz 30 dBm	Modellist
Channel width	2.4 GHz 20 MHz	•
	5 GHz 80 MHz	Why you should not use channel width 160MHz in 5GHz?
	6 GHz 160 MHz	Modellist
DCS setting	DCC Have Internal	720 X ¹⁰ (60-1440 minutes)
	DCS time interval:	(60-1440 minutes)
	DCS schedule	
	✓ Select all	
	Monday 🔽 Tuesday	
	Wednesday 🔽 Thursday	
	Friday Saturday	
	Sunday	
	03:00 👻	
	DCS client aware	
	Avoid 5G DFS channel	
	Blacklist DFS channels in the pr	resence of radar
	2.4 GHz channel deployment	Three-Channel Deployment 🔹
	5 GHz channel deployment:	All available channels 🔹
	6 GHz channel deployment:	All available channels
llow 802.11ax/ac/n stations only		
	If turned ON, legacy clients including 802.11a	/b/g will not be allowed to associate.
Smart steering	Enable this function will make A	AP steer the client to the better signal AP.
	ADVANCED OPTIONS	
	2.4G Setting	
	Disassociate Station Threshold:	-88 × dBm (-20 ~ -105)
	Optimization aggressiveness: Beta	Modellist O High
		Standard
		O Low

Figure 70 Site-wide > Configure > Access points > Radio settings

	5G Setting					
	Disassociate Station Threshold:	-88	× * d	iBm (-20 ~ -105)		
	Optimization aggressiveness: Beta	Model list	🔵 High			
			O Standard			
			O Low			
	6G Setting					
	Disassociate Station Threshold:	-88	× * d	iBm (-20 ~ -105)		
	Optimization aggressiveness: Beta	Model list	🔿 High			
			Standard			
			O Low			
802.11d 👔	Enable this function will make AP of	ndvertise 80211d ca	inability			
WI AN Pate Control Softing Rate						
WEAN Rule Control Setting Beta						
2.4Ghz 🚺	Lower Density					High Density
	0 1 Mbps					54 Mbps
5Ghz ()	Lower Density					High Density
	0					
	6 Mbps					54 Mbps
6Ghz 1 Model list	Lower Density					High Density
	C. Albert		1			Ed Mana
	o mups					04 MDP5
	EQUA COUR Development	lion				
						Hide transmit circles
Access point Radio # Mo	del Radio mode Channel	Transmit power	Channel width	Smart steering	Antenna	Airtime fairness
	AUTO (DCS)	30 UBIII		Disubled		

Table 52	Site-wide >	Configure > Access points > Radio settings
LABEL		DESCRIPTION

LABEL	DESCRIPTION
Country	Select the country where the Nebula Device is located or installed.
	The available channels vary depending on the country you selected. Be sure to select the correct or same country for both radios on a Nebula Device and all connected Nebula Devices in order to prevent roaming failure and interference with other systems.
Deployment selection	Select High-density (More than 10 APs) for the lowest output power for 10 or more Access Points.
	Select Moderate-density (6-9 APs) for moderate output power for 5 to 9 Access Points.
	Select Low-density (2-5 APs) for higher concentration of output power for less than 5 Access Points.
	Select Single AP for highest concentration of output power for a single Access Point.
Maximum output power	Selecting any of the options in the Deployment selection field will automatically set the maximum output power for 2.4 / 5 / 6 GHz. But you can change the setting (1 – 30 dBm).

LABEL	DESCRIPTION
Channel width	Select the wireless channel bandwidth you want the access point to use.
	A standard 20 MHz channel offers transfer speeds of up to 144 Mbps (2.4 GHz) or 217 Mbps (5 GHz) whereas a 40 MHz channel uses two standard channels and offers speeds of up to 300 Mbps (2.4 GHz) or 450 Mbps (5 GHz). An IEEE 802.11ac-specific 80 MHz channel offers speeds of up to 1.3 Gbps.
	40 MHz (channel bonding or dual channel) bonds two adjacent radio channels to increase throughput. An 80 MHz channel consists of two adjacent 40 MHz channels. The WiFi clients must also support 40 MHz or 80 MHz. It is often better to use the 20 MHz setting in a location where the environment hinders the WiFi signal.
	Note: It is suggested that you select 20 MHz when there is more than one 2.4 GHz Nebula Device in the network.
	Note: It is not possible to set channel bandwidth to 160 MHz for the whole site. To configure a Nebula Device to use 160 MHz, select a supported Nebula Device in the table at the bottom of the screen, click Edit , and then select 160 MHz under Channel width .
DCS setting	
DCS time interval	Select ON to set the DCS time interval (in minutes) to regulate how often the Nebula Device surveys the other Nebula Devices within its broadcast radius. If the channel on which it is currently broadcasting suddenly comes into use by another Nebula Device, the Nebula Device will then dynamically select the next available clean channel or a channel with lower interference.
DCS schedule	Select ON to have the Nebula Device automatically find a less-used channel within its broadcast radius at a specific time on selected days of the week.
	You then need to select each day of the week and specify the time of the day (in 24-hour format) to have the Nebula Device use DCS to automatically scan and find a less-used channel.
DCS client aware	Select ON to have the Nebula Device wait until all connected clients have disconnected before switching channels.
Avoid 5G DFS channel	If your Nebula Devices are operating in an area known to have RADAR devices, the Nebula Device will choose non-DFS channels to provide a stable WiFi service.
Blacklist DFS channels in the presence of radar	Select ON to blacklist a channel if RADAR is detected. After being blacklisted, the Nebula Device will not use the channel again until the Nebula Device is rebooted. However, the Nebula Device can still use other DFS channels.
2.4 GHz channel deployment	Select Three-Channel Deployment to limit channel switching to channels 1, 6, and 11, the three channels that are sufficiently attenuated to have almost no impact on one another. In other words, this allows you to minimize channel interference by limiting channel-hopping to these three "safe" channels.
	Select Four-Channel Deployment to limit channel switching to four channels. Depending on the country domain, if the only allowable channels are 1 – 11 then the Nebula Device uses channels 1, 4, 7, 11 in this configuration; otherwise, the Nebula Device uses channels 1, 5, 9, 13 in this configuration. Four channel deployment expands your pool of possible channels while keeping the channel interference to a minimum.
	Select All available channels to allow channel-hopping to have the Nebula Device automatically select the best channel.
	Select Manual to select the individual channels the Nebula Device switches between.

 Table 52
 Site-wide > Configure > Access points > Radio settings (continued)

LABEL	DESCRIPTION
5 GHz channel deployment	Select how you want to specify the channels the Nebula Device switches between for 5 GHz operation.
	Select All available channels to have the Nebula Device automatically select the best channel.
	Select Manual to select the individual channels the Nebula Device switches between.
	Note: The method is automatically set to All available channels when no channel is selected or any one of the previously selected channels is not supported.
6 GHz channel deployment	Select how you want to specify the channels the Nebula Device switches between for 6 GHz operation.
	Select All available channels to have the Nebula Device automatically select the best channel.
	Select Manual to select the individual channels the Nebula Device switches between.
	Note: The method is automatically set to All available channels when no channel is selected or any one of the previously selected channels is not supported.
Allow 802.11ax/ac/n stations only	Select ON to have the Nebula Device allow only IEEE 802.11n/ac/ax clients to connect, and reject IEEE 802.11a/b/g clients.
Smart Steering	Select ON to enable smart client steering on the Nebula Device. Client steering helps monitor WiFi clients and drop their connections to optimize the bandwidth when the clients are idle or have a low signal. When a WiFi client is dropped they have the opportunity to steer to an Nebula Device with a strong signal. Additionally, dual band WiFi clients can also steer from one band to another.
	Select OFF to disable this feature on the Nebula Device.
ADVANCED OPTIONS	Click this to display a greater or lesser number of configuration fields.
2.4G/5G/6G Settin	g
Disassociate Station Threshold	Set a minimum kick-off signal strength. When a WiFi client's signal strength is lower than the specified threshold, the Nebula Device disconnects the WiFi client.
	–20 dBm is the strongest signal you can require and –105 dBm is the weakest.
Optimization aggressiveness	High , Standard and Low stand for different traffic rate threshold levels. The level you select here decides when the Nebula Device takes action to improve the access point's WiFi network performance. The Nebula Device will postpone the actions implemented on access points until your network is less busy if the threshold is exceeded.
	Select a suitable traffic rate threshold level for your network.
	High : Select this if you want the Nebula Device to postpone the action set when the access point network traffic is heavy.
	Standard : Select this if you want the Nebula Device to postpone the action set when the access point network traffic is medium.
	Low: Select this if you want the Nebula Device to postpone the action set when the access point network traffic is low.
802.11d	Click this to enable 802.11d on the access point.
	802.11d is a WiFi network specification, for use in countries where 802.11 WiFi is restricted. Enabling 802.11d causes the Nebula Device to broadcast the country where it is located, which is determined by the Country setting.
WLAN Rate Contro	l Setting

Table 52 Site-wide > Configure > Access points > Radio settings (continued)

LABEL	DESCRIPTION
2.4Ghz/5Ghz/ 6Ghz	Sets the minimum data rate that 2.4 GHz, 5 GHz, and 6 GHz WiFi clients can connect to the Nebula Device, in Mbps.
	Increasing the minimum data rate can reduce network overhead and improve WiFi network performance in high density environments. However, WiFi clients that do not support the minimum data rate will not be able to connect to the Nebula Device.
Edit	Click this button to modify the channel, output power, channel width, airtime fairness (the same setting will apply to both 2.4 GHz and 5 GHz), and smart steering settings for the selected Nebula Devices. On the Nebula Device that comes with internal antennas and also has an antenna switch, you can adjust coverage depending on the orientation of the antenna for the Nebula Device radios. Select Wall if you mount the Nebula Device to a wall. Select Ceiling if the Nebula Device is mounted on a ceiling. You can switch from Wall to Ceiling if there are still WiFi dead zones, and so on. If you select Hardware Switch , you use the physical antenna switch to adjust coverage and apply the same antenna orientation settings to both radios.
	Edit ×
	Access Paint: BCCP4FE37C.99 Radio #: 1 Model: NWATDAX Band: 24 0Hz Radio mode: 1 ØDnnel: 1 Ohannel: 1 Othannel: 1 Othannel: 1 Othannel: 1 Band: 1 Optimization aggressiveness: 1 Optimization aggressiveness: 1 Optimization aggressiveness: 1 Optimization aggressiveness: 1 <tr< td=""></tr<>
	Note: On this screen, you can set channel width to 160 MHz for the 5/6 GHz channel, if the Nebula Device supports it.
DCS Now	Click this button to have the selected Nebula Devices immediately scan for and select a channel that has least interference.
List	Click this to display a list of all connected Nebula Devices.
Мар	Click this to display the locations of all connected Nebula Devices on the Google map.
2.4GHz	Click this to display the connected Nebula Devices using the 2.4 GHz frequency band.
5GHz	Click this to display the connected Nebula Devices using the 5 GHz frequency band.
6GHz	Click this to display the connected Nebula Devices using the 6 GHz frequency band.
BandFlex	Click this to display the connected Nebula Devices that supports BandFlex (5 GHz or 6 GHz frequency bands).
Hide transmit circles	Click this button to not show the transmission range on the Map.
Access point	This displays the descriptive name or MAC address of the connected Nebula Device.

Table 52 Site-wide > Configure > Access points > Radio settings (continued)

LABEL	DESCRIPTION
Radio #	This displays the number of the connected Nebula Device's radio.
Model	This displays the model name of the connected Nebula Device.
Radio mode	This displays the type of WiFi radio the Nebula Device is currently using, for example 802.11b/g/n.
Channel	This displays the channel ID currently being used by the connected Nebula Device's radio.
Transmit power	This displays the current transmitting power of the connected Nebula Device's radio. If the Nebula Device is offline, this shows the maximum output power you configured for the Nebula Device.
Channel width	This displays the wireless channel bandwidth the connected Nebula Device's radio is set to use.
Smart steering	This displays whether smart client steering is enabled or disabled on the connected Nebula Devices.
Antenna	This displays the antenna orientation settings for the Nebula Device that comes with internal antennas and also has an antenna switch.
Airtime fairness	This displays whether airtime fairness is enabled or disabled on the connected Nebula Device.
R	Click this icon to display a greater or lesser number of configuration fields. For faster loading of data, select only the configuration fields listed that do NOT take a long time to fetch data.

 Table 52
 Site-wide > Configure > Access points > Radio settings (continued)

The following table describes the pre-defined deployments and the related output power, channel width, DFS (Dynamic Frequency Selection) setting, rate control, and channel deployment.

Table 53	Radio Deployment	Selection and	Corresponding	Parameters
----------	------------------	---------------	---------------	------------

DEPLOYMENT		HIGH DENSITY	MODERATE DENSITY	LOW DENSITY	SINGLE AP
Number of APs		More than 10	6 – 9	2 – 5	1
Power (dBm)	2G	12	15	20	30
					20 (EU)
	5G	15	18	30	30
	6G	18	21	30	30
Channel width (MHz)	5G	20	40	80	80
	6G	80	160	160	160
Avoid 5G DFS channel / Blacklist DFS channels in the presence of	radar	Disabled / Enabled	Enabled / Disabled	Enabled / Disabled	Enabled / Disabled
Rate control (Mbps)	2.4G	11	1	1	1
	5G	12	6	6	6
2.4G channel deployment		All channels	Three-channel	Three-channel	Three-channel

5.3.5 Traffic Shaping

This feature is for dynamic VLAN application. The data limit set here applies to the VLAN on a per WiFi client basis. This has a higher priority than the data limit set in **Site-wide** > **Configure** > **Access points** > **SSID advanced settings**, which is applied on a per station basis. Use this screen to configure maximum bandwidth on the Nebula Device.

Click Site-wide > Configure > Access points > Traffic shaping to access this screen.

WLAN traffic shaping Beta Mon	<u>del list</u>												
Rule Name			× *										
VLAN id			× *										
Rate-limit		 Download	0	214	914	454	514	6M	75.4	814	QM.	1014	unlimited (Mb/s) (1 - 160
	6		(Per	client	device t	raffic ro	ite)	UN	/14/	OW	5141	IOW	
	L	 Upload	0										unlimited (Mb/s) (1 - 16
			1M	2M	ЗМ	4M	5M	6M	7M	8M	9M	10M	

LABEL	DESCRIPTION
WLAN traffic shaping	
Rule Name	Enter the name of the traffic shaping rule. The name is used to refer to the traffic shaping rule. You may use 1 – 31 alphanumeric characters, underscores(_), or dashes (-). This value is case-sensitive.
VLAN ID	Enter the VLAN ID. This 12-bit number uniquely identifies each VLAN. Allowed values are 1 – 4094. (0 and 4095 are reserved.)
Rate-limit	Set the maximum data download and upload rate in Mb/s, on a per WiFi client basis. Allowed values are 1 – 160. Click the lock icon to change the lock state. If the lock icon is locked, the data limit you set applies to both download and upload traffic. If the lock is unlocked, you can set download and upload traffic to have different data limits.
Add	Click this button to create a new rule.

Table 54 Site-wide > Configure >	Access points > Traff	ic shaping
----------------------------------	-----------------------	------------

5.3.6 Security Service

Use this screen to enable or disable the features available in the security pack for your Nebula Device, such as application visibility and optimization and/or IP reputation filter.

Click Site-wide > Configure > Access points > Security service to access this screen.

urity service	
pplication Visibility & Optimization Beta	Model list
Application visibility & Optimization	Application visibility optimizes wireless experience via application level thrattling on a per-user basis.
	View Application and configure Bandwidth limit <u>here</u>
hreat Protection Beta Model list	
Enabled	DNS Threat/IP Reputation Filter screens out unsafe phishing sites or botnets to provide users a trustworthy wireless service.
Block log	
Click to proceed	Allow users to browse unsafe sites. Proceed to unsafe website at user own risk.
Denied access message	This high risk page is blocked by Zyxel Connect & Protect service due it may contain maliclous $~ imes~$
Redirect external URL	URL: ×
	To use custom captive portal page, please download the zip file and edit them.
	Download the customized captive portal page example.
Notification page	
Enable on	e-Nebula-MAC 💌
Access message	Zyxel AP proactively secure your network and establish a trustworthy Wireless LAN to protect $_{1}$ $_{ imes}$ *
Category list	✓ Tor Proxy ✓ Mobile Threats ✓ Anonymizers ✓ Phishing
	✓ Malicious Dowloads ✓ Denial of Service ✓ Scanners ✓ BotNets
	Web Attacks and Malicious Sites ✓ Exploits Spyware and Adware Keyloggers ✓ Spam URLs
IP Reputation exempt list	IP or CIDR
	×
DNS Threat exempt list	FQDN(support wildcard)
	×

Figure 72 Site-wide > Configure > Access points > Security service

Table 55	Site-wide >	Configure >	Access	points >	Security	y service

LABEL	DESCRIPTION
Application Visibility &	Optimization
Application visibility & Optimization	Select this option to turn on application visibility and optimization. Application visibility and optimization does the following:
	Detects the type of applications used by WiFi clients,Throttles specific applications to save WiFi bandwidth.
	Application visibility provides a way for a Nebula Device to manage the use of various applications on its WiFi network. It can detect the type of applications used by WiFi clients and how much bandwidth they use.
	Application optimization limits the applications bandwidth usage by their categories. You can manage and view the applications and their categories in Site-wide > Applications usage > Application view by Access Point .
Threat Protection	

LABEL	DESCRIPTION				
Enabled	Select this option to allow inspection of DNS queries made by clients on your network and turn on IP blocking on the Nebula Device.				
	When you enable the DNS threat service, your Nebula Device inspects the DNS queries against a database of blocked or allowed Fully Qualified Domain Names (FQDNs). You can have the Nebula Device reply to the user with a fake DNS response (where the user will see a "Web Page Blocked!" page).				
	When you enable the IP reputation service, your Nebula Device downloads signature files that identifies reputation of IPv4 addresses. You can have the Nebula Device forward, block, and/or log packets from IPv4 addresses based on these signatures and categories.				
Block log	Select this option to create a log on the Nebula Device when the packet comes from an IPv4 address with bad reputation.				
Click to proceed	Select this option to allow clients to browse unsafe websites. When enabled, the denied access message window includes the Proceed button. To continue, you must close and restart your web browser to visit the unsafe website.				
	 This page is blocked by Zyxel Connect & Protect service due to the destination IP address may contain malicious content, malware, or a phishing site, or other security threat. Detail information: Category: Web Attacks and Malicious Sites Time: Tue Oct 11 07:50:46 2022 URL: iranact.co 				
	Ignore the risk and go to this unsafe site. Proceed connection might lead to personal information breaches. Proceed				
	Powered by ZYXEL				
Denied access message	Enter a message to be displayed when IP reputation filter blocks access to a web page. Use up to 127 characters (0–9a–zA–Z;/?:@&=+\$\l~*'()%,''). For example, "Access to this web page is not allowed. Please contact the network administrator".				
	It is also possible to leave this field blank if you have a URL specified in the Redirect external URL field. In this case if the IP reputation filter blocks access to a web page, the Nebula Device just opens the web page you specified without showing a denied access message.				
Redirect external URL	Enter the URL of the web page to which you want to send users when their web access is blocked by IP reputation filter. The web page you specify here opens in a new frame below the denied access message.				
	Use "http://" or "https://" followed by up to 262 characters (0–9a–zA–Z;/?:@&=+\$\l~*'()%). For example, http://192.168.1.17/blocked access.				
Notification page	Select this option to display the notification page.				
Enable on	Select the SSID 1 – 8 that is allowed access to WiFi clients.				
Access message	Enter a message to be displayed when access to a web page is allowed. Use up to 127 characters (0–9a–zA–Z;/?:@&=+\$\!~*'()%,"). For example, "Access to this web page is not allowed. Please contact the network administrator".				

Table 55 Site-wide > Configure > Access points > Security service (continued)

LABEL	DESCRIPTION
Category list	Select the categories of packets that come from the Internet and are known to pose a security threat to users or their computers.
IP Reputation exempt list	Sites that you want to allow access to, regardless of their content rating, can be allowed by adding them to this list. Add the IPv4 addresses that the Nebula Device will allow the incoming and outgoing packets.
DNS Threat exempt list	Domain names that you want to allow access to, regardless of their reputation, can be allowed by adding them to this list. Add the Fully Qualified Domain Names (FQDNs) that the Nebula Device will allow the DNS query packets.

Table 55 Site-wide > Configure > Access points > Security service (continued)

5.3.7 AP & Port Settings

Use this screen to configure general Nebula Device settings and network traffic load balancing between the Nebula Devices in the site. This screen also allows you to enable or disable a port on the managed Nebula Device and configure the port's VLAN settings. The port settings apply to all Nebula Devices that are assigned to the site and have one or more than one Ethernet LAN port (except the uplink port).

Click Site-wide > Configure > Access points > AP & port settings to access this screen.

& port settings					Verride access point configu
General setting					
AP LED lights					
AP Smart mesh	Model list				
Ethernet failover					
Load balancing					
	O Disable				
	Enable "By client d	levice number" mode			
	Recommended for	general use			
	2.4G Mc	aximum client device number:		10 × (1~127)	
	5G Mo	aximum client device number:		10 × (1~127)	
	6G Mc	aximum client device number:	Model list	10 × (1~127)	
				Disassociate client device when	overloaded
	C Enable "Smart Cla	ssroom" mode			
	Recommended for	E-learning only			
	2.4G Mc	ximum client device number:		10 × (1-127)	
	5G Mc	ximum client device number:		10 × (1-127)	
	6G Mc	ximum client device number:	Model list	10 × (1-127)	
Port setting					
LANT					
	PVID	1	×		
	Allowed VLANs ()	1	×		
LAN 2					
	PVID	1	×		
	Allowed VLANs ()	1	×		
LAN 3					
	PVID	1	×		
	Allowed VLANs (1	×		
Access point	Status			Port Setting	

Figure 73 Site-wide > Configure > Access points > AP & port settings

The following table describes the labels in this screen.

Table 56 Site-wide > Configure > Access points > AP & port settings

LABEL	DESCRIPTION			
General setting				
AP LED lights Click to turn on or off the LEDs on the Nebula Devices.				

LABEL	DESCRIPTION
AP Smart Mesh	Click to enable or disable the Nebula Smart Mesh feature on all Nebula Devices in the site.
	Click Model list to see whether your Nebula Device supports Nebula Smart Mesh.
	Note: Nebula Smart Mesh is a WiFi mesh solution for Nebula Devices. For details, see Section 5.1.1 on page 233.
	Note: You can override NCC settings and enable or disable Smart Mesh on individual Nebula Devices. For details, see Section 4.3.1.1 on page 159.
	Note: Disabling Nebula Device Smart Mesh automatically disables wireless bridge on all Nebula Devices in the site. For details on wireless bridge, see Section 4.3.1.1 on page 159.
Ethernet failover	When enabled, a wired Nebula Device in the site automatically changes its role from mesh controller to mesh extender if the Nebula Device is unable to reach the site's gateway.
	When disabled, a wired Nebula Device in the site automatically changes its role from mesh controller to mesh extender only if the Nebula Device's uplink Ethernet cable is unplugged.
	Note: For details on mesh controller and mesh extender, see Section 5.1.1 on page 233.
Load balancing	
Disable	Select this option to disable load balancing on the Nebula Device.
Enable "By client device number" mode	Select this option to balance network traffic based on the number of specified client devices connected to the Nebula Device.
Maximum client device number	Enter the threshold number of client devices at which the Nebula Device begins load balancing its connections.
Disassociate client device when overloaded	Select ON to disassociate WiFi clients connected to the Nebula Device when it becomes overloaded.
	Select OFF to disable this option, then the Nebula Device simply delays the connection until it can afford the bandwidth it requires, or it transfers the connection to another Nebula Device within its broadcast radius.
	The disassociation priority is determined automatically by the Nebula Device and is as follows:
	• Idle Time – Devices that have been idle the longest will be disassociated first. If none of the connected devices are idle, then the priority shifts to Signal Strength.
	• Signal Strength – Devices with the weakest signal strength will be disassociated first.
Enable "Smart Classroom" mode	Select this option to balance network traffic based on the number of specified client devices connected to the Nebula Device. The Nebula Device ignores association request and authentication request packets from any new client device when the maximum number of client devices is reached.
	The Disassociate client device when overloaded function is enabled by default and the disassociation priority is always Signal Strength when you select this option.
Maximum client device number	Enter the threshold number of client devices at which the Nebula Device begins load balancing its connections.
Port setting	•
LAN x	This is the name of the physical Ethernet port on the Nebula Device.
	This section lets you configure global port VLAN settings for all Nebula Devices in the site. To modify port settings for a specific Nebula Device, use its Edit button in the table below.
ON/OFF	Select ON to turn on the LAN port of the Nebula Device. Select OFF to disable the port.

Talala E/	Cite wide >	Configuras	1	in a limita >		in out a ottino ora	(a a lation of a d)
	Slie-wide 2	\cdot Connoure >	ACCESS	DOIDIS > P	4 P &	DOLLSETINGS	ICOMINUEOL
10010 00		Connigoro	,			porrigo	

LABEL	DESCRIPTION
PVID	Enter the port's PVID.
	A PVID (Port VLAN ID) is a tag that adds to incoming untagged frames received on a port so that the frames are forwarded to the VLAN group that the tag defines.
Allowed VLANs	Enter the VLAN ID numbers to which the port belongs.
	You can enter individual VLAN ID numbers separated by a comma or a range of VLANs by using a dash, such as 1,3,5–8.
Access Point	This displays the descriptive name or MAC address of the connected Nebula Device.
	Only the Nebula Device that has an extra Ethernet LAN port will be listed, such as NAP203 or NAP303.
Status	This shows whether the Nebula Device's Ethernet LAN port is enabled or disabled.
Port Setting	This displays the port's VLAN settings for the managed Nebula Device.

Table 56 Site-wide > Configure > Access points > AP & port settings (continued)

5.3.7.1 Edit Port Settings

Click an entry in the **Port setting** table of the **Site-wide** > **Configure** > **Access points** > **AP & port settings** screen to access this screen.

Select **NAT mode** to have the Nebula Device create a DHCP subnet with its own NAT for the SSID. This simplifies WiFi network management, as you do not need to configure a separate DHCP server. Otherwise, select **Local bridge**.

The following Nebula Device features do not work when NAT mode is enabled:

- 802.11r (see Table 48 on page 249 for more information on enabling 802.11r)
- Layer2 isolation
- Dynamic VLAN (cloud authentication, RADIUS server)

Note: In NAT mode, clients cannot communicate with clients connected to a different Nebula Device.

Only WAC500H supports Ethernet Traffic options Forwarding Mode at the time of writing.

By default, all Nebula Devices in the site use the global port settings. Use this screen to change the port settings on a per-device basis. You can turn on or off the port, modify its PVID or update the ID number of VLANs to which the port belongs.

Fiaure 74	Site-wide > Configure > A	.ccess points > AF	% port settings: Edit

Edit				×
Ethernet Traffic options Forwarding Mode	 Local bridge NAT mode Use Zyxel DHCP & NA Clients recieve IP add Client cannot communication 	AT 1 Beta <u>Model list</u> dresses in an isolated netw unicate with other cllients	/ork. associated with different /	ΔP.
LAN1				
PVID	1	×		
Allowed VLANs	1	×		
				Close OK

CHAPTER 6 Switch

6.1 Overview

This chapter discusses the menus that you can use to monitor the Nebula managed Switches in your network and configure settings even before a Nebula Device is deployed and added to the site.

Nebula Device refers to Zyxel Hybrid Switches (GS / XGS / XMG / XS Series) in this chapter. To view the list of Nebula Devices that can be managed through NCC, go to Help > Support tools > Device function table.

6.2 Monitor

Use the **Monitor** menus to check the Nebula Device information, client information, event log messages and summary report for Nebula Devices in the selected site.

6.2.1 Event Log

Use this screen to view Nebula Device log messages. You can enter the Nebula Device name or a key word, select one or multiple event types, or specify a date/time or even a time range to display only the log messages related to it.

Click Site-wide > Monitor > Switches > Event log to access this screen.

iwitch:		Keyword:		Pric	ority:		Category:		Tag:				
Any	×	Any	×	A	ny		Any		Any				
						From:		To:					
					Range 💌	2022-07-07	10:05 🔻	2022-07-07		11	.05	UTC+8	🗵 🔍 Sea
						Max range is 30 d	ays, the dates will be auto-adjusted.						
Kewer Older	555	Event logs											💙 🖹 Expo
Timo	Priority	Switch	Taa		Category	Dotail							
1111e	Frioricy	Switch	iug		Category	Detail							
2022-07-07 11:03:01	informatio	n <u>XS3800-1-1</u>	syster	n.	System	Auto re:	store back up configuration						
2022-07-07 11:02:56	Informatio	n <u>XS3800-1-1</u>	syster	n	System	Save sy	stem configuration						
2022-07-07 10:57:39	Informatio	n <u>XS3800-1-1</u>	syster	n	System	Save sy	stem configuration						
2022-07-07 10:57:01	Notice	XS3800-1-1	syster	n	System	Gets th	e time and <mark>d</mark> ate from a time se	erver successfully					
2022-07-07 10:56:54	Informatio	n <u>XS3800-1-1</u>	syster	n	System	Save sy	stem configuration						
2022-07-07 10:56:46	Informatio	n <u>XS3800-1-1</u>	switch		System	Cloud: [Device is online, VLAN 1, DHCP	IP 10.214.48.34					
2022-07-07 10:56:45	Informatio	n <u>XS3800-1-1</u>	switch		System	Cloud: 8	Set IP 10.214.48.34 on VLAN 1 by	Local					
2022-07-07 10:56:25	Informatio	n <u>XS3800-1-1</u>	switch		Switch	Cloud N	letconf connection has been t	erminated					
2022-07-07 10:53:43	Informatio	n <u>XS3800-1-1</u>	syster	n	System	Save sy	stem configuration						
2022-07-07 10:52:08	Notice	XS3800-1-1	syster	n	System	Gets th	e time and date from a time se	erver successfully					

Figure 75 Site-wide > Monitor > Switches > Event log

276

6.2.2 Surveillance

Use this screen to view information about Powered Devices (PDs) connected to ports on the Nebula Device.

Click Site-wide > Monitor > Switches > Surveillance to access this screen.

Figure 76 Site-wide > Monitor > Switches > Surveillance

Search ports	6) Switch ports						, x0 @ x0	₩ × 0 ?
vitch/Port	Port name	PD health	Link speed	PoE draw(W)	Bandwidth (Kbps)	CRC	Extended range	Device type	System name
D:CF:4F:47:7D:F1(Port1		Auto-1000M	0.0W	Tx: 10.68 Rx: 2.23	0	Disable		
D:CF:4F:47:7D:F1(Port2		Offline	0.0W	Tx: 0.00 Rx: 0.00	0	Disable		
D:CF:4F:47:7D:F1(Port3		Auto-1000M	0.0W	Tx: 2.60 Rx: 9.06	0	Disable	Others	XGS4600
D:CF:4F:47:7D:F1(Port4		Offline	0.0W	Tx: 0.00 Rx: 0.00	0	Disable		
D:CF:4F:47:7D:F1(Port5		Offline	0.0W	Tx: 0.00 Rx: 0.00	0	Disable		
D:CF:4F:47:7D:F1(Port6		Offline	0.0W	Tx: 0.00 Rx: 0.00	0	Disable		

LABEL	DESCRIPTION
Search ports	Enter a keyword to filter the list of ports or devices.
N switch ports	This shows the number of Nebula Device ports (N) in the list.
\triangle	This shows the number of connected PDs that did not respond to an automatic PD alive check.
0	This shows the number of ONVIF-compatible IP camera devices connected to Nebula Devices in the site.
•**	This shows the number of ONVIF-compatible NVR devices connected to Nebula Devices in the site.
?	This shows the number of connected devices that did not respond to an ONVIF discovery query, or are of an unknown type.
Switch/Port	This shows the port number of the Nebula Device.
Port name	This shows the port description on the Nebula Device.
PD health	This shows the status of auto PD recovery on this port.
	 Red: The Nebula Device failed to get information from the PD connected to the port using LLDP, or the connected PD did not respond to the Nebula Device's ping requests. Yellow: The Nebula Device is restarting the connected PD by turning the power off and turning it on again. Green: The Nebula Device successfully discovered the connected PD using LLDP or ping. : Auto PD Recovery is not enabled on the Nebula Device and/or the port, or the switch is not supplying power to the connected PD.
	page 288.
Link speed	This shows the speed (either 10M for 10 Mbps, 100M for 100 Mbps, or 1G for 1 Gbps) and the duplex (F for full duplex or H for half). This field displays Down if the port is not connected to any device.

Table 57 Site-wide > Monitor > Switches > Surveillance

LABEL	DESCRIPTION
PoE draw(W)	This shows the total power that the connected PD draws from the port, in watts. This allows you to plan and use within the power budget of the Nebula Device.
Bandwidth (Kbps)	Tx shows the number of kilobytes per second transmitted on this port. Rx shows the number of kilobytes per second received on this port.
CRC	This shows the number of packets received with CRC (Cyclic Redundant Check) errors.
Extended range	This shows whether extended range is enabled on the port.
Device type	This shows the device type of the PD, as reported by ONVIF discovery.
System name	This shows the name of the connected PD, as reported by ONVIF or LLDP.
IP	This shows the IP address of the connected PD, as reported by ONVIF or LLDP.
Discovered devices	This shows how many devices are connected to the port.
	Click the number to go to the Surveillance Port Details screen.

Table 57 Site-wide > Monitor > Switches > Surveillance (continued)

6.2.3 Surveillance Port Details

Use this screen to view detailed information about a port on the Surveillance screen.

Go to Site-wide > Monitor > Switches > Surveillance and click on a value in the Discovered Devices column to access this screen.

Figure 77	Site-wide >	Monitor >	Switches >	Surveillance	> Port Details
	0.1.0 1110.0		0	0011011000	

Surveillance / BC:CF:4	1F:47:7D:F1(GS1350-6HP) / Po	ort 3 Last 2 hours	• Č				
Status							
Link speed:	Auto-1000M			Bandwidth Tx/Rx(Kbps):	2.46/8.97		
PoE draw:	0.0 W			CRC:	0		
PD health:				Power cycle:			
Extended range:	Disable						
Neighbor detail							
Q Search clients	1 clients						Flush
Status	System name	Device type	Port	IP	Firmware	Description	Ę
•	XGS4600	Others	2	192.168.30.15	V4.70(ABBH.3) 04/27/2	2022	

Table 58 Site-wide > Monitor > Switches > Surveillance > Port Details

LABEL	DESCRIPTION
Status	
Link speed	This shows the speed (either 10M for 10 Mbps, 100M for 100 Mbps, or 1G for 1 Gbps) and the duplex (F for full duplex or H for half). This field displays Down if the port is not connected to any device.
PoE draw	This shows the total power that the connected PD draws from the port, in watts. This allows you to plan and use within the power budget of the Nebula Device.

LABEL	DESCRIPTION
PD health	This shows the status of auto PD recovery on this port.
	 Red: The Nebula Device failed to get information from the PD connected to the port using LLDP, or the connected PD did not respond to the Nebula Device's ping requests. Yellow: The Nebula Device is restarting the connected PD by turning the power off and turning it on again. Green: The Nebula Device successfully discovered the connected PD using LLDP or ping. : Auto PD Recovery is not enabled on the Nebula Device and/or the port, or the Nebula Device is not supplying power to the connected PD.
	For details on configuring auto PD recovery on a port, see Section 6.3.1 on page 288.
Extended range	This shows whether extended range is enabled on the port.
Bandwidth Tx/Rx (%)	Tx shows the number of kilobytes per second transmitted on this port. Rx shows the number of kilobytes per second received on this port.
CRC	This shows the number of packets received with CRC (Cyclic Redundant Check) errors.
Power cycle	Click Reset to power off the PD connected to the port, by temporarily disabling then re- enabling PoE.
Neighbor detail	This section shows all clients connected to the port.
Search clients	Search for one or more clients in the list by keyword, status, system name, port, IP address, or firmware version.
clients	This shows the number of clients connected to this port.
Flush	Click this to remove all offline clients from the list.
Status	This shows whether the client is online (green) or offline (red), and whether the client is wired or wireless.
System name	This displays the system name of the Nebula Device.
Port	This displays the number of the Nebula Device port that is connected to the Nebula Device.
IP	This shows the IP address of the Nebula Device.
Firmware	This shows the firmware version currently installed on the Nebula Device.
Description	This shows the descriptive name of the Nebula Device.

Table 58	Site-wide > Monitor	> Switches > Surveillance	> Port Details	(continued)
101010 00				100111100001

6.2.4 IPTV Report

Use this screen to view available IPTV channels and client information.

Click Site-wide > Monitor > Switches > IPTV report to access this screen.

		16 Total channels	6 Channel in use		Curre	11 ent viewers		
hanne	I summary Top 1	0 channels *	Channel in Use		Curr			
lefore	▼ 2022-03-28	📋 14:22 💌 24 h	ou ▼ UTC+8 🗵 Search					
]							
	90							
(%)	80							
ularity	60 56.91 50							
Pop	40 · 30 ·							
	20 · · · · · · · · · · · · · · · ·	9,49 9,49	9.49	4.79				
	0 HBO-123	239.12.4 239.12.3	239122 239121 2	39.255.255.250	239.1.2.6	0.03	0.03 239.1.2.15	0.03
22-03-2 nP pac nP pac 22-03-2 ceptior h ban h ban h ban 22-03-2 22-03-2	28 14:15:29 skets have been detu skets may interfere v 28 10:10:08 nal bandwidth utiliz dwidth utilization m <u>witch nort settings</u> 28 11:15:13	ected on the IPTV network. with IPTV traffic and cause pixilati ation is detected on (XS1930-12HP ay impact video quality. Either ad;	ion. You can use IP Filtering to block -++igmp/port 4) just uplink topology or change the c	UPnP packets. <u>L</u> lownlink max gro	Update filter rule roup value to fix h	s to drop UPnP tra	affic by destinat	ion addr
22-03-2 inP pac inP pac 22-03-2 ception sh ban- heck sv 22-03-2 ined gr ined gr hannel channel	28 14:15:29 ckets have been detr ckets may interfere v 28 10:10:08 nal bandwidth utiliz dwidth utilization m witch port settings. 28 11:15:13 roup requests excees attempted to join m figuration page. I information	ected on the IPTV network. with IPTV traffic and cause pixilati ation is detected on (X\$1930-12HP ay impact video quality. Either ad) d the threshold on (X\$1930-12HP-4 ore channels than allowed. The ea	ion. You can use IP Filtering to block tigmp/port 4) just uplink topology or change the c -tigmp/port 4) arliest channel learned will be replac	UPnP packets. <u>(</u> lownlink max gra	Update filter rule roup value to fix l	s to drop UPnP tri bandwidth utilizat	affic by destinat	ion addr
22-03-2 P P pace 22-03-2 22-03-2 22-03-2 ined gr ined gr ined gr hannel channel Channe	28 14:15:29 ckets have been detu- ckets may interfere v 28 10:10:08 mal bandwidth utiliz- dwidth utilization m witch port settings. 28 11:15:13 oroup requests excee- attempted to join m figuration page. I information	ected on the IPTV network. with IPTV traffic and cause pixilati ation is detected on (XS1930-12HP ay impact video quality. Either adj d the threshold on (XS1930-12HP-+ ore channels than allowed. The ea Channels Switch	ion. You can use IP Filtering to block -+igmp/port 4) just uplink topology or change the c +igmp/port 4) arliest channel learned will be replace	UPnP packets. § Iownlink max gro ed by the new c	Update filter rule roup value to fix l channel. Please c	a to drop UPnP tri bandwidth utilizat heck the group lin	offic by destinat	ion addr
22-03-2 nP pac 22-03-2 ception gh banned set system thannel channel channel channel	28 14:15:29 skets have been detr skets may interfere v 28 10:10:08 hal bandwidth utiliz dwidth utilization m witch port settings. 28 11:15:13 roup requests excees attempted to join m fisuration poge. I information I management I 3.	ected on the IPTV network. with IPTV traffic and cause pixilati ation is detected on (XS1930-12HP ay impact video quality. Either ad) d the threshold on (XS1930-12HP-+ ore channels than allowed. The ea Channels Switch XS1930-12HP-+igmy	ion. You can use IP Filtering to block -+igmp/port 4) just uplink topology or change the c +igmp/port 4) arliest channel learned will be replace Port name p Port4	UPnP packets. <u>(</u> lownlink max gra ed by the new c Port 4	Update filter rule coup value to fix l channel. Please c VID Client 5 19819	a to drop UPnP tri bandwidth utilizat heck the group lir : Vie 1.2 03:	affic by destinat tion. mit on the w-time 13:26	ion addr
22-03-2 PnP pac 22-03-2 22-03-2 22-03-2 22-03-2 22-03-2 22-03-2 22-03-2 22-03-2 22-03-2 22-03-2 22-03-2 22-03-2 2 2-03-2 2 2-03-2 2 2-03-2 2 2 2-03-2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	28 14:15:29 ckets have been detu- ckets may interfere vi- 28 10:10:08 mal bandwidth utilizion dwidth utilization million attempted to join million at	ected on the IPTV network. with IPTV traffic and cause pixilati ation is detected on (XS1930-12HP ay impact video quality. Either add d the threshold on (XS1930-12HP-+ ore channels than allowed. The ea Channels Switch XS1930-12HP-+igmp	ion. You can use IP Filtering to block +-+igmp/port 4) ust uplink topology or change the o +igmp/port 4) arliest channel learned will be replace p Port name p Port4	UPnP packets. s lownlink max gro ed by the new c Port 4	VID Client 5 19819	a to drop UPnP tre bandwidth utilizat heck the group lir 12 03: 122 03:	affic by destinat tion. mit on the 13:26	ion addi
22-03-2 PnP pac 22-03-2 cception gh bann 22-03-2 ined gr ined gr to annel channel Channel HBO-12	28 14:15:29 ckets have been detu- ckets may interfere vi- 28 10:10:08 mal bandwidth utiliz dwidth utilization m witch nort settings. 28 11:15:13 roup requests exceer attempted to join m figuration page. I information I management 1 3.	ected on the IPTV network. with IPTV traffic and cause pixilati ation is detected on (XS1930-12HP ay impact video quality. Either ad) d the threshold on (XS1930-12HP-+ ore channels than allowed. The ear Channels Switch XS1930-12HP-+igmp	ion. You can use IP Filtering to block -+igmp/port 4) use uplink topology or change the c igmp/port 4) arliest channel learned will be replace p Port name p Port4	UPnP packets. § iownlink max gro ed by the new c Port 4 4 4	VID Client 5 19819 5 19819	a to drop UPnP tre bandwidth utilizat heck the group lin 12 03: 122 03:	w-time 13:26 13:26	ion addr
22-03-2 PnP pac 22-03-2 cception 22-03-2 22-03-2 22-03-2 ined gr 22-03-2 ined gr 22-03-2 ined gr port confi hannel Channe HBO-12	28 14:15:29 skets have been detuins skets may interfere violation of the skets may interfere violation of the sketness of th	ected on the IPTV network. with IPTV traffic and cause pixilati ation is detected on (XS1930-12HP ay impact video quality. Either ad) d the threshold on (XS1930-12HP-+ ore channels than allowed. The ea Channels Switch XS1930-12HP-+igmp	ion. You can use IP Filtering to block ++igmp/port 4) ust uplink topology or change the c +igmp/port 4) urliest channel learned will be replace po Port name po Port4 po	UPnP packets. (lownlink max gra ed by the new c 4 4 4 4	VID Client 5 19819 5 19819 5 19819	s to drop UPnP tre pandwidth utilizat heck the group lin 12 03: 122 03: 123 03:	w-time 13.26 13.26	ion addr
22-03-2 PnP pac 22-03-2 22-03-2 22-03-2 22-03-2 22-03-2 22-03-2 ined gr ined gr ined gr ined gr ined gr hannel hannel hannel HBO-12	28 14:15:29 ckets have been detuins 28 10:10:08 anal bandwidth utilizioni dwidth utilization m witch port settings 28 11:15:13 oroup requests exceent attempted to join m figuration page. I information attempted to join m figuration page. I information attempted to join m figuration page.	ected on the IPTV network. with IPTV traffic and cause pixilati ation is detected on (XS1930-12HP ay impact video quality. Either ad) d the threshold on (XS1930-12HP-+ ore channels than allowed. The ea Channels Switch XS1930-12HP-+igmp	ion. You can use IP Filtering to blockigmp/port 4) just uplink topology or change the cigmp/port 4) arliest channel learned will be replace p Port name p Port4 P	UPnP packets. § lownlink max gra ed by the new c 4 4 4 4 4 4 4 4 4	VID Client 5 19819 5 19819 5 19819	a to drop UPnP tri bandwidth utilizat heck the group lin 12 03: 122 03: 123 03: 124 03:	w-time 13:26 13:26 13:26	In the second se Second second sec
22-03-2 PnP pac 22-03-2 cception gh bann hack sy 222-03-2 hined gr vined gr hannel Channel Channel	28 14:15:29 skets have been detu- skets may interfere vi 28 10:10:08 mai bandwidth utilizi dwidth utilizitons utilizations 28 11:15:13 roup requests exceen- attempted to join m figuration page. 1 information il management 1 3.	ected on the IPTV network. with IPTV traffic and cause pixilati ation is detected on (XS1930-12HP ay impact video quality. Either adj d the threshold on (XS1930-12HP-+ ore channels than allowed. The ea Channels Switch XS1930-12HP-+igmp	ion. You can use IP Filtering to block	UPnP packets. § lownlink max gra ed by the new c 4 4 4 4 4 4 4 4 4 4 4	VID Client 5 19819 5 19819 5 19819 5 19819	a to drop UPnP tri boandwidth utilizat heck the group lin 122 03: 123 03: 124 03: 125 03:	affic by destinat tion. mit on the 13.26 13.26 13.26 13.26 13.26	ion addr

_

LABEL	DESCRIPTION		
IPTV report	Click Model list to show the Non-supported model list. Click See more to go to the Help > Support tools > Device function table screen.		
Email report	Click this button to send channel summary report by email, change the report logo and set email schedules.		
Total channels	This shows the total number of IPTV channels that match the search criteria.		
Channel in use	This shows the number of channels that are being watched by IPTV clients.		
Current viewers	This shows the number of clients who are watching the IPTV channels.		
Channel Summary			
	 Select to view the channels according to the ranking. Alternatively, select Select channels to choose specific channels and click Apply. Top 10 channels Top 11 to 20 channels Bottom 11 to 20 channels Bottom 10 channels Select channels (10 channels max) 		
Search	Specify a date/time and select to view the channels available in the past day, week or month before the specified date/time after you click Search . You can also select Range in the second field, set a time range and click Search to display only the channels available within the specified period of time.		
y-axis	The y-axis represents the Popularity (%) of IPTV channels.		
x-axis	The x-axis shows the name of the IPTV channel. It shows the channel's multicast group address by default.		
Network Analytic Alert	This shows the alerts the NCC generates when an error or something abnormal is detected on the IPTV network.		
	For example, the maximum number of the IGMP multicast groups (TV channels) a Nebula Device port can join is reached and new groups replace the earliest ones, UPnP packets are detected on the IPTV network and may interfere with IPTV traffic to cause TV pixelation, or high bandwidth usage on a certain Nebula Device port results in loss of video quality.		
Channel Information			

Table 59 Site-wide > Monitor > Switches > IPTV Report

LABEL	DESCRIPTION				
Channel Management	Download the channel list and import multiple records for faster channel naming. Click Add to add new channels.				
	Channel management X				
	You can download the <u>channel list</u> here and <u>import</u> multiple records for faster channel naming				
	Channel address Channel name				
	239.1.2.1 × * HBO-123 × *				
	239.1.2.10 × * 239.1.2.10-123 × *				
	+ Add				
	Close				
Channel	This shows the name of the channel. Click the edit icon to change the channel name.				
	Click the channel name to display the channel's client statistics. See Section 6.2.4.2 on page 284.				
Switch	This shows the name of the Nebula Device to which the client is connected.				
Port Name	This shows the name of the Nebula Device port to which the client is connected.				
Port	This shows the number of the Nebula Device port to which the client is connected.				
VID	This shows the ID number of the VLAN to which the Nebula Device port belongs.				
Client	This shows the IP address of the client who is watching the TV program on the channel.				
View-time	This shows the amount of time the client has spent watching the IPTV channel.				

Table 59 Site-wide > Monitor > Switches > IPTV Report (continued)

6.2.4.1 Email Report

Use this screen to configure the email recipient's address, change the logo and set email schedules. To access this screen, click the **Email report** button in the **Site-wide** > **Monitor** > **Switches** > **IPTV Report** screen.

Figure 79	Site-wide >	 Monitor > 	Switches >	 IPTV Report 	: Email report

Email report							×
Email Channel Summary	report - 20)22-03-31 to 2022-04	¥-01				
Address:	samuel.	yu@zyxel.com.tw					*
Format:	HTML	-					Send now
Schedule reports							
Current logo Upload new logo: No logo	Choose F	ile					
Email address		Subject		Frequency	Туре	Channel summary	
y@zyxel.com.tw	×	HTML-test	×	Weekly 🔻	HTML -	Selected: Top 10 channels, Top 11 to 20 channels, Bottom 11 to 20 channels, Bottom 10 channels, 224.0.0.252, 224.0.0.251, 239.255.255.250, 239.121/HBO-123.239.12.3	st channels - 🗂
						Selected: Top 10 channels, Top 11 to 20 channels, Bottom 11 to 20	
Y@zyxel.com.tw	*	plain-test	×	Weekly 🔻	Plain text 🔻	channels, Bottom 10 channels, 224.0.0.252, 224.0.0.251, 239.255.255.250, 239.12.1/HBO-123, 239.12.3	et channels 👻 👘
Y@zyxel.com.tw	× *	plain-test	×	Weekly 🔻	Plain text 🔻	channels, Bottom 10 channels, 224.0.0.252, 224.0.0.251, 239.255.255.250, 239.12.1/HBO-123, 239.12.3	st channeis 🕶 📆

LABEL	DESCRIPTION
Email Channel Summary report	This shows the range of the date/time you specified in the Site-wide > Monitor > Switches > IPTV Report screen.
Address	Enter the recipient's email address of the IPTV channel summary report.
Format	Select to send the IPTV channel summary report in HTML or Plain text format.
Send now	Click this button to send the IPTV channel summary report now.
Schedule reports	
logo	This shows the logo image that you uploaded for the customized IPTV channel summary report.
	Select Current logo to continue using the present logo.
	Select Upload new logo and click Choose File to locate the logo graphic. You can use the following image file formats: GIF, PNG, or JPG. File size must be less than 200 KB, and images larger than 244 x 190 will be resized.
	Select No logo if you do not want a logo to appear on the IPTV channel summary report.

Table 60 Site-wide > Monitor > Switches > IPTV Report: Email report

LABEL	DESCRIPTION		
+ Add	Click this button to add a scheduled IPTV channel summary report profile.		
Email address	Enter the recipient's email address of the IPTV channel summary report.		
Subject	Enter the subject of the IPTV channel summary report.		
Frequency	Select to send the IPTV channel summary report Monthly, Weekly, or Daily.		
Туре	Select to send the IPTV channel summary report in HTML or Plain text format.		
Channel summary			
	channels to choose specific channels and click Update. Image: Top 10 channels Top 10 channels Top 11 to 20 channels Bottom 11 to 20 channels Bottom 10 channels Select channels (10 channels max) 1		
Remove	Click this to delete a scheduled profile.		
Save	Click Save to save the new scheduled profile.		

Table 60	Site-wide > I	Monitor >	Switches >	IPTV Report:	Email report	(continued)
101010 00	0110 111010 1		0.111.011.00			100111100001

6.2.4.2 Channel Information

Use this screen to view the IPTV channel's client information and statistics. To access this screen, click a channel name from the **Channel Information** list in the **Site-wide** > **Monitor** > **Switches** > **IPTV Report** screen.



Figure 80 Site-wide > Monitor > Switches > IPTV Report: Channel Information

 Table 61
 Switches > Monitor > Switches > IPTV Report: Channel Information

LABEL	DESCRIPTION
	Select a specific date to display only the clients who watch the IPTV channel on that day.
Current Viewer	This shows the number of clients who are currently watching the IPTV channel.
y-axis	The y-axis shows the number of clients watching the IPTV channel.
x-axis	The x-axis shows the hour of the day in 24-hour format.
Switch	This shows the name of the Nebula Device to which the client is connected.
Port Name	This shows the name of the Nebula Device port to which the client is connected.
Port	This shows the number of the Nebula Device port to which the client is connected.
VID	This shows the ID number of the VLAN to which the Nebula Device port belongs.
Client	This shows the IP address of the client who is watching the TV program on the channel.
View-time	This shows the amount of time the client has spent watching the IPTV channel.

6.2.5 Summary Report

This screen displays network statistics for Nebula Devices of the selected site, such as bandwidth usage, top ports and/or top Nebula Devices.

Click Site-wide > Monitor > Switches > Summary Report to access this screen.

Figure 81 Site-wide > Monitor > Switches > Summary Report



	Table 62	Site-wide	> Monitor >	Switches >	Summary	/ Report
--	----------	-----------	-------------	------------	---------	----------

LABEL	DESCRIPTION		
Switch – Summary report	Select to view the report for the past day, week or month. Alternatively, select Custom range to specify a time period the report will span. You can also select the number of results you want to view in a table.		
	 Last 24 hours Last 7 days Last 30 days Custom range 2022-07-06 in to 2022-07-07 in (Max range is 30 days, the dates will be auto-adjusted.) 		
	Report size: 10 ▼ results per table		
Email report	Click this button to send summary reports by email, change the logo and set email schedules.		
Consumption			
Total	This shows the total power consumption of the Nebula Device ports.		
Current Consumption	This shows the current power consumption of the Nebula Device ports.		
Max Consumption	This shows the maximum power consumption of the Nebula Device ports.		
Min Consumption	This shows the minimum power consumption of the Nebula Device ports.		
y-axis	The y-axis shows how much power is used in Watts.		
x-axis	The x-axis shows the time period over which the power consumption is recorded.		
Top power consumption	on line line line line line line line lin		
#	This shows the ranking of the Nebula Device.		
Name	This shows the descriptive name of the Nebula Device.		
Model	This shows the model number of the Nebula Device.		
Power Usage	This shows the total amount of power consumed by the Nebula Device's connected PoE devices during the specified period of time.		
Peak Power			
#	This shows the ranking of the Nebula Device.		
Name	This shows the descriptive name of the Nebula Device.		
Model	This shows the model number of the Nebula Device.		
Max Power	This shows the maximum power consumption for the Nebula Device's connected PoE devices during the specified period of time.		
Power %	This shows what percentage of the Nebula Device's total power budget has been consumed by connected PoE powered devices.		
Top uplink port			
#	This shows the ranking of the Nebula Device.		
Name	This shows the descriptive name of the Nebula Device.		
Model	This shows the model number of the Nebula Device.		
Usage	This shows the amount of data that has been transmitted through the Nebula Device's uplink port.		
Top port			

NCC User's Guide

LABEL	DESCRIPTION
#	This shows the ranking of the Nebula Device port.
Name	This shows the descriptive name of the Nebula Device.
Port	This shows the port number on the Nebula Device.
Model	This shows the model number of the Nebula Device.
Usage	This shows the amount of data that has been transmitted through the Nebula Device's port.
Location	
This shows the locatior	n of the Nebula Devices on the map.

Table 62 Site-wide > Monitor > Switches > Summary Report (continued)

6.3 Configure

Use the **Configure** menus to configure port setting, IP filtering, RADIUS policies, PoE schedules, and other Nebula Device settings for Nebula Devices of the selected site.

6.3.1 Switch Ports

Use this screen to view port summary and configure Nebula Device settings for the ports. To access this screen, click Site-wide > Configure > Switches > Switch ports or click the Configure ports button in the Site-wide > Devices > Switch: Switch Details screen.

	Figure 82	Site-wide > Configure > Switches > Switch ports
--	-----------	---

lit A	ggregate • , Split Tag • Reset Q Search	n ports	*	2) selec	cted in (16) Switcl	n ports				🕒 Exp
	Switch / Port	Port name	# Port	LLDP	Received bytes	Sent bytes	Connection	PoE Tag	IPSG protected	Manageme
2	BC:CF:4F:47:7D:F1(GS1850-6HP)/1 Uplink details	Port1	1	nsg50	1.28 MB	3.13 MB		Enabled	N/A	Enabled
1	BC:CF:4F:477D:F1(GS1350-6HP)/2 details	Port2	2	XGS4600	773.77 KB	664.21 KB		Enabled	N/A	Enabled
]	BC:CF:4F:47:7D:F1(GS1350-6HP)/3 details	Port3	з	NAP102	733.08 KB	1.25 MB		3.20	N/A	Enabled
)	BC:CF:4F:47:7D:F1(GS1350-6HP)/4 details	Port4	4	Enabled	0 bytes	0 bytes		Enabled	N/A	Enabled
)	BC:CF:4F:47:7D:F1(GS1350-6HP)/5 details	Port5	5	Enabled	0 bytes	0 bytes		Enabled	N/A	Enabled
ו	BC:CF:4F:47:7D:F1(GS1350-6HP)/6 details	Port6	6	Enabled	0 bytes	0 bytes	1	N/A	N/A	Enabled
]	D8:EC:E5:60:99:8A/1 details	Port1	1	Enabled	0 bytes	0 bytes	j.	N/A	N/A	Enabled
)	D8:EC:E5:60:99:8A/2. details	Port2	2	Enabled	0 bytes	0 bytes		N/A	N/A	Enabled
ן	D8:EC:E5:60:99:8A/3 details	Port3	з	Enabled	0 bytes	0 bytes	j.	N/A	N/A	Enabled
1	D8:EC:E5:60:99:8A/4 details	Port4	4	Enabled	0 bytes	0 bytes	1	N/A	N/A	Enabled

The following table describes the labels in this screen.

Table 63 Site-wide > Configure > Switches > Switch ports

LABEL	DESCRIPTION
Switch ports	Select to view the detailed information and connection status of the Nebula Device port in the past two hours, day, week or month.
C	Click this button to reload the data-related frames on this page.
Edit	Select the ports you want to configure and click this button to configure Nebula Device settings on the ports, such as link aggregation, PoE schedule, LLDP and STP.

NCC User's Guide
LABEL	DESCRIPTION
Aggregate	Select more than one port and click this button to group the physical ports into one logical higher-capacity link.
Split	Select a trunk group and click this button to delete the trunk group. The ports in this group then are not aggregated.
	A trunk group is one logical link containing multiple ports.
Tag	Click this button to create a new tag or delete an existing tag.
Reset	Click this button to reboot the PD (powered device) connected to the PoE port. Follow the prompt and click Confirm to reboot the PD connected to this port.
	Note: This button is not available for an uplink port.
Search	Specify your desired filter criteria to filter the list of Nebula Device ports.
	You can filter the search by selecting one or more Nebula Devices. Under Ports, you can search for multiple ports separated by a comma, or a range separated by a hyphen. For example: 1,2,4–6.
Switch ports	This shows the number of ports on the Nebula Device.
Export	Click this button to save the Nebula Device port list as a CSV or XML file to your computer.
CRC alert icon	This prompt appears if CRC errors are detected in the port(s). Go to Site-wide > Devices > Switches: Switch Details: Port Details for the details. See Section 4.3.2.1 on page 169 for more information.
Switch / Port	This shows the Nebula Device name and port number.
	If the port is added to a trunk group, this also shows whether it is configured as a static member of the trunk group (Static) or configured to join the trunk group through LACP (LACP). If the port is connected to an uplink gateway, it shows Uplink .
	Click details to display the port details screen. See Section 4.3.2.1 on page 169.
Port name	This shows the descriptive name of the port.
#Port	This shows the port number.
LLDP	This shows whether Link Layer Discovery Protocol (LLDP) is supported on the port.
Received broadcast packets	This shows the number of good broadcast packets received.
Received bytes	This shows the number of bytes received on this port.
Received packets	This shows the number of received frames on this port.
Sent broadcast packets	This shows the number of good broadcast packets transmitted.
Sent bytes	This shows the number of bytes transmitted on this port.
Sent multicast packets	This shows the number of good multicast packets transmitted.
Received multicast packets	This shows the number of good multicast packets received.
Sent packets	This shows the number of transmitted frames on this port.
Total bytes	This shows the total number of bytes transmitted or received on this port.
Enabled	This shows whether the port is enabled or disabled.
Link	This shows the speed of the Ethernet connection on this port.
	Auto (auto-negotiation) allows one port to negotiate with a peer port automatically to obtain the connection speed and duplex mode that both ends support.

Table 63 Site-wide > Configure > Switches > Switch ports (continued)

LABEL	DESCRIPTION
Connection	This shows the connection status of the port.
	 Gray (#888888): The port is disconnected. Orange (#FF8900): The port is connected and is transmitting data at 10 or 100 Mbps. Green (#64BE00): The port is connected and is transmitting data at 1000 Mbps (1 Gbps). Azure (#0079FF): The port is connected and is transmitting data at 2.5 Gbps. Violet (#8800FF): The port is connected and is transmitting data at 5 Gbps. Blue (#004FEE): The port is connected and is transmitting data at 10000 Mbps (10 Gbps).
	When the port is in the STP blocking state, failed LACP negotiation state, or failed port authentication state, a blocked icon displays.
	Move the cursor over a time slot to see the actual date and time when a port is connected or disconnected.
Auth. policy	This shows the name of authentication policy applied to the port.
Allowed VLAN	This shows the VLANs from which the traffic comes is allowed to be transmitted or received on the port.
PoE	This shows whether PoE is enabled on the port.
RSTP	This shows whether RSTP is enabled on the port.
Status	If STP/RSTP is enabled, this field displays the STP state of the port.
	If STP/RSTP is disabled, this field displays FORWARDING if the link is up, otherwise, it displays Disabled .
Schedule	This shows the name of the PoE schedule applied to the port.
Туре	This shows the port type (Trunk or Access).
PVID	This shows the port VLAN ID. It is a tag that adds to incoming untagged frames received on the port so that the frames are forwarded to the VLAN group that the tag defines.
Tag	This shows the user-specified tag that the Nebula Device adds to the outbound traffic on this port.
Storm Control	This shows whether traffic storm control is enabled or disabled on the port.
Broadcast Limit (pps)	This shows the maximum number of broadcast packets the Nebula Device accepts per second on this port.
Multicast Limit (pps)	This shows the maximum number of multicast packets the Nebula Device accepts per second on this port.
DLF Limit (pps)	This shows the maximum number of Destination Lookup Failure (DLF) packets the Nebula Device accepts per second on this port.
Loop Guard	This shows whether loop guard is enabled or disabled on the port.
Network Analytic Alert	An amber alert icon displays if the NCC generates alerts when an error or something abnormal is detected on the port for the IPTV network. Move the cursor over the alert icon to view the alert details.
IPSG protected	This shows whether IP source guard protection is enabled on this port.
Received CRC packets	This shows the number of CRC (Cyclic Redundancy Check) errors received on the port.
Number of IGMP Group	This shows the number of IGMP groups the port has joined.
Management control	This shows if management control is enabled on this port. See Table 64 on page 292 for more information.
	Click this icon to display a greater or lesser number of configuration fields.

Table 63 Site-wide > Configure > Switches > Switch ports (continued)

6.3.1.1 Update ports

Click to select the port you want to configure in the Site-wide > Configure > Switches > Switch ports screen.

Update 1 port						×
General settings						~
Switch ports	D8:EC:E5:60:99:8A/10					
Name	Port10	×	Bandwidth control	Enabled		•
Tags	None			Ingress 1000000	Kbps 🗙	
Port enabled	Enabled	•		Egress		
RSTP	Enabled	-		1000000	Kbps ×	
STP guard	None	•	Loop guard	Enabled		•
LLDP	Enabled		Storm control	Enabled		•
Link	Auto			Broadcast Limit (pps)	100	×
Port isolation	Enabled			Multicast Limit (pps)	100	×
Auth. policy	Open			DLF Limit (pps)	100	×
			Туре	Access		•
			Management control	Enabled		•
			VLAN type	None		-
			PVID	1		×
PoE settings						\sim
PTV setting Override adva	anced IGMP setting					``
Leave mode 🚺	Normal leave 🔻 4000	ms _×				
Maximum Group 🚺	Enabled 👻 1	×				
IGMP filtering profile	No Select	•				
Fixed router port	Auto	•				
					Clos	se Update

Figure 83 Site-wide > Configure > Switches > Switch ports: Edit

Tabla (1	Sita wida > Can	figuro > Switchor	> Switch	norte Edit
	SILE-WIDE - COL			DOUS. EQU
		0		

LABEL	DESCRIPTION
Switch ports	This shows the Nebula Device name and port number for the ports you are configuring in this screen.
Name	Enter a descriptive name for the ports.
Tags	Select or create a new tag for outgoing traffic on the ports.
Port enabled	Select to enable or disable the ports. A port must be enabled for data transmission to occur.
RSTP	Select to enable or disable RSTP on the ports.
STP guard	This field is available only when RSTP is enabled on the ports.
	Select Root guard to prevent the Nebula Devices attached to the ports from becoming the root bridge.
	Select BPDU guard to have the Nebula Device shut down the ports if there is any BPDU received on the ports.
	Otherwise, select None.
LLDP	Select to enable or disable LLDP on the ports.
Link	Select the speed and the duplex mode of the Ethernet connection on the ports. Choices are 10M/Half Duplex, 10M/Full Duplex, 100M/Full Duplex, 100M/Full Duplex, Auto, 10M/AN, and 100M/AN (Gigabit connections only).
Extended range	Select to enable or disable extended range.
	Extended range allows the port to transmit power and data at a distance of 250 meters.
	Note: When enabled, the port's PoE Power up mode is locked to 802.3at, and the port's link speed is limited to 10M/Full Duplex.
Media type	You can insert either an SFP+ transceiver or an SFP+ Direct Attach Copper (DAC) cable into the 10 Gigabit interface of the Nebula Device.
	Select the media type (SFP+or DAC 10G) of the SFP+ module that is attached to the 10 Gigabit interface.
Port Isolation	Select to enable or disable port isolation on the ports.
	The ports with port isolation enabled cannot communicate with each other. They can communicate only with the CPU management port of the same Nebula Device and the Nebula Device's other ports on which the isolation feature is not enabled.
IPSG protected	Select to enable or disable IP source guard protection on the port.
Auth. policy	This field is available only when you select Access in the Type field.
	Select the authentication policy type and name of the pre-configured authentication policy that you want to apply to the ports. See Table 73 on page 308 for more information on authentication policy type. See Section 6.3.6 on page 306 for more information on configuring authentication policy.
	Select Open if you do NOT want to enable port authentication on the ports.
Bandwidth Control	Select to enable or disable bandwidth control on the port.
Ingress	Specify the maximum bandwidth allowed in kilobits per second (Kbps) for the incoming traffic flow on the ports.
Egress	Specify the maximum bandwidth allowed in kilobits per second (Kbps) for the out-going traffic flow on the ports.

LABEL	DESCRIPTION
Loop guard	Select to enable or disable loop guard on the ports.
	Note: The loop guard feature cannot be enabled on the ports that have Spanning Tree Protocol (RSTP, MRSTP or MSTP) enabled.
Storm Control	Select to enable or disable broadcast storm control on the ports.
Broadcast Limit (pps)	Specifies the maximum number of broadcast packets the Nebula Device accepts per second on the ports.
Multicast Limit (pps)	Specifies the maximum number of multicast packets the Nebula Device accepts per second on the ports.
DLF Limit (pps)	Specifies the maximum number of DLF packets the Nebula Device accepts per second on the ports.
Туре	Set the type of the port.
	Select Access to configure the port as an access port which can carry traffic for just one VLAN. Frames received on the port are tagged with the port VLAN ID.
	Select Trunk to configure the port as a trunk port which can carry traffic for multiple VLANs over a link. A trunk port is always connected to a Nebula Device or router.
Management control	Select Enabled to configure the port as a management port. The default is Enabled . This allows the administrator to set the Nebula Device ports through which the device management VLAN traffic is allowed.
	Note: Make sure to enable this for an uplink port to maintain connection with Nebula.
VLAN type	This field is available only when you select Access in the Type field.
	None: This port is a regular access port and follows the device's access port rules.
	Vendor ID based VLAN: Apply the Vendor ID based VLAN settings from Switch > Configure > Switch settings to this port.
	Voice VLAN: Apply the Voice VLAN settings from Site-wide > Configure > Switches > Switch settings to this port.
	Note: For details on configuring Vendor ID based VLAN and Voice VLAN settings, see Section 6.3.8 on page 310.
PVID	A PVID (Port VLAN ID or native VLAN) is a tag that adds to incoming untagged frames received on a port so that the frames are forwarded to the VLAN group that the tag defines.
	Enter a number between 1 and 4094 as the port VLAN ID.
Allowed VLANs	This field is available only when you select Trunk in the Type field.
	Specify the VLANs from which the traffic comes. You can then transmit or receive traffic on the ports. See Section 3.24 on page 136 for the steps in setting up dynamic VLAN with RADIUS. See Section 3.25 on page 138 for more information on monitoring dynamic VLANs using event logs.
PoE Settings	
PoE	Select Enabled to provide power to a PD connected to the ports.

Table 64	Site-wide >	Configure >	Switches >	Switch ports:	Edit (continued)

LABEL	DESCRIPTION
PoE schedule	This field is available only when you enable PoE.
	Select a pre-defined schedule (created using the Site-wide > Configure > Switches > PoE schedules screen) to control when the Nebula Device enables PoE to provide power on the ports.
	Note: You must select Unschedule in the PoE schedule field before you can disable PoE on the ports.
	If you enable PoE and select Unschedule , PoE is always enabled on the ports.
	Note: The Nebula Device will follow the PoE schedule even when the Nebula Device is not connected to NCC.
	Click Edit to go to Site-wide > Configure > Switches > PoE schedules screen to create a new PoE schedule.
PoE priority	When the total power requested by the PDs exceeds the total PoE power budget on the Nebula Device, you can set the PD priority to allow the Nebula Device to provide power to ports with higher priority.
	Select Low to set the Nebula Device to assign the remaining power to the port after all critical and medium priority ports are served.
	Select Medium to set the Nebula Device to assign the remaining power to the port after all critical priority ports are served.
	Select Critical to give the highest PD priority on the port.
Power up mode	Set how the Nebula Device provides power to a connected PD at power-up.
	802.3at – the Nebula Device supports the IEEE 802.3at High Power over Ethernet standard and can supply power of up to 30W per Ethernet port. IEEE 802.3at is also known as PoE+ or PoE Plus. An IEEE 802.3at compatible device is referred to as Type 2. Power Class 4 (High Power) can only be used by Type 2 devices. If the connected PD requires a Class 4 current when it is turned on, it will be powered up in this mode.
	802.3af – the Nebula Device follows the IEEE 802.3af Power over Ethernet standard to supply power to the connected PDs during power-up.
	Legacy – the Nebula Device can provide power to the connected PDs that require high inrush currents at power-up. Inrush current is the maximum, instantaneous input current drawn by the PD when first turned on.
	Pre-802.3at – the Nebula Device initially offers power on the port according to the IEEE 802.3af standard, and then switches to support the IEEE 802.3at standard within 75 milliseconds after a PD is connected to the port. Select this option if the Nebula Device is performing 2-event Layer-1 classification (PoE+ hardware classification) or the connected PD is NOT performing Layer 2 power classification using Link Layer Discovery Protocol (LLDP).
	Force 802.3at – the Nebula Device provides PD Wide Range Detection (WRD) with power of up to 33 W on the port without performing PoE classification. Select this if the connected PD does not comply with any PoE standard.
	802.3bt – the Nebula Device follows the IEEE 802.3bt standard to supply power of up to 60 W per Ethernet port to the connected PDs at power-up.
	Pre-802.3bt – the Nebula Device offers power on the port according to the IEEE 802.3bt standard. Select this if the connected PD was manufactured before the IEEE 802.3bt standard was implemented on September 2018, but requires power between 33 W and 60 W. IEEE 802.3bt is also known as PoE++ or PoE Plus Plus.

Table 64 Site-wide > Configure > Switches > Switch ports: Edit (continued)

LABEL	DESCRIPTION
Auto PD recovery	Select to enable or disable automatic PD recovery on the port.
	Automatic PD recovery allows the Nebula Device to restart a Powered Device (PD) connected to the port by turning the device on and off again.
Detecting mode	Select LLDP to have the Nebula Device passively monitor current status of the connected Powered Device (PD) by reading LLDP packets from the PD on the port.
	Select Ping to have the Nebula Device ping the IP address of the connected Powered Device (PD) through the designated port to test whether the PD is reachable or not.
Action	Set the action to take when the connected Powered Device (PD) has stopped responding.
	Select Reboot-Alarm to have the Nebula Device send an SNMP trap and generate a log message, and then turn off the power of the connected PD and turn it back on again to restart the PD.
	Select Alarm to have the Nebula Device send an SNMP trap and generate a log message.
Neighbor IP	Set the IPv4 address of the Powered Device (PD) connected to this port.
	Note: If Detecting Mode is set to Ping and the PD supports LLDP, the connected PD's IPv4 address to which the Nebula Device sends ping requests is displayed automatically.
Polling Interval	Specify the number of seconds the Nebula Device waits for a response before sending another ping request.
	For example, the Nebula Device will try to detect the PD status by performing ping requests every 20 seconds.
Polling Count	Specify how many times the Nebula Device resends a ping request before considering the PD unreachable.
Resume Polling interval (sec)	Specify the number of seconds the Nebula Device waits before monitoring the PD status again after it restarts the PD on the port.
PD Reboot Count	Specify how many times the Nebula Device attempts to restart the PD on the port.
	The PD Reboot Count resets if any of the following conditions are true:
	 The Nebula Device successfully pings the PD. You modify any Auto PD Recovery settings and apply them. The Nebula Device restarts.
Resume Power Interval (sec)	Specify the number of seconds the Nebula Device waits before supplying power to the connected PD again after it restarts the PD on the port.
IPTV Setting	
Overwrite advanced IGMP setting	Select ON to overwrite the port's advanced IGMP settings (configured in the Site-wide > Configure > Switches > Advanced IGMP screen) with the settings you configure in the fields below. Otherwise, select OFF.

LABEL	DESCRIPTION
Leave Mode	Select Immediate Leave to remove this port from the multicast tree immediately when an IGMP leave message is received on this port. Select this option if there is only one host connected to this port.
	Select Normal Leave or Fast Leave and enter an IGMP normal/fast leave timeout value to have the Nebula Device wait for an IGMP report before the leave timeout when an IGMP leave message is received on this port. You need to specify how many milliseconds the Nebula Device waits for an IGMP report before removing an IGMP snooping membership entry when an IGMP leave message is received on this port from a host.
	In Normal Leave mode, when the Nebula Device receives an IGMP leave message from a host on a port, it forwards the message to the multicast router. The multicast router then sends out an IGMP Group-Specific Query (GSQ) message to determine whether other hosts connected to the port should remain in the specific multicast group. The Nebula Device forwards the query message to all hosts connected to the port and waits for IGMP reports from hosts to update the forwarding table.
	In Fast Leave mode, right after receiving an IGMP leave message from a host on a port, the Nebula Device itself sends out an IGMP Group-Specific Query (GSQ) message to determine whether other hosts connected to the port should remain in the specific multicast group. This helps speed up the leave process.
Maximum Group	Select Enable and enter the maximum number of multicast groups this port is allowed to join. Once a port is registered in the specified number of multicast groups, any new IGMP join report received on this port will replace the earliest group entry in the multicast forwarding table.
	Otherwise, select Disable to turn off multicast group limits.
IGMP filtering profile	An IGMP filtering profile specifies a range of multicast groups that clients connected to the Nebula Device are able to join.
	Select the name of the IGMP filtering profile to use for this port. Otherwise, select No Select to remove restrictions and allow the port to join any multicast group.
Fixed router port	Select Auto to have the Nebula Device use the port as an IGMP query port if the port receives IGMP query packets. The Nebula Device forwards IGMP join or leave packets to an IGMP query port.
	Select Fixed to have the Nebula Device always use the port as an IGMP query port. This helps prevent IGMP network topology changes when query packet losses occur in the network.

Table 64 Site-wide > Configure > Switches > Switch ports: Edit (continued)

6.3.2 ACL

ACL lets you allow or block traffic going through the Nebula Devices according to the rule settings. Use this screen to configure ACL rules on the Nebula Devices.

Click Site-wide > Configure > Switches > ACL to access this screen.



Manaaen	nent rule	e What is thi	ie?					
nanagen	ione and	Wilde to en	<u>a:</u>					
To ensur implies '	re manag that traff	gement con fics to and fr	nectivity with N rom the listed	Nebula Control Center (NCC), IP Address s management IP address are permitted on	specified for manageme the devices.	nt rules are added	to the IP filtering list by default configurat	tion. This
Note: Se	curity po	olicy informa	ation are perm	itted on the devices at all time to ensure s	smooth network operatio	on.		
Vebula cor	ntrol cente	er IP address						
2.19.85.221								
Sustomiz	ation rul	les Model list						
Sustomiz	ation rul	les <u>Model list</u>	Protocol	6	Sauran ID	Designed month	Durbing the C	Destination II
Sustomiz	ation rul Enabled	les <u>Model list</u> Policy	Protocol	Source MAC	Source IP	Source port	Destination MAC	Destination IF
Customiz I ¢∲≁ 1	ation rul Enabled	Policy Allow –	Protocol Any -	Source MAC eg.00123400.00/00/ff.ff.ff.00.00.00 ×	Source IP e.g.:192168:1.0/24 × *	Source port	Destination MAC	Destination IF e.g.1921681.0
Dustomiz ¢∲≁ 1	ation rul Enabled	Policy Allow	Protocol Any -	Source MAC e.g.:00.12.34:00:00:00/ff:ff:ff:00:00:00 × * Any	Source IP e.g.:192:1681.0/24 × *	Source port	Destination MAC [e.g.:00.12.34:00:00:00/ff:ff:ff:00:00:00 ×]* Any	Destination IF e.g. 1921681.0 Any
Customiz ¢∲> 1	ation rul Enabled	Policy Allow -	Protocol Any V	Source MAC e.g.:00:12:34:00:00:00/ff:ff:ff:00:00:00 × * Any	Source IP e.g.:192.1681.0/24 × * Any	Source port any a f	Destinction MAC e.g.:00:12:34:00:00:00/ff:ff:ff:00:00:00 × * Any	Destination 1 e.g. 1921681 Any

LABEL	DESCRIPTION
Management rules	The NCC automatically creates rules to allow traffic from/to the Nebula Control Center IP addresses in the list.
Customization rules	
¢€	Click the icon of a rule and drag the rule up or down to change the order.
Enabled	Select the check box to turn on the rule. Otherwise, clear the check box to turn off the rule.
Policy	Select to allow or deny traffic that matches the filtering criteria in the rule.
Protocol	Select the type of IP protocol used to transport the traffic to which the rule is applied.
Source MAC	Enter the source MAC address of the packets that you want to filter.
Source IP	Enter the source IP address of the packets that you want to filter.
Source port	Enter the source port numbers that defines the traffic type.
Destination MAC	Enter the destination MAC address of the packets that you want to filter.
Destination IP	Enter the destination IP address of the packets that you want to filter.
Destination port	Enter the destination port numbers that defines the traffic type.
VLAN	Enter the ID number of the VLAN group to which the matched traffic belongs.
Description	Enter a descriptive name for the rule.
Delete	Click the delete icon to remove the rule.
Add	Click this button to create a new rule.

Table 65 Site-wide > Configure > Switches > ACL

6.3.3 IP & Routing

This screen enables you to create IP interfaces and static routes on Nebula Devices in the site. This allows you to do the following:

- Create IP interfaces on a L2 Nebula Device for management or monitoring services, such as IGMP querier, auto PD recovery, ping, and ONVIF discovery.
- Create multiple IP interface on a L3 Nebula Device to route across VLANs.

• Create an IP interface and static route to specify the next hop to a specific destination subnet.

Click Site-wide > Configure > Switches > IP & Routing to access this screen.

IP & Routing						
IP interface						
	Switch	Name	IP address	Subnet mask	VLAN ID	
	XMG1930-30HP_Agg	Name	192.168.123.255	255.255.255.0	123	2
	+ Add					
Static route						
	Switch	Name	Destination	Subnet mask	Next hop IP	
	XMG1930-30HP_Agg	Test Route	12.13.14.15	255.255.255.0	16.17.18.19	2
	+ Add					

Figure 85 Site-wide > Configure > Switches > IP & Routing

The following table describes the labels in this screen.

LABEL	DESCRIPTION
IP interface	
Switch	This shows the name of the Nebula Device.
Name	This shows the name of the interface (network) on the Nebula Device.
IP address	This shows the IP address of the interface (network).
Subnet mask	This shows the subnet mask of the interface (network).
2	Click this icon to modify the interface.
	Click this icon to delete the interface.
VLAN ID	This shows the ID number of the VLAN with which the interface (network) is associated.
+ Add	Click this button to create a new interface on a Nebula Device in the site.
Static route	
Switch	This shows the name of the Nebula Device.
Name	This shows the name of the static route.
Destination	This shows the destination IP address.
Subnet mask	This shows the IP subnet mask.
Next hop IP	This shows the IP address of the next-hop gateway or the interface through which the traffic is routed. The gateway is a router or Nebula Device on the same segment as your Security Appliance's interfaces. It helps forward packets to their destinations.
2	Click this icon to modify the static route.
t	Click this icon to delete the static route.
+ Add	Click this button to create a new static route on a Nebula Device in the site.

Table 66 Site-wide > Configure > Switches > IP & Routing

6.3.3.1 Add IP Interface

Click the + Add button on the Site-wide > Configure > Switches > IP & Routing > IP Interface screen to access this screen.

Interface edit	or	×
Switch	XMG1930-30HP_Aggregator -	
Name	×	
Interface IP	×*	
Subnet mask	×*	
VLAN	*	
		Close Create

Figure 86 Site-wide > Configure > Switches > IP & Routing > IP Interface > Add

Tabla 47	Sito wido >	Configuro	Switchos > IP	& Pouting		Intorfaco	PPV <
	sile-wide ≥	Conligure >	Switches > IP	& KOUIING -	> IF	menace	> AUU

LABEL	DESCRIPTION
Switch	Select a Nebula Device in the site on which to create the interface.
Name	Enter a name of the interface (network) on the Nebula Device.
IP address	Inter the IP address of the interface (network).
Subnet mask	Enter the subnet mask of the interface (network).
VLAN	Enter the ID number of the VLAN with which the interface (network) is associated.
Close	Click Close to exit this screen without saving.
Create	Click Create to save your changes and create the interface.

6.3.3.2 Add Static Route

Click the + Add button on the Site-wide > Configure > Switches > IP & Routing > Static Route screen to access this screen.



Static rout	e	×
Switch	XMG1930-30HP_Aggregator -	
Name	×	
Destination	×*	
Subnet mask	*	
Next hop IP	*	
	Close	Create

Table 68	Site-wide >	Configure >	Switches > IF	% Routing >	Static Route 2	> Add

LABEL	DESCRIPTION
Switch	Select a Nebula Device in the site on which to create the interface.
Name	Enter a descriptive name for this route.
Destination	Specifies the IP network address of the final destination.
Subnet mask	Enter the IP subnet mask.
Next hop IP	Enter the IP address of the next-hop gateway.
Close	Click Close to exit this screen without saving.
Create	Click Create to save your changes and create the static route.

6.3.4 ONVIF Discovery

IP-based security products use a specific protocol for communication. One of the most common protocols is ONVIF (Open Network Video Interface Forum). ONVIF is a standard interface for interoperability of IP-based security products. When ONVIF is enabled and configured on a Nebula Device, the Nebula Device can obtain information from connected ONVIF-compatible devices, such as a device's system name and IP address.

In NCC, you can configure ONVIF-compatible Nebula Devices (for example, G\$1350) in a site to discover ONVIF-compatible devices in one designated VLAN.

Note: ONVIF and UPnP are similar protocols and may conflict with each other. If NCC detects UPnP packets on the same network as ONVIF, then it will prompt you to automatically create an ACL rule that blocks UPnP traffic (UDP, port 1900).

UPnP packets have been detected on the IPTV network.
UPnP packets may interfere with IPTV traffic and cause pixilation. You can use IP Filtering to block UPnP packets. Update filter rule to drop UPnP traffic by destination address.

NCC User's Guide

6.3.4.1 Configuring ONVIF Discovery

Follow these steps to configure ONVIF discovery within a site.

- 1 Decide on the VLAN ID you want to use for ONVIF discovery within the site. This VLAN is the ONVIF discovery VLAN.
- 2 Go to Site-wide > Configure > Switches > IP & Routing. For each Nebula Device that you want to enable ONVIF discovery on, add an IP interface for the Nebula Device on the ONVIF discovery VLAN.
- 3 Go to Site-wide > Configure > Switches > ONVIF discovery. Enable ONVIF discovery, and then set ONVIF VLAN ID to the ID of your ONVIF discovery VLAN.
- 4 For each Nebula Device that you want to enable ONVIF discovery on, click + Add. Select the Nebula Device, and then enter the ports that you want to listen for ONVIF devices.

6.3.4.2 ONVIF Discovery Screen

Click Site-wide > Configure > Switches > ONVIF discovery to access this screen.

Figure 88 Site-wide > Configure > Switches > ONVIF discovery

ONVIF discovery				
ONVIF configuration Model list				
ONVIF discovery ()				
ONVIF VLAN ID ()	1	×		
Surveillance switch	Switch name	Port list	Description	Model
	1 GS1350-6HP	•	*	GS1350-6HP

LABEL	DESCRIPTION
Model list	Click this to view a list of Zyxel Nebula Device models that support ONVIF discovery.
ONVIF discovery	Enable this to allow ONVIF-compatible Nebula Devices in the site to send ONVIF packets to discover or scan for ONVIF-compatible IP-based security devices.
ONVIF VLAN ID	Enter the ID number of the VLAN to run ONVIF. You can enter multiple VLAN IDs separated by a comma (,). For example, enter "1,2" for VLAN IDs 1 and 2.
Switch name	Select the Nebula Device that you want to enable ONVIF discovery on.
Port list	Enter the port numbers to allow discovery of ONVIF-compatible devices. You can enter multiple ports separated by comma (,) or hyphen (-) without spaces. For example, enter "3-5" for ports 3, 4, and 5. Enter "3,5,7" for ports 3, 5, and 7.
Description	Enter a descriptive name for this Nebula Device.
Model	This shows the Nebula Device model.
1	Click this icon to delete the ONVIF configuration for the Nebula Device.
+ Add	Click this to configure ONVIF discovery on another Nebula Device in the site.

Table 69 Site-wide > Configure > Switches > ONVIF discovery

6.3.5 Advanced IGMP

A Nebula Device can passively snoop on IGMP packets transferred between IP multicast routers/Nebula Devices and IP multicast hosts to learn the IP multicast group membership. It checks IGMP packets passing through it, picks out the group registration information, and configures multi-casting accordingly. IGMP snooping allows the Nebula Device to learn multicast groups without you having to manually configure them.

The Nebula Device forwards multicast traffic destined for multicast groups (that it has learned from IGMP snooping or that you have manually configured) to ports that are members of that group. IGMP snooping generates no additional network traffic, allowing you to significantly reduce multicast traffic passing through your Nebula Device.

Use this screen to enable IGMP snooping on the Nebula Devices in the site, create IGMP filtering profiles and configure advanced IGMP snooping settings that apply to all ports on the Nebula Device for your IPTV network. Click **Site-wide** > **Configure** > **Switches** > **Advanced IGMP** to access this screen. You can make adjustments on a per-port basis using the **Site-wide** > **Configure** > **Switches** > **Switc**

Vancea IGMP	
IGMP snooping	
IGMP-snooping VLAN <u>Model list</u>	Auto-detect X User Assign VLANs.
Unknown multicast drop <u>Model list</u> Drop on VLAN	All ×
IGMP filtering profiles 🟮	0 IGMP filtering profiles
	(There are no IGMP filtering profiles for this site) + Add
IPTV topology setup	IGMP snooping • Role • Port settings IGMP topology tips
	Switch name IGMP snooping IGMP report proxy Role Port settings
	Switch name IGMP snooping IGMP report proxy Role Port settings XIMG1930-30HP Querier Advanced setup

Figure 89 Site-wide > Configure > Switches > Advanced IGMP

Tabla 70	Sito wido >	Configura	Switchors	AdvancedICMP
TUDIE / U	slie-wide >	Conligue /	2MICHE2 ~	Advanced IGMF

LABEL	DESCRIPTION
IGMP snooping	Select ON to enable and configure IGMP snooping settings on all Nebula Devices in the site. Select OFF to disable it.
IGMP-snooping VLAN	Select Auto-detect to have the Nebula Device learn multicast group membership information of any VLANs automatically.
	Select User Assigned VLANs and enter the VLAN IDs to have the Nebula Device only learn multicast group membership information of the VLANs that you specify.
	Click Model List to view a list of Zyxel Nebula Device models that do not support this feature.
	Note: The Nebula Device can perform IGMP snooping on up to 16 VLANs.
Unknown multicast drop	Specify the action to perform when the Nebula Device receives an unknown multicast frame. Select ON to discard the frames. Select OFF to send the frames to all ports.
	Click Model List to view a list of Zyxel Nebula Device models that do and do not support this feature.
Drop on VLAN	This allows you to define the VLANs in which unknown multicast packets can be dropped.
	Note: The Nebula Device can drop unknown multicast packets on up to 8 VLANs.
IGMP filtering profiles	An IGMP filtering profile specifies a range of multicast groups that clients connected to the Nebula Device are able to join.
	You can set the Nebula Device to filter the multicast group join reports on a per-port basis by configuring an IGMP filtering profile and associating a port to the profile.
2	Click the edit icon to change the profile settings. See Section 6.3.5.1 on page 304.
1	Click the remove icon to delete the profile.
+Add	Click this button to create a new profile. See Section 6.3.5.1 on page 304.
IPTV topology setup The following three bu administrator account	ttons are available only when there are multiple Nebula Devices in the site and your has full access to this screen.
IGMP snooping	Select the Nebula Devices you want to configure and click this button to turn on or off IGMP snooping on the selected Nebula Devices.
Role	Select the Nebula Devices you want to configure and click this button to change the IGMP role of the selected Nebula Devices.
Port settings	Select the Nebula Devices you want to configure and click this button to open the Port settings screen, where you can change IGMP leave mode and IGMP filtering profile for the ports on the selected Nebula Devices. See Section 6.3.5.2 on page 305.
IGMP topology tips	Click this to view information about configuring your network and device roles to optimize IPTV performance.
The following list shows	you the IGMP settings for each Nebula Device in the site.
Switch Name	This shows the name of the Nebula Device in the site.
IGMP snooping	Click this to enable IGMP snooping on the Nebula Device. See Section 6.3.5 on page 302 for more information on IGMP snooping.

LABEL	DESCRIPTION
IGMP report proxy	Click this to enable IGMP report proxy on the Nebula Device. An IGMP report is generated when monitoring multicast address or membership query.
	It is highly recommended to disable this in the following conditions:
	 When the Nebula Device is deployed in a Networked AV environment. A Networked AV environment is specifically designed to simplify configuration and management of the Nebula Device for AVoIP (Audio-Video over Internet Protocol) application. When the Nebula Device is connected to CPEs (customer premise equipment) that require a specific IPTV source. Some CPEs validate IPTVs based on the source IP and MAC address of their IGMP join request. IGMP report proxy trims down the amount of IGMP join packets and sends its own IGMP join request.
Role	This shows whether the Nebula Device is acting as an IGMP snooping querier, aggregation Nebula Device or access Nebula Device in the IPTV network.
Port settings	Click Advanced setup to open the Port settings screen, where you can change IGMP leave mode and IGMP filtering profile for the ports on the Nebula Device. See Section 6.3.5.2 on page 305.
The following fields disp	olay when the IGMP role of a Nebula Device is set to Querier .
VLAN	Enter the ID number of the VLAN on which the Nebula Device learns the multicast group membership.
Querier IP Interface	Enter the IP address of the Nebula Device interface in IGMP querier mode.
	The Nebula Device acts as an IGMP querier in that network/VLAN to periodically send out IGMP query packets with the interface IP address and update its multicast forwarding table.
Mask	Enter the subnet mask of the Nebula Device interface in IGMP querier mode.
	Click the remove icon to delete the rule.
Add	Click this button to create a new rule.

	· ·		- ·· ·		
Table 70	Site-wide >	Contigure >	·Switches>	Advanced IGMP	(continued)
1001070		Gormgoro	011101100	/ (a) an oo a ronn	

6.3.5.1 Add/Edit IGMP Filtering Profiles

Use this screen to create a new IGMP filtering profile or edit an existing profile. To access this screen, click the Add button or a profile's Edit button in the IGMP filtering profiles section of the Site-wide > Configure > Switches > Advanced IGMP screen.

Figure 90 Site-wide > Configure > Switches > Advanced IGMP: Add IGMP Filtering Profile

IGMP filter				×
Profile name	New Name	×	*	
Start IP addr	955	End IP address		
1	*		× *	
+ Add				
			Close Sa	

LABEL	DESCRIPTION
Profile name	Enter a descriptive name for this profile for identification purposes.
	This shows the index number of the rule.
Start IP address	Enter the starting multicast IP address for a range of multicast IP addresses that you want to belong to the IGMP filter profile.
End IP address	Enter the ending multicast IP address for a range of IP addresses that you want to belong to the IGMP filter profile.
	If you want to add a single multicast IP address, enter it in both the Start IP Address and End IP Address fields.
.	Click the remove icon to delete the rule.
+Add	Click this button to create a new rule in this profile.
Close	Click this button to exit this screen without saving.
Save & Back	Click this button to save your changes and close the screen.

Table 71 Site-wide > Configure > Switches > Advanced IGMP: Add/Edit IGMP Filtering Profile

6.3.5.2 IGMP Port Settings

Use this screen to modify the IGMP snooping settings, such as IGMP leave mode and filtering profile for all ports on the Nebula Device. To access this screen, select one or more Nebula Devices and click the **Port settings** button or click a Nebula Device's **Advanced setup** button in the **IPTV topology setup** section of the **Site-wide > Configure > Switches > Advanced IGMP** screen.

Figure 91 Site-wide > Configure > Switches > Advanced IGMP: Port settings

Port settings				×
Switch name	XS3800-1-1			
Role	Querier			
Leave mode	Normal leave	▼ 20000	*	
Maximum group	Enable	▼ 1	*	
IGMP filtering profile	No select	•		
				Close Save

Table 72	Site-wide >	Configure >	Switches >	Advanced	IGMP: Port set	tings
----------	-------------	-------------	------------	----------	----------------	-------

LABEL	DESCRIPTION
Switch name	This shows the name of the Nebula Devices that you select to configure.
Role	This shows whether the Nebula Devices you selected is an IGMP snooping querier, aggregation Nebula Device or access Nebula Device in the IPTV network.

LABEL	DESCRIPTION
Leave mode	Select Immediate Leave to set the Nebula Device to remove this port from the multicast tree immediately when an IGMP leave message is received on this port. Select this option if there is only one host connected to this port.
	Select Normal Leave or Fast Leave and enter an IGMP normal/fast leave timeout value to have the Nebula Device wait for an IGMP report before the leave timeout when an IGMP leave message is received on this port. You need to specify how many milliseconds the Nebula Device waits for an IGMP report before removing an IGMP snooping membership entry when an IGMP leave message is received on this port from a host.
	In Normal Leave mode, when the Nebula Device receives an IGMP leave message from a host on a port, it forwards the message to the multicast router. The multicast router then sends out an IGMP Group-Specific Query (GSQ) message to determine whether other hosts connected to the port should remain in the specific multicast group. The Nebula Device forwards the query message to all hosts connected to the port and waits for IGMP reports from hosts to update the forwarding table.
	In Fast Leave mode, right after receiving an IGMP leave message from a host on a port, the Nebula Device itself sends out an IGMP Group-Specific Query (GSQ) message to determine whether other hosts connected to the port should remain in the specific multicast group. This helps speed up the leave process.
Maximum group	Select Enable and enter the maximum number of multicast groups this port is allowed to join. Once a port is registered in the specified number of multicast groups, any new IGMP join report received on this port will replace the earliest group entry in the multicast forwarding table.
	Otherwise, select Disable to turn off multicast group limits.
IGMP filtering profile	An IGMP filtering profile specifies a range of multicast groups that clients connected to the Nebula Device are able to join.
	Select the name of the IGMP filtering profile to use for this port. Otherwise, select No Select to remove restrictions and allow the port to join any multicast group.
Reset	Click this button to return the screen to its last-saved settings.
Close	Click this button to exit this screen without saving.
Save	Click this button to save your changes and close the screen.

 Table 72
 Site-wide > Configure > Switches > Advanced IGMP: Port settings (continued)

6.3.6 Authentication

Use this screen to configure authentication servers and policies to validate access to ports on the Nebula Device using the Nebula cloud authentication server or an external RADIUS server.

Note: Network traffic from clients will be denied when the Nebula cloud authentication server (NCAS) cannot be reached.



The following figure shows an example Nebula Device with ports enabled for MAC authentication. Clients 1 and 2 (C1, C2) passes MAC authentication (authorized). Client 3 (C3) fails MAC authentication (not authorized).



Click **Site-wide > Configure > Switches > Authentication** to access this screen.



uthentication Server		
Server type: 🚺	External radius server 👻	
	Please make sure you have created the corresponding VLAN setting in	the switches before enabling VLAN assignment in the external radius se
	Host Port Secret	
	= 1 x * x*	×)* 💼
	+ Add	
uthentication policy		
Password for MAC-Base Auth:	*	
	Nama Aith tuna Cuart /	I AN Dertsequity MAC limitation Auth parts
	1 Test	
	MAC-Buse *	

Table 73	Site-wide >	Configure > Switc	hes > Authentication
1001070		ooringere vinie	

LABEL	DESCRIPTION
Authentication Server	
Server type	Select External radius server to have both IEEE 802.1x (WPA-Enterprise) authentication and MAC-based authentication. The Nebula Device sends a request message to a RADIUS server in order to authenticate clients. The administrator must enter the IP address of the RADIUS server. The default port is 1812.
	Note: Make sure to configure VLAN for the Nebula Device before enabling VLAN assignment in the external RADIUS server.
	Select Nebula cloud authentication to have MAC-based authentication only. The Nebula Device sends HTTPS message to NCAS (Nebula Cloud Authentication Server) to authenticate clients. The default port is 443. See Section 3.23 on page 135 for the steps in setting up MAC authentication with NCAS.
	Blocked clients do not appear in the Nebula Device MAC address table. The Nebula Device re-authenticates blocked clients when:
	5 minutes after blocked client failed authenticationBlocked client disconnects and reconnects to the Nebula Device port.
	Note: The Blocked client in the Site-wide > Clients > Client list screen has a higher priority than MAC-based authentication. All network traffic from clients will be denied when the NCAS cannot be reached.
The following fields app	pear when you select External radius server as the Server type.
<ç̂→	Click the icon of a rule and drag the rule up or down to change the order.
Host	Enter the IP address of the external RADIUS server.
Port	Enter the port of the RADIUS server for authentication (default 1812).
Secret	Enter a password (up to 32 alphanumeric characters) as the key to be shared between the external RADIUS server and the Nebula Device.
1	Click the remove icon to delete the entry.
Add	Click this button to create a new RADIUS server entry.
Authentication policy	You apply the policy to a port in Site-wide > Configure > Switches > Switch ports : Edit (a selected port).
Password for MAC- Base Auth	Enter the password the Nebula Device sends along with the MAC address of a client for authentication with the RADIUS server. You can enter up to 32 printable ASCII characters.
Name	Enter a descriptive name for the policy.
Auth. type	Select MAC-Base if you want to validate access to the ports based on the MAC address and password of the client.
	Select 802.1X if you want to validate access to the ports based on the user name and password provided by the client.
	Note: 802.1X is not supported when you select Nebula cloud authentication in Server type .
Guest VLAN	A guest VLAN is a pre-configured VLAN on the Nebula Device that allows non- authenticated users to access limited network resources through the Nebula Device.
	Enter the number that identifies the guest VLAN.
Port security	Click On to enable port security on the ports. Otherwise, select Off to disable port security on the ports.

LABEL	DESCRIPTION
MAC limitation	This field is configurable only when you enable port security.
	Specify the maximum number of MAC addresses that may be learned on a port.
Auth. ports	This shows the number of the Nebula Device ports to which this policy is applied.
1	Click the remove icon to delete the profile.
Add	Click this button to create a new policy.

Table 73 Site-wide > Configure > Switches > Authentication (continued)

6.3.7 PoE Schedules

Use this screen to view and configure Power over Ethernet (PoE) schedules which can be applied to the ports. PoE is enabled at the specified time/date. Click **Site-wide > Configure > Switches > PoE schedules** to access this screen.

Note: The NCC will not generate an alert when PoE is disabled and the connected APs go offline because of the pre-defined PoE schedules.

The table shows the name of the existing schedules and the number of ports to which a schedule is applied. Click a schedule's edit icon to modify the schedule settings or click the **Add** button to create a new schedule. See Section 6.3.7.1 on page 309.

Figure 95	Site-wide > Configure > Switches > PoE schedules	
		-

PoE schedules		
PoE Schedule	Local time zone: Asia - Taipei (You can set this on $\ \underline{\mbox{General settings}}$)	3 PoE Schedule
	New Schedule used by () port(s)	2
	Always ON used by (0) port(s)	2
	Always OFF used by () port(s)	2
	+ Add Each site can have at most 5 PoE schedules	

6.3.7.1 Create new schedule

Click the Add button in the Site-wide > Configure > Switches > PoE schedules screen to access this screen.

New Sch	edule		×			Schedu	le temp	lates		Custo	m scheo	lule	•
Day A	Availability												
Sunday	00:00	02:00	04:00	06:00	08:00	10:00	12:00	14:00	16:00	18:00	20:00	22:00	24:00
Monday	00:00	02:00	04:00	06:00	08:00	10:00	12:00	14:00	16:00	18:00	20:00	22:00	24:00
Tuesday	00:00	02:00	04:00	06:00	08:00	10:00	12:00	14:00	16:00	18:00	20:00	22:00	24:00
Wednesday	00:00	02:00	04:00	06:00	08:00	10:00	12:00	14:00	16:00	18:00	20:00	22:00	24:00
Thursday	00:00	02:00	04:00	06:00	08:00	10:00	12:00	14:00	16:00	18:00	20:00	22:00	24:00
Friday	00:00	02:00	04:00	06:00	08:00	10:00	12:00	14:00	16:00	18:00	20:00	22:00	24:00
Saturday	00:00	02:00	04:00	06:00	08:00	10:00	12:00	14:00	16:00	18:00	20:00	22:00	24:00

Figure 96 Site-wide > Configure > Switches > PoE schedule: Add

The following table describes the labels in this screen.

LABEL	DESCRIPTION
Name	Enter a descriptive name for this schedule for identification purposes.
Schedule templates	Select a pre-defined schedule template or select Custom schedule and manually configure the day and time at which PoE is enabled.
Day	This shows the day of the week.
Availability	Click On to enable PoE at the specified time on this day. Otherwise, select Off to turn PoE off on the day and at the specified time.
	Specify the hour and minute when the schedule begins and ends each day.
Close	Click this button to exit this screen without saving.
Add	Click this button to save your changes and close the screen.

6.3.8 Switch Settings

Use this screen to configure global Nebula Device settings, such as (R)STP, QoS, port mirroring, voice VLAN, DHCP server guard, and IP source guard.

Click Site-wide > Configure > Switches > Switch settings to access this screen.

e-wide > Configure > Switches > <u>Switch settings</u> vitch settings	
Auto configuration recovery Model list	sta
Auto configuration recovery 🚯	
VLAN configuration	
	X Before changing management VLAN, please check that uplink port enable management control and belongs to management VLAN member for avoiding disconnect with NCC. To configure management control port in <u>Switch ports</u> .
STP configuration	
Rapid spanning tree protocol (RSTP):	
STP bridge priority: 0	Switches Bridge priority Default 32768 + Set the bridge priority for another switch
Quality of service	
Quality of service:	VLAN Priority Description
	+ Add What is this?
Port mirroring	
Port mirroring:	Switch Destination Port Source Port 1 × * × * <l< td=""></l<>
Err-disable recovery Model list	
Err-disable recovery	Recovery type Recover Expiration time (seconds)
	BPDU guard 300 ×
Voice VLAN	
Voice VLAN () Voice VLAN ID:	
Priority:	1
Assign VLAN by:	
OUE	OUI Description
	1 ×* ×* 💼

Figure 97	Site-wide > Configure	> Switches >	Switch settings
-----------	-----------------------	--------------	-----------------

Vendor ID based VLAN	
Vendor ID based VLAN Model list	
	Vendor OUI VLAN Priority Description
	$= 1$ x^* x^* x
	+ Add Vendor-ID on this network
Access management	
Access management Model list	
Allow IP range 🚺	Start IP End IP address address
	Default Deny all
	+ Add allow IP range
DHCP Server Guard	
DHCP Server Guard: 1	
IP source guard Model list	
IP source guard	
Protected switch	
	address will need to be inserted to "Permitted client entry", others need to renew their DHCP-IP
	adaress to successfully access the network.
	Switch name IP source guard Protected ports Client table
	XGS220-30HP_Test Null 👩 🕨 Run
Allowed client list 🚺	Action Q, IP, MAC, VLAN
	VLAN
	✓ 172.21.40.6 00.19:CB:00:00.01 1

Table 75	Site-wide >	Configure	> Switches >	Switch settings
----------	-------------	-----------	--------------	-----------------

LABEL	DESCRIPTION
Auto configuration red	covery
Auto configuration recovery	 When On, connectivity check to NCC is done 5 minutes after any configuration change. If an NCC connection problem is detected, the Nebula Device will return to its last saved custom default configuration. The Nebula Device will be locked by NCC and the banner N Switches are currently protected by Auto Configuration Recovery will be displayed. Otherwise, the latest configuration will be saved as the new custom default configuration. Note: If the NCC connectivity error occur 5 minutes after a configuration change, the Nebula Device will not return to its last saved configuration. Note: When Auto configuration recovery is turned Off, a pop-up message appears informing you that the locked Nebula Device(s) will be unlocked. Click Confirm if you wish to continue.
VLAN configuration	·

LABEL	DESCRIPTION
Management VLAN	Enter the VLAN identification number associated with the Nebula Device IP address. This is the VLAN ID of the CPU and is used for management only. The default is "1". All ports, by default, are fixed members of this "management VLAN" in order to manage the device from any port. If a port is not a member of this VLAN, then users on that port cannot access the device. To access the Nebula Device make sure the port that you are connected to is a member of Management VLAN.
	Before changing the management VLAN for an uplink port, check the following to avoid disconnection with NCC:
	 Management Control is enabled in Site-wide > Configure > Switches > Switch ports The uplink port belongs to the management VLAN in Site-wide > Configure > Switches > Switch ports: PVID.
STP configuration	
Rapid spanning tree protocol (RSTP)	Select On to enable RSTP on the Nebula Device. Otherwise, select Off .
STP bridge priority	Bridge priority is used in determining the root Nebula Device, root port and designated port. The Nebula Device with the highest priority (lowest numeric value) becomes the STP root Nebula Device. If all Nebula Devices have the same priority, the Nebula Device with the lowest MAC address will then become the root Nebula Device.
	The lower the numeric value you assign, the higher the priority for this bridge.
	Click Set the bridge priority for another switch to create a new entry. Select the Nebula Devices for which you want to configure the bridge priority, and select a value from the drop-down list box.
Quality of service	
Quality of service	Enter a VLAN ID and select the priority level that the Nebula Device assigns to frames belonging to this VLAN. Enter a descriptive name for the QoS (Quality of Service).
	Click Add to create a new entry.
Port mirroring	
Port mirroring	Click Add to create a new entry.
	Select the Nebula Device for which you want to configure port mirroring, specify the destination port you copy the traffic to in order to examine it in more detail without interfering with the traffic flow on the original ports, and also enter the source port on which you mirror the traffic.
Err-disable recovery	
Err-disable recovery	Enter the number of seconds (from 30 to 86400) to wait to activate a port or allow specific packets on a port, after the loop guard / BPDU guard error was gone.
	The loop guard feature shuts down a port if it detects that packets sent out on that port loop back to the Nebula Device.
	The BPDU guard feature allows you to prevent any new STP-aware (Spanning Tree Protocol) switch from connecting to an existing network and causing STP topology changes in the network. If there is any BPDU detected on the ports on which BPDU guard is enabled, the Nebula Device disables the ports automatically.
	 Loop guard recovery is always enabled. Click the switch to enable BPDU guard recovery. Default setting is disabled. The range of Expiration time (seconds) for both Loop guard recovery and BPDU guard recovery is 30 to 86400.
Voice VLAN	
Voice VLAN	Select On to enable the Voice VLAN feature on the Nebula Device. Otherwise, select Off .
	It groups the voice traffic with defined priority into an assigned VLAN which enables the separation of voice and data traffic coming into the Nebula Device port.

Table 75 Site-wide > Configure > Switches > Switch settings (continued)

LABEL	DESCRIPTION
Voice VLAN ID	Enter a VLAN ID number.
Priority	Select the priority level of the Voice VLAN from 1 to 6.
Assign VLAN by	Select how the Nebula Device assigns ports connected to VoIP devices to the Voice VLAN.
	OUI (Organizationally Unique Identifier): The Nebula Device assigns a port connected to a VoIP device to the Voice VLAN if the connected device's OUI matches any OUI in the list.
	LLDP-MED : The Nebula Device assigns a port connected to a VoIP device to the voice VLAN if the connected device is identified as a VoIP device using the LLDP-MED protocol.
	Note: The connected device must support LLDP-MED and have LLDP-MED enabled.
OUI	This field appears when you select OUI in the Assign VLAN by field.
	Click Add OUI on this network to add an OUI and a description for the OUI.
	An Organizationally Unique Identifier identifies a manufacturer. Typically, a device's OUI is the first three octets of the device's MAC address.
	For example, if you have an IP phone from Company A with MAC address 00:0a:95:9d:68:16, you can enter OUI <i>00:0a:95</i> to match all devices from Company A.
DSCP	This field appears when you select LLDP-MED in the Assign VLAN by field.
	Enter the Differentiated Services Code Point (DSCP) value for traffic on the voice VLAN. The value is defined from 0 through 63, and 0 is the default.
Vendor ID based VLAN	N
Vendor ID based VLAN	Select On to enable the Vendor ID based VLAN feature on the Nebula Device. Otherwise, select Off .
	Click the Add Vendor-ID on this network button to define the vendor MAC address OUI, assign to which VLAN, and set the priority. Enter a descriptive name for the Vendor ID based VLAN. Enter up to 64 characters for this field including special characters inside the square quotes [$\sim!@#$ \$%^&*()_+{} :"<>=[]\;',/].
Access management	
Access management	Select On to enable the access management feature on the Nebula Device. Otherwise, select Off .
Allow IP range	Click the Add allow IP range button to set the connected devices' starting and ending IP addresses that will be allowed to access the Nebula Devices through telnet, SSH, HTTP, HTTPS, and FTP.
DHCP Server Guard	
DHCP Server Guard	Select On to enable the DHCP server guard feature on the Nebula Device in order to prevent illegal DHCP servers. Only the first DHCP server that assigned the Nebula Device IP address is allowed to assign IP addresses to devices in this management VLAN.
	Otherwise, select Off to disable it.
IP source guard	
IP source guard	Select On to enable IP source guard protection. IP source guard uses a binding table to distinguish between authorized and unauthorized DHCP and ARP packets in your network. When the client does not exist in the binding table, the client is unauthorized and traffic will be blocked.
	To successfully access the network:
	 Client with static IP address will need to be added to the Allowed client list Client with dynamic IP address will need to get their IP address from an authorized DHCP server.

Table 75 Site-wide > Configure > Switches > Switch settings (continued)

LABEL	DESCRIPTION
Protected switch	This shows the Nebula Device(s).
	 Select On to enable IP source guard protection on the Nebula Device. Then click Save. Click the edit icon to go to Site-wide > Configure > Switches > Switch ports to configure Protected ports (see Section 6.3.1 on page 288 for more information). Click Run to display a pop-up window showing the current client table. Select the DHCP-snooping or Block entries and click Transfer to add these to the allowed client list. Then click Save.
Allowed client list	This allows the administrator to define a set of clients. Click Add client to define the IPv4 address , MAC address , and VLAN of the static client. A previous entry will be overwritten when you enter a duplicate MAC address and VLAN ID.
	Click Actions > Edit to modify the static client entry. Then click Update . The MAC address and VLAN ID will appear in red when you enter a duplicate entry.
	Click Actions > Delete to remove the static client entry.
	Click Save to activate the settings.
	Note: Maximum of 128 static entries is allowed per site.

Table 75 Site-wide > Configure > Switches > Switch settings (continued)

CHAPTER 7 Security Router

7.1 Overview

This chapter discusses the menus that you can use to monitor the Nebula-managed Security Routers in your network and configure settings even before a Security Router is deployed and added to the site.

A Nebula Security Router is a router firewall that can be managed by Nebula. It is referred to as a Nebula Device in this chapter.

7.2 Monitor

Use the **Monitor** menus to check the Nebula Device information, client information, event log messages and threat report for the Nebula Device in the selected site.

7.2.1 Event Log

Use this screen to view Nebula Device log messages. You can enter a key word, select one or multiple event types, or specify a date/time or a time range to display only the log messages that match these criteria.

Select **Range** to set a time range or select **Before** to choose a specific date/time and the number of hours/minutes to display only the log messages generated within a certain period of time (before the specified date/time). Then click **Search** to update the list of logs based on the search criteria. The maximum allowable time range is 30 days.

Click Site-wide > Monitor > Security router > Event log to access this screen.

vent log											
Keyword:		Category:									
Any	×	Any		Before 🔻	2023-03-30		11:26	• 1h	▼ UTC+8	\otimes	Q Search
/ Newar Older >	Event logs										C Export
Citati Videi 7	Litericity .		No matching events	s found between	2023-03-30 10:26 and 20	23-03-30 11:26	5			_	Carbone
Time C	ategory Source IP	Destination IP	Detail								E

Figure 98 Site-wide > Monitor > Security router > Event log

7.2.2 Threat Report

Use this screen to view statistics for threat management categories. Click **Site-wide** > **Monitor** > **Security** router > Threat report to access this screen.



Figure 99 Site-wide > Monitor > Security router > Threat report

LABEL	DESCRIPTION
Threat report	Select to view the report for the past day, week or month. Alternatively, select Custom range to specify a time the report will span. You can also select the number of results you want to view in a table. Then, click Update .
Email report	Click this button to send threat reports by email, change the logo and set email schedules.
Location	This shows the location on the map where the blocked threats occurred by category.
Threat Management b	by category
Category	This shows the name of the category to which the threat belongs.
	Click the Category to display the individual category statistics table. The category statistics table includes the following fields:
	Domain/Website/IP
	• Hits
	• %Hits
Hits	This shows the amount of hits on a specific threat category.
% Hits	This shows the percentage of the hit counts for the threat encountered by a specific category.
Threat Management b	by client
Description	This shows the name of the client device who encountered a threat.
	Click the name to display the individual client statistics table. The client statistics table includes the following fields:
	Domain/Website/IP
	Category
	• Hits
	• %Hits
IPv4 address	This shows the IPv4 address of the client device who encountered a threat.
MAC address	This shows the MAC address of the client device who encountered a threat.
Hits	This shows the number of threat hits of the client device.
% Hits	This shows the percentage of the hit counts encountered by a specific client device.

Table 76 Site-wide > Monitor > Security router > Threat report

7.2.3 VPN Connections

Use this screen to view the status of site-to-site IPSec VPN connections.

Note: If the peer gateway is not a Nebula Device, go to the **Site-wide** > **Configure** > **Security** router > **Site-to-Site VPN** screen to view and configure a VPN rule. See Section 7.3.5 on page 332 for more information.

Click Site-wide > Monitor > Security router > VPN connections to access this screen.

Figure 100 Site-wide > Monitor > Security router > VPN connections

onnection stu	tus					
Configuration:		This security gateway	is exporting 4 subnet over	the VPN: 192.168.128.0/24, 19	2.168.2.0/24, 192.168.10.0/24, 19	2.168.100.0/24
Site connectivit	εy.					
Location	Subnet	Status	Inbound	Outbound	Tunnel Up Time	Last Heartbeat
Non-Nebula VP	N peers connectivity					
	or Minister Carlo					

LABEL	DESCRIPTION
5	Click this button to reload the data on this page.
Connection Status	
Configuration	This shows the number and address of the local networks behind the Nebula Device, on which the computers are allowed to use the VPN tunnel.
Site Connectivity	
Location	This shows the name of the site to which the Nebula peer gateway is assigned.
	Click the name to view the VPN usage and connectivity status screen.
Subnet	This shows the address of the local networks behind the Nebula peer gateway.
Status	This shows whether the VPN tunnel is connected or disconnected.
Last heartbeat	This shows the last date and time a heartbeat packet is sent to determine if the VPN tunnel is up or down.
Non-Nebula VPN peer	's connectivity
Location	This shows the name of the site to which the Non-Nebula peer gateway (Zyxel or non-Zyxel IPSec VPN gateway and Cloud VPN (Azure VPN or AWS VPN)) is assigned.
	Click the name to go to the Site-wide > Configure > Security router > Site-to-Site VPN screen, where you can modify the VPN settings.
Subnet	This shows the address of the local networks behind the Non-Nebula peer gateway.
Status	This shows whether the VPN tunnel is connected or disconnected.
Inbound	This shows the amount of traffic that has gone through the VPN tunnel from the Non-Nebula peer gateway to the Nebula Device since the VPN tunnel was established.
Outbound	This shows the amount of traffic that has gone through the VPN tunnel from the Nebula Device to the Non-Nebula peer gateway since the VPN tunnel was established.

Table 77 Site-wide > Monitor > Security router > VPN connections

LABEL	DESCRIPTION
Tunnel up time	This shows how many seconds the VPN tunnel has been active.
Last heartbeat	This shows the last date and time a heartbeat packet was sent to determine if the VPN tunnel is up or down.

Table 77 Site-wide > Monitor > Security router > VPN connections (continued)

7.3 Configure

Use the **Configure** menus to configure interface addressing, firewall, site-to-site VPN, captive portal, traffic shaping, authentication server and other gateway settings for the Nebula Device of the selected site.

Note: Only one Security Router is allowed per site.

7.3.1 Interface

Use this screen to configure network interfaces on the Nebula Device. An interface consists of a port group, a VLAN ID, and an IP address, plus other configuration settings.

To access this screen, click **Site-wide** > **Configure** > **Security router** > **Interface**.

nterface				
		WAN Interface		
iner.		contra de contra		
Name	IP address	Subnet mask	VLAN ID	
WAN				2
		LAN Interface		
Name	IP address	Subnet mask		
LAN	192.168.168.1	255.255.255.0		2
Static Route				
Destination	Subnet mask	Next hop interface Next hop IP	Description	
	*	× * wan -	×	×
₩				
1 Add				

Figure 101 Site-wide > Configure > Security router > Interface

Table 78 Site-wide > Configure > Security router > Interface

LABEL	DESCRIPTION
WAN Interface	
Name	This field is read-only.
IP address	This shows the IP address for this interface.

LABEL	DESCRIPTION						
Subnet mask	This shows the subnet mask of this interface in dot decimal notation. The subnet mask indicates what part of the IP address is the same for all computers in the network.						
VLAN ID	This shows the VLAN ID. This 12-bit number uniquely identifies each VLAN. Allowed values are 2 – 4094. (0, 1 and 4095 are reserved.)						
2	Click the edit icon to modify the interface.						
LAN Interface							
Name	This field is read-only if you are editing an existing LAN interface.						
	Specify a name for the interface.						
	The format of interface names is strict. Each name consists of $2 - 4$ letters (interface type), followed by a number (x). For most interfaces, x is limited by the maximum number of the type of interface. For VLAN interfaces, x is defined by the number you enter in the VLAN name field. For example, VLAN interfaces are vlan0, vlan1, vlan2, and so on.						
IP address	This is the IP address for this interface.						
Subnet mask	This is the subnet mask of this interface in dot decimal notation. The subnet mask indicates what part of the IP address is the same for all computers in the network.						
2	Click the edit icon to modify it.						
Static Route							
Destination	Enter the destination IP address.						
Subnet mask	Enter an IP subnet mask. The route applies to all IP addresses in the subnet.						
Next hop interface	Select the interface you want to send all traffic to.						
Next hop IP	Enter the IP address of the next-hop gateway.						
Description	This is the descriptive name of the static route.						
2	Click this icon to modify a static route.						
.	Click this icon to remove a static route.						
Add	Click this button to create a new static route.						

Table 78 Site-wide > Configure > Security router > Interface (continued)

7.3.1.1 WAN Interface Configuration

Click the Edit button in the WAN Interface section to open the Security router > Configure > Interface > WAN interface configuration screen.

Figure 102	Site wide >	Configuras	Coourity	routor >	Interfaces	N/ANI interface	aanfiguration
FIGULE IUZ	Sile-wide ≥	Connourez	secum	100 mer	menace -	2 WAN INTENDCE	CONTICUTATION
	00	001110					00111000

WAN interface configuration			×
Interface properties			
Interface name	wan		
SNAT			
VLAN			
VLAN ID	2	× (2 - 4094)	
Туре	DHCP	•	
	ADVANCED OPTIONS		
IGMP proxy			
		Cancel	Ok

LABEL	DESCRIPTION
Interface properties	
Interface name	Specify a name for the WAN interface.
SNAT	Select this to enable SNAT. When enabled, the Nebula Device rewrites the source address of packets being sent from this interface to the interface's IP address.
VLAN	Select On to enable the VLAN feature on the WAN interface. Otherwise, select Off.
VLAN ID	Enter the VLAN ID. This 12-bit number uniquely identifies each VLAN. Allowed values are 2 – 4094. (0, 1 and 4095 are reserved.)
Туре	Select the type of interface to create.
	DHCP: The interface will automatically get an IP address and other network settings from a DHCP server.
	Static: You must manually configure an IP address and other network settings for the interface.
	PPPoE : The interface will authenticate with an Internet Service Provider, and then automatically get an IP address from the ISP's DHCP server. You can use this type of interface to connect to a DSL modem.
	PPPoE with static IP : Assign a static IP address to the WAN interface and your WAN interface is getting an Internet connection from a PPPoE server.
IP address assignment	These fields are displayed if you select Static .
IP address	Enter the static IP address of this interface.
Subnet mask	Enter the subnet mask for this interface's IP address.
Default gateway	Enter the IP address of the Nebula Device through which this interface sends traffic.

 Table 79
 Site-wide > Configure > Security router > Interface > WAN interface configuration

LABEL	DESCRIPTION
First DNS server	Enter a DNS server's IP address.
	The Domain Name System (DNS) maps a domain name to an IP address and vice versa. The Nebula Device uses the first and second DNS servers, in that order to resolve domain names for VPN, DDNS and the time server. Leave the field blank if you do not want to configure DNS servers.
Second DNS server	Enter the IP address of another DNS server. This field is optional.
These fields are displayed if	you selected PPPoE or PPPoE with static IP.
Username	Enter the user name provided by your ISP. You can use up to 31 alphanumeric characters and the underscore. Spaces are not allowed.
Password	Enter the password provided by your ISP. You can use up to 64 alphanumeric characters and the underscore. Spaces are not allowed.
IP address assignment	
IP address	Enter the static IP address of this interface.
DNS server	Enter a DNS server's IP address.
	The Domain Name System (DNS) maps a domain name to an IP address and vice versa. The Nebula Device uses the first and second DNS servers, in that order to resolve domain names for VPN, DDNS and the time server. Leave the field blank if you do not want to configure DNS servers.
ADVANCED OPTIONS	
Connection trigger	Select when to have the Nebula Device establish the PPP connection.
	Auto connect – select this to not let the connection time out.
	On demand – select this to automatically bring up the connection when the Nebula Device receives packets destined for the Internet.
PPPoE passthrough	Select this to allow up to ten hosts on the LAN to use PPPoE client software on their computers to connect to the ISP through the Nebula Device. Each host can have a separate account and a public WAN IP address.
	PPPoE passthrough is an alternative to NAT for application where NAT is not appropriate.
	Disable PPPoE passthrough if you do not need to allow hosts on the LAN to use PPPoE client software on their computers to connect to the ISP.
IGMP proxy	Select this to allow the Nebula Device to act as an IGMP proxy for hosts connected on the IGMP downstream interface.
Cancel	Click Cancel to exit this screen without saving.
OK	Click OK to save your changes.

T-1-1- 70	Cita unital a s	Configuration of the second se	C	Late of a set a	VALANT STATES OF A STATE		(line
IODIE / Y	NITE-WIDE 2	> (ontiduire >	\sim Security router >	interrace >	WAN INTERIOCE	contiauration	ICONTINUED
		Connigoro	000011,100101	initionaco ,		coningoranon	

7.3.1.2 LAN Interface Configuration

Click the Add button or click the Edit button in the LAN interface section to open the Site-wide > Configure > Security router > Interface > LAN interface configuration screen.

N interface configuration	1					
Interface properties						
Interface name	LAN					
IP address assignment						
IP address	192.168.168.1		>	<		
Subnet mask	255.255.255.0		>	<		
DHCP setting						
DHCP	DHCP server		•	•		
IP pool start address	192.168.168.33		>	< Pool size	200	×
First DNS server	This Router			•		
Lease time	2	× 0	×	0 ×		
	days	hours(Op	tional) n	ninutes(Optional)		

E' 400	C11		C		1		
Figure 103	>ite-wide >	Configure >	Security r	outer > in	iterrace >	LAN INTERTACE	configuration

LABEL	DESCRIPTION			
Interface properties				
Interface name	Specify a name for the LAN interface.			
IP address assignment				
IP address	Enter the IP address for this interface.			
Subnet mask	Enter the subnet mask of this interface in dot decimal notation. The subnet mask indicates what part of the IP address is the same for all computers in the network.			
DHCP setting DHCP	Select what type of DHCP service the Nebula Device provides to the network. Choices are: None – the Nebula Device does not provide any DHCP services. There is already a DHCP server on the network. DHCP relay – the Nebula Device routes DHCP requests to one or more DHCP servers you specify. The DHCP servers may be on another network. DHCP server – the Nebula Device assigns IP addresses and provides subnet mask, gateway, and DNS server information to the network. The Nebula Device is the DHCP server for the network.			
This field appear if the Nebula Device is a DHCP Relay.				
DHCP server	Enter the IP address of a DHCP server for the network.			

Table 80 Site-wide > Configure > Security router > Interface > LAN interface configuration

NCC User's Guide

LABEL	DESCRIPTION						
These fields appear if the Ne	These fields appear if the Nebula Device is a DHCP Server.						
IP pool start address	Enter the IP address from which the Nebula Device begins allocating IP addresses. If you want to assign a static IP address to a specific computer, use the Static DHCP Table .						
	If this field is blank, the Pool Size must also be blank. In this case, the Nebula Device can assign every IP address allowed by the interface's IP address and subnet mask, except for the first address (network address), last address (broadcast address) and the interface's IP address.						
First DNS Server	Specify the IP addresses of up to two DNS servers for the DHCP clients to use. Use one of the following ways to specify these IP addresses.						
	Custom Defined – enter a static IP address.						
	From ISP – select the DNS server that another interface received from its DHCP server.						
	This Router – the DHCP clients use the IP address of this interface and the Nebula Device works as a DNS relay.						
Second DNS Server	Enter the IP address of another DNS server. This field is optional.						
Lease Time	Specify how long each computer can use the information (especially the IP address) before it has to request the information again.						
	days, hours, and minutes (Optional) – enter how long IP addresses are valid.						
Cancel	Click Cancel to exit this screen without saving.						
OK	Click OK to save your changes.						

	Cite wide >	Configuras	Coourity routors	Interferes >	I A N I interfered	a a refigue restince	
	M = W = V	• (()))())()()()())()()()()()()()()()()	• \earlier >	inienace 2			(CONTINUE(1)
10010-00		Coningoio	000001191001019	innonaco ,		coningoranon	

7.3.2 Threat Management

Use this screen to enable the threat management categories such as:

- Ransomware and malware prevention that protects LAN clients from accessing or downloading harmful web contents.
- Intrusion blocker that prevents personal data theft in your network.
- Dark Web blocker that prevents unauthorized access from TOR proxies to the LAN clients.
- Stop mail fraud and phishing that blocks access by your LAN clients to phishing websites and SPAM URLs.
- Ads blocker that prevents access to websites containing annoying advertisements with links to harmful programs.
- VPN proxy blocker that prevents LAN clients connected to the Nebula Device from sending personal data to a cybercriminal's VPN gateway.

You can also configure the following:

- Up to 50 exception list, using the Nebula Device connected client device's name or IP address
- Up to 50 allowed domain name list
- Up to 50 blocked domain name list.

Click Site-wide > Configure > Security router > Threat management to access this screen.
reat management		
Threat Management		
Ransomware / Malware Prevention		
Intrusion blocker		
Dark Web blocker		
Stop mail fraud & phishing		
Block Ads		
Block VPN Proxy		
xception list		
By Client	Enabled Client Description	
	1 🖸 👻	×
	+Add	
By IP Address	Enabled Direction Ip Address Description	
	1 🗹 Both 👻 🔍 🔍 19226831	×
	+Add	
ustom allowed/blocked domain 🜖		
Allowed Domain	Domain Description	
	1	×
	+Add	
Blocked Domain	Domain Description	
	1	×

	· ·		· · ·	
Figure 104	Site-wide >	Contigure >	> Security router >	• Threat management

LABEL	DESCRIPTION
Threat management	
Ransomware / Malware	Ransomware and malware prevention protects the LAN clients connected to the Nebula Device from accessing or downloading harmful web content. These contents may contain files that could harm your operating system and personal files.
	Click the switch to enable ransomware/malware protection on the Nebula Device.
Intrusion blocker	Intrusion blocker prevents cybercriminals from harming, spying, or stealing personal data in your network.
	Click the switch to enable intrusion blocker protection on the Nebula Device.
Dark Web blocker	The Dark Web is an anonymous network accessed by browsers such as TOR. The purpose of the Dark Web is to enable anonymous access to content and prevent the identification of both the request and destination. The dark web blocker prevents unauthorized access from TOR proxies to the LAN clients connected to the Nebula Device.
	Click the switch to enable dark web blocker protection on the browsers of LAN clients connected to the Nebula Device.
Stop mail fraud & phishing	Mail fraud and phishing sites protection blocks access by your LAN clients to phishing websites and spam URLs.
	Click the switch to enable mail fraud and phishing protection on the browsers of LAN clients connected to the Nebula Device.

LABEL	DESCRIPTION
Block Ads	Ad blocking or ad filtering prevents exposure to websites containing advertisements with links to harmful programs.
	Click the switch to enable ads blocker protection on the browsers of LAN clients connected to the Nebula Device.
Block VPN Proxy	VPN proxy blocker prevents the LAN clients connected to the Nebula Device from sending personal data to a cybercriminal's VPN gateway.
	Click the switch to enable VPN proxy blocker protection on the browsers of LAN clients connected to the Nebula Device.
Exception list	Both wired and WiFi LAN clients connected to the Nebula Device in this list will bypass the threat management category check.
	Note: A maximum of 50 entries can be added to the exception list.
By Client	Enabled – Select this option to turn on this client exception profile. This allows both wired and WiFi LAN clients connected to the Nebula Device to bypass the threat management category check.
	Select the Client from the drop-down list. See Section 4.5.0.1 on page 194 and Section 4.5.0.2 on page 196 for more information on WiFi and wired clients.
	Enter a Description of the allowed client. You can use alphanumeric and ()+/:=?!*#@ $_{\infty}$ -characters, and it can be up to 512 characters long.
1	Click this icon to remove the client exception profile.
Add	Click this to create a client exception profile.
By IP Address	Enabled – Select this option to turn on this IPv4 address exception profile. This allows the client with this IPv4 address to bypass the threat management category check.
	Direction – Select Both to allow incoming/outgoing packets to/from the Nebula Device that match this IPv4 address. Select Source to allow incoming packets to the Nebula Device that match this IPv4 address. Select Destination to allow outgoing packets from the Nebula Device that match this IPv4 address.
	Add the IP Address that the Nebula Device will allow incoming and/or outgoing packets.
	Enter a description of the allowed IPv4 address. The description can be up to 512 characters long.
1	Click this icon to remove the IPv4 address exception profile.
Add	Click this icon to create an IPv4 address exception profile.
Custom allowed/ blocked domain	Create a list of host names to allow access to, or block access to, regardless of their content rating.
	Note: A maximum of 50 entries can be added to the Allowed Domain and Blocked Domain lists.

 Table 81
 Site-wide > Configure > Security router > Threat management (continued)

LABEL	DESCRIPTION
Allowed Domain	If you want to access any site, regardless of their content rating, add them to this list.
	Domain – Enter the host name, such as www.good-site.com into this text field. Do not enter the complete URL of the site – that is, do not include "http://". All sub-domains are allowed. For example, entering "zyxel.com" also allows "www.zyxel.com", "partner.zyxel.com", "press.zyxel.com", and so on. You can also enter just a top level domain. For example, enter .com to allow all .com domains.
	Use up to 127 characters (0–9 a–z). The casing does not matter.
	Enter a Description of the allowed domain. You can use alphanumeric and ()+/ :=?!*#@\$_%- characters, and it can be up to 60 characters long.
	Click Add to create a domain name profile.
Blocked Domain	If you want to block specific sites, regardless of their content rating, add them to this list.
	Domain – Enter the host name, such as www.bad-site.com into this text field. Do not enter the complete URL of the site – that is, do not include "http://". All sub-domains are also blocked. For example, entering "bad-site.com" also blocks "www.badsite.com", "partner.bad-site.com", "press.bad-site.com", and so on. You can also enter just a top level domain. For example, enter .com to block all .com domains.
	Enter a Description of the blocked domain. You can use alphanumeric and ()+/ :=?!*#@\$_%- characters, and it can be up to 60 characters long.
	Click Add to create a domain name profile.

 Table 81
 Site-wide > Configure > Security router > Threat management (continued)

7.3.3 Traffic Management

Application management allows you to manage the use of various applications on the network. Content Filtering allows you to control access to specific web sites or web content.

Click Site-wide > Configure > Security router > Traffic management to access this screen. Use this screen to control application usage and configure content filtering.

Enabled	Client App	lication	Description	
	•			×
	All		Default profile	
Add				
om allow/block don	nain (This list is same as content !	liter page configure allow/block list)		
ow bomain		Domain	Description	
	1		×]*	>
	+	Add		
ick Domain		Domain	Description	
	1		×	X
	4	004		
ent filtering st URL	• E	nter a url to know website category	X Test	
ent filtering st URL Enabled Client (• El Block category Parental control	nter a url to know website category Descript	X Test	
ent filtering st URL Enabled Cliant I	• El Block category Parental control	nter a uri to know website category Descript	X Test	
ent filtering tt URL Enabled Client (Block category Parental control Category list Adult Topics	nter a uri to know website category Descript	Anonymizing Utilities	Art/Culture/Heritage
ent filtering It URL Enabled Client		nter a uri to know website category Descript	ion Anonymizing Utilities Business Content Server	Art/Culture/Heritage
ent filtering It URL		nter a url to know website category	Anonymizing Utilities Business Content Server Dating/Social Networking	Art/Culture/Heritage Controversial Opinions . Digital Postcards
ent filtering It URL Enabled Client (Alcohol Stopp://wki Consumer Protection Datage: Druge: Druge:	Anonymizing Utilities Business Content Berver Content Berver Duting/Social Networking Education/Reference Encoment Renters	Art/Culture/Heritage Chot Controversial Opinions Digital Postcords For Kine
ent filtering It URL Enabled Client (nter a uri to know website category Descript Alcohal Alcohal Blogs/Wiki Consumer Protection Dating/Personals Drugs Fashion/Beauty Gambling Gambling	X Test	Art/Culture/Heritage Chot Controversial Opinions Digital Postcards Hertainment Gr Kids Game/Cartaon Violence
ent filtering It URL Enabled Client I	Block category Parental control Category list Adult Topics Canputing/Internet Cult/Occult Discrimination Estreme Forum/Sulletin Boards Games	nter a uri to know website category Descript Alcohol Alcohol Sloge/Wiki Consumer Protection Dating/Personals Dating/Personals Fashion/Beauty General News	Anonymizing Utilities Business Contont Server Dating/Social Networking Education/Reference Finance/Banking Gambing Related Gambing Related	> Art/Culture/Heritage Chot Controversial Opinions Digital Postards Entertainment Fark Kids Game/Cartoao Violence Gruesome Content
ent filtering It URL Enabled Client (nter a url to know website category	X Test.	Art/Culture/Heritage Chat Controversial Opinions Digital Postcards Fictrainment For Kids Game/Cartoon Violence Gruesome Content Humor/Comics Information Poster Union
ent filtering tt URL Enabled Client (nter a url to know website category Descript Alcohol Alcohol Biogs/Wiki Consumer Protection Dating/Personals Dating/Personals Gambiling General News Historical Revisionism Incidental Nuclity Interractive Web Applications	X Test.	Art/Culture/Heritage Chat Controversial Opinions Digital Postcards Entertainment For Kids Game/Cartoan Violence Gruesame Content. Humor/Carnics Information Socurity New Information Socurity New Information Socurity New
ent filtering tt URL Enabled Client (nter a url to know website category Discript Alcohol Alcohol Blogs/Wiki Consumer Protection Dating/Perisanals Dating/Perisanals General News Historical Revisionism Incidental Nudity Interactive Web Applications Major Global Religions	Anonymizing Utilities Business Content Server Dating/Social Networking Education/Reference Finance/Banking Gambling Related Government/Military History Information Security Information Security Information Adding	Art/Culture/Heritage Chtat Controversial Opinions Digital Postcards Far Kids Garme/Cartons Violence Gruesome Content Humor/Comics Information Security New Information Security New Media Downloads
ent filtering tt URL		nter a url to know website category	X Test	Art/Culture/Heritage Chat Controversial Opinions Digital Postcards Gravesana Contart Gravesana Contart Gravesana Contart Humar/Cornics Information Security New Information Security New Media Dewnloads Media Dewnloads Gravesana
ent filtering It URL		nter a url to know website category	X Test	Art/Culture/Heritage Chat Controversial Opinions Digital Postards Entertainment Garmon/Cantoa Violence Garmo/Cantoa Violence Garmo/Cantes Information Security New Internet Services Media Downloads Media Downloads Media Downloads Personal Network Storene
ent filtering tt URL		nter a url to know website category	X Test	Art/Culture/Heritage Chat Chat Cantroversial Opinions Digital Postcards Futertainment Gr Kids Game/Cartaon Violence Gruesoms Content Humor/Camics Information Socurity New Internet Services Media Downloads Mederoted Online Shooping Pornography
ent filtering It URL	Black catagory Parental control Category list Adult Topics Adult Topics CaronyClassifieds CaronyCla	nter a url to know website category	X Test	Art/Culture/Heritage Chot Controversial Opinions Digital Postaarda Entertainment Gr Kids Gruesome Content Gumor/Cartica Violence Gruesome Content Humor/Cartics Information Security New Internet Services Moderated Online Schopping Personal Network Storage Pomography Potential Illegal Software
ent filtering It URL	Block category Parental control Category list Category list Adult Topics Cassifieds Carporuling/Internet Call/Cacult Estreme Forum/Bulletin Boards Games Health Elliegol UK Category List Media Sharing Dubb Snarch Media Sharing Parsonal Pages Portal Sites Portual Sites Portual Sites Portual Information	nter d url to know website category	X Test.	Art/Culture/Heritage Chat Controversial Opinions Digital Postcards Entertainment For Kids Gamw/Cartaan Violence Grussame Content Humor/Carnics Humor/Carnics Humor/Carnics Media Downloads Med
ent filtering truture inabled Client		nter a url to know website category	Anonymizing Utilities Businoss Content Server Dating/Social Networking Gambling Related Government/Millary History Information Security Information Security History Plance/Banking Mobile Phone Nudity Porked Domain Potental Hacking/Computer Crime Professional Networking Recreation/Nobbles Resciedtal Photolessa	Art/Culture/Heritage Chat Controversial Opinions Digital Postcards Far Kids Gamw/Carton Violence Gruesome Contont Humor/Comics Information Security New Internet Services Media Downloads Media Media Media M
ent filtering st URL Frobled Client (nter a url to know website category	X Test	Art/Culture/Heritage Chat: Controversial Opinions . Digital Postcards . Digital Postcards . Structure Context . Humor/Cornics Information Security New Internet Services Information Security New Internet Services Media Downloads . Media Context
ent filtering st URL Enabled Client (X Test	Art/Culture/Heritage Chat Cantroversial Opinions Digital Postards For Kids Game/Cartoan Violence Game/Cartoan
ent filtering et URL Enabled Client (Client (Clien			X Test	Art/Culture/Heritage Chit Controversial Opinions . Digital Postcards Entertainment Gruesome Content Humor/Camics Information Security New Internet Services Media Downloads Media Downloads Media Downloads Mederoted Online Shopping Porvocative Attire Provocative Attire Peligion/deology Resource Sharing Sexual Materials Sports Taxwal
ent filtering st URL		nter a url to know website category	X Test	Art/Culture/Heritage Chat Chat Controversial Opinions Digital Postcards Entertainment For Kids Game/Cartoan Violence Grussome Content Humor/Comics Humor/Comics Game/Cartoa Security New Informst Services Media Downloads Med

Figure 105 Site-wide > Configure > Security router > Traffic management

Table 82 Site-wide > Configure > Security router > Traffic management

LABEL	DESCRIPTION
Application managem	nent
Application identification &	Click this to enable the Nebula Device to control usage of applications for a client or all clients.
control	When disabled:
	 the Security router network applications widget in the Site-wide > Dashboard screen will show Application monitor disabled
	 the Site-wide > Applications usage screen will show Application identification is turned off.
Enabled	Select the check box to turn on the rule. Otherwise, clear the check box to turn off the rule.
Client	Select All or select a client to apply the rule.
Application	Select All or select an application to apply the rule.
Description	Enter a description for this profile. The description can be up to 512 characters long.
1	Click this icon to remove the entry.
Add	Click this button to create up to five application management profiles.
Custom allow/block d	omain
Allowed Domain	Sites that you want to allow access to, regardless of their content rating, can be allowed by adding them to this list.
	Domain – Enter host names such as www.good-site.com into this text field. Do not enter the complete URL of the site – that is, do not include "http://". All sub-domains are allowed. For example, entering "zyxel.com" also allows "www.zyxel.com", "partner.zyxel.com", "press.zyxel.com", and so on. You can also enter just a top level domain. For example, enter .com to allow all .com domains.
	Use up to 127 characters (0–9 a–z). The casing does not matter.
	Enter a Description of the allowed domain. The description can be up to 60 characters long.
	Click Add to create a domain name profile.
Blocked Domain	Sites that you want to block access to, regardless of their content rating, can be blocked by adding them to this list.
	Domain – Enter host names such as www.bad-site.com into this text field. Do not enter the complete URL of the site – that is, do not include "http://". All sub-domains are also blocked. For example, entering "bad-site.com" also blocks "www.badsite.com", "partner.bad-site.com", "press.bad-site.com", and so on. You can also enter just a top level domain. For example, enter .com to block all .com domains.
	Enter a Description of the blocked domain. The description can be up to 60 characters long.
	Click Add to create a domain name profile.
Content filtering	
Test URL	You can check which category a web page belongs to. Enter a web site URL in the text box.
	When the content filter is active, you should see the web page's category. The query fails if the content filter is not active.
	Content Filtering can query a category by full URL string (for example, http:// www.google.com/picture/index.htm), but HTTPS Domain Filter can only query a category by domain name ('www.google.com'), so the category may be different in the query result. URL to test displays both results in the test.

LABEL	DESCRIPTION
Enabled	Select the check box to turn on the rule. Otherwise, clear the check box to turn off the rule.
Client	Select All or select a client to apply the rule.
Block category	Select the block category. Choices are Parental control, Productivity and Custom.
Description	Enter a description for this profile. You can use alphanumeric and ()+/:=?!*#@\$_%- characters, and it can be up to 512 characters long.
Category list	Click to display or hide the category list.
	These are categories of web pages based on their content. Select categories in this section to control access to specific types of Internet content.
1	Click this icon to remove the entry.
Add	Click this button to create up to five application categories and set actions for specific applications within the category.

T. I. I. 00	C'I	C C C C C C C C C C	C	T	/ · · · · · · · · · · · · · · · · · · ·
1 able 82	site-wide >	Contigure >	> Security router >	Irattic management	(continued)

7.3.4 Firewall

By default, a LAN user can initiate a session from within the LAN and the Nebula Device allows the response. However, the Nebula Device blocks incoming traffic initiated from the WAN and destined for the LAN. Use this screen to configure firewall rules for outbound traffic.

In addition, this screen allows you to create new NAT rules and edit/delete existing NAT rules.

Note: When adding a NAT rule, based on the NAT setting NCC will automatically add the incoming security policy (firewall) rule.

Click Site-wide > Configure > Security router > Firewall to access this screen.

Note: The Nebula Device has the following hidden default firewall rules: LAN to WAN is allowed, WAN to LAN is blocked.

Figure 106	Site-wide > Configure > Security router > Firewall	
------------	--	--

ewall						
ondi						
Country Restriction						
Action	O Disable					
	O Allow					
	O Block					
Directions	Both					
Country						
Security policy						
Security policy	2 X V		_	12 m 14		
	Action	Protocol	Source	Destination	Dst Port	Description
= 1 🗹	× Allow •	Any 👻	Any 👻	Any 🔻	Any 💌	
Implicit allow rules 🔺						
	Allow	Any	lan_192.168.168.1	Any	Any	All
	Allow	Any	lan_192.168.168.1	Device	Any	All
Implicit deny rule						
	Deny	Any	Any	Any	Any	De
4						
+ Add						
Nat						
Enabled Protocol Public	Port	LAN IP	Local Port	Allow Remote	IP Description	
= 1 🔽 Both 🔻		×	×	× Any		× 💼
+ Add						

Table 83	Site-wide > Configure > Security router > Fire	wall

LABEL	DESCRIPTION					
Country Restriction	Country Restriction					
Action	Choose one of the following actions:					
	• Disable: Select this to hide the Country Restriction settings.					
	• Allow: Select this to allow packets from the selected countries IP address in the Country field. Dropping of packets from countries not in the Allow list will occur.					
	• Block: Select this to drop packets from the selected countries IP address in the Country field.					
Directions	Select Both to allow incoming/outgoing packets to apply the firewall rules. Select Incoming to apply the firewall rules on incoming packets. Select Outgoing to apply the firewall rules on outgoing packets.					
Country	Select up to 10 countries or regions to apply the firewall rules configured in this screen.					
Security policy						
¢€	Click the icon of a rule and drag the rule up or down to change the order.					
Enabled	Select the check box to turn on the rule. Otherwise, clear the check box to turn off the rule.					
Name	Enter the name of the security policy.					
Action	Select what the Nebula Device is to do with packets that match this rule.					
	Select Deny to silently discard the packets without sending a TCP reset packet or an ICMP destination-unreachable message to the sender.					
	Select Allow to permit the passage of the packets.					

NCC User's Guide

LABEL	DESCRIPTION
Protocol	Select the IP protocol to which this rule applies. Choices are: ICMP, TCP, UDP, TCP and UDP and Any.
Source	Specify the source IP addresses (LAN interface / country) to which this rule applies. You can add a CIDR, or enter a new IP address by clicking Customize IP . Enter Any to apply the rule to all IP addresses.
Destination	Specify the destination IP addresses (LAN interface / country) or subnet to which this rule applies. You can add a CIDR, or enter a new IP address by clicking Customize IP . Enter Any to apply the rule to all IP addresses.
Dst Port	Specify the destination ports to which this rule applies. By default, Any applies the rule to all ports.
Description	Enter a descriptive name of up to 60 printable ASCII characters for the rule.
.	Click this icon to remove the rule.
Implicit allow rules	This shows the system generated Allow rules.
	 LAN interface / remote access VPN to Any LAN interface / remote access VPN to Nebula Device
Implicit deny rule	This shows the system generated Deny rule.
	Any to Any
Add	Click this button to create a new rule.
NAT – Virtual server	
¢	Click the icon of a rule and drag the rule up or down to change the order.
Enabled	Select the check box to turn on the rule. Otherwise, clear the check box to turn off the rule.
Protocol	Select the IP protocol to which this rule applies. Choices are: TCP, UDP, and Both.
Public Port	Enter the translated destination port or range of translated destination ports if this NAT rule forwards the packet.
LAN IP	Specify to which translated destination IP address this NAT rule forwards packets.
Local Port	Enter the original destination port or range of destination ports this NAT rule supports.
Allow Remote IPs	Specify the remote IP addresses that are allowed to access the public IP address.
	Select Any to allow all IP addresses.
Description	Enter the descriptive name of the policy of up to 60 printable ASCII characters.
1	Click this icon to remove the profile.
Add	Click this button to create a new schedule profile.

Table 83	Site-wide >	Configure >	Security	v router >	Firewall	(continued)
10016-03		Coninguie /	26000	y 100161 -	I II E WUII I	(COLINITOEQ)

7.3.5 Site-to-Site VPN

A virtual private network (VPN) provides secure communications between sites without the expense of leased site-to-site lines. Use this screen to configure VPN rules.

Note: Site-to-site VPN does not support both VPN sites behind NAT mode.

Click Site-wide > Configure > Security router > Site-to-Site VPN to access this screen.

Outgoing interface	WAN				
Local networks	Nama	Subnet		Use VPN	
	lan	192168168.0/2	14		
VPN Area	Default				
Nebula VPN enable					
Nebula VPN topology	Split tunnel (send only	site-to-site traffic over the V	PN)		
	Site-to-Site				
Area communication					
NAT traversal	O None				
	Custom NAT	raversal			
	Remote VPN peer con	nect to this Nebula gateway (using the public IP address you specify.		
Remote VPN participants	Network		Subnet(s)		
on-Nebula VPN peers te-wide settings ations in this section apply to this Nebuli inabled Name	a Batewañ oujň brapije ib	Private subnet 🚦	IPsec Preshared secr policy Preshared secr	et Availability	Addres
			× * Default	This site O	
		201. <u></u>			
+ Add					
ra-wide settings					
n this page is view only, please change th	e configure by <u>VPN Orchestrator</u>	Page			

Figuro	107	Sita wida >	Configura	> Sacurity	routor >	sita to sita	
riquie	107	sile-wide ≥	Conilquie	> 26C0III)		2 2116-10-2116	

	Table 84	Site-wide >	Configure >	Security	/ router >	Site-to-Site	VPN
--	----------	-------------	-------------	----------	------------	--------------	-----

LABEL	DESCRIPTION
Outgoing Interface	This displays WAN as the interface to which the VPN connection is going.
Local network	This shows the local network behind the Nebula Device.
Name	This shows the network name.
Subnet	This shows the IP address and subnet mask of the computer on the network.
Use VPN	Select ON to allow the computers on the network to use the VPN tunnel. Otherwise, select OFF.
VPN Area	Select the VPN area of the site.
	For details, see Section 11.4.4.2 on page 524.
Nebula VPN enable	Click this to enable or disable site-to-site VPN on the site's Nebula Device.
	If you disable this setting, the site will leave the VPN area.
Nebula VPN Topology	Click this to select a topology for the VPN area. For details on topologies, see Section 11.4.4.1 on page 523.
	Select disable to disable VPN connections for all sites in the VPN area.
Area communication	Enable this to allow the site to communicate with sites in different VPN areas within the organization.

NCC User's Guide

LABEL	DESCRIPTION
NAT traversal	If the Nebula Device is behind a NAT router, select Custom to enter the public IP address or Auto or the domain name that is configured and mapped to the Nebula Device on the NAT router.
	Note: To allow a site-to-site VPN connection, the NAT router must have the following ports open: UDP 500, 4500.
Remote VPN participants	This shows all sites within the VPN area.
Non-Nebula VPN peers	Configure this section to add a non-Nebula gateway to the VPN area.
+ Add	Click this button to add a non-Nebula gateway to the VPN area.
Enabled	Select the check box to enable VPN connections to the non-Nebula gateway.
Name	Enter the name of the non-Nebula gateway.
Public IP	Enter the public IPv4 address or FQDN of the non-Nebula gateway.
Private subnet	Enter the IP subnet that will be used for VPN connections. The IP range must be reachable from other devices in the VPN area.
IPSec policy	Click to select a pre-defined policy or have a custom one. See Section 7.3.5.1 on page 334 for detailed information.
Preshared secret	Enter a pre-shared key (password). The Nebula Device and peer gateway use the key to identify each other when they negotiate the IKE SA.
Availability	Select which sites the non-Nebula gateway can connect to in the VPN area.
	Select All sites to allow the non-Nebula gateway to connect to any site in the VPN area.
	Select This site and the non-Nebula gateway can only connect to the Nebula Device in this site.
Address	Enter the address (physical location) of the device.
1	Click this icon to remove the non-Nebula gateway.
Add	Click this button to create a new non-Nebula gateway.

Table 84 Site-wide > Configure > Security router > Site-to-Site VPN (continued)

7.3.5.1 IPsec Policy

Click the **Default** button in the **Non-Nebula VPN peers** section of the **Site-wide** > **Configure** > **Security** router > **Site-to-Site VPN** screen to access this screen.

Custom				×
Preset		Default	•	^
Phase 1				
IKE version		IKEv1	•	
Encryption		AES128	•	
Authentication		SHA128	•	
Diffie-Hellman g	roup	DH2	•	
Lifetime (second	is)	86400	×	
Advanced				
Phase 2				
Set	Encrypti	on	Authentication	
Set 1	AES128	-	SHA128	•
Set 2	None	•	None	•
Set 3	None	•	None	•
PFS group		DH2	•	
Lifetime (second	(0)	DTIZ	-	
Liteume (second	15/	28800	×	
				Close OK

Figure 108 Site-wide > Configure > Security router > Site-to-Site VPN: IPsec Policy

Table 85	Site-wide >	Configure >	Security	v router >	Site-to-Site	VPN: IPsec Policy	1

LABEL	DESCRIPTION
Preset	Select a pre-defined IPSec policy, or select Custom to configure the policy settings yourself.
Phase1	IPSec VPN consists of two phases: Phase 1 (Authentication) and Phase 2 (Key Exchange).
	A phase 1 exchange establishes an IKE SA (Security Association).
IKE version	Select IKEv1 or IKEv2.
	IKEv1 and IKEv2 applies to IPv4 traffic only. IKE (Internet Key Exchange) is a protocol used in setting up security associations that allows two parties to send data securely.
Encryption	Select which key size and encryption algorithm to use in the IKE SA. Choices are:
	DES – a 56-bit key with the DES encryption algorithm
	3DES – a 168-bit key with the DES encryption algorithm
	AES128 – a 128-bit key with the AES encryption algorithm
	AES192 – a 192-bit key with the AES encryption algorithm
	AES256 – a 256-bit key with the AES encryption algorithm
	The Nebula Device and the remote IPSec router must use the same key size and encryption algorithm. Longer keys require more processing power, resulting in increased latency and decreased throughput.

NCC User's Guide

LABEL	DESCRIPTION
Authentication	Select which hash algorithm to use to authenticate packet data in the IKE SA.
	Choices are SHA128 , SHA256 , SHA512 and MD5 . SHA is generally considered stronger than MD5, but it is also slower.
	The remote IPSec router must use the same authentication algorithm.
Diffie-Hellman group	Select which Diffie-Hellman key group (DHx) you want to use for encryption keys. Choices are:
	DH1 – use a 768-bit random number Modular Exponential (MODP) DH group
	DH2 – use a 1024-bit random number MODP
	DH5 – use a 1536-bit random number MODP
	DH14 – use a 2048-bit random number MODP
	DH19 – use a 256-bit random number elliptic curve group
	DH20 – use a 384-bit random number elliptic curve group
	DH21 – use a 521-bit random number elliptic curve group
	DH28 – use a 256-bit random number elliptic curve group
	DH29 – use a 384-bit random number elliptic curve group
	DH30 – use a 512-bit random number elliptic curve group
	Both routers must use the same DH key group.
Lifetime (seconds)	Enter the maximum number of seconds the IKE SA can last. When this time has passed, the Nebula Device and remote IPSec router have to update the encryption and authentication keys and re-negotiate the IKE SA. This does not affect any existing IPSec SAs, however.
Advanced	Click this to display a greater or lesser number of configuration fields.
Mode	Set the negotiation mode.
	Main encrypts the Nebula Device's and remote IPSec router's identities but takes more time to establish the IKE SA.
	Aggressive is faster but does not encrypt the identities.
Local ID	Enter an identifier used to identify the Nebula Device during authentication.
	This can be an IP address or hostname.
Peer ID	Enter an identifier used to identify the remote IPSec router during authentication.
	This can be an IP address or hostname.
Phase2	Phase 2 uses the SA that was established in phase 1 to negotiate SAs for IPSec.

Table 85	Site-wide > Configure	> Security router > S	Site-to-Site VPN: IPsec Pc	licv (continued)
	one mae · comgere			

LABEL	DESCRIPTION
Encryption	Select which key size and encryption algorithm to use in the IPSec SA. Choices are:
	(None) – no encryption key or algorithm
	DES – a 56-bit key with the DES encryption algorithm
	3DES – a 168-bit key with the DES encryption algorithm
	AES128 – a 128-bit key with the AES encryption algorithm
	AES192 – a 192-bit key with the AES encryption algorithm
	AES256 – a 256-bit key with the AES encryption algorithm
	The Nebula Device and the remote IPSec router must both have at least one proposal that uses the same encryption and the same key.
	Longer keys are more secure, but require more processing power, resulting in increased latency and decreased throughput.
PFS group	Select whether or not you want to enable Perfect Forward Secrecy (PFS) and, if you do, which Diffie-Hellman key group to use for encryption. Choices are:
	None – disable PFS
	DH1 – use a 768-bit random number Modular Exponential (MODP) DH group
	DH2 – use a 1024-bit random number MODP
	DH5 – use a 1536-bit random number MODP
	DH14 – use a 2048-bit random number MODP
	DH19 – use a 256-bit random number elliptic curve group
	DH20 – use a 384-bit random number elliptic curve group
	DH21 – use a 521-bit random number elliptic curve group
	DH28 – use a 256-bit random number elliptic curve group
	DH29 – use a 384-bit random number elliptic curve group
	DH30 – use a 512-bit random number elliptic curve group
	PFS changes the root key that is used to generate encryption keys for each IPSec SA. Both routers must use the same DH key group.
	PFS is ignored in initial IKEv2 authentication but is used when re-authenticating.
Lifetime (seconds)	Enter the maximum number of seconds the IPSec SA can last. Shorter life times provide better security. The Nebula Device automatically negotiates a new IPSec SA before the current one expires, if there are users who are accessing remote resources.
Close	Click this button to exit this screen without saving.
OK	Click this button to save your changes and close the screen.

 Table 85
 Site-wide > Configure > Security router > Site-to-Site VPN: IPsec Policy (continued)

7.3.6 SSID Advanced Settings

Use this screen to configure WiFi security, band selection, assisted roaming and U-APSD (Unscheduled automatic power save delivery) settings for the SSID profiles.

Click Site-wide > Configure > Security router > SSID advanced settings to access this screen. NCC will redirect to the Site-wide > Configure > Access points > SSID advanced settings screen.

iD advancea settings		
D: SSID1	*	
Network access		
Security options 🌖	O Open	
	Users can connect without entering a password	
	O WPA Personal With WPA2 ▼	
	Users must enter this key to associate: 💿 *	
	Wi-Fi Access QR Code Print	
Advanced settings		
Band mode	✓ 2.4GHz band	
	✓ 5GHz band	
	6GHz band Why can't I see WIFI in 6GHz?	
Assisted roaming	Enable 802.11k/v	
1100000		

Figure 109 Site-Wide > Configure > Access points > SSID advanced settings

The following table describes the labels in this screen.

LABEL	DESCRIPTION
SSID advanced settings	Select the SSID profile to which the settings you configure here is applied.
Network access	
Security options	Select Open to allow any client to associate this network without any data encryption or authentication.
	Select WPA Personal With (WPA2/WPA3) and enter a pre-shared key from 8 to 64 case- sensitive keyboard characters to enable WPA2/3-PSK data encryption. Upon selecting WPA Personal With WPA3, Nebula Devices that do not support it will revert to WPA2.
	Click Print to display the QR code that includes the password for quick access. You can save the QR code as PDF.
Advanced settings	
Band mode	Select to have the SSID use either 2.4GHz band, 5GHz band, or 6GHz band only.
Assisted roaming	Select to turn on or off IEEE 802.11k/v assisted roaming on the Nebula Device.
	When the connected clients request 802.11k neighbor lists, the Nebula Device will response with a list of neighbor Nebula Devices that can be candidates for roaming. When the 802.11v capable clients are using the 2.4 GHz band, the Nebula Device can send 802.11v messages to steer clients to the 5 GHz band.
U-APSD	Select to turn on or off Automatic Power Save Delivery. This helps increase battery life for battery-powered WiFi clients connected to the Nebula Device.

Table 86 Site-Wide > Configure > Access points > SSID advanced settings

7.3.7 Radio Settings

Use this screen to configure global radio settings for the Nebula Device in the site. Click **Site-wide** > **Configure** > **Security router** > **Radio settings** to access this screen. NCC will redirect to the **Site-wide** > **Configure** > **Access points** > **Radio settings** screen.

o settings				
Channel width	2.4 GHz 24	0 MHz 💌		
	5 GHz 8	0 MHz 🔻	Why you should not use channel width 160M	Hz in 5GHz?
	6 GHz 16	50 MHz 🔻	Model list	
DCS setting	2.4 GHz channel deployme	ent: Manual	✓ Hide	
	Channel ID			Sele
	1	2	3	4
	5	6	7	8
	9	10	11	
	5 GHz channel deploymer	nt: Manual	✓ Hide	
	Channel ID			Sele
	36	40	44	48
	52 (DFS)	56 (DFS)	60 (DFS)	64 (DFS)
	100 (DFS)	104 (DFS	108 (DFS)	112 (DFS)
	116 (DFS)	120 (DFS)	124 (DFS)	128 (DFS)
	132 (DFS)	136 (DFS)	140 (DFS)	144 (DFS)
	149	153	157	161
	6 GHz channel deploymer	nt: Manual	Hide Model list	
	Channel ID			

Figure 110 Site-wide > Configure > Access points > Radio settings

Table 87 Site-wide > Configure > Access points > Radio settings

LABEL	DESCRIPTION
Channel width	Select the wireless channel bandwidth you want the Nebula Device to use.
	A standard 20 MHz channel offers transfer speeds of up to 144 Mbps (2.4 GHz) or 217 Mbps (5 GHz) whereas a 40 MHz channel uses two standard channels and offers speeds of up to 300 Mbps (2.4 GHz) or 450 Mbps (5 GHz). An IEEE 802.11ac-specific 80 MHz channel offers speeds of up to 1.3 Gbps.
	40 MHz (channel bonding or dual channel) bonds two adjacent radio channels to increase throughput. An 80 MHz channel consists of two adjacent 40 MHz channels. The WiFi clients must also support 40 MHz or 80 MHz. It is often better to use the 20 MHz setting in a location where the environment hinders the WiFi signal.
	Note: It is suggested that you select 20 MHz when there is more than one 2.4 GHz Nebula Device in the network.
DCS setting	
2.4 GHz channel deployment	Select All available channels to allow channel-hopping to have the Nebula Device automatically select the best channel.
	Select Manual to select the individual channels the Nebula Device switches between.

LABEL	DESCRIPTION
5 GHz channel deployment	Select how you want to specify the channels the Nebula Device switches between for 5 GHz operation.
	Select All available channels to have the Nebula Device automatically select the best channel.
	Select Manual to select the individual channels the Nebula Device switches between.
	Note: The method is automatically set to All available channels when no channel is selected or any one of the previously selected channels is not supported.
6 GHz channel deployment	Select how you want to specify the channels the Nebula Device switches between for 6 GHz operation.
	Select All available channels to have the Nebula Device automatically select the best channel.
	Select Manual to select the individual channels the Nebula Device switches between.
	Note: The method is automatically set to All available channels when no channel is selected or any one of the previously selected channels is not supported.

 Table 87
 Site-wide > Configure > Access points > Radio settings (continued)

7.3.8 Router Settings

Use this screen to configure DNS settings.

Click Site-wide > Configure > Security router > Router settings to access this screen.

÷	
1	Site-wide > Configure > Security router > Router settings
đ	Router settings
	DNS

Figure 111 Site-wide > Configure > Security router > Router settings

Address Record				
FQDN	IP Address		Description	
×	*	*	×	.
+ Add				
ynamic DNS				
Dynamic DNS				
	Dynamic DNS updates a	DNS record each time the public IP address of t	the security appliance changes.	
DDNS provider	Dynamic DNS updates a	DNS record each time the public IP address of t	the security appliance changes.	
DDNS provider	Dynamic DNS updates a DynDNS	DNS record each time the public IP address of t	the security appliance changes.	
DDNS provider Hostname	Dynamic DNS updates a DynDNS	DNS record each time the public IP address of t	the security appliance changes.	
DDNS provider Hostname	Dynamic DNS updates a DynDNS	DNS record each time the public IP address of t	the security appliance changes.	
DDNS provider Hostname Username	Dynamic DNS updates a DynDNS	DNS record each time the public IP address of t	the security appliance changes.	
DDNS provider Hostname Username Password	Dynamic DNS updates a DynDNS	DNS record each time the public IP address of t	the security appliance changes.	

Table 88 Site-wide > Configure > Security router > Router settings

LABEL	DESCRIPTION
DNS	
Address Record	This record specifies the mapping of a Fully-Qualified Domain Name (FQDN) to an IP address. An FQDN consists of a host and domain name. For example, www.zyxel.com.tw is a fully qualified domain name, where "www" is the host, "zyxel" is the third-level domain, "com" is the second-level domain, and "tw" is the top level domain.
FQDN	Enter a host's fully qualified domain name.
	Use "*." as a prefix in the FQDN for a wildcard domain name (for example, *.example.com).
IP Address	Enter the host's IP address.
Description	Enter the descriptive name of the DNS record of up to 60 printable ASCII characters.
	Click this icon to remove the entry.
Add	Click this button to create a new entry.
Dynamic DNS	
Dynamic DNS	Click On to use dynamic DNS. Otherwise, select Off to disable it.
DDNS provider	Select your Dynamic DNS service provider from the drop-down list box.
	If you select User customize , create your own DDNS service.
Hostname	Enter the domain name you registered.
Username	Enter the user name used when you registered your domain name.
Password	Enter the password provided by the DDNS provider.

Chapter 8 Firewall

8.1 Overview

This chapter describes the menus used to monitor and configure the Hybrid Security Firewall devices that acts as a security gateway in the current organization.

Nebula Device (also called Security Firewall device) refers to ZyWALL ATP / USG FLEX / USG20(W)-VPN Series devices in this chapter. The **Firewall** menus are shown for Security Firewall devices only.

8.2 Monitor

Use the **Monitor** menus to check the Nebula Device information, client information, event log messages and summary report for the Nebula Device in the selected site.

8.2.1 Clients

This menu item redirects to Site-Wide > Monitor > Clients, with type set to Security firewall clients. For details, see Section 4.5 on page 192.

8.2.2 Event Log

Use this screen to view Nebula Device log messages. You can enter a key word, select one or multiple event types, or specify a date/time or a time range to display only the log messages that match these criteria.

Select **Range** to set a time range or select **Before** to choose a specific date/time and the number of hours/minutes to display only the log messages generated within a certain period of time (before the specified date/time). Then click **Search** to update the list of logs based on the search criteria. The maximum allowable time range is 30 days.

Click Site-wide > Monitor > Firewall > Event log to access this screen.

Figure 112 Site-wide > Monitor > Firewall > Eve	nt log
---	--------

Keyword:		Category:						
Any	×	Any		Before 💌	2023-03-30	11:26	▼ 1h ▼ UTC+8	🗵 🔍 Search
< Newer Older > 0	Event logs		No matching e	events found between	2023-03-30 10:26 and 20	23-03-30 11:26		Export
Time Categor	y Source IP	Destination IP	Detail					E

8.2.3 VPN Connections

Use this screen to view the status of site-to-site IPSec VPN connections and L2TP VPN connections.

Note: If the peer gateway is not a Nebula Device, go to the Firewall > Configure > Site-to-Site VPN screen to view and configure a VPN rule. See Section 8.3.5 on page 365 for more information.

Click Site-wide > Monitor > Firewall > VPN connections to access this screen.

Figure 113 Site-wide > Monitor > Firewall > VPN connections

Configuration:	15	This security gateway i	s exporting 4 subnet over	the VPN: 192.168.128.0/24, 19	92.168.2.0/24, 192.168.10.0/24, 1	92.168.100.0/24
ite connectivity						
_ocation	Subnet	Status	Inbound	Outbound	Tunnel Up Time	Last Heartbeat
lon-Nebula VPN	peers connectivity					
ocation	Subnet	Status	Inbound	Outbound	Tunnel Up Time	Last Heartbeat
emote AP VPN						
Name	Status	Inbound	Out	bound	Tunnel Up Time	Last Heartbeat
lient to site VPM	N login account					
		Hostname	As	signed IP	Public IP	

Table 89	Site-wide	> Monitor :	> Firewall >	VPN connections
----------	-----------	-------------	--------------	------------------------

LABEL	DESCRIPTION
C	Click this button to reload the data on this page.
Connection Status	
Configuration	This shows the number and address of the local networks behind the Nebula Device, on which the computers are allowed to use the VPN tunnel.
Site Connectivity	
Location	This shows the name of the site to which the Nebula peer gateway is assigned.
	Click the name to view the VPN usage and connectivity status screen.
Subnet	This shows the address of the local networks behind the Nebula peer gateway.
Status	This shows whether the VPN tunnel is connected or disconnected.
Last heartbeat	This shows the last date and time a heartbeat packet is sent to determine if the VPN tunnel is up or down.
Non-Nebula VPN peer	rs connectivity

LABEL	DESCRIPTION
Location	This shows the name of the site to which the Non-Nebula peer gateway (Zyxel or non-Zyxel IPSec VPN gateway and Cloud VPN (Azure VPN or AWS VPN)) is assigned.
	Click the name to go to the Site-wide > Configure > Firewall > Site-to-Site VPN screen, where you can modify the VPN settings.
Subnet	This shows the address of the local networks behind the Non-Nebula peer gateway.
Status	This shows whether the VPN tunnel is connected or disconnected.
Inbound	This shows the amount of traffic that has gone through the VPN tunnel from the Non-Nebula peer gateway to the Nebula Device since the VPN tunnel was established.
Outbound	This shows the amount of traffic that has gone through the VPN tunnel from the Nebula Device to the Non-Nebula peer gateway since the VPN tunnel was established.
Tunnel up time	This shows how many seconds the VPN tunnel has been active.
Last heartbeat	This shows the last date and time a heartbeat packet was sent to determine if the VPN tunnel is up or down.
Remote AP VPN	
Name	This shows the name of the remote access point (AP).
Status	This shows whether the VPN tunnel is connected or disconnected.
Inbound	This shows the amount of traffic that has gone through the VPN tunnel from the remote AP to the Nebula Device since the VPN tunnel was established.
Outbound	This shows the amount of traffic that has gone through the VPN tunnel from the Nebula Device to the remote AP since the VPN tunnel was established.
Tunnel up time	This shows how many seconds the VPN tunnel has been active.
Last heartbeat	This shows the last date and time a heartbeat packet is sent to determine if the VPN tunnel is up or down.
Client to site VPN login	account
User Name	This shows the remote user's login account name.
Hostname	This shows the name of the computer that has this L2TP VPN connection with the Nebula Device.
Tunnel up time	This shows how many seconds the VPN tunnel has been active.
Assigned IP	This shows the IP address that the Nebula Device assigned for the remote user's computer to use within the L2TP VPN tunnel.
Public IP	This shows the public IP address that the remote user is using to connect to the Internet.

Table 89	Site-wide > Monite	or > Firewall > VPN	connections	(continued)
	2116-MIGE > MIOLIII		CONTRECTIONS	(commoed)

8.2.4 SecuReporter

Click **Site-wide** > **Monitor** > **Firewall** > **SecuReporter** to open SecuReporter for the current organization and site. SecuReporter allows you to view statistics for the following Nebula Security Services (NSS): Content filtering, Intrusion Detection and Prevention (IDP), application patrol, anti-virus, anti-malware, URL threat filter.

Note: For more details, see the SecuReporter User's Guide.



Figure 114 Site-wide > Monitor > Firewall > SecuReporter

8.2.5 Summary Report

This screen displays network statistics for the Nebula Device of the selected site, such as WAN usage, top applications and/or top clients.

Click Site-wide > Monitor > Firewall > Summary report to access this screen.







|--|

LABEL	DESCRIPTION			
Security gateway – Summary report	Select to view the report for the past day, week or month. Alternatively, select Custom range to specify a time period the report will span. You can also select the number of results you want to view in a table.			
	🔘 Last 24 hours			
	💎 🖸 Last 7 days			
	💎 🔿 Custom range			
	C Update			
Email report	Click this button to send summary reports by email, change the logo and set email schedules.			
WAN usage				
y-axis	The y-axis shows the transmission speed of data sent or received through the WAN connection in kilobits per second (Kbps).			
x-axis	The x-axis shows the time period over which the traffic flow occurred.			
VPN usage				
y-axis	The y-axis shows the transmission speed of data sent or received through the VPN tunnel in kilobits per second (Kbps).			
x-axis	The x-axis shows the time period over which the traffic flow occurred.			
Nebula VPN usage				
y-axis	The y-axis shows the transmission speed of data sent or received through the VPN tunnels, in kilobits per second (Kbps).			
x-axis	The x-axis shows the time period over which the traffic flow occurred.			
Non-Nebula VPN usag	e			
y-axis	The y-axis shows the transmission speed of data sent or received through VPN tunnels, in kilobits per second (Kbps).			
x-axis	The x-axis shows the time period over which the traffic flow occurred.			
Remote AP VPN usage				
y-axis	The y-axis shows the transmission speed of data sent or received through the VPN tunnel between the Nebula Device and remote APs, in kilobits per second (Kbps).			
x-axis	The x-axis shows the time period over which the traffic flow occurred.			
Security gateway by u	sage			
	This shows the index number of the Nebula Device.			
Name	This shows the descriptive name of the Nebula Device.			
Model	This shows the model number of the Nebula Device.			
Usage	This shows the amount of data that has been transmitted through the Nebula Device's WAN port.			
Client	This shows the number of clients currently connected to the Nebula Device.			
Location				
This shows the location	of the Nebula Devices on the map.			
Top applications by us	age			
	This shows the index number of the application.			
Application	This shows the application name.			

Taila 00	Cita wida >	Manitars	Firewalls	C	· · · · · · · · · · · ·	
100ie 90	sile-wide >	NOULOI >	rirewali >	Somman	/ report	(continued)

LABEL	DESCRIPTION			
Category	This shows the name of the category to which the application belongs.			
Usage	This shows the amount of data consumed by the application.			
% Usage	This shows the percentage of usage for the application.			
Top ports by usage				
	This shows the top ten applications/services and the ports that identify a service.			
Name	This shows the service name and the associated port numbers.			
Usage	This shows the amount of data consumed by the service.			
% Usage	This shows the percentage of usage for the service.			
Clients per day				
y-axis	The y-axis represents the number of clients.			
x-axis	The x-axis represents the date.			
Top clients by usage				
	This shows the index number of the client.			
Description	This shows the descriptive name or MAC address of the client.			
Usage	This shows the total amount of data transmitted and received by the client.			
% Usage	This shows the percentage of usage for the client.			
Top operating systems	by usage			
	This shows the index number of the operating system.			
OS	This shows the operating system of the client device.			
# Client	This shows how many client devices use this operating system.			
% Client	This shows the percentage of top client devices which use this operating system.			
% Usage	This shows the percentage of usage for top client devices which use this operating system.			
Top client device man	ufacturers by usage			
	This shows the index number of the client device.			
Manufacturer	This shows the manufacturer name of the client device.			
Client	This shows how many client devices are made by the manufacturer.			
% Client	This shows the percentage of top client devices which are made by the manufacturer.			
Usage	This shows the total amount of data transmitted and received by the client device.			
% Usage	This shows the percentage of usage for the client device.			
CPU usage				
y-axis	The y-axis shows what percentage of the Nebula Device's processing capability is currently being used.			
x-axis	The x-axis shows the time period over which the traffic flow occurred.			
Memory usage				
y-axis	The y-axis shows what percentage of the Nebula Device's RAM is currently being used.			
x-axis	The x-axis shows the time period over which the traffic flow occurred.			
Sessions usage	·			
y-axis	The y-axis shows how many sessions, both established and non-established, that were create from, to, or within the Nebula Device, or passed through the Nebula Device.			
x-axis	The x-axis shows the time period over which the traffic flow occurred.			

8.3 Configure

Use the **Configure** menus to configure interface addressing, firewall, site-to-site VPN, captive portal, traffic shaping, authentication server and other gateway settings for the Nebula Device of the selected site.

Note: Only one Security Appliance is allowed per site.

8.3.1 Port

Use this screen to configure port groups on the Nebula Device. To access this screen, click **Firewall** > **Configure** > **Port**.

Port Port Group P3 P4 P5 P6 P7 P1 P2 P8 Port Type Optional Option Optiona WAN Port Group WAN Group 1 0 WAN Group 2 0 $+ \operatorname{Add}$ LAN Port Group 0 \bigcirc 0 0 LAN Group 1 0 LAN Group 2 0 0 $+ \operatorname{Add}$

Figure 116 Site-wide > Configure > Firewall > Port

The following table describes the labels in this screen.

Table 91 Site-wide > Configure > Firewall > Port

LABEL	DESCRIPTION
Port Group	Port groups create a hardware connection between physical ports at the layer-2 (data link, MAC address) level.
	The physical LAN Ethernet ports, for example P1, P2, P3, are shown at the top of the screen. The port groups are shown at the left of the screen. Use the radio buttons to select which ports are in each port group.
	For example, to add port P3 to LAN Group 1, select P3's radio button in the LAN Group 1 row.
	Note: See Table 1 on page 13 for the list of Nebula Device that do NOT have a P1 port.
Port Type	This shows whether the port is a WAN port or a LAN port. Optional means the port can be assigned as either WAN or LAN, by adding it to a WAN or LAN group.
WAN Port Group	
WAN Group 1	This shows the name of the WAN port group.
	Note: Each WAN port group can only contain one port.
.	Click this icon to remove a WAN port group.

NCC User's Guide

LABEL	DESCRIPTION
Add	Click this button to create a new WAN port group.
LAN Port Group	
LAN Group 1	This shows the name of the LAN port group.
1	Click this icon to remove a LAN port group.
Add	Click this button to create a new LAN port group.
Close	Click Close to exit this screen without saving.
OK	Click OK to save your changes.

Table 91 Site-wide > Configure > Firewall > Port (continued)

8.3.2 Interface

Use this screen to configure network interfaces on the Nebula Device. An interface consists of a port group, a VLAN ID, and an IP address, plus other configuration settings.

To access this screen, click **Site-wide** > **Configure** > **Firewall** > **Interface**.

Figure 117 Site-wide > Configure > Firewall > Interface

			WAN interfac	ce				
Name	Status	IP address	Subnet mask	VLAN	ID I	Port group		
wan1						WAN Group 1	Ŧ	Z
wan2						WAN Group 2	*	2
$+ \operatorname{Add}$								
			LAN interfac	ce				
Name	Status	IP address	Subnet mask	VLAN ID	Port group	Guest		
	_	1001601001	255 255 255 0		LAN Group 1	•		Z
lan1		132.100.120.1						
lan1 lan2		192.168.2.1	255.255.255.0		LAN Group 2	- 0		2 1
lan1 lan2 VLAN10		192.168.2.1 192.168.10.1	255.255.255.0 255.255.255.0	10	LAN Group 2 LAN Group 1			2
lan1 lan2 VLAN10 VLAN100		192.168.2.1 192.168.10.1 192.168.100.1	255,255,255,0 255,255,255,0 255,255,255,0	10	LAN Group 2 LAN Group 1 LAN Group 1			

LABEL	DESCRIPTION
WAN Interface	
Name	This field is read-only if you are editing an existing WAN interface. Specify a name for the interface.
	The format of interface names is strict. Each name consists of $2 - 4$ letters (interface type), followed by a number (x). For most interfaces, x is limited by the maximum number of the type of interface. For VLAN interfaces, x is defined by the number you enter in the VLAN name field. For example, VLAN interfaces are vlan0, vlan1, vlan2, and so on.

Table 92 Site-wide > Configure > Firewall > Interface

LABEL	DESCRIPTION				
Status	Select this to activate the selected WAN interface.				
IP address	This shows the IP address for this interface.				
Subnet mask	This shows the subnet mask of this interface in dot decimal notation. The subnet mask indicates what part of the IP address is the same for all computers in the network.				
VLAN ID	This shows the VLAN ID. This 12-bit number uniquely identifies each VLAN. Allowed values are 1 – 4094. (0 and 4095 are reserved.)				
	Note: NCC will show an error message when the VLAN ID in the interface is configured to be the same as the WAN port's VLAN ID.				
Port group	Select the name of the port group to which you want the interface to (network) belong.				
2	Click the edit icon to modify the interface.				
m	Click the remove icon to delete the interface.				
Add	Click this button to create a virtual WAN interface, which associates a VLAN with a WAN port group.				
LAN Interface					
Name	This field is read-only if you are editing an existing LAN interface.				
	Specify a name for the interface.				
	The format of interface names is strict. Each name consists of $2 - 4$ letters (interface type), followed by a number (x). For most interfaces, x is limited by the maximum number of the type of interface. For VLAN interfaces, x is defined by the number you enter in the VLAN name field. For example, VLAN interfaces are vlan0, vlan1, vlan2, and so on.				
Status	Select this to activate the LAN interface.				
IP address	This is the IP address for this interface.				
Subnet mask	This is the subnet mask of this interface in dot decimal notation. The subnet mask indicates what part of the IP address is the same for all computers in the network.				
VLAN ID	This is the VLAN ID. This 12-bit number uniquely identifies each VLAN. Allowed values are 1 – 4094. (0 and 4095 are reserved.)				
	Note: NCC will show an error message when the VLAN ID in the Security Firewall interface is configured to be the same as the WAN port's VLAN ID.				
Port group	Select the name of the port group to which you want the interface to (network) belong.				
Guest	Select On to configure the interface as a Guest interface. Client devices connected to a Guest interface have Internet access but cannot communicate with each other directly or access networks behind the Nebula Device.				
2	Click the edit icon to modify it.				
1	Click the remove icon to delete it.				
Add	Click this button to create a virtual LAN interface, which associates a VLAN with a LAN port group.				

Table 92 Site-wide > Configure > Firewall > Interface (continued)

8.3.2.1 WAN Interface Configuration

Click the Add button or click the Edit button in the WAN Interface section to open the Site-wide > Configure > Firewall > Interface > WAN interface configuration screen.

N interface configuation	n	×
Enabled		
Interface properties		
Interface name	VLAN1 ×	
Port group	WAN Group 1 👻	
SNAT		
VLAN ID	× (1 - 4094)	
Туре	DHCP 👻	
Downstream bandwidth	x	
Upstream bandwidth	×	
MTU	1500 × (Bytes)	
	ADVANCED OPTIONS	
Connectivity check	None	
	O Default gateway	
	Check the two addresses below	

Figure 118	Site-wide >	Configure 2	> Firewall >	Interface >	> WAN interface	configuration
		oor mgoro		111011000		oornigoranorr

LABEL	DESCRIPTION			
Enable	Select this to enable the WAN interface.			
Interface properties	•			
Interface name	Specify a name for the WAN interface.			
Port group	Select the name of the port group to which you want the interface to (network) belong.			
SNAT	Select this to enable SNAT. When enabled, the Nebula Device rewrites the source address of packets being sent from this interface to the interface's IP address.			
VLAN ID	Enter the VLAN ID. This 12-bit number uniquely identifies each VLAN. Allowed values are 1 – 4094. (0 and 4095 are reserved.)			

Table 02 Si	o wido >	Configura	Eirowall >	Interface >	. \A/ A N I	intorface	configuration
	e-wide /		riewui -	Intendce -		Intendce	Composition

LABEL	DESCRIPTION
Туре	Select the type of interface to create.
	DHCP : The interface will automatically get an IP address and other network settings from a DHCP server.
	Static: You must manually configure an IP address and other network settings for the interface.
	PPPoE : The interface will authenticate with an Internet Service Provider, and then automatically get an IP address from the ISP's DHCP server. You can use this type of interface to connect to a DSL modem.
	PPPOE with static IP : Assign a static IP address to the WAN interface and your WAN interface is getting an Internet connection from a PPPOE server.
IP address assignment	These fields are displayed if you select Static .
IP address	Enter the static IP address of this interface.
Subnet mask	Enter the subnet mask for this interface's IP address.
Default gateway	Enter the IP address of the Nebula Device through which this interface sends traffic.
First DNS server	Enter a DNS server's IP address.
	The Domain Name System (DNS) maps a domain name to an IP address and vice versa. The Nebula Device uses the first and second DNS servers, in that order to resolve domain names for VPN, DDNS and the time server. Leave the field blank if you do not want to configure DNS servers.
Second DNS server	Enter the IP address of another DNS server. This field is optional.
These fields are displayed if	you selected PPPoE or PPPoE with static IP.
Authentication Type	Select an authentication protocol for outgoing connection requests. Options are:
	Chap/PAP – The Nebula Device accepts either CHAP or PAP when requested by the remote node.
	Chap – The Nebula Device accepts CHAP only. PAP The Nebula Device accepts PAP only.
	 MSCHAP – The Nebula Device accepts Al Only. MSCHAP – The Nebula Device accepts MSCHAP only.
	• MSCHAP-V2 – The Nebula Device accepts MSCHAP-V2 only.
Username	Enter the user name provided by your ISP. You can use up to 31 alphanumeric characters and the underscore. Spaces are not allowed.
Password	Enter the password provided by your ISP. You can use up to 64 alphanumeric characters and the underscore. Spaces are not allowed.
Retype password	Enter the password again to confirm it.
Downstream bandwidth	Enter the downstream bandwidth of the WAN connection. This value is used for WAN load balancing by algorithms such as weighed round robin.
Upstream bandwidth	Enter the upstream bandwidth of the WAN connection. This value is used for WAN load balancing by algorithms such as weighed round robin.
MTU	Maximum Transmission Unit. Enter the maximum size of each data packet, in bytes, that can move through this interface. If a larger packet arrives, the Nebula Device divides it into smaller fragments. Allowed values are 576 – 1500.
ADVANCED OPTIONS	
Connectivity check	The interface can periodically check whether it can connect to its default gateway (Default gateway), or to two user-specified servers (Check the two addresses below). If the check fails, the interface's status changes to Down.
	You specify how often the interface checks the connection, how long to wait for a response before the attempt is a failure, and how many consecutive failures are required before the Nebula Device stops routing to the gateway.

Table 93	Site wide > Continuire '	> Firewall > Intertace >	WAN intertace	continuration	(continued)
				coningoration	

LABEL	DESCRIPTION
Probe Succeeds When	This field applies when you select Check the two addresses and specify two domain names or IP addresses for the connectivity check.
	Select any one if you want the check to pass if at least one of the domain names or IP addresses responds.
	Select all if you want the check to pass only if both domain names or IP addresses respond.
Proxy ARP	Proxy ARP (RFC 1027) allows the Nebula Device to answer external interface ARP requests on behalf of a device on its internal interface.
	Click Add new to add the IP address or IP range of devices that the interface will answer proxy ARP requests for.
IP Address	Enter a single IPv4 address, an IPv4 CIDR (for example, 192.168.1.1/24) or an IPv4 Range (for example, 192.168.1.2–192.168.1.100).
	The Nebula Device answers external ARP requests if they match one of these target IP addresses. For example, if the IPv4 address is 192.168.1.5, then the Nebula Device will answer ARP requests coming from the WAN only if it contains 192.168.1.5 as the target IP address.
1	Click the remove icon to delete the proxy ARP IP address.
MAC address Setting	Have the interface use either the factory-assigned default MAC address, or a manually specified MAC address.
DHCP client mode	Choices are Auto, Unicast and Broadcast.
DHCP option 60	DHCP Option 60 is used by the Security Firewall for identification to the DHCP server using the VCI (Vendor Class Identifier) on the DHCP server. The Nebula Device adds it in the initial DHCP discovery message that a DHCP client broadcasts in search of an IP address. The DHCP server can assign different IP addresses or options to clients with the specific VCI or reject the request from clients without the specific VCI.
	Enter a string using up to 63 of these characters $[a-z A-Z 0-9 !\"#$%&\'()*+,/:;<=>?@\[\\\]^_`{}] to identify this Nebula Device to the DHCP server. For example, Zyxel-TW.$
IGMP proxy	Select this to allow the Nebula Device to act as an IGMP proxy for hosts connected on the IGMP downstream interface.
IGMP Upstream	Enable IGMP Upstream on the interface which connects to a router running IGMP that is closer to the multicast server.
IGMP Downstream	Enable IGMP Downstream on the interface which connects to the multicast hosts.
Close	Click Close to exit this screen without saving.
OK	Click OK to save your changes.

Table 93	Site-wide > Configure :	> Firewall > Interface :	> WAN interface	configuration	(continued)
				conngoranon	

8.3.2.2 LAN Interface Configuration

Click the Add button or click the Edit button in the LAN interface section to open the Site-wide > Configure > Firewall > Interface > LAN interface configuration screen.

N interface configuation				×
Enabled				
Interface properties				
Interface name	VLAN1	×		
Port group	LAN Group 1	•		
VLAN ID		×	(1 - 4094)	
IP address assignment				
IP address		×		
Subnet mask		×		
DHCP setting				
DHCP	None	•		
MTU	1500	×	(Bytes)	
	ADVANCED OPTIONS			
DHCP extended options				
First WINS server		×		
Second WINS server				

Figure 110	Cite wide >	Configuras	Fireworld	listerfeise s	> I A NI interferee	a a sofice waties
Figure 119	Sile-wide >	Conlidure >	· FIGWOII 2	• intenace .	> I AIN INTERIOCE	CONTIQUIGITON
	0110 1110.0	001110				00111000

	Table 94	Site-wide >	Configure >	· Firewall >	Interface >	LAN interface	configuration
--	----------	-------------	-------------	--------------	-------------	---------------	---------------

LABEL	DESCRIPTION				
Enable	Select this to enable the LAN interface.				
Interface properties					
Interface name	Specify a name for the LAN interface.				
Port group	Select the name of the port group to which you want the interface to (network) belong.				
VLAN ID	Enter the VLAN ID. This 12-bit number uniquely identifies each VLAN. Allowed values are 1 – 4094. (0 and 4095 are reserved.)				
IP address assignment					
IP address	Enter the IP address for this interface.				
Subnet mask	Enter the subnet mask of this interface in dot decimal notation. The subnet mask indicates what part of the IP address is the same for all computers in the network.				

LABEL	DESCRIPTION
DHCP setting	Select what type of DHCP service the Nebula Device provides to the network. Choices are:
	None – the Nebula Device does not provide any DHCP services. There is already a DHCP server on the network.
	DHCP Relay – the Nebula Device routes DHCP requests to one or more DHCP servers you specify. The DHCP servers may be on another network.
	DHCP Server – the Nebula Device assigns IP addresses and provides subnet mask, gateway, and DNS server information to the network. The Nebula Device is the DHCP server for the network.
These fields appear if the Net	bula Device is a DHCP Relay.
DHCP server 1	Enter the IP address of a DHCP server for the network.
DHCP server 2	This field is optional. Enter the IP address of another DHCP server for the network.
These fields appear if the Net	pula Device is a DHCP Server.
IP pool start address	Enter the IP address from which the Nebula Device begins allocating IP addresses. If you want to assign a static IP address to a specific computer, use the Static DHCP Table .
	If this field is blank, the Pool Size must also be blank. In this case, the Nebula Device can assign every IP address allowed by the interface's IP address and subnet mask, except for the first address (network address), last address (broadcast address) and the interface's IP address.
First DNS Server, Second DNS Server, Third DNS Server	Specify the IP addresses of up to three DNS servers for the DHCP clients to use. Use one of the following ways to specify these IP addresses.
	Custom Defined – enter a static IP address.
	From ISP – select the DNS server that another interface received from its DHCP server.
	This Gateway – the DHCP clients use the IP address of this interface and the Nebula Device works as a DNS relay.
Lease Time	Specify how long each computer can use the information (especially the IP address) before it has to request the information again. Choices are:
	infinite – select this if IP addresses never expire.
	days, hours, and minutes (Optional) – select this to enter how long IP addresses are valid.
Static DHCP table	Configure a list of static IP addresses the Nebula Device assigns to computers connected to the interface. Otherwise, the Nebula Device assigns an IP address dynamically using the interface's IP Pool Start Address and Pool Size.
IP address	Enter the IP address to assign to a device with this entry's MAC address.
MAC	Enter the MAC address to which to assign this entry's IP address.
Description	Enter a description to help identify this static DHCP entry. You can use alphanumeric and ()+/:=?!*#@\$_%- characters, and it can be up to 60 characters long.
1	Select an entry in this table and click this to delete it.
Add New	Click this to create an entry in the Static DHCP table.
MTU	Maximum Transmission Unit. Enter the maximum size of each data packet, in bytes, that can move through this interface. If a larger packet arrives, the Nebula Device divides it into smaller fragments. Allowed values are 576 – 1500. Usually, this value is 1500.
ADVANCED OPTIONS	

T.L. OA			1.	A N I * 1 I		
I able 94	Site-wide > Configure >	> Firewali >	· Interface > L	AN INTERTACE	configuration (continuea)

LABEL	DESCRIPTION
DHCP extended options	This table is available if you select ADVANCED OPTIONS.
	Configure this table if you want to send more information to DHCP clients through DHCP packets.
	Click Add new to create an entry in this table. See Section 7.3.2.3 on page 189 for detailed information.
First WINS server	Enter the IP address of the WINS (Windows Internet Naming Service) server that you
Second WINS server	computer names on your network and the IP addresses that they are currently using.
PXE server	PXE (Preboot eXecution Environment) allows a client computer to use the network to boot up and install an operating system through a PXE-capable Network Interface Card (NIC).
	PXE is available for computers on internal interfaces to allow them to boot up using boot software on a PXE server. The Nebula Device acts as an intermediary between the PXE server and the computers that need boot software.
	The PXE server must have a public IPv4 address. You must enable DHCP server on the Nebula Device so that it can receive information from the PXE server.
PXE Boot loader file	A boot loader is a computer program that loads the operating system for the computer. Enter the exact file name of the boot loader software file, including filename extension, that is on the PXE server. If the wrong filename is entered, then the client computers cannot boot.
Default gateway	If you set this interface to DHCP server, you can select to use either the interface's IP address or another IP address as the default router. This default router will become the DHCP clients' default gateway.
IGMP proxy	Select this to allow the Nebula Device to act as an IGMP proxy for hosts connected on the IGMP downstream interface.
IGMP Upstream	Enable IGMP Upstream on the interface which connects to a router running IGMP that is closer to the multicast server.
IGMP Downstream	Enable IGMP Downstream on the interface which connects to the multicast hosts.
Close	Click Close to exit this screen without saving.
OK	Click OK to save your changes.

Table 94	Site-wide >	Configure >	Firewall >	Interface >	I AN interface	configuration	(continued)
	Sile-wide >	Conngole -		intendce -	LANIMENUCE	conngoranon	

8.3.2.3 DHCP Option

Click the Add new button in the DHCP extended options section to open the Site-wide > Configure > Firewall > Interface > LAN interface configuration: DHCP option screen.

DHCP option		×
Option	User defined	•
Name	User_Defined	×
Code	× (1-254)	
Туре	IP	•
First IP address		×
Second IP address		×
Third IP address		×
	Clos	еОК

Figure 120 Site-wide > Configure > Firewall > Interface: LAN interface configuration: DHCP option

The following table describes the labels in this screen.

LABEL	DESCRIPTION
Option	Select which DHCP option that you want to add in the DHCP packets sent through the interface.
Name	This field displays the name of the selected DHCP option. If you selected User defined in the Option field, enter a descriptive name to identify the DHCP option.
Code	This field displays the code number of the selected DHCP option. If you selected User defined in the Option field, enter a number for the option. This field is mandatory.
Туре	This is the type of the selected DHCP option. If you selected User defined in the Option field, select an appropriate type for the value that you will enter in the next field. Misconfiguration could result in interface lockout.
Value	Enter the value for the selected DHCP option. For example, if you selected TFTP Server Name (66) and the type is TEXT , enter the DNS domain name of a TFTP server here. This field is mandatory.
First/Second/Third IP address	If you selected User defined / Time/NTP/SIP/TFTP server / CAPWAP AC in the Option field, enter up to three IP addresses.
Close	Click Close to exit this screen without saving.
ОК	Click OK to save your changes.

Table 95 Site-wide > Configure > Firewall > Interface: LAN interface configuration: DHCP option

8.3.3 Routing

Use policy routes and static routes to override the Nebula Device's default routing behavior in order to send packets through the appropriate next-hop gateway, interface or VPN tunnel.

A policy route defines the matching criteria and the action to take when a packet meets the criteria. The action is taken only when all the criteria are met. Use this screen to configure policy routes.

Click Site-wide > Configure > Firewall > Routing: Policy Route/Traffic Shaping to access this screen.

irewall > Configure > <u>Routin</u> Couting	1						
Policy Route/Traffic Sha	ping						
Enabled	Source	Destination	Service	Next-Hop	Traffic Shaping	Description	
←੍ਹੈ→ 1 🔽	Any	Any	Any	Internet: wan1	Download Limit: unlimited Upload Limit: unlimited Priority: Medium(4)	PR-1	2
+ Add							

Figure 121 Site-wide > Configure > Firewall > Routing: Policy Route/Traffic Shaping

The following table describes the labels in this screen.

LABEL	DESCRIPTION
¢€	Click the icon of a rule and drag the rule up or down to change the order.
Enabled	Select the check box to turn on the rule. Otherwise, clear the check box to turn off the rule.
Source	This shows the source IP addresses to which this rule applies. This could be an IP, CIDR, FQDN, or GEO IP (country) object.
Destination	This shows the destination IP addresses to which this rule applies. This could be an IP, CIDR, FQDN, or GEO IP (country) object.
Service	This is the name of the service object (port) or application. Any means all services.
	Select Protocol to specify a protocol by protocol ID number, as defined in the IPv4 header. For example, $1 = ICMP$, $2 = IGMP$.
Next Hop	This is the next hop to which packets are directed. It helps forward packets to their destinations and can be a router, VPN tunnel, or outgoing interface.
Traffic Shaping	This displays the maximum downstream and upstream bandwidth for traffic from an individual source IP address and the priority level.
Description	This is the descriptive name of the policy.
2	Click this icon to change the profile settings.
1	Click this icon to remove the profile.
Add	Click this button to create a new policy route. See Section 8.3.7.1 on page 380 for more information.

Table 96 Site-wide > Configure > Firewall > Routing: Policy Route/Traffic Shaping

8.3.3.1 Add/Edit Policy Route / Traffic Shaping Rule

Click the Add button or an edit icon in the Site-wide > Configure > Firewall > Routing: Policy Route/Traffic Shaping: Add/Edit screen to access this screen.
latching Criteria		
Description:		×
Source:		
Destination:	Any 😒	
Service:	Any	•
olicy Route 🗾		
Туре:	Internet Traffic	•
Next-Hop:	wan1	-
raffic Shaping 🛑		
Download Limit:	O Unlimited	
	Kbps	
Upload Limit:	O Unlimited	
	Кърз	

Figure 122 Site-wide > Configure > Firewall > Routing: Policy Route/Traffic Shaping: Add/Edit

Table 97	Site-wide > Configure >	Firewall > Routing: Policy	Route/Traffic Shaping: Add/Edit
	0	.	

LABEL	DESCRIPTION						
Matching Criteria							
Description	Enter a descriptive name for the rule.						
Source	 Specify the source IP addresses (LAN interface / country) to which this rule applies. You can add multiple IP, CIDR, GEO IP (country) objects or a single FQDN object by pressing 'Enter', or enter a new IP address by clicking Add. Select Any to apply the rule to all IP addresses. Note: IP/CIDR, FQND, and GEO IP objects cannot be used at the same time. Multiple FQDNs are not supported. The IP FQDN does NOT support wildcards. 						
Destination	Specify the destination IP addresses (LAN interface / country) or subnet to which this rule applies. You can add multiple IP, CIDR, GEO IP (country) objects or a single FQDN object by pressing 'Enter', or enter a new IP address by clicking Add . Select Any to apply the rule to all IP addresses. Note: IP/CIDR, FQND, and GEO IP objects cannot be use at the same time. Multiple FQDNs are not supported.						

LABEL	DESCRIPTION
Service	Select a protocol to apply the policy route to.
	TCP, UDP, TCP & UDP, ICMP – Match packets from the specified network protocol, going to the optional destination port.
	Protocol – Match packets for the specified custom protocol. Enter the Protocol ID , 1 – 143 (1 for ICMP, 6 for TCP, 17 for UDP; the Service will automatically select ICMP / TCP / UDP respectively).
	Application – Match packets from the application.
	Otherwise, select Any .
Policy Route	Select this to enable policy route.
Туре	Select Internet Traffic to route the matched packets through the specified outgoing interface to a gateway (which is connected to the interface).
	Select Intranet Traffic to route the matched packets to the next-hop router or Switch you specified in the Next-Hop field.
	Select VPN Traffic to route the matched packets through the VPN tunnel you specified in the Next-Hop field.
Next-Hop	If you select Internet Traffic in the Type field, select the WAN interface to route the matched packets through the specified outgoing interface to a gateway connected to the interface.
	If you select Intranet Traffic in the Type field, enter the IP address of the next-hop router or Switch.
	If you select VPN Traffic in the Type field, select the remote VPN gateway's site name.
	 Only the VPN gateway sites belonging to the same VPN Area that you set in Organization- wide > Organization-wide manage > VPN orchestrator will be available. See Section 11.4.4.3 on page 524 for more information).
	 Setting a Policy Route to force traffic over a VPN tunnel between a Security Firewall and Nebula Security Gateway (NSG) is not supported. Both front/back end Nebula Devices must be the same type.
Traffic Shaping	Select this to restrict maximum downstream and upstream bandwidth for traffic in the policy route.
Download Limit	Set the maximum downstream bandwidth for traffic that matches the policy.
Upload limit	Set the maximum upstream bandwidth for traffic that matches the policy.
Priority	Enter a number between 1 and 6 to set the priority for traffic that matches this policy. The lower the number, the higher the priority.
	Traffic with a higher priority is given bandwidth before traffic with a lower priority.
Close	Click this button to exit this screen without saving.
Create	Click this button to save your changes and close the screen.

 Table 97
 Site-wide > Configure > Firewall > Routing: Policy Route/Traffic Shaping: Add/Edit (continued)

8.3.3.2 Static Route

Click the Add button in the Static Route section of the Site-wide > Configure > Firewall > Routing: Static Route screen to open the following screen.

Figure 123 Site-wide > Configure > Firewall > Routing: Static Route

Static Route					
Subnet		Next Hop Type	Next Hop	Metric(0-127)	Description
	× *	IP Address 💌		× 1	×* ×
+ Add					

	<u>.</u>	a r		D 11	<u></u>
Table 98	Site-wide >	Contigure >	Firewall >	Routina:	Static Route
	0.1.0 1.1.0.0	00			0.000

LABEL	DESCRIPTION
Subnet	Enter an IP subnet mask. The route applies to all IP addresses in the subnet.
Next Hop Type	Select IP Address or Interface to specify if you want to send all traffic to the gateway or interface.
Next Hop	Enter the IP address of the next-hop gateway.
Metric (0-127)	Metric represents the "cost" of transmission for routing purposes.
	IP routing uses hop count as the measurement of cost, with a minimum of 1 for directly connected networks. Enter a number that approximates the cost for this link. The number need not be precise, but it must be $0 - 127$. In practice, 2 or 3 is usually a good number.
Description	This is the descriptive name of the static route.
	Click this icon to remove a static route.
Add	Click this button to create a new static route.

8.3.3.3 WAN Load Balancing

Go to Site-wide > Configure > Firewall > Routing: WAN Load Balancing to configure WAN load balancing.

By default, the Nebula Device adds all WAN interfaces to a load balancing group, and balances the traffic load between interfaces based on their respective weights (upload bandwidth). An interface with a larger weight gets more chances to transmit traffic than an interface with a smaller weight.

For example, if the weight ratio of WAN 1 and WAN 2 interfaces is 2:1, the Nebula Device chooses WAN 1 for two sessions' traffic and WAN 2 for one session's traffic in each round of three new sessions.

Figure 124 Site-wide > Configure > Firewall > Routing: WAN Load Balancing

WAN Load Balancing		
Weight Round Robin ()	Load balancing interfaces: wan2	
Backup interface	• •	
	want •	

The following table describes the labels in this section.

Table 99	Site-wide >	Configure >	> Firewall >	Routing:	WAN Load	Balancing
		0		0		<u> </u>

LABEL	DESCRIPTION
Weight Round Robin	Displays the WAN interfaces that are in the WAN load balancing group.
Backup interface	Select this to assign one WAN interface as the backup interface.
	The backup interface is removed from the WAN load balancing group, and handles all traffic if all load balancing interfaces are down.

8.3.4 NAT

The NAT summary screen provides a summary of all NAT rules and their configuration. In addition, this screen allows you to create new NAT rules and edit and delete existing NAT rules.

Note: When adding a NAT rule, based on the NAT setting NCC will automatically add the incoming security policy (firewall) rule.

To access this screen, click **Site-wide** > **Configure** > **Firewall** > **NAT**. The following screen appears, providing a summary of the existing NAT rules.

Enable Upli	ink	Protocol	Public IP			Public Port		LAN IP		Local Port	
¢ 1 ☑ wa	n1 🔻	Both 💌	Any		× *		× *		× *		
+ Add								_			
INAT											
C Enable											
ame	4	IN_		*							
ublic IP			×	*							
AN IP			×	*							
plink	1	van1	×								
lowed inbound		Enal	ole Protocol			Local Port			Remote IPs		
		¢\$> 1 🔽	Both		•			*	any	× *	
		1.0.11									

Figure 125 Site-wide > Configure > Firewall > NAT

LABEL	DESCRIPTION
Virtual Server	
¢∲	Click the icon of a rule and drag the rule up or down to change the order.
Enable	Select the check box to turn on the rule. Otherwise, clear the check box to turn off the rule.
Uplink	Select the interface of the Nebula Device on which packets for the NAT rule must be received.
Protocol	Select the IP protocol to which this rule applies. Choices are: TCP, UDP, and Both.
Public IP	Enter the destination IP address of the packets received by the interface specified in this NAT rule.
	Note: To enable NAT loop-back, enter a specific IP address instead of Any in this field. NAT loop-back allows communications between two hosts on the LAN behind the Nebula Device through an external IP address,
Public Port	Enter the translated destination port or range of translated destination ports if this NAT rule forwards the packet.
LAN IP	Specify to which translated destination IP address this NAT rule forwards packets.
Local Port	Enter the original destination port or range of destination ports this NAT rule supports.

Table 100 Site-wide > Configure > Firewall > NAT

LABEL	DESCRIPTION
Allow Remote IPs	Specify the remote IP addresses that are allowed to access the public IP address. You can add multiple IP, specify a range of IP addresses (CIDR), or GEO IP (country) objects.
	Select Any to allow all IP addresses.
	Note: IP/CIDR, and GEO IP objects cannot be used at the same time.
Description	This is the descriptive name of the policy.
	Click the remove icon to delete it.
Add	Click this to create a new entry.
1:1 NAT	
Enable	Select this to turn on the rule. Otherwise, turn off the rule.
Name	Enter the name of the NAT rule. The name is used to refer to the NAT rule. You may use $1-31$ alphanumeric characters, underscores(_), or dashes (-). This value is case-sensitive.
Public IP	Enter the destination IP address of the packets received by the interface specified in this NAT rule.
LAN IP	Specify to which translated destination IP address this NAT rule forwards packets.
Uplink	Select the interface of the Security Firewall on which packets for the NAT rule must be received.
Allowed Inbound con	nections
¢	Click the icon of a rule and drag the rule up or down to change the order.
Enable	Select the check box to turn on the rule. Otherwise, clear the check box to turn off the rule.
Protocol	Select the IP protocol to which this rule applies. Choices are: TCP, UDP, and Both.
Local Port	Enter the original destination port or range of destination ports this NAT rule supports.
Remote IPs	Specify the remote IP addresses that are allowed to access the public IP address. You can add multiple IP, specify a range of IP addresses (CIDR), or GEO IP (country) objects.
	Select Any to allow all IP addresses.
	Note: IP/CIDR, and GEO IP objects cannot be used at the same time.
	Click the remove icon to delete it.
Add	Click this to create a new entry.

Table 100 Site-	wide > Contigure	> Firewall > NAT	(continued)

8.3.5 Site-to-Site VPN

A virtual private network (VPN) provides secure communications between sites without the expense of leased site-to-site lines. Use this screen to configure a VPN rule.

Note: Site-to-site VPN does not support both VPN sites behind NAT mode.

The following figure shows two routers (R1, R2) with NAT mode enabled. Site-to-site VPN between the two Firewall devices (F1, F2) is not allowed.





Click Site-wide > Configure > Firewall > Site-to-Site VPN to access this screen.

guring VPN with multiple sites is o	sumbersome. Use <u>VPN Orchestrator</u> to save your time:	
Outgoing interface	AUTO 👻	
Preferred uplink	want 👻	
Local networks	Name	Subnet Use VPN
	lan1	192.168.128.0/24
	lan2	192.168.2.0/24
	VLAN10	192.168.10.0/24
	VLAN100	192.168.100.0/24
	IPSec remote client VPN	192.168.200.0/24
		Save or Cancel
/PN Area	Default 👻	(Please allow 1-2 minutes for changes to take errect
Nebula VPN enable		
Nebula VPN topology	Split tunnel (send only site-to-site traffic over the VPN)	
	Hub-and-Spoke 👻	
Branch to branch VPN		
Hubs (peers connect to)		
Area communication		
NAT traversal	O None	
	Custom NAT traversal IP	
Remote VPN participants		
	Network	Subnet
n-Nebula VPN peers		
:e-wide settings	the first and a second s	
tions in this section upply to the	Nebula gateway only.	- IPsec
habled Name	Public IP Private subnet	policy Pre-shared secret Avail
2	× * × *	× * Default
Add		

Figure 127	Site-wide > Configure	> Firewall >	Site-to-Site VPN
iguic izi			

T I I 101	<u>.</u>	0 1		
Table 101	Site-wide >	Configure	> Firewall >	Site-to-Site VPN

LABEL	DESCRIPTION
Outgoing Interface	Select the WAN interface to which the VPN connection is going.
	Select AUTO to use all available WAN interfaces to build the VPN tunnel.
Preferred uplink	Specify the primary WAN interface through which the Nebula Device forwards VPN traffic when you set Outgoing Interface to AUTO .
Local networks	This shows the local networks behind the Nebula Device.
	Note: Non-Nebula VPN peers use the first interface with a local policy. For example, both lan1 and lan2 are enabled. The first interface in the list 'lan1' will be used. Regardless of the order they are created.
Name	This shows the network name.
Subnet	This shows the IP address and subnet mask of the computer on the network.
Use VPN	Select ON to allow the computers on the network to use the VPN tunnel. Otherwise, select OFF .
VPN Area	Select the VPN area of the site.
	For details, see Section 11.4.4.2 on page 524.
Nebula VPN enable	Click this to enable or disable site-to-site VPN on the site's Nebula Device.
	If you disable this setting, the site will leave the VPN area.
Nebula VPN Topology	Click this to select a topology for the VPN area. For details on topologies, see Section 11.4.4.1 on page 523.
	Select disable to disable VPN connections for all sites in the VPN area.
Branch to branch VPN	Enable this to allow spoke sites to communicate with each other in the VPN area. When disabled, spoke sites can only communicate with hub sites.
Hubs (peers to connect to)	This field displays the hub sites that the current site is connected to, when Topology is set to Hub-and-Spoke .
	You can configure hub sites at Organization-wide > Organization-wide manage > VPN orchestrator.
Area communication	Enable this to allow the site to communicate with sites in different VPN areas within the organization.
NAT traversal	If the Nebula Device is behind a NAT router, select Custom to enter the public IP address or the domain name that is configured and mapped to the Nebula Device on the NAT router.
	Note: To allow a site-to-site VPN connection, the NAT router must have the following ports open: UDP 500, 4500.
Remote VPN participants	This shows all sites within the VPN area.
Non-Nebula VPN peers	Configure this section to add a non-Nebula gateway to the VPN area.
+ Add	Click this button to add a non-Nebula gateway to the VPN area.
Enabled	Select the check box to enable VPN connections to the non-Nebula gateway.
Name	Enter the name of the non-Nebula gateway.
Public IP	Enter the public IPv4 address or FQDN of the non-Nebula gateway.
Private subnet	Enter the IP subnet that will be used for VPN connections. The IP range must be reachable from other devices in the VPN area.
IPSec policy	Click to select a pre-defined policy or have a custom one. See Section 8.3.8.1 on page 394 for detailed information.

LABEL	DESCRIPTION
Preshared secret	Enter a pre-shared key (password). The Nebula Device and peer gateway use the key to identify each other when they negotiate the IKE SA.
Availability	Select which sites the non-Nebula gateway can connect to in the VPN area.
	Select All sites to allow the non-Nebula gateway to connect to any site in the VPN area.
	Select This site and the non-Nebula gateway can only connect to the Nebula Device in this site.
Address	Enter the address (physical location) of the device.

Table 101 Site-wide > Configure > Firewall > Site-to-Site VPN (continued)

8.3.5.1 IPsec Policy

Click the **Default** button in the **Non-Nebula VPN peers** section of the **Site-wide** > **Configure** > **Firewall** > **Site-to-Site VPN** screen to access this screen.

Figure 128 Site-wide > Configure > Firewall > Site-to-Site VPN: IPsec Policy

Custom						×
Preset		Default	-			1
Phase 1						
IKE version		IKEv1	•			
Encryption		AES128	•			
Authentication		SHA128	•			
Diffie-Hellman gro	up	DH2	•			
Lifetime (seconds)		86400	×			
Advanced						
Phase 2						
Set	Encryption	1	Authenticatio	n		
Set 1	AES128	•	SHA128		•	
Set 2	None	•	None		•	
Set 3	None	•	None		•	
PFS group		DH2	•			
Lifetime (seconds)		28800	\times			
Connectivity chec	k		×			
					Close	ж

Table 102	Sita-wida >	Configure >	Firewalls	Sita_to_Sita	VPNI · IPsec	Policy
	slie-wide >	Conigure /	riewai -	3116-10-3116	VEIN. IESEC	FUICY

LABEL	DESCRIPTION			
Preset	Select a pre-defined IPSec policy, or select Custom to configure the policy settings yourself.			
Phase1	IPSec VPN consists of two phases: Phase 1 (Authentication) and Phase 2 (Key Exchange).			
	A phase 1 exchange establishes an IKE SA (Security Association).			
IKE version	Select IKEv1 or IKEv2.			
	IKEv1 and IKEv2 applies to IPv4 traffic only. IKE (Internet Key Exchange) is a protocol used in setting up security associations that allows two parties to send data securely.			
Encryption	Select which key size and encryption algorithm to use in the IKE SA. Choices are:			
	DES – a 56-bit key with the DES encryption algorithm			
	3DES – a 168-bit key with the DES encryption algorithm			
	AES128 – a 128-bit key with the AES encryption algorithm			
	AES192 – a 192-bit key with the AES encryption algorithm			
	AES256 – a 256-bit key with the AES encryption algorithm			
	The Nebula Device and the remote IPSec router must use the same key size and encryption algorithm. Longer keys require more processing power, resulting in increased latency and decreased throughput.			
Authentication	Select which hash algorithm to use to authenticate packet data in the IKE SA.			
	Choices are SHA128 , SHA256 , SHA512 and MD5 . SHA is generally considered stronger than MD5, but it is also slower.			
	The remote IPSec router must use the same authentication algorithm.			
Diffie-Hellman group	Select which Diffie-Hellman key group (DHx) you want to use for encryption keys. Choices are:			
	DH1 – use a 768-bit random number			
	DH2 – use a 1024-bit random number			
	DH5 – use a 1536-bit random number			
	DH14 – use a 2048-bit random number			
	The longer the key, the more secure the encryption, but also the longer it takes to encrypt and decrypt information. Both routers must use the same DH key group.			
Lifetime (seconds)	Enter the maximum number of seconds the IKE SA can last. When this time has passed, the Nebula Device and remote IPSec router have to update the encryption and authentication keys and re-negotiate the IKE SA. This does not affect any existing IPSec SAs, however.			
Advanced	Click this to display a greater or lesser number of configuration fields.			
Mode	Set the negotiation mode.			
	Main encrypts the Nebula Device's and remote IPSec router's identities but takes more time to establish the IKE SA.			
	Aggressive is faster but does not encrypt the identities.			
Local ID	Enter an identifier used to identify the Nebula Device during authentication.			
	This can be an IP address or hostname.			

LABEL	DESCRIPTION				
Peer ID	Enter an identifier used to identify the remote IPSec router during authentication.				
	This can be an IP address or hostname.				
Phase2	Phase 2 uses the SA that was established in phase 1 to negotiate SAs for IPSec.				
Encryption	Select which key size and encryption algorithm to use in the IPSec SA. Choices are:				
	(None) – no encryption key or algorithm				
	DES – a 56-bit key with the DES encryption algorithm				
	3DES – a 168-bit key with the DES encryption algorithm				
	AES128 – a 128-bit key with the AES encryption algorithm				
	AES192 – a 192-bit key with the AES encryption algorithm				
	AES256 – a 256-bit key with the AES encryption algorithm				
	The Nebula Device and the remote IPSec router must both have at least one proposal that uses the same encryption and the same key.				
	Longer keys are more secure, but require more processing power, resulting in increased latency and decreased throughput.				
PFS group	Select whether or not you want to enable Perfect Forward Secrecy (PFS) and, if you do, which Diffie-Hellman key group to use for encryption. Choices are:				
	None – disable PFS				
	DH1 – enable PFS and use a 768-bit random number				
	DH2 – enable PFS and use a 1024-bit random number				
	DH5 – enable PFS and use a 1536-bit random number				
	DH14 – enable PFS and use a 2048-bit random number				
	PFS changes the root key that is used to generate encryption keys for each IPSec SA. The longer the key, the more secure the encryption, but also the longer it takes to encrypt and decrypt information. Both routers must use the same DH key group.				
	PFS is ignored in initial IKEv2 authentication but is used when re-authenticating.				
Lifetime (seconds)	Enter the maximum number of seconds the IPSec SA can last. Shorter life times provide better security. The Nebula Device automatically negotiates a new IPSec SA before the current one expires, if there are users who are accessing remote resources.				
Connectivity check	Enter an IP address that the Nebula Device can ping, to check whether the non- Nebula VPN peer gateway is available.				
	Note: By default, NCC will use the private subnet IP address to do connectivity check.				
Close	Click this button to exit this screen without saving.				
OK	Click this button to save your changes and close the screen.				

Table 102	Site-wide >	Configure >	· Firewall >	Site-to-Site	VPN: IPsec	Policy	(continued)
						,	

8.3.6 Remote Access VPN

Use this screen to configure the VPN client settings on the Nebula Device. This allows incoming VPN clients to connect to the Nebula Device in order to access the site's network. The clients have dynamic IP addresses and are also known as dial-in users. Only the clients can initiate the VPN tunnel.

Click Site-wide > Configure > Firewall > Remote access VPN to access this screen.

note access VPN		
WAN interface	Auto 👻	
Domain name	alpha-6b734c86d2ns-nbl.com	
IPSec VPN server		2
Client VPN subnet	×]*	
IKE version	IKEv2 👻	
DNS name servers	Specify nameserver 🔻	
Custom name servers	×	
	One IP address in one line to specify your nameserver. Maximum number of nameservers is two. Example: 19216811 192168.3710	
Upload bandwidth limit	× Mbps 0	
Policy	Default	
Authentication	Nebula Cloud Authentication 👻	
SecuExtender IKEv2 VPN configuration provision	Two-factor authentication with Captive Portal samuel.yu@zyxel.com.tw Set Send Email	
L2TP over IPSec VPN server		
Client VPN subnet	×	
DNS name servers	Specify nameserver 💌	
Custom nameservers	×**	
	One IP address in one line to specify your nameserver. Maximum number of nameservers is two. Example: 192168.11 192168.3710	
Secret	©_*	
Authentication	Nebula Cloud Authentication	
Policy	Default	
VPN provision script ()		

Figure 129 Site-wide > Configure > Firewall > Remote access VPN

Table 102	Cite wide >	Configuras		Damata	
	Me-W/de >	· Connonre .		Remole	
		Connigoro	1 II O W OII F	KOIIIOIO	

LABEL	DESCRIPTION
WAN interface	Select the WAN interface which VPN users connect to.
Domain name	This displays the domain name that maps to a WAN interface IP address.
	Note: The mapping priority is WAN1, WAN2.
	This field is available only when you select AUTO in the WAN interface field.
IPsec VPN server	Select this to enable the IPsec VPN server.
Client VPN subnet	Specify the IP addresses that the Nebula Device uses to assign to the VPN clients.
IKE version	Select IKEv1 or IKEv2.
	IKE (Internet Key Exchange) is a protocol used in setting up security associations that allows two parties to send data securely.
DNS name servers	Specify the DNS servers to assign to the remote users. Or select Specify nameserver to enter a static IP address.
Custom nameservers	If you select Specify nameserver in the DNS name servers field, manually enter the DNS server IP addresses.
Upload Bandwidth Limit	This field is available only if you select IKEv2 in IKE version . Enter the maximum traffic load between VPN clients, 1 – 100 Mbps.
Secret	Enter the pre-shared key (password) which is used to set up the VPN tunnel. The password should be 8 – 32 characters.
Policy	Configure custom VPN tunnel settings.
	For details, see Section 8.3.6.1 on page 374.
Authentication	Select how the Nebula Device authenticates a remote user before allowing access to the VPN tunnel.
Two-factor authentication with Captive	Select this to require two-factor authentication for a user to access the Nebula Device through VPN.
Portal	Note: Two-factor authentication is only supported with Zyxel SecuExtender IPSec client.
SecuExtender IKEv2 VPN configuration provision	Enter the email address to send new IKEv2 Remote Access VPN configuration file to VPN client. Then click Send Email . The VPN client needs to replace the IPSec VPN client configuration by importing the configuration file.
L2TP over IPSec VPN server	Select this to enable the L2TP over IPSec VPN server.
Client VPN subnet	Specify the IP addresses that the Nebula Device uses to assign to the VPN clients.
DNS name servers	Specify the DNS servers to assign to the remote users. Or select Specify nameserver to enter a static IP address.
Custom nameservers	If you select Specify nameserver in the DNS name servers field, manually enter the DNS server IP addresses.
Secret	This field is available only if you select IKEv1 in IKE version . Enter the pre-shared key (password) which is used to set up the VPN tunnel. The password should be 8 – 32 characters.
Authentication	Select how the Nebula Device authenticates a remote user before allowing access to the VPN tunnel.

LABEL	DESCRIPTION
Policy	Configure custom VPN tunnel settings.
	For details, see Section 8.3.6.1 on page 374.
VPN provision script	Send an email to help automatically configure VPN settings on client devices so that the devices can remotely access this Nebula Device. The email contains two scripts; one for mac OS and iOS devices, and one for Windows 8 and Windows 10 devices.
	You can send the email to one or more email addresses.
	 If Authentication is set to Nebula Cloud Authentication, the default email address list contains all authorized VPN user email addresses and your email address.
	 If Authentication is set to AD and RADIUS Authentication, the default email address list contains your user email address.
	This field is available only when you select L2TP over IPSec client in the Client VPN server field.

Table 103 Site-wide > Configure > Firewall > Remote access VPN (continued)

8.3.6.1 Remote Access VPN > Custom VPN Policy

Click **Default** in **Site-wide** > **Configure** > **Firewall** > **Remote access VPN** > **Policy** to open the following screen.

Custom				×
Preset		Default	•	
Phase 1				
IKE version		IKEv1	•	
Encryption		3DES	•	
Authentication		SHA128	•	
Diffie-Hellman g	group	DH2	•	
Lifetime (secon	ds)	86400	×	
Advanced				
Phase 2				
Set	Encrypti	on	Authentication	
Set 1	3DES		▼ SHA128	•
Set 2	None		▼ None	•
Set 3	None		▼ None	•
PFS group		None		
		None	•	
Litetime (second	ds)	86400	×	
				Close OK

Figure 130 Site-wide > Configure > Firewall > Remote access VPN: Default

Table 104 Site-wide > Configure > Firewall > Remote access VPN: Default

LABEL	DESCRIPTION
Custom	
Preset	Select a pre-defined IPSec policy, or select Custom to configure the policy settings yourself.
Phase 1	

LABEL	DESCRIPTION
Encryption	Select which key size and encryption algorithm to use in the IPSec SA. Choices are:
	(None) – no encryption key or algorithm
	DES – a 56-bit key with the DES encryption algorithm
	3DES – a 168-bit key with the DES encryption algorithm
	AES128 – a 128-bit key with the AES encryption algorithm
	AES192 – a 192-bit key with the AES encryption algorithm
	AES256 – a 256-bit key with the AES encryption algorithm
	The Nebula Device and the remote IPSec router must both have at least one proposal that use the same encryption and the same key.
	Longer keys are more secure, but require more processing power, resulting in increased latency and decreased throughput.
Authentication	Select which hash algorithm to use to authenticate packet data in the IKE SA.
	Choices are SHA128 , SHA256 , SHA512 and MD5 . SHA is generally considered stronger than MD5, but it is also slower.
	The remote IPSec router must use the same authentication algorithm.
Diffie-Hellman group	Select the Diffie-Hellman key group (DHx) you want to use for encryption keys. Choices are:
	DH1 – use a 768-bit random number
	DH2 – use a 1024-bit random number
	DH5 – use a 1536-bit random number
	DH14 – use a 2048-bit random number
	The longer the key, the more secure the encryption, but also the longer it takes to encrypt and decrypt information. Both routers must use the same DH key group.
Lifetime (seconds)	Enter the maximum number of seconds the IPSec SA can last. Shorter life times provide better security. The Nebula Device automatically negotiates a new IPSec SA before the current one expires, if there are users who are accessing remote resources.
Phase 2	
Set	This shows the index number of the IPSec policy.
Encryption	Select which key size and encryption algorithm to use in the IPSec SA. Choices are:
	(None) – no encryption key or algorithm
	DES – a 56-bit key with the DES encryption algorithm
	3DES – a 168-bit key with the DES encryption algorithm
	AES128 – a 128-bit key with the AES encryption algorithm
	AES192 – a 192-bit key with the AES encryption algorithm
	AES256 – a 256-bit key with the AES encryption algorithm
	The Nebula Device and the remote IPSec router must both have at least one proposal that use the same encryption and the same key.
	Longer keys are more secure, but require more processing power, resulting in increased latency and decreased throughput.

 Table 104
 Site-wide > Configure > Firewall > Remote access VPN: Default (continued)

LABEL	DESCRIPTION
Authentication	Select which hash algorithm to use to authenticate packet data in the IKE SA.
	Choices are None , SHA128 , SHA256 , SHA512 and MD5 . SHA is generally considered stronger than MD5, but it is also slower.
	The remote IPSec router must use the same authentication algorithm.
PFS group	Select whether or not you want to enable Perfect Forward Secrecy (PFS) and, if you do, which Diffie-Hellman key group to use for encryption. Choices are:
	None – disable PFS
	DH1 – enable PFS and use a 768-bit random number
	DH2 – enable PFS and use a 1024-bit random number
	DH5 – enable PFS and use a 1536-bit random number
	DH14 – enable PFS and use a 2048 bit random number
	PFS changes the root key that is used to generate encryption keys for each IPSec SA. The longer the key, the more secure the encryption, but also the longer it takes to encrypt and decrypt information. Both routers must use the same DH key group.
	PFS is ignored in initial IKEv2 authentication but is used when re-authenticating.
Lifetime (seconds)	Enter the maximum number of seconds the IPSec SA can last. Shorter life times provide better security. The Security Firewall automatically negotiates a new IPSec SA before the current one expires, if there are users who are accessing remote resources.
Close	Click this button to exit this screen without saving.
ОК	Click this button to save your changes and close the screen.

Table 101	Cite wide >	Configuras	- Fires well >	Demote areas	VDNI Defeudt	
	Me-Wide >	· CONHOLITE 2	> FILEW/(111 >	Remole drivess	VPINIDEIGUII	CONTINUEOU
		Connigoro	1 II O W OII F	1011010 accoss		

8.3.7 Security Policy

By default, a LAN user can initiate a session from within the LAN and the Nebula Device allows the response. However, the Nebula Device blocks incoming traffic initiated from the WAN and destined for the LAN. Use this screen to configure firewall rules for outbound traffic, application patrol and content filtering, schedule profiles and port forwarding rules for inbound traffic.

Click Site-wide > Configure > Firewall > Security policy to access this screen.

Note: The Nebula Device has the following hidden default firewall rules: LAN to WAN is allowed, WAN to LAN is blocked.

Enabled Name	Action	Application Patrol / Content Filtering Policy	Protocol	Source	D	Destination	Dst Port
	* Deny •		Any	▼ IP, IP rang	e, CIDR, or FQDN	IP, IP range, CIDR, or FQDN	L. Any
	* Allow •		Any	▼ IP, IP rang	e, CIDR, or FQDN	IP, IP range, CIDR, or FQDN	Any
nplicit allow rules 🔺							
	Allow		Any	lan1_192.168.1 lan2_192.168.	10/24 A	ληγ	Any
	Allow		Any	lan1_192.168.* lan2_192.168	1.0/24 2.0/24 D	Device	Any
nplicit deny rule				1000	L.O. a. t		
	Deny		Any	Any	A	Any	Any
+ Add							
+ Add nomaly Detection and Prevention Enable Anomaly Detection and Prevent	ion 💽						
+ Add nomaly Detection and Prevention Enable Anomaly Detection and Prevent Ission Control	ion 💽						
+ Add nomaly Detection and Prevention Enable Anomaly Detection and Prevent ession Control UDP Session Time Out:	tion ()	× (1-28800 sec	ond)				
+ Add nomaly Detection and Prevention Enable Anomaly Detection and Prevent ession Control UDP Session Time Out: Session per Host	tion 60	× (1-28800 sec	ond) unlimited)				
+ Add nomaly Detection and Prevention Enable Anomaly Detection and Prevent ssion Control UDP Session Time Out: Session per Host :hedule profiles	60 1000	× (1-28800 sec × (0-8192, 0 is r	ond) unlimited)				

Figure 131 Site-wide > Configure > Firewall > Security policy

Table 105	Site-wide > Configure	> Firewall >	Security policy
-----------	-----------------------	--------------	-----------------

LABEL	DESCRIPTION
Security policy	•
¢€	Click the icon of a rule and drag the rule up or down to change the order.
Enabled	Select the check box to turn on the rule. Otherwise, clear the check box to turn off the rule.
Name	Enter the name of the security policy.
Action	Select what the Nebula Device is to do with packets that match this rule.
	Select Deny to silently discard the packets without sending a TCP reset packet or an ICMP destination-unreachable message to the sender.
	Select Allow to permit the passage of the packets.

LABEL	DESCRIPTION
Application Patrol/ Content Filtering	Click the "+" to add an Application Patrol or Content Filtering profile. The firewall takes the action set in the profile when traffic matches the profile's policy.
Policy	Application Patrol manages the use of various applications on the network. It manages general protocols (for example, HTTP and FTP) and instant messenger (IM), peer-to-peer (P2P), Voice over IP (VoIP), and streaming (RSTP) applications. You can even control the use of a particular application's individual features (like text messaging, voice, video conferencing, and file transfers). See Section 8.3.7.1 on page 380 for how to create an Application Patrol profile.
	Content Filtering controls access to specific web sites or web content. See Section 8.3.7.2 on page 381 for how to create a Content Filtering profile.
Protocol	Select the IP protocol to which this rule applies. Choices are: ICMP, TCP, UDP, TCP and UDP and Any.
Source	Specify the source IP addresses (LAN interface / country) to which this rule applies. You can add multiple IP, CIDR, FQDN, GEO IP (country) objects, or a single FQDN object by pressing 'Enter', or enter a new IP address by clicking Add . Enter any to apply the rule to all IP addresses.
	Note: IP/CIDR, FQDN, and GEO IP objects cannot be used at the same time. Multiple FQDNs are not supported. The IP FQDN does NOT support wildcards.
Destination	Specify the destination IP addresses (LAN interface / country) or subnet to which this rule applies. You can add multiple IP, CIDR, GEO IP (country) objects or a single FQDN object by pressing 'Enter', or enter a new IP address by clicking Add . Enter any to apply the rule to all IP addresses.
	Note: IP/CIDR, FQDN, and GEO IP objects cannot be use at the same time. Multiple FQDNs are not supported.
Dst Port	Specify the destination ports to which this rule applies. You can specify multiple ports by pressing 'Enter', or enter a new port by clicking Add . Enter any to apply the rule to all ports.
User	Select the External User Group name configured in Site-wide > Configure > Firewall > Firewall settings.
Schedule	Select the name of the schedule profile that the rule uses. Always means the rule is active at all times if enabled.
Description	Enter a descriptive name of up to 60 printable ASCII characters for the rule.
Log	Select whether to have the Nebula Device generate a log (ON) or not (OFF) when traffic matches the profile's policy.
	Note: By default, Log is ON when the Action field is Deny. Log is OFF when the Action field is Allow .
	Click this icon to remove the rule.
Implicit allow rules	This shows the system generated Allow rules.
	 LAN interface / remote access VPN to Any Guest interface to WAN interface LAN interface / remote access VPN to Nebula Device Guest interface to Nebula Device TCP (TCP:443, 80, 53) Guest interface to Nebula Device UDP (UDP:53)
Implicit deny rule	This shows the system generated Deny rule.
	Any to Any
Add	Click this button to create a new rule.
Anomaly Detection ar	nd Prevention

T. L.L. 105	C11			C		/ I [*] I	•
100 I 105	>ite-wide >	Configure >	+irewali >	Security	policy	(continued))

LABEL	DESCRIPTION
Enable Anomaly Detection and Prevention	Select this to enable traffic anomaly and protocol anomaly detection and prevention.
Session Control	
UDP Session Time Out	Set how many seconds the Nebula Device will allow a UDP session to remain idle (without UDP traffic) before closing it.
Session per Host	Use this field to set a common limit to the number of concurrent NAT/Security Policy sessions each client computer can have.
	If only a few clients use peer to peer applications, you can raise this number to improve their performance. With heavy peer to peer application use, lower this number to ensure no single client uses too many of the available NAT sessions.
Schedule profiles	
Schedule name	This shows the name of the schedule profile and the number of the outbound rules that are using this schedule profile.
2	Click this icon to change the profile settings.
.	Click this icon to remove the profile.
Add	Click this button to create a new schedule profile. See Section 8.3.7.3 on page 384 for more information.

Table 105 Site-wide > Configure > Firewall > Security policy (continued)

8.3.7.1 Add an Application Patrol Profile

Application patrol provides a convenient way to manage the use of various applications on the network. It manages general protocols (for example, HTTP and FTP) and instant messenger (IM), peer-to-peer (P2P), Voice over IP (VoIP), and streaming (RSTP) applications. You can even control the use of a particular application's individual features (like text messaging, voice, video conferencing, and file transfers).

An application patrol profile is a group of categories of application patrol signatures. For each profile, you can specify the default action the Nebula Device takes once a packet matches a signature (forward, drop, or reject a service's connections and/or create a log alert).

Click "+" in the **Application Patrol/Content Filtering Policy** field of the **Organization-wide** > **Organization-wide** manage > **Security profile sync** screen to access this screen. Use the application patrol profile screens to customize action and log settings for a group of application patrol signatures.

Figure 132 Organization-wide > Organization-wide manage > Security profile sync > Application patrol: Add an Application Profile

Add profile		×
Name Description (Optional)		×
Application Management Enabled Category	Application	Action
1 🗹 Antivirus 🔻	All 😵	Reject 🝵
+ Add Search Application	•	
		Close Create

The following table describes the labels in this screen.

Table 106Organization-wide > Organization-wide manage > Security profile sync > Application patrol:Add an Application Profile

LABEL	DESCRIPTION					
Name	Enter a name for this profile for identification purposes.					
Description (Optional)	Enter a description for this profile.					
Log	Select whether to have the Nebula Device generate a log (ON) or not (OFF) by default when traffic matches an application signature in this category.					
Application Managem	nent					
Enabled	Select the check box to turn on the rule. Otherwise, clear the check box to turn off the rule.					
Category	Select an application category.					
Application	Select All or select an application within the category to apply the policy.					
Action	Select the default action for the applications selected in this category.					
	Reject – the Nebula Device drops packets that matches these application signatures and sends notification to clients.					
1	Click this icon to remove the entry.					
Add	Click this button to create a new application category and set actions for specific applications within the category.					
	Enter a name to search for relevant applications and click Add to create an entry.					
Close	Click this button to exit this screen without saving.					
Create	Click this button to save your changes and close the screen.					

8.3.7.2 Add a Content Filtering Profile

Click "+" in the Application Patrol/Content Filtering Policy section of the Organization-wide > Organization-wide manage > Security profile sync > Content filtering: Create content filtering profile screen to access this screen.

Figure 133 Organization-wide > Organization-wide manage > Security profile sync > Content filtering: Create content filtering profile

Create content filtering prot	file		×
Add profile			
Name			×*
Description (Optional)			×
Log			
Enabled			
Block Web Pages			
Action for Unrated Web Pages	Warn		
Action When Service is Unavailable	Warn 🔻		
Block Category			
Templates	Parental control 💌		
Test URL	Enter a url to know website category	×	Test
Search category	✓ Category list		×
Block web site	There are no block web site rules defined for this site.		
	+ Add		
Allow web site	There are no allow web site rules defined for this site.		
	+ Add		
		Cancel	Create

The following table describes the labels in this screen.

Table 107Organization-wide > Organization-wide manage > Security profile sync > Content filtering:Create content filtering profile

LABEL	DESCRIPTION
Name	Enter a name for this profile for identification purposes.
Description (Optional)	Enter a description for this profile.
Log	Select whether to have the Nebula Device generate a log (ON) or not (OFF) by default when traffic matches an application signature in this category.
DNS Content Filtering	Select whether to enable DNS content filtering, in addition to web content filtering. The DNS Content Filter allows the Nebula Device to block access to specific websites by inspecting DNS queries made by users on your network.
Block Web Pages	

LABEL	DESCRIPTION
Action for Unrated Web Pages	Select Pass to allow users to access web pages that the external web filtering service has not categorized.
	Select Block to prevent users from accessing web pages that the external web filtering service has not categorized. When the external database content filtering blocks access to a web page, it displays the denied access message that you configured in the Content Filter General screen along with the category of the blocked web page.
	Select Warn to display a warning message before allowing users to access web pages that the external web filtering service has not categorized.
Action When Service is Unavailable	Select Pass to allow users to access any requested web page if the external content filtering database is unavailable.
	Select Block to block access to any requested web page if the external content filtering database is unavailable.
	Select Warn to display a warning message before allowing users to access any requested web page if the external content filtering database is unavailable.
	The following are possible causes for the external content filtering server not being available:
	• There is no response from the external content filtering server within the time period specified in the Content Filter Server Unavailable Timeout field.
	• The Nebula Device is not able to resolve the domain name of the external content filtering database.
	• There is an error response from the external content filtering database. This can be caused by an expired content filtering registration (External content filtering's license key is invalid").
Block Category	
Templates	Select the block category. Choices are Parental control, Productivity and Custom.
Test URL	You can check which category a web page belongs to. Enter a web site URL in the text box.
	When the content filter is active, you should see the web page's category. The query fails if the content filter is not active.
	Content Filtering can query a category by full URL string (for example, http:// www.google.com/picture/index.htm), but HTTPS Domain Filter can only query a category by domain name ('www.google.com'), so the category may be different in the query result. URL to test displays both results in the test.
Search category	Click to display or hide the category list.
	These are categories of web pages based on their content. Select categories in this section to control access to specific types of Internet content.
Custom block web site	Sites that you want to block access to, regardless of their content rating, can be blocked by adding them to this list.
	Enter host names such as www.bad-site.com into this text field. Do not enter the complete URL of the site – that is, do not include "http://". All sub-domains are also blocked. For example, entering "bad-site.com" also blocks "www.badsite.com", "partner.bad-site.com", "press.bad-site.com", and so on. You can also enter just a top level domain. For example, enter .com to block all .com domains.
	Use up to 127 characters $(0 - 9 a - z)$. The casing does not matter.
Add	Click this button to create a new application category and set actions for specific applications within the category.
	Click this icon to remove the entry.

Table 107Organization-wide > Organization-wide manage > Security profile sync > Content filtering:Create content filtering profile (continued)

Table 107	Organization-wide >	Organization-wide manage > Security profile sync > Content filtering:
Create co	ntent filtering profile	continued)

LABEL	DESCRIPTION
Custom allow web site	Sites that you want to allow access to, regardless of their content rating, can be allowed by adding them to this list.
	Enter host names such as www.good-site.com into this text field. Do not enter the complete URL of the site – that is, do not include "http://". All sub-domains are allowed. For example, entering "zyxel.com" also allows "www.zyxel.com", "partner.zyxel.com", "press.zyxel.com", and so on. You can also enter just a top level domain. For example, enter .com to allow all .com domains.
	Use up to 127 characters (0 – 9 a – z). The casing does not matter.
Add	Click this button to create a new application category and set actions for specific applications within the category.
1	Click this icon to remove the entry.
Cancel	Click this button to exit this screen without saving.
Create	Click this button to save your changes and close the screen.

8.3.7.3 Create a New Schedule

Click the Add button in the Schedule Profiles section of the Site-wide > Configure > Firewall > Security policy > Schedule profiles screen to access this screen.

Figure 134 Site-wide > Configure > Firewall > Security policy > Schedule profiles: Create new schedule

Create new s	chedule	è											×
Local time zone: (You Name: NewSchedule	ı can set th	is on <u>Gen</u>	eral setting	<u>a</u>)		Temple X Alwar	ate: ys on						•
Day Av	vailability												
Sunday	00:00	02:00	04:00	06:00	08:00	10:00	12:00	14:00	16:00	18:00	20:00	22:00	24:00
Monday	00:00	02:00	04:00	06:00	08:00	10:00	12:00	14:00	16:00	18:00	20:00	22:00	24:00
Tuesday	00:00	02:00	04:00	06:00	08:00	10:00	12:00	14:00	16:00	18:00	20:00	22:00	24:00
Wednesday 🧲	00:00	02:00	04:00	06:00	08:00	10:00	12:00	14:00	16:00	18:00	20:00	22:00	24:00
Thursday 🦲	00:00	02:00	04:00	06:00	08:00	10:00	12:00	14:00	16:00	18:00	20:00	22:00	24:00
Friday 🧲	00:00	02:00	04:00	06:00	08:00	10:00	12:00	14:00	16:00	18:00	20:00	22:00	24:00
Saturday 🧲	00:00	02:00	04:00	06:00	08:00	10:00	12:00	14:00	16:00	18:00	20:00	22:00	24:00
												Close	

Taile 1.00				
10016 108	Site-wide > Configure > Fir	rewall > security policy >	> schedule profiles: Cre	ate new schedule

LABEL	DESCRIPTION
Name	Enter a descriptive name for this schedule for identification purposes.
Templates	Select a pre-defined schedule template or select Custom schedule and manually configure the day and time at which the associated firewall outbound rule is enabled.
Day	This shows the day of the week.
Availability	Click On to enable the associated rule at the specified time on this day. Otherwise, select Off to turn the associated rule off at the specified time on this day.
	Specify the hour and minute when the schedule begins and ends each day.
Close	Click this button to exit this screen without saving.
Add	Click this button to save your changes and close the screen.

8.3.8 Security Service

Use this screen to enable or disable the features available in the security pack for your Nebula Device, such as content filtering, Intrusion Detection and Prevention (IDP) and/or anti-virus. As to application patrol, go to the **Firewall** screen to configure it since you need to have a firewall rule for outbound traffic.

Content filtering allows you to block access to specific web sites. It can also block access to specific categories of web site content. IDP can detect malicious or suspicious packets used in network-based intrusions and respond instantaneously. Anti-virus helps protect your connected network from virus/spyware infection.

Click Site-wide > Configure > Firewall > Security service to access this screen.

- Note: Packet inspection signatures examine packet content for malicious data. Packet inspection applies to OSI (Open System Interconnection) layer-4 to layer-7 contents. You need to subscribe for IDP service in order to be able to download new signatures.
- Note: If Security Profile Sync (SPS) is enabled, you cannot configure security settings on this screen. For details, see Section 11.4.5 on page 527.

Content filtering Model list	
Drop connection when there is an HTTPS connection with SSL v3(or previous version)	
Denied Access Message	Web access is restricted. Please contact the administrator
Redirect URL	×
	There are no content filtering rules defined for this site.
	.+ Add
Application Patrol Model list	
Application profiles	There are no profiles defined for this alte.
	+ Add
IP Exception Model int	
Enabled Source IP	Destination IP Description
+ Add	
DNS/URL Threat Filter Model list	
Signature Information	Current Version.
	Released Date: - (UTC+08:00)
Log	
DNS Threat Filter	
DNS Threat Filter policy	Redirect 👻
DNS Threat Filter Redirect IP	Default 👻
URL Threat Filter	
URL Threat Filter policy	Block
URL Threat Filter Denied Access	Web access is perioded. Blasse control the administrator
Message URL Threat Filter Redirect URL	THE WOOD IS HER PROVIDE FRANK OF THE WITHOUT MALE.
Toth Thread Cotogoni	×
lest medic colegoly	X Ind
Category list	Anonymizers Browser Exploits Molicious Downloads Molicious Sites S Molicious Source State Molicious Sites S Molicious Source Sour
Block list	Ask Question
	FQDN(support wildbard)
Allow list	
	FQDN(support wildcard)
URI. Threat Filter external block list	Enabled Name External D8 Description
	+ Add
Schedule update	External DB schedule update
	Daily 👻

Figure 135 Site-wide > Configure > Firewall > Security service

F IP Reputation Model list					
Signature information	Ourraph Varsion				
	Released Date: - (UTC+08)	50)			
Enabled					
Log					
Policy	Block 👻				
Threat level threshold	Medium and above 👻				
Test Category			Y Test		
Category list 🚺	Anonymous Proxies Negotive Reputation	Denial of Service Scanners Web Attacks	Spam Sources		
Block list	BotNets				
DRAM H9	1P or CIDR			3	×
Allow list	IP or OIDR				
				ž	Ŕ
External block list	Enabled Name	External DB	Description		
		×	× *	×	7
	- add				
Schedule undote	Estarcal DB school do undeter				
n norther was well	Belle				
	Losny				
	0300				
F Anti-Malware Modellist					
Signature Information	Current Version.				
	Released Date (UTC+08)	00)			
Enabled	•				
Lag	•				
Scan mode	Stream mode Express mode ()				
Cloud Query					
Block list	Hile Types				
10-10-10-10-10-10-10-10-10-10-10-10-10-1	File Pattern				
Allow list					
	File Pattern				
Sandboxing Model list					
Emphand					
Loo					
209					
Folicy	Allow				
Inspect selected downloaded files ()					
File submission options	ZIP Archives (zip) Executor PDF Document (pdf) RTF File Types	bles (exe) OMS Office Document	s (doc) 🧿 👘 Macromedia Flash Data	(awf) O	

Intrusion Prevention System	(IPS) <u>Model list</u>
Signature Information	Current Version:
	Released Date: - (UTC+08:00)
Detection	
Prevention	
	finitive or Cancel
	(Please allow 1-2 minutes for changes to take effect.)

LABEL	DESCRIPTION
Content Filtering	
Drop connection when HTTPS connection with SSL V3 or previous version	Select On to have the Nebula Device block HTTPS web pages using SSL V3 or a previous version.
Denied Access Message	Enter a message to be displayed when content filter blocks access to a web page. Use up to 127 characters (0–9 a–z A–Z;/?:@&=+\$\!~*'()%,"). For example, "Access to this web page is not allowed. Please contact the network administrator".
	It is also possible to leave this field blank if you have a URL specified in the Redirect URL field. In this case if the content filter blocks access to a web page, the Nebula Device just opens the web page you specified without showing a denied access message.
Redirect URL	Enter the URL of the web page to which you want to send users when their web access is blocked by content filter. The web page you specify here opens in a new frame below the denied access message.
	Use "http://" or "https://" followed by up to 262 characters (0–9 a–z A–Z;/?:@&=+\$\!~*'()%). For example, http://192.168.1.17/blocked access.
Name	This shows the name of this content filtering profile.
Description	This shows the description for this profile.
2	Click this icon to change the profile settings.
1	Click this icon to remove the profile.
Add	Click this to create a content filtering profile. See Section 8.3.7.2 on page 381 for more information.
Application Patrol	
Application profiles	
Name	This shows the name of this Application Patrol profile.
Description	This shows the description for this profile.
2	Click this icon to change the profile settings.
1	Click this icon to remove the profile.
Add	Click this to create an Application Patrol profile. See Section 8.3.8.2 on page 397 for more information.
IP Exception	
Enabled	Select the check box to enable IP Exception.
	IP addresses listed here are not checked by security services.
Source IP	This field displays the source IP address of incoming traffic. It displays any if there is no restriction on the source IP address.

LABEL	DESCRIPTION		
Destination IP	This field displays the destination IP address of incoming traffic. It displays any if there is no restriction on the destination IP address.		
Description	Enter a description for this profile.		
1	Click this icon to remove the entry.		
Add	Click this button to create a new entry.		
DNS/URL Threat Filter	DNS filtering inspects DNS queries made by clients on your network and compares the queries against a database of blocked or allowed Fully Qualified Domain Names (FQDNs). If a user attempts to connect to a suspect site, where the DNS query packet contains an FQDN with a bad reputation, then a DNS query is sent from the user's computer and detected by the DNS Filter. The Nebula Device DNS filter will either drop the DNS query or reply to the user with a fake DNS response using the default dnsft.cloud.zyxel.com IP address (where the user will see a "Web Page Blocked!" page) or a custom IP address.		
	When you enable the URL Threat filtering service, your Nebula Device downloads signature files that contain known URL Threat domain names and IP addresses. The Nebula Device will also access an external database, Cloud Query, that has millions of web sites categorized based on content. You can have the Nebula Device allow, block, warn and/ or log access to web sites or hosts based on these signatures and categories.		
Signature information	This shows the Current Version of the DNS/URL threat definition and the Released Date .		
Log	Select whether to have the Nebula Device generate a log when the policy is matched to the criteria listed above.		
DNS Threat Filter	Select On to turn on the rule. Otherwise, select Off to turn off the rule.		
DNS Threat Filter Policy	Select Pass to have the Nebula Device allow the DNS query packet and not reply with a DNS reply packet containing a default or custom-defined IP address.		
	Select Redirect to have the Nebula Device reply with a DNS reply packet containing a default or custom-defined IP address.		
DNS Threat Filter Redirect IP	Enter the IP address to have the Nebula Device reply with a DNS reply packet containing a default or custom-defined IP address when a DNS query packet contains an FQDN with a bad reputation. The default IP is the dnsft.cloud.zyxel.com IP address. If you select a custom-defined IP, then enter a valid IPv4 address in the text box.		
URL Threat Filter	Select On to turn on the rule. Otherwise, select Off to turn off the rule.		
URL Threat Filter Policy	Select Pass to allow users to access web pages that the external web filtering service has not categorized.		
	Select Block to prevent users from accessing web pages that the external web filtering service has not categorized. When the external database content filtering blocks access to a web page, it displays the denied access message that you configured in the Content Filter General screen along with the category of the blocked web page.		
	Select Warn to display a warning message before allowing users to access web pages that the external web filtering service has not categorized.		
URL Threat Filter Denied Access Message	Enter a message to be displayed when content filter blocks access to a web page. Use up to 127 characters (0–9 a–z A–Z;/?: $@$ &=+ $\!~*'()$ %,"). For example, "Access to this web page is not allowed. Please contact the network administrator".		
	It is also possible to leave this field blank if you have a URL specified in the Redirect URL field. In this case if the content filter blocks access to a web page, the Nebula Device just opens the web page you specified without showing a denied access message.		
URL Threat Filter Redirect URL	Enter the URL of the web page to which you want to send users when their web access is blocked by content filter. The web page you specify here opens in a new frame below the denied access message.		
	Use "http://" or "https://" followed by up to 262 characters (0–9 a–z A–Z;/?:@&=+\$\ !~*'()%). For example, http://192.168.1.17/blocked access.		

Table 109	Site-wide >	Configure >	Firewall >	Security	service	(continued)
-----------	-------------	-------------	------------	----------	---------	-------------

LABEL	DESCRIPTION
Test Threat Category	Enter a URL using http://domain or https://domain and click the Test button to check if the domain belongs to a URL threat category.
Category List	These are categories of web pages based on their content. Select categories in this section to control access to specific types of Internet content.
Block list	Sites that you want to block access to, regardless of their content rating, can be blocked by adding them to this list.
	Enter host names such as www.bad-site.com into this text field. Do not enter the complete URL of the site – that is, do not include "http://". All sub-domains are also blocked. For example, entering "bad-site.com" also blocks "www.badsite.com", "partner.bad-site.com", "press.bad-site.com", and so on. You can also enter just a top level domain. For example, enter .com to block all .com domains.
	Use up to 127 characters (0–9 a–z). The casing does not matter.
Allow list	Sites that you want to allow access to, regardless of their content rating, can be allowed by adding them to this list.
	Enter host names such as www.good-site.com into this text field. Do not enter the complete URL of the site – that is, do not include "http://". All sub-domains are allowed. For example, entering "zyxel.com" also allows "www.zyxel.com", "partner.zyxel.com", "press.zyxel.com", and so on. You can also enter just a top level domain. For example, enter .com to allow all .com domains.
	Use up to 127 characters (0–9 a–z). The casing does not matter.
URL Threat Filter external block list	The Nebula Device uses black list entries stored in a file on a web server that supports HTTP or HTTPS. The Nebula Device blocks incoming and outgoing packets from the black list entries in this file.
Enabled	Select this to have the Nebula Device block the incoming packets that come from the listed addresses in the block list file on the server.
Name	Enter an identifying name for the block list file. You can use alphanumeric and ()+/ :=?!*#@\$_%- characters, and it can be up to 60 characters long.
External DB	Enter the exact file name, path and IP address of the server containing the block list file. The file type must be 'txt'.
	For example, http://172.16.107.20/blacklist-files/myip-ebl.txt
	The server must be reachable from the Nebula Device.
Description	Enter a description of the block list file. You can use alphanumeric and ()+/:=?!*#@\$_%-characters, and it can be up to 60 characters long.
	Click this icon to remove the entry.
Add	Click this button to create a new entry.
Schedule update	The signatures for DNS Filter and URL Threat Filter are the same. These signatures are continually updated as new malware evolves. New signatures can be downloaded to the Nebula Device periodically if you have subscribed for the URL Threat filter signatures service.
	You need to create an account at myZyxel, register your Nebula Device and then subscribe for URL Threat filter service in order to be able to download new signatures from myZyxel.
	Select Daily to set the time of the day, or Weekly to set the day of the week and the time of the day.
	Schedule signature updates for a day and time when your network is least busy to minimize disruption to your network.
IP Reputation	

Table 100	Site wide >	Configura	Firewalls	Security	(convior	(continued)	
	slie-wide >	Conligure 2	> Firewali >	secon	y service	(coninuea)	1

LABEL	DESCRIPTION
Signature information	This shows the Current Version of the signature set the Nebula Device is using and the Released Date .
Enabled	Select this option to turn on IP blocking on the Nebula Device.
Log	Select this option to create a log on the Nebula Device when the packet comes from an IPv4 address with bad reputation.
Policy	Select Pass to have the Nebula Device allow the packet to go through.
	Select Block to have the Nebula Device deny the packets and send a TCP RST to both the sender and receiver when a packet comes from an IPv4 address with bad reputation.
Threat level threshold	Select the threshold threat level to which the Nebula Device will take action (High , Medium and above , Low and above).
	The threat level is determined by the IP reputation engine. It grades IPv4 addresses.
	 High: an IPv4 address that scores 0 to 20 points. Medium and above: an IPv4 address that scores 0 to 60 points. Low and above: an IPv4 address that scores 0 to 80 points.
	For example, a score of "10" will cause the Nebula Device to take action whether you set the Threat level threshold at High , Medium and above , or Low and above .
	But a score of "61" will not cause the Nebula Device to take any action if you set the Threat level threshold at Medium and above .
Test Category	Enter an IPv4 address of a website, and click the Test button to check if the website associates with suspicious activities that could pose a security threat to users or their computers.
Category list	Select the categories of packets that come from the Internet and are known to pose a security threat to users or their computers.
Block list	Sites that you want to block access to, regardless of their content rating, can be blocked by adding them to this list.
	Add the IPv4 addresses that the Nebula Device will block the incoming packets.
Allow list	Sites that you want to allow access to, regardless of their content rating, can be allowed by adding them to this list.
	Add the IPv4 addresses that the Nebula Device will allow the incoming packets.
External block list	
Enabled	Select this check box to have the Nebula Device block the incoming packets that come from the listed addresses in the block list file on the server.
Name	Enter the identifying name for the block list file. You can use alphanumeric and ()+/:=?!*#@\$_%- characters, and it can be up to 60 characters long.
External DB	Enter the file name, path and IP address of the server containing the block list file. For example, http://172.16.107.20/blacklist-files/myip-ebl.txt
Description	Enter a description of the block list file. You can use alphanumeric and ()+/:=?!*#@\$_%-characters, and it can be up to 60 characters long.
.	Click this icon to remove the entry.
Add	Click this button to create a new entry.

Table 109	Site-wide >	Configure 2	> Firewall >	Security	service	(continued)
		Configure		Jecomy	301 4100	

LABEL	DESCRIPTION
Schedule update	New IP reputation signatures can be downloaded to the Nebula Device periodically if you have subscribed for the IP reputation signatures service. You need to create an account at myZyxel, register your Nebula Device and then subscribe for IP reputation service in order to be able to download new signatures from myZyxel.
	Select Daily to set the time of the day, or Weekly to set the day of the week and the time of the day.
	Schedule signature updates for a day and time when your network is least busy to minimize disruption to your network.
Anti-Malware	
Signature information	This shows the Current Version of the signature set the Nebula Device is using and the Released Date .
Enabled	Select On to turn on the rule. Otherwise, select Off to turn off the rule.
Log	Select whether to have the Nebula Device generate a log when the policy is matched to the criteria listed above.
Scan Mode	
Express Mode	In this mode you can define which types of files are scanned using the File Type For Scan fields. The Nebula Device then scans files by sending each file's hash value to a cloud database using cloud query. This is the fastest scan mode.
Stream Mode	In this mode the Nebula Device scans all files for viruses using its anti-malware signatures to detect known virus pattens. This is the deepest scan mode.
Hybrid Mode (for ATP devices only)	In this mode you can define which types of files are scanned using the File Type For Scan fields. The Nebula Device then scans files by sending each file's hash value to a cloud database using cloud query. It also scans files using anti-malware signatures, and Threat Intelligence Machine Learning. This mode combines Express Mode and Stream Mode to offer a balance of speed and security.
File decompression (ZIP and RAR)	Select this check box to have the Nebula Device scan a compressed file (the file does not need to have a "zip" or "rar" file extension). The Nebula Device first decompresses the file and then scans the contents for malware.
	Note: The Nebula Device decompresses a compressed file once. The Nebula Device does NOT decompress any files within a compressed file.
Destroy compressed files that could not be decompressed	When you select this check box, the Nebula Device deletes compressed files that use password encryption.
	Select this check box to have the Nebula Device delete any compressed files that it cannot decompress. The Nebula Device cannot decompress password protected files or a file within another compressed file. There are also limits to the number of compressed files that the Nebula Device can concurrently decompress.
	Note: The Nebula Device's firmware package cannot go through the Nebula Device with this check box enabled. The Nebula Device classifies the firmware package as a file that cannot be decompressed and then deletes it. Clear this check box when you download a firmware package from the Zyxel website. It is okay to upload a firmware package to the Nebula Device with the check box selected.
Cloud Query	Select the Cloud Query supported file types for the Nebula Device to scan for viruses.

Table 109 Site-wide > Configure > Firewall > Security service (continued)

LABEL	DESCRIPTION
Block list	This field displays the file or encryption pattern of the entry. Enter an MD5 hash or file pattern that would cause the Nebula Device to log and modify this file.
	File patterns:
	•Use up to 80 characters. Alphanumeric characters, underscores (_), dashes (-), question marks (?) and asterisks (*) are allowed.
	•A question mark (?) lets a single character in the file name vary. For example, use "a?.zip" (without the quotation marks) to specify aa.zip, ab.zip and so on.
	•Wildcards (*) let multiple files match the pattern. For example, use "*a.zip" (without the quotation marks) to specify any file that ends with "a.zip". A file named "testa.zip would match. There could be any number (of any type) of characters in front of the "a.zip" at the end and the file name would still match. A file named "test.zipa" for example would not match.
	• A * in the middle of a pattern has the Nebula Device check the beginning and end of the file name and ignore the middle. For example, with "abc*.zip", any file starting with "abc" and ending in ".zip" matches, no matter how many characters are in between.
	•The whole file name has to match if you do not use a question mark or asterisk.
	• If you do not use a wildcard, the Security Firewall checks up to the first 80 characters of a file name.
Allow list	Enter the file or encryption pattern for this entry. Enter an MD5 hash or file pattern to identify the names of files that the Nebula Device should not scan for viruses.
	File patterns:
	•Use up to 80 characters. Alphanumeric characters, underscores (_), dashes (-), question marks (?) and asterisks (*) are allowed.
	•A question mark (?) lets a single character in the file name vary. For example, use "a?.zip" (without the quotation marks) to specify aa.zip, ab.zip and so on.
	•Wildcards (*) let multiple files match the pattern. For example, use "*a.zip" (without the quotation marks) to specify any file that ends with "a.zip". A file named "testa.zip would match. There could be any number (of any type) of characters in front of the "a.zip" at the end and the file name would still match. A file named "test.zipa" for example would not match.
	• A * in the middle of a pattern has the Nebula Device check the beginning and end of the file name and ignore the middle. For example, with "abc*.zip", any file starting with "abc" and ending in ".zip" matches, no matter how many characters are in between.
	•The whole file name has to match if you do not use a question mark or asterisk.
	• If you do not use a wildcard, the Nebula Device checks up to the first 80 characters of a file name.
Sandboxing	Sandboxing provides a safe environment to separate running programs from your network and host devices. Unknown or untrusted programs/codes are uploaded to the Defend Center and executed within an isolated virtual machine (VM) to monitor and analyze the zero-day malware and advanced persistent threats (APTs) that may evade the Nebula Device's detection, such as anti-malware. Results of cloud sandboxing are sent from the server to the Nebula Device.
Enabled	Select this option to turn on sandboxing on the Nebula Device
Log	Enable this option to allow the Security Firewall to create a log when a suspicious file is detected.
Policy	Specify whether the Nebula Device deletes (Destroy) or forwards (Allow) malicious files. Malicious files are files given a high score for malware characteristics by the Defend Center.

Table 109 Site-wide > Configure > Firewall > Security service (continued)

LABEL	DESCRIPTION		
Inspect selected downloaded files	 Select this option to have the Nebula Device hold the downloaded file for up to 2 seconds if the downloaded file has never been inspected before. The Nebula Device will wait for the Defend Center's result and forward the file in 2 seconds. Sandbox detection may take longer than 2 seconds, so infected files could still possibly be forwarded to the user. Note: The Nebula Device only checks the file types you selected for sandbox inspection. The scan result will be removed from the Nebula Device cache after the Nebula Device restarts. 		
File submission options	Specify the type of files to be sent for sandbox inspection.		
Intrusion Detection/Prevention			
Signature information	This shows the Current Version of the signature set the Nebula Device is using and the Released Date .		
Detection	Select On to enable Detection.		
Prevention	Select On to enable Prevention.		

Table 109 Site-wide > Configure > Firewall > Security service (continued)

8.3.8.1 Create a Content Filtering Profile

Click the Add button in the Content Filtering section of the Site-wide > Configure > Firewall > Security service screen to access this screen.

ld profile			
Name			×
Description (Optional)			×
Log			
NS content filtering			
Enabled			
ock Web Pages			
Action for Unrated Web Pages	Warn 🔻		
Action When Service is Unavailable	Warn 👻		
ock Category			
Templates	Parental control 🔹		
	Test IDI		
Test URL	Enter a url to know website category		X lest
	▲ Category list		^
	Adult Topics	Alcohol	Anonymizing Utilities
	Art/Culture/Heritage	Auctions/Classifieds Chat	Blogs/Wiki Computing/Internet
	Consumer Protection	Content Server	Controversial Opinions
	Cult/Occult	Dating/Personals	Dating/Social Networking
	Education/Reference		Extreme
	Fashion/Beauty	Finance/Banking	For Kids
	Forum/Bulletin Boards	 Gambling 	Gambling Related
	Game/Cartoon Violence	Games	General News
	Government/Military	Gruesome Content	Health
		Incidental Nudity	
	Information Security New	 Instant Messaging 	Interactive Web Applications
	Internet Radio/TV	Internet Services	Job Search
	Major Global Religions	Marketing/Merchandising	Media Downloads
Search category	Media Sharing	Messaging	Mobile Phone
	Nudity	Motor venicies Online Shopping	P2P/File Sharing
	Parked Domain	Personal Network Storage	Personal Pages
	Pharmacy	Politics/Opinion	Pornography
	Portal Sites	Potential Criminal Activities	Potential Hacking/Computer Crime
	Potential Illegal Software	Private IP Address	Profanity Rublic Information
		Real Estate	Recreation/Hobbies
	Religion/Ideology	Remote Access	Residential IP Addresses
	Resource Sharing	Restaurants	School Cheating Information
	Search Engines	Sexual Materials	Shareware/Freeware
	Social Networking	Streaming Media	Sports Technical Information
	Technical/Business Forums	Text Translators	Text/Spoken Only
	Tobacco	Travel	Usenet News
	Violence	Visual Search Engine	Veapons
	Web Ads	Web Mail	Browser Evolute
	Malicious Downloads	Malicious Sites	Phishing
	Spam URLs	Spyware/Adware/Keyloggers	
Block web site	Wob Site		
	- Web Site	*	
	1	×	•
	+ Add		
Allow web site			
	Web Site		
	1	× *	Û
	+ Add		

Figure 136	Site-wide >	Configure >	> Firewall >	 Security 	/ service >	Content Filtering:	Add/Edit
				/			

LABEL	DESCRIPTION		
Add profile			
Name	This column lists the names of the content filter profile rule.		
Description (Optional)	This column lists the description of the content filter profile rule.		
Log	Select whether to have the Nebula Device generate a log when the policy is matched to the criteria listed above.		
DNS content filtering	Select this option to turn on DNS filtering on the Nebula Device.		
	DNS filtering inspects DNS queries made by clients on your network and compares the queries against a database of blocked or allowed Fully Qualified Domain Names (FQDNs). The Nebula Device DNS content filtering will either drop the DNS query or reply to the user with a fake DNS response.		
Block Web Pages			
Action for Unrated Web Pages	Select Pass to allow users to access web pages that the external web filtering service has not categorized.		
	Select Block to prevent users from accessing web pages that the external web filtering service has not categorized. When the external database content filtering blocks access to a web page, it displays the denied access message that you configured in the Content Filter General screen along with the category of the blocked web page.		
	Select Warn to display a warning message before allowing users to access web pages that the external web filtering service has not categorized.		
Action when service is Unavailable	Select Pass to allow users to access any requested web page if the external content filtering database is unavailable.		
	Select Block to block access to any requested web page if the external content filtering database is unavailable.		
	Select Warn to display a warning message before allowing users to access any requested web page if the external content filtering database is unavailable.		
	The following are possible causes for the external content filtering server not being available:		
	•There is no response from the external content filtering server within the time period specified in the Content Filter Server Unavailable Timeout field.		
	•The Nebula Device is not able to resolve the domain name of the external content filtering database.		
	•There is an error response from the external content filtering database. This can be caused by an expired content filtering registration (External content filtering's license key is invalid").		
Block Category			
The Nebula Device prevents When external database cor that you configured in the De	users from accessing web pages that match the categories that you select below. Intent filtering blocks access to a web page, it displays the denied access message enied access message field along with the category of the blocked web page.		
Templates	Web pages are classified into a category based on their content. You can choose a pre-defined template that has already selected certain categories. Alternatively, choose Custom and manually select categories in this section to control access to specific types of Internet content.		
LABEL DESCRIPTION			
---	--	--	
Test URL	You can check which category a web page belongs to. Enter a web site URL in the text box.		
	When the content filter is active, you should see the web page's category. The query fails if the content filter is not active.		
	Content Filtering can query a category by full URL string (for example, http:// www.google.com/picture/index.htm), but HTTPS Domain Filter can only query a category by domain name ('www.google.com'), so the category may be different in the query result. Test URL displays both results in the test.		
Search Category	Specify your desired filter criteria to filter the list of categories.		
Category List	Click to display or hide the category list.		
	These are categories of web pages based on their content. Select categories in this section to control access to specific types of Internet content.		
Block web site	Sites that you want to block access to, regardless of their content rating, can be blocked by adding them to this list.		
	Enter host names such as www.bad-site.com into this text field. Do not enter the complete URL of the site – that is, do not include "http://". All sub-domains are also blocked. For example, entering "bad-site.com" also blocks "www.badsite.com", "partner.bad-site.com", "press.bad-site.com", and so on. You can also enter just a top level domain. For example, enter .com to block all .com domains.		
	Use up to 127 characters (0–9 a–z). The casing does not matter.		
Add	Click this button to add a new entry.		
Allow web site	Sites that you want to allow access to, regardless of their content rating, can be allowed by adding them to this list.		
	Enter host names such as www.good-site.com into this text field. Do not enter the complete URL of the site – that is, do not include "http://". All sub-domains are allowed. For example, entering "zyxel.com" also allows "www.zyxel.com", "partner.zyxel.com", "press.zyxel.com", and so on. You can also enter just a top level domain. For example, enter .com to allow all .com domains.		
	Use up to 127 characters (0–9 a–z). The casing does not matter.		
Add	Click this button to add a new entry.		
.	Click this icon to remove the entry.		
Cancel	Click this button to exit this screen without saving.		
Create Click this button to save your changes and close the screen.			

Table 110	Site-wide >	Configure > Firev	all > Security service	> Content Filtering: Add/Edit (continued)	

8.3.8.2 Add Application Patrol Profile

Click the Add button in the Application Patrol section of the Site-wide > Configure > Firewall > Security service screen to access this screen.

Add profil	e			×
Name				
				X
Description (Optional)				×
Log				
Application Ma	nagement			
Enabled	Category	Application	Action	
1 🗸	Anti 💌	All 😂	Reject	ŵ
+ Add Set	arch Application		•	
			Close	Create

Figure 137 Site-wide > Configure > Firewall > Security service > Application Patrol: Add/Edit

Table 111 Site-wide > Configure > Firewall > Security service > Applicati	on Patrol: Add/Edit
---	---------------------

LABEL	DESCRIPTION
Add profile	
Name	This column lists the names of the application patrol profile rule.
Description (Optional)	This column lists the description of the application patrol profile rule.
Log	Select whether to have the Nebula Device generate a log when the policy is matched to the criteria listed above.
Application Management	
Enabled	Select the check box to turn on the rule. Otherwise, clear the check box to turn off the rule.
Category	Select an application category.
Application	Select All or select an application within the category to apply the policy.
Action	Displays the default action for the applications selected in this category.
	Reject – the Nebula Device drops packets that matches these application signatures and sends notification to clients.
1	Click this icon to remove the entry.
Add	Click this button to create a new application category and set actions for specific applications within the category.
Search Application	Enter a name to search for relevant applications and click Add to create an entry.
Close	Click this button to exit this screen without saving.
Create	Click this button to save your changes and close the screen.

8.3.9 Captive Portal

Use this screen to configure captive portal settings for each interface. A captive portal can intercept network traffic until the user authenticates his or her connection, usually through a specifically designated login web page.

Click Site-wide > Configure > Firewall > Captive portal to access this screen.

Figure 138 Site-wide > Configure > Firewall > Captive portal

terface	VLAN100	¥	
	Captive portal on this interface is	e is direct access. You can change this setting here,	
Themes			
	0		
BUTT	ON		
O Defout	Modern		
	2 2 2 2		
Click-to-continue	/Voucher/Sign-on page		
Logo			
		No logo	
Message			
Success page			
Message			
	Successi		
External captive p	oortal URL		
Use URL:	URL:	×	
	To use sustem on	cartine partal page place download the zin file and edit them	
	Download the c	e customized captive portal page example.	
Captive portal bel	havior		
After the captive p user should go?	oortal page where the 🛛 Stay on Capt	aptive portal authenticated successfully page	
	O To promotion	tion URL: X	

LABEL	DESCRIPTION
Interface	Select the Nebula Device's interface (network) to which the settings you configure here is applied.
Themes	This section is not configurable when External captive portal URL is set to ON.
	 Click the Preview icon at the upper right of a theme image to display the portal page in a new frame. Click the Copy icon to create a new custom theme (portal page). Click the Edit icon of a custom theme to go to a screen, where you can view and configure the details of the custom portal pages. See Section 8.3.9.1 on page 400. Click the Remove icon to delete a custom theme.
	Select the theme you want to use on the specified interface.
Click-to-continue/Sign	-on page
This section is not confi	gurable when External captive portal URL is set to ON.
Logo	This shows the logo image that you uploaded for the customized login page.
	Click Upload a logo and specify the location and file name of the logo graphic or click Browse to locate it. You can use the following image file formats: GIF, PNG, or JPG.
Message	Enter a note to display below the title. Use up to 1024 printable ASCII characters. Spaces are allowed.
Success page	
Message	Enter a note to display on the page that displays when a user logs in successfully. Use up to 1024 printable ASCII characters. Spaces are allowed.
External captive porta	I URL
Use URL	Select On to use a custom login page from an external web portal instead of the one built into the NCC. You can configure the look and feel of the web portal page.
	Specify the login page's URL; for example, http://IIS server IP Address/login.asp. The Internet Information Server (IIS) is the web server on which the web portal files are installed.
Captive portal behavi	or
After the captive portal page where the user should go?	Select To promotion URL and specify the URL of the web site/page to which the user is redirected after a successful login. Otherwise, select Stay on Captive portal authenticated successfully page .

Table 112 Site-wide > Configure > Firewall > Captive portal

8.3.9.1 Custom Theme Edit

Use this screen to check what the custom portal pages look like. You can also view and modify the CSS values of the selected HTML file. Click a custom login page's **Edit** button in the **Site-wide** > **Configure** > **Firewall** > **Captive portal** screen to access this screen.

Captive portal / Copy of Modern						← Back to config
Theme name	success.htm	l user_login.html	click_to_continue.html	color.css	icon.css	layout.css
Copy of Modern 🖾						
Font						
Color R 255 B 255 # FFFFFF Select		Welcome to lan2				
				P	Agree	_
			F	Powered by Z	YXEL	

Figure 139 Site-wide > Configure > Firewall > Captive portal: Edit

LABEL	DESCRIPTION
Back to config	Click this button to return to the Captive portal screen.
Theme name	This shows the name of the theme. Click the edit icon to change it.
Font	Click the arrow to hide or display the configuration fields.
	To display this section and customize the font type and/or size, click an item with text in the preview of the selected custom portal page (HTML file).
Color	Click the arrow to hide or display the configuration fields.
	Click an item in the preview of the selected custom portal page (HTML file) to display this section and customize its color, such as the color of the button, text, window's background, links, borders, and so on.
	Select a color that you want to use and click the Select button.
HTML/CSS	This shows the HTML file name of the portal page created for the selected custom theme. This also shows the name of the CSS files created for the selected custom theme.
	Click an HTML file to display the portal page. You can also change colors and modify the CSS values of the selected HTML file.
$\langle \rangle$	Click this button to view and modify the CSS values of the selected HTML file. It is recommended that you do NOT change the script code to ensure proper operation of the portal page.
	Click this button to preview the portal page (the selected HTML file).
Save	Click this button to save your settings for the selected HTML file to the NCC.
Apply	Click this button to save your settings for the selected HTML file to the NCC and apply them to the Nebula Device in the site.

Table 113 Site-wide > Configure > Firewall > Captive portal: Edit

8.3.10 Authentication Method

Use this screen to enable or disable web authentication on an interface.

Click Site-wide > Configure > Firewall > Authentication method to access this screen.

Figure 140	Site-wide >	Configure >	Firewall >	Authent	ication me	thod

erfaces: Ian1 💌		
Network Access		
	O Disable	
	Users can access the network directly	
	Olick-to-continue	
	Users must view and agree the captive portal page then can access the network	
	Sign-on-with Nebula Cloud Authentication	
	Two-factor authentication	
Nalled garden		
Walled garden ranges		
)
	What do lenter here?	
Captive portal access attribute		
Self-registration	Don't allow users to create accounts	
Login on multiple client devices	Multiple devices access simultaneously 🔹	
NCAS disconnection behavior 🜖		
	Allowed: Client devices can access the network without signing in excent they are explicitly blocked.	
	Limited:	

LABEL	DESCRIPTION
Interfaces	Select the Nebula Device's interface (network) to which the settings you configure here is applied.
Network Access	Select Disable to turn off web authentication.
	Select Click-to-continue to block network traffic until a client agrees to the policy of user agreement.
	Select Sign-on with to block network traffic until a client authenticates with an external RADIUS or AD server through the specifically designated web portal page. Select Nebula Cloud Authentication or an authentication server that you have configured in the Site-wide > Configure > Firewall > Firewall settings screen (see Section 8.3.12 on page 405).
	Select Two-Factor Authentication to require that the user log in using both their password and a Google Authenticator code. To log in, users must have Two-Factor Authentication enabled on their account and have setup Google Authenticator on their mobile device.
Walled garden	This field is not configurable if you set Network Access to Disable.
	Select to turn on or off the walled garden feature.
	With a walled garden, you can define one or more web site addresses that all users can access without logging in. These can be used for advertisements for example.
Walled garden ranges	Specify walled garden web site links, which use a domain name or an IP address for web sites that all users are allowed to access without logging in.
Captive portal access	attribute
Self-registration	This field is available only when you select Sign-on with Nebula Cloud authentication in the Network Access field.
	Select Allow users to create accounts with auto authorized or Allow users to create accounts with manual authorized to display a link in the captive portal login page. The link directs users to a page where they can create an account before they authenticate with the NCC. For Allow users to create accounts with manual authorized, users cannot log in with the account until the account is authorized and granted access. For Allow users to create accounts with auto authorized, users can just use the registered account to log in without administrator approval.
	Select Don't allow users to create accounts to not display a link for account creation in the captive portal login page.
Login on multiple	This field is available only when you select Sign-on with in the Network Access field.
	Select Multiple devices access simultaneously if you allow users to log in as many times as they want as long as they use different IP addresses.
	Select One device at a time if you do not allow users to have simultaneous logins.
NCAS disconnection behavior	This field is available only when you select Sign-on with Nebula Cloud Authentication in the Network Access field.
	Select Allowed to allow any users to access the network without authentication when the NCAS (Nebula Cloud Authentication Server) is not reachable.
	Select Limited to allow only the currently connected users or the users in the white list to access the network.

Table 114 Site-wide > Configure > Firewall > Authentication method

8.3.11 Wireless

This screen allows you to configure different SSID profiles for your Nebula Device. An SSID, or Service Set IDentifier, is the name of the WiFi network to which a WiFi client can connect. The SSID appears as

readable text to any device capable of scanning for WiFi frequencies (such as the WiFi adapter in a laptop), and is displayed as the WiFi network name when a person makes a connection to it.

Click Site-wide >	Configure >	Firewall	> Wireless to	access this screen.
-------------------	-------------	----------	---------------	---------------------

4o.	1		2		3		4	
lame	Private Network (Zycamp)	× *	Guest Network (Zycamp)	× *	SSID3	× *	SSID4	>
nabled								
uthentication								
WLAN Security	WPA2-PSK	•	Open	•	Open	•	Open	
Associate Key		•		•		0		0
Band	Concurrent operation(2.4G	•	Concurrent operation(2.4G	•	Concurrent operation(2.4G	•	Concurrent operation(2.4G.	. ,
Outgoing Interface	VLAN10	•	lan1	-	lan1	•	lan1	2
adio Settings								
adio Settings Maximum output power	2.4GHz 5GHz	30	0 dBm 🔹 0 dBm 👻					
adio Settings Maximum output power Channel width	2.4GHz 5GHz 2.4GHz	30	0 dBm O dBm O dBm O dBm V					
adio Settings Maximum output power Channel width	2.4GHz 5GHz 2.4GHz 5GHz	31 31 20 81	0 dBm					
adio Settings Maximum output power Channel width 24 GHz channel deployment	2.4GHz 5GHz 2.4GHz 5GHz Three-Channel	30 34 20 81 I Deploym	0 dBm O dBm O dBm O MHz O MHz ent					

Figure 141 Site-wide > Configure > Firewall > Wireless

Table 115	Site-wide > Configure > Firewall > Wireless
-----------	---

LABEL	DESCRIPTION
SSID Settings	
No.	This shows the SSID number.
Name	This shows the SSID name as it appears to WiFi clients.
Enabled	Click this to enable the SSID to be discoverable by WiFi clients.
Authentication	
WLAN Security	Select Open to allow any WiFi client to associate with this network without any data encryption nor authentication.
	Select WPA2-PSK to enable WPA2-PSK data encryption.
Associate Key	Enter a pre-shared key from 8 to 64 case-sensitive keyboard characters to enable WPA2- PSK data encryption.
Band	Select to have the SSID use either 2.4 GHz band only or the 5 GHz band only.
	If you select Concurrent operation (2.4 GHz and 5 GHz) , the SSID uses both frequency bands.
Outgoing Interface	Select the interface for outgoing traffic from the Nebula Device to the Internet.

LABEL	DESCRIPTION
Radio Settings	
Maximum output power	Enter the maximum output power of the radio (in dBm).
Channel width	Select the WiFi channel bandwidth you want the Nebula Device to use.
	A standard 20 MHz channel offers transfer speeds of up to 144 Mbps (2.4 GHz) or 217 Mbps (5 GHz) whereas a 40 MHz channel uses two standard channels and offers speeds of up to 300 Mbps (2.4 GHz) or 450 Mbps (5 GHz). An IEEE 802.11ac-specific 80 MHz channel offers speeds of up to 1.3 Gbps.
	40 MHz (channel bonding or dual channel) bonds two adjacent radio channels to increase throughput. An 80 MHz channel consists of two adjacent 40 MHz channels. The WiFi clients must also support 40 MHz or 80 MHz. It is often better to use the 20 MHz setting in a location where the environment hinders the WiFi signal.
	Note: It is suggested that you select 20 MHz when there is more than one 2.4 GHz Nebula Device in the network.
2.4 GHz channel deployment	Select Three-Channel Deployment to limit channel switching to channels 1, 6, and 11, the three channels that are sufficiently attenuated to have almost no impact on one another. In other words, this allows you to minimize channel interference by limiting channel-hopping to these three "safe" channels.
	Select Four-Channel Deployment to limit channel switching to four channels. Depending on the country domain, if the only allowable channels are 1 – 11 then the Nebula Device uses channels 1, 4, 7, 11 in this configuration; otherwise, the Nebula Device uses channels 1, 5, 9, 13 in this configuration. Four-Channel Deployment expands your pool of possible channels while keeping the channel interference to a minimum.
	Select Manual to choose the allowable channels 1 – 11.
5 GHz channel deployment	Select how you want to specify the channels the Nebula Device switches between for 5 GHz operation.
	Select Auto to have the Nebula Device automatically select the best channel.
	Select Manual to choose from the allowable channels.

Table 115 Site-wide > Configure > Firewall > Wireless (continued)

8.3.12 Firewall Settings

Use this screen to configure DNS settings and external AD (Active Directory), RADIUS, or LDAP server that the Nebula Device can use for authenticating users.

AD (Active Directory) is a directory service that is both a directory and a protocol for controlling access to a network. The directory consists of a database specialized for fast information retrieval and filtering activities. You create and store user profile and login information on the external server.

This screen also lets you configure the addresses of walled garden web sites that users can access without logging into the Nebula Device. The settings in this screen apply to all networks (interfaces) on the Nebula Device. If you want to configure walled garden web site links for a specific interface, use the **Authentication method** screen.

Click Site-wide > Configure > Firewall > Firewall settings to access this screen.

ewall settings					
SNG					
Address Record					
FQDN			IP Address		
×	<[×		
+ Add					
Domain Zone Forwarder					
Domain Zone	IP Address	Interface			
	4	× auto			•
+ Add					
ynamic DNS					
Automatic registration					
	Dynamic D	NS updates a DNS record ec	ach time the public IP address of the sec	urity appliance changes.	
uthentication Server					
My AD Server					
Nama	Server address	Bookup serv	er address Port	AD domain	Do
×	¢.	×	× 389	×	×
My LDAP Server					
Nome	Server oddress	Bookup serv	ar oddrass Port	Base DN	Bir
×	e.	×	× 389	× *	× *
4.8					
+ Add					
My RADIUS Server				19.500 million	
Name	Server address	Bookup serv	er oddress Port	Secret	Ad
-		×	× 1812	×,	© <u>D</u>
+ Add					
kternal User Group					
Please create outh	nentication server before	add external user group			
alled garden					
Global walled garden	This is glot interface v	val walled garden configurat valled garden policy.	ion. All web authentication interface will	match this policy first and the sec	ond priority is the
	If needed o	only allow specify interface, p	lease go to Network access method cor	nfigure -	
	What do	enter here?			
PALG					
OID N.A.	-				
sir n53 SiP Signaling Port					
	5000				
ADVANCED OPTIONS					
dvanced Options					
Isolate unwanted traffic betwee	in tunnel				

Figure 142 Site-wide > Configure > Firewall > Firewall settings

NCC User's Guide

Table 116	Sita wida >	Configure	Firewalls	Firewall settings
	slie-wide >	Configure -	Filewali /	ritewali setiings

LABEL	DESCRIPTION
DNS	
Address Record	This record specifies the mapping of a Fully-Qualified Domain Name (FQDN) to an IP address. An FQDN consists of a host and domain name. For example, www.zyxel.com.tw is a fully qualified domain name, where "www" is the host, "zyxel" is the third-level domain, "com" is the second-level domain, and "tw" is the top level domain.
FQDN	Enter a host's fully qualified domain name.
	Use "*." as a prefix in the FQDN for a wildcard domain name (for example, *.example.com).
IP Address	Enter the host's IP address.
	Click this icon to remove the entry.
Add	Click this button to create a new entry.
Domain Zone Forwarder	This specifies a DNS server's IP address. The Nebula Device can query the DNS server to resolve domain zones for features like VPN, DDNS and the time server. When the Nebula Device needs to resolve a domain zone, it checks it against the domain zone forwarder entries in the order that they appear in this list.
Domain Zone	A domain zone is a fully qualified domain name without the host. For example, zyxel.com.tw is the domain zone for the www.zyxel.com.tw fully qualified domain name. Whenever the Nebula Device receives needs to resolve a zyxel.com.tw domain name, it can send a query to the recorded name server IP address.
IP Address	Enter the DNS server's IP address.
Interface	Select the interface through which the Nebula Device sends DNS queries to the specified DNS server.
	Click this icon to remove the entry.
Add	Click this button to create a new entry.
Authentication Server My AD Server	
Name	Enter a descriptive name for the server.
Server address	Enter the address of the AD server.
Backup server address	If the AD server has a backup server, enter its address here.
Port	Specify the port number on the AD server to which the Nebula Device sends authentication requests. Enter a number between 1 and 65535.
AD domain	Specify the Active Directory forest root domain name.
Domain admin	Enter the name of the user that is located in the container for Active Directory Users, who is a member of the Domain Admin group.
Password	Enter the password of the Domain Admin user account.
Advanced	Click to open a screen where you can select to use Default or Custom advanced settings. See Section 8.3.12.3 on page 413.
m	Click this icon to remove the server.
Add	Click this button to create a new server.
My LDAP Server	
Name	Enter the description of each server, if any. You can use up to 60 printable ASCII characters.
Server address	Enter the address of the LDAP server.
Backup server address	If the LDAP server has a backup server, enter its address here.

LABEL	DESCRIPTION
Port	Specify the port number on the LDAP server to which the Nebula Device sends authentication requests. Enter a number between 1 and 65535.
Base DN	Specify the directory (up to 127 alphanumerical characters). For example, o=Zyxel, c=US.
Bind DN	Specify the bind DN for logging into the AD or LDAP server. Enter up to 127 alphanumerical characters.
	For example, cn=zywallAdmin specifies zywallAdmin as the user name.
Password	If required, enter the password (up to 15 alphanumerical characters) required to bind or log in to the LDAP server.
Advanced	Click to open a screen where you can select to use Default or Custom advanced settings. See Section 8.3.12.3 on page 413.
	Click this icon to remove the entry.
Add	Click this button to create a new server.
My RADIUS Server	
Name	Enter a descriptive name for the server.
Server address	Enter the address of the RADIUS server.
Backup server address	If the RADIUS server has a backup server, enter its address here.
Port	Specify the port number on the RADIUS server to which the Nebula Device sends authentication requests. Enter a number between 1 and 65535.
Secret	Enter a password (up to 15 alphanumeric characters) as the key to be shared between the external authentication server and the Nebula Device.
	The key is not sent over the network. This key must be the same on the external authentication server and the Security Firewall.
Advanced	Click to open a screen where you can select to use Default or Custom advanced settings. See Section 8.3.12.3 on page 413.
.	Click this icon to remove the server.
Add	Click this button to create a new server.
External User Group	
Group Name	Enter a descriptive name for the group, up to 31 characters [0–9][a–z][A–Z][@] but the first character must be an alphabet.
Authentication Server	Select the Name of the Authentication Server you added in My AD Server , My LDAP Server , or My RADIUS Server .
Group ID	Enter the name of the attribute that the Nebula Device checks to determine to which group an external user belongs. The value for this attribute is called a group identifier; it determines to which group an external user belongs.
Add	Click this button to create a new group. The maximum number of external user groups is 20.
Walled garden	
Global Walled garden	With a walled garden, you can define one or more web site addresses that all users can access without logging in. These can be used for advertisements for example.
	Specify walled garden web site links, which use a domain name or an IP address for web sites that all users are allowed to access without logging in.
Advanced Options	
Isolate unwanted traffic between tunnel mode APs	Select On to block broadcast and multicast traffic coming from Remote APs (RAPs).

Talala 11/	Cite Mide >	Configures		Fire well estimate	(a a lation , a al)
	slie-wide >	Conligure .	> Firewali >	Firewall settings	(continued)

8.3.12.1 Dynamic DNS

Enable Dynamic DNS to open the Site-wide > Configure > Firewall > Firewall settings: Dynamic DNS screen.

Figure 143	Site-wide > Contigure	> Firewall > Firewal	I settings: Dynamic DNS

Dynamic DNS	
Automatic registration	
	Dynamic DNS updates a DNS record each time the public IP address of the security appliance changes.
General settings	
DDNS provider	DynDNS -
DDNS type	DynDNS
DDNS account	
Username	×
Password	•
Confirm password	Image: A start of the start
DDNS settings	
Domain name	×
Primary binding address	
Interface	
IP address	Auto
Backup binding address	
Interface	•
IP address	Auto 👻
Enable wildcard	
Mail exchanger	× (Optional)
Backup mail exchanger	

The following table describes the labels in this screen.

Table 117 Site-wide > Configure > Firewall > Firewall settings: Dynamic DNS

LABEL	DESCRIPTION
Dynamic DNS	
Automatic registration	Click On to use dynamic DNS. Otherwise, select Off to disable it.
General Settings	
DDNS provider	Select your Dynamic DNS service provider from the drop-down list box.
	If you select User customize, create your own DDNS service.

LABEL	DESCRIPTION			
DDNS type	Select the type of DDNS service you are using.			
	Select DynDNS custom to create your own DDNS service and configure the DynDNS and DDNS static fields below.			
	If the DDNS provider is Dynu , you can select the account type of DynuBasic or DynuPremium .			
DDNS account				
Username	Enter the user name used when you registered your domain name.			
Password	Enter the password provided by the DDNS provider.			
Confirm password	Enter the password again to confirm it.			
DDNS settings				
Domain name	Enter the domain name you registered.			
Primary binding address	Use these fields to set how the Nebula Device determines the IP address that is mapped to your domain name in the DDNS server. The Nebula Device uses the Backup binding address if the interface specified by these settings is not available.			
Interface	Select the interface to use for updating the IP address mapped to the domain name.			
IP address	Select Auto if the interface has a dynamic IP address. The DDNS server checks the source IP address of the packets from the Nebula Device for the IP address to use for the domain name. You may want to use this if there are one or more NAT routers between the Nebula Device and the DDNS server.			
	Note: The Nebula Device may not determine the proper IP address if there is an HTTP proxy server between the Nebula Device and the DDNS server.			
	Select Custom if you have a static IP address. Enter the IP address to use it for the domain name.			
	Select Interface to have the Nebula Device use the IP address of the specified interface.			
Backup binding address	Use these fields to set an alternate interface to map the domain name to when the interface specified by the Primary binding address settings is not available.			
Interface	Select the interface to use for updating the IP address mapped to the domain name.			
IP address	Select Auto if the interface has a dynamic IP address. The DDNS server checks the source IP address of the packets from the Nebula Device for the IP address to use for the domain name. You may want to use this if there are one or more NAT routers between the Nebula Device and the DDNS server.			
	Note: Note: The Nebula Device may not determine the proper IP address if there is an HTTP proxy server between the gateway and the DDNS server.			
	Select Custom if you have a static IP address. Enter the IP address to use it for the domain name.			
	Select Interface to have the Security Firewall use the IP address of the specified interface.			
Enable wildcard	This option is only available with a DynDNS account.			
	Enable the wildcard feature to alias sub-domains to be aliased to the same IP address as your (dynamic) domain name. This feature is useful if you want to be able to use, for example, www.yourhost.dyndns.org and still reach your hostname.			

 Table 117
 Site-wide > Configure > Firewall > Firewall settings: Dynamic DNS (continued)

LABEL	DESCRIPTION		
Mail exchanger	This option is only available with a DynDNS account.		
	DynDNS can route email for your domain name to a mail server (called a mail exchanger). For example, DynDNS routes email for john-doe@yourhost.dyndns.org to the host record specified as the mail exchanger.		
	If you are using this service, type the host record of your mail server here. Otherwise, leave the field blank.		
Backup mail exchanger	This option is only available with a DynDNS account.		
	Select this check box if you are using DynDNS's backup service for email. With this service, DynDNS holds onto your email if your mail server is not available. Once your mail server is available again, the DynDNS server delivers the mail to you. See www.dyndns.org for more information about this service.		
DYNDNS Server	This field displays when you select User customize from the DDNS provider field above. Enter the IP address of the server that will host the DDNS service.		
URL	This field displays when you select User customize from the DDNS provider field above. Enter the URL that can be used to access the server that will host the DDNS service.		
Additional DDNS Options	This field displays when you select User customize from the DDNS provider field above. These are the options supported at the time of writing:		
	 dyndns_system to specify the DYNDNS Server type – for example, dyndns@dyndns.org 		
	 ip_server_name which should be the URL to get the server's public IP address – for example, http://myip.easylife.tw/ 		

 Table 117
 Site-wide > Configure > Firewall > Firewall settings: Dynamic DNS (continued)

8.3.12.2 SIP ALG

Application Layer Gateway (ALG) allows the following applications to operate properly through the NCC's NAT.

SIP (Session Initiation Protocol) is an application-layer protocol that can be used to create voice and multimedia sessions over Internet.

Go to SIP ALG in the Site-wide > Configure > Firewall > Firewall settings screen to access this screen. Use this screen to turn the ALG off or on, configure the port numbers to which they apply, and configure SIP ALG time outs.

Note: If the NCC provides an ALG for a service, you must enable the ALG in order to use the application patrol on that service's traffic.

Figure 144	Site-wide >	Configure >	• Firewall >	· Firewall	settings:	SIP	ALG

IP ALG				
SIP ALG				
SIP Signaling Port	5060	*		
ADVANCED OPTIONS				
SIP Inactivity Timeout				
SIP Media Inactivity Timeout	120	× seconds		
SIP Signaling Inactivity Timeout	1800	× * seconds		
Restrict Peer to Peer Signaling Connection				
Restrict Peer to Peer Media Connection				

LABEL	DESCRIPTION		
SIP ALG	Turn on SIP ALG to detect SIP traffic and help build SIP sessions through the Nebula Device NAT. Enabling the SIP ALG also allows you to use the application patrol to detect SIP traff and manage SIP traffic bandwidth.		
SIP Signaling Port	If you are using a custom UDP port number (not 5060) for SIP traffic, enter it here. Use the Add icon to add fields if you are also using SIP on additional UDP port numbers.		
ADVANCED OPTIONS	Click the arrow to show the fields for setting the SIP inactivity timeout and restrict peer-to- peer connection.		
SIP Inactivity Timeout	Select this to have the Nebula Device apply SIP media and signaling inactivity time out limits. These timeouts will take priority over the SIP session time out "Expires" value in a SIP registration response packet.		
SIP Media Inactivity Timeout	Use this field to set how many seconds (1 – 86400) the Nebula Device will allow a SIP session to remain idle (without voice traffic) before dropping it.		
	If no voice packets go through SIP ALG before the timeout period expires, the Nebula Device deletes the audio session. You cannot hear anything and you will need to make a new call to continue your conversation.		
SIP Signaling Inactivity Timeout	Most SIP clients have an "expire" mechanism indicating the lifetime of signaling sessions. The SIP user agent sends registration packets to the SIP server periodically and keeps the session alive in the Nebula Device.		
	If the SIP client does not have this mechanism and makes no calls during the Nebula Device SIP timeout, the Nebula Device deletes the signaling session after the timeout period. Enter the SIP signaling session timeout value (1 – 86400).		
Restrict Peer to Peer	A signaling connection is used to set up the SIP connection.		
Signaling Connection	Enable this if you want signaling connections to only arrive from the IP addresses you have already registered with. Signaling connections from other IP addresses will be dropped.		
Restrict Peer to Peer	A media connection is the audio transfer in a SIP connection.		
	Enable this if you want media connections to only arrive from the IP addresses you registered with. Media connections from other IP addresses will be dropped.		

Table 118	Site-wide >	Configure >	Firewall >	Firewall	settinas: SIP	ALG
		Conngolor		110000	30111193.01	,

8.3.12.3 Advanced Settings

Click the Advanced column in the Site-wide > Configure > Firewall > Firewall settings screen to access this screen.

Figure 145	Site-wide > Configure > Firewall >	Firewall settings: Advanced
riguic 145		ricovali seriligs. Aavaneee

Advanced			×
Preset:	Default	.	
Timeout:	5	× (1-300 seconds)	
Case-Sensitive User Name:	off		
NAS IP Address	127.0.0.1	×	
			Close OK

LABEL	DESCRIPTION
Preset	Select Default to use the pre-defined settings, or select Custom to configure your own settings.
Timeout	Specify the timeout period (between 1 and 300 seconds) before the Nebula Device disconnects from the server. In this case, user authentication fails.
	Search timeout occurs when either the user information is not in the servers or the AD or server is down.
Case-Sensitive User Name	Click ON if the server checks the case of the user name. Otherwise, click OFF to not configure your user name as case-sensitive.
Group Membership Attribute	Enter the name of the attribute that the gateway checks to determine to which group a user belongs. The value for this attribute is called a group identifier; it determines to which group a user belongs. You can add ext-group-user user objects to identify groups based on these group identifier values.
	For example you could have an attribute named "memberOf" with values like "sales", "RD", and "management". Then you could also create a ext-group-user user object for each group. One with "sales" as the group identifier, another for "RD" and a third for "management".
LDAP-only Fields	
Login Name Attribute	Enter the type of identifier the users are to use to log in. For example "name" or "email address".
RADIUS-only Fields	
NAS IP Address	Enter the IP address of the NAS (Network Access Server).
NAS Identifier	If the RADIUS server requires the Nebula Device to provide the Network Access Server identifier attribute with a specific value, enter it here.
Close	Click this button to exit this screen without saving.
OK	Click this button to save your changes and close the screen.

Table 119	Site-wide >	Configure >	Firewall >	Firewall s	ettinas: A	Advanced
		Gormgoro			onn 1951 /	ariooa

CHAPTER 9 Security Gateway

9.1 Overview

This chapter discusses the menus that you can use to monitor the Nebula managed Security Gateways in your network and configure settings even before a gateway is deployed and added to the site.

Nebula Device refers to Nebula NSG devices in this chapter. The **Security gateway** menus are shown for Nebula NSG devices only.

9.2 Monitor

Use the **Monitor** menus to check the Nebula Device information, client information, event log messages and summary report for the Nebula Device in the selected site.

9.2.1 Event Log

Use this screen to view Nebula Device log messages. You can enter a key word, select one or multiple event types, or specify a date/time or a time range to display only the log messages that match these criteria.

Select **Range** to set a time range or select **Before** to choose a specific date/time and the number of hours/minutes to display only the log messages generated within a certain period of time (before the specified date/time). Then click **Search** to update the list of logs based on the search criteria. The maximum allowable time range is 30 days.

Click Site-wide > Monitor > Security gateway > Event log to access this screen.

×			Category: Any
Before	2019-10-29	10:56	▼ 1h ▼ UTC+8 🖾 🔍 Se
338 Event log			💎 🕒 Exp
Category	Source	Destination	Detail
VPN	192.168.11.74	61.216.142.42	ISAKMP SA [S201711070315] is disconnected
VPN	192.168.11.74	61.216.142.42	The cookie pair is : 0xa8c4726c50064617 / 0x6f
VPN	61.216.142.42	192.168.11.74	Recv:[NOTIFY:NO_PROPOSAL_CHOSEN]
VPN	61.216.142.42	192.168.11.74	The cookie pair is : 0x6f8f47eb7aac5173 / 0xa8c
VPN	192.168.11.74	61.216.142.42	Send:[SA][VID][VID][VID][VID][VID][VID][VID][VID
VPN	192.168.11.74	61.216.142.42	Send Main Mode request to [61.216.142.42]
VPN	192.168.11.74	61.216.142.42	Tunnel [S201711070315] Sending IKE request
VPN	192.168.11.74	61.216.142.42	The cookie pair is : 0xa8c4726c50064617 / 0x00
VPN	192.168.11.74	61.216.142.42	ISAKMP SA [S201711070315] is disconnected
VPN	192.168.11.74	61.216.142.42	The cookie pair is : 0x2d752e6167623ee9 / 0x53
	Before Before Before Before SSB Event log VPN VPN VPN VPN	Before 2019-10-29 Before 2019-10-29 State 2019-10-29 VPN 19216811.74 VPN 19216811.74	Before 2019-10-29 Destination 338 Event log 50urce Destination VPN 192.168.11.74 61.216.142.42 VPN 192.168.11.74 61.216.142.42 VPN 61.216.142.42 192.168.11.74 VPN 61.216.142.42 192.168.11.74 VPN 61.216.142.42 192.168.11.74 VPN 61.216.142.42 192.168.11.74 VPN 192.168.11.74 61.216.142.42 VPN 192.168.11.74 61.216.142.42 <

Figure 146 Site-wide > Monitor > Security gateway > Event log

9.2.2 VPN Connections

Use this screen to view the status of site-to-site IPSec VPN connections and L2TP VPN connections.

Note: If the peer gateway is not a Nebula Device, go to the Site-wide > Configure > Security gateway > Site-to-Site VPN screen to view and configure a VPN rule. See Section 9.3.6 on page 449 for more information.

Click Site-wide > Monitor > Security gateway > VPN Connections to access this screen.

Figure 147 Site-wide > Monitor > Security gateway > VPN Connections

Configuration:	j	This security gateway	is exporting 1 subnet o	ver the VPN: 100.25.1.0/2	24	
NAT type:		Manual. This security gateway has a publicly accessible IP address and is using 211.22.54.173 as a contact point.				
ite connectivity						
ocation	Subnet(s)	Status	Inbound(Bytes)	Outbound(Bytes)	Tunnel up time	Last heartbeat
Hub	10.0.1.0/24 172.16.0.0/12 10.251.0.0/16 10.253.0.0/16	disconnected	0 bytes	0 bytes	-	-
<u>Site25 NCC AE B</u>	-	-	0 bytes	0 bytes	-	-
lient to site VPN	login account					
Jser Name		Hostname	Assign	ed IP	Public IP	

NCC User's Guide

Table 120	Site wide > Monitor >	Security anteway >	VPN Connections
	2116-MICE > MICHINOL >	> seconily galeway >	

LABEL	DESCRIPTION
C	Click this button to reload the data-related frames on this page.
Connection Status	
Configuration	This shows the number and address of the local networks behind the Nebula Device, on which the computers are allowed to use the VPN tunnel.
NAT Type	This shows the public IP address or the domain name that is configured and mapped to the Nebula Device on the NAT router.
Site Connectivity	
Location	This shows the name of the site to which the peer gateway is assigned.
	Click the name to go to the Site-wide > Configure > Security gateway > Site-to-Site VPN screen, where you can modify the VPN settings.
Subnet(s)	This shows the address of the local networks behind the Nebula Device.
Status	This shows whether the VPN tunnel is connected or disconnected.
Inbound (Bytes)	This shows the amount of traffic that has gone through the VPN tunnel from the remote IPSec router to the Nebula Device since the VPN tunnel was established.
Outbound (Bytes)	This shows the amount of traffic that has gone through the VPN tunnel from the Nebula Device to the remote IPSec router since the VPN tunnel was established.
Tunnel up time	This shows how many seconds the VPN tunnel has been active.
Last heartbeat	This shows the last date and time a heartbeat packet is sent to determine if the VPN tunnel is up or down.
Client to site VPN login	account
User Name	This shows the remote user's login account name.
Hostname	This shows the name of the computer that has this L2TP VPN connection with the Nebula Device.
Assigned IP	This shows the IP address that the Nebula Device assigned for the remote user's computer to use within the L2TP VPN tunnel.
Public IP	This shows the public IP address that the remote user is using to connect to the Internet.

9.2.3 NSS Analysis Report

Use this screen to view the statistics report for NSS (Nebula Security Service), such as content filtering, Intrusion Detection and Prevention (IDP), application patrol, and anti-virus. The screen varies depending on the service type (**Application**, **Content Filtering**, or **Anti-Virus**) you select.

Click Site-wide > Monitor > Security gateway > NSS analysis report to access this screen.





Table 121	Site-wide >	Monitor >	 Security 	gateway	> NSS	Analysis	Report
-----------	-------------	-----------	------------------------------	---------	-------	----------	--------

LABEL	DESCRIPTION
Security Appliance – NSS Analysis	Select to view the report for the past day, week or month. Alternatively, select Custom range to specify a time period the report will span. You can also select the number of results you want to view in a table.
	2022-07-06 iii to 2022-07-07 iii (Max range is 30 days, the dates will be auto-adjusted.) Report size: 10 results per table C Update
	Select the type of service for which you want to view the statistics report.
Email report	Click this button to send summary reports by email, change the logo and set email schedules.

LABEL	DESCRIPTION
Application	
The following field information about previous screen.	ds displays when you select to view the application statistics. Click an application name to view It the clients who use that application. Click Top Application under the chart to switch back to the
y-axis	The y-axis shows the amount of the application's traffic which has been transmitted or received.
x-axis	The x-axis shows the time period over which the traffic flow occurred.
Application	This shows the name of the application. Click an application name to view the IPv4 addresses of the clients who used the application.
Description	This shows the name of the client who used the application.
	This field is available when you click the application name. Click the name to display the individual client statistics. See Section 9.2.1 on page 414.
IPv4 Address	This shows the IPv4 address of the client who used the application.
	This field is available when you click the application name.
MAC Address	This shows the MAC address of the client who used the application.
	This field is available when you click the application name.
Category	This shows the name of the category to which the application belongs.
Usage	This shows the total amount of data consumed by the application used by all or a specific IPv4 address.
% Usage	This shows the percentage of usage for the application used by all or a specific IPv4 address.
The following field information about back to the previ	ds display when you select to view the content filtering statistics. Click a website URL to view the clients who tried to access that web page. Click Content Filtering under the chart to switch ious screen.
y-axis	The y-axis shows the number of hits on web pages that the Nebula Device's content filter service has blocked.
x-axis	The x-axis shows the time period over which the web page is checked.
Website	This shows the URL of the web page to which the Nebula Device blocked access. Click a website URL to view the IPv4 addresses of the clients who tried to access the web page.
Description	This shows the name of the client who tried to access the web page.
	This field is available when you click the website URL. Click the name to display the individual client statistics. See Section 9.2.1 on page 414.
IPv4 Address	This shows the IPv4 address of the client who tried to access the web page.
	This field is available when you click the website URL.
MAC Address	This shows the MAC address of the client who tried to access the web page.
	This field is available when you click the website URL.
Category	This shows the name of the category to which the web page belongs.
Hits	This shows the number of hits on the web page visited by all or a specific IPv4 address.
% Hits	This shows the percentage of the hit counts for the web page visited by all or a specific IPv4 address.
Anti-Virus	•

Table 121	Site-wide > Mor	nitor > Security g	gateway > NSS	Analysis Report	(continued)
-----------	-----------------	--------------------	---------------	-----------------	-------------

The following fields are displayed when you select **Anti-Virus**. Click a virus name to view information about the clients who sent the virus. Click the number in the center of the donut chart or **Anti-Virus** under the chart to switch back to the previous screen.

LABEL	DESCRIPTION				
y-axis	The y-axis shows the total number of viruses that the gateway has detected.				
x-axis	The x-axis shows the time period over which the virus is detected.				
Virus Name	This shows the name of the virus that the Nebula Device has detected and blocked. Click a virus name to view the IPv4 addresses of the clients who sent the virus.				
Description	This shows the name of the client who sent the virus.				
	This field is available when you click the virus name. Click the name to display the individual client statistics. See Section 9.2.1 on page 414.				
IPv4 Address	This shows the IPv4 address of the virus sender.				
	This field is available when you click the virus name.				
MAC Address	This shows the MAC address of the virus sender.				
	This field is available when you click the virus name.				
Hits	This shows how many times the gateway has detected the virus sent by all or a specific IPv4 address.				
% Hits	This shows the percentage of the hit counts for the virus sent by all or a specific IPv4 address.				
Intrusion Detection	/ Prevention				
The following fields	are displayed when you select Intrusion Detection / Prevention.				
The donut chart sh Prevention (IDP) se attacks blocked b	ows the number of potential network attacks detected by the Intrusion Detection and rvice, if any. The number in the center of the donut chart indicates the number of network y the IDP service.				
Signature Name	The name of the IDP signature that triggered the hit. The signature name identifies the type of intrusion pattern				
Hits	This shows the total number of network attacks blocked by the IDP service.				
% Hits	This shows the number of network attacks blocked as a percentage of the total number of network requests scanned by the IDP service.				

Tabla 101	Site wide > Meniter > Security actoway > NSS Analysis Depart (continued)
	- 2116-MIDE > MOUTIOL > 26COTTA DOTEMON > 1222 VUOLARIS KEDOLI (COLTINUED)

9.2.4 Summary Report

This screen displays network statistics for the Nebula Device of the selected site, such as WAN usage, top applications and/or top clients.

Click Site-wide > Monitor > Security gateway > Summary report to access this screen.



Figure 149 Site-wide > Monitor > Security gateway > Summary report

NCC User's Guide

Table 122 Site-wide > Monitor > Security gateway > Summary report

LABEL	DESCRIPTION				
Security gateway – Summary report	Select to view the report for the past day, week or month. Alternatively, select Custom range to specify a time period the report will span. You can also select the number of results you want to view in a table.				
	 Last 24 hours Last 7 days 				
	💗 🔘 Last 30 days				
	💎 🔕 Custom range				
	2022-07-06 to 2022-07-07 to				
	(Max range is 30 days, the dates will be auto-adjusted.)				
	Report size: 10 🔻 results per table				
Email report	Click this button to send summary reports by email, change the logo and set email schedules.				
WAN1/WAN2 usage					
y-axis	The y-axis shows the transmission speed of data sent or received through the WAN connection in kilobits per second (Kbps).				
x-axis	The x-axis shows the time period over which the traffic flow occurred.				
VPN usage					
y-axis	The y-axis shows the transmission speed of data sent or received through the VPN tunnel in kilobits per second (Kbps).				
x-axis	The x-axis shows the time period over which the traffic flow occurred.				
Security gateway by u	sage				
	This shows the index number of the Nebula Device.				
Name	This shows the descriptive name of the Nebula Device.				
Model	This shows the model number of the Nebula Device.				
Usage	This shows the amount of data that has been transmitted through the Nebula Device's WAN port.				
Client	This shows the number of clients currently connected to the Nebula Device.				
Location					
This shows the location	of the Nebula Devices on the map.				
Top applications by us	age				
	This shows the index number of the application.				
Application	This shows the application name.				
Category	This shows the name of the category to which the application belongs.				
Usage	This shows the amount of data consumed by the application.				
% Usage	This shows the percentage of usage for the application.				
Top ports by usage	This shows the top ten applications/services and the ports that identify a service.				
Name	This shows the service name and the associated port numbers.				
Usage	This shows the amount of data consumed by the service.				
% Usage	This shows the percentage of usage for the service.				

NCC User's Guide

LABEL	DESCRIPTION
Clients per day	
y-axis	The y-axis represents the number of clients.
x-axis	The x-axis represents the date.
Top operating systems	by usage
	This shows the index number of the operating system.
OS	This shows the operating system of the client device.
# Client	This shows how many client devices use this operating system.
% Client	This shows the percentage of top client devices which use this operating system.
# Usage	This shows the amount of data consumed by the client device on which this operating system is running.
% Usage	This shows the percentage of usage for top client devices which use this operating system.
Top clients by usage	
	This shows the index number of the client.
Description	This shows the descriptive name or MAC address of the client.
Usage	This shows the total amount of data transmitted and received by the client.
% Usage	This shows the percentage of usage for the client.
Top client device man	ufacturers by usage
	This shows the index number of the client device.
Manufacturer	This shows the manufacturer name of the client device.
Client	This shows how many client devices are made by the manufacturer.
% Client	This shows the percentage of top client devices which are made by the manufacturer.
Usage	This shows the total amount of data transmitted and received by the client device.
% Usage	This shows the percentage of usage for the client device.

Table 122	Site-wide > Mo	nitor > Security	/ gateway >	> Summary	report	(continued)
-----------	----------------	------------------	-------------	-----------	--------	-------------

9.3 Configure

Use the **Configure** menus to configure interface addressing, firewall, site-to-site VPN, captive portal, traffic shaping, authentication server and other Nebula Device settings for the Nebula Device of the selected site.

Note: Only one Security Appliance is allowed per site.

9.3.1 Interface Addressing

Use this screen to configure network mode, port grouping, interface address, static route and DDNS settings on the Nebula Device. To access this screen, click **Site-wide** > **Configure** > **Security gateway** > **Interface addressing**.

Note: If the gateway device of the site supports link aggregation, for example model NSG300, then the **Interface addressing** screen changes to allow you to configure link aggregation groups. For details, see Section 9.3.5 on page 446.

Mode:		 Network address translation Client traffic to the Intern 	n (NAT) eet is modified so that it	appears to have the se	curity gateway as its s	iource.		
		Router Client traffic to the intern interfaces to external inte	et is by routing result, v erfaces.	vhich means, the gatew	ay will not automatica	lly use SNAT for tr	affic it routes from	n internal
ort Group Setting			P3	P4	P5		PR	
Port Group 1:			0	0	0		0	
Port Group 2:			0	0	0		0	
nterface								
Name	IP address	Subnet mask	VLAN ID	Port Gr	oup	Guest		
LANI	100.25.11	255.255.255.0		Port 0	Froup 1		×	
LAN2	173:16:25:1	255.255.255.0		Port 0	iroup 2	on 🔵	×	
VLANIOO	1921681001	255,255,255,0	100	Port		_		
VI ANIO	192168101	255 255 255 0	10	Pole	(Please allow 1-2	Save or Co 2 minutes for cl	noel hanges to take	effect.
VI 41/250	102168 2501	255 255 255 0	250	Port	roup i	-		
- Add	102.100.200.1	2012002000	200	Port G	roup 1 👻	-		
- Add								
Static Route								
Name	Destination	Subnet m	vnek					
				Next	iop II ²			
s5 + Add bynamic DNS Automatic registratio	eres on	255.255.25	55.0	NUXT	199 P*		x •	
s5 Add Dynamic DNS Automatic registratio	9 H)	265 265 20 Contraction of the second	55.0	lic IP oddress of the sec	unity appliance chang	e1.	2 8	
eS Add Add Automatic registratic Beneral settings DDNS provider	on	285.285.20 C Dynamic DNS updates a DNS re DynDNS	cord each time the pub	Lic IP address of the sec	unty appliance chang	es.	2	
=5	90960 90	265 265 20 C Dynomic DNS updates a DNS re DynDNS DynDNS	cord each time the put	ic IP address of the sec	unity appliance chang	e3.	ef e	
+Add -Add Automatic registration Automatic registration DONS previder DONS type DONS type DONS account	on	265 265 20 C Dynamic DNS updates a DNS re DynDNS DynDNS	cord each time the pub	ic IP address of the sec	unity appliance chang	61	₫ •	
Add Automatic registratic Automatic registratic Beneral settings DDNS provider DDNS type DDNS account Username	se de la	255.255.25 Dynamic DNS updates a DNS ne DynDNS DynDNS	cord each time the put	iic IP address of the sec	unity appliance chang	63.	⊻ ∎	
45 Automatic PNS Automatic registratis Peneral settings DDNS provider DDNS type DDNS type Username Password	90960 90	255 255 25 Dynamic DNS updates a DNS re DynDNS DynDNS Immediates	cord each time the put	Re IP address of the sec	urity appliance chang	62	⊻ ∎	
	50	255 255 25 Dynomic DNS updates a DNB ref DynDNS DynDNS	cord each time the public cord each time the	Bc IP address of the sec	unity appliance chang	93.	⊻ ∎	
s5	909969 20	255.255.25	cord each time the put	lic IP address of the sec	unity appliance chang	93.	⊻ ∎	
Add Add	90960 201	285 285 28 C Dynamic DNS updates a DNS re DynDNS UynDNS G G G S S S S S S S S S S S S S	cord each time the put	in: IP address of the sec	urity appliance chang	61	ו	
s5	nets	285.285.25	cord each time the public cord each time the	BC IP address of the sec	urity appliance chang	93.	⊻ •	
s5 →Add Dynamic DNS Automatic registratio Deneral settings DDNS provider DDNS provider DDNS soccount Username Confirm password Domain name Primary binding add interface IP address	n ress	285.95.25	cord each time the put	lic IP address of the sec	unity appliance chang	93.	⊻ •	
s5	90966 on	255 255 25	cord each time the put	lic IP address of the sec	unity appliance chang	91	ו	
	ress	285.295.20	cord each time the pub	Road (urity appliance chang	63.	ו	
Add	900 00 00 00 00 00 00 00 00 00 00 00 00	255 255 25	S50 Cord each time the put	Re IP address of the sec	urity appliance chang	e1.	⊻ •	
s s s s s s s s s s s s s	1000 C	255 255 25	cord each time the put	ic IP address of the sec	urity appliance chang	e1.	ו	
s5	100 mess ress	285.985.25	2550	Rott	urity appliance chang	63.	ו	
s5	20 Pess 765	285.98.20	cord each time the put	nal)	urity appliance chong	63.	⊻ •	

Figure 150 Site-wide > Configure > Security gateway > Interface addressing

NCC User's Guide

LABEL	DESCRIPTION		
Network wide			
Mode	Select Network address translation (NAT) to have the Nebula Device automatically use SNAT for traffic it routes from internal interfaces to external interfaces.		
	Select Router to have the Nebula Device forward packets according to the routing policies. The Nebula Device does not automatically convert a packet's source IP address.		
Port Group Setting	Port groups create a hardware connection between physical ports at the layer-2 (data link, MAC address) level.		
	The physical LAN Ethernet ports are shown at the top (P3, P4, and so on) and the port groups are shown at the left of the screen. Use the radio buttons to select which ports are in each port group.		
	For example, select a port's Port Group 1 radio button to use the port as part of the first port group. The port will use the first group's IP address.		
	Note: You cannot select ports 1 and 2, as these ports are reserved for WAN usage.		
Interface	<u>.</u>		
By default, LAN1 is	created on top of port group 1 and LAN2 is on top of port group 2.		
Name	This shows the name of the interface (network) on the Nebula Device.		
IP address	This shows the IP address of the interface (network).		
Subnet mask	This shows the subnet mask of the interface (network).		
VLAN ID	This shows the ID number of the VLAN with which the interface (network) is associated.		
	If you have associated an SSID with the VLAN ID, the Smart VLAN screen displays after you change or delete the VLAN ID and click Save . You can exit the screen without saving, or apply your changes directly. If the Smart guest/VLAN network feature is enabled in the Site-Wide > Configure > General settings screen, you can select to apply the changes and update the SSID's VLAN setting as well.		
	Smart VLAN X		
	The VLAN interfaces: 220, 4095, 4096 are being used in the SSIDs settings detailed below. By modifying these interfaces, the SSIDs might not work properly. Smart VLAN allows to automatically update SSID settings with the new VLAN ID. Do you wish to continue with the changes? SSIDs SMame Interface Facebook wifi VLAN220 Close Update SSID & continue Continue		
Port group	This shows the name of the port group to which the interface (network) belongs.		

Table 123 Site-wide > Configure > Security gateway > Interface addressing

LABEL	DESCRIPTION
Guest	Select On to configure the interface as a Guest interface. Devices connected to a Guest interface will have Internet access but cannot communicate with each other directly or access network sources behind the Nebula Device.
	Otherwise, select Off to not use the interface as a Guest interface.
	Note: If the Smart guest/VLAN network feature is enabled in the Site-Wide > Configure > Site settings screen, the guest settings you configure for an interface also apply to the WiFi networks (SSIDs) associated with the same VLAN ID. For example, if you set an interface in VLAN 100 as a guest interface, the SSID that belongs to VLAN 100 will also act as a guest network.
2	Click this button to modify the network settings. See Section 9.3.1.1 on page 427 for detailed information.
1	Click this icon to remove a VLAN entry.
Add	Click this button to create a VLAN, which is then associated with one Ethernet interface (network). See Section 9.3.1.1 on page 427 for detailed information.
Static Route	
Name	This shows the name of the static route.
Destination	This shows the destination IP address.
Subnet mask	This shows the IP subnet mask.
Next hop IP	This shows the IP address of the next-hop gateway or the interface through which the traffic is routed. The gateway is a router or switch on the same segment as your Nebula Device's interfaces. It helps forward packets to their destinations.
2	Click this button to modify the static route settings. See Section 9.3.2.4 on page 437 for detailed information.
1	Click this icon to remove a static route.
Add	Click this button to create a new static route. See Section 9.3.2.4 on page 437 for detailed information.
Dynamic DNS	
Automatic registration	Click On to use dynamic DNS. Otherwise, select Off to disable it.
General Settings	
DDNS provider	Select your Dynamic DNS service provider from the drop-down list box.
	If you select User custom , create your own DDNS service.
DDNS type	Select the type of DDNS service you are using.
	Select User custom to create your own DDNS service and configure the DYNDNS Server , URL , and Additional DDNS Options fields below.
DDNS account	
Username	Enter the user name used when you registered your domain name.
Password	Enter the password provided by the DDNS provider.
Confirm password	Enter the password again to confirm it.
DDNS settings	
Domain name	Enter the domain name you registered.
Primary binding address	Use these fields to set how the Nebula Device determines the IP address that is mapped to your domain name in the DDNS server. The Nebula Device uses the Backup binding address if the interface specified by these settings is not available.

 Table 123
 Site-wide > Configure > Security gateway > Interface addressing (continued)

LABEL	DESCRIPTION
Interface	Select the interface to use for updating the IP address mapped to the domain name.
IP address	Select Auto if the interface has a dynamic IP address. The DDNS server checks the source IP address of the packets from the Nebula Device for the IP address to use for the domain name. You may want to use this if there are one or more NAT routers between the Nebula Device and the DDNS server.
	Note: The Nebula Device may not determine the proper IP address if there is an HTTP proxy server between the Nebula Device and the DDNS server.
	Select Custom if you have a static IP address. Enter the IP address to use it for the domain name.
	Select Interface to have the Nebula Device use the IP address of the specified interface.
Backup binding address	Use these fields to set an alternate interface to map the domain name to when the interface specified by the Primary binding address settings is not available.
Interface	Select the interface to use for updating the IP address mapped to the domain name.
IP address	Select Auto if the interface has a dynamic IP address. The DDNS server checks the source IP address of the packets from the Nebula Device for the IP address to use for the domain name. You may want to use this if there are one or more NAT routers between the Nebula Device and the DDNS server.
	Note: The Nebula Device may not determine the proper IP address if there is an HTTP proxy server between the Nebula Device and the DDNS server.
	Select Custom if you have a static IP address. Enter the IP address to use it for the domain name.
	Select Interface to have the Nebula Device use the IP address of the specified interface.
Enable wildcard	This option is only available with a DynDNS account.
	Enable the wildcard feature to alias sub-domains to be aliased to the same IP address as your (dynamic) domain name. This feature is useful if you want to be able to use, for example, www.yourhost.dyndns.org and still reach your hostname.
Mail exchanger	This option is only available with a DynDNS account.
	DynDNS can route email for your domain name to a mail server (called a mail exchanger). For example, DynDNS routes email for john-doe@yourhost.dyndns.org to the host record specified as the mail exchanger.
	If you are using this service, type the host record of your mail server here. Otherwise, leave the field blank.
Backup mail	This option is only available with a DynDNS account.
exchanger	Select this check box if you are using DynDNS's backup service for email. With this service, DynDNS holds onto your email if your mail server is not available. Once your mail server is available again, the DynDNS server delivers the mail to you. See www.dyndns.org for more information about this service.
DYNDNS Server	This field displays when you select User custom from the DDNS provider field above.
	Enter the IP address of the server that will host the DDNS service.

 Table 123
 Site-wide > Configure > Security gateway > Interface addressing (continued)

LABEL	DESCRIPTION
URL	This field displays when you select User custom from the DDNS provider field above.
	Enter the URL that can be used to access the server that will host the DDNS service.
Additional DDNS	This field displays when you select User custom from the DDNS provider field above.
Options	These are the options supported at the time of writing:
	 dyndns_system to specify the DYNDNS Server type – for example, dyndns@dyndns.org ip_server_name which should be the URL to get the server's public IP address – for example, http://myip.easylife.tw/

 Table 123
 Site-wide > Configure > Security gateway > Interface addressing (continued)

9.3.1.1 Local LAN (Add VLAN)

Click the Add button or click the Edit button in the Interface section of the Site-wide > Configure > Security gateway > Interface addressing screen.

Local LAN					>
Interface properties					
starface ture					
itendee type	VLAN	•			
nterface name					
	VLAN1				
IP address assignment					
^p address		~			
		^			
ubnet mask		×			
		^			
LAN ID	1	×	(1 - 4096)		
		^			
lort group	LAN2	-			
	0.041	-			
DHCP setting					
HCP	DHCP Server	-			
pool start address		×	Pool size	200	×
irst DNS server	NSG	•			
econd DNS server					
	None	•			
hird DNS server	No.				
	None	Ŧ			
irst WINS server		×	(Optional)		
econd WINS server		×	(Optional)		
ease time	O Infinite				

Figure 151	Site-wide > Configure 3	> Security gateway >	· Interface addressing:	Local LAN (VLAN)

LABEL	DESCRIPTION
Interface properties	
Interface type	Select VLAN to add a virtual interface. Note: This field only appears if the Nebula Device supports Link Aggregation Groups (LAGs). If the Nebula Device does not support LAGs, then VLAN is the default interface type.

LABEL	DESCRIPTION
Interface name	This field is read-only if you are editing an existing interface.
	Specify a name for the interface.
	The format of interface names is strict. Each name consists of $2 - 4$ letters (interface type), followed by a number (x). For most interfaces, x is limited by the maximum number of the type of interface. For VLAN interfaces, x is defined by the number you enter in the VLAN name field. For example, VLAN interfaces are vlan0, vlan1, vlan2, and so on.
IP address assignment	
IP address	Enter the IP address for this interface.
Subnet mask	Enter the subnet mask of this interface in dot decimal notation. The subnet mask indicates what part of the IP address is the same for all computers in the network.
VLAN ID	Enter the VLAN ID. This 12-bit number uniquely identifies each VLAN. Allowed values are 1 – 4094. (0 and 4095 are reserved.)
	Note: NCC will show an error message when the VLAN ID in the NSG interface is configured to be the same as the WAN port's VLAN ID.
Port group	Select the name of the port group to which you want the interface to (network) belong.
DHCP setting	
DHCP	Select what type of DHCP service the Nebula Device provides to the network. Choices are:
	None – the Nebula Device does not provide any DHCP service. There is already a DHCP server on the network.
	DHCP Relay – the Nebula Device routes DHCP requests to one or more DHCP servers you specify. The DHCP servers may be on another network.
	DHCP Server – the Nebula Device assigns IP addresses and provides subnet mask, gateway, and DNS server information to the network. The Nebula Device is the DHCP server for the network.
These fields appear if t	he Nebula Device is a DHCP Relay .
Relay server 1	Enter the IP address of a DHCP server for the network.
Relay server 2	This field is optional. Enter the IP address of another DHCP server for the network.
These fields appear if t	he Nebula Device is a DHCP Server .
IP pool start address	Enter the IP address from which the Nebula Device begins allocating IP addresses. If you want to assign a static IP address to a specific computer, click Add new under Static DHCP Table .
Pool size	Enter the number of IP addresses to allocate. This number must be at least one and is limited by the interface's Subnet mask . For example, if the Subnet mask is 255.255.255.0 and IP pool start address is 10.10.10.10, the Nebula Device can allocate 10.10.10.10 to 10.10.254, or 245 IP addresses.
First DNS server	Specify the IP addresses up to three DNS servers for the DHCP clients to use. Use one of the following ways to specify these IP addresses.
Second DNS server	Custom Defined – enter a static IP address.
Third DNS server	From ISP – select the DNS server that another interface received from its DHCP server
	NSG – the DHCP clients use the IP address of this interface and the Nebula Device works as a DNS relay.
First WINS server Second WINS server	Type the IP address of the WINS (Windows Internet Naming Service) server that you want to send to the DHCP clients. The WINS server keeps a mapping table of the computer names on your network and the IP addresses that they are currently using.

Table 121	Site wide > Configure	s Security aateway s l	Interface addressing, Local LAN ($\sqrt{ \Delta N }$
	JIC-WILC - COLLIGUIC	- Juconny guicwuy - i		V L/ (I N)

LABEL	DESCRIPTION
Lease time	Specify how long each computer can use the information (especially the IP address) before it has to request the information again. Choices are:
	infinite – select this if IP addresses never expire.
	days, hours, minutes – select this to enter how long IP addresses are valid.
Extended options	This table is available if you selected DHCP server .
	Configure this table if you want to send more information to DHCP clients through DHCP packets.
	Click Add new to create an entry in this table. See Section 9.3.2.3 on page 435 for detailed information.
Name	This is the option's name.
Code	This is the option's code number.
Туре	This is the option's type.
Value	This is the option's value.
	Click the edit icon to modify it.
	Click the remove icon to delete it.
Static DHCP Table	Configure a list of static IP addresses the Nebula Device assigns to computers connected to the interface. Otherwise, the Nebula Device assigns an IP address dynamically using the interface's IP pool start address and Pool size .
	Click Add new to create an entry in this table.
IP address	Enter the IP address to assign to a device with this entry's MAC address.
MAC	Enter the MAC address to which to assign this entry's IP address.
Description	Enter a description to help identify this static DHCP entry.
Close	Click Close to exit this screen without saving.
OK	Click OK to save your changes.

Talala 104 Cita videla S. Carafiavina S. Caravilla analazzaria S. Latarifazzaria and diservice and the	
-10000 ± 01 NTO WIDO ST ODIDUTO SNOCHTIN DOTOWOV SIDTOTOCO DODOCTORI OCOLLO	
TUDIE 124 - SIIE-WIGE / COLINQUIE / SECULIY QUIEWUY / INTENDUCE QUUIESSING. LOCULA	

9.3.2 Link Aggregation Groups

A Link Aggregation Group (LAG) combines multiple Ethernet ports into a single logical interface, in order to increase network bandwidth and/or availability.

Ports in the group can all connect to a target simultaneously, combining their bandwidth. A LAG can also offer higher network availability; if any port in the group becomes disconnected, the LAG can continue sending data using another port.

9.3.2.1 Interface Addressing with Link Aggregation Groups

If the Nebula Device of the selected site supports Link Aggregation Groups (LAGs), for example NSG300, you can create a LAG by clicking **Add**.

After you create a LAG, the **Port Group Settings** and **Interface** sections of the **Interface addressing** screen change. The new screen layout allows you to view and configure which ports are in a LAG.

Mode: Network address translation (NAT) Client traffic to the Internet is modif Router Client traffic to the Internet is by rou traffic it routes from Internal internal					adified so that it appears to have the security gateway as its source. routing result, which means, the gateway will not automatically use SNAT rraces to external interfaces.				
ort Grou	p Setting								
	P3	P4	P5	P6	P7	P8			
AN1~6	0	0	0	0	0	0			
AG1	0	0	0	0	0	0			
AG2	0	0	0	0	0	o			
iterface									
Name	IP address	Subnet mask	VLAN ID	Port Group		Guest			
LAN1	0.0.0.0	0.0.0.0		LANI		-			
LAN2	0.0.0.0	0.0.0.0		LAN2		-			
LAN3	0.0.0.0	0.0.0.0		LAN3		-			
LAN4	0.0.0.0	0.0.0.0		LAN4		•			
LAN5	0.0.0.0	0.0.0.0		LAN5		-			
LAN6	0.0.0.0	0.0.0.0		LAN6		-			
LAG1	192.168.1.10	255.255.255.0		LANI O					
LAG2	192.168.20.1	255.255.255.0		LANS @ LANG	8				

Figure 152 Site-wide > Configure > Security gateway > Interface addressing (LAG Interface Type)

Table 125	Site-wide >	Configure >	• Security gateway	> Interface	addressing (LAG	Interface Type)
-----------	-------------	-------------	--------------------	-------------	-----------------	-----------------

LABEL	DESCRIPTION				
Port Group	Select which port group or Link Aggregation Group (LAG) an Ethernet port belongs to.				
Sening	When LAGs are enabled, NCC adds each available LAN Ethernet port (port 3 and higher) to a separate port group, named LAN1, LAN2, LAN3, and so on. These default port groups cannot be modified or renamed.				
Interface					
Name	This shows the name of the interface (network) on the Nebula Device.				
IP address	This shows the IP address of the interface (network).				
Subnet mask	This shows the subnet mask of the interface (network).				

LABEL	DESCRIPTION					
VLAN ID	This shows the ID number of the VLAN with which the interface (network) is associated.					
	Note: If you have associated an SSID with the VLAN ID, the Smart VLAN screen displays after you change or delete the VLAN ID and click Save. You can exit the screen without saving, or apply your changes directly. If the Smart guest/ VLAN network feature is enabled in the Site-Wide > Configure > General settings screen, you can select to apply the changes and update the SSID's VLAN setting as well.					
	Smart VLAN X					
	The VLAN interfaces: 220, 4095, 4096 are being used in the SSIDs settings detailed below. By modifying these interfaces, the SSIDs might not work properly. Smart VLAN allows to automatically update SSID settings with the new VLAN ID. Do you wish to continue with the changes?					
	SSIDs Name Interface					
	Facebook wifi VLAN220					
	Close Update SSID & continue Continue					
Port group	For an Ethernet port, this shows the name of the port group to which the port belongs.					
	For a link aggregation group, this shows its member port groups.					
Guest	Select On to configure the interface as a Guest interface. Devices connected to a Guest interface will have Internet access but cannot communicate with each other directly or access network sources behind the Nebula Device.					
	Otherwise, select Off to not use the interface as a Guest interface.					
	Note: If the Smart guest/VLAN network feature is enabled in the Site-Wide > Configure > General settings screen, the guest settings you configure for an interface also apply to the WiFi networks (SSIDs) associated with the same VLAN ID. For example, if you set an interface in VLAN 100 as a guest interface, the SSID that belongs to VLAN 100 will also act as a guest network.					
2	Click this button to modify the network settings. See Section 9.3.1.1 on page 427 for detailed information.					
	If the interface is a member of a link aggregation group, you cannot edit the interface's network settings.					
1	Click this icon to delete a VLAN entry or link aggregation group.					
Add	Click this button to create a VLAN or link aggregation group.					
	 For details on creating a VLAN, see Section 9.3.1.1 on page 427. For details on creating a link aggregation group, see Section 9.3.2.2 on page 432. 					

Table 125 Site-wide > Configure > Security gateway > Interface addressing (LAG Interface Type)

9.3.2.2 Local LAN (LAG Interface Type)

Click the Add button or click the Edit button in the Interface section of the Site-wide > Configure > Security gateway > Interface addressing screen.
Figure 153	Site-wide > Configure > Security gateway > Interface addressing: Local LAN (LAG Interface
Type)	

Local LAN				×
Interface properties				
Interface type	LAG	-		
Interface name	LAG3	×		
LAG configuration				
Mode	active-backup	-		
Link monitoring	Mii	•		
Miimon	100	×		
Updelay	1	×		
Downdelay	0	×		
IP address assignment				
IP address		×		
Subnet mask		×		- 1
Port group	LAN2 ©			
DHCP setting				
DHCP	DHCP Server	-		
IP pool start address		×	Pool size 200	×
First DNS server	NSG	-		
Second DNS server				-
			C	lose OK

Table 126 Site-wide > Configure > Security gateway > Interface addressing: Local LAN (LAG Interface Type)

LABEL	DESCRIPTION
Interface properties	
Interface type	Select LAG to add a link aggregation group.
	Note: This field only appears if the Nebula Device supports Link Aggregation Groups (LAGs). If the Nebula Device does not support LAGs, a VLAN is created by default.
Interface name	Specify a name for the interface.
	This must be "LAG" plus a number, for example "LAG1".

NCC User's Guide

LABEL	DESCRIPTION
LAG Configuration	
Mode	Select a mode for this Link Aggregation Group (LAG) interface. Choices are as follows:
	active-backup: Only one port in the LAG interface is active and another port becomes active only if the active port fails.
	 802.3ad (IEEE 802.3ad Dynamic link aggregation): Link Aggregation Control Protocol (LACP) negotiates automatic combining of ports and balances the traffic load across the LAG link by sending LACP packets to the directly connected device that also implements LACP. The ports must have the same speed and duplex settings.
	 balance-alb (adaptive load balancing): Traffic is distributed according to the current load on each port by ARP negotiation. Incoming traffic is received by the current port. If the receiving port fails, another port takes over the MAC address of the failed receiving port.
Link Monitoring	Select how each link is monitored.
	mii (Media Independent Interface) – The Nebula Device monitors the state of the local interface only. The Nebula Device cannot tell if the link can transmit or receive packets.
	arp – The Nebula Device monitors the link by sending ARP queries. The Nebula Device then uses the reply to know if the link is up and that traffic is flowing through the link.
Miimom	This field displays for mii Link Monitoring. Set the interval in milliseconds that the system polls the Media Independent Interface (MII) to get the link's status.
Updelay	This field displays for mii Link Monitoring. Set the waiting time in milliseconds to confirm that a member interface link is up.
Downdelay	This field displays for mii Link Monitoring. Set the waiting time in milliseconds to confirm that a member interface link is down.
IP address assignment	
IP address	Enter the IP address for this interface.
Subnet mask	Enter the subnet mask of this interface in dot decimal notation. The subnet mask indicates what part of the IP address is the same for all computers in the network.
VLAN ID	Enter the VLAN ID. This 12-bit number uniquely identifies each VLAN. Allowed values are 1 – 4094. (0 and 4095 are reserved.)
	Note: NCC will show an error message when the VLAN ID in the NSG interface is configured to be the same as the WAN port's VLAN ID.
Port group	Select the name of the port group to which you want the interface to (network) belong.
DHCP setting	
DHCP	Select what type of DHCP service the Nebula Device provides to the network. Choices are:
	None – the Nebula Device does not provide any DHCP services. There is already a DHCP server on the network.
	DHCP Relay – the Nebula Device routes DHCP requests to one or more DHCP servers you specify. The DHCP servers may be on another network.
	DHCP Server – the Nebula Device assigns IP addresses and provides subnet mask, gateway, and DNS server information to the network. The Nebula Device is the DHCP server for the network.
These fields appear if t	he Nebula Device is a DHCP Relay .
Relay server 1	Enter the IP address of a DHCP server for the network.
Relay server 2	This field is optional. Enter the IP address of another DHCP server for the network.
These fields appear if t	he Nebula Device is a DHCP Server .

Table 126 Site-wide > Configure > Security gateway > Interface addressing: Local LAN (LAG Interface Type) (continued)

LABEL	DESCRIPTION					
IP pool start address	Enter the IP address from which the Nebula Device begins allocating IP addresses. If you want to assign a static IP address to a specific computer, click Add new under Static DHCP Table .					
Pool size	Enter the number of IP addresses to allocate. This number must be at least one and is limited by the interface's Subnet mask . For example, if the Subnet mask is 255.255.255.0 and IP pool start address is 10.10.10.10, the Nebula Device can allocate 10.10.10.10 to 10.10.10.254, or 245 IP addresses.					
First DNS server	Specify the IP addresses of up to three DNS servers for the DHCP clients to use. Use one of					
Second DNS server	The following ways to specify these iP addresses.					
Third DNS server	Custom Defined – enter a static IP address.					
	From ISP – select the DNS server that another interface received from its DHCP server.					
	NSG – the DHCP clients use the IP address of this interface and the Nebula Device works as a DNS relay.					
First WINS server	Enter the IP address of the WINS (Windows Internet Naming Service) server that you want to					
Second WINS server	send to the DHCP clients. The WINS server keeps a mapping table of the computer names on your network and the IP addresses that they are currently using.					
Lease time	Specify how long each computer can use the information (especially the IP address) before it has to request the information again. Choices are:					
	infinite – select this if IP addresses never expire					
	days, hours, minutes – select this to enter how long IP addresses are valid.					
Extended options	This table is available if you selected DHCP server.					
	Configure this table if you want to send more information to DHCP clients through DHCP packets.					
	Click Add new to create an entry in this table. See Section 9.3.2.3 on page 435 for detailed information.					
Name	This is the option's name.					
Code	This is the option's code number.					
Туре	This is the option's type.					
Value	This is the option's value.					
	Click the edit icon to modify it.					
	Click the remove icon to delete it.					
Static DHCP Table	Configure a list of static IP addresses the Nebula Device assigns to computers connected to the interface. Otherwise, the Nebula Device assigns an IP address dynamically using the interface's IP pool start address and Pool size .					
	Click Add new to create an entry in this table.					
IP address	Enter the IP address to assign to a device with this entry's MAC address.					
МАС	Enter the MAC address to which to assign this entry's IP address.					
Description	Enter a description to help identify this static DHCP entry.					
Close	Click Close to exit this screen without saving.					
ОК	Click OK to save your changes.					

Table 126	Site-wide > 0	Configure > :	Security go	ateway >	Interface	addressing:	Local LAN	(LAG Interfo	ace
Type) (co	ntinued)								

9.3.2.3 DHCP Option

Click the Add new button under Extended options in the Site-wide > Configure > Security gateway > Interface addressing: Local LAN screen.

DHCP Option		×
Option	User Defined	-
Name	User_Defined	×
Code	0	×
Туре	IP	-
First IP address		×
Second IP address		×
Third IP address		×
		Close OK

Figure 154 Site-wide > Configure > Security gateway > Interface addressing: Local LAN: DHCP Option

LABEL	DESCRIPTION
Option	Select which DHCP option that you want to add in the DHCP packets sent through the interface.
Name	This field displays the name of the selected DHCP option. If you selected User_Defined in the Option field, enter a descriptive name to identify the DHCP option.
Code	This field displays the code number of the selected DHCP option. If you selected User_Defined in the Option field, enter a number for the option. This field is mandatory.
Туре	This is the type of the selected DHCP option. If you selected User_Defined in the Option field, select an appropriate type for the value that you will enter in the next field. Misconfiguration could result in interface lockout.
Value	Enter the value for the selected DHCP option. For example, if you selected TFTP Server Name (66) and the type is TEXT , enter the DNS domain name of a TFTP server here. This field is mandatory.
First IP address	If you selected Time Server (4), NTP Server (41), SIP Server (120), CAPWAP AC (138), or TFTP
Second IP address	Server (150), you have to enter at least one IP address of the corresponding servers in these fields. The servers should be listed in order of your preference.
Third IP address	
First enterprise ID	If you selected VIVC (124) or VIVS (125), you have to enter at least one vendor's 32-bit
Second enterprise ID	company.

Table 127 Site-	vide > Configure	> Security gateway >	 Interface addressing: 	Local LAN: DHCP	Option
-----------------	------------------	----------------------	---	-----------------	--------

LABEL	DESCRIPTION
First class	If you selected VIVC (124) , enter the details of the hardware configuration of the host on
Second class	which the client is furning, or of industry consonium compliance.
First information	If you selected VIVS (125) , enter additional information for the corresponding enterprise
Second information	
First FQDN	If the Type is FQDN , you have to enter at least one domain name of the corresponding
Second FQDN	
Third FQDN	
Close	Click Close to exit this screen without saving.
ОК	Click OK to save your changes.

 Table 127
 Site-wide > Configure > Security gateway > Interface addressing: Local LAN: DHCP Option

9.3.2.4 Static Route

Click the Add button in the Static Route section of the Site-wide > Configure > Security gateway > Interface addressing screen.

atic Route		
Name:		×
Destination:		×
Subnet mask:		×
Next hop IP address:		>
	Close	

Figure 155 Site-wide > Configure > Security gateway > Interface addressing: Static Route

The following table describes the labels in this screen.

	Table 128	Site-wide >	Configure :	> Security	gateway	> Interface	addressing:	Static Route
--	-----------	-------------	-------------	------------	---------	-------------	-------------	--------------

LABEL	DESCRIPTION
Name	Enter a descriptive name for this route.
Destination	Specifies the IP network address of the final destination. Routing is always based on network number.
Subnet mask	Enter the IP subnet mask.
Next hop IP address	Enter the IP address of the next-hop gateway.
Close	Click Close to exit this screen without saving.
OK	Click OK to save your changes.

NCC User's Guide

9.3.3 Policy Route

Use policy routes and static routes to override the Nebula Device's default routing behavior in order to send packets through the appropriate next-hop gateway, interface or VPN tunnel.

A policy route defines the matching criteria and the action to take when a packet meets the criteria. The action is taken only when all the criteria are met. Use this screen to configure policy routes.

Click Site-wide > Configure > Security gateway > Policy route to access this screen.

Figure 156 Site-wide > Configure > Security gateway > Policy route

Policy ro	oute								
	Enabled	Туре	Protocol	Source IP	Source Port	Destination IP	Destination Port	Next-Hop	
↔ 1	\checkmark	VPN	Any	Any	Any	10.253.81.6	Any	Hub	2
$+ \operatorname{Add}$	Each site can	have at mos	st 50 policy route	es					

The following table describes the labels in this screen.

LABEL	DESCRIPTION
¢\$	Click the icon of a rule and drag the rule up or down to change the order.
Enabled	Select the check box to turn on the rule. Otherwise, clear the check box to turn off the rule.
Туре	This shows whether the packets will be routed to a different gateway (INTRANET), VPN tunnel (VPN) or outgoing interface (INTERNET).
Protocol	This displays the IP protocol that defines the service used by the packets. Any means all services.
Source IP	This is the source IP addresses from which the packets are sent.
Source Port	This displays the port that the source IP addresses are using in this policy route rule. The gateway applies the policy route to the packets sent from the corresponding service port. Any means all service ports.
Destination IP	This is the destination IP addresses to which the packets are transmitted.
Destination Port	This displays the port that the destination IP addresses are using in this policy route rule. Any means all service ports.
Next-Hop	This is the next hop to which packets are directed. It helps forward packets to their destinations and can be a router, VPN tunnel or outgoing interface.
2	Click this icon to change the profile settings.
1	Click this icon to remove the profile.
Add	Click this button to create a new policy route. See Section 9.3.4.1 on page 444 for more information.

Table 129 Site-wide > Configure > Security gateway > Policy route

9.3.3.1 Add/Edit policy route

Click the Add button or an edit icon in the Site-wide > Configure > Security gateway > Policy route screen to access this screen.

Type:	Internet Traffic	•
Protocol:	Any	-
Source IP:		×
Source Port:	Any	×
Destination IP:		×
Destination Port:	Any	×
Next-Hop:	WAN1	-

Fiaure 157	Site-wide >	Configure >	Security aateway	> Policy route: Add/Edit
inguio io,		Gormagoro	galona,	1 0107 1001017 (007) 2011

LABEL	DESCRIPTION
Туре	Select Internet Traffic to route the matched packets through the specified outgoing interface to a gateway (which is connected to the interface).
	Select Intranet Traffic to route the matched packets to the next-hop router or switch you specified in the Next-Hop field.
	Select VPN Traffic to route the matched packets through the VPN tunnel you specified in the Next-Hop field.
Protocol	Select TCP or UDP if you want to specify a protocol for the policy route. Otherwise, select Any.
Source IP	Enter a source IP address from which the packets are sent.
Source Port	Enter the port number (1 – 65535) from which the packets are sent. The Nebula Device applies the policy route to the packets sent from the corresponding service port. Any means all service ports.
Destination IP	Enter a destination IP address to which the packets go.
Destination IP Destination Port	Enter a destination IP address to which the packets go. Enter the port number (1 – 65535) to which the packets go. The Nebula Device applies the policy route to the packets that go to the corresponding service port. Any means all service ports.
Destination IP Destination Port Next-Hop	Enter a destination IP address to which the packets go. Enter the port number (1 – 65535) to which the packets go. The Nebula Device applies the policy route to the packets that go to the corresponding service port. Any means all service ports. If you select Internet Traffic in the Type field, select the WAN interface to route the matched packets through the specified outgoing interface to a Nebula Device connected to the interface.
Destination IP Destination Port Next-Hop	Enter a destination IP address to which the packets go. Enter the port number (1 – 65535) to which the packets go. The Nebula Device applies the policy route to the packets that go to the corresponding service port. Any means all service ports. If you select Internet Traffic in the Type field, select the WAN interface to route the matched packets through the specified outgoing interface to a Nebula Device connected to the interface. If you select Intranet Traffic in the Type field, enter the IP address of the next-hop router or switch.
Destination IP Destination Port Next-Hop	Enter a destination IP address to which the packets go. Enter the port number (1 – 65535) to which the packets go. The Nebula Device applies the policy route to the packets that go to the corresponding service port. Any means all service ports. If you select Internet Traffic in the Type field, select the WAN interface to route the matched packets through the specified outgoing interface to a Nebula Device connected to the interface. If you select Intranet Traffic in the Type field, enter the IP address of the next-hop router or switch. If you select VPN Traffic in the Type field, select the remote VPN gateway's site name.
Destination IP Destination Port Next-Hop Close	 Enter a destination IP address to which the packets go. Enter the port number (1 – 65535) to which the packets go. The Nebula Device applies the policy route to the packets that go to the corresponding service port. Any means all service ports. If you select Internet Traffic in the Type field, select the WAN interface to route the matched packets through the specified outgoing interface to a Nebula Device connected to the interface. If you select Intranet Traffic in the Type field, enter the IP address of the next-hop router or switch. If you select VPN Traffic in the Type field, select the remote VPN gateway's site name. Click this button to exit this screen without saving.

Table 130 Site-wide > Configure > Security gateway > Policy Route: Add/Edit

9.3.4 Firewall

By default, a LAN user can initiate a session from within the LAN and the Nebula Device allows the

response. However, the Nebula Device blocks incoming traffic initiated from the WAN and destined for the LAN. Use this screen to configure firewall rules for outbound traffic, application patrol, schedule profiles and port forwarding rules for inbound traffic.

Click Site-wide > Configure > Security gateway > Firewall to access this screen.

Note: The Nebula Device has the following hidden default firewall rules: LAN to WAN is allowed, WAN to LAN is blocked.

ecurity policy					
Policy rules	Destination	Dst port	Schedule	Description	
	× * 10.253.61.5	× Any	× * Always	REDMINE ACCESS	׍
	Any	Any	Always	Default rule	
	4				
Security gateway services	- Adu		Allowed cometer IDs		
	Service		any		
	Ping				>
			none		
	Web (local status & configuration))
Application Patrol					
Application monitor	Enable this setting to allow traffic				
Application profiles	Enable this option to ollow traffic analysi				
	There are no profiles defined for this site.				
hedule profiles					
	NewSchedule-1 used by 0 outb	ound rules			e i
	+ Add				
PALG					
SIP ALG					
SIP Signaling Port	5060 × *				
ADVANCED OPTIONS					
SIP Inactivity Timeout					
SIP Media Inactivity Timeout	120 × seconds				
SIP Signaling Inactivity Timeout	1800 \times seconds				
11 NAT					
	Enabled Uplink Pu	blic IP LA	AN IP	Allowed Remote IP	D
	€ 1 V WAN1 ▼	×	×	any	×
	+ Add				
				101700-000	
Virtual Server	Enabled Uplink Pro	otocol Public IP	Public port	LAN IP	

Figure 158 Site-wide > Configure > Security gateway > Firewall

Table 131	Sito wido >	Configuro N	Socurity	aatoway	> Firowall
	2116-MIGE ~		SECOM	yuiewuy -	

LABEL	DESCRIPTION			
Security Policy				
Policy rules				
¢	Click the icon of a rule and drag the rule up or down to change the order.			
Enabled	Select the check box to turn on the rule. Otherwise, clear the check box to turn off the rule.			
Policy	Select what the Nebula Device is to do with packets that match this rule.			
	Select Deny to silently discard the packets without sending a TCP reset packet or an ICMP destination-unreachable message to the sender.			
	Select Allow to permit the passage of the packets.			
	Select a pre-defined application patrol profile to have the Nebula Device take the action set in the profile when traffic matches the application patrol signatures. See Section 9.3.4.1 on page 444 for how to create an application patrol profile.			
Protocol	Select the IP protocol to which this rule applies. Choices are: TCP, UDP, and Any.			
Source	Specify the source IP addresses to which this rule applies. You can specify multiple IP addresses or subnets in the field separated by a comma (","). Enter any to apply the rule to all IP addresses.			
Destination	Specify the destination IP addresses or subnet to which this rule applies. You can specify multiple IP addresses or subnets in the field separated by a comma (","). Enter any to apply the rule to all IP addresses.			
Dst Port	Specify the destination ports to which this rule applies. You can specify multiple ports separated by a comma (","). Enter any to apply the rule to all ports.			
Schedule	Select the name of the schedule profile that the rule uses. Always means the rule is active at all times if enabled.			
Description	Enter a descriptive name of up to 60 printable ASCII characters for the rule.			
ŧ	Click this icon to remove the rule.			
Add	Click this button to create a new rule.			
Security gateway servi	ices			
Service	This shows the name of the service.			
Allowed remote IPs	Specify the IP address or a range of IP addresses (CIDR) with which the computer is allowed to access the Nebula Device using the service.			
	Any allows all IP addresses.			
Application Patrol				
Application monitor	Click On to enable traffic analysis for all applications and display information about the top 10 applications in the Site-wide > Dashboard: Traffic summary screen. Otherwise, select Off to disable traffic analysis for applications.			
Application profiles				
Name	This shows the name of the application patrol profile.			
Description	This shows the description of the application patrol profile.			
	Click this icon to change the profile settings.			
	Click this icon to remove the profile.			
Add	Click this button to create a new application patrol profile. See Section 9.3.4.1 on page 444 for more information.			
Schedule profiles				
	This shows the name of the schedule profile and the number of the outbound rules that are using this schedule profile.			

LABEL	DESCRIPTION
2	Click this icon to change the profile settings.
1	Click this icon to remove the profile.
Add	Click this button to create a new schedule profile. See Section 9.3.4.2 on page 445 for more information.
SIP ALG	
SIP ALG	Session Initiation Protocol (SIP) is an application-layer protocol that can be used to create voice and multimedia sessions over the Internet.
	Application Layer Gateway (ALG) allows the following applications to operate properly through the Nebula Device's NAT.
	Turn on the SIP ALG to detect SIP traffic and help build SIP sessions through the Nebula Device's NAT. Enabling the SIP ALG also allows you to use the application patrol to detect SIP traffic and manage the SIP traffic's bandwidth.
SIP Signaling Port	If you are using a custom UDP port number (not 5060) for SIP traffic, enter it here.
ADVANCED OPTIONS	
SIP Inactivity Timeout	Select this option to have the Nebula Device apply SIP media and signaling inactivity time out limits.
SIP Media Inactivity Timeout	Use this field to set how many seconds (1 – 86400) the Nebula Device will allow a SIP session to remain idle (without voice traffic) before dropping it.
	If no voice packets go through the SIP ALG before the timeout period expires, the Nebula Device deletes the audio session. You cannot hear anything and you will need to make a new call to continue your conversation.
SIP Signaling Inactivity Timeout	Most SIP clients have an "expire" mechanism indicating the lifetime of signaling sessions. The SIP user agent sends registration packets to the SIP server periodically and keeps the session alive in the Nebula Device.
	If the SIP client does not have this mechanism and makes no calls during the Nebula Device SIP timeout, the Nebula Device deletes the signaling session after the timeout period. Enter the SIP signaling session timeout value (1 – 86400).
NAT	
1:1 NAT	
A 1:1 NAT rule maps a	public IP address to the private IP address of a LAN server to give WAN users access.
If a private network se source IP address of th the server.	rver will initiate sessions to the outside clients, 1:1 NAT lets the Nebula Device translate the e server's outgoing traffic to the same public IP address that the outside clients use to access
د ۇ،	Click the icon of a rule and drag the rule up or down to change the order.
↓ Enabled	Select the check box to turn on the rule. Otherwise, clear the check box to turn off the rule.
Uplink	Select the interface of the Nebula Device on which packets for the NAT rule must be received.
Public IP	Enter the destination IP address of the packets received by the interface specified in this NAT rule.
	Note: To enable NAT loop-back, enter a specific IP address instead of any in this field. NAT loop-back allows communications between two hosts on the LAN behind the Nebula Device through an external IP address.
LAN IP	Specify to which translated destination IP address this NAT rule forwards packets.
Allowed Remote IP	Specify the remote IP address with which the computer is allowed to use the public IP address to access the private network server. You can specify a range of IP addresses.
	any allows all IP addresses.

Tabla 121	Site wide > Configure >	Soourity and avan	C S Eirough	(aantinuad)
IODIE IST	She-Mide > Coundrie >	seconiv dalewa	v > riewaii	ICOMINUED
	ene mele eengere	00000, go	,	

NCC User's Guide

LABEL	DESCRIPTION
Description	Enter a description for the rule.
1	Click this icon to remove the rule.
Add	Click this button to create a new 1:1 NAT mapping rule.
Virtual server	
¢	Click the icon of a rule and drag the rule up or down to change the order.
Enabled	Select the check box to turn on the rule. Otherwise, clear the check box to turn off the rule.
Uplink	Select the interface of the Nebula Device on which packets for the NAT rule must be received.
Protocol	Select the protocol (TCP, UDP, or Any) used by the service requesting the connection.
Public IP	Enter the destination IP address of the packets received by the interface specified in this NAT rule. Note: To enable NAT loop-back, enter a specific IP address instead of any in this field. NAT loop-back allows communications between two hosts on the
	LAN bening the Nebula Device through an external IP address.
Public port	Enter the translated destination port or range of translated destination ports if this NAT rule forwards the packet.
LAN IP	Specify to which translated destination IP address this NAT rule forwards packets.
Local port	Enter the original destination port or range of destination ports this NAT rule supports.
Allowed Remote IP	Specify the remote IP address with which the computer is allowed to use the public IP address to access the private network server. You can specify a range of IP addresses. any allows all IP addresses.
Description	Enter a description for the rule.
1	Click this icon to remove the rule.
Add	Click this button to create a new virtual server mapping rule.

Table 191 Sile-Wae > Cornigole > Seconty galeway > Filewaii (Corninoca)

9.3.4.1 Add application patrol profile

Application patrol provides a convenient way to manage the use of various applications on the network. It manages general protocols (for example, HTTP and FTP) and instant messenger (IM), peer-to-peer (P2P), Voice over IP (VoIP), and streaming (RSTP) applications. You can even control the use of a particular application's individual features (like text messaging, voice, video conferencing, and file transfers).

An application patrol profile is a group of categories of application patrol signatures. For each profile, you can specify the default action the Nebula Device takes once a packet matches a signature (forward, drop, or reject a service's connections and/or create a log alert).

Click the Add button in the Application patrol section of the Site-wide > Configure > Security gateway > Firewall screen to access this screen. Use the application patrol profile screens to customize action and log settings for a group of application patrol signatures.



Name Description (Option	nal)					×
Application Manag	ement					
Enabled	Category	Application		Policy		
Enabled	Category	Application All	•	Policy Drop	Ŧ	Û
Enabled	Category Instant mess	Application All	¥	Policy Drop	Ţ	ũ

LABEL	DESCRIPTION
Name	Enter a name for this profile for identification purposes.
Description	Enter a description for this profile.
Log	Select whether to have the Nebula Device generate a log (ON) or not (OFF) by default when traffic matches an application signature in this category.
Application managen	nent
Enabled	Select the check box to turn on the rule. Otherwise, clear the check box to turn off the rule.
Category	Select an application category.
Application	Select All or select an application within the category to apply the policy.
Policy	Select the default action for the applications selected in this category.
	Forward – the Nebula Device routes packets that matches these application signatures.
	Drop – the Nebula Device silently drops packets that matches these application signatures without notification.
	Reject – the Nebula Device drops packets that matches these application signatures and sends notification to clients.
1	Click this icon to remove the entry.
Add	Click this button to create a new application category and set actions for specific applications within the category.
	Enter a name to search for relevant applications and click Add to create an entry.
Close	Click this button to exit this screen without saving.
Create	Click this button to save your changes and close the screen.

Table 132 Site-wide > Configure > Security gateway > Firewall: Add an application profile

9.3.4.2 Create new schedule

Click the Add button in the Schedule Profiles section of the Site-wide > Configure > Security gateway > Firewall screen to access this screen.

me:							Templa	te:						
ewSchedule							× Alway	son						•
Day	Ava	ilability												
Sunday	on	00:00	02:00	04:00	06:00	08:00	10:00	12:00	14:00	16:00	18:00	20:00	22:00	0 24:00
Vonday	on	00:00	02:00	04:00	06:00	08:00	10:00	12:00	14:00	16:00	18:00	20:00	22:00	24:00
Fuesday	on	00:00	02:00	04:00	06:00	08:00	10:00	12:00	14:00	16:00	18:00	20:00	22:00	24:00
Vednesday	on	00:00	02:00	04:00	06:00	08:00	10:00	12:00	14:00	16:00	18:00	20:00	22:00	24:00
「hursday	on	00:00	02:00	04:00	06:00	08:00	10:00	12:00	14:00	16:00	18:00	20:00	22:00	24:00
Friday	on	00.00	02:00	04.00	06:00	08:00	10.00	12:00	14:00	16:00	18:00	20:00	22:00	24:00

Figure 160 Site-wide > Configure > Security gateway > Firewall: Add a schedule profile

Table 133 Site-wide > Configure > Security gateway > Firewall: Add a schedule profile

LABEL	DESCRIPTION
Name	Enter a descriptive name for this schedule for identification purposes.
Templates	Select a pre-defined schedule template or select Custom schedule and manually configure the day and time at which the associated firewall outbound rule is enabled.
Day	This shows the day of the week.
Availability	Click On to enable the associated rule at the specified time on this day. Otherwise, select Off to turn the associated rule off at the specified time on this day.
	speciry me noor and minore when me schedole begins and ends each day.
Close	Click this button to exit this screen without saving.
Add	Click this button to save your changes and close the screen.

9.3.5 Security Service

Use this screen to enable or disable the features available in the security pack for your Nebula Device, such as content filtering, Intrusion Detection and Prevention (IDP) and/or anti-virus. As to application patrol, go to the **Firewall** screen to configure it since you need to have a firewall rule for outbound traffic.

Content filtering allows you to block access to specific web sites. It can also block access to specific categories of web site content. IDP can detect malicious or suspicious packets used in network-based intrusions and respond instantaneously. Anti-virus helps protect your connected network from virus/spyware infection.

Click Site-wide > Configure > Security gateway > Security service to access this screen.

Note: Packet inspection signatures examine packet content for malicious data. Packet inspection applies to OSI (Open System Interconnection) layer-4 to layer-7 contents. You need to subscribe for IDP service in order to be able to download new signatures.

Figure 161	Site-wide >	Configure >	> Security	gateway	/ > Security	/ service
------------	-------------	-------------	------------	---------	--------------	-----------

Security service		
두 Content filtering		
Enabled		
	Interface	
	LAN2	
	VLANIOO CO	
	VLANIO	
	VLAN250	
Denied access message		
	This cotegory has been blocked_Predse contact the network dumin.	
Redirect URL		×
Black list		
		×
	FQDN(support wildcord)	
White list		×
	FQDN(support wildcard)	
Block Category		
Templates	Security 👻	
Test URL	× Test	
	Enter a uri to know website category	
Search category		×
	✓ Category list	
💎 Anti-Virus		
Signature Information	Current Version: 10.0.20200106.0	
	Signature Number: 632627	
	Released Date: 2020-01-06 08:33 (UTC+08:00)	
Enabled	(4)	
Black list		
		×
	File Pattern	
White list		
		×
	File Pattern	
• Intrusion Detection / Prevention		
Signature Information	Current Version: 314.391	
	Signature Number: 2143	
	Released Date: 2020-01-06 08:33 (UTC+08:00)	
Detection		
Prevention		

NCC User's Guide

Table 134 Site-wide > Configure > Security gateway > Security service

LABEL	DESCRIPTION
Content Filtering	
Enabled	Click ON to enable the content filtering feature on the Nebula Device. Otherwise, click OFF to disable it.
Interface	This shows the name of the interfaces created on the Nebula Device. Click ON to enable content filtering on the interfaces.
Denied access message	Enter a message to be displayed when content filter blocks access to a web page. Use up to 127 characters (0–9a–zA–Z;/?:@&=+ $\!-*'()$ %,"). For example, "Access to this web page is not allowed. Please contact the network administrator".
	It is also possible to leave this field blank if you have a URL specified in the Redirect URL field. In this case if the content filter blocks access to a web page, the Nebula Device just opens the web page you specified without showing a denied access message.
Redirect URL	Enter the URL of the web page to which you want to send users when their web access is blocked by content filter. The web page you specify here opens in a new frame below the denied access message.
	Use "http://" or "https://" followed by up to 262 characters (0–9a–zA–Z;/?:@&=+\$\ _!~*'()%). For example, http://192.168.1.17/blocked access.
Black list	Sites that you want to block access to, regardless of their content rating, can be blocked by adding them to this list.
	Enter host names such as www.bad-site.com into this text field. Do not enter the complete URL of the site – that is, do not include "http://". All sub-domains are also blocked. For example, entering "bad-site.com" also blocks "www.badsite.com", "partner.bad-site.com", "press.bad-site.com", and so on. You can also enter just a top level domain. For example, enter .com to block all .com domains.
	Use up to 127 characters (0–9a–z–). The casing does not matter.
White list	Sites that you want to allow access to, regardless of their content rating, can be allowed by adding them to this list.
	Enter host names such as www.good-site.com into this text field. Do not enter the complete URL of the site – that is, do not include "http://". All sub-domains are allowed. For example, entering "zyxel.com" also allows "www.zyxel.com", "partner.zyxel.com", "press.zyxel.com", and so on. You can also enter just a top level domain. For example, enter .com to allow all .com domains.
	Use up to 127 characters (0–9a–z–). The casing does not matter.
Block Category	
The Nebula Device pre When external databo that you configured in	events users from accessing web pages that match the categories that you select below. ase content filtering blocks access to a web page, it displays the denied access message the Denied access message field along with the category of the blocked web page.
Templates	Web pages are classified into a category based on their content. You can choose a pre- defined template that has already selected certain categories. Alternatively, choose Custom and manually select categories in this section to control access to specific types of Internet content.
Test URL	You can check which category a web page belongs to. Enter a web site URL in the text box.
	When the content filter is active, you should see the web page's category. The query fails if the content filter is not active.
	Content Filtering can query a category by full URL string (for example, http:// www.google.com/picture/index.htm), but HTTPS Domain Filter can only query a category by domain name ('www.google.com'), so the category may be different in the query result. Test URL displays both results in the test.

LABEL	DESCRIPTION
Search Category	Specify your desired filter criteria to filter the list of categories.
Category List	Click to display or hide the category list.
	These are categories of web pages based on their content. Select categories in this section to control access to specific types of Internet content.
Anti-Virus	
Signature Information	This shows the Current Version of the anti-virus definition, its Signature Number and the Released Date .
Enabled	Click On to enable anti-virus on the Nebula Device. Otherwise, select Off to disable it.
Black/White List	Use this to set up anti-virus black (blocked) and white (allowed) lists of virus file patterns.
File Pattern	For a black list entry, specify a pattern to identify the names of files that the Nebula Device should log and delete.
	For a white list entry, specify a pattern to identify the names of files that the Nebula Device should not scan for viruses.
	 Use up to 80 characters. Alphanumeric characters, underscores (_), dashes (-), question marks (?) and asterisks (*) are allowed.
	 A question mark (?) lets a single character in the file name vary. For example, use "a?.zip" (without the quotation marks) to specify aa.zip, ab.zip and so on.
	 Wildcards (*) let multiple files match the pattern. For example, use "*a.zip" (without the quotation marks) to specify any file that ends with "a.zip". A file named "testa.zip would match. There could be any number (of any type) of characters in front of the "a.zip" at the end and the file name would still match. A file named "test.zipa" for example would not match.
	 An * in the middle of a pattern has the Nebula Device check the beginning and end of the file name and ignore the middle. For example, with "abc*.zip", any file starting with "abc" and ending in ".zip" matches, no matter how many characters are in between.
	The whole file name has to match if you do not use a question mark or asterisk.
	• If you do not use a wildcard, the Nebula Device checks up to the first so characters of a file name.
Intrusion Detection / Pr	revention System
Signature Information	This shows the Current Version of the anti-intrusion definition, its Signature Number and the Released Date .
Detection	Click On to detect malicious or suspicious packets. Otherwise, select Off to disable it.
Prevention	Click On to identify and respond to intrusions. Otherwise, select Off to disable it.

Table 121	Site wide > Configure	> Socurity actoway	> Socurity convico	(continued)
10010134	sile-wide / Conliguie	> seconny galeway		(commueu)

9.3.6 Site-to-Site VPN

A virtual private network (VPN) provides secure communications between sites without the expense of leased site-to-site lines. Use this screen to configure a VPN rule.

Note: Site-to-site VPN do not support both sites behind NAT scenario.

Click Site-wide > Configure > Security gateway > Site-to-Site VPN to access this screen.

guring VPN with multiple sites is cu	mbersome. Use <u>VPN Orchestrator</u>	to save your time.		
Outgoing interface	WAN1	T.		
Local networks	Name	Subnet		Use VPN
	LAN1	192.168.1.0/24		
	LAN2	192.168.2.0/24		
VPN Area	Default	•		
Nebula VPN enable				
Nebula VPN topology	Split tunnel (send only	site-to-site traffic over the VPN)		
	Hub-and-Spoke	•		
Branch to branch VPN				
Hubs (peers connect to)		•		
Area communication				
NAT traversal	×	IP or FQDN		
	Remote VPN peer conn	nect to this Nebula gateway using the publ	ic IP address you specify.	
Remote VPN participants	Network		Subnet(s)	
e-wide settings				
tions in this section apply to this Nr	ebula gateway only.			
n-Nebula VPN peers				
nabled Name	Public IP	Private subnet 1	IPsec Preshared secr policy	et Availability
	*	× * ×	* Default	💿 * 🛛 This site 💿

Figure 162	Site-wide >	Configure >	 Security 	aateway >	 Site-to-Site 	VPN
		Connigoro	000000000000000000000000000000000000000	galonay		

|--|

LABEL	DESCRIPTION
Outgoing Interface	Select the WAN interface to which the VPN connection is going.
	Select AUTO to send VPN traffic through a different WAN interface when the primary WAN interface is down or disabled.
Preferred uplink	Specify the primary WAN interface through which the Nebula Device forwards VPN traffic when you set Outgoing Interface to AUTO .
Local networks	This shows the local networks behind the Nebula Device.
Name	This shows the network name.
Subnet	This shows the IP address and subnet mask of the computer on the network.
Use VPN	Click this to allow or disallow the computer connected to the LAN port to use VPN.
VPN Area	Select the VPN area of the site. For details, see Section 11.4.4.2 on page 524.

LABEL	DESCRIPTION					
Nebula VPN enable	Click this to enable or disable site-to-site VPN on the site's Nebula Device.					
	If you disable this setting, the site will leave the VPN area.					
Nebula VPN	This shows the VPN mode supported by the Nebula Device.					
lobology	Select a VPN topology.					
	Select Disable to not set a VPN connection.					
	In the Site-to-Site VPN topology, the remote IPSec device has a static IP address or a domain name. This Nebula Device can initiate the VPN tunnel.					
	In the Hub-and-Spoke VPN topology, there is a VPN connection between each spoke router and the hub router, which uses the VPN concentrator. The VPN concentrator routes VPN traffic between the spoke routers and itself.					
	In the Server-and-Client VPN topology, incoming connections from IPSec VPN clients are allowed. The clients have dynamic IP addresses and are also known as dial-in users. Only the clients can initiate the VPN tunnel.					
Branch to branch VPN	Enable this to allow spoke sites to communicate with each other in the VPN area. When disabled, spoke sites can only communicate with hub sites.					
Hubs (peers to connect to)	This field is available when you set Topology to Hub-and-Spoke . The field is configurable only when the Nebula Device of the selected site is the hub router.					
	You can select another site's name to have the Nebula Device of that site act as the hub router in the Hub-and-Spoke VPN topology.					
Area communication	Enable this to allow the site to communicate with sites in different VPN areas within the organization.					
NAT traversal	If the Nebula Device is behind a NAT router, enter the public IP address or the domain name that is configured and mapped to the Nebula Device on the NAT router.					
Server (client connect to)	This field is available when you set Topology to Server-and-Client . The field is configurable only when the Nebula Device of the selected site is the VPN server.					
	You can select another site's name to have the Nebula Device of that site act as the VPN server.					
Client-to-Client communication	Select On to allow VPN traffic to transmit between VPN clients by going through the server. The field is configurable only when the Nebula Device of the selected site is the VPN server.					
Remote VPN participants	This shows the remote (peer) Nebula Device's network name and address.					
Non-Nebula VPN peers	If the remote VPN gateway is not a Nebula Device, use this section to set up a VPN connection between it and the Nebula Device.					
+ Add	Click this button to add a non-Nebula gateway to the VPN area.					
Enabled	Select the check box to turn on the rule. Otherwise, clear the check box to turn off the rule.					
Name	Enter the name of the peer gateway.					
Public IP	Enter the public IP address of the peer gateway.					
Private subnet	Enter the local network address or subnet behind the peer gateway.					
IPSec policy	Click to select a pre-defined policy or have a custom one. See Section 9.3.6.1 on page 452 for detailed information.					
Preshared secret	Enter a pre-shared key (password). The Nebula Device and peer gateway use the key to identify each other when they negotiate the IKE SA.					

Table 135	Site-wide >	Configure >	Security	aateway >	Site_to_Site \	VPN	(continued)
		Configure	Jecomy	guiciay			

LABEL	DESCRIPTION
Availability	Select All sites to allow the peer gateway to connect to any Nebula Device in the organization through a VPN tunnel.
	Select This site and the peer gateway can only connect to the Nebula Device in this site through a VPN tunnel.
	You can also configure any specific sites in the organization,
Address	Enter the address (physical location) of the device.
Remove	Click the remove icon to delete the entry.
Add	Click this button to add a peer VPN gateway to the list.

 Table 135
 Site-wide > Configure > Security gateway > Site-to-Site VPN (continued)

9.3.6.1 Custom IPSec Policy

Click an existing IPSec Policy button in the Non-Nebula VPN peers section of the Site-wide > Configure > Security gateway > Site-to-Site VPN screen to access this screen.

Justom					×
reset		Default		•	
hase 1					
(E version		IKEv1		•	
ncryption		3DES		Ŧ	
uthenticati	on	SHA128		•	
iffie-Hellmo	an group	DH2		•	
ifetime (sec	onds)	86400			
hase 2					
hase 2 Set	Encrypt	ion		Authentication	
hase 2 Set Set 1	Encrypt 3DES	ion	•	Authentication SHA128	
hase 2 Set Set 1 Set 2	Encrypt 3DES None	ion	•	Authentication SHA128 None	•
hase 2 Set Set 1 Set 2 Set 3	Encrypt 3DES None None	ion	* *	Authentication SHA128 None None	•
hase 2 Set Set 1 Set 2 Set 3 FS group	Encrypt 3DES None	ion	•	Authentication SHA128 None None	•
hase 2 Set Set 1 Set 2 Set 3 FS group fetime (sec	Encrypt 3DES None None	ion None 86400	•	Authentication SHA128 None	•

Figure 163 Site-wide > Configure > Security gateway > Site-to-Site VPN: Custom IPSec Policy

Table 136	Site-wide >	Configure >	Security	gateway >	> Site-to-Site	VPN:	Custom	IPSec	Policy
-----------	-------------	-------------	----------	-----------	----------------	------	--------	-------	--------

LABEL	DESCRIPTION
Preset	Select a pre-defined IPSec policy, or select Custom to configure the policy settings yourself.
Phase 1	IPSec VPN consists of two phases: Phase 1 (Authentication) and Phase 2 (Key Exchange).
	A phase 1 exchange establishes an IKE SA (Security Association).

LABEL	DESCRIPTION
IKE version	Select IKEv1 or IKEv2.
	IKEv1 applies to IPv4 traffic only. IKEv2 applies to both IPv4 and IPv6 traffic. IKE (Internet Key Exchange) is a protocol used in setting up security associations that allows two parties to send data securely.
Encryption	Select which key size and encryption algorithm to use in the IKE SA. Choices are:
	DES – a 56-bit key with the DES encryption algorithm
	3DES – a 168-bit key with the DES encryption algorithm
	AES128 – a 128-bit key with the AES encryption algorithm
	AES192 – a 192-bit key with the AES encryption algorithm
	AES256 – a 256-bit key with the AES encryption algorithm
	The Nebula Device and the remote IPSec router must use the same key size and encryption algorithm. Longer keys require more processing power, resulting in increased latency and decreased throughput.
Authentication	Select which hash algorithm to use to authenticate packet data in the IKE SA.
	Choices are SHA128 , SHA256 , SHA512 and MD5 . SHA is generally considered stronger than MD5, but it is also slower.
	The remote IPSec router must use the same authentication algorithm.
Diffie-Hellman group	Select which Diffie-Hellman key group (DHx) you want to use for encryption keys. Choices are:
	DH1 – use a 768-bit random number
	DH2 – use a 1024-bit random number
	DH5 – use a 1536-bit random number
	DH14 – use a 2048-bit random number
	The longer the key, the more secure the encryption, but also the longer it takes to encrypt and decrypt information. Both routers must use the same DH key group.
Lifetime (seconds)	Type the maximum number of seconds the IKE SA can last. When this time has passed, the Nebula Device and remote IPSec router have to update the encryption and authentication keys and re-negotiate the IKE SA. This does not affect any existing IPSec SAs, however.
Advanced	Click this to display a greater or lesser number of configuration fields.
Mode	Select the negotiation mode to use to negotiate the IKE SA. Choices are:
	Main – this encrypts the Nebula Device's and remote IPSec router's identities but takes more time to establish the IKE SA
	Aggressive – this is faster but does not encrypt the identities
	The Nebula Device and the remote IPSec router must use the same negotiation mode.
Local ID	Enter the identity of the Nebula Device during authentication. Any indicates that the remote IPSec router does not check the identity of the Nebula Device.
Peer ID	Enter the identity of the remote IPSec router during authentication. Any indicates that the Nebula Device does not check the identity of the remote IPSec router.
Phase 2	Phase 2 uses the SA that was established in phase 1 to negotiate SAs for IPSec.

Table 136	Site-wide > Configure >	> Security gateway >	Site-to-Site VPN: Custom	IPSec Policy (continued)
-----------	-------------------------	----------------------	--------------------------	--------------------------

LABEL	DESCRIPTION			
Encryption	Select which key size and encryption algorithm to use in the IPSec SA. Choices are:			
	(none) – no encryption key or algorithm			
	DES – a 56-bit key with the DES encryption algorithm			
	3DES – a 168-bit key with the DES encryption algorithm			
	AES128 – a 128-bit key with the AES encryption algorithm			
	AES192 – a 192-bit key with the AES encryption algorithm			
	AES256 – a 256-bit key with the AES encryption algorithm			
	The Nebula Device and the remote IPSec router must both have at least one proposal that uses use the same encryption and the same key.			
	Longer keys are more secure, but require more processing power, resulting in increased latency and decreased throughput.			
Authentication	Select which hash algorithm to use to authenticate packet data in the IPSec SA.			
	Choices are None , MD5 , SHA128 , SHA256 , and SHA512 . SHA is generally considered stronger than MD5, but it is also slower.			
	The Nebula Device and the remote IPSec router must both have a proposal that uses the same authentication algorithm.			
PFS group	Select whether or not you want to enable Perfect Forward Secrecy (PFS) and, if you do, which Diffie-Hellman key group to use for encryption. Choices are:			
	None – disable PFS			
	DH1 – enable PFS and use a 768-bit random number			
	DH2 – enable PFS and use a 1024-bit random number			
	DH5 – enable PFS and use a 1536-bit random number			
	DH14 – enable PFS and use a 2048-bit random number			
	PFS changes the root key that is used to generate encryption keys for each IPSec SA. The longer the key, the more secure the encryption, but also the longer it takes to encrypt and decrypt information. Both routers must use the same DH key group.			
	PFS is ignored in initial IKEv2 authentication but is used when re-authenticating.			
Lifetime (seconds)	Enter the maximum number of seconds the IPSec SA can last. Shorter life times provide better security. The Nebula Device automatically negotiates a new IPSec SA before the current one expires, if there are users who are accessing remote resources.			
VPN tunnel interface (optional)			
IPSec VPN Tunnel Interface (VTI) encrypts or decrypts IPv4 traffic from or to the interface according to the IP routing table.				
VTI allows static routes interface. Therefore mo to the IPSec tunnel as Create a trunk using V	to send traffic over the VPN. The IPSec tunnel endpoint is associated with an actual (virtual) any interface capabilities such as Policy Route, Static Route, Trunk, and BWM can be applied soon as the tunnel is active. IPSec VTI simplifies network management and load balancing. PN tunnel interfaces for load balancing.			
This section is available when you select IKEv2 in the IKE Version field.				
IP address	Enter the IP address of the VPN tunnel interface.			
Subnet mask	Enter the subnet mask of this interface in dot decimal notation. The subnet mask indicates what part of the IP address is the same for all computers in the network			
Close	Click this button to exit this screen without saving.			
ОК	Click this button to save your changes and close the screen.			

Table 136 Site-wide > Configure > Security gateway > Site-to-Site VPN: Custom IPSec Policy (continued)

9.3.7 Remote Access VPN

Use this screen to configure the VPN client settings.

Internet Protocol Security (IPSec) VPN connects IPSec routers or remote users using IPSec client software. This standards-based VPN offers flexible solutions for secure data communications across a public network. IPSec is built around a number of standardized cryptographic techniques to provide confidentiality, data integrity and authentication at the IP layer.

The Layer 2 Tunneling Protocol (L2TP) works at layer 2 (the data link layer) to tunnel network traffic between two peers over another network (like the Internet). In L2TP VPN, an IPSec VPN tunnel is established first and then an L2TP tunnel is built inside it.

Click Site-wide > Configure > Security gateway > Remote access VPN to access this screen.

Figure 164	Site-wide > Configure > S	ecurity gateway > Remote access VPN

IPSec VPN server		2 Downlos
Outgoing interface	WAN1	
NAT traversal	× (IP or FQDN)	
Client VPN subnet	× *	
DNS name servers	Use Google Public DNS	
WINS	No WINS servers	
Secret	© *	
Authentication	Nebula Cloud Authentication 🔹	
L2TP over IPSec VPN server		
Client VPN subnet	*	
DNS name servers	Use Google Public DNS	
WINS	No WINS servers	
Secret	© *	
Authentication	Nebula Cloud Authentication	

Tabla 127	Site wide > Configure >	Socurity actoryou	> Domoto gooon V/DN
	ne-wide > Coundrie >	s seconi s dalewas	Zernole occess veix
10101010	ene male eengere	<i>secon, gener, e.,</i>	

LABEL	DESCRIPTION
Download VPN Client	Click this icon to download VPN client software.
IPSec VPN server	Select to enable the IPSec client feature on the Nebula Device. Otherwise, select Disable to turn it off.
Outgoing interface	Select the WAN interface to which the IPSec VPN connection is going.
NAT traversal	Enter the IP address or domain name of the NAT router if the IPSec VPN tunnel must pass through NAT (there is a NAT router between the IPSec devices).
Client VPN subnet	Specify the IP addresses that the Nebula Device uses to assign to the IPSec VPN clients.
DNS name servers	Specify the IP addresses of DNS servers to assign to the remote users.
	Select Use Google Public DNS to use the DNS service offered by Google. Otherwise, select Specify nameserver to enter a static IP address.
Custom nameservers	If you select Specify nameserver in the DNS name servers field, manually enter the DNS server IP addresses.
WINS	The WINS (Windows Internet Naming Service) server keeps a mapping table of the computer names on your network and the IP addresses that they are currently using.
	Select No WINS Servers to not send WINS server addresses to the users. Otherwise, select Specify nameserver to enter the IP addresses of WINS servers to assign to the remote users.
Custom nameservers	If you select Specify nameserver in the WINS field, manually enter the WINS server IP addresses.
Secret	Enter the pre-shared key (password) which is used to set up the IPSec VPN tunnel.
Authentication	Select how the Nebula Device authenticates a remote user before allowing access to the IPSec VPN tunnel.
L2TP over IPSec VPN server	Select to enable the L2TP over IPSec VPN feature on the Nebula Device. Otherwise, select Disable to turn it off.
Client VPN subnet	Specify the IP addresses that the Nebula Device uses to assign to the L2TP over IPSec VPN clients.
DNS name servers	Specify the IP addresses of DNS servers to assign to the remote users.
	Select Use Google Public DNS to use the DNS service offered by Google. Otherwise, select Specify nameserver to enter a static IP address.
Custom nameservers	If you select Specify nameserver in the DNS name servers field, manually enter the DNS server IP addresses.
WINS	The WINS (Windows Internet Naming Service) server keeps a mapping table of the computer names on your network and the IP addresses that they are currently using.
	Select No WINS Servers to not send WINS server addresses to the users. Otherwise, select Specify nameserver to enter the IP addresses of WINS servers to assign to the remote users.
Custom nameservers	If you select Specify nameserver in the WINS field, manually enter the WINS server IP addresses.
Secret	Enter the pre-shared key (password) which is used to set up the L2TP over IPSec VPN tunnel.

LABEL	DESCRIPTION
Authentication	Select how the Nebula Device authenticates a remote user before allowing access to the L2TP over IPSec VPN tunnel.
VPN provision script	Send an email to help automatically configure VPN settings on client devices so that the devices can remotely access this Nebula Device. The email contains two scripts; one for mac OS and iOS devices, and one for Windows 8 and Windows 10 devices.
	You can send the email to one or more email addresses.
	• If Authentication is set to Nebula Cloud Authentication, the default email address list contains all authorized VPN user email addresses and your email address.
	• If Authentication is set to AD and RADIUS Authentication, the default email address list contains your user email address.

Table 137	Site-wide > Configure >	Security aateway >	Remote access	VPN (continued)
	one mae comgere	galona,		

9.3.8 Captive Portal

Use this screen to configure captive portal settings for each interface. A captive portal can intercept network traffic until the user authenticates his or her connection, usually through a specifically designated login web page.

Click Site-wide > Configure > Security gateway > Captive portal to access this screen.

Captive portal	
nterface	LAN1 Captive portal on this interface is direct access. You can change this setting <u>here.</u>
Themes	
BUTTON	
Default Modern	Copy of Modern Copy of Modern
Click-to-continue/Sign-	-on page
Logo	Upload a logo No logo
Message	Terms go here!
Success page	
Message	Success!
External captive portal	URL
Use URL:	
	To use custom captive portal page, please download the zip file and edit them.
	Download the customized captive portal page example.
Captive portal behavio	r
After the captive portal p where the user should go	2399 Stay on Captive portal authenticated successfully page
	O To promotion URL: X
	Save or Cancel
	(Please allow 1-2 minutes for changes to take effect.)

Figure 165 Site-wide > Configure > Security gateway > Captive portal

LABEL	DESCRIPTION		
Interface	Select the Nebula Device's interface (network) to which the settings you configure here is applied.		
Themes	This section is not configurable when External captive portal URL is set to ON.		
	 Click the Preview icon at the upper right of a theme image to display the portal page in a new frame. Click the Copy icon to create a new custom theme (portal page). Click the Edit icon of a custom theme to go to a screen, where you can view and configure the details of the custom portal pages. See Section 9.3.8.1 on page 460. Click the Remove icon to delete a custom theme. 		
Click-to-continue/Sian	-on page		
This section is not confi	aurable when External captive portal URL is set to ON.		
Logo	This shows the logo image that you uploaded for the customized login page.		
	Click Upload a logo and specify the location and file name of the logo graphic or click Browse to locate it. You can use the following image file formats: GIF, PNG, or JPG.		
Message	Enter a note to display below the title. Use up to 1024 printable ASCII characters. Spaces are allowed.		
Success page			
Message	Enter a note to display on the page that displays when a user logs in successfully. Use up to 1024 printable ASCII characters. Spaces are allowed.		
External captive porta	External captive portal URL		
Use URL	Select On to use a custom login page from an external web portal instead of the one built into the NCC. You can configure the look and feel of the web portal page.		
	Specify the login page's URL; for example, http://IIS server IP Address/login.asp. The Internet Information Server (IIS) is the web server on which the web portal files are installed.		
Captive portal behavi	or		
After the captive portal page where the user should go?	Select To promotion URL and specify the URL of the web site/page to which the user is redirected after a successful login. Otherwise, select Stay on Captive portal authenticated successfully page .		

9.3.8.1 Custom Theme Edit

Use this screen to check what the custom portal pages look like. You can also view and modify the CSS values of the selected HTML file. Click a custom login page's **Edit** button in the **Site-wide** > **Configure** > **Security gateway** > **Captive portal** screen to access this screen.

Captive portal / Copy of Modern	> Seconry	gu	ieway > cap				← Back to config
Theme name	success.ht	tml	user_login.html	click_to_continue.html	color.css	icon.css	layout.css
Copy of Modern 🗹	<> sc						
Font							
Color			Welcome to lan2				
B 255 # FFFFFF Select							
						Agree	
				F	Powered by Z	YXEL	

Figure 166 Site-wide > Configure > Security gateway > Captive portal: Edit

LABEL	DESCRIPTION
Back to config	Click this button to return to the Captive portal screen.
Theme name	This shows the name of the theme. Click the edit icon to change it.
Font	Click the arrow to hide or display the configuration fields.
	To display this section and customize the font type and/or size, click an item with text in the preview of the selected custom portal page (HTML file).
Color	Click the arrow to hide or display the configuration fields.
	Click an item in the preview of the selected custom portal page (HTML file) to display this section and customize its color, such as the color of the button, text, window's background, links, borders, and so on.
	Select a color that you want to use and click the Select button.
HTML/CSS	This shows the HTML file name of the portal page created for the selected custom theme. This also shows the name of the CSS files created for the selected custom theme.
	Click an HTML file to display the portal page. You can also change colors and modify the CSS values of the selected HTML file.
$\langle \rangle$	Click this button to view and modify the CSS values of the selected HTML file. It is recommended that you do NOT change the script code to ensure proper operation of the portal page.
	Click this button to preview the portal page (the selected HTML file).
Save	Click this button to save your settings for the selected HTML file to the NCC.
Apply	Click this button to save your settings for the selected HTML file to the NCC and apply them to the Nebula Device in the site.

Table 139 Site-wide > Configure > Security gateway > Captive portal: Edit

9.3.9 Network Access Method

Use this screen to enable or disable web authentication on an interface.

Click Site-wide > Configure > Security gateway > Network access method to access this screen.

Figure 167 Site-wide > Configure > Security gateway > Network access method

letwork access method	
Interfaces: LAN1 -	
Network Access	
	O Disable Users can access the network directly
	Click-to-continue
	 Sign-on-with Nebula Cloud Authentication
Walled garden	on
Walled garden ranges	×
	What do I enter here? One IP address/domain in one line to specify your walled garden. Example: *.zyxel.com www.zyxel.com 192.1681.0/24
Captive portal access attribute	
Self-registration	Allow users to create accounts with auto authorized
Login on multiple client devices	Multiple devices access simultaneously
NCAS disconnection behavior	Ð
	 Allowed: Client devices can access the network without signing in, except they are explicitly blocked Limited: Only currently authorized clients and whitelisted client devices will be able to access the network
	Client devices can access the network without signing in, except they are explicitly blocked Limited: Only currently authorized clients and whitelisted client devices will be able to access the network

LABEL	DESCRIPTION	
Interfaces	Select the Nebula Device's interface (network) to which the settings you configure here is applied.	
Network Access	Select Disable to turn off web authentication.	
	Select Click-to-continue to block network traffic until a client agrees to the policy of user agreement.	
	Select Sign-on with to block network traffic until a client authenticates with an external RADIUS or AD server through the specifically designated web portal page. Select Nebula Cloud Authentication or an authentication server that you have configured in the Site-wide > Configure > Security gateway > Gateway settings screen (see Section 9.3.11 on page 466).	
	Select Two-Factor Authentication to require that the user log in using both their password and a Google Authenticator code. To log in, users must have Two-Factor Authentication enabled on their account and have setup Google Authenticator on their mobile device.	
Walled garden	This field is not configurable if you set Network Access to Disable.	
	Select to turn on or off the walled garden feature.	
	With a walled garden, you can define one or more web site addresses that all users can access without logging in. These can be used for advertisements for example.	
Walled garden ranges	Specify walled garden web site links, which use a domain name or an IP address for web sites that all users are allowed to access without logging in.	
Captive portal access	attribute	
Self-registration	This field is available only when you select Sign-on with Nebula Cloud authentication in the Network Access field.	
	Select Allow users to create accounts with auto authorized or Allow users to create accounts with manual authorized to display a link in the captive portal login page. The link directs users to a page where they can create an account before they authenticate with the NCC. For Allow users to create accounts with manual authorized, users cannot log in with the account until the account is authorized and granted access. For Allow users to create accounts with auto authorized, users can just use the registered account to log in without administrator approval.	
	Select Don't allow users to create accounts to not display a link for account creation in the captive portal login page.	
Login on multiple	This field is available only when you select Sign-on with in the Network Access field.	
	Select Multiple devices access simultaneously if you allow users to log in as many times as they want as long as they use different IP addresses.	
	Select One device at a time if you do NOT allow users to have simultaneous logins.	
NCAS disconnection behavior	This field is available only when you select Sign-on with Nebula Cloud Authentication in the Network Access field.	
	Select Allowed to allow any users to access the network without authentication when the NCAS (Nebula Cloud Authentication Server) is not reachable.	
	Select Limited to allow only the currently connected users or the users in the white list to access the network.	

Table 140 Site-wide > Configure > Security gateway > Network access method

9.3.10 Traffic Shaping

Use this screen to configure maximum bandwidth and load balancing on the Nebula Device.

Click Site-wide > Configure > Security gateway > Traffic shaping to access this screen.

plink comgaration	
WAN1	486623 Up(kb/s)
WANT	466623 Down(kb/s)
	O unlimited Up(kb/s)
WAN2	Unlimited Down(kb/s)
WAN load balancing algorithm:	Failover
Prefer WAN:	WAN1 -
WAN Connectivity check:	Check Default Gateway Check this address
lobal bandwidth limits	
lobal bandwidth limits Per-client limit:	Source First IP Source Last IP Destination IPs Port(s)
lobal bandwidth limits Per-client limit:	Source First IP Source Last IP Destination IPs Port(s) 192168.100.1 × * [192168.100.254 × * [any × * [any
lobal bandwidth limits Per-client limit:	Source First IP Source Last IP Destination IPs Port(s) 192.168.100.1 × * 192.168.100.254 × * any × * any + Add
lobal bandwidth limits Per-client limit:	Source First IP Source Last IP Destination IPs Port(s) 192.168.100.1 × * 192.168.100.254 × * any × * any + Add
Iobal bandwidth limits Per-client limit: ession Control UDP Session Time Out:	Source First IP Source Last IP Destination IPs Port(s 192168100.1 × * 192168100.254 × * any × * any + Add

Figure 168 Site-wide > Configure > Security gateway > Traffic shaping

LABEL	DESCRIPTION				
Uplink configuration					
WAN 1	Set the amount of upstream/downstream bandwidth for the WAN interface.				
WAN 2	Click a lock icon to change the lock state. If the lock icon for a WAN interface is locked, the bandwidth limit you set applies to both inbound and outbound traffic. If the lock is unlocked, you can set inbound and outbound traffic to have different transmission speeds.				
WAN load balancing algorithm	 Select a load balancing method to use from the drop-down list box. Select Least Load First to send new session traffic through the least utilized WAN interface. Select Round Robin to balance the traffic load between interfaces based on their respective weights (bandwidth). An interface with a larger weight gets more chances to transmit traffic than an interface with a smaller weight. For example, if the weight ratio of WAN 1 and WAN 2 interfaces is 2:1, the Nebula Device chooses WAN 1 for two sessions' traffic and WAN 2 for one session's traffic in each round of three new sessions. Select Failover to send traffic through a second WAN interface when the primary WAN interface is down or disabled. 				
Prefer WAN	Specify the primary WAN interface through which the Nebula Device forwards traffic.				
	This field is available when you set WAN load balancing algorithm to Failover .				
WAN Connectivity check	The interface can regularly check the connection to the gateway you specified to make sure it is still available. The Nebula Device resumes routing to the gateway the first time the gateway passes the connectivity check.				
	If the WAN connection is down (the check fails), the Nebula Device will switch (failover) to use a redundant WAN connection.				
	 Select Check Default Gateway to use the default gateway for the connectivity check. Select Check this address to specify a domain name or IP address for the connectivity check. 				
	Note: If you select Check this address but the IP address you specified cannot be reached through the primary WAN interface, the Nebula Device will switch to the other one even if the primary WAN connection is still up. Make sure your Nebula Device supports multiple WAN interfaces and both WAN connections are configured properly before you select Check this address .				
	This field is available when you set WAN load balancing algorithm to Failover.				
Global bandwidth limits					
Per-client limit	You can limit a client's outbound or inbound bandwidth.				
Source First IP	Enter the first IP address in a range of source IP addresses for which the Nebula Device applies the rule.				
Source Last IP	Enter the last IP address in a range of source IP addresses for which the Nebula Device applies the rule.				
Destination IPs	Enter the destination IP addresses for which the Nebula Device applies the rule.				
	Enter any if the rule is effective for every destination.				
Port(s)	Enter the port numbers (1 – 65535) to which the packets go. The Nebula Device applies the rule to the packets that go to the corresponding service port. any means all service ports.				
Protocol	Select TCP or UDP if you want to specify a protocol for the rule. Otherwise select Any.				
	Any means the rule is applicable to all services.				

LABEL	DESCRIPTION	
Down/Up	Set the maximum upstream/downstream bandwidth for traffic from an individual source IP address.	
	Click a lock icon to change the lock state. If the lock icon is locked, the bandwidth limit you set applies to both inbound and outbound traffic. If the lock is unlocked, you can set inbound and outbound traffic to have different transmission speeds.	
Priority	Enter a number between 1 and 7 to set the priority for traffic that matches this policy. The smaller the number, the higher the priority.	
	Traffic with a higher priority is given bandwidth before traffic with a lower priority.	
	Click this icon to remove the rule.	
Add	Click this button to create a new rule.	
Session Control		
UDP Session Time Out	Set how many seconds the Nebula Device will allow a UDP session to remain idle (without UDP traffic) before closing it.	
Default Session per Host	Set a common limit to the number of concurrent NAT/Security Policy sessions each client computer can have.	
	If only a few clients use peer to peer applications, you can raise this number to improve their performance. With heavy peer to peer application use, lower this number to ensure no single client uses too many of the available NAT sessions.	

Table 141 Site-wide > Configure > Security gateway > Traffic shaping (continued)

9.3.11 Gateway Settings

Use this screen to configure DNS settings and external AD (Active Directory) server or RADIUS server that the Nebula Device can use in authenticating users.

AD (Active Directory) is a directory service that is both a directory and a protocol for controlling access to a network. The directory consists of a database specialized for fast information retrieval and filtering activities. You create and store user profile and login information on the external server.

This screen also lets you configure the addresses of walled garden web sites that users can access without logging into the Nebula Device. The settings in this screen apply to all networks (interfaces) on the Nebula Device. If you want to configure walled garden web site links for a specific interface, use the **Network access method** screen.

Click Site-wide > Configure > Security gateway > Gateway settings to access this screen.

Figure 169	Site-wide >	Configure >	Security	gateway	> Gateway	y settings
			/	0 /		

Address Record				
FQDN		IP Address		
d.nebula.zyxel.com		* 52.19.85.221		× * 🝵
www.nebula.zyxel.com		× * 52.84.248.13		× * 💼
s.nebula.zyxel.com		× * 18.202.42.142		× * 🍵
+ Add				
Domain Zone Forwarder				
Domain Zone		IP Address		Interface
		×)*	*	LAN1 🔻 💼
I thentication Server My AD Server				
thentication Server My AD Server Name	Server address	Backup server addres	is Port	AD domo
thentication Server My AD Server Name ADTest	Server address	Backup server addres	ss Port	AD domo
thentication Server My AD Server Name ADTest	Server address	Backup server addres	ss Port	AD domo ×]* [zyxel.co
thentication Server My AD Server ADTest ADTest ADTest ADTest ADTest	Server address	Backup server addres	ss Port	AD domo
thentication Server My AD Server ADTest ADTEST ADTE	Server address * 192.168.8.1 Server address	Backup server addres	ss Port	AD domo ×)* [zyxel.co Secret
thentication Server My AD Server ADTest	Server address × 192168.8.1 Server address ×	Backup server addres	ss Port X 389 ss Port X 1812	AD domo ×)* [zyxel.co Secret ×]*
thentication Server My AD Server ADTest ADTEST ADT	Server address × 192168.8.1 Server address ×	Backup server addres	ss Port X 389 ss Port X 1812	AD domo × * zyxel.co Secret × *
thentication Server My AD Server ADTest ADTEST ADT	Server address × 192168.8.1 Server address ×	Backup server addres	ss Port × 389 ss Port × 1812	AD domo ×)* [zyxel.co Secret ×]* [
thentication Server My AD Server ADTest ADTEST ADT	Server address × 192168.8.1 Server address ×	Backup server addres	ss Port × 389 ss Port × 1812	AD domo × * zyxel.co Secret × *
thentication Server My AD Server ADTest ADTEST ADT	Server address × 192168.8.1 Server address × * This is global walled second priority is the if needed only allow	Backup server addres × * Backup server addres × * garden configuration. All web authne interface walled garden policy. specify interface, please go to Netw	ss Port × 389 ss Port × 1812 entication interface will match vork access method configure	AD domo
ADTest ADTest ADTest ADTest ADTest ADTest ADTest	Server address × 192168.8.1 Server address × * This is global walled second priority is the fineded only allow	Backup server addres X * Backup server addres X * Backup server addres X * garden configuration All web authne interface walled garden policy. 's specify interface, please go to Network	ss Port × 389 ss Port × 1812 entication interface will match vork access method configure	AD doma × * zyxel.co Secret × *

Table 142	Site-wide >	Configure >	Security	gateway >	Gateway	settings
			/	0 /	/	

LABEL	DESCRIPTION				
DNS					
Address Record	This record specifies the mapping of a Fully-Qualified Domain Name (FQDN) to an IP address. An FQDN consists of a host and domain name. For example, www.zyxel.com.tw i a fully qualified domain name, where "www" is the host, "zyxel" is the third-level domain, "com" is the second-level domain, and "tw" is the top level domain.				
FQDN	Enter a host's fully qualified domain name.				
	Use "*." as a prefix in the FQDN for a wildcard domain name (for example, *.example.com).				
IP Address	Enter the host's IP address.				
 	Click this icon to remove the entry.				
Add	Click this button to create a new entry.				
Domain Zone Forwarder	This specifies a DNS server's IP address. The Nebula Device can query the DNS server to resolve domain zones for features like VPN, DDNS and the time server. When the Nebula Device needs to resolve a domain zone, it checks it against the domain zone forwarder entries in the order that they appear in this list.				
Domain Zone	A domain zone is a fully qualified domain name without the host. For example, zyxel.com.tw is the domain zone for the www.zyxel.com.tw fully qualified domain name. Whenever the Nebula Device needs to resolve a zyxel.com.tw domain name, it can send a query to the recorded name server IP address.				
IP Address	Enter the DNS server's IP address.				
Interface	Select the interface through which the Nebula Device sends DNS queries to the specified DNS server.				
	Click this icon to remove the entry.				
Add	Click this button to create a new entry.				
Authentication Server					
My AD Server					
Name	Enter a descriptive name for the server.				
Server address	Enter the address of the AD server.				
Backup server address	If the AD server has a backup server, enter its address here.				
Port	Specify the port number on the AD server to which the Nebula Device sends authentication requests. Enter a number between 1 and 65535.				
AD domain	Specify the Active Directory forest root domain name.				
Domain admin	Enter the name of the user that is located in the container for Active Directory Users, who is a member of the Domain Admin group.				
Password	Enter the password of the Domain Admin user account.				
Advanced	Click to open a screen where you can select to use Default or Custom advanced settings. See Section 9.3.11.1 on page 469.				
	Click this icon to remove the server.				
Add	Click this button to create a new server.				
My RADIUS server					
Name	Enter a descriptive name for the server.				
Server address	Enter the address of the RADIUS server.				
Backup server address	If the RADIUS server has a backup server, enter its address here.				
LABEL	DESCRIPTION				
-------------------------	---				
Port	Specify the port number on the RADIUS server to which the Nebula Device sends authentication requests. Enter a number between 1 and 65535.				
Secret	Enter a password (up to 15 alphanumeric characters) as the key to be shared between the external authentication server and the Nebula Device.				
	The key is not sent over the network. This key must be the same on the external authentication server and the Nebula Device.				
Advanced	Click to open a screen where you can select to use Default or Custom advanced settings. See Section 9.3.11.1 on page 469.				
.	Click this icon to remove the server.				
Add	Click this button to create a new server.				
Walled garden					
Global Walled garden	With a walled garden, you can define one or more web site addresses that all users can access without logging in. These can be used for advertisements for example.				
	Specify walled garden web site links, which use a domain name or an IP address for web sites that all users are allowed to access without logging in.				

Table 142 Site-wide > Configure > Security gateway > Gateway settings (continued)

9.3.11.1 Advanced Settings

Click the Advanced column in the Site-wide > Configure > Security gateway > Gateway settings screen to access this screen.

Elguro 170	Sito wido >	Configuro	Socurity	aatoways	Catoway	(cottings:	Avancad
rigule 170	slie-wide /	Configure -	Secomy	guiewuy -	Gulewu	/ seminys. /	Auvunceu

Advanced			×
Preset:	Default		
Timeout:	5 ×	(1-300 seconds)	
Case-Sensitive User Name:	off		
NAS IP Address	1270.0.1 ×		
		Close	ОК

The following table describes the labels in this screen.

 Table 143
 Site-wide > Configure > Security gateway > Gateway settings: Advanced

LABEL	DESCRIPTION
Preset	Select Default to use the pre-defined settings, or select Custom to configure your own settings.
Timeout	Specify the timeout period (between 1 and 300 seconds) before the Nebula Device disconnects from the server. In this case, user authentication fails.
	Search timeout occurs when either the user information is not in the servers or the AD or server is down.
Case-Sensitive User Name	Click ON if the server checks the case of the user name. Otherwise, click OFF to not configure your user name as case-sensitive.
NAS IP Address	This field is only for RADIUS.
	Enter the IP address of the NAS (Network Access Server).



LABEL	DESCRIPTION
Close	Click this button to exit this screen without saving.
OK	Click this button to save your changes and close the screen.

Tabla 112	Sito wido >	Configuros	Socurity actory	Catoway cattings	Advancad	(continued)
10010 143	Slie-wide ~		- Seconi y gulewuy -	Gulewuy senings.	Auvunceu	
			, , , , , ,			1

CHAPTER 10 Mobile Router

10.1 Overview

This chapter discusses the menus that you can use to monitor the Nebula-managed Mobile Routers in your network and configure settings even before a Mobile Router is deployed and added to the site.

A Nebula Mobile Router is an LTE or NR cellular 5G indoor or outdoor router that can be managed by Nebula. It is referred to as a Nebula Device in this chapter. To identify whether your Nebula Device is an outdoor or indoor device and view the list of the Nebula Devices that can be managed through the NCC, go to Help > Support tools > Device function table.

10.2 Configuration

From the navigation panel, click **Site-wide** > **Devices** > **Mobile router** and the following screen appears. The **Configuration** screen allows you to view the information of your indoor or outdoor Nebula Device in a selected site. To edit the **Name**, **MAC** address, Serial number, **Description**, **Address**, and **Tags** of your Nebula Device, click the edit icon ($[\mathcal{O}]$) in the **Configuration** field.

Note: Only one Mobile Router is allowed per site.



Figure 171 Site-wide > Devices > Mobile router > Configuration (Indoor)

Figuro 172	Sita wida >	Davicas >	Mobile router >	Configuration	(Outdoor)	
rigule 172	slie-wide /	Devices /		Configuration	(OUIGOOI)	

		Map Photo
Configuration 🖉		
Nome:	D8.EC.E5.20.80.56	O Q Position device 1 O Floor plan Map Satellite
MAC address.	D8ECE5208056	This device is being located by GEO IP
Serial number:	\$210Z45007757 (NR7101)	 IP-based geolocation services can only provide an approximate measure of geolocation accuracy. <u>Acknowledge</u>
Description		
Address:		
Togs		
Cellular IP Passthrough:	Enabled 🖾	
Firmware availability:	N/A	
Ourseast Managine	N/A	興訊科技設份有限公司
Current version:		
Configuration status:	Not up to date	Ŷ

10.2.1 Configuration: Edit

The following screen displays after you click the edit icon. Use the **Site-wide** > **Devices** > **Mobile router** > **Configuration**: **Edit** screen to configure your indoor and outdoor Nebula Device information. You can also move the Nebula Device to another site.

Configuration		
comgaration		
Name:	DR-EC-E5-01-76-DR	×
MAC address:	DR-EC-E5-04-78-DR	
Serial number:	52107470051012 (NR5101)	
	$\widehat{\square}$ Remove from this site	
Description:		
		×
Taos		
Tuga.	lag	
Address:		×
	Move map marker	

Figure 173 Site-wide > Devices > Mobile router > Configuration: Edit

Table 144 Site-wide > Devices > Mobile router > Configuration: Edit

LABEL	DESCRIPTION
Configuration	
Name	Enter a descriptive name for the Nebula Device.
MAC address	This shows the MAC address of the Nebula Device.
Serial number	This shows the serial number of the Nebula Device.
Description	Enter a user-specified description for the Nebula Device.
Tags	Enter a user-specified tag for the Nebula Device.
Address	Enter a user-specified address for the Nebula Device.
Save	Click Save to save your changes.
Cancel	Click Cancel to exit this screen without saving.

10.2.2 Home Networking

To configure the Home networking setting, click the edit icon (2) in the Home networking field.

Note: Home Networking is only available for the LTE3301-PLUS and NR5101.

Figure 174 Site-wide > Devices > Mobile router > Configuration: Home networking (Indoor)

bbile router		
Configuration 🗹		
Name:	NR5101	
MAC address:	20-21-12-0A-00-20	
Serial number:	202112040020 (NR5101)	
Description:		
Address:		
Tags:		
Home networking:	192.168.1.1	
WiFi settings:	Configure on local GUI	
Firewall settings:	Configure on local GUI	
Firmware availability:	NA	
Current Version:	N/A	
Configuration status:	Not up to date	
WAN Usage:	0 bytes in last day	
LAN Usage:	0 bytes in last day	
2.4G WLAN Usage:	0 bytes in last day	
5G WLAN Usage:	0 bytes in last day	

The following Site-wide > Devices > Mobile router > Configuration > Home networking: Edit screen displays. Use this screen to configure the LAN IP address and DHCP server settings of your indoor Nebula Device.

Figure 175	Site-wide > Devices >	Mobile router > Configuration > Ho	ome networking: Edit
		0	0

Edit		×
IP address assignment		
IP address	192.168.3.1 × *	
Subnet mask	255.255.255.0 × *	
DHCP setting		
DHCP Server		
IP pool start add	dress 192.168.3.33 × Pool size 20	× *
Lease time	O Infinite	
	0 1 × * 0 × * 0	× *
	days hours(Optional) minutes(Optional)	
	Close	e OK

The following table describes the labels in this screen.

LABEL	DESCRIPTION
IP address assignment	
IP address	Enter the IP address for this interface.
Subnet mask	Enter the subnet mask of this interface in dot decimal notation. The subnet mask indicates what part of the IP address is the same for all computers in the network.
DHCP setting	
DHCP Server	Select this to disable or enable the DHCP server.
IP pool start address	Enter the IP address from which the Nebula Device begins allocating IP addresses.
Pool size	Enter the number of IP addresses to allocate. This number must be at least one and is limited by the interface's Subnet mask . For example, if the Subnet mask is 255.255.255.0 and IP pool start address is 10.10.10.10, the security gateway can allocate 10.10.10.10 to 10.10.254, or 245 IP addresses.
Lease time	Specify how long each computer can use the information (especially the IP address) before it has to request the information again. Choices are: Infinite – select this if IP addresses never expire; days , hours , minutes – select this to enter how long IP addresses are valid.
Close	Click Close to exit this screen without saving.
ОК	Click OK to save your changes.

Table 145	Site wide > Devices	Mabile routers	Configuration	Llama naturating Edit
10010143	SIIG-MIDE - DEVICES		Configuration /	nome nerworking. Eur

10.2.3 Cellular IP Passthrough

To configure the cellular IP passthrough setting, click the edit icon (2) in the **Cellular IP Passthrough** field. IP passthrough allows a LAN computer on the local network of the Nebula Device to have access to web services using a public IPv4 address. When IP passthrough is configured, all traffic is forwarded to the LAN computer and will not go through NAT.

Note: As of this writing, cellular IP passthrough is for NR7101 and LTE7461 only.

Configuration \square		
Name:	D0-E0-E5-20-00-56	
MAC address:	D0-E0-E5-30-00-56	
Serial number:	010745007757 (NR7101)	
Description:		
Address:		
Tags:		
Cellular IP Passthrough:	Enablec 🗹	
Firmware availability:	N/A	
Current Version:	N/A	
Configuration status:	Not up to date	
WAN Usage	0 bytes in last day	

Figure 176 Site-wide > Devices > Mobile router > Configuration: Cellular IP Passthrough (Outdoor)

The following **Site-wide > Devices > Mobile router > Configuration > Cellular IP Passthrough: Edit** screen displays. Use this screen to disable or enable IP passthrough on your outdoor Nebula Device. Slide the switch to the right to enable IP passthrough.

Figure 177 Site-wide > Devices > Mobile router > Configuration > Cellular IP Passthrough: Edit

Edit	×
IP Passthrough mode: Note: Enable IP Passthrough to allow internet traffic to go to the LAN computer be the router without going through NAT.	ehind
Close	ок

Table 116	Sita wida	> Davicas >	Mobile router >	Configuration >	Cellular IP Passtbrough: Edit
	ane-wide -	- Devices -		Configuration /	Celiular ir rassiniougn. Eali

LABEL	DESCRIPTION
IP Passthrough mode	This displays if IP passthrough is enabled on the Nebula Device.
Close	Click Close to exit this screen without saving.
OK	Click OK to save your changes.

10.2.4 Firmware Status

Go back to the **Site-wide** > **Devices** > **Mobile router** > **Configuration** screen to view the firmware version and WAN/LAN/WLAN usage of your indoor or outdoor Nebula Device.

Note: LAN Usage, 2.4G WLAN Usage and 5G WLAN Usage are only available for indoor Nebula Devices.

Figure 178	Site-wide > Devices > Mobile router > Configuration > Firmwa	re status

bile router	
Configuration 🗹	
Name:	NR5101
MAC address:	2U-21-12-UN-UU-2U
Serial number:	202112040020 (NR5101)
Description:	
Address:	
Tags:	
Home networking:	192.168.1.1 🗹
WiFi settings:	Configure on local GUI
Firewall settings:	Configure on local GUI
Firmware availability:	<u>N/A</u>
Current Version:	N/A
Configuration status:	Not up to date
WAN Usage:	0 bytes in last day
LAN Usage:	0 bytes in last day
2.4G WLAN Usage:	0 bytes in last day
5G WLAN Usage:	0 bytes in last day

LABEL	DESCRIPTION
WiFi settings	Configure the Nebula Device's WiFi settings using its Web Configurator. Refer to the Nebula Device's User's Guide for more information.
	Note: This field is NOT configurable.
Firewall settings	Configure the Nebula Device's firewall settings using its Web Configurator. Refer to the Nebula Device's User's Guide for more information.
	Note: This field is NOT configurable.
Firmware availability	The NCC automatically detects whether the firmware is up-to-date or not. Click the value in the Firmware availability field to go to the Site-wide > Configure > Firmware management screen and configure your Firmware management settings.
Current Version	This shows the firmware version currently installed on the Nebula Device.
Configuration status	This shows whether the configuration on the Nebula Device is up-to-date.
WAN Usage	This shows the total amount of data consumed by the Nebula Device on the WAN (uplink/ downlink) in the past 24 hours.
LAN Usage (indoor NCCs only)	This shows the total amount of data consumed by the Nebula Device on the LAN (upllink/ downlink) in the past 24 hours.
2.4G WLAN Usage (indoor NCCs only)	This shows the total amount of data consumed by the Nebula Device on the 2.4G WiFi network (uplink/downlink) in the past 24 hours.
5G WLAN Usage (indoor NCCs only)	This shows the total amount of data consumed by the Nebula Device on the 5G WiFi network (uplink/downlink) in the past 24 hours.

Table 147 Site-wide > Devices > Mobile router > Configuration > Firmware status

10.3 Map/Photo

Click the **Map** tab. This shows the location of the Nebula Device on Google map. To upload a photo of the Nebula Device, select the **Photo** tab.



Figure 179 Site-wide > Devices > Mobile router > Map

Table 148	Site-wide > Devices > Mobile router > Map/Photo	
-----------	---	--

LABEL	DESCRIPTION
Мар	This shows the location of the Nebula Device on Google Maps (Map view or Satellite imagery view) or on a floor plan. Click Floor plan to display a list of existing floor plans. Each floor plan has a drawing that shows the rooms scaled and viewed from above. Drag-and-drop your Nebula Device directly on the Google map or click Position device to update the Nebula Device's address (physical location).
	Position device X
	Update my device's location. <u>What is this?</u> Use the device's IP address (GEO IP).
	 Get my location from web browser. Use the following address or coordinates.
	×
	Cancel Update
	 Select GEO IP to use the public IP address of the Nebula Device. Select Get my location from web browser to use the public IP address of the computer accessing the NCC portal.
	Select Use the following address or coordinates to enter the complete address or coordinates of the Nebula Device.
	Note: Nebula Devices that are offline cannot use GEO IP.
Photo	This shows the photo of the Nebula Device. Click Add to upload up to five photos of your Nebula Device. Click the remove icon () to delete a photo.

10.4 Live Tools

Use live tools to view various interface information, system/security logs, perform diagnostics, reboot or establish a remote connection to the Nebula Device.

Figure 180	Site-wide >	Devices >	Mobile router >	> Live tools >	Traffic	(Example)
------------	-------------	-----------	-----------------	----------------	---------	-----------

VAN status	Cellular info	Traffic	LAN stations 🕨	WLAN stations 🕨	Ping	Traceroute	DNS lookup	Reboot	Log	💎 Remote configurato
Traffic:								2	5.7 Kbps	(23.7 Kbps 🕢 2.0 Kbps 🅐)
28.8 Kbps 7					\wedge	~		~~	~ ~	
19.2 Kbps										
9.6 Kbps										
10 bos						<u> </u>		<u> </u>		
	14:30	14:32	14:34	14:36		14:38	14:40		14:4	2 14:44

Note: In the Traffic, LAN stations, and WLAN stations screens, click the pause icon (1) to stop getting data for the respective screens. Alternatively, click the play icon () to continue.

The following table describes the labels in this screen.

LABEL	DESCRIPTION
WAN Status	This shows the connection status of the Ethernet WAN interface. See Section 10.4.1 on page 480 for more information.
Cellular info	This shows the connection status of the cellular WAN interface. See Section 10.4.2 on page 481 for more information.
Traffic	This shows the Nebula Device traffic statistics.
	The y-axis represents the transmission rate for uplink and downlink traffic.
	The x-axis represents the time period over which the traffic flow occurred.
LAN stations	This shows the Nebula Device's connected LAN clients' MAC address and IPv4 Address.
WLAN stations (indoor NCCs only)	This shows the Nebula Device's connected WiFi clients' MAC address, SSID name, IPv4 address, Signal strength, Security, Channel, Tx rate, Rx rate, Tx/Rx, and Capability. See Section 10.4.4 on page 488 for more information.
Ping	Enter the hostname or IP address of a computer that you want to perform ping from the Nebula Device in order to test a connection and click Ping .
	This can be used to determine if the Nebula Device and the computer are able to communicate with each other.
Traceroute	Enter the domain name or IP address of a computer that you want to perform traceroute from the Nebula Device and click Run . This determines the path a packet takes to the specified computer.
DNS lookup	Enter a host domain name and click Run to resolve the IP address for the specified domain name.
Reboot	Click this button to restart the Nebula Device.
Log	Select this to display System log and Security log entries in the past 24 hours.
Remote configurator	Click Establish to use TCP (Transmission Control Protocol) port 443 to establish a remote connection to this Nebula Device. The Nebula Device will create a reverse SSH (Secure SHell) connection.
	After clicking Ok , NCC will provide a remote connection IPv4 address and service port number. For example, https://63.35.218.205:31479. Use this IPv4 address and port to connect to the Nebula Device to open the Web Configurator. The remote session will be available for 30 minutes.
	In case the connection cannot be established, confirm that the network allows Port 443.
	Note: Remote configuration is only available if the Nebula Device is running the latest firmware. Otherwise, Device firmware is not up to date , please update it . will appear when you click Establish .

10.4.1 WAN Status

Go to the Site-wide > Devices > Mobile router > Live tools > WAN status screen to view the Ethernet WAN status of the Nebula Device.

WAN status Cellular info	Traffic 🕨	LAN stations 🕨	WLAN stations 🕨	Ping	Traceroute	DNS lookup	Reboot	Log	💎 Remote configurato
C									
Mode	R	outer Mode							
Status	L	P							
IP Address	1	0.253.40.59							
Primary DNS server	1	0.253.40.254							
IPv6 Address	2	001:b030:710f:0:daec:e	5ff:fe1a:9e3						
Access Technology	E	thernet WAN							

Figure 181 Site-wide > Devices > Mobile router > Live tools > WAN status

Table 150 Site-wide > Devices > Mobile router > Live tools > WAN sta
--

LABEL	DESCRIPTION
C	Click this button to reload the data-related frames on this page.
Mode	This displays which operating mode the Nebula Device is assigned to.
Status	This displays whether the Nebula Device is online/offline.
IP Address	This shows the LAN IPv4 address of the Nebula Device.
Primary DNS server	The shows the first DNS server address assigned by the ISP.
IPv6 Address	This shows the LAN IPv6 address of the Nebula Device.
Access Technology	This displays the type of the network (such as NR, LTE, Ethernet WAN) to which the Nebula Device is connecting.
Signal Strength	This show the signal strength of the Nebula Device.

10.4.2 Cellular Info

Go to the Site-wide > Devices > Mobile router > Live tools > Cellular Info screen to view the cellular WAN status of the Nebula Device.

N status Cellular info Tr	affic 🕨 LAN stations 🕨 Ping Trac	eroute DNS lookup Reboot Log	g 👎 Remote configurator 📴
Module Information		Service Information	
IMEI	357927100010811	Access Technology	LTE
Module SW Version SIM Status	EG06ALAR02A07M4G	Band	LTE_BC7
SIM Card Status	Available	RSSI	-57
IMSI	466924000089642	Cell ID	81552675
ICCID	89886920040000896422	Physical Cell ID	95
PIN Protection	Disable	UL Bandwidth (MHz)	10
PIN Remaining Attempts	3	DL Bandwidth (MHz)	10
IP Passthrough Status		RFCN	3400
IP Passthrough Enable	Enable	RSRP	-87
IP Passthrough Mode Cellular Status	Dynamic	RSRQ	-10
Cellular Status	Up	RSCP	0
Data Roaming	Disable	EcNo	0
Operator	Chunghwa Telecom	TAC	13700
PLMN	46692	LAC	0
NR-NSA Information		RAC	0
MCC		BSIC	0
MNC		SINR	14
Physical Cell ID	0	CQI	8
RFCN	0	MCS	0
Band		RI	2
RSRP	0	PMI	0
RSRQ	0	SCC Information	
SINR GNSS Information	0		
Enable	true		
Scan OnBoot	false		
Scan Status			
HDOP	0.0		
Display Format			
Latitude	0		
Longitude	0		
Elevation	0.0		
Positioning Mode	0		
Course Over Ground	0.0		
Speed Over Ground	0.0		
Last Fix Time	None		
Number Of Satellites	0		
11/2011/2012 1970 1970 1970 1970 1970	\$\$		

Figure 182 Site-wide > Devices > Mobile router > Live tools > Cellular Info

Table 151 Site-wide > Devices > Mobile router > Live tools > Cellular Info

LABEL	DESCRIPTION
Module Information	
IMEI	This shows the International Mobile Equipment Identity of the Nebula Device.

LABEL	DESCRIPTION
Module SW Version	This shows the software version of the cellular network module.
SIM Status	
SM Card Status	This displays the SIM card status:
	None – the Nebula Device does not detect that there is a SIM card inserted.
	Available – the SIM card could either have or does not have PIN code security.
	Locked – the SIM card has PIN code security, but you did not enter the PIN code yet.
	Blocked – you entered an incorrect PIN code too many times, so the SIM card has been locked. Call the ISP (Internet Service Provider) for a PUK (Pin Unlock Key) to unlock the SIM card.
	Error – the Nebula Device detected that the SIM card has errors.
IMSI	This displays the International Mobile Subscriber Identity (IMSI) of the installed SIM card. An IMSI is a unique ID used to identify a mobile subscriber in a mobile network.
ICCID	Integrated Circuit Card Identifier (ICCID). This is the serial number of the SIM card.
PIN Protection	A PIN (Personal Identification Number) code is a key to a SIM card.
	This field shows Enable if PIN Protection is enabled. Otherwise, this field shows Disable.
PIN Remaining Attempts	This is how many more times you can try to enter the PIN code before the ISP blocks your SIM card.
IP Passthrough Status	
IP Passthrough	This displays if IP passthrough is enabled on the Nebula Device.
ENGDIE	IP passthrough allows a LAN computer on the local network of the Nebula Device to have access to web services using the public IP address. When IP passthrough is configured, all traffic is forwarded to the LAN computer and will not go through NAT.
IP Passthrough Mode	This displays the IP passthrough mode.
	This displays Dynamic and the Nebula Device will allow traffic to be forwarded to the first LAN computer requesting an IP address from the Nebula Device.
	This displays Fixed and the Nebula Device will allow traffic to be forwarded to a specific LAN computer on the local network of the Nebula Device.
Cellular Status	
Cellular Status	This displays the status of the cellular Internet connection.
Data Roaming	This displays if data roaming is enabled on the Nebula Device.
	4G roaming is to use your NCC in an area which is not covered by your service provider.
	Enable roaming to ensure that your Nebula Device is kept connected to the Internet when you are traveling outside the geographical coverage area of the network to which you are registered.
Operator	This displays the name of the service provider.
PLMN	This displays the PLMN (Public Land Mobile Network) number.
NR-NSA Information	This displays the status of the cellular Internet connection.
МСС	This shows the Mobile Country Code (MCC). MCC is a unique code that identifies the country where a Public Land Mobile Network (PLMN) is at.
MNC	This shows the Mobile Network Code (MNC). MNC is a unique code that identifies a Public Land Mobile Network (PLMN) in a country. MCC and MNC combined together are used to identify a globally unique PLMN.
Physical Cell ID	This shows the Physical Cell ID (PCI), which are queries and replies between the Nebula Device and the mobile network it is connecting to. The normal range is 1 to 504.

Table 151	Site-wide > Devices > Mobile router > Live tools > Cellular Info	(continued)

LABEL	DESCRIPTION
RFCN	This displays the Radio Frequency Channel Number of DL carrier frequency used by the mobile network to which the Nebula Device is connecting.
	The value depends on the type of the mobile network (such as LTE, UMTS, GSM) to which the Nebula Device is connecting:
	• For UMTS (3G), it is the UARFCN (UTRA Absolute Radio-Frequency Channel Number) as specified in 3GPP-TS.25.101.
	• For LTE/5G, it is the EARFCN (E-UTRA Absolute Radio-Frequency Channel Number) as specified in 3GPP-TS.36.101. The value is '0' (zero) or 'N/A' if there is no network connection.
Band	This displays the current cellular band of your Nebula Device.
RSRP	This displays the Reference Signal Receive Power (RSRP), which is the average received power of all Resource Element (RE) that carry cell-specific Reference Signals (RS) within the specified bandwidth.
	The received RSRP level of the connected E-UTRA cell, in dBm, is as specified in 3GPP-TS.36.214.
	The reporting range is specified in 3GPP-TS.36.133.
	An undetectable signal is indicated by the lower limit, example –140 dBm.
	This parameter is for LTE only. The normal range is –30 to –140. The value is –140 if the Current Access Technology is not LTE. The value is 'N/A' if there is no network connection.
RSRQ	This displays the Reference Signal Receive Quality (RSRQ), which is the ratio of RSRP to the E- UTRA carrier RSSI and indicates the quality of the received reference signal.
	The received RSRQ level of the connected E-UTRA cell, in 0.1 dB, is as specified in 3GPP-TS.36.214.
	An undetectable signal is indicated by the lower limit, example –240.
	This parameter is for LTE only. The normal range is –30 to –240. The value is –240 if the Current Access Technology is not LTE. The value is 'N/A' if there is no network connection.
SINR	This displays the Signal to Interference plus Noise Ratio (SINR) of the SCC.
Service Information	If the cellular service provider supports carrier aggregation (CA), then this section displays statistics for the connection's primary component carrier (PCC).
Access Technology	This displays the type of the network (such as NR, LTE, Ethernet WAN) to which the Nebula Device is connecting.
Band	This displays the current cellular band of your Nebula Device.
RSSI	This displays the cellular signal strength between an associated cellular station and the Nebula Device for this SCC.
Cell ID	This shows the cell ID, which is a unique number used to identify the Base Transceiver Station to which the Nebula Device is connecting.
	The value depends on the Current Access Technology:
	• For GPRS, it is the Cell Identity as specified in 3GPP-TS.25.331.
	• For UMTS, it is the Cell Identity as defined in SIB3 3GPP-TS.25.331, 3GPP-TS.24.008.
	• For LTE/5G, it is the 28-bit binary number Cell Identity as specified in SIB1 in 3GPP-TS.36.331.
	The value is '0' (zero) or 'N/A' if there is no network connection.
Physical Cell ID	This displays the Physical Cell ID (PCI) of the SCC.

Table 151 Site-wide > Devices > Mobile router > Live tools > Cellular Info (continued)

LABEL	DESCRIPTION
UL Bandwidth (MHz)	This shows the uplink cellular channel bandwidth from the Nebula Device to the base station.
	According to 3GPP specifications, the bandwidths defined by the standard are 1.4, 3, 5, 10, 15, and 20 MHz. The wider the bandwidth the higher the throughput.
DL Bandwidth (MHz)	This shows the downlink cellular channel bandwidth from the base station to the Nebula Device.
	According to 3GPP specifications, the bandwidths defined by the standard are 1.4, 3, 5, 10, 15, and 20 MHz. The wider the bandwidth the higher the throughput.
RFCN	This displays the Radio Frequency Channel Number of DL carrier frequency used by the mobile network to which the Nebula Device is connecting.
	The value depends on the type of the mobile network (such as LTE, UMTS, GSM) to which the Nebula Device is connecting:
	• For UMTS (3G), it is the UARFCN (UTRA Absolute Radio-Frequency Channel Number) as specified in 3GPP-TS.25.101.
	• For LTE/5G, it is the EARFCN (E-UTRA Absolute Radio-Frequency Channel Number) as specified in 3GPP-TS.36.101.
	The value is '0' (zero) or 'N/A' if there is no network connection.
RSRP	This displays the Reference Signal Receive Power (RSRP), which is the average received power of all Resource Element (RE) that carry cell-specific Reference Signals (RS) within the specified bandwidth.
	The received RSRP level of the connected E-UTRA cell, in dBm, is as specified in 3GPP-TS.36.214.
	The reporting range is specified in 3GPP-TS.36.133.
	An undetectable signal is indicated by the lower limit, example –140 dBm.
	This parameter is for LTE only. The normal range is -30 to -140 . The value is -140 if the Current Access Technology is not LTE. The value is 'N/A' if there is no network connection.
RSRQ	This displays the Reference Signal Receive Quality (RSRQ), which is the ratio of RSRP to the E- UTRA carrier RSSI and indicates the quality of the received reference signal.
	The received RSRQ level of the connected E-UTRA cell, in 0.1 dB, is as specified in 3GPP-TS.36.214.
	An undetectable signal is indicated by the lower limit, example –240.
	This parameter is for LTE only. The normal range is –30 to –240. The value is –240 if the Current Access Technology is not LTE. The value is 'N/A' if there is no network connection.
RSCP	This displays the Received Signal Code Power, which measures the power of channel used by the Nebula Device.
	The received signal level, in dBm, is of the CPICH channel (Ref. 3GPP TS 25.133). An undetectable signal is indicated by the lower limit, example –120 dBm.
	This parameter is for UMTS only. The normal range is -30 to -120 . The value is -120 if the Current Access Technology is not UMTS. The value is 'N/A' if there is no network connection.
EcNo	This displays the ratio (in dB) of the received energy per chip and the interference level.
	The measured EcNo is in 0.1 dB and is received in the downlink pilot channel. An undetectable signal is indicated by the lower limit, example –240 dB.
	This parameter is for UMTS only. The normal range is –30 to –240. The value is –240 if the Current Access Technology is not UMTS or there is no network connection.

 Table 151
 Site-wide > Devices > Mobile router > Live tools > Cellular Info (continued)

LABEL	DESCRIPTION
TAC	This displays the Tracking Area Code (TAC), which is used to identify the country of a mobile subscriber.
	The physical cell ID of the connected E-UTRAN cell, is as specified in 3GPP-TS.36.101.
	This parameter is for LTE only. The value is '0' (zero) or 'N/A' if the Current Access Technology is not LTE or there is no network connection.
LAC	This displays the 2-octet Location Area Code (LAC), which is used to identify a location area within a PLMN.
	The LAC of the connected cell is as defined in SIB 1 [3GPP-TS.25.331]. The concatenation of PLMN ID (MCC+MNC) and LAC uniquely identifies the LAI (Location Area ID) [3GPP-TS.23.003].
	This parameter is for UMTS or GPRS. The value is '0' (zero) if the Current Access Technology is not UMTS or GPRS. The value is 'N/A' if there is no network connection.
RAC	This displays the RAC (Routing Area Code), which is used in mobile network "packet domain service" (PS) to identify a routing area within a location area.
	In a mobile network, the Nebula Device uses LAC (Location Area Code) to identify the geographical location for the old 3G voice only service, and uses RAC to identify the location of data service like HSDPA or LTE.
	The RAC of the connected UTRAN cell is as defined in SIB 1 [3GPP-TS.25.331]. The concatenation of PLMN ID (MCC+MNC), LAC, and RAC uniquely identifies the RAI (Routing Area ID) [3GPPTS. 23.003].
	This parameter is for UMTS or GPRS. The value is '0' (zero) if the Current Access Technology is not UMTS or GPRS. The value is 'N/A' if there is no network connection.
BSIC	The Base Station Identity Code (BSIC), which is a code used in GSM to uniquely identify a base station.
	This parameter is for GPRS only. The value is '0' (zero) if the Current Access Technology is not GPRS. The value is 'N/A' if there is no network connection.
SINR	This displays the Signal to Interference plus Noise Ratio (SINR) in dB. This is also a measure of signal quality and used by the UE (User Equipment) to calculate the Channel Quality Indicator (CQI) that it reports to the network. A negative value means more noise than signal.
CQI	This displays the Channel Quality Indicator (CQI). It is an indicator carrying the information on how good or bad the communication channel quality is.
MCS	MCS stands for modulation coding scheme. The base station selects MCS based on current radio conditions. The higher the MCS the more bits can be transmitted per time unit.
RI	This displays the Rank Indication, one of the control information that a UE will report to eNodeB (Evolved Node-B) on either PUCCH (Physical Uplink Control Channel) or PUSCH (Physical Uplink Shared Channel) based on uplink scheduling.
PMI	This displays the Precoding Matrix Indicator (PMI).
	PMI is for transmission modes 4 (closed loop spatial multiplexing), 5 (multi-user MIMO), and 6 (closed loop spatial multiplexing using a single layer).
	PMI determines how cellular data are encoded for the antennas to improve downlink rate.
SCC Information	If the cellular service provider supports carrier aggregation (CA), then this section displays statistics for the connection's secondary component carriers (SCCs).
GNSS Information	Global Navigation Satellite System (GNSS) sends position and timing data from high orbit artificial satellites. It works with GPS navigational satellites to provide better receiver accuracy and reliability than just using GPS alone. This is necessary for 5G networks that require very accurate timing for time and frequency synchronization. With GNSS, your can easily locate the Nebula Device with accurate information.

Table 151	Site-wide > Devices > Mobile router > Live tools >	Cellular Info	(continued)
			(Commoeu)

LABEL	DESCRIPTION
Enable	This shows if GNSS is enabled.
	Note: This can only be configured by a qualified service technician.
Scan OnBoot	This shows Enable if Scan OnBoot is enabled, so that GNSS runs automatically after the Nebula Device is turned on.
	Note: This can only be configured by a qualified service technician.
Scan Status	This shows GNSS error codes for debugging by a qualified service technician.
HDOP	Horizontal Dilution of Precision (HDOP) shows how accurate data collected by the Nebula Device is according to the current satellite configuration. A smaller value of HDOP means a higher precision.
Display Format	This shows the latitude and longitude display modes. There are three modes: 0, 1, and 2. Below are examples for these modes shown in latitude/longitude.
	0 – ddmm.mmmmN/S, dddmm.mmmmE/W
	1 – ddmm.mmmmmm, N/S, dddmm.mmmmmm, E/W 2 – (–)dd.ddddd, (–)ddd.ddddd
	N/S/E/W: North/South/East/West
	"-" : Negative values refer to South latitude/West longitude respectively. Positive values refer to North latitude/East longitude respectively.
Latitude	This shows the latitude coordinate of the Nebula Device. These positioning values (latitude, longitude, and altitude) help you locate the Nebula Device accurately.
Longtitude	This shows the longitude coordinate of the Nebula Device.
Elevation	This shows the altitude of the Nebula Device above sea level in meters.
Positioning Mode	This shows the GNSS positioning mode. 2D ("2") GNSS positioning mode displays latitude and longitude coordinates; 3D ("2") GNSS positioning mode displays latitude and longitude coordinates, and elevation.
Course Over Ground	This shows the course of the Nebula Device based on true North. Course Over Ground (COG) is different from the direction an object is headed, but the path derived from its actual motion (considered as Track), since the motion of an object is often with respect to other factors like wind and tides.
Speed Over Ground	This shows the Speed Over Ground (SOG) of the Nebula Device. SOG is the true object speed over the surface of the Earth.
Last Fix Time	This shows the last time in UTC format that the position of the Nebula Device was updated.
Number of Satellites	This shows the number of current active satellites. GNSS requires at least four satellites to determine the position of the Nebula Device.

Table 151	Site-wide >	· Devices >	 Mobile 	router >	Live	tools >	Cellular Info	(continued)	

10.4.3 LAN Stations

Go to the Site-wide > Devices > Mobile router > Live tools > LAN stations screen to view the LAN status of the Nebula Device. Click the pause icon () to stop scanning for LAN stations. Alternatively, click the play icon () to continue scanning.

Figure 183 Site-wide > Devices > Mobile router > Live tools > LAN stations

		LAN stations	WLAN stations >	Ping	Traceroute	DNS lookup	Reboot	Log	👎 Remote configurato
MAC address	IPv4 ad	dress							

The following table describes the labels in this screen.

Table 152 Site-wide > Devices > Mobile router > Live tools > LAN stations

LABEL	DESCRIPTION
MAC address	This field displays the MAC address of the LAN station.
IPv4 address	This indicate the IPv4 address of the LAN station.

10.4.4 WLAN Stations

Go to the Site-wide > Devices > Mobile router > Live tools > WLAN stations screen to view the WiFi status of the Nebula Device. Click the pause icon () to stop scanning for WiFi stations. Alternatively, click the play icon () to continue scanning.

Figure 184 Site-wide > Devices > Mobile router > Live tools > WLAN stations

VAIN Status	Cellular info	Traffic 🕨	LAN stations 🕨	WLAN stations	Ping	Traceroute	DNS lookup	Reboot	Log	👎 Remote configurato
MAC address	SSID nan	ne IPv4 addre	ss Capability	Security Ch	annel	Tx rate	Tx Rxr	ate Rx	_	Signal stren

The following table describes the labels in this screen.

LABEL	DESCRIPTION
MAC address	This field displays the MAC address of an associated WiFi station.
SSID name	This is the descriptive name used to identify the Nebula Device in a WiFi network.
IPv4 address	This indicate the IPv4 address of the gateway that helps forward this route's traffic.
Capability	This shows the WiFi standard supported by the client or the supported standards currently used by the client.
Security	This displays the type of security mode the WiFi interface is using in the WiFi network.
Channel	This is the channel number currently used by the WiFi interface.
Tx rate	This shows the maximum transmission rate of the client.
Tx	This shows the amount of data transmitted by the client since it last connected.
Rx rate	This shows the maximum reception rate of the client.
Rx	This shows the amount of data received by the client since it last connected.
Signal strength	This shows the RSSI (Received Signal Strength Indicator) of the client's WiFi connection.

Table 153 Site-wide > Devices > Mobile router > Live tools > WLAN stations

10.5 Backup & Restore

Use the **Site-wide** > **Devices** > **Mobile router** > **Backup & restore** screen to back up your configuration settings to the cloud or restore your current setting to the backup configuration.

Figure 185 Site-wide > Devices > Mobile router > Backup & restore

Site time	Admin	
2022-07-04 13:41:55	YaWen Lin	\$

The following table describes the labels in this screen.

LABEL	DESCRIPTION
Backup & restore	
Site time	This shows the date and time of the site, to which the change was applied, when the log was recorded.
Admin	This shows the name of the administrator who made the back up.
Backup	Click this button to create a new backup of the current configuration of the Nebula Device to the NCC.
	Click the Download icon (\bigcirc) to download the configuration file to your computer or laptop. Click the Delete icon ($\frac{1}{100}$) to remove the configuration file on the Nebula Device.
Restore	Click this button to overwrite the settings of the Nebula Device with the selected configuration backup.

Table 154 Site-wide > Devices > Mobile router > Backup & restore

10.6 Network Usage and Connectivity

Go to the Site-wide > Devices > Mobile router > Network usage and connectivity screen and then move the cursor to see the transmission rate (uplink/downlink) of a specific time.

oom: 2 hours 1 day	/ 💎 7 days < 30 days			Pan:	4 4 C > >>
	2022-07-12 13:42	~			1
	DL 23.23 Kbps				20.5 Kbp
	UL 264.07 bps				
					10.2 Kbp
					0 6
13:40	14:00	14:20	14:40	15:00	15:20

Figure 186 Site-wide > Devices > Mobile router > Network usage and connectivity

Table 155	Site wide > Devices > Mobile router > Network usage and connectivity
	Sile-wide > Devices > Mobile router > Nerwork usage and connectivity

LABEL	DESCRIPTION		
Network usage and connectivity			
Move the cursor over the chart to see the transmission rate at a specific time.			
Zoom	Select a time period to view the statistics in the past 2 hours, day, week, or month.		
Pan	Use this to move backward or forward by one day or a week.		

PART III Manage by Organization Deployment

CHAPTER 11 Organization-wide

11.1 Overview

This chapter discusses the menus that you can use to monitor your organization and manage sites, Nebula Devices, accounts, licenses, and VPN members for the organization.

11.2 License & inventory

The following section describes license management screens in NCC.

Unused licenses can be transferred from a Nebula Device in an Organization to another Nebula Device in an Organization.

11.2.1 License & Inventory Overview Screen

Use these screens to view licenses and Nebula Devices in the organization. Click **Organization-wide** > **License & inventory** > **Overview** to access this screen.



Figure 187 Organization-wide > License & inventory > Overview

The following table describes the labels in this screen.

Table 156 Organization-wide > License & inventory > Overview

LABEL	DESCRIPTION
Organization Statu	S
Actions	Click this button to add licenses and/or Nebula Devices to the organization. Choose one of the following actions:
	• Add more devices: Add new Nebula Devices to the organization, by serial number and MAC address. For details, see Section 11.2.2 on page 495.
	• Add more licenses: Add new licenses to the organization, by license key. For details, see Section 11.2.4 on page 496.
	• Install wizard: Add Nebula Devices and licenses to the organization, assign the licenses to the Nebula Devices, and then upgrade the organization if required. For details, see Section 11.2.5 on page 497.

LABEL	DESCRIPTION
Purchase License	Click this button to go to a window that will ask if you wish to be redirected to the Zyxel Circle web site (if the NCC account has a Circle account).
	If you do not have a Circle account, you can do the following:
	 Select what license to purchase and set the target expiration date to keep the Pro/Plus tier features/services running.
	2. You may export the list of required licenses to your computer.
	 After calculating the license to purchase, click the Zyxel license marketplace (Check out) button to complete your purchase. Purchased licenses are directly assigned to Nebula Device(s).
	Unused licenses assigned to your organization will not be counted as it is not yet assigned to a Nebula Device.
	This button is available only for the Full (Delegated) administrator privilege or Owner administrator account with a registered Nebula Device(s).
Upgrade Now	Click this button to upgrade the organization to Plus or Pro tier.
	The button is only available if you have a Plus or Pro license for every Nebula Device in the organization.
Downgrade Now	Click this button to downgrade the organization from Plus or Pro to Base tier, or from Pro to Plus tier.
	All active NCC licenses in the organization will stay active and continue to count down to their expiry time.
Organization type	This shows the licensing tier of the organization. Possible values are: Base , Plus Pack , Professional Pack , and Trial .
NCC license	This shows whether there are any Nebula Devices with near expiring licenses.
NSS/UTM license	This shows whether the current site has an active NSS or UTM license.
Secure WiFi license	This shows whether the current site has an active Secure WiFi license. A Secure WiFi license unlocks the Remote AP feature. Remote AP allows users connected to an off-site (remote) AP to connect to on-site resources behind the Nebula Device through a secure IPSec VPN tunnel.
Device status by expiration date	Click this button to select the data to be shown in the graph. Choose one from each of the following criteria:
	 All service name, Nebula Professional Pack, Nebula Plus Pack, Nebula Security Pack, UTM Security Pack, or Secure WiFi: select the category of licenses to display. All device type. Access Point Switch or Security Gateway: select the category of Nebula
	Device to display.
	Monthly, Quarterly, or Yearly: select the period of time to display.
Device defail statu	
License type	Gold Security Pack, Nebula Security Pack, UTM Security Pack, Content Filter Pack, Secure WiFi, Connect & Protect).
Device type	This shows the category of Nebula Device (Access points, Switches, Security appliance, Mobile router) and Nebula Device model.
# in org	This shows the total number of Nebula Devices of the specified category and model that are in the organization.
# unlicensed	This shows the total number of Nebula Devices of the specified category and model that have:
(expired)	No NCC Pro or Plus license.An expired NCC Pro or Plus license.
# near expiration in 90 days	This shows the total number of Nebula Devices of the specified category and model that have an NCC Pro or Plus license that will expire within 90 days.

Table 156	Organization-wide >	License & inventor	v > Overview	(continued)
	organization mao			001111100001

NCC User's Guide

LABEL	DESCRIPTION
# expiration over 90 days	This shows the total number of Nebula Devices of the specified category and model that have an NCC Pro or Plus license that have more than 90 days before expiration.
# inactive	This shows the total number of Nebula Devices of the specified category and model that have an NCC Pro or Plus license that has not been activated.

Table 156 Organization-wide > License & inventory > Overview (continued)

11.2.2 Add Devices Screen

Use this screen to add Nebula Devices to an organization. Click **Organization-wide** > **License & inventory** > **Overview** > **Actions** > **Add more devices** to access this screen.



Add devices							[] ×
Add devices	Devices						
Firmware upgrade	Enter one or more MAC o	address and serial num	ber.				
	Or you can download the	e <u>template</u> here and <u>im</u>	port multiple records f	or faster registration.			
	What Zyxel devices supp	ort Nebula?					
	Where can I find these n	umbers?					
	MAC address	Serial number	Name	Model	License info	Expiration date	Assign licenses from inventory
	MAC address	Serial number	Name	Model	License info	Expiration date	Assign licenses from inventory
	MAC address + Add another device	Serial number		X	License info	Expiration date	Assign licenses from inventor
	MAC address + Add another device	Serial number		X	License info	Expiration date	Assign licenses from inventor
	MAC address + Add another device	Serial number		X	License info	Expiration date	Assign licenses from inventor

The following table describes the labels in this screen.

LABEL	DESCRIPTION
template	Click this to download an XLSX file that you can use as a template to import a large number of Nebula Devices at once. Follow the instructions and formatting in the template to add the Nebula Device's serial numbers and MAC addresses.
import	Click this to upload a completed template XLSX file and import all Nebula Devices in the file.
MAC address	Enter the MAC address of the new Nebula Device.
Serial Number	Enter the serial number of the new Nebula Device.
Name	Enter a name for the new Nebula Device. It can consist of 1 – 64 characters.
Model	This shows the model number of the Nebula Device being added.
License info	This shows the type of NCC license activated on the Nebula Device, if there is one.
Expiration date	This shows the expiration date of the NCC license activated on the Nebula Device, if there is one.
Assign licenses from inventory	Click here to assign unassigned licenses already in the organization to the Nebula Device. Note: If the organization is a Pro or Plus tier, you must assign a Pro or Plus license to the Nebula Device within 15 days
^	Click the remove icon to delete the entry
W	
Add another device	Click this to add another Nebula Device to the organization.

Table 157 Organization-wide > License & inventory > Overview: Add devices: Add devices

LABEL	DESCRIPTION
Acknowledge	Select this to confirm that your NCC account will be the owner of the new Nebula Devices.
Finish	Click this to add the Nebula Devices to the organization.
Cancel	Click this to close the screen without saving.

Table 157 Organization-wide > License & inventory > Overview: Add devices: Add devices (continued)

11.2.3 Firmware Upgrade Screen

If a newer Nebula Device firmware is available, use this screen to upgrade it. Click **Organization-wide** > **License & inventory** > **Overview** > **Actions** > **Add more devices** > **Firmware upgrade** to access this screen.

Figure 189 Organization-wide > License & inventory > Overview: Add devices: Firmware upgrade

Add devices				×
Add devices Firmware upgrade	Firmware upgrade If a newer firmware is available, do you want to upgrade it during the device registration? The upgrade may take a few minutes. Please check the LED statue on your device(s). Yes No			
		Previous	Cancel	Finish

Note: If you choose not to upgrade the firmware, NCC will still perform an upgrade if the Nebula Device's firmware has security vulnerabilities, and/or lacks key performance improvements.

11.2.4 Add Licenses Screen

Use this screen to add licenses to an organization. Click **Organization-wide** > **License & inventory** > **Overview** > **Actions** > **Add more licenses** to access this screen.

Add licenses			×
Add licenses	Add licenses Enter one more license keys. G	Dr You can download the <u>template</u> here and <u>import</u> multip	le license keys for faster registration.
	License key	License information	
		×	ū
	+ Add		
			Cancel Finish

Figure 190 Organization-wide > License & inventory > Overview: Add licenses

Table 158 Organization-wide > License & inventory > Overview: Add	d licenses	
---	------------	--

LABEL	DESCRIPTION
template	Click this to download an XLSX file that you can use as a template to import a large number of licenses at once. Follow the instructions and formatting in the template to add the license keys.
import	Click this to upload a completed template XLSX file and import all licenses in the file.
License key	Enter the license key of the new license.
License information	This shows the license type and validity period of the license being added.
	Click the remove icon to delete the entry.
Add	Click this to add another license to the organization.
Finish	Click this to add the license to the organization.
Cancel	Click this to close the screen without saving.

11.2.5 Install Wizard

Use this wizard to add licenses and Nebula Devices to an organization, assign licenses to the new Nebula Devices, and then upgrade the organization if required. Follow the steps below to use the wizard.

1 Click Organization-wide > License & inventory > Overview > Actions > Install wizard. After the wizard window opens, click Next.

Wizard		[] ×
Before you start Add devices Firmware upgrade Add licenses Assign licenses Upgrade organization	Welcome install wizard The wizard will help you to add device/license and automatically assign the license to device. It will also provide upgrade options if the organization is available to upgrade Pro/Plus Pack.	
		Next Cancel

2 Add the MAC address and serial number of one or more Nebula Devices, select Acknowledge, and then click Next. For more information on this page, see Section 11.2.2 on page 495.

Wizard		[] ×
Before you start Add devices Firmware upgrade Add licenses Assign licenses	Devices Enter one or more MAC address and serial number. Or you can download the <u>template</u> here and <u>import</u> multiple records for faster registration. <u>What Zyxel devices support Nebula?</u> <u>Where can I find these numbers?</u>	
Upgrade organization	MAC address Serial number Name Model License info Expiration date X X X X	Ŵ
	Previous Next	Cancel

3 Click Yes (selected by default) to upgrade the Nebula Device firmware. If you select No, NCC will still perform an upgrade if the Nebula Device's firmware have security vulnerabilities, and/or lack key performance improvements. Click Next to continue.

Wizard		:: ×
Before you start Add devices Firmware upgrade Add licenses Assign licenses Upgrade organization	Firmware upgrade If a newer firmware is available, do you want to upgrade it during the device registration? The upgrade may take a few minutes. Please check the LED statue on your device(s). Yes No	
		Previous Next Cancel

4 Add the license keys of one or more licenses, and then click **Next**. For more information on this page, see Section 11.2.4 on page 496.

Wizard		[] ×
Before you start Add devices	Add licenses Enter one more license keys. Or You can download the <u>template</u> here and <u>import</u> multiple license keys for faster registration.	
Add licenses	License key License information	
Assign licenses	×	
Upgrade organization	-Add	
	Province Nant	Orneal
	Previous Next	Cancel

5 NCC automatically tries to assign an unused license to each matching Nebula Device. Reassign unused licenses for each Nebula Device manually by clicking **Select # of license**. Then click **Next**.

Wizard		[] ×
Before you start Add devices Firmware upgrade Add licenses Assign licenses Upgrade organization	Assign licenses There is no suitable license for selected device(s).	
	Previous Nex	Cancel

6 If the organization is on the base tier and you have added sufficient licenses for all Nebula Devices, you are given the option to upgrade to the Pro or Plus tier. Select **Yes** or **No**, and then click **Finish**.

Wizard				:: ×
Before you start Add devices Firmware upgrade Add licenses Assign licenses Upgrade organization	Upgrade organization No need to upgrade.			
		Previous	Cancel	Finish

11.2.6 License & Inventory Devices Screen

Use these screen to view and manage Nebula Devices in the organization. Click **Organization-wide** > **License & inventory** > **Devices** to access this screen.

Figure 191 Organization-wide > License & inventory > Devices

Overview Device	es Li	censes	Trial	c	change log	Purchase His	tory		
1 Access	Point		3 Switch	G) 1 Security Applia	nce		O Mobile Router	
actions → In use Unused Bo	th Q Search		• (1)s	elected in 5 de	vices.			+ Ac	id 🕒 Exp
Device	Device type	Site	Model	Serial number	MAC address	Claim date	Country	License expiration date	License info
and the second	A REPORT OF A R					Low-contraction to be	Contraction of the second second		Constant Provide Links & States
en-91-07-04-E1-04 (NAP102)	Access Point	2F Office	NAP102	9160705100070	R0-91-07-94-E1-94	2022-07-22	Taiwan	2024-07-22	Nebula Prof
60-91-07-04-51-04(NAP102)	Access Point Security Gateway	2F Office 2F Office	NAP102 NSG50	9167775100740 91771 97100056	60-91-07-97-51-97	2022-07-22 2022-10-18	Taiwan Taiwan	2024-07-22 2023-07-22	Nebula Profe
еп-эт-лэол-стол(NAP102) NSG50 ро-долгло-эл-л4	Access Point Security Gateway Switch	2F Office 2F Office 2F Office	NAP102 NSG50 XS3800-28	9182725100240 91201 97100058 91821 52080128	80-91-07.9.4-E1-9.4 50-E3-90-50-00-49 80-00-11-D8-9A-A.4	2022-07-22 2022-10-18 2022-10-17	Taiwan Taiwan Taiwan	2024-07-22 2023-07-22 2023-04-06	Nebula Profi Nebula Profi
	Access Point Security Gateway Switch Switch	2F Office 2F Office 2F Office 2F Office	NAP102 NSG50 XS3800-28 GS1350-6HP	S182725100240 S182725100240 S18215200058 S18215200058	60-31-07.97/E1-97 50-50-00-50-00-40 80-00-11-DR-3A-AA 90-05-45-47-70-51	2022-07-22 2022-10-18 2022-10-17 2022-07-01	Taiwan Taiwan Taiwan Taiwan	2024-07-22 2023-07-22 2023-04-06 2023-07-05	Nebula Profe Nebula Profe Nebula Profe

The following table describes the labels in this screen.

LABEL	DESCRIPTION
N Access Point	This shows the total number of access points (N) in the organization.
N Switch	This shows the total number of switches (N) in the organization.
N Security Appliance	This shows the total number of Security Gateway devices (N) in the organization.
N Mobile Router	This shows the total number of Mobile Router devices (N) in the organization.

Table 159 Organization-wide > License & inventory > Devices

LABEL	DESCRIPTION
Actions	Select one or more Nebula Devices and then click this button to perform one of the following actions:
	Change organization : Moves the Nebula Device to an organization. The organizations must have the same owners.
	Change site assignment: Moves the selected Nebula Devices to a site, or remove them from their current site while leaving them in the organization.
	Note: When you change the site for a Security Firewall (see Table 1 on page 13 for information on the supported Security Firewall devices), select the deployment method for management by Nebula (see Step 7: Set up the Deployment Method on page 51 for more information), configure the WAN settings and choose the installation method.
	Remove from organization : Remove the Nebula Devices from NCC. You can manage the Nebula Devices in standalone mode, or re-add them to NCC later.
	Assign license: Assign licenses to the selected Nebula Devices.
	Undo assign : Unlink the inactive licenses from the associated Nebula Devices. After unlinking, the license will be categorized as unused in Inventory . An inactive license is a license that has been assigned to a Nebula Device but is not yet in use or queued.
	Transfer license : Moves the unused licenses linked to a Nebula Device to another Nebula Device. Nebula Devices can be in the same organization or in a different organization. The Nebula Devices must have the same owner. Bundled, Trial, and Promotion licenses cannot be transferred.
	Purchase license : Select what license to purchase and target expiration date to keep the Pro/ Plus tier features/services running. You may export the list of required licenses to your computer. Then click the Zyxel license marketplace (Check out) button to complete your purchase.
	Unused licenses assigned to your organization will not count as it is not yet assigned to a Nebula Device.
	This button is available only for the Organization (Delegated) or Owner administrator account with a registered Nebula Device(s).
In use / Unused / Both	Select to display the Nebula Device currently in a site (In use), not current (Unused), or show all (Both).
Search	Enter a keyword or specify one or more filter criteria to filter the list of Nebula Devices.
+ Add	Add one or more new Nebula Devices to the organization, by entering the Nebula Device's MAC address and serial number. For details, see Section 11.2.2 on page 495.
Export	Click this button to save the Nebula Device list as a CSV or XML file to your computer.
	Select an entry's check box to select a specific Nebula Device. Otherwise, select the check box in the table heading row to select all Nebula Devices.
Device	This shows the hostname of the Nebula Device.
Device type	This shows the category of Nebula Device (Access points, Switches, Security appliance, Firewall, Mobile router) and Nebula Device model.
Site	This shows the site that the Nebula Device is currently in. If the Nebula Device is not in any site, the value is blank.
Model	This shows the Nebula Device's model.
Serial Number	This shows the Nebula Device's serial number.
MAC address	This shows the MAC address of the Nebula Device's first Ethernet port.
Device tag	This shows the tag created and added to the Nebula Device.

Table 159 Organization-wide > License & inventory > Devices (continued)

LABEL	DESCRIPTION
Claim date	This shows the date on which the Nebula Device was added to NCC. If the Security Firewall has NOT yet connected to NCC (see Table 1 on page 13 for the list of Security Firewalls):
	• Native mode. Click this button and select Nebula Native mode in the Deployment Method. Follow the instructions to connect the Security Firewall to NCC.
	 Waiting ZTP will be shown if Native mode is not available. Click the Waiting ZTP button and select Zero Touch Provisioning in Deployment Method to configure the ZTP settings.
Unused / In use	This shows Unused if the Nebula Device is not assigned to a site, or In use if the Nebula Device is currently in a site.
Country	This shows the country in which the Nebula Device is located.
License expiration date	This shows the date on which the Nebula Device's NCC license will expire.
License info	This shows the type of NCC license assigned to the Nebula Device.
	Note: Move the pointer over this field to see information about all licenses associated with this Nebula Device.
Action	Select one or more Nebula Devices and then click this button to perform one of the following actions:
	Change organization : Moves the Nebula Device to an organization. The organizations must have the same owners.
	Change site assignment : Moves the selected Nebula Devices to a selected site, or removes them from their current site while leaving them in the organization.
	Note: When you change the site for a Security Firewall (see Table 1 on page 13 for information on the supported Security Firewall devices), select the deployment method for management by Nebula (see Step 7: Set up the Deployment Method on page 51 for more information), configure the WAN settings and choose the installation method.
	Remove from organization : Remove the Nebula Devices from NCC. You can manage the Nebula Devices in standalone mode, or re-add them to NCC later.
	Assign license: Assign unassigned licenses to the selected Nebula Devices.
	Undo assign : Unlink the inactive licenses from the associated Nebula Devices. After unlinking, the license will be categorized as unused in Inventory . An inactive license is a license that has been assigned to a Nebula Device but is not yet in use or queued.
	Transfer license : Moves unused licenses linked from one Nebula Device to another Nebula Device. The Nebula Devices can be in the same organization or in a different organization. The Nebula Devices must have the same owner. Bundled, Trial, and Promotion licenses cannot be transferred.

Tabla 150	Organization w	ida > Liconco	8 invontory >	Dovicos	(continued)
	Organization-w				

11.2.7 License & Inventory Licenses Screen

Use these screen to view and manage licenses in the organization. Click **Organization-wide** > **License & inventory** > **Licenses** to access this screen.

Overview	Devices	Licenses	Trial	Change log	Purchase History			
		8 assigned		5 unu (Pro	sed Pack, 2YR)			
ctions → Q Sear	ch	1 selected , (19) matches i	n (19) licenses.			Show expired I	icenses + A	dd 🕒 Export
License Key		Service	License states	License expiration date	e Remaining days	Claim date	Activate date	Associated devi
1 IC NEED 779V0	05303106040001 16	Nebula Professional Pack License, 2YF	Active	2024-10-18	730 days	2022-10-18	2022-10-18	D0.E0.EE.00.00.F
1 IC NEED 779V0	0500006040001 17	Nebula Professional Pack License, 2YF	Active	2024-07-22	642 days	2022-07-22	2022-07-22	60-91-07-94-E1-94(
I IO NIDDO 771VO	0500010505101000	Nebula Professional Pack License, 1YR	Active	2023-07-22	276 days	2022-07-21	2022-07-21	NSG50
NI NIDDOTV OOSEE	000E/E/77DE1 00	Nebula Professional Pack License, 1YR	Active	2023-07-05	259 days	2021-03-31	2021-03-31	DO-OE-XE-X7-7D-E
N NEEO1V 00965	000EAEA77DE1 01	Nebula Professional Pack License, 1YR	Expired	2022-07-04	-	2021-03-31	2021-03-31	DO.OF. 4F. 47-7D.F
	0 010000144061	Nebula Professional Pack License, 1MC	D Expired	2022-01-20	-	2021-12-20	2021-12-20	NSG50
	DOOE/12/77DE1 02	Nebula Professional Pack License, 1MC	D Expired	2021-07-03	-	2021-03-31	2021-03-31	DO.OF.4E.47.7D.P
	DOOE4E477DE1 04	Nebula Professional Pack License, 1MC	D Expired	2021-06-26	-	2021-03-31	2021-03-31	DO.OF. 45-47-70-5
	DOOE/10/271001 00	Nebula Professional Pack License, 1MC	D Expired	2021-05-19	-	2021-03-19	2021-03-19	DO-OE-4E-47-7D-E
		Nebula Professional Pack License, 1M	D Expired	2021-04-19	-	2021-03-19	2021-03-19	DO-OE-4E-47-7D-E

Figure 192 Organization-wide > License & inventory > Licenses

The following table describes the labels in this screen.

LABEL	DESCRIPTION			
N assigned	This shows the total number of licenses (N) in the organization that are assigned to a Nebula Device and activated.			
N unused (Pro Pack, 1MO/1YR/ 2YR/4YR/7YR)	This shows the total number of Nebula Professional Pack or Nebula Plus Pack licenses (N) in the organization that are not assigned to a Nebula Device.			
or				
N unused (Plus Pack, 1MO/1YR/ 2YR)				
N unused (UTM Pack, 1MO/1YR/ 2YR)	This shows the total number of UTM Security Pack licenses (N) in the organization that are not assigned to a Nebula Device.			
Actions	Select one or more Nebula Devices and then click this button to perform one of the following actions:			
	Change organization : Moves the selected licenses to an organization. The organizations must have the same owners.			
	Assign License: Assign the selected licenses to one or more Nebula Devices. Only the licenses applicable for the Nebula Device can be selected.			
	Undo assign : Unlink the inactive licenses from the associated Nebula Devices. After unlinking, the license will be categorized as unused in Inventory . An inactive license is a license that has been assigned to a Nebula Device but is not yet in use or queued.			
	Transfer license : Moves the unused licenses linked to a Nebula Device to another Nebula Device. The Nebula Devices can be in the same organization or in a different organization. The Nebula Devices must have the same owner. Bundled, Trial, and Promotion licenses cannot be transferred.			

Table 160 Organization-wide > License & inventory > Licenses

LABEL	DESCRIPTION			
Search	Enter a keyword or specify one or more filter criteria to filter the list of licenses.			
N licenses	This shows the total assigned and unassigned licenses in the organization.			
Show expired licenses	Click this to display licenses that are past their validity.			
+ Add	Add one or more new licenses to the organization, by entering their license keys. For details, see Section 11.2.4 on page 496.			
Export	Click this to save the license list as a CSV or XML file to your computer.			
License Key	This shows the key of license, including bundled licenses.			
Service	This shows the service that license is for, for example "Nebula Professional Pack".			
License states	 This shows the current status of the license: Active: The license is assigned to a specific Nebula Device and activated. Inactive: The license is assigned to a specific Nebula Device but not activated. Expired: The license is past its validity. Queued: The license is assigned to a specific Nebula Device, and the license is waiting for the currently active license to expire. Unused: The license is not assigned to a specific Nebula Device. 			
License expiration date	This shows the date on which the license will expire. Queued means there are multiple licenses assigned to the Nebula Device, and the license is waiting for the currently active license to expire.			
Remaining days	This shows how days remain until the license expires.			
Claim date	This shows the date on which the license was added to NCC. If the Security Firewall has NOT yet connected to NCC:			
	 Native mode. Click this button and select Nebula Native mode in Deployment Method. Follow the instructions to connect the Security Firewall to NCC. Waiting ZTP will be shown if Native mode is not available. Click the Waiting ZTP button and select Zero Touch Provisioning in Deployment Method to configure the ZTP settings. 			
Activate date	This shows the date on which the license was activated.			
Associated device	This shows the name and model of the Nebula Device that the license is assigned to.			
Associated site	This shows the name of the site that the license is being used in. Click the site to go to its dashboard.			
Action	Click this button to perform the following actions:			
	Change organization: Moves the selected licenses to an organization. The organizations must have the same owners.			
	Assign License: Assign the selected licenses to one or more Nebula Devices. Only the licenses applicable for the Nebula Device can be selected.			
	Undo assign : Unlink the inactive licenses from the associated Nebula Devices. After unlinking, the license will be categorized as unused in Inventory . An inactive license is a license that has been assigned to a Nebula Device but is not yet in use or queued.			
	Transfer license: Moves the unused licenses linked to a Nebula Device to another Nebula Device. The Nebula Devices can be in the same organization or in a different organization. The Nebula Devices must have the same owner. Bundled, Trial, and Promotion licenses cannot be transferred.			

Table 160	Organization-wide >	License & inventory >	> Licenses	(continued)
-----------	---------------------	-----------------------	------------	-------------
11.2.8 License & Inventory Trial Screen

A free 30-day trial license is available for each Nebula organization you create. Trial licenses are available even if you have no Nebula Devices in the organization.

Note: Make sure services are usable by the Nebula Device before activating the trial license.

All trial licenses apply to all Nebula Devices in an organization. There is no limit to the number of organizations. You will lose access to related services or advanced NCC features when trial expires. You must then buy a standard license (not a trial) for each Nebula Device.

Activating a standard license during the trial period will add the remaining trial time to the standard license time. However, activating a Nebula Professional Pack standard license during the trial period will cancel the trial. NCC activates inactive licenses when the associated trial has expired.

If you activate the Nebula Pro Pack Trial, you can use advanced features in Nebula Devices in all organizations.

Moving a Nebula Device to another organization will cancel its trial license. However, a trial license is still available for the Nebula Device if you did not activate a trial or standard license of the same type in the new organization.

Note: Each trial license is not available if you previously activated a trial or standard license of the same type.

At the time of writing, trial licenses are associated with the following:

TRIAL LICENSE	ASSOCIATED FEATURES OR NEBULA DEVICES			
Nebula Pro Pack Trial	This is for advanced features, except open API access, within the Nebula Device's organization. See Section 4.9.7 on page 229 for more information on open API access.			
MSP Pack Trial	This is for new NCC accounts or NCC accounts that have not used MSP before. This allows you to manage multiple organizations.			
Gold Security Pack Trial	This is for ATP devices and USG FLEX devices except USG20-VPN / USG20W-VPN / USG FLEX 50. Note: The Gold Security Pack Trial also includes use of advanced features except open API access from the Nebula Pro Pack Trial.			
Secure WiFi Trial	This is for remote APs (access points) to securely connect a ZyWALL ATP / USG FLEX (except USG FLEX 50) in the office.			
Content Filter Pack Trial	This is for USG FLEX 50 /USG20-VPN / USG20W-VPN devices.			
Connect & Protect (CNP) Trial	This allows you to manage small business WiFi hotspots using an NWA1123-ACv3, WAC500, WAC500H, NWA110AX, NWA210AX, WAX510D, WAX610D, WAX630S, or WAX650S.			

Table 161 Trial Licenses Summary

See Table 2 on page 16 for detailed information on the licenses available in NCC.

Use this screen to view the status and activate trial licenses for Nebula Devices within the organization. Click **Organization-wide** > **License & inventory** > **Trial** to access this screen.

Overview	Devices	Licenses	Trial	Change log	Purchase History
We offer 30 day Note: You can a	ys free trial for each kind activate once per service	of services per demand. type and cancel it anytim	e during the trial period	L	
					Acti
lebula Pro Pack	Trial TRIAL EXPIRED			Content Filter	Pack Trial
full feature set/ser	vice with all advanced fu	nctionality and managem	ent features of	Apply to USG20(W	V)-VPN/USG FLEX 50(W)
lebula Control Cent	ter for device, site and or	ganization.		Content Filter Pac	ck include Web Filtering/Content Filtering, SecuReporter Premium,
rial expired on 2021	1-08-11			Security Profile Sy	/nc.
				Trial not activated	d yet!
ISP Pack Trial	TRIAL EXPIRED				Activ
lebula MSP Pack Lie	cense is a Per-admin use	r account license that incl	lude cross-org.		
lanagement featur	es of Nebula Control Cer	iter and can be used in co	njunction with	Connect & Pro	otect Trial
xisting Pack (Base,	Plus, Pro). Apply <u>here</u> .			Apply to NWA1123	ACv3, WAC500, WAC500H, NWA110AX, NWA210AX, WAX510D, WAX610D,
				WAX630S & WAX6	350S.
				Connect & Protec	t service is a cloud mode license focus on SB hotspot wifi scenarios.
old Security Pa	ick Trial			experience for wit	fi clients
pply to ATP/USG FL	EX series.			Trial not activated	d veti
old Security Pack i	ncl <mark>u</mark> des Sandboxing, We	b Filtering, Application Sec	curity, Malware	indifier doctrato,	a you
locker, Intrusion Pre	evention, Geo Enforcer, S	ecure WiFi Service, Collab	orative Detection &		ACTIV
esponse, SecuRepo	orter Premium, Security P	rofile Sync.			
rial not activated y	et!			Secure WiEi Tr	IN PROGRESS
			Activate		
				Secure tunnel & m	nanaged AP service
				Expiration date 2	022-11-19
					Deactiv

Figure 193 Organization-wide > License & inventory > Trial

LABEL	DESCRIPTION
Actions	Click this to perform one of the following actions:
	 Activate trial for all: select this to start using all trial licenses available for your organization. Then click Confirm to continue. Deactivate trial for all: select this to cancel all trial licenses currently in use in your organization. Then click Confirm to continue.
	Note: When you cancel any trial license, you cannot re-activate the unused portion of the trial license.
(Status)	The status displays next to the name of a trial license. If no status displays, it means you can activate the trial license. The trial license can be used on the Nebula Devices within the organization. Click Activate to start using the services of the trial license.
	Note: You can activate each type of 30-day trial license on each organization only once.
IN PROGRESS	The 30-day countdown for the trial license has begun. Click Deactivate if you want to cancel the trial license.
	Note: You can cancel the trial license anytime during the 30-day trial period, but you cannot re-activate it.
TRIAL EXPIRED	You have previously activated a trial or standard license and the license period has ended.
CANCELED	You have deactivated the trial license during the 30-day trial period.

LABEL	DESCRIPTION
Activate	Click this to start using the 30-day trial license. Then click Confirm to continue.
Deactivate	Click this to cancel the 30-day trial license anytime before it expires. Then click Confirm to continue.

Table 162 Organization-wide > License & inventory > Trial (continued)

11.2.9 License & Inventory Change Log Screen

Use this screen to view a record of Nebula Device and license actions within the organization. The log also shows the change in state of the organization, as a before and after, as a result of each action. Click **Organization-wide** > **License & inventory** > **Change log** to access this screen.



cense & inventory												
Overview	Devices	Licenses		Trial	Change	log	Purchase History					
Keyword:				From:			To:					
Q Search	-		Range 🔻	2022-10-06	(11)	02:32 🔻	2022-10-13		02:32 🔻	UTC+0	\propto	् Search
												D - 1
Newer Older > 4 Date and time	change logs withi	n the time filtered.	. Changes da	te back to 2022-03-	-21 03:20 (UTC) Be	efore	After				Admin	Export-
Newer Older > 4 Date and time 2022-10-12 07:15:25	change logs withi Action Claimed device: ⁻	n the time filtered.	. Changes da	te back to 2022-03-	-21 03:20 (UTC) Ba	efore	After Added 1	o Inventor	γ		Admin W Hua	Export •
Newer Older 4 Date and time 2022-10-12 07:15:25 2022-10-12 07:15:25	change logs withi Action Claimed device: ^{rn} Moved device: ^{rnn}	n the time filtered.		te back to 2022-03-	-21 03:20 (UTC) Ba	efore	After Added 1 added 1	:o Inventor :o Site: [AE	V :_Test]		Admin W Hua W Hua	Export →
Newer Older 4 Date and time 2022-10-12 07.15.25 2022-10-12 07.15.25 2022-10-12 02.15.54 2022-10-12 02.15.54 2022-10-12 02.15.54	change logs within Action Claimed device: ^^^ Moved device: ^^^ Removed device:		. Changes da / caral acada / caral acada / caral acada	te back to 2022-03-	-21 03:20 (UTC) Be In In	efore wentory	After Added t added t	:o Inventor :o Site: [AE	Y :_Test]		Admin W Hua W Hua W Hua	Export -

The following table describes the labels in this screen.

Table 163 Organization-wide > License & inventory > Change log

LABEL	DESCRIPTION
Keyword	Enter a keyword or specify one or more filter criteria to filter the list of log entries.
Range / Before	Select a filtering option, set a date, and then click Search to filter log entries by date.
	Range: Display log entries from the first specified date to the second specified date.
	Before: Display log entries from the beginning of the log to the selected date.
Search	Click this to update the list of logs based on the search criteria.
Reset filters 🖂	Click this to return the search criteria to the previously saved time setting.
Newer / Older	Click to view the list of log messages with the most recent or oldest message displayed first.
	This shows the total number of the log messages that match the search criteria. It also shows the date and time the very first log was created.
Export	Click this button to save the log list as a CSV or XML file to your computer.
Date and time	This shows the date and time in UTC+00:00 (or UTC+0) when the log was recorded.
	UTC is a standard time for use around the world (formerly known as Greenwich Mean Time or GMT). UTC is an international abbreviation that is neither French nor English. It means both "Temps Universel Coordonné" and "Coordinated Universal Time".
Action	This shows the action that triggered the log entry.
Before	This shows the old setting or state that was overwritten with the new value.
After	This shows the new setting or state.

Admin	This shows the name of the NCC administrator account that made the changes
	Click this icon to display a greater or lesser number of configuration fields.

Table 163 Organization-wide > License & inventory > Change log (continued)

11.2.10 License & Inventory Purchase History Screen

Use this screen to view a record of Nebula Device license purchased within the organization. Click **Organization-wide > License & inventory > Purchase history** to access this screen.



		Welcome to Nebula Professional Pack! Make the most of your network without limitations.					
Organization-wide > Configure	> License & inventory						
Overview	Devices	Licenses	Trial	Change log	Purchase History		
Q Search	• O purc	chases.					
Order ID	Purchase date		# licenses	Purchase by	Status	Export	

The following table describes the labels in this screen.

LABEL	DESCRIPTION				
Keyword	Enter a keyword or specify one or more filter criteria to filter the list of purchased license entries.				
Search	Click this to update the list of logs based on the search criteria.				
N purchases	This displays the total purchased licenses in the organization.				
Order ID	This displays a unique code that identifies the order. Clicking this link will take you to the Marketplace > Order History screen.				
Purchase date	This displays the date that the order was created.				
# licenses	This displays the number of licenses purchased for the specified license type.				
Purchase by	This displays the email address of the NCC account that created the order.				
Status	 This displays the current status of the order. Done: The order has been paid for and the license was successfully activated on the target Nebula Device. Processing: The license activation on the target Nebula Device is still under process. Failed: The license was not successfully activated on the target Nebula Device. 				
Export	Click this to download the order details as a CSV or XML file to your computer. This includes the Order ID and each license's assigned device information.				

Table 164 Organization-wide > License & inventory > Purchase history

11.3 Administrators

Use this screen to view, manage and create administrator accounts for the specified organization. Click **Organization-wide > Administrators** to access this screen.

vation = 🔄 🕞 Ford	ce logout 🛛 🗇 Delete 🛛 Q. Sea	rch administrators 👻	(10) administrators		💎 🕒 Import	$+ \operatorname{Add}$
Name	Email address	Merged privilege 🔒	Privilege	Account status	Last access time (UTC)	Cr 📃
NE-Worg	Assessing an application for	Owner	Owner	ОК	2021-04-19 23:52:25	2021-0
Thursdamore	Burnan narrang@isediantite	Organization (Full) more >	Organization (Delegated)	ОК	2021-04-20 01:53:35	2021-
0.00	ann changispai con tar	Organization (Full)	Organization (Full)	ОК	2021-04-19 02:52:10	2021-
(Frank Halles	(Franci Passes) (Speak Section (Sp.	Organization (Full)	Organization (Full)	Deactivated	2021-04-21 00:48:58	2021-
and the state of t	eller Filer Bryani son de	Organization (Full)	Organization (Full)	Deactivated	2021-04-20 04:02:04	2021-
States Augusta	Water Brand care	Organization (Read)	Organization (Read)	Deactivated	2021-04-06 01:50:17	2021-
.collect.	patholographic and the	Organization (Full)	Organization (Full)	Deactivated	2021-04-20 02:45:07	2021-
1077-035	warrey/infinity and cloth the	Organization (Full)	Organization (Full)	ОК	2021-04-20 08:33:50	2021-
80	whether all propriet sources	Organization (Full)	Organization (Full)	ОК	2021-04-21 01:56:51	2021-
	participation and the	Organization (Full) by MSP		OK	2019-01-14 09:25:10	2021-

Figure 196 Organization-wide > Administrators

LABEL	DESCRIPTION					
Activation	Click this button to Activate/Deactivate the selected accounts. Then click Update.					
Force logout	Click this button to force the selected accounts to log out of the NCC.					
Delete	Click this button to remove the selected accounts.					
Search	Specify your desired filter criteria to filter the list of administrate	or accounts.				
administrators	This shows the number of administrator accounts in the list.					
Change owner	Click this button to transfer ownership of the organization owner owner account must be an organization full administrator.	Inother user account. The new				
	Change organization owner	×				
	Please select current organization admin to become new owner.					
	Tam - thomas manning Signal com/lw	•				
	This action will cause you lose ownership rights include Nebula devic under this organization. Do you want to continue?	es				
	No	es				
	After transferring ownership, NCC performs the following actions:					
	 Changes your account from organization owner to organ Transfers all Nebula Devices and licenses in the organizati Sends the new owner an email, notifying them of the cha 	ization full administrator. on to the new owner. nge.				

Table 165 Organization-wide > Administrators

LABEL	DESCRIPTION					
Import	Click this button to create administrator accounts in bulk by importing a complete list of all new administrators in an Excel file.					
	Bulk Import X					
	"Bulk Import" supports for faster inputting. Please follow this template to import					
	Browse					
	Or drag file here					
	Close					
Add	Click this button to create a new administrator account. See Section 11.3.0.1 on page 510.					
Name	This shows the name of the administrator account.					
Email address	This shows the email address of the administrator account.					
Merged privilege	This shows the final privilege the account has in the organization, when organization privileges configured on different screens are combined and prioritized. Organization privileges can be configured on the following screens; the highest privilege level takes priority:					
	 MSP cross-org manage > Admins & teams > Admins MSP cross-org manage > Admins & teams > Teams Group-wide manage > Administrators Organization-wide > Administrators 					
	For more information, see Section 13.5.1 on page 581.					
Privilege	This shows whether the administrator account has read-only, monitor-only, guest ambassador, or read and write (full) access to the organization and sites.					
	Installer indicates that the administrator account can register Nebula Devices at a site.					
	Owner indicates that the administrator account is the creator of the organization, who has full access to that organization and cannot be deleted by other administrators.					
	Organization (Delegated) means that the administrator account has delegated owner privileges. This type of account can perform all of the same actions as the organization owner, except for the following:					
	 Delete organization Transfer organization ownership Assign delegate owner privileges to an administrator account. 					
Account status	This shows whether the administrator account has been validated (OK). It shows Deactivated if an administrator account has been created but cannot be used. This may happen since you can only have up to five active administrator account on Nebula (free).					
Last access time	This shows the last date and time traffic was sent from the administrator account.					
Create date	This shows the date and time the administrator account was created.					
Status change date	This shows the last date and time the administrator account status was changed.					
Ð	Click this icon to display a greater or lesser number of configuration fields.					

Table 165 Organization-wide > Administrators (continued)

11.3.0.1 Create/Update Administrator

In the **Organization-wide** > **Administrator** screen, click the **Add** button to create a new administrator account or double-click an existing account entry to modify the account settings.

Create administrator					
Name:		*			
Email:		×			
Organization access:	Read-Only	•			
	Delegate owner's authority 🟮				
Activate:	No	•			
Site	Privilege				
	▼ Monitor-only ▼				
+ Add					
	Close Create of				

Figure 197 Organization-wide > Administrator: Create/Update administrator

Table 166	Organization-wide >	Administrator: Create/U	pdate administrator
	organization mao	, tarrin isiraren. ereare, e	

LABEL	DESCRIPTION
Name	Enter a descriptive name for the administrator account.
Email	Enter the email address of the administrator account, which is used to log into NCC.
	This field is read-only if you are editing an existing account.
Organization	Set the administrator account's access to the organization.
access	When an administrator account has read and write (Full) access, the administrator can create or delete other administrator accounts, create or delete a site, and add or renew licenses for Nebula Devices in the organization.
	Note: The administrator account you use to create an organization is the organization creator account that has full access to that organization. The organization creator account cannot be deleted by other organization administrators.
	If you select Read-only , the administrator account can be the organization administrator (that has no write access to the organization) and also be a site administrator.
	If you select None , the administrator account can only be a site administrator.
Delegate	This setting is only available when Organization access is set to Full.
owner's authority	Select this setting to grant delegate owner privileges to an organization full administrator account. An account with delegate owner privileges can perform all of the same actions as the organization owner, except for the following:
	Delete organization
	 Transfer organization ownership Assign delegate owner privileges to an administrator account.
Activate	Select Yes to enable the account or No to temporarily disable the account.

LABEL	DESCRIPTION
YES, I want to do it.	The check box displays only when an administrator that has full access to the organization selects No in the Activate field to disable his/her own account.
	Note: After you select the check box and click Update admin , you lose administrator privileges and cannot manage the organization again. If you have other organizations created on your account, you can click and select another organization to manage in the MSP Portal screen.
Site	This field is available only when you set the account's organization access to Read-only or None .
	Select the site to which you want to set the account's access.
Privilege	This field is available only when you set the account's organization access to Read-only or None .
	Set the administrator account's access to the site.
	You can select from Read-only , Monitor-only , Guest Ambassador , Installer and Full (read and write).
	An administrator account that has Guest Ambassador access can create, remove or manage guest accounts using the Cloud authentication screen (see <u>Section on page 555</u>).
	Installer access allows an administrator to register Nebula Devices at this site.
Add	Click this button to create a new entry in order to configure the account's access to another site.
Close	Click this button to exit this screen without saving.
Create admin/ Update admin	Click this button to save your changes and close the screen.

Table 166	Organization-wide >	Administrator	Create/Update	administrator	(continued)
	organization what s	/ arringingingingingingingingingingingingingi	ciculo, opualo	aarmininararor	

11.4 Organization-wide Manage

Use the **Organization-wide manage** menus to create new sites, register or unregister a Nebula Device, change organization general settings, and manage licenses, user accounts, administrator accounts or VPN members in the organization.

11.4.1 Organization Portal

This screen shows you the site locations on a Google map and the summary of sites, site tags and connected Nebula Devices for the selected organization.

Click Organization-wide > Organization-wide manage > Organization portal to access this screen.



Figure 198 Organization-wide > Organization-wide manage > Organization portal

11.4.1.1 Sites

Click the **Sites** tab in the **Overview** screen to view detailed information of the sites which are associated with the selected organization.

Sites	Site tags	Devices								
Tog*		Q, Search	(41) Si	tes				Over the	e last day: 163 Clients, 356.56 GB 💎 🗈	Export -
	Status 🔋	Name	Usage	Client	Tag	Site health 🔋 📋	Device	Offline device	% Offline	R
	•	Site11	37.57 MB	0		e e	1	0	0.0 %	^
	•	Site09	0 bytes	0			1	1	100.0 %	
		Site17	0 bytes	0			0	0	N/A	
	•	Site04	0 bytes	0			1	1	100.0 %	
	•	Site25	12.09 GB	9	more test		4	0	0.0 %	
	•	Site05	204.27 MB	1			1	0	0.0 %	
	•	Site16	21.56 MB	0			1	1	100.0 %	
	•	Site01	0 bytes	0			1	1	100.0 %	
	•	Site14	0 bytes	0			1	1	100.0 %	
	•	Site30	11.36 GB	30			6	1	16.7 %	~
							ŀ	< < Page 1	of 5 > > Results per page:	10 👻

Figure 199 Organization-wide > Organization-wide manage > Organization portal: Sites

Table 167 Organization-wide > Organization-wide manage > Organization portal: Sites

LABEL	DESCRIPTION
Tag	Select one or multiple sites and click this button to create a new tag for the sites or delete an existing tag.
Delete	Select the sites and click this button to remove it.

LABEL	DESCRIPTION						
Search	Enter a key word as the filter criteria to filter the list of sites.						
Sites	This shows the number of sites in this organization.						
Over the last day	This shows how many clients are associated with the sites in this organization and the total amount of data transmitted or received by the clients in the past day.						
Export	Click this button to save the site list as a CSV or XML file to your computer.						
Status	This shows the status of Nebula Devices in the site.						
	 Green: All Nebula Devices are online and have no alerts. Amber: Some Nebula Devices have alerts. Red: Some Nebula Devices are offline. Gray: All Nebula Devices have been offline for 7 days or more. White: No Nebula Devices. 						
Name	This shows the descriptive name of the site.						
Usage	This shows the amount of data consumed by the site.						
Client	This shows the number of clients connected to Nebula Devices in the site.						
Tag	This shows the user-specified tag that is added to the site.						
Site Health	 This shows the percentage of uptime in a given time interval to indicate the site's network availability. Green: 95 – 100% network uptime Dark green: 75 – 95% network uptime Brown: 50 – 75% network uptime Red: < 50% network uptime Grey: No uptime data 						
Device	This shows the total number of Nebula Devices deployed in the site.						
Offline device	This shows the number of Nebula Devices which are added to the site but not accessible by the NCC now.						
% Offline	This shows what percentage of the connected clients are currently offline.						
	Click this icon to display a greater or lesser number of configuration fields.						

Tabla 177	Organization wide >	Organization wide	managa	ragnization	nortal: Sitor	(continued)
	VIGUIIZQIIOII-WIGE /	Oldanization-wide	munuue – O	raanizanon	DOLIAI. SILES	ICOMMUEU
	- 0			0		

11.4.1.2 Site tags

Click the **Site tags** tab in the **Overview** screen to view the tags created and added to the sites for monitoring or management purposes.

Figure 200 Organization-wide > Organization-wide manage > Organization portal: Site tags

Q, S	earch		2)Site tags				Ove	r the last day: 142 Clients, 199.50 GB 💎 🕒 Export
	Client	Device	% Offline	Offline device	Offline site	Site	Status	Tag	Usage
	10	5	0.0 %	0	0	1	•	more	7.93 GB
	10	5	0.0 %	0	0	1	•	test	7.93 GB

Table 168	Organization-wide >	Organization-wide manage 2	> Organization portal: Site tags

LABEL	DESCRIPTION
Search	Enter a key word as the filter criteria to filter the list of tags.
Site tags	This shows the number of site tags created and added to the sites in this organization.
Over the last day	This shows the number of clients associated with the sites in this organization and the total amount of data transmitted or received by the clients in the past day.
Export	Click this button to save the tag list as a CSV or XML file to your computer.
Status	This shows the status of Nebula Devices in sites with the specified tag.
	 Green: All Nebula Devices are online and have no alerts. Amber: Some Nebula Devices have alerts. Red: Some Nebula Devices are offline. Gray: All Nebula Devices have been offline for 7 days or more. White: No Nebula Devices.
Tag	This shows the name of the specified tag.
Site	This shows the total number of sites with the specified tag.
Offline device	This shows the number of offline Nebula Devices in all sites with the specified tag.
Client	This shows the number of clients in sites with the specified tag.
Usage	This shows the total amount of data consumed in all sites with the specified tag.
Device	This shows the total number of Nebula Devices deployed to all sites with the specified tag.
Offline site	This shows the number of offline sites with the specified tag.
% Offline	This shows what percentage of all sites with the specified tag are currently offline.
R	Click this icon to display a greater or lesser number of configuration fields.

11.4.1.3 Devices

Click the **Devices** tab in the **Organization portal** screen to view the detailed information about Nebula Devices which are connected to the sites in the selected organization.

ξ S	earch	7	5) Devices		Over the last day:	142 Clients, 200.58	GB 💎 🕒 Export
	Client	MAC address	Model	Name	Site	Status Tag	Usage 📑
	0	BREC ASBACD 9F	NSG50	B8:EC:A3:B4:CD:9F	Site11 Jack Market	•	0 bytes
	0	88ECA384CC67	NSG50	B8:EC:A3:B4:CC:67	Site09	•	0 bytes
	0	BEEC ASEA OF BE	NSG50	B8:EC:A3:B4:CF:B5	Site04	•	0 bytes
	9	50£240.50.00FE	NSG50	Home GW	Site25 MCC at Designed	•	0 bytes
	0	8860 A3-0F 08:34	NSW200-28P	Office NSW200	Site25 MCC at the second	•	0 bytes
	3	58.88573/914575	NAP102	OfficeNAP102-MESH	Site25 Million Dimension	•	0 bytes
	5	603137640715	NAP102	HomeNAP102	Site25	Home	2.61 GB
	9	BBEC A3157F4D	NSW100-10P	Home NSW100	Site25 Million and December	•	2.69 GB
	1	BEEC A3 B4 CD 87	NSG50	B8:EC:A3:B4:CD:87	Site05	•	0 bytes
	0	BRECASBACC43	NSG50	B8:EC:A3:B4:CC:43	Site16_1015_114	•	0 bytes
					K K Page 1 of 8	> > Results	per page: 10

Figure 201 Organization-wide > Organization-wide manage > Organization portal: Devices

Table 169	Organization-wide >	Organization-wide n	nanage > Organization	portal: Devices
	organization=wide >	organization-watch	nunuge - organization	portal. Devices

LABEL	DESCRIPTION			
Search	Enter a key word as the filter criteria to filter the list of connected Nebula Devices.			
Devices	This shows the number of Nebula Devices assigned to the sites in this organization.			
Over the last day	This shows the number of clients associated with the sites in this organization and the total amount of data transmitted or received by the clients in the past day.			
Export	Click this button to save the Nebula Device list as a CSV or XML file to your computer.			
Status	 This shows the status of the Nebula Device. Green: The Nebula Device is online. Amber: The Nebula Device recently had alerts. Red: The Nebula Device was recently offline. Gray: The Nebula Device has been offline for more than 6 days. 			
Model	This shows the model number of the Nebula Device.			
Name	This shows the descriptive name of the Nebula Device.			
Site	This shows the name of the site to which the Nebula Device is connected.			
MAC address	This shows the MAC address of the Nebula Device.			
Tag	This shows the user-specified tag for the Nebula Device.			
Client	This shows the number of the clients which are currently connected to the Nebula Device.			
Usage	This shows the amount of data consumed by the Nebula Device.			
Serial number	This shows the serial number of the Nebula Device.			
Configuration This shows whether the configuration on the Nebula Device is up-to-date.				
Connectivity	This shows the Nebula Device connection status.			
	The red time slot indicates the connection to the NCC is down, and the green time slot indicates the connection is up. Move the cursor over a time slot to see the actual date and time when a Nebula Device is connected or disconnected.			

LABEL	DESCRIPTION
Public IP	This shows the global (WAN) IP address of the Nebula Device.
R	Click this icon to display a greater or lesser number of configuration fields.

Table 169 Organization-wide > Organization-wide manage > Organization portal: Devices (continued)

11.4.2 Configuration Management

Configuration synchronization allows you to easily copy configurations from one site or Nebula Device to another. Use this screen to synchronize the configuration between sites or switch ports. You can also back up the current configurations for sites or switches to the NCC and restore the configuration at a later date.

Click Organization-wide > Organization-wide manage > Configuration management to access this screen.

Synchronization					
	Settings:	Site-wide general settings	•		
	From source site:	Hub		•	
	To site(s):	Select some sites		•	
	What will be sync	hronized?		Suno	
				Sync	
Switch settings clone					
	From source	Office NSW200		Ŧ	
	To device(s):	Calant name deviage			
	10 000100(0).			•	
	What will be close	Include uplink port settings		_	
	What will be clone	<u>50 :</u>		Clone	
Backup & restore Beta					
Site(s) settings	Backup Desc	cription	Date	(UTC) A	dmin
			*		-
			~ _		
	+ Add				
	What is this?			Pestera	
				Restore	
Switch acttings					
Switch settings	Backup Switch	Description	Model	Date (UTC)	Admin
	1	>	< *	Never	÷
	+ Add				
	What is this?			Restore	

Fiaure 202	Oragnization-wide >	· Oraanization-wide manage 3	> Configuration management

Table 170 Organiz	ation-wide > Oraanizatio	n-wide manaae >	 Configuration 	manaaement

LABEL	DESCRIPTION
Synchronization	
Settings	Specify whether general site configuration or just SSID settings of a site will be propagated to other sites. Click What will be synchronized? to view detailed information.
From source site	Select the site from which you want to copy its site configuration to other sites.
To Site(s)	Select one or more sites to which you want to import the copied site configuration. You can also select the site tags created using the Organization-wide > Organization-wide manage > Organization portal: Sites screen.

NCC User's Guide

LABEL	DESCRIPTION
Sync	Click this button to start synchronizing configuration settings between the selected sites.
Switch settings clone	
From source device	Select the Nebula Switch from which you want to copy its Switch port settings to other Nebula Devices.
To device(s)	Select one or more Nebula Switches to which you want to import the copied Switch port settings.
	Note: Only Nebula Switches of the same model can synchronize. Both Switches should be registered to a site in the organization.
Clone	Click this button to start synchronizing Switch port settings between the selected Nebula Devices.
Backup & Restore	
Note: To back up or full access to	restore a previously saved configuration, your administrator account should have the organization.
Site(s) settings	You can create up to three site configuration backups for the organization.
	The NCC automatically creates and saves one backup when you perform configuration restoration. The automatic backup cannot be deleted.
Backup	This shows the index number of the site configuration backup.
Description	This shows the descriptive name of the backup.
	Note: When you click Add to create a new backup, you need to enter a name for the backup in order to save it to the NCC.
Date (UTC)	This shows the date and time the backup was saved on the NCC server.
Admin	This shows the name of the administrator account who performed the backup.
Remove	Click the remove icon to delete the backup.
Add	Click this button to create a new configuration backup of all the sites in the organization.
Restore from backup	Select the backup you want to restore.
Restore to site(s)	Select one or more sites to which you want to restore the specified configuration backup.
Restore	Click this button to overwrite the settings of the sites with the selected configuration backup.
Switch settings	At the time of writing, only one backup is allowed per Nebula Device.
Backup	This shows the index number of the Switch configuration backup.
Switch	This shows the name of the Switch.
Description	This shows the descriptive name of the backup.
	Note: When you click Add to create a new backup, you need to enter a name for the backup in order to save it to the NCC.
Model	This shows the model number of the Switch.
Date (UTC)	This shows the date and time the backup was saved on the NCC server.
Admin	This shows the name of the administrator account who performed the backup.
Remove	Click the remove icon to delete the backup.
Add	Click this button to create a new configuration backup of a specific Switch.
	This button is selectable only when you have at least one Switch in the organization.
Restore from backup	Select the backup you want to restore.

Table 170	Organization-wide	> Organization-wide	manage >	Configuration	management	(continued)
	- 0		0 -			1 /

LABEL	DESCRIPTION
Restore to device(s)	Select one or more Nebula Switches to which you want to restore the specified configuration backup.
	Note: You can restore the backup to the same Switch or Switches of the same model and registered to a site in the organization.
Restore	Click this button to overwrite the settings of the Switches with the selected configuration backup.

Table 170	Organization-wide	> Organization-wide	manage >	Configuration	management (continued)
	0	0	0	0		/

11.4.3 Configuration Templates

A configuration template is a virtual site. The settings you configured in a template will apply to the real sites which are bound to the template. If you do not want to apply any new settings from the template to a site, just unbind that site. If you want to configure some specific settings directly in a site after the site is bound to a template, turn on the local override function (see Section 11.4.3.3 on page 522).

Use this screen to create and manage configuration templates. You can then bind or unbind a site from the template (see Section 11.4.3.1 on page 521).

Note: A site can only be bound to one template. The same template can be used by multiple sites. The sites and the template should belong to the same organization for binding.

Note: If the NCC service is downgraded from Nebula Professional Pack to Nebula Base, all the sites will be unbound from the templates but retain the settings already applied from the template.

Click Organization-wide > Organization-wide manage > Configuration templates to access this screen.

Figure 203 Organization-wide > Organization-wide manage > Configuration templates

nfiguration templates			
+ Create - Delete Q. Search	▼ 1 Template		
Name	# bound sites	Bound sites	
Template1	0		4

|--|

LABEL	DESCRIPTION				
Create	Click this button to create a new configuration template. You can copy settings from an existing site or configuration template, or have a new template with default settings. It is optional to bind one or more sites to the template when you are creating a template.				
	Create a new template X Template name: My Template X Import settings from: Create new configuration template Vou could also bind sites during create template: Target sites: Taipei London E Berlin E Close Create				
Delete	Click this button to remove the selected templates. A window pops up asking you to confirm that you want to delete the templates. If you remove a template that is being used by a site, the site will be unbound from the template automatically and retain the settings previously applied from the template. Delete template confirmation X Are you sure you wish to delete template(s) which bound site(s) as below: My Template 2 (5 sites bound)				
	Warning: Template will be deleted, any bound sites will be unbound and keep current setting. Clase Delete				
Search	Enter a key word as the filter criteria to filter the list of templates.				
Templates	This shows how many templates match the filter criteria and how many templates are created in total.				
Name	This shows the name of the template.				
# Bound sites	This shows the number of the sites bound to the template.				
Bound sites	This shows the name of the sites bound to the template.				

11.4.3.1 Site Binding

Use this screen to bind or unbind a site from a template. Click an existing template from the list in the **Organization-wide > Organization-wide manage > Configuration templates** screen to access this screen. To go back to the previous screen, click the **Configuration templates list** link.

nfigurat	ion templates				
onfigura	ition template list / My Template				
Bind add	litional site Unbind Q Search	• 5 Site			
	Name	Tag	Device	Local Override	
	APP	Test	0	SWITCH	
	Backup2		1		
	APP_L1_Test		0		
	APP1		1		
	000000		0		

Figure 204 Organization-wide > Organization-wide manage > Configuration templates: Template

The following table describes the labels in this screen.

Table 172 Organization-wide > Organization-wide manage > Configuration templates: Template

LABEL	DESCRIPTION		
Bind additional site	Click this button to bind more sites to the template. A window displays. Select the name of the sites in the Target sites field and click Bind.		
Unbind	Click this button to remove the selected sites from the template. The site which is unbound from the template still retains the settings applied from the template.		
Search	Enter a key word as the filter criteria to filter the list of sites.		
Sites	This shows how many sites match the filter criteria and how many sites are bound to the template in total.		
Name	This shows the name of the site bound to the template.		
Tag	This shows the tags added to the site.		
Device	This shows the number of Nebula Devices which are assigned to the site.		
Local override	This shows which settings in the template do not apply to the site.		

11.4.3.2 Template settings

An administrator that has full access to the organization can modify the template configurations. To access a template's configuration screen, select the template name from the **Site** field in the NCC title bar. It also shows the number of sites that are bound to the template on each configuration screen.

Note: At the time of writing, you can use a template to configure site-wide, Switches, and access points settings.

11.4.3.3 Local Override

When a site is bound to a template, you can see the name of the template on the site's configuration screens (which are also available in a template and can be configured).

There is also an option to make the changes you made locally to a site persist. If you select the override check box of the site's configuration screen, all the configuration screens under the same menu tab

(Site-Wide or Switches) are configurable. Settings in these screens will not be affected and modified by the template. If the override check box is not selected, any changes of the same configuration screen in the template apply to the site.

11.4.3.4 Switch Port Profile and Configuration

Just as a configuration template is a virtual site, so is a profile to a Switch. The settings you configured in a profile will apply to the Switches which are bound to the profile. If you do not want to apply any new settings from the profile to a Switch, just unbind that Switch. If you want to configure some specific settings directly in a Switch (For example, a port's **Broadcast (pps)** value. See Section 6.3.1.1 on page 291 for details.) after the Switch is bound to a profile, turn on the local override function (see Section 11.4.3.3 on page 522).

11.4.4 VPN Orchestrator

VPN Orchestrator enables you to automatically create Virtual Private Network (VPN) connections between sites within an organization. This allows the Security Gateway of each site and the Nebula Devices behind it to communicate securely.

Note: You can manually create VPN connections between sites at Site-wide > Configure > Security Gateway > Site-to-Site VPN or Site-wide > Configure > Firewall > Site-to-Site VPN.

11.4.4.1 Topology Overview

There are two topologies you can use when creating a site-to-site VPN.

- Fully Meshed: In a fully-meshed VPN topology (1 in the figure below), there is a VPN connection between every two sites in the organization. Sites can communicate directly with each other, but having permanent tunnels between every site takes up more resources.
- Hub-and-spoke: In a hub-and-spoke topology (2 in the figure below), every site is either a hub or a spoke. There is a VPN connection between each spoke site (B, C, D, and E) and the hub site (A). Traffic from each spoke site must first go through the hub site. If the hub site fails, the site-to-site VPN network fails. To avoid this, you can assign more than one hub site.





11.4.4.2 VPN Areas

An organization can contain multiple VPN areas. Each VPN area is an independent VPN with its own sites, settings, and topology. Every organization has a default VPN area called Default, which cannot be deleted. Sites in different VPN areas within the same organization can communicate if you enable the **Area communication** setting.

11.4.4.3 VPN Orchestrator Screen

Use this screen to manage and create site-to-site VPNs within the current organization. Click **Organization-wide > Organization-wide manage > VPN orchestrator** to access this screen.

PN Topology								
"N Area FLEX			•					
© Q Latrator for		Off Norgan Online Kingdom 12 ark	Finland Belarus		Russia		-	Map Sate
		0/1 Frage	tria Romania	Kazakhetan	Monapilia	5. 5		
		Cin Haly	Greece Turkey	Kyrgyzetan	X	and the of Jacob		
North Atlant Ocea	h ic	Portugal	Syria	Alghanistan	China	South Korea Japan		North
		Algeria	Libya Egypt	Iran Pakistan	Nepal	Creatiles		Pacific Ocean
PN connected / VPN d	lisconnected / Pa	rtial VPN connected	Non-Nebula VPN connec	ted 💉 Non-Nebula VPN dis	connected	Keyboard shortout	a Map data 62022 Google,	INEOR 1000 km t1 1
art VPN								
i Area		Default		*				
ology		Hub-and-Spa	iloa	-				
inch to Branch VPN:								
Q. Security Gatewa	• 1):	ecurity gateway						
	0							
o aite: 1	0							
osite: 1 Site	Modal	VPN enable	Subnet(s)	NAT troversal	Area commun	ication Gateway status	VPN status	WAN status
o aite:1 ■ 5ito \$• 1 <u>Sito</u>	Model	VPN enable	Subnet(s) 192368.10/24 @ 192368.20/24 @	NAT traversal	Area commun	Coline	VPN status Connected	WAN status wont: 192368188.48 Public IP: 111249.66 won2: Public IP:
alte:1 Site \$ 1 Site01.050 Q. Security Octaway	Model REX500 USO FLE + (d) sec	VPN enable x 500	Subnet(s) 19236810/24 @ 19236820/24 @	NAT troversal	Area commun	Coline	VPN status Connected	WAN status word: 192368.88.48 Public IP:11249.82 word: Public IP:
paite:1 Site Site C. Security Cotevery ke site: 4 Site	Model FLEX650 USO FLEX + (d) sec Model	VPN enable	Subnet(s) 192583.0/24 @ 192588.20/24 @	NAT traversal	Area communicati	on Goteway status	VPN stotus Connected VPN stotus	WAN status wort 192508-188.48 Public IP: 111249.86 worg Public IP:
exite: 1 Site Site Q. Security Coteway ke site: 4 Site	Model + (4) sec Model	VPN enable x soo • unity gateway vPN enable s	Subnet(a) 192368.0(24 @ 192368.2.0(24 @) subnet(a) 192368.8.0(24 @)	NAT traversal	Area communicati	on Gotaway status	VPN stotus Connected VPN stotus	WAN status wort 192308-188.4 Public IP: 11240.8 Work: Public IP:
site:1 Site SiteO2_USO_FLIXSO SiteO2_USO_FLIXSO	Model + (d) exc Model USB FLEX 500	VPN enable x soo • uurity gateway VPN enable s	Submet(a) 192368.10/24 © 192368.20/24 © submet(a) 192368.20/24 © 192368.20/24 ©	NAT traversal	Area communicati	on Gateway status Online Online	VPN status Connected VPN status Connected	WAN status want 192508-188.45 Public IP: 11240.86 wang, Public IP: WAN status want 192508-188.55 Public IP: 111249.863
site:1 Site C. Security Octaway ke site: 4 Site Site	Model + (d) sec Model USB FLEX 500	VPN enable X 500 • Lunhy poteway VPN enable 5	Subnet(s) 192368.0/24 @ 192368.2.0/24 @ 192368.2.0/24 @ 192368.8.0/24 @ 192368.8.0/24 @	NAT troversol	Area communicati	on Gateway status Online Online	VPN status Connected VPN status Connected	WAN status wont 192368188.48 Public IP: 11240.86 won2 Public IP: WAN status won1: 192368.188.53 Public IP: 11240.863
site:1 Site Site C. Security Outeway Site Site Site	Model + (4) sec Model USG FLEX 500 USG FLEX 500	VPN enable X 500 • • • • • • • • • • • • • • • • • •	Subnet(s) 192368.2.0/24 © 192368.2.0/24 © 192368.2.0/24 © 192368.8.0/24 © 192368.5.0/24 © 192368.15.0/24 ©	NAT troversol	Area communicati	ication Gateway status Online Online Online Online Online Online Online	VPN status Cennected VPN status Connected Connected	WAN status won1 192 188 188 48 Public IP: 11240 JB WaR Public IP: WAN status won1: 192 188 188 48 won1: 192 188 188 53 Public IP:
eite: 1 Site Site Site Site Site Site Site Sit	Model FLEX500 USG FLEX Model USG FLEX 500 USG FLEX 200	VPN enable X 500 C Unity gateway VPN enable S C	Subnet(a) 192368.0/24 @ 192368.20/24 @ 192368.50/24 @ 192368.50/24 @ 192368.50/24 @ 192368.50/24 @	NAT traversal	Area communication	iontion Goteway status Online on Goteway status Online Online	VPN status Cennected VPN status Connected	WAN status wont 1923/08/388.4 Public IP 11246.8 wont, Public IP: WAN status wont 1923/08/388.51 Public IP 111249.86. wont 1923/08/388.51
site: 1 Site Sit	Model TLEXESS USG FLEE Model USG FLEX 500 USG FLEX 200 USG FLEX 200	VPN enable X 500 C Unity gateway VPPN enable 8 C	Subnet(a) 192588.20/24 @ 192588.20/24 @ 192588.20/24 @ 192588.30/24 @ 192588.30/24 @ 192588.320/24 @	NAT traversal	Area communication	on Gateway status on Gateway status on Gateway status onine Online Online Online	VPN status Connected Connected Connected Connected	WAN status won1 192 168 188 48 Public IP: 11249 JB Public IP: WAN status Won1 192 168 188 49 Public IP: WAN status won1 192 168 188 49 Public IP: 11249 JB Won1 192 168 188 49
> site • 1	Model FLEXEDO USO FLEX USO FLEX 500 USO FLEX 500 USO FLEX 500 USO FLEX 500	VPN enable X 500 UITly Octewory VPN enable 0 0 0 0 0 0 0 0 0	Subnet(a) 102368.0/24 © 102368.2.0/24 © 102368.0/24 © 102368.6.0/24 © 102368.5.0/24 © 102368.26.0/24 ©	NAT traversal	Area communicada	Iostion Coteway status Online Online Online Online Online	VPN status Connected Connected Connected Connected	WAN status won1 192/108 188.41 Public IP: 111240.81 Public IP: 111240.81 WAN status WAN status won1: 102/68.188.51 Public IP: 111240.86.51 Public IP: 111240.86.51 won1: 102/68.188.51 Public IP: 111240.86.51 won1: 102/68.88.49 Public IP: 111249.86.52 won1: 102/68.88.49 Public IP: 111249.86.52 won1: 102/68.86.49 Public IP: 111249.86.51
Image: Site Image: Site Image: Site Image: Site Image: Site Site Image: Site <td>Model ILINESSO USG PLEX Wodel USG PLEX 500 USG PLEX 500 USG PLEX 200 NSG500</td> <td>VPN enable X 500 C</td> <td>Subnet(a) 122368.0(24 @ 122368.2.0(24 @) 122368.2.0(24 @) 122368.9.0(24 @) 122368.9.0(24 @) 122368.9.0(24 @) 122368.9.0(24 @) 122368.9.0(24 @)</td> <td>NAT troversol</td> <td>Area communicati</td> <td>Ioation Cateway status Online on Cateway status Online Online Online Online</td> <td>VPN status Cennected VPN status VPN status Connected Connected Connected Disconnected</td> <td>WAN status won1 192.108 188.45 Public IP: 111240 JB Public IP: WAN status WAN status Wan1 192.108 108.53 Public IP: WAN status Wan1 192.268 108.53 Public IP: Wan1 192.268 108.53 Public IP: Wan1 192.268 108.53 Wan1 192.268 108.53 Public IP: Wan1 192.268 108.53 Wan1 192.268 108.53 Wan1 192.168 108.51 Wan1: Wan1:</td>	Model ILINESSO USG PLEX Wodel USG PLEX 500 USG PLEX 500 USG PLEX 200 NSG500	VPN enable X 500 C	Subnet(a) 122368.0(24 @ 122368.2.0(24 @) 122368.2.0(24 @) 122368.9.0(24 @) 122368.9.0(24 @) 122368.9.0(24 @) 122368.9.0(24 @) 122368.9.0(24 @)	NAT troversol	Area communicati	Ioation Cateway status Online on Cateway status Online Online Online Online	VPN status Cennected VPN status VPN status Connected Connected Connected Disconnected	WAN status won1 192.108 188.45 Public IP: 111240 JB Public IP: WAN status WAN status Wan1 192.108 108.53 Public IP: WAN status Wan1 192.268 108.53 Public IP: Wan1 192.268 108.53 Public IP: Wan1 192.268 108.53 Wan1 192.268 108.53 Public IP: Wan1 192.268 108.53 Wan1 192.268 108.53 Wan1 192.168 108.51 Wan1: Wan1:
a Site a Site b 1 Site a Site Site a Site Site b Site Site	Model Model Model USB FLEX 500 USB FLEX 500 USB FLEX 200 USB FLEX 200 NSB200	VPN enable x soo Unity gateway VPN enable	Subnet(a) 192368.2.0/34 @ 192368.2.0/34 @ 192368.2.0/34 @ 192368.2.0/24 @ 192368.2.0/24 @ 192368.2.0/24 @ 192368.2.0/24 @	NAT troversol	Area communicati	ication Coteway status Online	VPN status Cennected VPN status Connected Connected Connected Disconnected	WAN status won1 192 588 188 48 Public IP: 111240 88 Public IP: WAN status WAN status won1 192 588 188 49 Public IP: WAN status won1 192 588 188 51 Public IP: 11249 86 won1 192 588 188 51 WAN: 192 188 188 51 WAN: 192 188 188 51
a Site a Site b Site c Site	Model LINSIGO USO PLEX Model USO PLEX 500 USO PLEX 500 USO PLEX 200 NS0300	VPN enable x soo VPN enable VPN enable	Submet(a) 192368.2.0/24 @ 192368.2.0/24 @ 192368.2.0/24 @ 192368.2.0/24 @ 192368.2.0/24 @ 192368.2.0/24 @ 192368.2.0/24 @ 192368.2.0/24 @	NAT troversol	Area communication	ication Cateway status Online on Cateway status Online Online Online Online Online Online Online	VPN status Connected Connected Connected Connected Disconnected	WAN stotus wont 192:58,188.48 Public IP: 111249.86 wong Public IP: 111249.86 WAN stotus Want 192:581.188.51 Public IP: 111249.86 Want 192:581.188.51 Public IP: 111249.86 Want 192:588.188.51 Want 192:588.188.51 Want 192:588.188.51 Want 192:588.188.51
b aite; 1 Site S	Model Model Model USG FLEX 500 USG FLEX 500 USG FLEX 200 NSG200 NSG200	VPN enable X 500 VPN enable S VPN enable S VPN enable S VPN enable S VPN enable	Submet(s) 192368.20/24 192368.20/24 192368.20/24 192388.20/24 1000000000000000000000000000000000000	NAT traversal	Area communication	iontion Coteway status Online Online Online Online Online Online Online Online Pre-shored secret	VPN status Connected Connected Connected Connected Disconnected Address	WAN status want 192368.188.48 Public IP: 111249.06 Want Status WAN status Want 192368.188.48 Wang 2000 Want Status Want 192368.188.51 Want 192368.188.51 Want 192368.188.51 Want 192368.188.51 Want 192268.188.51 Want 192268.188.51

Figure 206 Organization-wide > Organization-wide manage > VPN orchestrator

Table 173 Organization-wide > Organization-wide manage > VPN orchestrator

LABEL	DESCRIPTION
VPN Topology	
VPN Area	Select the name of a VPN area to view on the map.
	Select Overview to view all VPN areas in this organization on the map.
Smart VPN	
VPN Area	Select the name of a VPN to configure.
	Select + Create VPN area to create a new VPN within the organization.
1	Click the remove icon to delete the VPN area.



LABEL	DESCRIPTION
Topology	Click this to select a topology for the VPN area. For details on topologies, see Section 11.4.4.1 on page 523.
	Select Disable to disable VPN connections for all sites in the VPN area.
The following settings of	are shown when Topology is set to Hub-and-Spoke.
Branch to Branch VPN	Enable this to allow spoke sites to communicate with each other in the VPN area. When disabled, spoke sites can only communicate with hub sites.
Spoke	Select one or more sites and then click this to assign the sites as spokes. The sites are added to the spoke list.
Hub	Select one or more sites and then click this to assign the sites as hubs. The sites are added to the hubs list.
Security Gateway	Enter the name of a site or Nebula Device to filter the list of sites.
Hub site	This shows the number of hub site.
	Note: Only one hub site is supported
Spoke site: N	This shows the number of spoke sites (N) in the spoke list
#	This shows the priority of the hub site. If the VPN area contains multiple hub sites, then the
	spoke sites always send traffic through the available hub with the highest priority.
	You can change the priority of a site by clicking the move icon (\leftrightarrow), and then dragging the site up or down in the list.
Site	This shows the name of the site in the VPN area.
Model	This shows the model of the site's Security Gateway device.
VPN enable	Click this to enable or disable site-to-site VPN on the site's Security Gateway.
	If you disable this setting, the site will leave the VPN area.
Subnets	This shows the IP subnets of all LAN interfaces behind the site's Security Gateway.
NAT traversal	If the Security Gateway is behind a NAT router, enter the public IP address or the domain name that is configured and mapped to the Security Gateway on the NAT router.
Area communication	Enable this to allow the site to communicate with sites in different VPN areas within the organization.
	If Topology is set to Site-to-Site , then you must assign at least one site in each VPN area as the Area Leader . The area leaders create VPN tunnels between VPN areas.
Gateway status	This shows whether the site's Security Gateway is currently online.
VPN status	This shows whether the VPN is currently connected.
WAN status	This shows the IP address of the WAN interface and the public IP address of the site's Security Gateway.
Non-Nebula VPN peers	Configure this section to add a non-Nebula gateway, such as an on-premise ZyWALL series device or non-Zyxel gateway, to the VPN area.
+ Add	Click this button to add a non-Nebula gateway to the VPN area.
Enabled	Select the check box to enable VPN connections to the non-Nebula gateway.
Name	Enter the name of the non-Nebula gateway.
Public IP	Enter the public IP address of the non-Nebula gateway. The public IP address supports both FQDN (Fully Qualified Domain Name) and IP formats.
Private Subnet	Enter the IP subnet that will be used for VPN connections. The IP range must be reachable from other Nebula Devices in the VPN area.
IPSec policy	Click to select a pre-defined policy or have a custom one. See Section 9.3.6.1 on page 452 for detailed information.

Telela 172	Ore environtions wide >	Organization wide manages	V/DNL arabatratar	
100101017.5	Undani/alion-wide 2	• Urdanizaiion-wide manade 2	> VPIN OFCHESITOLOF	ICONIINUEOL
	erganization mae	organization machinariage		

LABEL	DESCRIPTION
Preshared secret	Enter a pre-shared key (password). The Nebula Security Gateway and peer gateway use the key to identify each other when they negotiate the IKE SA.
Address	Enter the address (physical location) of the Nebula Device.
1	Click the remove icon to delete the entry.

 Table 173
 Organization-wide > Organization-wide manage > VPN orchestrator (continued)

11.4.5 Security Profile Sync

Security profile sync allows you to share the same Security Firewall gateway device security service settings with multiple sites in an organization. This replaces the Unified Threat Management (UTM) settings configured for each site at **Site-wide** > **Configure** > **Firewall** > **Security service**.

11.4.5.1 Configuring Security Profile Sync

Follow the steps below to enable security profile sync in an organization.

1 Go to Organization-wide > Organization-wide manage > Security profile sync. Select Enabled, and then under Sync sites add the sites that you want to share security settings.

Note: You can only add sites that have a Security Firewall gateway device.

- 2 Configure security service settings for Content filtering, Application Patrol, URL Threat Filter, Anti-Malware, and Intrusion Detection / Prevention. Then click Save. All security settings are synced to the selected sites.
- 3 If you change the settings in the Security profile sync screen, the changes will be copied to all selected sites.
- 4 If you want to modify security settings for an individual site, go to Site-wide > Configure > Firewall > Security service and select Override security profile sync.

11.4.5.2 Security Profile Sync Screen

Use this screen to enable and configure security profile sync. Click **Organization-wide > Organization-wide manage > Security profile sync** to access this screen.

Security profile sync	
Enabled	
Sync sites	All sites O
Content filtering Model list	
Drop connection when there is an HTTPS connection with SSL v3(or previous version)	
Denied Access Message	Web access is restricted. Please contact the administrator. \times
Redirect URL	×
	There are no content filtering rules defined for this site. + Add
Application Patrol Model list	
Application profiles	There are no profiles defined for this site.
	+ Add
DNS/URL Threat Filter Model list	
Log	
DNS Threat Filter	
DNS Threat Filter policy	Redirect
DNS Threat Filter Redirect IP	Default 👻
URL Threat Filter	
URL Threat Filter policy	Block 👻
URL Threat Filter Denied Access Message	Web access is restricted. Please contact the administrator. $ imes$ *
URL Threat Filter Redirect URL	×
Test Threat Category	X Test
Category list	Anonymizers Anonymizers
Block list	
	FQDN(support wildcard)
Allow list	
	FQDN(support wildcard)
URL Threat Filter external block list	Enabled Name External DB Description
	+ Add
Schedule update	External DB schedule update
	factor of the second second

Figure 207 Organization-wide > Organization-wide manage > Security Profile Sync

VIP Reputation Model list		
Enabled		
Log		
Policy	Block 👻	
Threat level threshold	Medium and above 💌	
Test Category		
	X	
Category list 👔	Anonymous Proxies Openial of Service Exploits Negative Reputation Scanners Spam Sources Tor Proxies Web Attacks Phishing BotNets	
Block list	IP or CIDR	×
Allow list	IP or CIDR	×
External block list	Enabled Name External DB Description	
		ŵ
	+ Add	
Schedule update	External DB schedule update	
	Daily	
	0300	
	0.00	
Y Anti-Malware Model list		
Enabled		
Log		
Scan mode	Stream mode Express mode Hybrid mode	
Scan mode Cloud Query	Stream mode Express mode Hybrid mode	
Scan mode Cloud Query	Stream mode Express mode Hybrid mode	
Scan mode Cloud Query Block list	Stream mode Express mode Hybrid mode File Types	
Scan mode Cloud Query Block list	Stream mode Express mode Hybrid mode File Types	×
Scan mode Cloud Query Block list	Stream mode Express mode Hybrid mode File Types	×
Scan mode Cloud Query Block list Allow list	Stream mode Express mode Hybrid mode Image: Pile Types	×
Scan mode Cloud Query Block list Allow list	Stream mode Express mode Hybrid mode - - File Types	×
Scan mode Cloud Query Block list Allow list	Stream mode Express mode Hybrid mode File Types	×
Scan mode Cloud Query Block list Allow list	Stream mode Express mode Hybrid mode File Types File Pattern File Pattern File Pattern	×
Scan mode Cloud Query Block list Allow list	Stream mode Express mode Hybrid mode File Types File Pattern File Pattern File Pattern	×
Scan mode Cloud Query Block list Allow list Sandboxing Model list Enabled	Stream mode Express mode Hybrid mode File Types File Pattern File Pattern	×
Scan mode Cloud Query Block list Allow list Sandboxing Model list Enabled Log	Stream mode Express mode Hybrid mode File Types	×
Scan mode Cloud Query Block list Allow list Sandboxing Model list Enabled Log Policy	Stream mode Express mode File Types File Pattern File Pattern File Pattern	×
Scan mode Cloud Query Block list Allow list Sandboxing Model list Enabled Log Policy Inspect selected downloaded files	Stream mode Express mode File Types File Pattern File Pattern File Pattern	×
Scan mode Cloud Query Block list Allow list Sandboxing Model list Enabled Log Policy Inspect selected downloaded files File submission pations	Stream mode Express mode File Types File Pattern File Pattern File Pattern	×
Scan mode Cloud Query Block list Allow list Sandboxing Model list Enabled Log Policy Inspect selected downloaded files File submission options	Stream mode Express mode File Types File Pattern File Pattern File Pattern Ele Pattern Destroy Destroy Ele Types TIP Archives (zip) Executables (exe) Ms Office Documents (doc) Macromedia Flash Data (swf) File Types	×
Scan mode Cloud Query Block list Allow list Sandboxing Model list Enabled Log Policy Inspect selected downloaded files File submission options Vince (IPS) Model Intrusion Prevention System (IPS) Model	Stream mode Express mode File Types File Pattern File Pattern File Pattern File Pattern E File Pattern C Destroy C C Stroy C Destroy C File Types E File Types E Stroy C File Types E Stroy Time Document (rdf) Time Document (rdf) File Types E	×
Scan mode Cloud Query Block list Allow list Sandboxing Model list Enabled Log Policy Inspect selected downloaded files File submission options File submission options	Stream mode Express mode File Types File Pattern File Pattern File Pattern File Pattern C Destroy C C Destroy C C Destroy File Types Ele Types Ele Types	×

NCC User's Guide

Talala 171				Coourity uprofile autoo
Table 174	Organization-wid	e > Organization-\	vide manage >	Security profile sync

LABEL	DESCRIPTION			
Security profile sync				
Enabled	Click this to enable or disable security profile sync for the organization.			
Sync sites	Select one or more sites that you want to sync the security settings on this screen to. Select All sites to sync security settings with all sites in the organization.			
	Note: You can only add sites that have a Security Firewall gateway device.			
Content Filtering				
Drop connection when there is an HTTPS connection with SSL v3 (or previous version)	Select On to have the Security Gateway block HTTPS web pages using SSL V3 or a previous version.			
Denied Access Message	Enter a message to be displayed when content filter blocks access to a web page. Use up to 127 characters (0–9a–zA–Z;/?: $@&=+$ \!~*'()%,"). For example, "Access to this web page is not allowed. Please contact the network administrator".			
	It is also possible to leave this field blank if you have a URL specified in the Redirect URL field. In this case if the content filter blocks access to a web page, the security gateway just opens the web page you specified without showing a denied access message.			
Redirect URL	Enter the URL of the web page to which you want to send users when their web access is blocked by content filter. The web page you specify here opens in a new frame below the denied access message.			
	Use "http://" or "https://" followed by up to 262 characters (0–9a–zA–Z;/?:@&=+\$\ _!~*'()%). For example, http://192.168.1.17/blocked access.			
Enabled	Select the check box to enable the content filtering profile.			
Description	Enter a description for this profile.			
2	Click this icon to change the profile settings.			
1	Click this icon to remove the profile.			
Add	Click this to create a content filtering profile. See Section 8.3.8.1 on page 394 for more information.			
Application Patrol				
Application profiles				
Name	Enter a name for this profile for identification purposes.			
Description	Enter a description for this profile.			
2	Click this icon to change the profile settings.			
1	Click this icon to remove the profile.			
Add	Click this icon to create an application patrol profile. See Section 8.3.8.2 on page 397 for more information.			
DNS/URL Threat Filter				
Log	Select whether to have the Nebula Device generate a log (log), log and alert (log alert) or not (no) when the policy is matched to the criteria listed above.			
DNS Threat Filter	Select On to turn on the rule. Otherwise, select Off to turn off the rule.			
DNS Threat Filter policy	Select Pass to have the Nebula Device allow the DNS query packet and not reply with a DNS reply packet containing a default or custom-defined IP address.			
	Select Redirect to have the Nebula Device reply with a DNS reply packet containing a default or custom-defined IP address.			

LABEL	DESCRIPTION
DNS Threat Filter Redirect IP	Enter the IP address to have the Nebula Device reply with a DNS reply packet containing a default or custom-defined IP address when a DNS query packet contains an FQDN with a bad reputation. The default IP is the dnsft.cloud.zyxel.com IP address. If you select a custom-defined IP, then enter a valid IPv4 address in the text box.
URL Threat Filter	Select On to turn on the rule. Otherwise, select Off to turn off the rule.
URL Threat Filter Policy	Select Pass to allow users to access web pages that the external web filtering service has not categorized.
	Select Block to prevent users from accessing web pages that the external web filtering service has not categorized. When the external database content filtering blocks access to a web page, it displays the denied access message that you configured in the Content Filter General screen along with the category of the blocked web page.
	Select Warn to display a warning message before allowing users to access web pages that the external web filtering service has not categorized.
URL Threat Filter Denied Access Message	Enter a message to be displayed when content filter blocks access to a web page. Use up to 127 characters (0–9a–zA–Z;/?:@&=+ $\$ _!~*'()%,"). For example, "Access to this web page is not allowed. Please contact the network administrator".
	It is also possible to leave this field blank if you have a URL specified in the Redirect URL field. In this case if the content filter blocks access to a web page, the Nebula Device just opens the web page you specified without showing a denied access message.
URL Threat Filter Redirect URL	Enter the URL of the web page to which you want to send users when their web access is blocked by content filter. The web page you specify here opens in a new frame below the denied access message.
	Use "http://" or "https://" followed by up to 262 characters (0–9a–zA–Z;/?:@&=+ $\!~*'()$). For example, http://192.168.1.17/blocked access.
Test Threat Category	Enter a URL using http://domain or https://domain and click the Test button to check if the domain belongs to a URL threat category.
Category List	These are categories of web pages based on their content. Select categories in this section to control access to specific types of Internet content.
Block list	Sites that you want to block access to, regardless of their content rating, can be blocked by adding them to this list.
	Enter host names such as www.bad-site.com into this text field. Do not enter the complete URL of the site – that is, do not include "http://". All sub-domains are also blocked. For example, entering "bad-site.com" also blocks "www.badsite.com", "partner.bad-site.com", "press.bad-site.com", and so on. You can also enter just a top level domain. For example, enter .com to block all .com domains.
	Use up to 127 characters (0–9 a–z). The casing does not matter.
Allow list	Sites that you want to allow access to, regardless of their content rating, can be allowed by adding them to this list.
	Enter host names such as www.good-site.com into this text field. Do not enter the complete URL of the site – that is, do not include "http://". All sub-domains are allowed. For example, entering "zyxel.com" also allows "www.zyxel.com", "partner.zyxel.com", "press.zyxel.com", and so on. You can also enter just a top level domain. For example, enter .com to allow all .com domains.
	Use up to 127 characters (0–9 a–z). The casing does not matter.
URL Threat Filter external block list	The Nebula Device uses black list entries stored in a file on a web server that supports HTTP or HTTPS. The Nebula Device blocks incoming and outgoing packets from the black list entries in this file.
Enabled	Select this to have the Nebula Device block the incoming packets that come from the listed addresses in the block list file on the server.

Table 174	Organization-wide >	Organization-wide	manage > Security	profile sync (continued)
	organization mao	organization mao	manage	

LABEL	DESCRIPTION
Name	Enter an identifying name for the block list file. You can use alphanumeric and ()+/ :=?!*#@\$_%- characters, and it can be up to 60 characters long.
External DB	Enter the exact file name, path and IP address of the server containing the block list file. The file type must be 'txt'.
	For example, http://172.16.107.20/blacklist-files/myip-ebl.txt
	The server must be reachable from the Nebula Device.
Description	Enter a description of the block list file. You can use alphanumeric and ()+/:=?!*#@\$_%-characters, and it can be up to 60 characters long.
1	Click this icon to remove the entry.
Add	Click this button to create a new entry.
Schedule update	The signatures for DNS Filter and URL Threat Filter are the same. These signatures are continually updated as new malware evolves. New signatures can be downloaded to the Nebula Device periodically if you have subscribed for the URL Threat filter signatures service.
	You need to create an account at myZyxel, register your Nebula Device and then subscribe for URL Threat filter service in order to be able to download new signatures from myZyxel.
	Select Daily to set the time of the day, or Weekly to set the day of the week and the time of the day.
	Schedule signature updates for a day and time when your network is least busy to minimize disruption to your network.
IP Reputation	
Enabled	Select this option to turn on IP blocking on the Nebula Device.
Log	Select this option to create a log on the Nebula Device when the packet comes from an IPv4 address with bad reputation.
Policy	Select Pass to have the Nebula Device allow the packet to go through.
	Select Block to have the Nebula Device deny the packets and send a TCP RST to both the sender and receiver when a packet comes from an IPv4 address with bad reputation.
Threat level threshold	Select the threshold threat level to which the Nebula Device will take action (High , Medium and above, Low and above).
	The threat level is determined by the IP reputation engine. It grades IPv4 addresses.
	• High: an IPv4 address that scores 0 to 20 points.
	 Medium and above: an IPv4 address that scores 0 to 60 points. Low and above: an IPv4 address that scores 0 to 80 points.
	For example, a score of "10" will cause the Nebula Device to take action whether you set the Threat level threshold at High , Medium and above , or Low and above .
	But a score of "61" will not cause the Nebula Device to take any action if you set the Threat level threshold at Medium and above.
Test Category	Enter an IPv4 address of a website, and click the Test button to check if the website associates with suspicious activities that could pose a security threat to users or their computers.
Category list	Select the categories of packets that come from the Internet and are known to pose a security threat to users or their computers.
Block list	Sites that you want to block access to, regardless of their content rating, can be blocked by adding them to this list.
	Add the IPv4 addresses that the Nebula Device will block the incoming packets.

Table 174	Organization-wide >	Organization-wide manage >	Security profile sync	(continued)
	organization mao -	erganization manage -		

LABEL	DESCRIPTION
Allow list	Sites that you want to allow access to, regardless of their content rating, can be allowed by adding them to this list.
	Add the IPv4 addresses that the Nebula Device will allow the incoming packets.
External block list	
Enabled	Select this check box to have the Nebula Device block the incoming packets that come from the listed addresses in the block list file on the server.
Name	Enter the identifying name for the block list file. You can use alphanumeric and ()+/ :=?!*#@\$_%- characters, and it can be up to 60 characters long.
External DB	Enter the file name, path and IP address of the server containing the block list file. For example, http://172.16.107.20/blacklist-files/myip-ebl.txt
Description	Enter a description of the block list file. You can use alphanumeric and ()+/:=?!*#@\$_%-characters, and it can be up to 60 characters long.
1	Click this icon to remove the entry.
Add	Click this button to create a new entry.
Schedule update	New IP reputation signatures can be downloaded to the Nebula Device periodically if you have subscribed for the IP reputation signatures service. You need to create an account at myZyxel, register your Nebula Device and then subscribe for IP reputation service in order to be able to download new signatures from myZyxel.
	Select Daily to set the time of the day, or Weekly to set the day of the week and the time of the day.
	Schedule signature updates for a day and time when your network is least busy to minimize disruption to your network.
Anti-Malware	
Enabled	Select On to turn on the rule. Otherwise, select Off to turn off the rule.
Log	Select whether to have the Nebula Device generate a log when the policy is matched to the criteria listed above.
Scan mode	·
Express mode	In this mode you can define which types of files are scanned using the File Type For Scan fields. The Nebula Device then scans files by sending each file's hash value to a cloud database using cloud query. This is the fastest scan mode.
Stream mode	In this mode the Nebula Device scans all files for viruses using its anti-malware signatures to detect known virus pattens. This is the deepest scan mode.
Hybrid mode	In this mode you can define which types of files are scanned using the File Type For Scan fields. The Nebula Device then scans files by sending each file's hash value to a cloud database using cloud query. It also scans files using anti-malware signatures, and Threat Intelligence Machine Learning. This mode combines Express Mode and Stream Mode to offer a balance of speed and security.
Cloud Query	Select the Cloud Query supported file types for the Nebula Device to scan for viruses.

Table 174	Organization-wide > Organization-wide manage > Security profile sync		continued)

LABEL	DESCRIPTION
Block list	This field displays the file or encryption pattern of the entry. Enter a file pattern that would cause the Nebula Device to log and modify this file.
	•Use up to 80 characters. Alphanumeric characters, underscores (_), dashes (-), question marks (?) and asterisks (*) are allowed.
	•A question mark (?) lets a single character in the file name vary. For example, use "a?.zip" (without the quotation marks) to specify aa.zip, ab.zip and so on.
	•Wildcards (*) let multiple files match the pattern. For example, use "*a.zip" (without the quotation marks) to specify any file that ends with "a.zip". A file named "testa.zip would match. There could be any number (of any type) of characters in front of the "a.zip" at the end and the file name would still match. A file named "test.zipa" for example would not match.
	• A * in the middle of a pattern has the Nebula Device check the beginning and end of the file name and ignore the middle. For example, with "abc*.zip", any file starting with "abc" and ending in ".zip" matches, no matter how many characters are in between.
	•The whole file name has to match if you do not use a question mark or asterisk.
	• If you do not use a wildcard, the Nebula Device checks up to the first 80 characters of a file name.
Allow list	Enter the file or encryption pattern for this entry. Specify a pattern to identify the names of files that the Nebula Device should not scan for viruses.
	•Use up to 80 characters. Alphanumeric characters, underscores (_), dashes (-), question marks (?) and asterisks (*) are allowed.
	•A question mark (?) lets a single character in the file name vary. For example, use "a?.zip" (without the quotation marks) to specify aa.zip, ab.zip and so on.
	•Wildcards (*) let multiple files match the pattern. For example, use "*a.zip" (without the quotation marks) to specify any file that ends with "a.zip". A file named "testa.zip would match. There could be any number (of any type) of characters in front of the "a.zip" at the end and the file name would still match. A file named "test.zipa" for example would not match.
	• A * in the middle of a pattern has the Nebula Device check the beginning and end of the file name and ignore the middle. For example, with "abc*.zip", any file starting with "abc" and ending in ".zip" matches, no matter how many characters are in between.
	•The whole file name has to match if you do not use a question mark or asterisk.
	• If you do not use a wildcard, the Nebula Device checks up to the first 80 characters of a file name.
Sandboxing	Sandboxing provides a safe environment to separate running programs from your network and host devices. Unknown or untrusted programs/codes are uploaded to the Defend Center and executed within an isolated virtual machine (VM) to monitor and analyze the zero-day malware and advanced persistent threats (APTs) that may evade the Nebula Device's detection, such as anti-malware. Results of cloud sandboxing are sent from the server to the Nebula Device.
Enabled	Select this option to turn on sandboxing on the Nebula Device
Log	Enable this option to allow the Security Firewall to create a log when a suspicious file is detected.
Policy	Specify whether the Nebula Device deletes (Destroy) or forwards (Allow) malicious files. Malicious files are files given a high score for malware characteristics by the Defend Center.

Table 174	Organization-wide >	Organization-wide manage	> Security profile sync	(continued)
	organization what y	organization what manage		

LABEL	DESCRIPTION	
Inspect selected downloaded files	Select this option to have the Nebula Device hold the downloaded file for up to 2 seconds if the downloaded file has never been inspected before. The Nebula Device will wait for the Defend Center's result and forward the file in 2 seconds. Sandbox detection may take longer than 2 seconds, so infected files could still possibly be forwarded to the user. Note: The Nebula Device only checks the file types you selected for sandbox inspection. The scan result will be removed from the Nebula Device cache after the Nebula Device restarts.	
File submission options	Specify the type of files to be sent for sandbox inspection.	
Intrusion Detection/Prevention		
Detection	Select On to enable Detection.	
Prevention	Select On to enable Prevention.	

Table 171	Organization wide >	Organization wi	do managa > 1	Socurity profile	when (continued)
		Oldanization-wi	1e munuue / ,		
	- 0				

11.4.6 Firmware Management

Use this screen to upgrade Nebula Device firmware, or schedule a firmware upgrade for Nebula Devices within the sites in the organization. Click **Organization-wide > Organization-wide manage > Firmware management** to access this screen.

11.4.6.1 Firmware Management Overview Screen

Use this screen to view and/or schedule a firmware upgrade for Nebula Devices within each site in the organization. You can make different schedules for different sites in the organization. Click **Organization-wide > Organization-wide manage > Firmware management > Overview** to access this screen.

Figure 208 Organization-wide > Organization-wide manage > Firmware management > Overview



You can select Nebula Devices by device type and by site, but you cannot select individual Nebula Devices. For example, you can upgrade all Switches in Site A and all APs in Site B. To upgrade individual Nebula Devices, go to **Organization-wide > Organization-wide manage > Firmware management > Devices**.

Note: This is a Nebula Professional Pack feature. If your Nebula Professional Pack license expires, existing firmware upgrades will still run as scheduled.

11.4.6.2 Firmware Upgrade Priority

NCC prioritizes the different Nebula Device firmware upgrade schedules as follows, from highest to lowest:

1. Individual Nebula Device upgrade schedule (set at Organization-wide > Organization-wide manage > Firmware management > Devices).

2. Organization-wide or site-wide upgrade schedule. If both are set, the schedule that was most recently set takes priority.

3. NCC default per-device upgrade schedule and default site-wide upgrade schedule (14 days after new firmware is released).

11.4.6.3 Firmware Management Overview Screen

LABEL	DESCRIPTION				
Site	Select a site in your organization. By default, all the sites are displayed (Any).				
Device type	Select the type of Nebula Device. By default, all the Nebula Devices are displayed (Any).				
Status	Select the status of the Nebula Device's firmware. By default, all the status are displayed (Any).				
	Select Good to display the Nebula Devices running a stable firmware with no immediate action is required.				
	Select Warning to display the Nebula Devices with a newer firmware available and immediate action is recommended. The newer firmware may contain security enhancements, new features, and performance improvements.				
	Select Critical to display the Nebula Devices with a newer firmware available and immediate action is required. The existing firmware may have security vulnerabilities and/or lack key performance improvements.				
	Select N/A to display the Nebula Devices that are offline and its firmware status is not available.				
Availability	Select to show the Nebula Devices with Up to date firmware, there is firmware update available for the Nebula Device (Upgrade available), or a specific version of firmware has been installed by Zyxel customer support (Locked). By default, all the available firmware are displayed (Any).				
Upgrade Now	Click this to immediately upgrade the firmware on all selected sites.				
	This button is selectable only when there is firmware update available for the Nebula Devices for the selected sites.				

 Table 175
 Organization-wide > Organization-wide manage > Firmware management > Overview

LABEL	DESCRIPTION					
Schedule Upgrade	Click this to pop-up a window where you can set a specific date and time to upgrade the Nebula Devices firmware on the selected sites.					
	Schedule upgrade X					
	Note: Schedule upgrade will follow each site's time zone.					
	Upgrade policy Auto upgrade at Monday 02:00 Upgrade at 2023-01-04 12:00					
	O Upgrade now					
	Ignore upgrade					
	Firmware type Stable 🕶					
	Cancel Update					
	Note: Nebula Devices are upgraded according to the time zone of the site they are in.					
Reset	Select one or more Site-wide firmware upgrade Schedules , and then click Reset to restore the default site-wide settings (Every Monday at 02:00)					
	Select one or more Per device firmware upgrade Schedule s, and then click Reset to allow the Nebula Devices to follow the site-wide firmware management settings.					
Site-wide/Per device	Select your desired filter criteria to filter the list of firmware upgrade schedules.					
Note: Drag the follo change the s	owing column headings to change the order. Click the column heading to sorting, ascending or descending order.					
Status	This shows the status of the Nebula Device's firmware.					
	 Green: All Nebula Devices are running Stable or above firmware. Amber: One or more Nebula Devices is not running the Latest firmware. 					
	 Red: One or more Nebula Devices is running firmware that may have security vulnerabilities and/or lack key performance improvements. Gray: No schedule is set for upgrading the Nebula Device's firmware. 					
Site	This shows which site the Nebula Device is in.					
	Click the site name to go to the site's Dashboard.					
Device type	This shows the type of Nebula Device.					
Schedule	This shows the day and time when a new firmware upgrade is scheduled to occur. Site- wide settings means the Nebula Device is following the site-wide firmware schedule. Per device settings means a firmware schedule is set for the Nebula Device and it will not follow the site-wide firmware schedule.					
# of devices	This shows the number of Nebula Devices in the site for a particular Schedule status . Click this to change the schedule (see the Schedule upgrade field in Table 176 on page 538 for more information).					

 Table 175
 Organization-wide > Organization-wide manage > Firmware management > Overview

LABEL	DESCRIPTION
Availability	This shows whether the firmware on the Nebula Device is Up to date , there is firmware update available for the Nebula Device (Upgrade available), or a specific version of firmware has been installed by Zyxel customer support (Locked).
Ð	Click this icon to show and hide columns in the table.

 Table 175
 Organization-wide > Organization-wide manage > Firmware management > Overview

11.4.6.4 Firmware Management Devices Screen

Use this screen to make different firmware upgrade schedules for the Nebula Devices in the organization. Click **Organization-wide > Organization-wide manage > Firmware management > Devices** to access this screen.

Note: While installing a firmware update, the Nebula Device will continue to operate normally until it reboots. The reboot will take 3 to 5 minutes, so it is best to pick an upgrade time that has minimal impact on your network.

Figure 209 Organization-wide > Organization-wide manage > Firmware management > Devices

Overview	Dev	ices									
ite	Status	Device type	e Tag	Model		Current version	Firmware status	Firmware t	ype Av	vailability	Locked
Any	✓ Any	▼ Any	▼ Any	▼ Any		Any	Any	 Any 	•	Any 👻	Any
Status	Device type	Model	MAC address	s/N	Site	Current version	Firmware status	Availability	Firmware	e type Upgrad	le scheduled
Status	Device type Firewall	Model USG FLEX 500	MAC address	S/N 91631 45300133	Site ZyNet TW	Current version	Firmware status N/A ()	Availability N/A	Firmware N/A	e type Upgrad No	le scheduled
Status	Device type Firewall Access point	Model USG FLEX 500 NWA110AX	MAC address DO-CC-45-E9-70-00	S/N 01601 45000100 00001 450400004	Site ZyNet TW ZyNet TW	Current version N/A N/A	Firmware status N/A 1 N/A 1	Availability N/A N/A	Firmware N/A N/A	e type Upgrad No No	le scheduled
Status Status	Device type Firewall Access point Mobile router	Model USG FLEX 500 NWA110AX NR7101	MAC address DOEC AD 1979 EA DOEC AD 2970 CAS DOEC AD 2970 CAS	S/N 01601 45000100 00001 4504000.4 0010745007757	Site ZyNet TW ZyNet TW ZyNet TW	Current version N/A N/A N/A	Firmware status N/A (1) N/A (1) N/A (1)	Availability N/A N/A N/A	Firmware N/A N/A N/A	e type Upgrad No No No	de scheduled
Status O	Device type Firewall Access point Mobile router Switch	Model USG FLEX 500 NWA110AX NR7101 GS2220-10HP	MAC address B0-EC A010-70-EA B0-CE 4E E0 70-00 N0-EC-E0 70-00-58 30-22-00-03-20-01	S/N C1631 45300193 C3031 45340304 C3107 45007757 203200033001	Site ZyNet TW ZyNet TW ZyNet TW ZyNet TW	Current version N/A N/A N/A N/A	Firmware status N/A 1 N/A 1 N/A 1 N/A 1	Availability N/A N/A N/A N/A	Firmward N/A N/A N/A N/A	e type Upgrad No No No No	le scheduled
Status •	Device type Firewall Access point Mobile router Switch Access point	Model USG FLEX 500 NWA110AX NR7101 GS2220-10HP WAX650S	MAC address DOEC A010 TOEA DOEC A010 TOEA DOEC ES TO DOE 10 77-00 09 70-01 20 71-07-00 190	S/N 01601 45000100 00001 45040004 0010745007757 00000000001 000107000100	Site ZyNet TW ZyNet TW ZyNet TW ZyNet TW	Current version N/A N/A N/A N/A N/A	Firmware status N/A 1 N/A 1 N/A 1 N/A 1 N/A 1	Availability N/A N/A N/A N/A N/A	Firmward N/A N/A N/A N/A	e type Upgrad No No No No No	le scheduled
Status Status O	Device type Firewall Access point Mobile router Switch Access point Access point	Model USG FLEX 500 NWA110AX NR7101 GS2220-10HP WAX650S WAX650S	MAC address B0:E0.404037044 B0.0040E037040 N0:E0.004060 20.00406030410 20.01407000120 20.01407030120	S/N C1621 AE300132 C3001 AE300132 C3001 AE300132 C3001 AE300132 C30010 C30010 C3001 AE300132 C30010 C30010 C3001 C30010 C30010 C30010 C30010 C30010 C30010 C30010 C30010 C30010 C30010 C30010	Site ZyNet TW ZyNet TW ZyNet TW ZyNet TW ZyNet TW	Current version N/A N/A N/A N/A N/A	Firmware status N/A N/A N/A N/A N/A N/A N/A N/A	Availability N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A	Firmward N/A N/A N/A N/A N/A	etype Upgrad No No No No No No No	ie scheduled

Table 176 Organization-wide > Organization-wide manage > Firmware management > Devices

LABEL	DESCRIPTION
Site/Status/Device type/Tag/Model/ Current version/ Firmware status/ Firmware type/ Availability/Locked	Specify your desired filter criteria to filter the list of Nebula Devices.
Upgrade Now	Click this to immediately install the firmware on the selected Nebula Devices. This button is selectable only when there is firmware update available for the selected Nebula Devices.

LABEL	DESCRIPTION				
Schedule upgrade	Click this to pop up a window where you can create a new schedule for the selected Nebula Devices.				
	You can select to upgrade firmware according to the organization-wide schedule configured for the Nebula Device type in the site, create a recurring schedule, edit the schedule with a specific date and time when firmware update is available for all the selected Nebula Devices, or immediately install the firmware. With a recurring schedule, the NCC will check and perform a firmware update when a new				
	firmware release is available for any of the selected Nebula Devices. If the NCC service is downgraded from Nebula Professional Pack to Nebula Base, the Nebula Devices automatically changes to adhere to the organization-wide schedule.				
	Schedule upgrade X				
	Note: Schedule upgrade will follow each site's time zone.				
	Upgrade policy O Follow device type settings What is this?				
	Auto upgrade at every Week ▼ on Monday ▼ at 02.00 ▼				
	Upgrade at 2023-03-29 🛅 14:30 🔻				
	Upgrade now Device(s) below will be upgraded as required time.				
	Device type Model MAC address S/N Current version Schedule upgrade version				
	Firewall USG FLEX 500 00-50-2012/01-54 61621 45000120 N/A N/A				
	Access point NWATIOAX BPOPE AC EXPERIMENTATION N/A N/A N/A Access point WAX6505 20/21/07/03/01/28 20/21/07/03/01/28 N/A N/A				
	Access point WAX650S 20:21:07:03:0129 202107030129 N/A N/A				
Reset	Select one or more Nebula Devices, and then click Reset to allow the Nebula Devices to follow the site-wide firmware management settings.				
Status	This shows the status of the Nebula Device.				
	Green: The Nebula Device is online and has no alerts.				
	Amber: The Nebula Device has alerts.				
	 Red: The Nebula Device is offline. Grav: The Nebula Device has been offline for 7 days or more. 				
Device type	This shows the type of the Nebula Device.				
Model	This shows the model number of the Nebula Device.				
Тад	This shows the tag created and added to the Nebula Device				
Name	This shows the descriptive name of the Nebula Device.				
MAC address	This shows the MAC address of the Nebula Device.				
S/N	This shows the serial number of the Nebula Device.				
Site	This shows the descriptive name of the site.				
Current version	This shows the version number of the firmware the Nebula Device is currently running. It shows N/A when the Nebula Device goes offline and its firmware version is not available.				

 Table 176
 Organization-wide > Organization-wide manage > Firmware management > Devices

LABEL	DESCRIPTION
Firmware status	The status shows Good if the Nebula Device is running a stable firmware and no immediate action is required. See the description of a stable firmware on the next field Firmware type .
	The status shows Warning if a newer firmware is available and immediate action is recommended. The newer firmware may contain security enhancements, new features, and performance improvements.
	The status shows Critical if a newer firmware is available and immediate action is required. The firmware may have security vulnerabilities and/or lack key performance improvements.
	The status shows Custom if the Nebula Device is running a firmware with specialized features that is not available to the general public.
	The status changes to Upgrading after you click Upgrade Now to install the firmware immediately.
Firmware type	This shows Stable when the installed firmware may not have the latest features but has passed Zyxel internal and external testing.
	This shows Latest when the installed firmware is the most recent release with the latest features, improvements, and bug fixes.
	This shows General Availability when the installed firmware is a release before Latest , but is still undergoing Zyxel external testing.
	This shows Dedicated when the installed firmware is locked and Zyxel support is monitoring. Contact Zyxel customer support if you want to unlock the firmware in order to upgrade to a later one.
	This shows Beta when the installed firmware is a release version for testing the latest features and is still undergoing Zyxel internal and external testing.
	This shows N/A when the Nebula Device is offline and its firmware status is not available.
	Note: See Table 177 on page 541 for an example Firmware type version progression example scenario.
Availability	This shows whether the firmware on the Nebula Device is Up to date , there is firmware update available for the Nebula Device (Upgrade available), or a specific version of firmware has been installed by Zyxel customer support (Locked).
Upgrade scheduled	This shows the date and time when a new firmware upgrade is scheduled to occur. Otherwise, it shows Follow upgrade time and the Nebula Device sticks to the site-wide schedule or No when the firmware on the Nebula Device is up-to-date or the Nebula Device goes offline and its firmware status is not available.
	A lock icon displays if a specific schedule is created for the Nebula Device, which means the Nebula Device firmware will not be upgraded according to the schedule configured for all Nebula Devices in the site.
Last upgrade time	This shows the last date and time the firmware was upgraded on the Nebula Device.
Schedule upgrade version	This shows the version number of the firmware which is scheduled to be installed.
R	Click this icon to display a greater or lesser number of configuration fields.

Table 176	Organization-wide	Organization-wide	manaae > Firmware	management > Devices
	organization mao	organization mad	indiago - rinniaio	management benees
Firmware Type / Version Progression

The following table shows an example firmware version progression scenario.

VERSION NUMBER TIMELINE	FIRMWARE TYPE	VERSION NUMBER TIMELINE	FIRMWARE TYPE
V6	Latest	V5	General Availability
V7	Latest	V6	General Availability

Table 177	Firmware Ty	pe Version	Progression	Example

Note: Zyxel will select a previous version, (for example, V3) as a **Stable** release if no major issues have been reported by users.

There can only be one Latest and one Stable firmware.

11.4.7 Cloud Authentication

Use this screen to view and manage the user accounts which are authenticated using the NCC user database, rather than an external RADIUS server. Click **Organization-wide > Organization-wide manage > Cloud authentication** to access this screen.

Note: The changes you made in this screen apply to all sites in the organization. To change the cloud authentication settings for a specific site, go to **Site-wide** > **Configure** > **Cloud authentication** (see Section 4.9 on page 207).

11.4.7.1 User Account Types

NCC has the following types of user accounts. For details on using these accounts for WiFi and network authentication, see Section 5.3.1 on page 247.

ACCOUNT TYPE	DESCRIPTION	AUTHENTICATION METHODS
User	The user account can gain access to the networks by authenticating using a pre-created user name and password, or their email address. This type of user account also supports DPPSK and two-factor authentication.	 WiFi authentication (WPA-Enterprise) Network access through captive portal VPN Access WiFi authentication + network authentication through DPPSK
MAC	The Nebula Device account that can gain access to the networks by authenticating using its MAC address.	 MAC-based Nebula Device authentication (combined with DPPSK)
DPPSK	A user that can gain access to the network using a unique dynamic Personal Pre-Shared key that is linked to their user account.	WiFi authentication + network authentication through DPPSK

Table 178 Cloud Authentication: User Account Types

11.4.7.2 Cloud Authentication User Screen

Use this screen to view and manage regular NCC network user accounts. Click **Organization-wide** > **Organization-wide manage** > **Cloud authentication** > **User** to access this screen.

Figure 210 Organization-wide > Organization-wide manage > Cloud authentication > User

Jser	MAC [PPSK													
Auth	norization -	Remove us	ers VPN a	ccess *	VLAN attribute -	Q Sec	arch users selected in (2) Users	•				ि Im	iport + A	dd [⇒Export+
	Email	Userna	Descri	802.1X	VPN ac	Authori	Expire i	Login by	DPPSK 🗞	VLAN a	2FA St	Bypass	Authori	Create	Created at (📃
	Email abc@c.c	Userna abcde	Descri 12345678	802.1X Yes	VPN ac No	Authori All sites	Expire i Never	Login by Usernam	DPPSK 🗞	VLAN a	2FA St Not Enroll	Bypass No	Authori yawen.lin	Create yawen.lin	Created at (📃 2022-05-05 09:2

The following table describes the labels in this screen.

Note: Some of the actions on this screen are only available if your administrator account has full access to the organization.

 Table 179
 Organization-wide > Organization-wide manage > Cloud authentication > User

LABEL	DESCRIPTION			
Authorization	Select one or more than one user account and click this button to configure the authorization settings for the selected user accounts.			
	O Does not expire			
	O Expires in: X minutes 💌			
	Revoke authorization (Not authorized)			
	Update			
Remove users	Select one or more than one user account and click this button to remove the selected user accounts.			
VPN access	Select one or more than one user account and click this button to configure whether the accounts can be used to connect to the organization's networks through VPN.			
VLAN attribute	Select one or more than one user account and click this button to assign the users to a specific VLAN ID, or clear the VLAN ID. Then click Update .			
Print	Click this button to print information about each selected user account, such as their user name and password.			
Search users	Enter a key word as the filter criteria to filter the list of user accounts.			
N User	This shows how many user accounts (N) match the filter criteria and how many user accounts of the selected type are created in total.			

LABEL	DESCRIPTION				
Import	Click this button to create user accounts in bulk by importing a complete list of all new users in an Excel file.				
	Bulk Import X				
	"Bulk Import" supports for faster inputting. Please follow this template to import				
	Browse				
	Or drag file here				
	Close				
Add	Click this button to create a new user account. See Section 11.4.7.3 on page 544.				
Export	Click this button to save the account list as a CSV or XML file to your computer.				
Email	This shows the email address of the user account.				
Username	This shows the user name of the user account.				
Description	This shows the descriptive name of the user account.				
802.1X	This shows whether 802.1X (WPA-Enterprise) authentication is enabled on the account.				
VPN access	This shows whether the accounts can be used to connect to the organization's networks through VPN.				
Authorized	This shows whether the user has been authorized or not (No). If the user is authorized, it shows All sites or the name of the site to which the user is allowed access.				
Expire in (UTC)	This shows the date and time that the account expires.				
	This shows if authentication is disabled for this account.				
	This shows Never if the account never expires.				
	This shows Multiple value if the account has different Expire in values across different sites.				
Login by	This shows whether the user needs to log in with the email address and/or user name.				
DPPSK	This shows the account's dynamic personal pre-shared key (DPPSK), if one is set.				
VLAN assignment	This field is available only when the account type is set to User .				
	This shows the VLAN assigned to the user.				
2FA Status	This shows whether the account has set up two-factor authentication yet.				
Bypass 2FA	This shows whether the account is allowed to bypass two-factor authentication, if two-factor authentication is enabled on a captive portal or VPN gateway.				
Authorized by	This shows the email address of the administrator account that authorized the user.				
	If the account has been authorized by different admins across different sites, it shows Multiple value.				
Created by	This shows the email address of the administrator account that created the user.				
Created at	This shows the date and time that the account was created.				
	Click this icon to display a greater or lesser number of configuration fields.				

Table 179	Organization-wide >	Organization-wide	manage > Cloud	d authentication >	· User (continued)
	0	0	0		1 /

11.4.7.3 Create/Update User Account

In the Site-wide or Organization-wide > Organization-wide manage > Cloud authentication > User screen, click the Add button to create a new user account or double-click an existing account entry to modify the account settings.

Figure 211	Organization-wide > Organization-wide manage > Cloud authentication > User: Create/
Update	e User

Create user		×
Account type:	USER	
Email:	test@zyxel.com.tw \times *	
Username:	X	
Description:	X	
Password:	IAN6xmw1 C Generate	
VPPSK:	Generate	
802.1X:	Allow to use WPA-Enterprise to access network	
VPN Access:	Allow to use Remote VPN access	
Authorized:	Not authorized 🔹	
Login by:	Email	
💎 VLAN assignment: Beta	X	
Two-Factor Auth.:	Bypass two-factor authentication.	
Email to user:	Email account information to user.	
	Close Print Creat	e user

The following table describes the labels in this screen.

Table 180 Organization-wide > Organization-wide manage > Cloud authentication > User: Create/ Update user

LABEL	DESCRIPTION
Account type	This shows the type of the user account.
Email	Enter the email address of the user account, which is used to log into the networks.
Username	Enter a user name for this account.
	Note: This field is optional if Login by is set to Email.
Description	Enter a descriptive name for the account.
Password	Enter the password of this user account. It can consist of 4 – 31 alphanumeric characters.
	You can click Generate to have NCC create a password for the account automatically.

LABEL	DESCRIPTION
DPPSK	Enter a dynamic personal pre-shared key (DPPSK) for this DPPSK user account, if you want to be able to authenticate using DPPSK in addition to a user name and password. It can consist of 8 – 31 alphanumeric characters.
	You can click Generate to have the NCC create a DPPSK for the account automatically.
802.1X	Select this to allow the account to be used for single sign-on (SSO) network and WiFi authentication using 802.1X (WPA-Enterprise).
VPN Access	Select this to allow the account to be used to connect to the organization's networks through VPN.
Authorized	Set whether you want to authorize the user of this account.
	You can select to authorize the user's access to All Sites or Specified Sites in the organization. If you select Specified Sites , a field displays allowing you to specify the sites to which the user access is authorized.
Expire in	This field is available only when the user is authorized.
	Click Change to specify the number of minutes/hours/days/weeks the user can be logged into the network in one session before the user of this account has to log in again.
	Note: If the account has been set with different Expire in values across different sites, it will show Multiple value and the Change link.
	Otherwise, select Never and the user of this account will never be logged out.
Login by	Select whether the user needs to log in with the email address and/or user name.
VLAN assignment	This allows you to assign a user to a specific VLAN based on the user credentials instead of using a RADIUS server.
Bypass two- factor authentication	This shows whether the account is allowed to bypass two-factor authentication, if two-factor authentication is enabled on a captive portal or VPN gateway.
Email account information to user	Select this to send a copy of the information on this screen to the account email address, after the account has been created.
Close	Click this button to exit this screen without saving.
Print	Click this button to print the account information.
Create user	Click this button to save your changes and close the screen.

Table 180	Organization-wide > Organization-wide manage	> Cloud authentication > User: Create/
Update us	er (continued)	

11.4.7.4 Cloud Authentication MAC Screen

Use this screen to view and manage NCC Nebula Device user accounts, used for MAC-based authorization. Click **Organization-wide > Organization-wide manage > Cloud authentication > MAC** to access this screen.

ud a	uthentication						
er	MAC DPPSK						
Auth	orization - Remove u	sers	Q Search us	ers			🕒 Import 🛛 + Add 🕒 Exp
				a in 4 Users			
	MAC address	Description	Account type	Authorized	Authorized by	Expire in (UTC)	Created at (UTC)
	MAC address	Description	Account type MAC	Authorized All sites	Authorized by y@zyxel.com.tw	Expire in (UTC) Never	Created at (UTC) 2021-10-21 05:09:01
	MAC address 11:11:11:11:11:11 22:22:22:22:22:22:22	Description 11 22	Account type MAC MAC	Authorized All sites All sites	Authorized by y@zyxel.com.tw y@zyxel.com.tw	Expire in (UTC) Never	Created at (UTC) 2021-10-21 05:09:01 2022-03-31 15:52:09
	MAC address 11:11:11:11:11:11 22:22:22:22:22:22 40:CR:CD14:6EFE	Description 11 22 IPHONE135	Account type MAC MAC MAC	Authorized All sites All sites All sites	Authorized by y©zyxel.com.tw y©zyxel.com.tw y©zyxel.com.tw	Expire in (UTC) Never Never	Created at (UTC) 2021-10-21 05:09:01 2022-03-31 15:52:09 2022-04-07 08:28:08

Figure 212 Organization-wide > Organization-wide manage > Cloud authentication > MAC

Note: Some of the actions on this screen are only available if your administrator account has full access to the organization.

Table 181 Organization-wide > Organization-wide manage > Cloud authentication > MAC

LABEL	DESCRIPTION
Authorization	Select one or more than one account and click this button to configure the authorization settings for the selected user accounts.
Remove users	Select one or more than one user account and click this button to remove the selected user
	accounts.
Search users	Enter a key word as the filter criteria to filter the list of user accounts.
N User	This shows how many user accounts (N) match the filter criteria and how many user accounts of the selected type are created in total.
Import	Click this button to create user accounts in bulk by importing a complete list of all new users in an Excel file. Bulk Import "Bulk Import" supports for faster inputting. Please follow this template to import Browse Or drag file here Close
Add	Click this button to create a new user account. See Section 11.4.7.5 on page 547.
Export	Click this button to save the account list as a CSV or XML file to your computer.
Email	This shows the email address of the user account.

NCC User's Guide

LABEL	DESCRIPTION
MAC address	This shows the MAC address of the user account.
Description	This shows the descriptive name of the user account.
Account type	This shows the type of user account: USER, MAC, or DPPSK.
Authorized	This shows whether the user has been authorized or not (No). If the user is authorized, it shows All sites or the name of the site to which the user is allowed access.
Authorized by	This shows the email address of the administrator account that authorized the user.
	If the account has been authorized by different admins across different sites, it shows Multiple value.
Expire in (UTC)	This shows the date and time that the account expires.
	This shows if authentication is disabled for this account.
	This shows Never if the account never expires.
	This shows Multiple value if the account has different Expire in values across different sites.
Created at	This shows the date and time that the account was created.
	Click this icon to display a greater or lesser number of configuration fields.

Table 181 Organization-wide > Organization-wide manage > Cloud authentication > MAC (continued)

11.4.7.5 Create/Update MAC Account

In the Site-wide or Organization-wide > Organization-wide manage > Cloud authentication > MAC screen, click the Add button to create a new user account or double-click an existing account entry to modify the account settings.

Figure 213 Organization-wide > Organization-wide manage > Cloud authentication > MAC: Create/ Update user

Create user		×
Account type:	MAC	
Description:	×	
MAC address:	20:22:09:03:20:01 × *	
Authorized:	All sites 👻	
Expires:	Does not expire	
	O Expires in: 60 X minutes ▼	
	Close Print Cre	ate user

Table 182	Organization-wide > Organization-wide manage > Cloud authentication > MAC: Create/
Update us	er

LABEL	DESCRIPTION
Account type	This shows the type of the user account.
Description	Enter a descriptive name for the account.
MAC address	Enter a MAC address for this account.
Authorized	Set whether you want to allow the user of this account access to sites.
	Select All Sites or Specified sites to allow the user access to all or some sites in the organization. If you select Specified sites , a field displays allowing you to specify the sites to which the user access is authorized.
	Select Not authorized to prevent the user access to all the sites in the organization.
Expires	Specify the number of minutes/hours/days/weeks the user has access to site(s) in the organization.
Close	Click this button to exit this screen without saving.
Print	Click this button to print the account information.
Create user	Click this button to save your changes and close the screen.

11.4.7.6 Cloud Authentication DPPSK Screen

Use this screen to view and manage DPPSK network user accounts. Click **Organization-wide** > **Organization-wide manage** > **Cloud authentication** > **DPPSK** to access this screen.

	o · · · · · ·				
Figure 214	()raanization-wide	> ()raanization-wide	e manade >	Cloud authentic	ation > DPPSK
	organization mao	erganizanon mao	manage		Janon' Dirok

thoriz	ation - Remo	ve users Print		Q Search users	•			🕒 Imp	ort 🕂 Add 🕒 Exp
Em	ail	Username	Account type		VLAN ID	Authorized	Expire in (UTC)	Created by	Created at (UTC)
b1@	9bb.com2000	bb.2000	DPPSK	********		No		Y@zyxel.c	2022-02-11 06:14:03
] b1@	000 000 000 000 000 000 000 000 000 00	bb.2009	DPPSK	********		No		y@zyxel.c	2022-02-11 06:14:03
]			DPPSK	*********	100	No		Y@zyxel.c	2022-04-07 07:43:1
]			DPPSK	*****	100	No		Y@zyxel.c	2022-04-07 07:43:1
]			DPPSK	*********	100	No		Y@zyxel.c	2022-04-07 07:43:1
j			DPPSK	********	100	No		y@zyxel.c	2022-04-07 07:43:1
]			DPPSK	*********	100	No		Y@zyxel.c	2022-04-07 07:43:1
] bbl	b123@c.c	bbb123	DPPSK	********		No		Y@zyxel.c	2022-04-08 01:24:3
]			DPPSK	*********		No		y@zyxel.c	2021-10-06 05:48:1
] b1@	9bb.com1	bb1	DPPSK	******		No		y@zyxel.c	2022-02-11 06:13:37

Table 183	Organization-wide	> Oragnization-wide	e manaae >	Cloud authentication >	> DPPSK
	organization mac	r organization mad	managoz	Cloud domicritication -	DITUK

LABEL	DESCRIPTION						
Authorization	Select one or more than one user account and click this button to configure the authorization settings for the selected user accounts.						
	• Authorize users (All sites)						
	O Does not expire						
	O Expires in: X minutes V						
	Revoke authorization (Not authorized)						
	Update						
Remove users	Select one or more than one user account and click this button to remove the selected user accounts.						
Print	Click this button to print the unique dynamic personal pre-shared key (DPPSK) and expiry time of each selected user account.						
	The account details can be cut into cards, and then given to users in order to grant them WiFi network access.						
	DPPSK						
	Expired in: Never Never						
Searchusers	Enter a key word as the filter criteria to filter the list of user accounts						
N Users	This shows how many user accounts (N) match the filter criteria and how many user accounts of the selected type are created in total.						
Import	Click this button to create user accounts in bulk by importing a complete list of all new users in an Excel file.						
	Bulk Import X						
	"Bulk Import" supports for faster inputting. Please follow this template to import						
	Browse						
	Or drag file here						
	Close						
Add	Click this button to create a single new account, or a batch of accounts.						
	 Single DPPSK: See Section 11.4.7.7 on page 550. Batch create DPPSK: See Section 11.4.7.8 on page 551. 						
Export	Click this button to save the account list as a CSV or XML file to your computer.						
Email	This shows the email address of the user account.						
Username	This shows the user name of the user account.						

NCC User's Guide

LABEL	DESCRIPTION
Account type	This shows the type of user account: USER, MAC, or DPPSK.
DPPSK	This shows the account's dynamic personal pre-shared key (DPPSK).
VLAN ID	This shows the VLAN assigned to the account.
Description	This shows the descriptive name of the user account.
Authorized	This shows whether the user has been authorized or not (No). If the user is authorized, it shows All sites or the name of the site to which the user is allowed access.
Expire in (UTC)	This shows the date and time that the account expires.
	This shows if authentication is disabled for this account.
	This shows Never if the account never expires.
	This shows Multiple value if the account has different Expire in values across different sites.
Created by	This shows the email address of the administrator account that created the user.
Created at	This shows the date and time that the account was created.
	Click this icon to display a greater or lesser number of configuration fields.

Table 183 Organization-wide > Organization-wide manage > Cloud authentication > DPPSK

11.4.7.7 Add/Edit DPPSK Account

In the Site-wide or Organization-wide > Organization-wide manage > Cloud authentication > DPPSK screen, click Add > Single DPPSK to create a new user account or double-click an existing account entry to modify the account settings.

Figure 215 Organization-wide > Organization-wide manage > Cloud authentication > DPPSK: Create/ Update DPPSK user

Create single DPPSK user					
Account type:	DPPSK				
Email:	test2@zyxel.com.tw	*			
Username:		×			
DPPSK:	********	Generate			
VLAN id:		×			
Authorized:	All sites	•			
Expire in:	Never Change				
Email to user:	Email account infor	mation to user.			
		Close Print Cred	ate user		

Table 184	Organization-wide > Organization-wide manage > Cloud authentication > DPPSK: Create/
Update DF	PPSK user

LABEL	DESCRIPTION
Account type	This shows the type of the user account.
Email	Enter the email address of the user account, which is used to log into the networks.
Username	Enter a user name for this account.
Description	Enter a descriptive name for the account.
DPPSK	Enter a dynamic personal pre-shared key (DPPSK) for this DPPSK user account. It can consist of 8 – 31 alphanumeric characters.
	You can click Generate to have the NCC create a DPPSK for the account automatically.
VLAN id	Enter the ID of a VLAN to assign a user to a specific VLAN.
Authorized	Set whether you want to authorize the user of this account.
	You can select to authorize the user's access to All Sites or Specified Sites in the organization. If you select Specified Sites , a field displays allowing you to specify the sites to which the user access is authorized.
Expire in	This field is available only when the user is authorized.
	Click Change to specify the number of minutes/hours/days/weeks the user can be logged into the network in one session before the user of this account has to log in again.
	Note: If the account has been set with different Expire in values across different sites, it will show Multiple value and the Change link.
	Otherwise, select Never and the user of this account will never be logged out.
Email account information to user	Select this to send a copy of the information on this screen to the account email address, after the account has been created.
Close	Click this button to exit this screen without saving.
Print	Click this button to print the account information.
Create user	Click this button to save your changes and close the screen.

11.4.7.8 Batch Create DPPSK Accounts

To have NCC create multiple DPPSK user accounts, each with a unique dynamic personal pre-shared key (DPPSK), go to the **Site-wide** or **Organization-wide** > **Organization-wide** manage > **Cloud authentication** > **DPPSK** screen, click **Add**, and then select **Batch Create DPPSK**.

Figure 216 Organization-wide > Organization-wide manage > Cloud authentication: Batch Create DPPSK

Batch create DPPSK user				
Account type:	DPPSK			
Number of accounts:	20	× * (1~20)		
VLAN id:		×		
E-mail account info to:		×		
Authorized:	All sites	•		
Expire in:	Never <u>Change</u>			
		Close Create user		

The following table describes the labels in this screen.

LABEL	DESCRIPTION
Number of accounts	Enter how many DPPSK user accounts you want to create.
VLAN id	Assign the users to a specific VLAN based on the user's dynamic personal pre-shared key (DPPSK).
E-mail account info to	Send a copy of each user account's dynamic personal pre-shared key (DPPSK) and expiry date to the specified email address. This information is in a printable format.
	The expiry date includes a time and date in UTC format.
Authorized	Set whether you want to authorize the user of this account.
	You can select to authorize the user's access to All Sites or Specified Sites in the organization. If you select Specified Sites , a field displays allowing you to specify the sites to which the user access is authorized.
Expire in	This field is available only when the user is authorized.
	Click Change to specify the number of minutes/hours/days/weeks the user can be logged into the network in one session before the user of this account has to log in again.
	Note: If the account has been set with different Expire in values across different sites, it will show Multiple value and the Change link.
	Otherwise, select Never and the user of this account will never be logged out.
Close	Click this button to exit this screen without saving.
Create user	Click this button to save your changes and close the screen.

Table 185 Organization-wide > Organization-wide manage > Cloud authentication: Batch Create DPPSK

11.4.8 Change Log

Use this screen to view logged messages for changes in the specified organization. Click **Organization**wide > **Organization-wide manage** > **Change log** to access this screen.

When the log is full, it deletes older entries one by one to make room for new ones.

Figure 217 Organization-wide > Organization-wide manage > Change log

२ Search	· · ·	From:			To:					O Segral
	Range	2019-10-18		05:57 👻	2019-10-25	Ē	05:57 👻	UTC+0		Secilo
		Max range is a	o days, the dates will be	auto-aajuste	u.				_	
Newer Older >	211 change logs within	the time filtere	ed. Changes date l	back to 201	7-09-14 02:5	3 (UTC)				Export
lime (UTC)	Site time	A	dmin		Site	SSID	Page	Label	Old	N
2019-10-25 05:25:28	2019-10-25 13:25:28	(UTC +8.0) N			<u>Site25 N</u>		Capti	ADD:		Site25
2019-10-25 05:25:28	2019-10-25 13:25:28	(UTC +8.0) N			<u>Site25_N</u>		Capti	ADD:		5a7d51
2019-10-25 05:25:28	2019-10-25 13:25:28	(UTC +8.0) N			<u>Site25_N</u>		Authe	CHAN	SNS	CLICK
2019-10-25 02:51:40	2019-10-25 10:51:40	(UTC +8.0) N	1CC_		<u>Site30</u>		Firew	CHAN	60.24	60.248
2019-10-25 02:51:40	2019-10-25 10:51:40	(UTC +8.0) N	1CC_1000_00000		<u>Site30</u>		Firew	REMO	HUB	
2019-10-25 02:51:40	2019-10-25 10:51:40	(UTC +8.0) N	1CC_		<u>Site30</u>		Firew	REMO	WAN1	
2019-10-25 02:51:40	2019-10-25 10:51:40	(UTC +8.0) N	1CC_1000		<u>Site30</u>		Firew	REMO	false	
2019-10-25 02:51:40	2019-10-25 10:51:40	(UTC +8.0) N	1CC_		<u>Site30</u>		Firew	REMO	false	
2019-10-25 02:51:40	2019-10-25 10:51:40	(UTC +8.0) N	1CC_		<u>Site30</u>		Firew	REMO	86400	
2019-10-25 02:51:40	2019-10-25 10:51:40	(UTC +8.0) N	1CC_		<u>Site30_</u>		Firew	REMO	NONE	
						a 1 af 22		Regulta		10

The following table describes the labels in this screen.

LABEL	DESCRIPTION
Search	Click to enter one or more key words as the search criteria to filter the list of logs.
Range/Before	Select Range to set a time range or select Before to choose a specific date/time and the number of hours/minutes to display only the log messages generated within a certain period of time (before the specified date/time). The maximum allowable time range is 30 days.
Search	Click this to update the list of logs based on the search criteria.
Reset filters 🖂	Click this to return the search criteria to the previously saved time setting.
Newer/Older	Click to view a list of log messages with the most recent or oldest message displayed first.
	This shows the total number of the log messages that match the search criteria. It also shows the date and time the very first log was created.
Export	Click this button to save the log list as a CSV or XML file to your computer.
Time (UTC)	This shows the date and time in UTC+00:00 (or UTC+0) when the log was recorded.
	UTC is a standard time for use around the world (formerly known as Greenwich Mean Time or GMT). UTC is an international abbreviation that is neither French nor English. It means both "Temps Universel Coordonné" and "Coordinated Universal Time".
Site Time	This shows the date and time of the site, to which the change was applied, when the log was recorded.
Admin	This shows the name of the administrator who made the changes.
Site	This shows the name of the site to which the change was applied.

Table 186 Organization-wide > Organization-wide manage > Change log

NCC User's Guide

LABEL	DESCRIPTION
SSID	This shows the SSID name to which the change was applied.
Page	This shows the name of the NCC menu in which the change was made.
Label	This shows the reason for the log.
Old value	This shows the old setting that was discarded and overwritten with the new attribute value.
New value	This shows the new setting that was adopted.
Ð	Click this icon to display a greater or lesser number of configuration fields.

Table 186 Organization-wide > Organization-wide manage > Change log (continued)

11.4.9 Organization Settings

Use this screen to change your general organization settings, such as the organization name and security. Click **Organization-wide > Organization-wide manage > Organization settings** to access this screen.

Figure 218 Organization-wide > Organization-wide manage > Organization settings

5	
Organization information	
Name:	SVD ×
Country:	Taiwan 👻
Security	
Idle Timeout 1	0 x minutes of inactivity will logout users.
ኛ Login IP ranges	Only allow access to this organization from IP addresses in the specified ranges.
	This computer is using IP address : 61.222.86.26
	v [*]
	What do Lenter here?
	Acceptable IP ranges: A single IP address (e.g. 61.222.86.26) A CIDR subnet (e.g. 61.222.86.26/32)
Import certificate	Use my certificate
	Name: × (64 letters)
	File Path: Upload a PKCS#12 file that bundles a private key with its X.509 certificate.
	Password: (PKCS#12 only)
Override device ownership	Prevent other users take my ownership of this organization's device(s) from Nebula App.
Delete this organization	You can delete this organization only if it has no sites, administrators, users, licenses, or devices registered in this inventory.
	Please check your setting as below: <u>sites</u> , <u>administrators</u> , <u>users</u> , <u>licenses/devices</u> of devices.

NCC User's Guide

Table 187	Oragnization-wide	> Oraanization-v	vide manaae > (Draanization settinas

LABEL	DESCRIPTION
Name	Enter a descriptive name for the organization.
Country	Select the country where the organization is located.
	Note: This field is only for reference. It does not affect any other fields or features in NCC.
Security	
Idle timeout	Select ON and enter the number of minutes each user can be logged in and idle before the NCC automatically logs out the user.
	Select OFF if you do not want the NCC to log out idle users.
Login IP ranges	Select ON and specify the IP address range of the computers from which an administrator is allowed to log into the NCC.
	Select OFF to allow any IP address of the computer from which an administrator can log into the NCC.
Import certificate	
Use my certificate	Select ON to import a certificate that can be used by connected Nebula Access Points in WPA2 authentication.
Name	Enter a name for the certificate (up to 64 letters).
File Path	Click to find the certificate file you want to upload.
Import	Click this button to save a new certificate to the NCC.
Password	Enter the certificate file's password.
Override device ownership	By default, your Nebula Device can transfer to another administrator's organization by using the Nebula Mobile app to scan the QR code. Click this switch to the right to prohibit Nebula Device transfer between administrators.
Delete this organization	Click the Delete organization button to remove the organization when it does not have any sites, Nebula Devices or users.
	Note: You will be redirected to the Choose organization page after this organization is deleted.

PART IV Manage by Group Deployment

CHAPTER 12 Group-wide

12.1 Introduction

This chapter discusses the menus that you can use to monitor and manage your groups settings.

A group is a collection of two or more organizations. Groups allow you to view and manage multiple organizations, and create VPN links between groups in the organization.

12.1.1 Creating a Group

Follow the steps below to create a group.

- 1 Ensure that you are the owner of two or more Pro Pack organizations that are not currently in a group.
- 2 Click the Organization list, and then select Create Group.

nebula	Group: List All Groups	▼ Organization	n: Nebula_Org Site: Site:	01
Control Center			۹.	
	Site-wide > <u>Dashboard</u> Dashboard		MSP portal + Create group + Create organization	Prof
	AP Status	<u> Wireless</u> C	PRO 11 point than have no	
	No APs	No A	TRL Nebula_Org	•

3 In the **Create group** window, enter a group name and then select two or more organizations to add to the group. You must be the group owner, and each group must have a Pro Pack license. Then click **OK**.

Create group		×
Group name	Test Group	*
Group member	test 😵 TestOrg2 😵	
Note: You could selec	et organizations own by you to join G	roup.
	Cancel	ок

12.1.2 Group-Wide Menu

The **Group-wide** menu and the **Group** list appear when you create at least one group. You can select a group to manage by selecting it in the **Group** list.

Figure 219	Group-wide > Group-wide manage > Overview: Group	
		_

13/20	nebula	Group	TW Test-29	Organization:	TW Sep-2	-	\bigcirc
200	Control Center	oroup	Q	ite:	Overview •	-	Ч
			List All Groups TW Test		You have 24 day(s) left of yo	our Nebula Professional Pack trial. (<u>More Information</u>)	
		Ov	TW Test-29)			

12.2 Group Portal

The overview screen allows you to view the status of organizations in a group. Click **Group-wide** > **Group-wide manage** > **Group portal** to access this screen.

Figure 220 Group-wide > Group-wide manage > Group portal

Overview					
Q Search		 ▼ ③ Organizations 			
Status 👔	Organization	Туре	NCC license status	Payment mode	NCC license expiration (UTC) 🚍
0	Nebula_Org2	Nebula Professional Pack (Trial)	ок		2021-04-30
0	test	Nebula Professional Pack (Trial)	ОК		2021-04-30
0	TestOrg2	Nebula Professional Pack (Trial)	ок		2021-04-25
4					•

Table 188 Group-wide > Group-wide manage > Group portal

LABEL	DESCRIPTION
Search	Specify your desired filter criteria to filter the list of organizations.
matches in	This shows the number of organizations that match your filter criteria after you perform a search.

LABEL	DESCRIPTION
N Organizations	This shows the number of organizations (N) tin the group.
Status	This shows the status of Nebula Devices in the organization.
	 Green: All Nebula Devices are online and have no alerts. Amber: One or more Nebula Devices have alerts. Red: One or more Nebula Devices are offline. Gray: All Nebula Devices have been offline for 7 days or more. White: No Nebula Devices.
Organization	This shows the descriptive name of the organization.
Туре	This shows the NCC license type of the organization.
NCC License Status	This shows whether the license is valid (OK), the license has expired and the organization downgraded from NCC Pro or Plus Pack to the base tier (Expired), or this is a free organization and an NCC license is not required (N/A).
Payment mode	This shows the payment method of the organization's license if you arranged a special payment method with Zyxel.
	If you bought the license through the Zyxel web store or a third-party vendor, the value will be blank.
NCC License expiration (UTC)	This shows the date when the license will expire, or N/A when there are no Nebula Devices in the organization or if this is a free organization and an NCC license is not required.
Sites	This shows the number of sites belonging to this organization.
Devices	This shows the number of Nebula Devices in the organization that have one of the following status:
	Green: The Nebula Device is online and has no alerts.
	 Amber: The Nebula Device has alerts. Red: The Nebula Device has been offline for less than 7 days.
	 Gray: The Nebula Device has been offline for 7 days or more.
AP	This shows the number of Nebula Access Points in the organization.
SW	This shows the number of Nebula Switches in the organization.
SA	This shows the number of NSG and USG FLEX, ATP series, and USG20(W)-VPN Security Appliances connected to the sites in this organization.

Table 188	Group-wide >	Group-wide	manaae >	Group portal	(continued)
		Oloop=wide	manage -	Croop portai	

12.3 Org-to-Org VPN

Org-to-Org VPN allows devices in different organizations in a group to access each other's services, such as a website, database, or ERP server, through VPN tunnels.

12.3.1 Configure Org-to-Org VPN

Follow the steps below to configure Org-to-Org VPN in the group.

- 1 Configure Smart VPN for each organization you want included in the Org-to-Org VPN.
 - **1a** In the **Organization** list, select the organization.
 - **1b** Go to Organization-wide > Organization-wide manage > VPN orchestrator.

- **1c** Configure a VPN area with hub-and-spoke topology, and then assign at least one site as a hub. If a site contains a server that you want to share between organizations, then ensure the server is in a hub site or that **Branch to Branch VPN** is enabled.
- 2 Go to Group-wide > Group-wide manage > Org-to-Org VPN, and then enable Hub to Hub VPN.
- 3 Click + Hub. In the Select Hubs window, add at least one hub site from each organization to the Within Org-to-Org list.
- 4 Click + Org-to-Org Service, and add a server's fully qualified domain name (FQDN) and IP address.
- 5 Devices in the organizations included in the Org-to-Org VPN are now able to access the server by IP address or FQDN.

12.3.2 Org-to-Org VPN Example

Figure 221 shows organization O1 with two VPN areas and hubs H1 and H2. Area communication and Branch to Branch VPN are both enabled. It shows another organization O2 with its own set of sites and a hub. H1 and H3 belong to the Org-to-Org VPN. The server behind S9 is listed as an org-to-org service. If a Nebula Device behind S5 wants to access the server behind S9, traffic will pass through its hub H2 and then to H1 and H3.



12.3.3 Org-to-Org VPN Screen

Click Group-wide > Group-wide manage > Org-to-Org VPN to access this screen.



rg-to-Org VPN		
Reserved IP Address Pool	10.255.255.0/24 👻	
AutoVPN		
Hub to Hub VPN		
	Organization	Hub
	Hub	
Service		
	Organization FQDN	IP Address
	+ Org-to-Org Service	
		Save or Cancel
	(Please allow 1-2 m	inutes for changes to take effect.)

Figure 222 Group-wide > Group-wide manage > Org-to-Org VPN

LABEL	DESCRIPTION
Reserved IP Address Pool	Specify the IP addresses that Nebula Devices use to create the VPN tunnels between the gateway devices in the org-to-org VPN network. You can select a set or custom range.
	This IP address range must not overlap with any IP address ranges already in use within any sites in the org-to-org VPN.
AutoVPN	•
Hub to Hub VPN	Turn the switch to On to enable create VPN tunnels between the hubs in the list. This is required to enable Org-to-Org VPN.
	When this setting is disabled, Org-to-Org VPN will not work and can only be configured.
Organization	This column lists down the organization to which the hub site belongs.
Hub	This column lists down the names of the hub sites included in the Org-to-Org VPN.
+Hub	Click this to set up which hub site you want to add to the Org-to-Org VPN.
Service	
Organization	This displays the organization to which the network service belongs.
FQDN	This displays the Fully-Qualified Domain Name (FQDN) associated with the network service which Security Gateway devices and Nebula Devices behind them are given access.
IP Address	This displays the IP address of the network service which Security Gateway devices and Nebula Devices behind them are given access.
+Org-to-Org Service	Click this to add a service that can be accessed within the org-to-org VPN.
Save	Click this button to save your changes and close the screen.
Cancel	Click Cancel to exit this screen without saving.

Table 189	Group-wide >	Group-wide manage	e > Ora-to-Ora VPN

12.3.4 Add Hub

Click the **+Hub** button on the **Group-wide > Group-wide manage > Org-to-Org VPN** screen to access the following screen. If **Hub to Hub VPN** is enabled, use this screen to select which hubs you want to include in the **Org-to-Org VPN**.

Figure 223 Group-wide > Group-wide manage > Org-to-Org VPN: SD-WAN Hubs



Hubs are listed in this screen and you may choose whether to include them in the org-to-org network or not by clicking the "<" and ">" buttons. The "<<" and ">>" buttons move all hubs at once. Details about this screen are described in the table below.

The following table describes the labels in this screen.

LABEL	DESCRIPTION
All Organization Hubs	This box lists all hub sites in the group that are outside the org-to-org network. It shows the name of the hub followed by the Organization it belongs to in parentheses.
Within Org-to-Org	This box lists all hub sites inside the org-to-org network. It shows the name of the hub followed by the Organization it belongs to in parentheses.
Cancel	Click Cancel to exit this screen without saving.
Save	Click Save to add the hubs to the org-to-org network.

Table 190 Group-wide > Group-wide manage > Org-to-Org VPN: SD-WAN Hubs

12.3.5 Service

Use this screen to add a service accessible through the org-to-org VPN. Note that you can choose to add only the FQDN or only the IP address. Click **+Org-to-Org Service** and then the following screen appears.

Figure 224 Group-wide > Group-wide manage > Org-to-Org VPN: Service

Add service			
Organization	10 point then haas AP	•	
FQDN		× *	
IP Address		*	
	Cancel	Save	

LABEL	DESCRIPTION
Organization	Select the organization to which the service you want to add is linked to.
FQDN	Enter the Fully-Qualified Domain Name (FQDN) associated with the service. An FQDN starts with a host name and continues all the way up to the top-level domain name. For example, www.zyxel.com.tw is a fully qualified domain name, where "www" is the host, "zyxel" is the third-level domain, "com" is the second-level domain, and "tw" is the top level domain. Underscores are not allowed. Use "*." as a prefix in the FQDN for a wildcard domain name (for example, *.example.com).
IP Address	Enter the IP address of the service you want to add to the org-to-org VPN.
Save	Click Save to allow access to the service through the org-to-org VPN.
Cancel	Click Cancel to exit this screen without saving.

Table 191	Group-wide >	Group-wide	manaae >	Ora-to-Ora	VPNI Service

12.4 Inventory

Use this screen to view all Nebula Devices in the organizations of the selected group. Click **Group-wide** > **Group-wide manage** > **Inventory** to access this screen.

Figure 225	Group-wide > Group-wide manage > Inventory
rigule zzj	

nventory							
Show all devices	belong to each org	ganization in group					
Unused Used	Both Q Sea	rch	•	2) devices.			Export -
MAC address	Serial Number	Organization	Site	Model	Registered on (UTC)	Country	
10.0112-00.00	10.0101241903	0322		NWA1123-ACv2	2021-03-22 08:48:49		
312100820	20201030405200	10 point then haas AP		NWA1123-ACv2	2021-03-06 05:04:24		

LABEL	DESCRIPTION
Unused	Click this button to show the Nebula Devices which are not assigned to a site yet.
Used	Click this button to show the Nebula Devices which are assigned to a site.
Both	Click this button to show all Nebula Devices which are registered for the organizations in the group.
Search	Enter a key word as the filter criteria to filter the list of connected Nebula Devices.
	Open the search box drop-down list to filter the search results by site, model, and country.
Devices	This shows the number of the Nebula Devices in the list.
Export	Click this button to save the Nebula Device list as a CSV or XML file to your computer.
MAC address	This shows the MAC address of the Nebula Device.
	Click on the MAC address to view the Nebula Device details page.
Serial number	This shows the serial number of the Nebula Device.
Organization	This shows the organization of the Nebula Device.

 Table 192
 Group-wide > Group-wide manage > Inventory



LABEL	DESCRIPTION
Site	This shows the name of the site to which the Nebula Device is connected.
Model	This shows the model number of the Nebula Device.
Registered on (UTC)	This shows the date and time that the Nebula Device was registered at the NCC.
Country	This shows the country where the Nebula Device is located.

Table 192 Group-wide > Group-wide manage > Inventory (continued)

12.5 Administrators

Group Administrator accounts can be added, modified, or deleted through this screen. A group administrator has administrator privileges in all organizations in the group. Group administrators are registered using their NCC account email address.

Click Group-wide > Group-wide manage > Administrators to access this screen.

Figure 226 Group-wide > Group-wide manage > Administrators

1	Name	Email address	Privilege	Account status	Last access time (UTC)	Create date (UTC)	Statu 🗐
	and table	Justial amplitude of the producers	Owner	OK	2021-03-26 04:48:06	2021-03-15 07:21:22	2021-03
2	and departure.	and a subscription of the subscription of the	Organization (Delegated)	OK	2021-03-26 05:47:36	2021-03-23 03:28:03	2021-03
	anin diam Brazaki.	with changing with constant	Organization (Delegated)	OK	2021-03-26 06:01:22	2021-03-23 03:28:49	2021-03
	an ay self-spreak.	at gyalitymeticen.	Organization (Delegated)	OK	2021-03-26 06:25:57	2021-03-23 03:29:41	2021-03
)	Test with the	Burnas namingEspation (ar	Organization (Delegated)	OK	2021-03-26 03:36:07	2021-03-23 05:59:52	2021-03
	deeniharra@	distant ituang Espanican itu	Organization (Delegated)	OK	2021-03-26 03:00:16	2021-03-23 06:45:15	2021-03

|--|

LABEL	DESCRIPTION
Activation	Click this button to Activate/Deactivate the selected accounts. Then, click Update.
Force logout	Click this button to force the selected accounts to log out of NCC.
Delete	Click this button to remove group administrator privileges for the selected accounts.
Search	Specify your desired filter criteria to filter the list of administrator accounts.
administrators	This shows the number of administrator accounts in the list.

LABEL	DESCRIPTION				
Import	Click this button to create administrator accounts in bulk by importing a complete list of all new administrators in an Excel file.				
	Bulk Import X				
	"Bulk Import" supports for faster inputting. Please follow this <u>template</u> to import				
	Browse				
	Or drag file here				
	Close				
Add	Click this button to create a new group administrator account. See Section 12.5.1 on page 565.				
Name	This shows the name of the administrator account.				
Email address	This shows the email address of the administrator account.				
Privilege	This shows the privileges the administrator has within all organizations in the group.				
	Full: the administrator can edit settings, create or delete other administrator accounts, create or delete a site, and add or renew licenses for Nebula Devices in the organization.				
	Read-only : the administrator account has no write access to the organization, but can be a site administrator.				
	Delegate owner's authority : The administrator account has delegated owner privileges. This type of account can perform all of the same actions as the organization owner, except for the following:				
	 Delete organization Transfer organization ownership Assign delegate owner privileges to an administrator account. 				
Account status	This shows whether the administrator account has been validated (OK). It shows Deactivated if an administrator account has been created but cannot be used. This may happen since you can only have up to five active administrator accounts in the NCC base tier.				
Last access time	This shows the last date and time traffic was sent from the administrator account.				
Create date	This shows the date and time the administrator account was created.				
Status change date	This shows the last date and time the administrator account status was changed.				
	Click this icon to display a greater or lesser number of configuration fields.				

Table 193 Group-wide > Group-wide manage > Administrators (continued)

12.5.1 Create/Update Administrator

In the Group-wide > Group-wide manage > Administrators screen, click the Add button to add a new group administrator account or double-click an existing account entry to modify the account settings.

Figure 227	Group-wide >	Group-wide manage >	Administrators.	Create/Update	administrator
11guic 227		oroup mac manage -	/ arrining arous.	cicule, opuale	aarminisharor

Create administrator				
Name:		× *		
Email:		× *		
Organization access:	Full	•		
	Delegate owner's authority 🚺			
Activate:	No	•		
	_	_		
	Close Create ad	dmin		

LABEL	DESCRIPTION
Name	Enter a descriptive name for the administrator account.
Email	Enter the email address of the administrator account, which is used to log into the NCC.
	This field is read-only if you are editing an existing account.
Organization	This shows the privileges the administrator has within all organizations in the group.
access	Full: the administrator can edit settings, create or delete other administrator accounts, create or delete a site, and add or renew licenses for Nebula Devices in the organization.
	Read-only : the administrator account has no write access to the organization, but can be a site administrator.
Delegate	This setting is only available when Organization access is set to Full.
owner's duinonly	Select this setting to grant delegate owner privileges to an organization full administrator account. An account with delegate owner privileges can perform all of the same actions as the organization owner, except for the following:
	Delete organization
	Transfer organization ownership Assign delegate owner privileges to an administrator account
Activate	Select Ves to enable the account or No to temporarily disable the account
Activate	
Close	Click this button to exit this screen without saving.
Create admin/ Update admin	Click this button to save your changes and close the screen.

Table 194	Group-wide >	Group-wide r	manaae >	Administrators:	Create/Updat	e administrator
			nanago -	/ (011111111101013)	ciculo, opual	c durining alor

12.6 Change Log

Use this screen to view logged messages for changes in all organizations in the group. Click **Group-wide** > **Group-wide** manage > **Change log** to access this screen.

When the log is full, it deletes older entries one by one to make room for newer ones.

nange log									
Keyword:									
Q Search.		-							
	From:			To:					
Range 🔻	2021-03-16		03:59 💌	2021-03-26		03:59 💌	UTC+0	× Q	Search
	Max range is 30 days	s, the dates will be a	uto-adjusted.						
	Older > 9	change logs w	vithin the tin	ne filtered. Changes d	late back to	2021-03-1	5 07:21 (U	ITC)	
Time (UTC)		Admin		Page	Label		Old	value	New value
2021-03-23	06:45:19	svd nsbu		Administrator	Addeo	d Alexandri iliy	100 a		Added, Organizati
2021-03-23	06:07:51	svd nsbu		Administrator	Updat	ted Tech-w	vri Rem	noved: Organi	z Added: Organizati
2021-03-23	06:02:12	with charries of	cell com the	Administrator	Chan	ged Tech-v	wr Org	anization: Red	a Organization: Full
2021-03-23	05:59:56	with charries of	nel com the	Administrator	Addeo	d Tech-writ	te		Added, Organizati
2021-03-23	03:29:45	svd nsbu		Administrator	Addeo	t in synakti			Added, Organizati
2021-03-23	03:28:51	svd nsbu		Administrator	Addeo	dan tin cêlam			Added, Organizati
2021-03-23	03:28:14	svd nsbu		Administrator	Updat	ted sdd9.rd	d Rem	noved: Organi	z Added: Organizati
2021-03-23	03:28:05	svd nsbu		Administrator	Addeo	d administratio	10.		Added, Organizati

Figure 228 Group-wide > Group-wide manage > Change log

Table 195 Group-wide > Group-wide manage > Change log

LABEL	DESCRIPTION
Keyword	Enter a keyword or specify one or more filter criteria to filter the list of log entries.
Range/Before	Select a filtering option, set a date, and then click Search to filter log entries by date.
	Range: Display log entries from the first specified date to the second specified date.
	Before: Display log entries from the beginning of the log to the selected date.
Search	Click this to update the list of logs based on the search criteria.
Reset filters 🖂	Click this to return the search criteria to the previously saved time setting.
Newer/Older	Click to sort the log messages by most recent or oldest.
N change logs within the time filtered.	This shows the total number of the log messages that match the search criteria. It also shows the date and time the very first log was created.
Export	Click this button to download the log list as a CSV or XML file to your computer.
Time (UTC)	This shows the date and time in UTC+00:00 (or UTC+0) when the log was recorded.
	UTC is a standard time for use around the world (formerly known as Greenwich Mean Time or GMT). UTC is an international abbreviation that is neither French nor English. It means both "Temps Universel Coordonné" and "Coordinated Universal Time".
Admin	This shows the name of the NCC administrator account that made the changes.
Page	This shows the name of the NCC menu in which the change was made.
Label	This shows the action that triggered the log entry
Old value	This shows the old setting or state that was overwritten with the new value.
New value	This shows the new setting or state.
	Click this icon to display a greater or lesser number of configuration fields.

12.7 Group Settings

Use this screen to change your general group settings, such as the group name and members. Click **Group-wide > Group-wide manage > Group settings** to access this screen.

up-wide > Configure > <u>Settir</u>	201	
tings		
roup information		
Group name	Zyxel ×	
Description		
	×	
roup members		
Organizations	Nebula_Org2 😵	
	Note: You could select organizations own by you to join Group.	
Delete this group	You can delete this aroun only when	
0, o o b	+ No any Pro Pack organization belong to it	
	+ AutoVPN is off	

LABEL	DESCRIPTION
Group name	Enter a descriptive name for the group.
Description	Enter a description for the group.
Group members	Click in the box to add an organization to the group. Click X to remove an organization from the group. Note: You must be the group owner, and each group must have a Pro license.
Delete this group	Click this to delete the group. Note: You can only delete a group if it contains no organizations, and Hub to Hub VPN is disabled at Group-wide > Group-wide manage > Org-to-Org VPN .

Part V MSP

Chapter 13 MSP

13.1 Overview

The **MSP** (Managed Services Provider) menus allow you to view the summary of organizations and change the branding on NCC.

An MSP license that expires will keep the previous settings in MSP but disable the MSP features.

An MSP license can be transferred to another MSP administrator. Click the More icon at the top righthand corner of the **Dashboard** screen and click the **Services** tab to view the **Status** of MSP licenses. To transfer an MSP license, select the MSP license and click **Actions** > **Transfer license**. Alternatively, click **Transfer license** under **Actions**.

Figure 230 Transfer an MSP License



Note: To see these menus, assign an MSP license to your NCC login account.

13.2 MSP Portal

This screen lists every organization to which your account has at least read-only access.

To access this screen, select MSP portal from the Organization drop-down list box in the title bar, or click MSP cross-org > MSP cross-org manage > MSP portal in the navigation panel.

org	anizations w	vill expire in 90 days.	Click here to see detail.								
	iour										
	iew				Device li						
Jrg	anization	type summary			Device lic	cense s	tatus sur	nmary			
					@@@YaW	en_Mobil	le router/ F	LEX, Hote	el Maeir, N	IY HOME 🔻	
										Healthy	
	8		Nebula Professional Pack	5						Normal	
V			Nebula Base Pack	10			6			Near expiration	
						-				Unlicensed (Expired)	
ani	zations	Sites									
ani act	zations livate CSM	Sites Tag - Q. Searc	h • (14) Organizations	1							Exp
ani act	zations tivate CSM Status	Sites Tag - Q Searc NCC license statu	h • (14) Organizations us () Type • Tag Nebula Base Pack	s Site	es Devid	ces online	e 0 •0	AP 0	sw	Payment mode	Next NCC lia
ani act	zations ivate CSM Status	Sites Tag - Q. Searc NCC license statu N/A	h • 14 Organizations 15 1 Type • Tag Nebula Base Pack Nebula Base Pack	; Site 5 Test2 1	es Devie 0 0	ces online 0	e ● 0 ● 0 ● 0 ● 0	AP 0	sw 0	Payment mode	Next NCC line N/A N/A
ani act))	zations	Sites Tag - Q. Searce NCC license statu N/A N/A	h Type Type Tog Nebula Base Pack	Site 5 Test2 1 1	es Devi 0 0 0	ces online 0 0 0 0	• • • • • • • • • • •	АР 0 0	sw 0 0	Payment mode	Next NCC lid N/A N/A N/A
ani act)))	zations ivate CSM Status () O O O O	Sites Tag • Q Searce NCC license statu N/A N/A N/A N/A	hv (14) Organizations us () Type Tag Nebula Base Pack Nebula Base Pack Nebula Base Pack Nebula Base Pack	Site 5 Test2 1 1	es Devia 0 0 0 0 0 0	Ces online 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0	AP 0 0 0	sw 0 0 0	Payment mode	Next NCC lia N/A N/A N/A N/A
ani act)))	zations	Sites Tag • Q Searce NCC license statu N/A N/A N/A N/A N/A N/A N/A N/A	L Type Ta Organizations Nebula Base Pack Nebula Base Pack Nebula Base Pack Nebula Base Pack Nebula Base Pack Nebula Base Pack	5 5 Test2 1 1 1 1 1	es Devia 0 0 0 0 0 0 0 0 0 0 0 0	Ces online O O O O O O O O O O O O O O O O	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	АР 0 0 0 0	sw 0 0 0 0 0	Payment mode	Next NCC lin N/A N/A N/A N/A N/A
ani act)))))	zations	Sites Tag Q Search NCC license statu N/A	Image: Second	Site 5 Test2 1 1 1 1 1 1 2	Bes Devia 0 0 0 0 0 0 0 0 0 0 0 0 0 0	ces online 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	AP 0 0 0 0 0 0	sw 0 0 0 0 0 0	Payment mode	Noxt NCC fiel N/A N/A N/A N/A N/A N/A
	zations ivate CSM Status j O O O O O O O O O O O O O O O O O O O	Sites NCC license statu N/A	Image: Straig	site 5 Test2 1 1 1 1 2 1	es Devi • 0 • 0 • 0 • 0 • 0 • 0 • 0 • 0 • 0 • 0 • 0 • 0 • 0 • 0 • 0 • 0	 ces online 0 0<td>0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0</td><td>АР О О О О О О О</td><td>sw 0 0 0 0 0 0 0 0</td><td>Payment mode</td><td>Next NCC lin N/A N/A N/A N/A N/A N/A N/A N/A</td>	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	АР О О О О О О О	sw 0 0 0 0 0 0 0 0	Payment mode	Next NCC lin N/A N/A N/A N/A N/A N/A N/A N/A
	ivate CSM	Sites Q Search NGC Iicense statu N/A N/A - -	Image: Straig	s Test2 1 1 1 1 2 1 1 1 1 1 1 1	BS Devia 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	Cos online 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	AP 0 0 0 0 0 0 0 0 0 0	sw 0 0 0 0 0 0 0 0 0 0	Payment mode	Next NCC lin N/A N/A N/A N/A N/A N/A N/A N/A
	ivote CSM	Sites Sites Case of the second N/A N/A N/A N/A N/A N/A N/A N/A	 Type Type Toge Toge Toge Nebula Base Pack 	s Test2 1 1 1 1 1 1 2 1 1 1 1 1 2 1 1 1 0	as Devia • 0 • 0 • 0 • 0 • 0 • 0 • 0 • 0 • 0 • 0 • 0 • 0 • 0 • 0 • 0 • 0 • 0 • 0 • 0 • 0	Cos online 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	АР О О О О О О О О О О	sw 0 0 0 0 0 0 0 0 0 0 0	Payment mode	Next NCC lin N/A N/A N/A N/A N/A N/A N/A N/A N/A
	ivate CSM	Sites Tag • Q. Search NCC license statu N/A N/A - - N/A - -	Type Ta Organizations Nebula Base Pack	5 Test2 1 1 1 1 1 1 1 2 1 1 1 0 1 1 1 1 1 1 1 1 1 1 1 1 1	as Devia 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	Ces online O	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	АР 0 0 0 0 0 0 0 0 0 0 0	sw 0 0 0 0 0 0 0 0 0 0 0 0 0	Payment mode	Noxt NCC lit N/A N/A

Figure 231 MSP cross-org > MSP cross-org manage > MSP portal

Table 197	MSP c	ross-org > N	ЛSP	cross-org	manage	> MSP	portal
		0		0	0		

LABEL	DESCRIPTION
Organization type summary	This pie chart shows the total number of the organization mode (for example, x PRO, x Plus, x Base organizations).
Device license status summary	This pie chart shows the total number of Nebula managed devices with NCC and ATP licenses only. You can select the organization to display in the drop-down list. Click a particular color in the pie chart to show the details of the licenses of the selected organizations.
Organizations	
Deactivate CSM	Select the organization(s) and click this button to disable CSM (Cloud-Saving Mode). See Section 1.6 on page 45 for more information on Cloud-saving mode.

LABEL	DESCRIPTION
Tag	Assign a name to an organization or to a group of organizations.
	1. Select the organizations. The Tag button will be enabled.
	2. Click Tag.
	 In the Add field, enter a tag (up to 32 alphanumeric characters and spaces are allowed).
	4. Click +Add new. Then Add to confirm.
	To remove the tag assigned to an organization or to a group of organizations.
	1. Select the organization with an assigned tag.
	2. Click Tag.
	3. Enter the name of the tag. As you type along, NCC will automatically show the names of tags that matches.
	4. Select the tag. Then click Remove .
Search	Specify your desired filter criteria to filter the list of organizations and organization status.
matches in	This shows the number of organizations that match your filter criteria after you perform a search.
Organizations	This shows the number of organizations that you can manage.
*	Click this to select all rows.
	Alternatively, click a row to go to the Sites tab that will show the sites belonging to the organization.
Status	This shows the status of Nebula Devices in the organization.
	 Green: All Nebula Devices are online and have no alerts. Orange: Some Nebula Devices have alerts. Red: Some Nebula Devices are offline. Gray: All Nebula Devices have been offline for 7 days or more. White: No Nebula Devices in this organization. (a): This organization is in Cloud-saving mode.
NCC license	This shows the license status of Nebula Devices in the organization.
SIGIOS	 Green: All Nebula Devices with over 1 year licenses. Blue: Any Nebula Device with over 90 days but less than 1 year license together with another Nebula Device with over 1 year license. Orange: Any Nebula Device with license that will expire in 90 days together with another Nebula Device with over 90 days license. Red: Any Nebula Device with an expired license or is unlicensed. Gray: No Nebula Devices in this organization.
Organization	This shows the descriptive name of the organization. Click an Organization to go to the Organization-wide > Organization-wide manage > Organization portal screen. Hover the mouse over the name of the Organization to display the site information window. Clicking a Site name will go to the Site-wide > Dashboard screen.
Туре	This shows your NCC version type.
Tag	This shows the tag name assigned to this organization. Otherwise, the organization does not have a tag.
Sites	This shows the number of sites belonging to this organization.

 Table 197
 MSP cross-org > MSP cross-org manage > MSP portal (continued)

LABEL	DESCRIPTION
Devices online	This shows the number of Nebula Devices in this organization which are online (green), have recently had alerts (orange), recently went offline (red), or have been offline for more than 6 days (gray).
AP	This shows the number of Nebula access points connected to the sites in this organization.
SW	This shows the number of Nebula switches connected to the sites in this organization.
Security appliance	This shows the number of Nebula security appliances connected to the sites in this organization.
MR	This shows the number of Nebula mobile routers connected to the sites in this organization.
Payment mode	This shows the payment method of the NCC license if you arranged a special payment method with Zyxel.
	If you bought the license through the Zyxel webstore or a third-party vendor, the value will be blank.
Next NCC license expiration date	This shows the date when the license will expire, or N/A when there is no Nebula-managed device in the organization.
	For example, if you have two Nebula Devices in the organization:
	 Nebula Device 1 is with NCC license expiration date on 2022/10/1 Nebula Device 2 is with NCC license expiration date on 2022/11/1
	This field will show the nearest expiration date '2022/10/1'.
# devices will expire in 90 days	This shows the number of Nebula-managed devices with licenses that will expire in 90 days or less in this organization.
# unused NCC license	This shows the number of unused NCC (Nebula Control Center) licenses in this organization.
	Click this icon to display a greater or lesser number of configuration fields.
Export	Click this button to save the MSP Portal list as a CSV or XML file to your computer.
Sites	
Search	Specify your desired filter criteria to filter the list of sites.
matches in	This shows the number of sites that match your filter criteria after you perform a search.
sites	This shows the number of sites that you can manage.
*	Click this to select all rows.
Status	This shows the status of Nebula Devices in the site.
	Green: All Nebula Devices are online and have no alerts.
	Orange: Some Nebula Devices have alerts. Red: Some Nebula Devices are offline.
	 Gray: All Nebula Devices have been offline for 7 days or more.
	White: No Nebula Devices in this site.
Organization	This shows the descriptive name of the organization.
Site	This shows the descriptive name of the site. Clicking a site name will go to the Site-wide > Dashboard screen.
Tags	This shows the tag name assigned to this site. Otherwise, the site does not have a tag.
Devices	This shows the number of Nebula Devices connected to the site.
Offline devices	This shows the number of Nebula Devices in this site which are offline.
% Offline	This shows the percentage of Nebula Devices in this site which are offline.
Template	This shows the name of the template that is bound to a site.
	Click this icon to display a greater or lesser number of configuration fields.
Export	Click this button to save the MSP Portal list as a CSV or XML file to your computer.

 Table 197
 MSP cross-org > MSP cross-org manage > MSP portal (continued)

NCC User's Guide

13.3 Cross-org synchronization

The Cross-org synchronization screen allows you to copy settings or a site from one organization to another. You can also move Nebula Devices with its settings to another organization.

13.3.1 Cross-Org setting sync

Cross-org sync copies the following items from one organization to another organization:

- Organization-wide settings
- Administrators
- Cloud Authentication accounts (Users and MAC)
- Configuration templates

Your account must have **owner** or **organization-full** privileges in both source and destination organizations. When copying organization-wide settings, the following settings will not be overwritten if they are already configured in the destination organization:

- Organization-wide > Organization-wide manage > Organization settings > Country
- Organization-wide > Organization-wide manage > Organization settings > Login IP ranges
- Administrators privileges (when source and destination organizations have the same admin account)
- Cloud Authentication account privileges (when source and destination organizations have the same Cloud Authentication account)

When copying configuration templates:

- No sites are bound to the new template site.
- If the destination organization has a template with the same name, then the new template will have a number appended to the end of its name.

13.3.2 Cross-Org site clone

Cross-org site clone copies a site and all of its settings from one organization to another. Your account must have **owner** or **organization-full** privileges in both source and destination organizations.

If the destination organization has a site with the same name, then the new site will have a number appended to the end of its name.

The following table describes the Nebula Device (Access Point, Switch, Security Firewall) during cross-org site clone.

NEBULA DEVICE	CROSS-ORG SITE CLONE	MOVE NEBULA DEVICE TO CLONED SITE – ENABLED	KEEP MANAGEMENT/WAN INTERFACE – ENABLED
Access Point (AP)	 When enabled: AP site-wide configuration is cloned Individual AP configuration is NOT cloned (for example, radio settings) 	When enabled: AP site-wide configuration and individual AP configuration are cloned (for example, radio settings)	When enabled: AP site-wide configuration and individual AP configuration are cloned (for example, radio settings)
Switch	 When enabled: Switch site-wide configuration is cloned Individual Switch configuration is NOT cloned (for example, IGMP) Switch port configuration is NOT cloned 	 When enabled: Switch site-wide configuration is cloned Individual Switch configuration is cloned (for example, IGMP) Switch port configuration is cloned 	 When enabled: Switch site-wide configuration is cloned Individual Switch configuration is cloned (for example, IGMP) Switch port configuration is cloned
Security Firewall	When enabled, the site-to-site VPN settings are reset.	When enabled, the site-to- site VPN settings are reset.	When enabled, the site-to- site VPN settings are reset.

Table 198 Nebula Device Cross-org Site Clone

13.3.3 Cross-org synchronization Screen

Use this screen to configure cross-org synchronization and cross site clones.

Figure 232 MSP cross-org > MSP cross-org manage > Cross-org synchronization

s-org synchronization						
oss-Org setting sync						
From source organization	Test_July	•				
Org. setting	All org-wide settin	gs, Org 🔻				
To dest. organization	Nebula_Org			•		
		Sync				
oss-Org site clone with devic	e movement					
From source organization	Test_July	•	ZyNet TW	•		
	Move site device	es to cloned sit	e in destination org	anization. <u>What is</u>	it?	
	When you mavin organization, you Management/W/ environment is si	g site include o u could select r AN Interface or milar or the so	levices to another eset device keep it if your netv me.	vorking		
	Keep Ma	nagement/WA	N Interface.			
To dest. organization	TWTest	•				
		Clone				

NCC User's Guide

575

Table 199 MSP cross-org > MSP cross-org manage > Cross-org synchronization

LABEL	DESCRIPTION						
Cross-Org setting sync	Cross-Org setting sync						
From source organization	Select the organization to copy settings from.						
Org. setting	Select the settings that you want to copy from the source to the destination organization.						
	Select All org-wide settings to copy everything.						
To dest. organization	Select the organization to copy settings to.						
Sync	Click this to copy the selected settings from the source to the destination organization.						
Cross-Org site clone w	ith device movement						
From source	Select the organization to copy settings from.						
organization	Then select one or more sites. Select All sites to copy all sites from the source to the destination organization.						
	Select Move site devices to cloned site in destination organization to include the Nebula Devices.						
	Enable Keep Management/WAN interface to copy the WAN connection settings for the Nebula Devices to the destination organization.						
To dest. organization	Select the organization to copy the selected sites to.						
Clone	Click this to copy the selected organization and sites from the source to the destination organization.						

13.4 MSP Alert Templates

The MSP administrator can configure MSP alert template to monitor Nebula Devices for unexpected events (for example, online / offline events). This screen will list the alert templates you have created. See Section 13.4.1 on page 577 for details on creating an alert template.

To access this screen, click **MSP cross-org** > **MSP cross-org manage** > **Alert templates** in the navigation panel.

+ 4	Create	🕆 Dele	ete Q Search	• (1)selected in 1 Temp	late	
	Name	-	Description	Creator	Bound organizations	Exclude sites	Enable
	Test-temp	olate i	For testing purposes.	samuel.yu@zyxel.com.tw	Test July	ZyNet TW-2	
						Save or Cancel	
				(Please allow 1-2 mir	nutes for chang	ges to take effect.)

Table 200	MSP cross	-org > MSF	° cross-org	manage >	Alert templates	

LABEL	DESCRIPTION
+ Create	Click this button to add a new alert template (see Section 13.4.1 on page 577).
Delete	Click this button to remove alert templates already created.
LABEL	DESCRIPTION
---	---
Search	Specify your desired search criteria to filter the list of alerts.
selected in	This shows the number of alerts that match your filter criteria after you perform a search.
Template	This shows the number of alert templates you have created.
Name	This shows a descriptive name of the alert template.
Description	This shows more details on the alert template.
Creator	This shows your email address.
Bound organizations	This shows All organizations or a list of the selected organizations to send alerts to.
Exclude sites	This shows the sites that will not receive any alerts.
Enable	Click this to activate the alert template.
Note: To edit the N a the field and	ame, Description, Creator, Bound organizations, and Exclude sites fields, just click the Update alert screen will appear.

 Table 200
 MSP cross-org > MSP cross-org manage > Alert templates (continued)

13.4.1 Alert Settings

Use this screen to set which alerts are created and emailed, and set the email addresses to which an alert is sent. Click **MSP cross-org** > **MSP cross-org manage** > **Alert templates** > **Create** to access this screen.

Note: NCC's Smart Alert Engine uses knowledge of network topology and cross-device functionality to only generate alerts for unexpected events. This helps avoids unnecessary emails and notifications.

For example, an AP is receiving power from a PoE switch. If the AP loses power because its Ethernet cable is disconnected, NCC generates an alert. If the AP loses power because the switch has a PoE schedule that disables power to the AP, NCC does not generate an alert.

reate alert		
General		
Template name		×
Description		~
Description		~
Email recipient 🚺	E.g. nebula@zyxel.com	
Apply to	All organizations	
Evoludo ottoo	Select organizations	
Exclude sites	Select organizations Select sites Add to exclude list	
Enable		
System alerts 🚹		
Wireless	Disable 🔻 5 💌 minutes after AP goes offline	
	Show additional recipients	
Switches	Disable 🔻 5 💌 minutes after Switches goes offline	
	Show additional recipients	
	Disable T E minutes Any switch part T ages down	
	Show additional recipients	
Security gateway		
	Disable 5 minutes after the gateway goes offline	
	Disable Any DHCP lease pool is exhausted	
	Show datitional recipients	
	Disable A VPN connection is established or disconnected	
	Show additional recipients	
	Disable WAN connectivity status changed	
	Show additional recipients	
Mobile router	Disable	
	Show additional recipients	
Other	Disable Configuration settings are changed	
	Show additional recipients	
Security alerts		
	Disable Email to receive containment alerts	
CDR containment 🚹	Show additional recipients	
Security Report		
Notification mode	Disable Email to receive security alerts by SecuReporter	
	Show additional recipients	
Email subject	× (Optional, maximum character is 64.)	
Email description		
	× (Optional, maximum character is 255.)	
Notification interval	1 hour Select notification interval if events were triggered	

Figure 234 MSP cross-org > MSP cross-org manage > Alert templates > Create/Update alert

NCC User's Guide

Category		Severity		
Network Security	Attack counts	High	1	times of highest severity attacks within 5 minutes.
Network Security	Attack counts	High	10	times attacks within 5 minutes.
Network Security	Alert counts	High	10	count(s) of Malware/IPS(highest severity)/ADP(protocol anomaly) within 1 minute.
Network Security	Malware/virus detection	Medium	2	times of same malware/virus is detected within 15 minutes.
Network Security	Malware/virus detection	High	10	count(s) of malware/virus attack within 5 minutes.
Network Security	URL Threat Filter	High	5	times of connection to threat websites within 60 minutes.
Network Security	DNS Threat Filter	High	5	times of connection to threat/block DNS domain within 60 minutes.
Network Security	Sandboxing	High	10	times destoryed malicious files within 5 minutes.
Network Security	Sandboxing	High	10	times destoryed suspicious files within 5 minutes.
Network Security	IP Reputation-Incoming	High	10	times over of attacks to the internal network from external threat IP address within 10 minutes
Network Security	IP Reputation-Outgoing	High	1	times over of connections to threat websites within 60 minutes
Anomaly	Login failure	Medium	10	times of login failures within 1 minute.
Anomaly	Traffic anomaly	High	1	times of traffic anomaly scans/floods detected within 5 minutes.
Anomaly	Protocol anomaly	High	1	times of protocol anomaly TCP/UDP/ICMP/IP decoders within 5 minutes.

The following table describes the labels in this screen.

Table 201	MSP cross-ora > MSP	cross-ora manaae > Alert template	es > Create/Update alert

LABEL	DESCRIPTION
General	
Template name	Enter a descriptive name for the alert template (up to 64 alphanumeric characters including spaces).
Description	Enter more details of the alert template (up to 64 alphanumeric characters including spaces).
Email recipient	Enter the email addresses to which you want to send alerts.
	Note: Recipients belonging to Base organizations will not receive email alerts, except if the recipient's account includes an MSP license. In general, only the organizations with activated MSP license will receive email alerts.
	For example, ORG 1 is a Base tier organization, and ORG 2 is a Professional tier organization. An MSP alert template is created to monitor AP offline events. If there are three email recipients in both ORG 1 and ORG 2 with the following licenses:
	 REP 1 (recipient 1) has an account which includes an MSP license. REP 2 (recipient 2) and REP 3 (recipient 3) has accounts which does not include an MSP license.
	When an AP offline event occurs, an email alert will only be sent to REP 1 in ORG 1 . While an email alert will be sent to all recipients (REP 1 , REP 2 , and REP 3) in ORG 2 .
Apply to	Select All organizations or specify the selected organizations to send alerts to.
Exclude sites	Select the sites in organizations that will not receive any alerts.

LABEL	DESCRIPTION
Enable	Click this to activate the alert template.
System alerts	
Notification Type	For each alert, you can set how to receive alert notifications:
	• Email: Alert notifications are sent by email to configured recipients.
	In-app Push: Alert notifications are sent to site administrators who are logged into the Nebula Mobile app. This type of notification is not available for some features.
	Both: Alert notifications are sent by email and app notification.
	Disabled: No alerts are sent.
Show additional recipients	Add additional user accounts who will receive email and in-app notifications for the alert.
System Alerts	
Wireless	Specify how long in minutes the NCC waits before generating and sending an alert when an access point goes offline.
Switches	Specify how long in minutes the NCC waits before generating and sending an alert when a port or a switch goes offline.
Security gateway	Specify how long in minutes the NCC waits before generating and sending an alert when the following events occur:
	 A gateway device goes offline. Any DHCP pool on the acteway device runs out of IP addresses to assign
	 A VPN connection to or from the gateway device is created or terminated.
	The WAN connectivity goes offline.
Mobile router	Specify how long in minutes the NCC waits before generating and sending an alert when an mobile router goes offline.
Other	Specify whether to send an alert each time configuration settings are changed.
Security alerts	
CDR containment	Specify whether to send an alert each time a CDR block or containment action is triggered.
Show additional recipients	Add additional user accounts who will receive email and in-app notifications for the alert.
Security Report	
Notification mode	Select whether to receive email security reports from SecuReporter.
Show additional recipients	Add additional user accounts who will receive email and in-app notifications for the alert.
Email subject	Enter an email title here.
Email description	Enter a description of the emails to be sent here. For example, maybe these emails are just for high severity events.
Notification interval	Specify how often to receive a SecuReporter report.
	If no security events were triggered, SecuReporter will not send a report.
Event severity	Select the severity level of events that will be included in each report.
Event threshold	This table lists the events that trigger SecuReporter security alerts.
	You can set the alert threshold. For example, X count(s) of malware/virus attack within 5 minutes means SecuReporter includes a report in the email if the total number of combined malware and virus detection events exceed X within a 5 minute time period.

Table 201	MSP cross ord > MSP	cross ora managa	Alort tomplator >	Croate/Undate alort	(continued)
	10131 CIU33-OIY - 10131	closs-olg mulluge /	Alen lemplules -	Cieule/opuule uleri	(Commoed)

13.5 Admins & Teams

The Admins & teams enables you to assign an administrator or a group of administrators (a team) to multiple organizations at the same time. This is faster than configuring administrators for each organization at **Organization-wide** > **Administrators**, especially if you have a large number of organizations.

13.5.1 Administrator Privilege Priority

You can configure organization administrator privileges on the following screens:

- MSP cross-org > MSP cross-org manage > Admins & teams > Admins
- MSP cross-org > MSP cross-org manage > Admins & teams > Teams
- Group-wide > Group-wide manage > Administrators
- Organization-wide > Administrators

If an NCC account has different administrator privileges configured on different screens, then the highest privilege level takes priority.

Example, account User1 has four different privilege levels configured for organization Org1 on the four screens above: None, Read-Only, Full, Full (Delegate). User1's final privilege level for Org1 is Full (Delegate).

13.5.2 Admins Screen

The admins screen allows you to assign an administrator account to multiple organizations. To access this screen, click MSP cross-org > MSP cross-org manage > Admins & teams > Admins.

Figure 235 MSP cross-org > MSP cross-org manage > Admins & teams > Admins

	Q Search adm	inistrators	• 2 administrators		ि Impor	t + Add
Em	nail address	Orgnization	Orgnization type	Org. privile	Account status	Last a 📃
348	ing@Zoud.ion.iw	Nebula_Org	Nebula Professional Pack	Full	ок	2019-01-14
308	ing@itenationer.ba	Test	Nebula Base Pack	Full	OK	2019-01-14
	En	Email address	Email address Orgnization Nebula_Org Test	Email address Orgnization Orgnization type Nebula_Org Nebula Professional Pack Test Nebula Base Pack	Email address Orgnization Orgnization type Org. privile Nebula_Org Nebula Professional Pack Full Test Nebula Base Pack Full	Email address Orgnization Orgnization type Org. privile Account status Nebula_Org Nebula Professional Pack Full OK Test Nebula Base Pack Full OK

The following table describes the labels in this screen.

Table 202	MSP cross-org > MSP	cross-org manage >	Admins & teams >	Admins
-----------	---------------------	--------------------	------------------	--------

LABEL	DESCRIPTION
Activation	Click this button to Activate/Deactivate the selected accounts. Then, click Update.
Delete	Click this button to remove group administrator privileges for the selected accounts.
Search	Specify your desired filter criteria to filter the list of administrator accounts.
N administrators	This shows the number of administrator accounts (N) in the list.

LABEL	DESCRIPTION
Import	Click this button to create administrator accounts in bulk by importing a complete list of all new administrators in an Excel file. Click template to view the file format.
	Bulk Import X
	"Bulk Import" supports for faster inputting. Please follow this <u>template</u> to import
	Browse Or drag file here
	L
	Close
Add	Click this button to create a new group administrator account.
Name	This shows the name of the administrator account.
Email address	This shows the email address of the administrator account.
Organization	This shows the name of the organization in which the privileges apply.
Organization type	This shows the license tier of the organization.
Org. privilege	This shows the privileges the administrator has within the specified organization.
	Full: the administrator can edit settings, create or delete other administrator accounts, create or delete a site, and add or renew licenses for Nebula Devices in the organization.
	Read-only : the administrator account has no write access to the organization, but can be a site administrator.
	Delegate owner's authority : The administrator account has delegated owner privileges. This type of account can perform all of the same actions as the organization owner, except for the following:
	 Delete organization Transfer organization ownership Assign delegate owner privileges to an administrator account
Account status	This shows whether the administrator account has been validated (OK). It shows Deactivated if an administrator account has been created but cannot be used. This may happen since you can only have up to 5 active administrator account in NCC base tier.
Last access time (UTC)	This shows the last date and time traffic was sent from the administrator account.
Create date (UTC)	This shows the date and time the administrator account was created.
Status change date (UTC)	This shows the last date and time the administrator account status was changed.
Creator	This shows the name of the MSP user account that added the privilege settings.
R	Click this icon to display a greater or lesser number of configuration fields.

Table 202 MSP cross-org > MSP cross-org manage > Admins & teams > Admins (continued)

13.5.3 Create/Update Administrator

In the MSP cross-org > MSP cross-org manage > Admins & teams > Admins screen, click the Add button to add a new administrator account, or double-click an existing account entry to modify the account settings.

Figure 236 MSP cross-org > MSP cross-org manage > Admins & teams > Admins: Create/Update administrator

Create administra	ator			×
Name:				× *
Email address:				× *
Assign privilege:	Organization	Privilege	Activate	
	Select organizations	Full	Yes 🔻	ŵ
	+ Add			
			Close Create o	

The following table describes the labels in this screen.

Table 203 MSP cross-org > MSP cross-org manage > Admins & teams > Admins: Create/Update administrator

LABEL	DESCRIPTION
Name	Enter a descriptive name for the administrator account. Enter up to 100 characters in this field including special characters inside the square quotes $[\sim!@#\$\%^&*()_+{}]:"<>?=[],:',./].$
Email address	Enter the email address of the administrator account, which is used to log into the NCC.
	This field is read-only if you are editing an existing account.
Assign privilege	
Organization	Select one or more organizations to assign the account privileges to. Only organizations belonging to an MSP account with full privileges can be selected.
	Note: If no organization is selected, then the administrator cannot access any organization until an organization is assigned full privileges.
Privilege	Select the privileges the administrator has within the selected organizations.
	Full: the administrator can edit settings, create or delete other administrator accounts, create or delete a site, and add or renew licenses for Nebula Devices in the organization.
	Read-only : the administrator account has no write access to the organization, but can be a site administrator.
Activate	Select Yes to enable the account or No to temporarily disable the account.
1	Click the remove icon to delete the current set of admin privileges.
Add	Add administrator privileges for an organization.
Close	Click this button to exit this screen without saving.
Create admin/ Update admin	Click this button to save your changes and close the screen.

13.5.4 Teams Screen

The team screen allows you to assign administrator privileges to a group of NCC accounts (a team). To access this screen, click **MSP cross-org > MSP cross-org manage > Admins & teams > Teams**.

Figure 237 MSP cross-org > MSP cross-org manage > Admins & teams > Teams

and the second s				
dmins Teams				
Delete Q Sec	irch teams 🔻 🔿	1)teams		+ Ad
Name	Description	Org. privilege	Orgnization	Admini 🗐
Name	Description Tech Writing team	Org. privilege Full	Orgnization Nebula_Org, Nebula_Org2	Admini (

The following table describes the labels in this screen.

LABEL	DESCRIPTION
Delete	Click this button to remove the selected teams.
Search	Specify your desired filter criteria to filter the list of teams.
N teams	This shows the number of teams (N) in the list.
Add	Click this button to create a new administrator team.
	Select an entry's check box to select a specific team. Otherwise, select the check box in the table heading row to select all teams.
Name	This shows the name of the team.
Description	This shows a description of the team.
Org. privilege	This shows the privileges the team has within the specified organizations.
	Full: the administrator can edit settings, create or delete other administrator accounts, create or delete a site, and add or renew licenses for Nebula Devices in the organization.
	Read-only : the administrator account has no write access to the organization, but can be a site administrator.
Organization	This shows the names of the organizations in which the privileges apply.
Administrator	This shows a list of the administrators in the team.
Create date (UTC)	This shows the date and time the team was created.
Status change date (UTC)	This shows the last date and time the team status was changed.
Creator	This shows the name of the MSP user account that added the privilege settings.
R	Click this icon to display a greater or lesser number of configuration fields.

Table 204 MSP cross-org > MSP cross-org manage > Admins & teams > Teams

13.5.5 Create/Update Team

In the MSP cross-org > MSP cross-org manage > Admins & teams > Teams screen, click the Add button to add a new administrator team, or double-click an existing team entry to modify its settings.

Figure 238	MSP cross-ora > MSP	cross-ora manaae >	Admins & teams >	Teams: Create/Update team
inguic 200	11101 CI 033 OI g + 11101	cross org manage -	/ arring a roarns -	rearris. creare, epadre rearri

Create team				×
Name:				× *
Description:				×
Assign privilege:	Full Read-only			
Organizations:	Select organizatio	ons		
Members: ()	Name	Email		
		× *	×	ŵ
		×	× *	ŵ
	+ Add			
			Close	

The following table describes the labels in this screen.

- 10016 203 - M31 C1033-019 / M31 C1033-019 M0101096 / A0MIN & 1601115 / 1601115, C16016/000016 160	Table 205	MSP cross-org >	MSP cross-org manage >	Admins & teams >	Teams: Create/Update tea
---	-----------	-----------------	------------------------	------------------	--------------------------

LABEL	DESCRIPTION
Name	Enter a descriptive name for the team. Enter up to 15 characters in this field including special characters inside the square quotes [$\sim!@#$ \$%^&*()_+{} :"<>?==[]\;',./].
Description	Enter a description of the team, for example their role or membership. Enter up to 64 characters for this field including special characters inside the square quotes [$\sim!@#$ \$%^&*()_+{} :"<>?==[]\:'/].
Assign privilege	Select the privileges the team members have within the selected organizations.
	Full: Each member of the team can edit settings, create or delete other administrator accounts, create or delete a site, and add or renew licenses for Nebula Devices in the organization.
	Read-only: Each member of the team has no write access to the organization, but can be a site administrator.
Organization	Select one or more organizations to assign the team privileges to. An organization can belong to multiple teams.
Members	·
Name	Enter a descriptive name for the members. Enter up to 15 characters for this field including special characters inside the square quotes [$-!@#$ \$%^&*()_+{} :"<>?==[]\;',./].
Email address	Enter the email address of the members who can log into the NCC.
1	Click the remove icon to delete the current set of admin privileges.
Add	Add another NCC account to this team.

Table 205 MSP cross-org > MSP cross-org manage > Admins & teams > Teams: Create/Update team

LABEL	DESCRIPTION
Close	Click this button to exit this screen without saving.
Create/Update	Click this button to save your changes and close the screen.

13.6 Change Log

Use this screen to view logged messages for changes in the Admins & teams and Cross-org synchronization screens. Click MSP cross-org > MSP cross-org manage > Change log to access this screen.

When the log is full, it deletes older entries one by one to make room for newer ones.

Figure 239 MSP cross-org > MSP cross-org manage > Change log

(eyword:		From:		To:		
Q Search	*	Range 🔻 2022	2-01-06 🛗 03:2	26 🕶 2022-01-13	🛗 03:26 🤜	UTC+0 🗵 🔍 Sea
<pre>Colder > Time (UTC)</pre>	1 change logs withi	in the time filtered. Changes Page	date back to 2021-06-21 13:37 (U Label	JTC) Old value		Expo
		Admin & Too	ms Add team			Added: Tech Writing, Description: Technical writi team, Privilege: Full, Organizations: Test July,

The following table describes the labels in this screen.

DESCRIPTION
Enter a keyword or specify one or more filter criteria to filter the list of log entries.
Select a filtering option, set a date, and then click Search to filter log entries by date.
Range: Display log entries from the first specified date to the second specified date.
Before: Display log entries from the beginning of the log to the selected date.
Click this to update the list of logs based on the search criteria.
Click this to return the search criteria to the previously saved time setting.
Click to sort the log messages by most recent or oldest.
This shows the total number of the log messages that match the search criteria. It also shows the date and time the very first log was created.
Click this button to download the log list as a CSV or XML file to your computer.
This shows the date and time in UTC+00:00 (or UTC+0) when the log was recorded.
UTC is a standard time for use around the world (formerly known as Greenwich Mean Time or GMT). UTC is an international abbreviation that is neither French nor English. It means both "Temps Universel Coordonné" and "Coordinated Universal Time".
This shows the name of the NCC menu in which the change was made.
This shows the action that triggered the log entry

Table 206 MSP cross-org > MSP cross-org manage > Change log

NCC User's Guide



LABEL	DESCRIPTION
Old value	This shows the old setting or state that was overwritten with the new value.
New value	This shows the new setting or state.
	Click this icon to display a greater or lesser number of configuration fields.

Table 206 MSP cross-org > MSP cross-org manage > Change log (continued)

13.7 MSP Branding

The **Dashboard logo** section of this screen allows organization owners to replace the Nebula Control Center logo with a new MSP logo. The **Support contact** section allows addition of a customized message or MSP contact information in the **Help > Support** request page. To access this screen, click **MSP crossorg > MSP cross-org manage > MSP branding**.

Figure 240 MSP cross-org > MSP cross-org manage > MSP branding

Dashboard logo 🚺		
Upload new logo:		Upload a logo
	No logo	
Apply to:	All current and new organizations	
	O None	
	O Custom Nebula Org2	
Support contact 🏮	1	
Support contact ① Support request page	Show default Zyxel support cases 🜖	
Support contact 1 Support request page	Show default Zyxel support cases 🜖	mation
Support contact 1 Support request page Apply to:	Show default Zyxel support cases () Customized MSP support contact infor All current and new organizations	mation
Support contact () Support request page Apply to:	Show default Zyxel support cases ① Customized MSP support contact infor All current and new organizations None None	mation

The following table describes the labels in this screen.

Table 207 MSP cross-org > MSP cross-org manage > MSP branding

LABEL	DESCRIPTION
Dashboard logo	·
Upload new logo	Click this to browse for the location of the image file to be used as your dashboard logo.
	 Allowed image file formats: JPG/JPEG, PNG, GIF. Maximum image file size: 200 KB. NCC converts the image file to a 160 x 44 pixel logo after uploading.
Replace this logo	Click this to browse for the location of the image file to replace your current dashboard logo.
Remove this logo	Click this to remove your current dashboard logo.
Apply to	Select All current and new PRO organizations to apply the logo to all Nebula Professional Pack organization dashboards.
	Select Custom to choose which Nebula Professional Pack organization to apply the logo.
	Select None if you only wish to upload the image file but will not apply it yet.
Support contact	
Support request page	
Show default Zyxel support cases	Select ON to display the standard Zyxel support contact information in the Help > Support request screen. Organization owners can choose to hide the default Help > Support screen section to only show their information to clients. But the organization owner and administrators with full privilege will still see the hidden default screen section.
Customized MSP support contact information	Create your own support contact information. Enter up to 1000 characters in this field including special characters inside the square quotes [$\sim!@#$ \$%^&*()_+{} :"<>?==[]\;',./].
Apply to	Select All current and new PRO organizations to apply the support contact information to all Nebula Professional Pack organization Help > Support request screens.
	Select Custom to choose which Nebula Professional Pack organization to apply the support contact information.
	Select None if you only wish to save the settings but will not apply it yet.

PART VI Troubleshooting and Appendices

Chapter 14 Help

14.1 Online documents

Click Help > Online documents to view the documentation for NCC and NCC-compatible devices. For example, to view the Security Firewall Series configuration and hardware information, locate the documents under Firewall.

Figure 241	Help > Online documents		
		Choose the kind of help you need Online documents	
	NCC General Overview > MSP > Group-wide > Organization-wide > Site-wide >	Access Point Monitor in NCC > Configure in NCC > Device GUI (Cloud mode) >	Switch Monitor in NCC > Configure in NCC > Device GUI (Prior to v4.70) > Device GUI (v4.80 and later) >
	Firewall Monitor in NCC > Configure in NCC > Device GUI (Cloud mode) >	Security Router Monitor in NCC > Configure in NCC > Device GUI (Cloud mode) >	Mobile Router

The following summarizes how to navigate the online document screen. The online document screen is divided into these parts:

	Enter search term or phrase D				
B E C	H Introduction				
Getting Started					
NCC Portal Overview	NCC Overview				
Create Organization	The Zyxel Nebula Control Center (NCC) is a cloud-based network management system that allows you to				
Choose Organization	remotely manage and monitor Zyxel Nebula Mobile Routers, Access Points, Ethernet Switches, and Security				
Cloud-Saving Mode	by Nebula. You need to set up a myZyxel account in order to log into the NCC and manage your Nebula				
Setup Wizard	Devices, as discussed in <u>Access the NCC Portal</u> .				
Tutorials	NCC feature support includes:				
MSP	 System accounts with different privilege levels 				
Group-wide	 Site Administrator: manage one site, which is a network that contains Nebula Devices 				
Organization-wide	 Organization Administrator: manage one or more organizations, which are sets of sites 				
Site-wide	Multi-tenant management				
Mobile Router	 Inventory and license management 				
Firewall	Alerts to view events, such as when a device goes down				
Security Gateway	Graphically monitor individual devices Securely page and block a Devices by using the Matwork Configuration Protocol (NETCONEL over TUS)				
Switch	Second mondle report pances by ostig the retwork Costilingtion Liptocol [NELCONE] 046LIP2				
Access Point	The following table describes the supported Nebula Devices.				
Help	Supported Nebula Devices				
Troubleshooting	CATEGORY INCLUDED ZYXEL DEVICES				
	Hybrid Mobile Rauters LTE/NR Indoor/Outdoor Models				
	NSG (Nebula Security Gateway) devices NSG Series				
	Hybrid Security Firewall devices ZyWALL ATP / USG FLEX / USG20(W)-VPN Series				

Figure 242 Online Document Overview

- A Hide/Show the Contents Menu/Index
- B Contents Menu
- C Index
- D-Search Bar
- E Navigation Buttons
- F Google Translate Button
- G Download Content PDF Button
- H Content Page

The following table shows the description of the online document parts.

LABEL	DESCRIPTION
А	Click to hide or show the contents menu and Index.
В	This shows a menu of the content topics. Click a topic heading to display its content in the main screen.
С	Click this to show the Index panel. Click an index entry to view its description.
D	Enter a keyword to search and display the related section(s) in the online document.
E	 These are the navigation buttons. Click the Previous button to display the previous chapter in the online document. Click the Next button to display the next chapter in the online document. Click the Home button to display the first chapter in the online document.
F	Click this to view the translated content page. You can click Google Translate anywhere in a content page, but you must be at the top of the content page to choose a language. The bottom right of the content page has a 'Back to top' arrow to get there.

Table 208 Online Document Overview

Table 208 Online Document Overview (continued)

LABEL	DESCRIPTION
G	Click this to download content in a PDF file. You must be at the top of the content page to click the PDF icon.
Н	The content of the online document is displayed here.

14.2 Troubleshooting Tips

To find suggestions to solve problems you might encounter with NCC and Nebula Devices, go to Chapter 15 on page 597 for more information.

14.2.1 Firewall Information

Click Help > Support tools > Firewall information to view information required for firewall rules to allow management traffic between NCC and Nebula Devices on your sites. Click Export to export the information to a CSV or XML file.

Note: The **Firewall Information** page for a Security Gateway will show its FQDN (fully qualified domain name) and service ports. The FQDN is the complete domain name of Nebula Cloud Management on the Internet.

The following table shows the sample information required for firewall rules at the time of writing.

SERVICE	FQDN	IP ADDRESS	PORT	PROTOCOL
Nebula Cloud Management (NETCONF)	d.nebula.zyxel.com	34.247.112.130, 52.210.12.1, 52.48.115.44, 54.73.103.137, 63.32.141.172, 63.35.107.114	4335 / 6667	TCP
Nebula Cloud Management	s.nebula.zyxel.com	Dynamic	443	TCP
Network Time Protocol	*.pool.ntp.org	Dynamic	123	UDP
Nebula Cloud Management (Zero Touch Provisioning)	d-a.nebula.zyxel.com	Dynamic	443	TCP
Nebula Cloud Management (Configure related service for USG FLEX series)	d-cp.nebula.zyxel.com	34.254.181.105, 52.212.114.133	4335	TCP
Nebula Cloud Management (Monitor related service for USG FLEX series)	d-mp.nebula.zyxel.com	52.18.204.70, 54.220.154.85, 63.34.155.16	443	TCP

Table 209 Sample Information Required for Firewall Rules

14.2.2 Data Policy

Click Help > Support tools > Data Policy to view and download NCC GDPR data policy, privacy policy, and terms of use.

Figure 243 Help > Support tools > Data Policy

Help center > <u>Data policy</u> Data policy			
Nebula Data Policies	Zyxel Privacy Policy Zyxel Nebula GDPR Data Processing Agreement	Nebula Terms of Use	

14.3 Device Function Table

Click Help > Support tools > Device function table to view a list of NCC-compatible Access Points, Switches, Security Gateway, and Security Firewall devices at the time of writing. The table also includes which features each Nebula Device supports.

enter > <u>Device funct</u> ce function table	ion table							
ccess point S	witch Security appliant	ce						
Feature	Model	NWA220AX-6E	WAX620D-6E	WAX640S-6E	WAX655E	NWA110AX	NWA210AX	WAX510D
	Open	•	•	•	•	•	•	•
	Enhanced-Open	•						
	WPA2	•	•	•	•	•	•	•
	WPA2-MIX		•	•	•	•		
Wireless security	WPA3	•	•		•			
	Dynamic personal pre- shared key (DPPSK)	·		•	•			·
	MAC authentication							•
	MAC authentication RADIUS accounting	•	•	•	•			
	Smart mesh	•	•			•	•	•
	Smart mesh manual uplink	•	•	•	•			
Mesh	Smart mesh dedicated band	•	•	•	·	·	·	·
	Wireless bridge		•	•	•			
	Remote AP (RAP) wireless secure tunnel		•	•	•			•
Remote AP (RAP)	Remote AP (RAP) ethernet secure tunnel							
4								

Figure 244 Help > Support tools > Device function table

14.4 Support Forum

Click Help > Still need help? > Support community to go to Zyxel Nebula Community, where you can get the latest Nebula information and have conversations with other people by posting your messages.

14.5 Support Request

If you need Zyxel customer support to help you find answers and/or solve problems, you can submit a ticket through the NCC.

Note: It is suggested that you check this user's guide first to seek help and then go to the Zyxel Nebula Community before you use this screen to send a ticket.

Click Help > Still need help? > Support request to access this screen. The screen varies depending on whether you select to view the ticket details or create a new ticket.

Note: Direct Support for opening a ticket to get direct assistance from the Nebula technical support team is only available for Nebula Pro Pack license.

Zyxel Support Access	Invite Zyxel support as a	dmin	istrat	or	
	By enabling this, you are grantin	g ten	npora	ry ac	cess (21 days by default) to Zyxel support as admin
	check your configuration & logs.	This	will au	utom	atically be switched off after specified days, or you
	might also edit the access privile	eges	here,		
	CSO account will be expired in:	0	21	\$	days
		0	Neve	r	

Figure 245 Help > Still need help?: Support request

lyxel Support Access	Invite Zyxel support as administrator
	By enabling this, you are granting temporary access (21 days by default) to Zyxel support as administrator of your Organization. So they can help
	check your configuration & logs. This will automatically be switched off after specified days, or you could turn it off right after your issue is solved.
	might also edit the access privileges <u>here.</u>
	CSO account will be expired in: • • days
	O Never
Direct Support	You're able to open a ticket to get direct assisstance from the Nebula technical support team.
	Alternately, you can contact your local/regional Zyxel office for support:
	Europe, the Middle East and Africa (EMEA), click <u>here</u> .
	North and Central America, click <u>here</u> .
New Case	
ас. Г	
Subject* :	
Carbon Copy (CC) :	
Device":	v
Issue Description* :	
	h
Priority	low Petinition of priority
Thoney.	
Unload site's topology to s	need up the process:
Add Apother File	
Choose File No file chose	n Delete
Total File Upload Limit 7M	3(The maximum length of a filename is 60 characters)
Cancel Submit	

The following table describes the labels in this screen.

LABEL	DESCRIPTION					
Zyxel Support Access Invite Zyxel support as administrator	Select ON to allow the Zyxel customer support account to access your organization temporarily, so that they can help check your configurations and log messages. At the time of writing, the support account will be deactivated automatically after 21 days. You can set the number of days, or select Never . If you select ON , you can click here to change the support account's name and access right to the organization and sites.					
	Update administrator X					
	Name: Zyxel Support × *					
	Email: nebula.cso@zyxel.com.tw ×					
	Organization access: Read-only					
	Activated: Ves 💌					
	Site Privilege Monitor-only + Add Close Update admin					
My Cases						
C	Click this button to reload the data-related frames for this section on the page.					
Open/Closed	Select to view the details about the tickets that are still open or closed.					
Case Number	This shows the number of the eITS ticket.					
Created	This shows the first date and time the ticket was created.					
Last Updated	This shows the last date and time the ticket was updated.					
Creator	This shows the account name of the administrator that created this ticket.					
Subject	This shows the subject of the ticket.					
Priority	This shows the severity level of the ticket.					
Status	This shows whether the ticket is open or closed.					
Engineer	This shows the name of the support person who handles the ticket.					
New Case	Click this button if you want to issue a new ticket. The following fields then appear allowing you to provide the necessary information and describe the issue encountered.					
Subject	Enter the subject of the ticket.					
Carbon Copy (CC)	Enter the email address of the person you would like to receive a copy of the case.					
Device	Select the NCC or the name of the Nebula Device that cannot work properly.					
lssue Description	Enter a complete and detailed description of your issue.					

Table 210 Help > Still need help?: Support Request

NCC User's Guide

Table 210 Help > Still need help?: Support Request (contin	iued)
--	-------

LABEL	DESCRIPTION
Priority	Select the severity level of the ticket. Click the Definition of priority link to see how to correctly identify a ticket's severity level. This can help to get your problem solved quickly.
Add Another File	Click this button to upload another file.
Choose File/ Browse	Click this button to locate the file you want to upload for reference.
Delete	Click this button to remove the file you just uploaded before submitting the ticket.
Cancel	Click this button to close the New Case section without saving.
Submit	Click this button to send your ticket to the Zyxel customer support.

CHAPTER 15 Troubleshooting

This chapter offers some suggestions to solve problems you might encounter with NCC and Nebula Devices.

- To see how to do things in NCC, go to the Tutorials section.
- To know how to manage Mobile Routers in NCC, go to Section 10.2 on page 471 for more information.
- To know how to monitor Security Appliances in NCC, go to Section 8.2 on page 342 (Security Firewalls) or Section 9.2 on page 414 (Security Gateways) for more information.
- To know how to configure Security Appliances in NCC, go to Section 8.3 on page 350 (Security Firewalls) or Section 9.3 on page 422 (Security Gateways) for more information.
- To know how to monitor Switches in NCC, go to Section 6.2 on page 276 for more information.
- To know how to configure Switches in NCC, go to Section 6.3 on page 288 for more information.
- To know how to monitor Access Points in NCC, go to Section 5.2 on page 235 for more information.
- To know how to configure Access Points in NCC, go to Section 5.3 on page 247 for more information.

I cannot register the Zyxel Device in NCC.

Check if your Zyxel Device supports Nebula by locating the Nebula QR code on the Zyxel Device label or package box.

I cannot access the NCC portal.

- Check that you are using the correct URL:
 - NCC: https://nebula.zyxel.com/
- Make sure your computer's Ethernet card is installed and functioning properly.
- Check that you have Internet access. In your computer, click **Start**, **(All) Programs**, **Accessories** and then **Command Prompt**. In the **Command Prompt** window, type 'ping' followed by a website such as 'zyxel.com'. If you get a reply, try to ping 'nebula.zyxel.com'.
- Make sure you are using the correct web browser that supports HTML5. View the browser in full screen mode to display the NCC portal properly. Browsers supported are:
 - Google Chrome
 - Microsoft Edge
 - Mozilla Firefox

I cannot log into the NCC portal.

Open your web browser and go to *https://nebula.zyxel.com*. Sign in with the correct email and password. Click **Sign Up** if you do not have a myZyxel account and create an account.

I cannot access a Nebula Device that I have registered in NCC.

- Check if the TCP/UDP port is blocked by your network's firewall rule or ISP. Click Help > Support tools >
 Firewall information to view information required for firewall rules to allow management traffic
 between NCC and Nebula Devices on your sites.
- Check the Nebula Device's hardware connections, and make sure the LEDs are behaving as expected. See the Quick Start Guide.
- Make sure the Nebula Device is connected to the Internet.
- For Mobile Routers, make sure a valid SIM card is inserted in the SIM card slot.
- Make sure the Mobile Router is located where the cellular signal is strong.
- For ZyWALL USG FLEX / ATP / USG20(W)-VPN Series devices with Nebula native mode as the deployment method, make sure you perform the steps for Nebula native mode on the Nebula Device; see Section 2.1.7.1 on page 51 for information.
 If you select Zero Touch Provision mode as the deployment method. Make sure you perform the steps for Zero Touch Provision mode on the Nebula Device, see Section 2.1.7.2 on page 51 for information.
- Check if the WAN IP address is configured on the Nebula Device.
- Check if the Nebula Device can access the NCC server's domain through SSH/Console and enter 'nslookup d.nebula.zyxel.com'. If the Nebula Device shows 'unknown host', check your DNS server setting or use '8.8.8.8' as the DNS server on the Nebula Device.
- The Nebula Devices will apply the site-wide password after getting online on NCC. Check the login credential by going to Site-wide > Configure > Site settings: Local credentials.
- Specify the **Port** number and click **Establish** using **Remote Access** in the following screens to obtain real-time logs and data from the Nebula Device.
 - Site-wide > Devices > Firewall
 - Site-wide > Devices > Security gateway
 - Site-wide > Devices > Access points

Note: Remote Access to Nebula Access Points is available to the organization owner,

organization administrators with full privileges, and site administrators with full privileges in Nebula Pro Pack license only.

Remote Access to Nebula Security Firewalls and Security Gateways is available to the organization owner in Nebula Pro Pack license only.

I cannot see my Nebula Devices in the NCC Dashboard or the corresponding Nebula Device monitor page.

- If your Nebula Device is a Zyxel Hybrid Switch (GS / XGS / XMG / XS Series), make sure that the Nebula Device is working in Nebula cloud management mode with NCC Discovery enabled.
 - For the Web Configurator version 4.70: Active is enabled in Basic Setting > Cloud Management > Nebula Control Center Discovery.
 - For the Web Configurator version 4.80: Nebula Control Center (NCC) Discovery is enabled in SYSTEM > Cloud Management.
- Make sure that your Nebula Device can connect to the NCC by checking your network's firewall/ security settings. The following ports must be allowed:
 - TCP: 22, 443, 4335 and 6667
 - UDP: 123

Note: Go to Help > Support tools > Firewall information to find the latest port information.

- Make sure that you have registered your Nebula Devices with the NCC. See Section 11.2.1 on page 492.
- Make sure that you have created an organization and site and add the Nebula Devices to the site. See Create Organization on page 43.

My organization is now in Cloud-saving mode; how can I disable it?

There are two ways to disable Cloud-saving mode.

- Click the **Cloud-saving mode** switch in the **Welcome back** pop-up window. Then click **Close** to turn off Cloud-saving mode for the organization.
- A banner displays when NCC is in Cloud-saving mode. Click the You could change mode <u>here</u> link in the NCC banner. Click the Cloud-saving mode switch in the Cloud-saving mode pop-up window. Then click Close to turn off Cloud-saving mode for the organization.

I want to place my Nebula Device on the right location on Google maps.

If your Nebula Device has a public IPv4 address, Google Maps can use Geo IP to approximatively locate your Nebula Device. If your Nebula Device has an IPv6 address or a private IPv4 address or you want locate the Nebula Device more exactly, use one of the following methods.

- Select Use the following address or coordinates to enter the complete address or coordinates of the Nebula Device in Site-wide > Devices > Firewall / Security gateway / Switches / Access points: details: Map: Position device.
- Select Get my location from web browser to use the public IP address of the computer accessing the NCC portal.
- Drag-and-drop your Nebula Device directly on the Google map.

I cannot set up Secure WiFi in NCC.



- Make sure the Nebula Security Firewall and Nebula Access points are in the same NCC site.
- Make sure a Secure WiFi license is assigned to the Nebula Security Firewall.
- Make sure to configure the **Remote AP Setting** of each Remote Access Point before booting up the Remote Access Point in the remote site. See Table 14 on page 161.
- The maximum number of Remote Access points depends on the Nebula Security Firewall.

Table 211 Maximum Remote Access Points (at the time of writing)

CAPACITY	USG FLEX 50 / USG20-VPN / USG20W-VPN	USG FLEX 100 / USG FLEX 100W / ATP100 / ATP100W	USG FLEX 200 / ATP200	USG FLEX 500 / ATP500	ATP700	USG FLEX 700 / ATP800
Maximum IPSec Tunnel	10	40	90	250	450	450
Maximum Remote AP	No support	6	10	18	66	130

The mesh extender does not appear online on Status in Site-wide > Devices > Access points.

- Click Reconnect in Site-wide > Devices > Access points: Uplink AP to re-establish connection.
- Make sure your Nebula Device supports smart mesh. To view the list of Nebula Devices that support smart mesh, go to Help > Device function table.

After adding a mesh extender to a site, the mesh extender cannot connect to a mesh controller.

• Make sure you enable AP Smart Mesh in Site-wide > Configure > Access points > AP & port settings. See Section 5.3.7 on page 271 for more information.

Note: For more information about smart mesh, see Section 5.1.1 on page 233.

The mesh extender does not broadcast the mesh controller SSID.

 Make sure you enable Downlink in Site-wide > Monitor > Access points: Details. See Section 4.3.1.1 on page 159 for more information.

None of the Nebula Device LEDs turn on.

- Make sure that you have the power cord connected to the Nebula Device and plugged in to an appropriate power source. Make sure you have the Nebula Device turned on.
- Check all cable connections. See the related Quick Start Guide.
- If the LEDs still do not turn on, you may have a hardware problem. In this case, you should contact your local customer support.

The Nebula Device PWR LED is red.

- The Nebula Device has a power-related error. Disconnect and reconnect the power cord. Make sure that you are using the included power cord for the Nebula Device and it is plugged into an appropriate power source. See the related Quick Start Guide.
- If the LED is still red, you may have a hardware problem. In this case, you should contact your local customer support.

15.1 Getting More Troubleshooting Help

Go to *support.zyxel.com* at the Zyxel website for other technical information on the NCC.

15.2 NCC Live Chat

Clicking the **Ask Question** button at the bottom of NCC window prompts you to search for a solution on the Zyxel forum, and then connects you to a Zyxel technical support agent. If a technical support agent is not available, you can fill in a form to send your question to Zyxel by email.



Note: This is an NCC Professional Pack feature.

Live chat might be limited to a certain number of hours per day. The time that live chat is available varies depending on your country.

APPENDIX A Customer Support

In the event of problems that cannot be solved by using this manual, you should contact your vendor. If you cannot contact your vendor, then contact a Zyxel office for the region in which you bought the device.

For Zyxel Communication offices, see *https://service-provider.zyxel.com/global/en/contact-us* for the latest information.

For Zyxel Network offices, see *https://www.zyxel.com/index.shtml* for the latest information.

Please have the following information ready when you contact an office.

Required Information

- Product model and serial number.
- Warranty Information.
- Date that you received your device.
- Brief description of the problem and the steps you took to solve it.

Corporate Headquarters (Worldwide)

Taiwan

- Zyxel Communications (Taiwan) Co., Ltd.
- https://www.zyxel.com

Asia

China

- Zyxel Communications Corporation–China Office
- https://www.zyxel.com/cn/sc

India

- Zyxel Communications Corporation–India Office
- https://www.zyxel.com/in/en-in

Kazakhstan

- Zyxel Kazakhstan
- https://www.zyxel.com/ru/ru

Korea

- Zyxel Korea Co., Ltd.
- http://www.zyxel.kr/

Malaysia

- Zyxel Communications Corp.
- https://www.zyxel.com/global/en

Philippines

- Zyxel Communications Corp.
- https://www.zyxel.com/global/en

Singapore

- Zyxel Communications Corp.
- https://www.zyxel.com/global/en

Taiwan

- Zyxel Communications (Taiwan) Co., Ltd.
- https://www.zyxel.com/tw/zh

Thailand

- Zyxel Thailand Co., Ltd.
- https://www.zyxel.com/th/th

Vietnam

- Zyxel Communications Corporation–Vietnam Office
- https://www.zyxel.com/vn/vi

Europe

Belarus

- Zyxel Communications Corp.
- https://www.zyxel.com/ru/ru

Belgium (Netherlands)

- Zyxel Benelux
- https://www.zyxel.com/nl/nl
- https://www.zyxel.com/fr/fr

Bulgaria

• Zyxel Bulgaria

https://www.zyxel.com/bg/bg

Czech Republic

- Zyxel Communications Czech s.r.o.
- https://www.zyxel.com/cz/cs

Denmark

- Zyxel Communications A/S
- https://www.zyxel.com/dk/da

Finland

- Zyxel Communications
- https://www.zyxel.com/fi/fi

France

- Zyxel France
- https://www.zyxel.com/fr/fr

Germany

- Zyxel Deutschland GmbH.
- https://www.zyxel.com/de/de

Hungary

- Zyxel Hungary & SEE
- https://www.zyxel.com/hu/hu

Italy

- Zyxel Communications Italy S.r.l.
- https://www.zyxel.com/it/it

Norway

- Zyxel Communications A/S
- https://www.zyxel.com/no/no

Poland

- Zyxel Communications Poland
- https://www.zyxel.com/pl/pl

Romania

- Zyxel Romania
- https://www.zyxel.com/ro/ro

Russian Federation

- Zyxel Communications Corp.
- https://www.zyxel.com/ru/ru

Slovakia

- Zyxel Slovakia
- https://www.zyxel.com/sk/sk

Spain

- Zyxel Iberia
- https://www.zyxel.com/es/es

Sweden

- Zyxel Communications A/S
- https://www.zyxel.com/se/sv

Switzerland

- Studerus AG
- https://www.zyxel.com/ch/de-ch
- https://www.zyxel.com/fr/fr

Turkey

- Zyxel Turkey A.S.
- https://www.zyxel.com/tr/tr

UK

- Zyxel Communications UK Ltd.
- https://www.zyxel.com/uk/en-gb

Ukraine

- Zyxel Ukraine
- https://www.zyxel.com/ua/uk-ua

South America

Argentina

- Zyxel Communications Corp.
- https://www.zyxel.com/co/es-co

Brazil

• Zyxel Communications Brasil Ltda.

https://www.zyxel.com/br/pt

Colombia

- Zyxel Communications Corp.
- https://www.zyxel.com/co/es-co

Ecuador

- Zyxel Communications Corp.
- https://www.zyxel.com/co/es-co

South America

- Zyxel Communications Corp.
- https://www.zyxel.com/co/es-co

Middle East

Israel

- Zyxel Communications Corp.
- https://il.zyxel.com

North America

USA

- Zyxel Communications, Inc. North America Headquarters
- https://www.zyxel.com/us/en-us



APPENDIX B Legal Information

Copyright

Copyright © 2023 by Zyxel and/or its affiliates.

The contents of this publication may not be reproduced in any part or as a whole, transcribed, stored in a retrieval system, translated into any language, or transmitted in any form or by any means, electronic, mechanical, magnetic, optical, chemical, photocopying, manual, or otherwise, without the prior written permission of Zyxel and/or its affiliates.

Published by Zyxel and/or its affiliates. All rights reserved.

Disclaimer

Zyxel does not assume any liability arising out of the application or use of any products, or software described herein. Neither does it convey any license under its patent rights nor the patent rights of others. Zyxel further reserves the right to make changes in any products described herein without notice. This publication is subject to change without notice.

Viewing Certifications

Go to http://www.zyxel.com to view this product's documentation and certifications.

Zyxel Limited Warranty

Zyxel warrants to the original end user (purchaser) that this product is free from any defects in material or workmanship for a specific period (the Warranty Period) from the date of purchase. The Warranty Period varies by region. Check with your vendor and/or the authorized Zyxel local distributor for details about the Warranty Period of this product. During the warranty period, and upon proof of purchase, should the product have indications of failure due to faulty workmanship and/or materials. Zyxel will, at its discretion, repair or replace the defective products or components without charge for either parts or labor, and to whatever extent it shall deem necessary to restore the product of equal or higher value, and will be solely at the discretion of Zyxel. This warranty shall not apply if the product has been modified, misused, tampered with, damaged by an act of God, or subjected to abnormal working conditions.

Note

Repair or replacement, as provided under this warranty, is the exclusive remedy of the purchaser. This warranty is in lieu of all other warranties, express or implied, including any implied warranty of merchantability or fitness for a particular use or purpose. Zyxel shall in no event be held liable for indirect or consequential damages of any kind to the purchaser.

To obtain the services of this warranty, contact your vendor. You may also refer to the warranty policy for the region in which you bought the device at *http://www.zyxel.com/web/support_warranty_info.php*.

Registration

Register your product online at www.zyxel.com to receive email notices of firmware upgrades and related information.

Index

Symbols

+Hub button 562 +Org-to-Org Service 562

Numbers

2FA 27 enable 104 set up 110 3GPP specifications 485 802.11k neighbor lists 255, 338 802.11r fast roaming 250 802.11v capable client 255, 338 802.1X (WPA-Enterprise) authentication 221, 543 802.1X authentication 250

Α

access point connection status 158 access point list export 157 Access points screen 156 access port 293 account disable 566, 583 expire 547, 550 account list export 220, 222, 224, 543, 546, 549 account status 510, 565, 582 ACL rule configure 296 ACL screen 296 Active license state 504 Active Directory 405, 466

AD server 405, 466 add a device tutorial 58 Add devices screen 495, 496 Add licenses screen 496 Add more devices action 493 Add more licenses action 493 address gateway 188 administrator create 582 delete 99 privilege 581 update 582 administrator account password 143 administrator accounts 508 Administrator screen 510, 565 Administrators screen 508 Admins & teams 581 Admins screen 581 advanced 534 AES encryption algorithm 376 Alert Settings screen 210 ALG 411 allow list 393, 531, 534 antenna orientation 164 antenna switch 164 anti-malware 533 Anti-Malware signature 225 AP 267 AP & Port Settings screen 271 AP Google Map Dashboard 154 AP photo 166 AP Smart Mesh 233 **AP Status** Dashboard 153 **AP** Traffic

NCC User's Guide

Dashboard 154 AP/switch setting overwrite 90 API token generate 232 App ID 251 Appliance Clients (by Usage) Dashboard 154 **Appliance Network Applications** Dashboard 154 **Appliance Status** Dashboard 154 Application Layer Gateway, see ALG application management profile add 327 application patrol 379, 530 application patrol profile 444 add 380 Applications screen 200 APs (by Usage) Dashboard 154 ARP packet 130 Ask Question button 601 assign administrator tutorial 76 Assign License action 501, 502, 503, 504 assisted roaming 255, 337, 338 ATP device license 17 trial license 505 Authentication Method screen 402 Authentication screen 307 Automatic Power Save Delivery 255, 338 auto-negotiation 289

В

Backup & restore screen 489 backup code 29 download 29 inactive 29 band selection

WiFi 337 bandwidth utilization 177 Base Station Identity Code (BSIC) 486 Base tier 17 base transceiver station 484 Batch Create DPPSK 551 battery life 255, 338 Beta firmware type 159, 169, 218, 540 binding table 131 block list 393, 531, 534 blocked client release 205 Branch to Branch VPN enable 560 bridge priority 313 set 313 browser supported 597 browser support 24 BSIC 486 bundled license 23 bypass mode 251

С

captive portal 399, 458 Captive portal customization screen 256 Captive portal screen 399, 458 captive portal setting SSID profile 256 carrier aggregation (CA) 484, 486 CDR configure 140 tutorial 140 CDR (Collaborative Detection & Response) 17 CDR containment 142 CDR security service 205 cell ID 484 Cell Identity 484 cellular band 484 Cellular Info screen 481 cellular IP passthrough

setting 475 supported model 475 certificate import 555 update 555 certifications viewing 607 Change log screen 507, 553, 566, 586 Change organization action 501, 502, 503, 504 change owner 509 Change site assignment action 501, 502 channel bandwidth 242, 243 channel ID 182, 185 channel list 128 channel profile and naming configure 127 Channel Quality Indicator (CQI) 486 channel width 264, 339 Classification mode 175 client IP traffic block 140 client statistics display 194 client steering 265 Clients list screen 194 Clients screen 192 Cloud Authentication account privilege 574 Cloud Authentication screen 219, 541 Cloud authentication screen 102 cloud mode 143 cloud sandboxing 534 Cloud-Saving mode 45 **CNP** security service enable 93 Collaborative Detection & Response (CDR) 22, 225 Collaborative detection & response screen 141 configuration backup 517 configuration management 555 configuration synchronization 517 configuration template 520 manage 79 tutorial 79

Configuration templates screen 520 Configure menu 207, 247 Connect & Protect (CNP) license 17 Connect & Protect (CNP) trial license 505 Connect & Protect Plus (CNP+) 17 connected device summary 512 connection speed port 188 connection status port 188 Connection Test button 144 connectivity 516 Consumption mode 175 contact information 602 containment action 227 Containment List screen 205 Content Filter Pack license 17 Content Filter Pack trial license 505 content filtering 379, 385, 446, 530 content filtering profile add 381 controller mesh 234 Coordinated Universal Time (UTC) 48 copyright 607 CPU usage 185, 189 CQI 486 CRC (Cyclic Redundant Check) error 278, 279 CRC error 178 create user account 547, 550 Create Group window 557 Create organization screen 43 create user account 544 Critical firmware status 218, 536, 540 cross-org site clone 574 cross-org sync 574 Cross-org synchronization screen 575 CSS value 400, 460 Custom firmware status 218, 540 custom portal page 258, 460 custom theme 257, 460

customer support 595, 602 Cyclic Redundant Check 178

D

dark mode 35 dark web blocking 325 Dashboard logo 587 specs 588 dashboard logo upload/replace/remove 14 Dashboard screen 31, 152 change default view 62 Data Policy screen 592 data roaming 483 DCS time interval 264 DCS (Dynamic Channel Selection) 242, 243 DDMI 179 Dedicated firmware type 159, 169, 218, 540 Defend Center 534 Delegate owner's authority privilege 565 delete icon 489 **Deployment Method** Setup Wizard 146 deployment method 501, 502, 504 set up 51 DES encryption algorithm 376 description gateway 188 destination lookup failure (DLF) 163 device add to site 48 view 34 Device function table 13, 233, 276, 471 Device function table screen 593 device list export 501, 516, 563 Device screen 500 Devices menu 156 Devices screen 215, 538 Devices tab 515

DH key 454 DHCP relay 429, 434 DHCP server 429, 434 DHCP server guard 314 DHCP service 429, 434 Differentiated Services Code Point (DSCP) value 314 Diffie-Hellman key group 454 Diffie-Hellman key group (DHx) 336, 370, 376 Digital Diagnostics Monitoring Interface 179 disable account 511, 566, 583 disclaimer 607 DNS content filtering enable 382 DNS lookup 480 DNS setting 340, 405, 466 DNS/URL threat filter 530 domain allow or block list 326, 329 domain name 480 domain zone 407, 468 download icon 489 **DPPSK** account add/edit 550 batch creation 551 DPPSK screen 223, 548 DPPSK third-party integration 232 Dynamic Personal Pre-Shared Key (DPPSK) 20, 221, 224, 239, 251, 543, 545, 550, 551, 552 print 224, 549 DynDNS account 426

Ε

EARFCN (E-UTRA Absolute Radio-Frequency Channel Number) 484, 485 EcNo 485 Edit floor plans tab 191 eITS ticket 595 email alert notification 212 email address user account 224 email recipient MSP alerts 579 email report 281, 287, 348, 417, 421 Email Verification 29 encryption algorithm 337, 371, 376 encryption method SSID network 204 eNodeB (Evolved Node-B) 486 Ethernet WAN status 480 E-UTRA Absolute Radio-Frequency Channel Number 484, 485 event log 235, 276, 316, 342, 414 Event Log screen 235, 276, 316, 342, 414 Expired license state 504 expired license renew 63 extender mesh 234

F

Facebook app ID 251 Facebook fan page 251 Facebook login 251 Facebook WiFi 251 factory-default setting reset to 143 fan page Facebook 251 fast roaming 250 firewall information export 592 port information 599 Firewall information screen 592 firewall rule 440 default 330, 377 Firewall screen 182, 330, 377, 444 **Firewall settings** Dynamic DNS screen 409 Firewall settings screen 405 firmware maintain 74 upgrade 49 Firmware management screen 212, 535 firmware upgrade priority 536 schedule 212, 536, 537

Firmware upgrade screen 496 firmware version view 476 floor plan 190 force logout 509, 564 FQDN 341, 407, 468, 561 FQDN (Fully-Qualified Domain Name) 592 Full (read and write) access 512 full access 510 Full privilege 565, 566 Fully-Qualified Domain Name 341, 407, 468 Fully-Qualified Domain Name (FQDN) 560

G

gateway log message 414 Gateway settings screen 466, 469 General Availability firmware type 159, 169, 218, 540 get started 24 Gold Security Pack license 17 Gold Security Pack trial license 505 Good firmware status 218, 536, 540 Google Authenticator app 28 install 111 Google authenticator passcode 104 Google map 188, 477 client location 196 site location 512 grace period organization license 23 group create 557 definition 557 Group Administrator account 564 Group list 558 Group-wide overview 558 Group-wide menu 558 guest ambassador access 510, 512
Guest interface 425, 432 guest VLAN 308 guest WiFi network 50

Η

hash algorithm 377 Hit for AP Network IP Reputation Filter Dashboard 154 Hit for Collaborative Detect & Response Dashboard 154 Home networking setting 473 hub 523 add 562 hub to hub VPN 560, 561 hub-and-spoke topology 523, 560 Hub-and-Spoke VPN 451

ICCID 483 icon delete 489 download 489 More 33 pause 480 photo remove 479 play 480 idle timeout 555 IDP 385, 446 IEEE 802.11k/v 255, 338 IEEE 802.11r 250 IEEE 802.3af Power over Ethernet standard 294 IEEE 802.3at High Power over Ethernet standard 294 IGMP filtering profile 296 IGMP Group-Specific Query (GSQ) message 296, 306 IGMP multicast groups 281 IGMP query port 296 IGMP snooping 302 IGMP status 172 IKE (Internet Key Exchange) 373, 454

IKE SA 334, 369 IKE SA (Security Association) 453 IKE version 103, 335, 370 image maximum file size 240 upload/replace/remove 240 IMEI 482 import certificate 555 IMSI 483 Inactive license state 504 in-app push alert notification 212 Install wizard action 493 installation wizard 497 Installer access 512 installer privilege 510 instant messenger (IM) 380, 444 Insufficient Licenses status 23 Integrated Circuit Card Identifier (ICCID) 483 Interface addressing screen 422 Interface screen 319, 351 Interfaces addressing Local LAN screen 435 Interfaces addressing screen 437 internal antenna 164 International Mobile Equipment Identity (IMEI) 482 International Mobile Subscriber Identity (IMSI) 483 Internet access voucher 236 Internet Information Server (IIS) 258 Internet Protocol Security 456 Internet Service Provider 483 intra-BSS traffic blocking 247, 255 Intrusion Detection and Prevention 385, 446 Intrusion Detection and Prevention (IDP) 344, 416, 419 intrusion detection/prevention 535 Inventory screen 563 IP & Routing screen 298 **IP** address add 560 client 197

IP passthrough 483 IP source guard setup 130 IPS (IDP) signature 225 IPS (Intrusion Prevention System) 17 IPSec 456 IPSec SA 337, 371, 376 **IPSec** tunnel maximum 600 IPSec VPN 453 IPSec VPN client enable remote access VPN rule 103 IPSec VPN tunnel 494 IPTV manage 123 set up VLAN 124 IPTV channel 38, 127, 281 IPTV client 281 IPTV report 38 IPTV report screen 279 IPTV topology setup 125 IPTV traffic 281 IPv4 address gateway 488 LAN station 488 ISP (Internet Service Provider) 483

Κ

key size 376

L

L2 isolation 247 L2TP 456 L2TP VPN 456 LAC 486 LAI (Location Area ID) 486 LAN interface configuration DHCP option screen 358 LAN interface configuration screen 322, 355 LAN stations screen 487 LAN usage view 476 language select 35 Latest firmware type 159, 169, 218, 540 Layer 2 Tunneling Protocol 456 layer-2 isolation 255 leave mode fast 296 normal 296 license activate/assign 58 activation 59 assign to Nebula Device in new organization 66 Circle 494 feature difference 18 general information 23 monitor expiration 62 payment method 559 purchase 494 states 59 transfer 64, 98 transfer to a different organization 65 transfer to a Nebula Device in same/new organization 68 undo assign 65 validity 23 license concept 15 License Expired status 23 license list export 504 License screen 24, 502 license summary table 16 license tier organization 17 License transfer action 502, 503, 504 license types 59 licenses purchase export 494, 501 Link Aggregation Control Protocol (LACP) 434 Link Aggregation Group (LAG) 430 Link Layer Discovery Protocol 177 Link Layer Discovery Protocol (LLDP) 289 live tools 479 LLDP 177

NCC User's Guide

LLDP (Link Layer Discovery Protocol) 197 LLDP-MED protocol 314 load balancing 162, 273 load balancing method 465 Local credentials password 140 local override 520, 522 enable 88 switch 523 Location Area Code (LAC) 486 Location Area ID 486 locator LED 174 log full 566, 586 log list export 206, 507, 553, 567, 586 log message 235, 276 login account menu 33 login information 105 login page 257 login password 140 logo remove 588 replace 588 upload 588 loop guard 290

Μ

MAC account create/update 547 MAC address client 197 gateway 188 LAN station 488 port 188 user account 222 WiFi station 488 MAC authentication 250 MAC screen 221, 545, 547 Managed Service Provider (MSP) 14, 570 management control enable 290 management port 293 management VLAN 313 map 190 pin a device 190 Map & floor plans screen 190 Max Power (mW) 175 MCC 483 MCS 486 Media Independent Interface (MII) 434 Memory usage 185, 189 merged privilege 510 mii (Media Independent Interface) 434 MNC 483 Mobile Country Code (MCC) 483 Mobile Network Code (MNC) 483 Mobile router Dashboard 154 mobile router restart 480 Mobile router screen 189 modulation codina scheme 486 Monitor menu 204, 235, 276, 316, 342, 414 Monitor-only access 510, 512 More icon 33 MSP introduction 14 MSP administrator 576 MSP alerts create 578 email recipient 579 notification type 580 update 578 MSP alerts screen 576 MSP branding 587 MSP license 16, 570 activate 93 MSP Pack trial license 505 MSP portal 570 MSP Trial license 16 My RADIUS server 251 myZyxel account 13, 15, 24 email address 29, 70 myZyxel portal 72

Ν

N/A firmware type 159, 169, 218, 536, 540 name port 188 user account 222, 224 NAS 413, 469 NAS identifier 252 NAS IP address 413, 469 NAT rule 363 NAT screen 364 Native mode 502, 504 NCAS 253 NCAS (Nebula Cloud Authentication Server) 403 NCC access 24 account settings 36 alert 34 change device owner 43 create organization 43 dark mode 35 Dashboard 152 display language 35 example network 15 features 13 license expiration 559, 573 license status 559 live chat 601 log message view 34 login 24 menu 36 notification 34 organization 14 overview 13, 57 portal website 24 sample network topology 15 settings icon 34 site 14 two-factor authentication 27 NCC license 494 category 16 status 572 NCC logo replace 587 NCC menu summary 37 NCC portal

access 24 no access 597 overview 31 parts 31 title bar 31 NCC, Nebula Control Center 13, 57 Near Expiring status 23 Nebula account login 30 Nebula cloud authentication select 251 Nebula Cloud Authentication Server 253 Nebula configuration 145 Nebula Device location 477 power consumption 175 remove 97 status 63 Nebula Device list export 168 Nebula Device port list export 289 Nebula managed device connect 24 Nebula management mode select 143 Nebula Mode select 143 Nebula native mode 146 deploy 142 deployment method 51 Nebula portal 145 Nebula SD-WAN 26 Nebula Security Service 38, 344, 416 Nebula Security Service (NSS) license 16 Nebula Smart Mesh 233, 273 NETCONF 13 NETCONF over TLS 13 Network Access Server 413, 469 Network Access Server identifier 252 Network address translation (NAT) 424 Network Configuration Protocol (NETCONF) 13 network connection check 140 Network Interface Card (NIC) PXE-capable 358

network topology 155 fully-meshed 523 Network usage and connectivity screen 489 next hop 360, 438 NSG device license 16 NSS 38, 344, 416 NSS Analysis Report screen 416 NSS/UTM license 494

0

online document parts 590 Online documents screen 590 ONVIF (Open Network Video Interface Forum) 300 **ONVIF** discovery configure 301 ONVIF discovery screen 301 Orchestrator 26 Orchestrator Management 26 organization choose 44 create 43, 47, 145 create new 27 delete 97, 100 privilege 584 summary view 33 organization (delegated) privilege 510 organization access 511, 566 organization administrator 13, 15 organization license 16 grace period 23 organization name change 73 Organization Trial license 16 organization-wide schedule firmware upgrade 539 Org-to-Org Service 560 Org-to-Org VPN 561, 568 configure 559 example 560 Org-to-Org VPN screen 560 OSI (Open System Interconnection) 385, 447

OUI (Organizationally Unique Identifier) 314 output power radio 263 override site-wide configuration enable 88 Overview screen 212, 492, 512, 535 owner change 509 owner privilege 510

Ρ

passcode 2FA 112 password administration account 143 login 229 reset 69 pause icon 480 payment method license 559 PCC 484 PCI 483, 484 PD priority 294 peer-to-peer (P2P) 380, 444 Perfect Forward Secrecy 455 Perfect Forward Secrecy (PFS) 337, 371, 377 Personal Identification Number 483 PFS 455 photo upload 477 photo remove icon 479 Physical Cell ID (PCI) 483, 484 PIN (Personal Identification Number) 483 PIN code 483 Pin Unlock Key 483 ping perform 480 play icon 480 PLMN (Public Land Mobile Network) 483 Plus Pack license 16 Plus tier 17 PMI 486

PoE 309 PoE mode 175 PoE Power Dashboard 154 PoE schedule 294 PoE schedules screen 309 PoE status 172 policy route 359, 438 Policy Route screen 438 port disabled 313 gateway 188 port group 350, 424 port isolation 292 port mirroring 177, 313 Port screen 350 port security 308 port setting Nebula AP 271 port VLAN ID 290, 293 power consumption Nebula Device 175 port 178, 287 power management mode 172, 175 Power over Ethernet 309 Power over Ethernet (PoE) 155 power-up mode 294 Precoding Matrix Indicator (PMI) 486 pre-shared key 334, 369, 451, 527 pre-shared key (PSK) 208, 250 primary component carrier (PCC) 484 privilege 510 admin in organization 565 administrator 581 assign 585 merged 510 organization 582, 584 privilege priority 581 Pro Pack license 16 Pro Pack trial license 505 Pro tier 17 problems 597 product registration 607 profile switch 523

protocol ID 362 Public Land Mobile Network 483 PUCCH (Physical Uplink Control Channel) 486 PUK (Pin Unlock Key) 483 purchase history export 508 Purchase History screen 508 Purchase license action 501 PUSCH (Physical Uplink Shared Channel) 486 PVID 163, 293 PXE (Preboot eXecution Environment) 358

Q

QR code 28, 111 QSG (Quick Start Guide) Nebula Device 140 Quality of Service (QoS) 313 Quarantine screen 228 quarantine VLAN 228 Queued license state 504

R

RAC (Routing Area Code) 486 radio output power 263 Radio Frequency Channel Number 484, 485 radio setting AP 261, 338 Radio settings screen 261, 338 RADIUS accounting 252 RADIUS server 252, 466 RAI (Routing Area ID) 486 Rank Indication 486 rate limiting 210 read and write (Full) access 511 read and write access 510 read-only access 510, 511

Read-only privilege 565, 566 reboot mobile router 480 Received Signal Code Power 485 recurring schedule firmware upgrade 217, 539 Reference Signal Receive Power (RSRP) 484, 485 Reference Signal Receive Quality (RSRQ) 484, 485 Reference Signals (RS) 484, 485 register a device 37 registration product 607 remote access VPN setup 101 Remote access VPN screen 103, 371 remote AP maximum number of 600 Remote AP (RAP) 408 remote AP feature enable 162 remote configurator mobile router 480 Remove from organization action 501, 502 renewal license 24 repeater AP 234 RESET button 143 Resource Element (RE) 484, 485 restore configuration 520 RFCN 484, 485 roaming mobile router 483 root AP 234 root bridge 172 Routerl settings screen 340 Routing Policy Routes/Traffic Shaping screen 360 Static Route screen 362 Routing Area Code 486 Routing Area ID 486 RSCP 485 RSRP 484, 485 RSRQ 484, 485 RSSI 484 RSSI (Received Signal Strength Indicator) 488

RSTP status 172

S

sandboxing 534 schedule firmware upgrade 41, 42, 212, 535 schedule profile add 446 schedule template 261, 310 SCR Pro Pack license 17 SD-WAN license 26 search for NCC-managed device 33 SecuExtender license key activate 105 SecuExtender software install 105 Secure WiFi license 17, 494 Secure WiFi trial license 505 SecuReporter screen 344 Security Alert Dashboard 154 Security Gateway screen 186 security log display 480 security policy 331, 378 security profile sync 530 configure 527 Security Profile Sync (SPS) 385 Security profile sync screen 527 Security router screen 180 Security Service license 16 Security service screen 268, 385 security services 189 serial number 168 gateway 188 Server-and-Client VPN 451 service provider 483 Service Set IDentifier 207 Settings screen 554, 568 Setup Wizard 139, 145 setup wizard 30, 46 steps 46

Setup Wizard screen 143 severity level 596 SFP (Small Form Factor Pluggable) port 179 signal strength mobile router 481 Signal to Interference plus Noise Ratio (SINR) 484, 486 SIM card install 140 status 483 single sign-on (SSO) 545 SINR 484, 486 SIP ALG 411 site 523 create 47, 145 delete 98 summary 512 site administrator 13, 15 site binding 521 Site features logs screen 206 site list export 514 site name change 73 Site settings screen 229 site tag summary 512 Site tags tab 514 Sites tab 513 Site-to-Site VPN screen 332, 366, 449, 452 site-wide schedule firmware upgrade 217 Smart Alert Engine 211, 577 smart client steering 267 Smart Mesh 164, 233 network topology 234 wireless hops 234 Smart VPN 559 spanning tree 172 SSID 49, 207 mobile router 488 SSID availability screen 260 SSID network encryption method 204 SSID overview screen 207, 337

SSID profile 207 settings 247, 337 SSID schedule 260 SSID settings screen 247, 337 SSIDs (by Usage) Dashboard 154 Stable firmware type 159, 169, 218, 540 static route 359, 438 status voucher 238 submit ticket 593 Summary Report screen 202, 244, 285, 345, 419 support contact detail 14 support account 595 Support contact 587 Support forum 593 Support Request screen 594 supported browsers 597 supported Nebula Devices 13, 233, 276, 471 Surveillance screen 277 switch define role 125 switch connection status 168 Switch ports screen 291 Switch settings screen 310 Switch Status Dashboard 153 Switches screen 167 Syslog server 231 system log display 480

Т

```
TAC 486
tag
gateway 188
tag list
export 515
team
create 584
name 584
```

NCC User's Guide

update 584 Teams screen 584 template configuration 520 setting 522 template setting duplicate/import to a site 83 template site/setting create/bind 79 Threat management screen 325 ticket support request 593 ticket details 594 time zone set 48 site location 231 topology overview 523 Topology screen 155 traceroute perform 167, 480 traceroute function 182, 185 Tracking Area Code (TAC) 486 traffic log AP 232 traffic shaping 267, 463 Traffic shaping screen 267, 464 transfer license action 501 tutorial 64 transfer license to another organization tutorial 65 transferable license select 64 transmitting power 267 trial license activate 55 expiration date 43 trial license summary table 505 Trial screen 505 troubleshooting 597 more help 601 trunk group 289 trunk port 293 TV pixelation 281 Two factor authentication screen 110

two-factor authentication bypass 221, 543 enable 27, 104 type user account 224

U

U/TM Security Pack license 17 U-APSD 255, 338 U-APSD (Unscheduled automatic power save delivery) 337 UARFCN (UTRA Absolute Radio-Frequency Channel Number) 484, 485 UE (User Equipment) 486 Undo assign action 501, 502, 503, 504 Unused license state 504 Upgrading firmware status 218, 540 uplink AP 158 uplink port 133, 293, 313 UPnP conflict with ONVIF 300 user remove 100 user account bind 110 create 544 remove 542, 546 type 541 User screen 219, 541, 544 **USG FLEX device** license 17 trial license 505 **USG VPN device** license 17 trial license 505 UTRA Absolute Radio-Frequency Channel Number 484, 485

V

virtual machine (VM) 534 virtual private network 332, 365, 449 Virtual Private Network (VPN) create automatically 523 VLAN attribute 220, 542 VLAN for IPTV 124 VLAN ID 49 client 197 VLAN settings guest 50 Voice over IP (VoIP) 380, 444 voice VLAN 313 configure 121 voucher 236 create 238 status 238 voucher code login 251 Voucher Settings screen 239 voucher table export 237 voucher-based WiFi access 236 Vouchers screen 237 VPN 332, 365, 449 VPN access 220, 542 VPN area 333, 368, 524 VPN client setup 105 VPN client setting 371 VPN client software download 457 VPN configuration file 105 import steps 105 VPN Connections screen 318, 343, 415 VPN connections screen 112 VPN gateway site 362 VPN Orchestrator screen 524, 559 **VPN** rule enable 103 **VPN** setup by user 105 VPN tunnel 416 open steps 108

VPN Tunnel Interface (VTI) 455 VPN user create 102

W

walled garden 247, 253, 403, 405, 408, 466 WAN interface configuration screen 320, 352 WAN load balancing configure 363 WAN settings 143 configure 143 WAN status screen 480 WAN Throughput Dashboard 154 WAN usage view 476 Warning firmware status 218, 536, 540 warranty 607 note 607 web authentication 402, 462 Web Filtering signature 225 widget rearrange 152 WiFi guest 50 WiFi aid 199 WiFi Aid screen 198 WiFi frequency band 182, 185 WiFi mesh solution 233 WiFi network name enter 49 WiFi password enter 49 WiFi security configure 337 WiFi settings 49 WiFi status mobile router 488 wildcard domain name 253 WINS (Windows Internet Naming Service) server 429, 435 wired clients 192

wireless bridge use 235 wireless channel bandwidth 264, 339 Wireless Clients 192 Dashboard 153, 154 Wireless Clients (by Usage) Dashboard 154 Wireless Clients Manufacturer Dashboard 154 Wireless Clients OS Dashboard 154 Wireless Health screen 240 Wireless screen 404 wizard installation 497 WLAN stations screen 488 WLAN usage view 476 world map 190 WPA Enterprise 251 WPA2-PSK data encryption 404

Ζ

Zero Touch Provision 142 Zero Touch Provision mode 146 Zero Touch Provisioning 51 ZTP (Zero Touch Provision) 142 ZTP (Zero Touch Provisioning) 51 ZTP deployment method 146 ZyWALL VPN device configure 26 Zyxel Device 155 Zyxel license marketplace 494, 501 Zyxel webstore 559